

UNITED STATES ENVIRONMENTAL PROTECTION AGENCY REGION IX 75 Hawthorne Street San Francisco, CA 94105

JAN 30 2002

Dennis A. Rea, Colonel, USAF Vice Commander, 56 FW 13970 W. Lightning Street Luke AFB, Arizona 85309-1149

Re: First Five Year Review, Luke Air Force Base, Arizona, January 18, 2002

Dear Colonel Rea:

The U.S. Environmental Protection Agency (EPA) Region 9 has received the First Five Year Review, Luke Air Force Base (AFB), Arizona, dated January 18, 2002. We have reviewed the aforementioned document and the Air Force's responses to our comments on the draft final version dated November 29, 2001. Based on this review, EPA agrees with the findings, conclusions, and recommendations provided in the Report, and concurs with the Air Force that the remedies at Luke AFB remain protective of human health and the environment and exposure pathways that could result in unacceptable risks are being controlled through implementation of institutional controls and monitoring.

We understand the Declaration of Environmental Use Restriction (DEUR) for Potential Source of Contamination (PSC) DP-23 has been filed with Arizona Department of Environmental Quality (ADEQ) (filed in November 2001). The DEUR for PSC ST-18 will also be filed to ensure long-term protectiveness. The groundwater monitoring wells at PSC SS-42 will be sampled annually until the ROD requirement of 5 annual sampling events is completed, after which these wells can be sampled at five year intervals. Continued groundwater monitoring will be conducted for PSCs RW-02, FT-07, ST-18, SD-20 and SS-42 as part of future five-year reviews. If there are any questions, please contact Xuan-Mai Tran, Remedial Project Manager, at (415) 972-3002.

Sincerely,

En

Jane Diamond Acting Director Superfund Division

cc: Lou Minkler, ADEQ Belle Matthews, Luke AFB Jeff Rothrock, Luke AFB Daniel Salzler (CAB Co-Chair) Ed Cayous, EPA-HQ Ronald McRobbie, Air Force Regional Environmental Office

Final First Five-Year Review

Nicole Cherry

Scientist I

(Kent Lang **Principal Scientist**

R Charles Leichner, P.E

Senior Vice President

Prepared for: Luke Air Force Base, Arizona

Prepared by: ARCADIS G&M, Inc. 8222 S. 48th St. Suite 140 Phoenix Arizona 85044 Tel 602 438 0883 Fax 602 438 0102

Our Ref.: AZO00891.0001

Date: January 18, 2002

List of Acronyms and Abbreviations

ADEQ	Arizona Department of Environmental Quality
ADWR	Arizona Department of Water Resources
AMSL	Above Mean Sea Level
AWQS	Aquifer Water Quality Standard
BEP	Bis (2-Ethyl Hexyl) Phthalate
BGS	Below Ground Surface
BGP	Base General Plan
BNA	Base/Neutral and Acid Extractable Organic Compound
BTEX	Bezene, Toluene, Ethylbinzene, Total Xylene
BX	Base Exchange
CERCLA	Comprehensive Environmental Response Compensation and Liability Act
CFR	Federal Code of Regulations
COC	Constituents of Concern
COEC	Constituents of Ecological Concern
COPC	Constituents of Potential Concern
DBCP	1,2-dibromo-3-chlorpropane
DCA	1,2-dichloroethane
DCP	Dichloropropane
DOD	Department of Defense
DPDO	Defense Property Disposal Office
DRO	Diesel – Range Organics
DEUR	Declaration of Environmental Use Restriction
ECC	Environmental Chemical Corporation
EI	Ecological Inventory
ELCR	Excess Lifetime Cancer Risk
EPA	Environmental Protection Agency
FCOR	Final Close Out Report
FCOR	Final Closeout Report
FFA	Federal Facilities Agreement
FS	Feasibility Study
GWNT	Groundwater Not Threatened
GPL	Groundwater Protection Level
HDPE	High Density Polyethylene
HI	Hazard Index
HQ	Hazard Index
HRS	Hazard Ranking System
ICE	Internal Combustion Engine
ICP	Institutional Control Plan
IEUBK	Integrated Exposure Uptake Biokinetic
IRP	Installation Restoration Program
LCU	Lower Conglomerate Unit
LTM	Long Term Monitoring

LUST	Leaking Underground Storage Tank
MCL	Maximum Contaminate Level
MFU	Middle Fine Unit
NCP	National Contingency Plan
NPL	National Properties List
OSWER	Office of Solid Waste and Emergency Response
OU	Operable Unit
PAH	Polycyclic Aromatic Hydrocarbon
PCB	Polycholrinated Biphenyls Chromium
PCE	Tetracholoethane
PPE	Personal Protective Equipment
PRG	Preliminary Remediation Goal
PSC	Potential Source of Contamination
QA/QC	Quality Control and Assurance
OU	Operable Unit
RCRA	Resource Conservation and Recovery Act
RCVA	Resource Conservation Recovery Act
RFA	Facility Assessment
RFI	Facility Investigation
RI	Remedial Investigation
RI/FS	Remedial Investigation/Feasibility Study
ROD	Record of Decision
SARA	Super fund Amendments and Reauthorization Act
SRL	Soil Remediation Level
SVE	Soil Vapor Extraction
TCE	Trichloroethene
TPH	Total Petroleum Hydrocarbons
TPH	Total Petroleum Hydrocarbons
TRPH	Total Recoverable Petroleum Hydrocarbons
TRPH	Total Recoverable Petroleum Hydrocarbons
UAU	Upper Alluvial Unit
USACE	United States Army Corps of Engineers
USAF	United States Air Force
USEPA	United States Environmental Protection Agency
UST	Underground Storage Tank
UTL	Upper Threshold Limit
VEMUR	Voluntary Environmental Mitigation Use Restriction
VES	Vapor Extraction System
VOC	Volatile Organic Compound
WSRV	West Salt River Valley
WWTP	Waste Water Treatment Plant

1.0	Introduction	1
1.1	Background	1
1.1.1	Physical Characteristics	1
1.1.2	Land and Resource Use	2
1.1.3	Project History	2
1.2	Purpose	4
2.0	Site Chronology	5
2.1	Operable Unit 2	5
2.2	Operable Unit 1	6
3.0	RI/FS Results and ROD Findings	8
3.1	OU-2 RI/FS Results	8
3.2	OU-1 RI/FS Results	12
3.3	OU-2 ROD Summary	19
3.4	OU-1 ROD Summary	19
4.0	Summary of Base-wide Risk Assessment	20
5.0	Groundwater Summary	24
5.1	Groundwater Hydrology	24
5.1.1	Aquifer Units	24
5.1.2	Regional Recharge	25
5.1.3	Regional Discharge	25
5.1.4	Historical Trends in Regional Groundwater Levels	26
5.1.5	Groundwater Occurrence, Apparent Gradient, and Estimated Flow Directions in the Luke AFB Area	26
5.2	PSC Groundwater Summary	30

i

5.2.1	PSC RW-02	38
5.2.2	PSC DP-05	38
5.2.3	PSC FT-06	39
5.2.4	PSC FT-07	40
5.2.5	PSC ST-18	41
5.2.6	PSC SD-20	42
5.2.7	PSC SD-21	43
5.2.8	PSC SD-38	44
5.2.9	PSC SS-42	44
6.0	Investigative Site History	45
6.1	Five Year Review Process Summary	45
6.2	Review of OU-1 PSCs	46
6.2.1	PSC RW-02 Wastewater Treatment Annex	47
6.2.2	PSC LF-03 Outboard Runway Landfill	53
6.2.3	PSC FT-07E Eastern Portion of North Fire Training Area	58
6.2.4	6.2.4 DP-13 Drainage Ditch Disposal Area	
6.2.5	PSC LF-14 Old Salvage Yard Burial Site	68
6.2.6	PSC LF-25 Northwest Landfill	73
6.2.7	PSC SD-38 Southwest Oil/Water Separator at the Auto Hobby Shop	77
6.2.8	PSC SS-42 Bulk Fuels Storage	82
6.3	Review Process for OU-2 PSCs	88
6.3.1	PSC ST-18 Facility 993	89
6.3.2	PSC DP-23 Old Surface Impoundment West of Facility 993	94
6.4	Groundwater Review	103
6.5	Interviews	113

ii

Tables

- 2-1 Chronology of Events for the construction of the concrete cap at PSC ST-18
- 2-2 Chronology of events for the ex-situ bioremediation (soil composting) at PSC DP-23
- 2-3 Chronology of events for the OU-1 remedial action.
- 2-4 Chronology of Events for the SVE at PSC SS-42
- 3-1 Summary of RI/FS Results for OU-2
- 3-2 Summary of OU-1 RI/FS Results
- 3-3 Description of the Remedy for OU-2
- 3-4 Description of the Remedy for OU-1
- 5-1 Status of Groundwater Monitoring Wells
- 5-2 Groundwater Sampling Summary
- 6-1 Reasonable Maximum Soil Exposure Point Concentrations in Soil and USEPA Region IX PRGs for Surface Soil (Base Worker) PSC RW-02
- 6-2 Reasonable Maximum Soil Exposure Point Concentrations in Soil and USEPA Region IX PRGs for Surface and Subsurface (Excavation Worker) PSC RW-02
- 6-3 Reasonable Maximum Soil Exposure Point Concentrations in Soil and USEPA Region IX PRGs for Surface Soil (Base Worker) PSC LF-03
- 6-4 Reasonable Maximum Soil Exposure Point Concentrations in Soil and USEPA Region IX PRGs for Surface and Subsurface Soil (Excavation Worker) PSC LF-03
- 6-5 Reasonable Maximum Soil Exposure Point Concentrations in Soil and USEPA Region IX PRGs for Surface and Subsurface Soil (Excavation Worker) PSC FT-07
- 6-6 Reasonable Maximum Soil Exposure Point Concentrations in Soil and USEPA Region IX PRGs for Surface Soil (Base Worker) PSC FT-07
- 6-7 Reasonable Maximum Soil Exposure Point Concentrations in Soil and USEPA Region IX PRGs for Surface Soil (Base Worker) PSC SP-13
- 6-8 Reasonable Maximum Soil Exposure Point Concentrations in Soil and USEPA Region IX PRGs for Surface and Subsurface Soil (Excavation Worker) PSC DP-13
- 6-9 Reasonable Maximum Soil Exposure Point Concentrations in Soil and USEPA Region IX PRGs for Surface Soil (Base Worker) PSC LF-14

g:\env\proj\800\891\finalfiveyearreview.rtf

- 6-10 Reasonable Maximum Soil Exposure Point Concentrations in Soil and USEPA Region IX PRGs for Surface and Subsurface Soil (Excavation Worker) PSC LF-14
- 6-11 Reasonable Maximum Soil Exposure Point Concentrations in Soil and USEPA Region IX PRGs for Surface and Subsurface Soil (Base and Excavation Worker) PSC LF-25
- 6-12 Reasonable Maximum Soil Exposure Point Concentrations in Soil and USEPA Region IX PRGs for Surface and Subsurface Soil (Excavation Worker) PSC SD-38
- 6-13 Reasonable Maximum Soil Exposure Point Concentrations in Soil and USEPA Region IX PRGs for Surface and Subsurface Soil (Excavation Worker) PSC SS-42
- 6-14 Reasonable Maximum Soil Exposure Point Concentrations in Soil and USEPA Region IX PRGs for Surface and Subsurface Soil (Excavation Worker) PSC ST-18
- 6-15 Reasonable Maximum Soil Exposure Point Concentrations in Soil and USEPA Region IX PRGs for Surface Soil (Base Worker) PSC DP-23 (Northern Portion)
- 6-16 Reasonable Maximum Soil Exposure Point Concentrations in Soil and USEPA Region IX PRGs for Surface and Subsurface Soil (Excavation Worker) PSC DP-23 (Northern Portion)
- 6-17 Reasonable Maximum Soil Exposure Point Concentrations in Soil and USEPA Region IX PRGs for Surface and Subsurface Soil (Base and Excavation Worker) PSC SP-23 (Southern Portion)
- 6-18 Maximum Exposure Point Concentrations in Groundwater and USEPA Region IX PRGS for PSC RW-02 (MW-115; MW-116; MW-124)
- 6-19 Maximum Exposure Point Concentrations in Groundwater and USEPA Region IX PRGS for PSC DP-05 (MW-104; MW-105; MW-106)
- 6-20 Maximum Exposure Point Concentrations in Groundwater and USEPA Region IX PRGS for PSC FT-06 (MW-107; MW-108
- 6-21 Maximum Exposure Point Concentrations in Groundwater and USEPA Region IX PRGS for PSC FT-07 (MW-109; MW-110; MW-111; MW-118; MW-12)
- 6-22 Maximum Exposure Point Concentrations in Groundwater and USEPA Region IX PRGS for PSC ST-18 (MW-2; MW-3; MW-4; MW-5; MW-114; MW-122)
- 6-23 Maximum Exposure Point Concentrations in Groundwater and USEPA Region IX PRGS for PSC SD-20 (MW-102; MW-103; MW-112D; MW-112S; MW-113)
- 6-24 Maximum Exposure Point Concentrations in Groundwater and USEPA Region IX PRGS for PSC SD-21 (MW-101)
- 6-25 Maximum Exposure Point Concentrations in Groundwater and USEPA Region IX PRGS for PSC SD-38 (MW-117)
- 6-26 Maximum Exposure Point Concentrations in Groundwater and USEPA Region IX PRGS for SS-42 (MW-119; MW-120; MW-121; MW-125)

g:\env\proj\800\891\finalfiveyearreview.rtf

Figures

1-1	Location of Luke Air Force Base, Arizona
1-2	Location of OU-2 PSCs
1-3	Location of OU-1 PSCs
5-1	Location of OU-1 and OU-2 PSCs and Monitor Wells
5-2	Groundwater Analytical Results

Appendices

Appendix A	PSC - Detailed Descriptions and Background Information
Appendix B	Groundwater Monitoring Results for the Period of Record
Appendix C	Changed Standards Comparison for No-Action PSCs
Appendix D	USEPA Region IX PRGs for 1991, 1996 and 2000
Appendix E	Photo Log
Appendix F	Interview Notes

Executive Summary

In 1990, the United States Environmental Protection Agency (USEPA) placed Luke Air Force Base (Luke AFB) on the National Priorities List (NPL) pursuant to the Comprehensive Environmental Response, Compensation, and Liability Act (CERCLA also known as Superfund) of 1980, as amended by the Superfund Amendments and Reauthorization Act (SARA) of 1986. Luke AFB (the site) was added to the NPL as a result of past hazardous material handling and disposal practices.

This five-year review was conducted pursuant to the Comprehensive Environmental Response Compensation and Liability Act (CERCLA) Section 121(c), 42 U.S.C. § 9621(c), the National Contingency Plan (NCP) (40 CFR § 300.430 (f)(4)(ii)), Office of Solid Waste and Emergency Response (OSWER) Directive 9355.7-03B-P Final. This report summarizes the remedial actions and data collected since the beginning of the project in August 1990 through November 2001 and provides an evaluation of the effectiveness of the remedial actions relative to remedial objectives and to verify that remedial actions remain protective of human health and the environment.

Luke AFB, which is an advanced fighter pilot training institution, covers approximately 4,000 acres west of the Phoenix metropolitan area in Glendale, Arizona. Aircraft maintenance and light industrial operations in support of training missions have been in existence at Luke AFB since its inception in 1941. The results of these activities generated potentially hazardous wastes such as petroleum residues, cleaning solvents, and other related wastes.

Subsequent to the listing of Luke AFB, remedial investigation/feasibility studies were performed to determine the nature and extent of contamination. A total of 33 potential sources of contamination (PSCs) were initially identified for investigation purposes. To aid in the management of the investigations, the PSCs were divided into two operable units, OU-1 and OU-2. OU-2, the first to be investigated, included the investigation of soils at eight sites at which only petroleum-related wastes were disposed. OU-2 PSCs include the following:

- PSC OT-04 Old Perimeter Road POL Waste Site
- PSC DP-05 POL Waste Disposal Trench
- PSC FT-06 South Fire Training Area
- PSC FT-07W Western Portion of the North Fire Training Area
- PSC ST-18 Facility 993
- PSC DP-22 POL Trench at Northeast Runway
- PSC DP-23 Old Surface Impoundment West of Facility 999
- PSC SD-40 Taxiway Discharge Area

OU-1 included the investigation of the soils at 25 PSCs and the Base-wide investigation of air, surface water, and groundwater resources. OU-1 PSCs include the following:

- Old Incinerator Site (PSC OT-01).
- Wastewater Treatment Annex Landfill (PSC RW-02).
- Outboard Runway Landfill (PSC LF-03).
- Eastern Portion of North Fire Training Area (PSC FT-07E).
- F-15 Burial Site (PSC OT-08).
- Canberra Burial Site (PSC OT-09).
- Concrete Rubble Burial Site (PSC OT-10).
- Former Outside Transformer Storage (PSC SS-11).
- Old Explosive Ordnance Division (EOD) Burial Site (PSC OT-12).
- Drainage Ditch Disposal Area (PSC DP-13).
- Old Salvage Yard Burial Site (PSC LF-14).
- Facility 328 Spill Site (PSC SS-15).
- Facility 321 Underground Storage Tank (UST) (PSC SS-16).
- Former Defense Property Disposal Office (DPDO) Yard (PSC SS-17).
- Base Exchange (BX) Leaking USTs (PSC ST-19).
- Oil/Water Separator Canal and Earth Fissures (PSC SD-20).
- Sewage Treatment Plant Effluent Canal (PSC SD-21).
- Base Ammunition Storage Area (PSC DP-24).
- Northwest Landfill (PSC LF-25).
- Hush House Canal (PSC SD-26).
- Northeast Landfill (PSC LF-37).
- Southwest Oil/Water Separator at the Auto Hobby Shop (SD-38).
- Waste Discharge at the Old Lockheed Site (SD-39).
- Skeet Range (OT-41).
- Bulk Fuels Storage (SS-42).

A more detailed description and background information for OU-1 and OU-2 PSCs is in Appendix A. It is important to note that PSCs 27 through 36 do not exist because there was a break in the numbering between PSC SD-26 and PSC LF-37.

In addition to the investigation of identified PSCs, a RCRA facility assessment (RFA) and RCRA facility investigation (RFI) were conducted to determine if any of the current operational facilities at Luke AFB should be included as PSCs in the CERCLA program. Remedial alternatives were identified, and remedial actions were designed and implemented as part of clean up activities.

As part of the OU-1 Feasibility Study (FS), a risk-based assessment was performed regarding acceptability of PSCs for residential land use given current conditions. Residential land use implies that a site can be developed and used for any purpose, including residential development. If a PSC was deemed unsuitable for residential land use, remedial alternatives were developed for that site.

Remedial alternatives were also developed for any site that could potentially impact underlying groundwater resources in the future.

Potentially exposed populations considered in the risk assessment included the following:

- Base workers
- Excavation workers
- Military personnel
- Child visitors for sites which extend off the base property
- Base residents

The risk assessment considered both average and reasonable maximum exposure conditions to characterize current and future risks.

Risk from residential exposure to combined surface and subsurface soil were calculated using both the USEPA Region IX PRGs and the ADEQ SRLs. Based on the results of the evaluation, all of the PSC areas evaluated were determined to be suitable for unrestricted, or residential land use with the exception of the following PSCs:

- RW-02
- LF-03
- FT-07 E
- DP-13
- LF-14
- ST-18
- LF-25
- DP-23 N
- SD-38

In addition to evaluating potential human exposure at Luke AFB, an ecological risk assessment was also performed.

This five-year review report provides a historical and five-year review process summary for OU-1 and OU-2 PSCs for which remedies were selected. These PSCs include the following:

- RW-02
- LF-03
- FT-07 E
- DP-13
- LF-14

- ST-18
- DP-23
- LF-25
- SD-38
- SS-42

The historical review and evaluation process also includes PSCs for which it was determined no action was required. These PSCs include the following:

- OT-01
- OT-04
- DP-05
- FT-06
- FT-07 W
- OT-08
- OT-09
- OT-10
- SS-15
- SS-16
- SS-19
- DP-22
- SD-21
- DP-24
- SD-26
- LF-37SD-39
- SD-39 • SD-40
- OT-41

The five-year review process primarily consisted of a site inspection, interviews and a review of relevant documents and data. Jeff Rothrock of Luke AFB led the five-year review for the site. The following team members assisted in the review:

- Jeff Rothrock, Luke AFB
- Jon Sherrill, ARCADIS G&M, Inc.
- Kent Lang, ARCADIS G&M, Inc.
- Stephanie Armijo, ARCADIS G&M, Inc.
- Monique Ostemann, USACE
- Greg Mellema, USACE

- Dan Stralka, USEPA
- Nancy Lou Minkler, ADEQ

The five-year review process includes the following primary elements:

- Remedy selection and implementation is reviewed and summarized for each OU-1 and OU-2 PSC for which a remedy was selected.
- Changes in standards were evaluated with respect to the continued effectiveness of the remedies that were implemented based on cancer risks and non-cancer hazards for applicable constituents of concern (COCs) for base worker or excavation worker scenarios.
- Groundwater monitoring results are compared to groundwater standards established for the project.
- Representatives of Luke AFB, USEPA, US Army Corps. of Engineers, ADEQ and ARCADIS G&M performed a site inspection of each PSC for which a remedy was selected on May 22, 2001.
- The results of interviews with individuals knowledgeable about the project.

OU-1 PSCs for which remedies were selected based on the results of risk assessment include the following:

- RW-02
- LF-03
- FT-07 E
- DP-13
- LF-14
- LF-25
- SD-38
- SS-42

For OU-1 PSCs, PRGs were not established. Alternatively, PSC specific cancer risks and non-cancer hazards were calculated using 1996 USEPA Region IX PRG guidance to develop a site-specific industrial scenario. To evaluate changes in standards as part of this five-year review, cancer risk and non-cancer hazards were recalculated using 2000 USEPA Region IX industrial PRGs and post remediation exposure point concentrations for base worker and excavation worker scenarios as applicable. ADEQ SRLs were also reviewed since they were used to determine risk under a residential land use scenario.

Changes in standards are evaluated with respect to the continued effectiveness of the remedies that were implemented based on a non-cancer HI less than or equal to 1.0 or an ELCR greater than the risk range of 1×10^{-6} to 1×10^{-4} .

OU-2 PSCs for which remedies were selected based on the results of risk assessment include the following:

- ST-18
- DP-23

For OU-2 PSCs, 1991 USEPA Region IX PRGs were originally used to establish performance standards. To evaluate changes in standards as part of this five-year review, cancer risks and non-cancer hazards were recalculated for each COC using 2000 USEPA Region IX industrial PRGs and post remediation exposure point concentrations for base worker and excavation worker scenarios as applicable. The analysis of standard changes also included a review of 1996 USEPA industrial PRGs. ADEQ SRLs were also reviewed since they were used to evaluate residential use standards. Changes in standards are evaluated with respect to the continued effectiveness of the remedies that were implemented based on a non-cancer HI less than or equal to 1.0 or an ELCR greater than the risk range of 1×10^{-6} to 1×10^{-4} .

The comparison indicates that PSCs for which changes in standards were evaluated are still within the acceptable risk range. It was concluded that selected remedies are protective of human health and the environment and exposure pathways that could result in unacceptable risks are being controlled with the following exceptions:

- The northern portion of PSC DP-23 is within the acceptable risk range for industrial or non-residential land use, but outside the risk range for residential land use.
- The remedy at PSC ST-18 Facility 993 currently protects human health and the environment because the cap prevents exposure in the short term. However, in order for the remedy to be protective in the long term, a (Declaration of Environmental Use Restriction) DEUR is needed at the site to ensure long-term protectiveness.

A DEUR for PSC ST-18 and the northern portion of DP-23 has been filed with the ADEQ to resolve these issues.

A comparison of exposure point concentrations in groundwater (maximum concentrations for the period of record) and USEPA Region IX 2000 PRGs for tap water and ADEQ aquifer water quality standards were evaluated for the following PSCs:

- RW-02
- DP-05
- FT-06
- FT-07
- ST-18
- SD-20
- SD-21
- SD-38
- SS-42

The comparison indicates that exposure to groundwater results in risk that are within the acceptable risk range for these PSCs. A review of groundwater data for the period of record indicates that groundwater at Luke AFB is not impacted as there are no constituents in groundwater that currently exceed applicable water quality standards. All potential sources of constituents have been controlled or eliminated through the institution of pollution prevention measures or remedial activities

The following individuals were solicited for interviews by questionnaire as part of this five-year review:

- Belle Matthews, Luke AFB Project Manager
- Sean Hogan, EPA Project Manager
- Nancy Lou Minkler, ADEQ Project Manager
- Dan Salzler, Citizens Advisory Board (CAB) Community Co-Chairperson
- Joyce Clark, CAB member
- Martin Jeffries, CAB member

In addition to solicitation of interviews by questionnaire, the following individuals were interviewed in person as part of the five-year review site inspection:

- Chris Christoffer, Luke AFB Environmental Analyst
- Sergeant Anthony Michels, Luke AFB Infrastructure Superintendent

Chris Christoffer and Sergeant Michels were interviewed relative to procedures that ensure compliance with the Base General Plan (BGP) and Institutional Control Plan (ICP). As part of these interviews, the BGP was reviewed and it was verified that the ICP had been implemented. Also verified were approval and record keeping procedures for digging permits relative to environmental constraints at Luke AFB.

Luke AFB Five-Year Review Signature Cover Preliminary Information

Site name: Luke Air Force Base		EPA ID: AZ0570024133		
Region: 09	State: Arizona City/County: Luke AFB/Maricopa		unty: Luke AFB/Maricopa	
LTRA* (highlight)		Constru	Construction completion date: December 17, 1999	
Fund/PRP Lead: Luke AFB			NPL status: Final	
Lead agency: USEPA Region IX				
Who conducted the review (EPA Region, state, Federal agencies or contractor): USEPA Region IX, ADEQ, USACE, ARCADIS G&M, Inc.				
Dates review conducted: From: April 2001 through December 2001		Date(s) of site visit: May 22, 2001		
Whether first or successive review: First				
Circle: Regional Discretion Due dat		te: January 21, 2002		
Trigger for this review: Final close-out process (2000-2001) and time that has lapsed since finalization of the OU-2 Record of Decision (ROD) in January 1994.				
Recycling, reuse, redevelopment site (highlight): N/A				

Issues: Northern portion of PSC DP-23 not remediated to residential soil standards and requires deed restriction. PSC ST-18 requires deed restriction to prevent future removal of cap and excavation of soil. Continued of monitoring at specific PSC to confirm protectiveness of remedies.

Recommendations: Declaration of Environmental Use Restrictions (DEURs) for PSC ST-18 and DP-23 have been filed with ADEQ (filed in 2001). Continued monitoring of groundwater will be conducted for PSCs RW-02, FT-07, ST-18, SD-20 and SS-42 as part of future five-year reviews.

Protectiveness Statement(s): The remedies at Luke AFB are protective of human health and the environment and exposure pathways that could result in unacceptable risks are being controlled through implementation of remediation, institutional controls and monitoring.

Other Comments: None

Signature of Luke AFB Environmental Protection Committee Chairman

DENNIS A. REA, Colonel, USAF

Vice Commander, 56 FW

Jan 02

1.0 Introduction

In 1990, the United States Environmental Protection Agency (USEPA) placed Luke Air Force Base (Luke AFB) on the National Priorities List (NPL) pursuant to the Comprehensive Environmental Response, Compensation, and Liability Act (CERCLA also known as Superfund) of 1980, as amended by the Superfund Amendments and Reauthorization Act (SARA) of 1986. Luke AFB (the site) was added to the NPL as a result of past hazardous material handling and disposal practices. The location of the site is shown in Figure 1-1. On behalf of the United States Air Force, ARCADIS G&M, Inc. (ARCADIS G&M) has prepared this final first five-year review of remedial actions at Luke AFB, Arizona.

This five-year review was conducted pursuant to the Comprehensive Environmental Response Compensation and Liability Act (CERCLA) Section 121(c), 42 U.S.C. § 9621(c), the National Contingency Plan (NCP) (40 CFR § 300.430 (f)(4)(ii)), Office of Solid Waste and Emergency Response (OSWER) Directive 9355.7-03B-P Final.

1.1 Background

The following sections provide a general overview of site conditions, and project history. This information is intended to give the reader of the final first five-year review report for Luke AFB adequate background information with which to evaluate current conditions at the site.

1.1.1 Physical Characteristics

Luke AFB, which is an advanced fighter pilot training institution, covers approximately 4,000 acres west of the Phoenix metropolitan area in Glendale, Arizona. Aircraft maintenance and light industrial operations in support of training missions have been in existence at Luke AFB since its inception in 1941. Luke AFB lies in the Salt River Valley (SRV), which lies within the Basin and Range physiographic province. Elevations at Luke AFB range from 1,250-feet above mean sea level (amsl) at the northwest corner to 995-feet amsl at the southeast corner. The climate at Luke AFB is characterized as a desert climate. Rainfall at Luke AFB averages about 7.7 inches per year.

1.1.2 Land and Resource Use

The eastern portion of Luke AFB currently consists of a variety of light industrial facilities, office buildings occupied by administrative and community services, base barracks, and outdoor recreation centers. The central and western portions of Luke AFB include the runways, open spaces, and aircraft operations, training and maintenance facilities. Base residential housing and commercial areas are located east of the fenced areas of the main portions Luke AFB. Aircraft maintenance and light industrial operations in support of training missions have been in existence at Luke AFB since its inception in 1941. The results of these activities generated potentially hazardous wastes such as petroleum residues, cleaning solvents, and other related wastes.

1.1.3 Project History

Subsequent to the listing of Luke AFB on the NPL, remedial investigation/feasibility studies were performed to determine the nature and extent of contamination. Remedial alternatives were identified, and remedial actions were designed and implemented as part of clean up activities. A record of the remedial actions implemented and how cleanup was accomplished at Luke AFB are summarized in the Remedial Action Report. The following is a background summary relative to the Superfund project at Luke AFB:

- Prior to 1976 and the Resource Recovery and Conservation Act (RCRA), potentially hazardous wastes, such as petroleum residues, cleaning solvents, and other related materials, were disposed on Base through fire department training exercises, road oiling for dust suppression, and in shallow trenches.
- In 1981, the Department of Defense (DOD) initiated the IRP to investigate and remediate past hazardous materials handling and disposal practices at all military institutions.
- Before the passage of SARA, the USEPA did not supervise the IRP program at Luke AFB. Subsequent to the passage of SARA, the USEPA was required to establish and maintain a docket of potentially contaminated federal facilities, perform Hazard Ranking System (HRS) scoring on these facilities, and list those facilities exceeding the HRS threshold score on the NPL.

- The USEPA audited Luke AFB in 1987, and scored the institution using the HRS.
- Because the Luke AFB HRS score of 37.93 exceeded the threshold value of 28.5, the USEPA added Luke AFB to the NPL in August 1990.
- On September 27, 1990, the USEPA, Arizona Department of Environmental Quality (ADEQ), Arizona Department of Water Resources (ADWR), and the United States Air Force (USAF) signed a FFA to establish the procedural framework for conducting the required environmental investigations at Luke AFB.
- Environmental investigations at Luke AFB were implemented in accordance with regulations established in the NCP at Title 4, part 300 of the Federal Code of Regulations (CFR).

Based on the results of the Installation Restoration Program (IRP) and other information compiled during the initial planning stages, the Federal Facilities Agreement (FFA) identified 33 potential sources of contamination (PSCs). To aid in the management of the investigations, the FFA parties divided the PSCs into two operable units (OU). OU-1 included the investigation of the soils at 25 PSCs and the Base-wide investigation of air, surface water, and groundwater resources. OU-2 included the investigation of soils at eight sites at which only petroleum-related wastes were disposed. The FFA created this special grouping to put the eight OU-2 sites on a "fast-track;" the idea being that sites with common wastes would allow for a timely investigation and cleanup. The eight OU-2 PSCs are listed below. The location of PSCs in OU-2 is in Figure 1-2.

- PSC OT-04 Old Perimeter Road POL Waste Site
- PSC DP-05 POL Waste Disposal Trench
- PSC FT-06 South Fire Training Area
- PSC FT-07W Western Portion of the North Fire Training Area
- PSC ST-18 Facility 993
- PSC DP-22 POL Trench at Northeast Runway
- PSC DP-23 Old Surface Impoundment West of Facility 999
- PSC SD-40 Taxiway Discharge Area

OU-1 was the last of two operable units to be addressed at Luke AFB and was defined to govern the investigation and potential remediation of air, surface water, and

groundwater resources Base-wide. In addition, the soils at 25 PSCs believed to have been impacted primarily by non-petroleum related wastes were included in OU-1. The 25 PSCs included in OU-1 are listed below. The location of PSCs in OU-1 is in Figure 1-3.

- Old Incinerator Site (PSC OT-01).
- Wastewater Treatment Annex Landfill (PSC RW-02).
- Outboard Runway Landfill (PSC LF-03).
- Eastern Portion of North Fire Training Area (PSC FT-07E).
- F-15 Burial Site (PSC OT-08).
- Canberra Burial Site (PSC OT-09).
- Concrete Rubble Burial Site (PSC OT-10).
- Former Outside Transformer Storage (PSC SS-11).
- Old Explosive Ordnance Division (EOD) Burial Site (PSC OT-12).
- Drainage Ditch Disposal Area (PSC DP-13).
- Old Salvage Yard Burial Site (PSC LF-14).
- Facility 328 Spill Site (PSC SS-15).
- Facility 321 Underground Storage Tank (UST) (PSC SS-16).
- Former Defense Property Disposal Office (DPDO) Yard (PSC SS-17).
- Base Exchange (BX) Leaking USTs (PSC ST-19).
- Oil/Water Separator Canal and Earth Fissures (PSC SD-20).
- Sewage Treatment Plant Effluent Canal (PSC SD-21).
- Base Ammunition Storage Area (PSC DP-24).
- Northwest Landfill (PSC LF-25).
- Hush House Canal (PSC SD-26).
- Northeast Landfill (PSC LF-37).
- Southwest Oil/Water Separator at the Auto Hobby Shop (SD-38).
- Waste Discharge at the Old Lockheed Site (SD-39).
- Skeet Range (OT-41).
- Bulk Fuels Storage (SS-42).

A more detailed description and background information for OU-1 and OU-2 PSCs is in Appendix A. It is important to note that PSCs 27 through 36 do not exist because there was a break in the numbering between PSC SD-26 and PSC LF-37.

1.2 Purpose

This report summarizes the remedial actions and data collected since the beginning of the project in August 1990 through November 2001 and provides an evaluation of the

effectiveness of the remedial actions relative to remedial objectives and to verify that remedial actions remain protective of human health and the environment. The need for this five-year review was identified during preparation of the Final Close Out Report (FCOR)¹ as part of the delisting process. This review is required because hazardous substances, pollutants, or contaminants remain in the subsurface at concentrations that are above levels that allow unrestricted land use. As the delisting process progressed, it was determined that the five-year review would be required because of the amount of time that has lapsed since finalization of the OU-2 Record of Decision (ROD) in January 1994.

This five-year review report is intended to be a concise summary of the work that was conducted at OU-1 and OU-2 to meet the statutory requirements of the Superfund process at Luke AFB. Numerous references are provided as part of this report however, not all support documents may be referenced. Rather, the most relevant documents are referenced in support of the objectives of the five-year review.

2.0 Site Chronology

2.1 Operable Unit 2

This section of the five-year review report provides a summary of the chronology of events for the implementation of the remedial alternatives for OU-2 at Luke AFB. The chronology of events for PSCs ST-18 and DP-23 are summarized in Tables 2-1 and 2-2, respectively.

Date	Event
October 19, 1983	RCRA closure of facility 993 begins.
April 19, 1988	Final inspection of concrete cap construction.
September 27, 1990	Signing of the FFA transferring jurisdiction of ST-18 to CERCLA.

Table 2-1. Chronology of Events for the Construction of the Concrete Cap at PSC ST-18

¹ ARCADIS Geraghty & Miller. 2001. Final Close-Out Report, Luke Air Force Base, Arizona. April 5, 2001.

Date	Event
January 28, 1994	Signing of the OU-2 ROD.
Annually	Cap inspection and maintenance at PSC ST-18.
Five year review	Groundwater monitoring.

Table 2-2 Chronology of Events for the Ex-situ Bioremediation (Soil Composting) at PSC DP-23

Date	Event
January 28, 1994	Signing of the OU-2 ROD.
April 11, 1995	Conduct preliminary soil sampling to further characterize the site.
May, 1994	Submittal and agency approval of the remedial design Report.
July 7, 1995	Excavation of contaminated soil and mixing in treatment cell.
October, 1995	Interim sampling to check status of bioremediation.
April 3, 1997	Addition of optimized soil amendment mix and continued soil composting.
June 5, 1997	Final sampling and begin construction demobilization.
August 1, 1997	Site restoration; re-grading and hydro seeding.
August 6, 1997	Conduct final site inspection.
August 27, 1997	Submit final closure report.

2.2 Operable Unit 1

This section of the five-year review report provides a summary of the chronology of events for the implementation of the remedial alternatives for OU-1 at Luke AFB. The chronology of events for the eight OU-1 sites are summarized in Table 2-3. The chronology for the SVE at PSC SS-42 is summarized in Table 2-4.

Date	Event
September 7, 1999	Final signatures on the OU-1 ROD.
December 16, 1999	Remedial design workplan for PSC LF-25 submitted.
December 17, 1999	Conducted metal shot recovery at PSC LF-25.
December 21, 1999	Radiological monitoring points installed at PSC RW-02.
December 29,1999	Perimeter fencing installed around containment structure at PSC RW-02.
January 5, 2000	Revisions to base general plan implemented and policy letter established to implement required institutional controls.
January 12, 2000	Radiological LTM plan for PSC RW-02 submitted.
June 15, 2000	VEMURs filed for PSCs RW-02, LF-03, FT-07E, DP-13, LF-14, LF-25, and SD-38 to restrict residential development of the sites.
November 13, 2000	Institutional Control Plan (ICP) developed and submitted.
Annually	Radiological monitoring at RW-02.

Table 2-3 Chronology of Events for the OU-1 Remedial Action

Table 2-4 Chronology of Events for the SVE at PSC SS-42

Date	Event
May 1995	Wells installed for bioventing treatability study.
August 6, 1996	Initiation of SVE using Internal Combustion Engine (ICE).
June 1997	Soil Boring CB-1 advanced to determine effectiveness of ICE.
November 2, 1998	Shut down of SVE system.
January 7, 1999	Second boring advanced to determine effectiveness of SVE.
September 7, 1999	OU-1 ROD signed requiring five-year groundwater monitoring.
May 12, 2000	Groundwater LTM plan for PSC SS-42 submitted.
May 16, 2000	First groundwater sampling event of five-year monitoring completed.

Date	Event
May 22, 2000	Soil vapor extraction and confirmation sampling summary report submitted.
Annually	Groundwater monitoring.

3.0 RI/FS Results and ROD Findings

The section of the report summarizes RI/FS results as recorded in the RODs for $OU-1^2$ and $OU-2^3$. The purpose of this section of the five year review report is to identify what COCs were evaluated as part of the RIs, which COCs exceeded standards established for the project, and what remedies were selected to address impacts for applicable PSCs.

3.1 OU-2 RI/FS Results

OU-2 included the investigation of soils at eight PSCs at which only petroleum-related wastes were believed to have been disposed. The location of the OU-2 PSCs are in Figure 1-2. The OU-2 RI/FS was conducted in accordance with USEPA guidance⁴ and approved work plans^{5,6,7,8,9}. The OU-2 field activities were limited to soil evaluations.

² Geraghty & Miller, 1999. Final Record of Decision, Operable Unit 1. Luke Air Force Base, Arizona. January 1999.

³ Geraghty & Miller, 1994. Final Record of Decision, Operable Unit 2. Luke Air Force Base, Arizona. January 1994.

⁴ USEPA, 1988. Guidance for Conducting Remedial Investigations and Feasibility Studies Under CERCLA, Interim Final: Office of Solid Waste and Emergency Response, OSWER Directive 9355.3-01.

⁵ Geraghty & Miller, 1991. Luke Air Force Base, Arizona, Final Base-Wide Remedial Investigations/Feasibility Study Work Plan, August 1991.

⁶ Geraghty & Miller, 1991. Luke Air Force Base, Arizona, Final Base-Wide Sampling and Analysis Plan, August 1991.

⁷ Geraghty & Miller, 1991. Luke Air Force Base, Arizona, Final OU-2 Remedial Investigations/Feasibility Field Sampling and Analysis Plan, November 1991.

⁸ Geraghty & Miller, 1992. Final Addenda for the Remedial Investigation/Feasibility Study Planning Documents, May 1992.

⁹ U.S. Army Corps of Engineers, 1991. Scope of Services, Operable Unit #2, Remedial Investigation/Feasibility Study (RI/FS), Luke Air Force Base, Arizona, August 9, 1991.

OU-2 RI results are detailed in the OU-2 RI report¹⁰. Part of the FS, USEPA guidance¹¹ was used to calculate Preliminary Remediation Goals (PRGs) for OU-2 soils. OU-2 FS results are detailed in the OU-2 FS report¹². OU-2 RI/FS results are summarized in Table 3-1.

¹⁰ Geraghty & Miller, 1992. Final Remedial Investigation Report, Operable Unit 2, Luke Air Force Base, Arizona, October 20, 1992.

¹¹ USEPA, 1991. Human Health Evaluation Manual Part B: Development of Risk-Based Preliminary Remediation Goals. Office of Emergency and Remedial Response, Washington, DC.

¹² Geraghty & Miller, 1993. Final Feasibility Study Report, Operable Unit 2, Luke Air Force Base, Arizona, May 12, 1993.

PSC	Description	COCs evaluated	COCs in excess of Industrial PRGs	Selected Remedial Alternative
OT-04	Old perimeter road POL waste site	Bis (2-ethylhexyl) phthalate, butylbenzylphthalate, TRPH, copper, lead	Less than PRGs	No action
DP-05	POL Waste Disposal Trench	Ethylbenzene, xylenes, bis (2-ethylhexyl) phthalate, 2-methylnaphthalene, naphthalene, TRPH, copper, lead	Less than PRGs	No action
FT-06	South fire training area	2-butanone (MEK), ethylbenzene, 2-hexanone (MBK), 1,1,2,2-tetrachlorethane, tetrachloroethene, toluene, trichloroethene, xylenes, acenaphthene, anthracene, benzo(a)anthracene, benzo(b)fluoranthene, benzo(k)fluoranthene, benzo(g,h,i)perylene, benzo(a)pyrene, bis(2-ethylhexyl) phthalate, butylbenzylphthalate, chrysenedibenzo(a,h)anthracene, dibenzofuran, di-n- butylphthalate, fluoranthene, fluorene, indeno(1,2,3-c,d)pyrene, 2-methylnaphthalene, 4-Methylphenol, Naphthalene, Pentachlorophenol, Phenanthrene, Phenol, Pyrene, TRPH, Metals, Copper, Lead	Trichloroethene, Benzo(a)anthracene, Benzo(b)fluoranthene, Benzo(k)fluoranthene, Benzo(a)pyrene, Indeno(1,2,3-cd)pyrene, Dibenzo(a,h)anthracene	Below the risk range
FT-07 W	Western portion of the north fire training area	benzo(b)fluoranthene, benzo(g,h,i)perylene, bis(2- ethylhexyl)phthalate, chrysene, fluoranthene, 2- methylnaphthalene, naphthalene pyrene, TRPH, copper, lead	Less than PRGs	No action

Table 3-1 Summary of RI/FS Results for OU-2

PSC	Description	COCs evaluated	COCs in excess of Industrial PRGs	Selected Remedial Alternative
ST-18	Facility 993	benzene, 1,1 dichloroethene, ethylbenzene, 1,1,2,2-tetrachlorethane, tetrachloroethene, toluene, trichloroethene, xylenes, benzo(a)anthracene, benzo(b)fluoranthene, benzo(g,h,i)perylene, benzo(a)pyrene, benzyl alcohol, bis(2-ethylhexyl)phthalate, chrysene, fluoranthene, indeno(1,2,3-cd)pyrene, 2-methylnaphthalene, naphthalene, phenanthrene, pyrene, TRPH, copper, lead	Benzene, 1,1,2,2- Tetrachlorethane, Benzo(a)pyrene	Maintain concrete cap, groundwater monitoring during each five-year review
DP-22	POL trench at northeast runway	acetone, TRPH, copper, lead	Less than PRGs	No action
DP-23	Old surface impoundment west of facility 999	ethylbenzene, toluene, anthracene, benzo(a)anthracene, benzo(b)fluoranthene, benzo(k)fluoranthene, benzo(g,h,i)perylene, benzo(a)pyrene, bis(2-ethylhexyl)phthalate, chrysene, dibenzo(a,h)anthracene, fluoranthene, indeno(1,2,3-cd)pyrene, phenanthrene, pyrene, TRPH, copper, lead	Benzo(a)anthracene Benzo(k)fluoranthene Benzo(a)pyrene Dibenz(a,h)anthracene Indeno(1,2,3-cd)pyrene	Ex-situ biological treatment
SD-40	Taxiway discharge area	acetone, benzene, ethylbenzene, toluene, xylenes, 2-methylnaphthalene, naphthalene, TRPH, copper, lead	Less than PRGs	No action

3.2 OU-1 RI/FS Results

OU-1 included the investigation of soils at 25 PSCs and the Base-wide investigation of air, surface water, and groundwater resources. In addition to the investigation of identified PSCs, a RCRA facility assessment (RFA) and RCRA facility investigation (RFI) were conducted to determine if any of the current operational facilities at Luke AFB should be included as PSCs in the CERCLA program. The results of the RCRA investigation are in Appendix A of the OU-1 report¹³. The location of PSCs in OU-1 are in Figure 3-2.

Prior to the beginning of the OU-1 RI field activities, the FFA parties determined that "no further remedial investigations" were needed at eight OU-1 PSCs, as follows:

- PSCs OT-01, OT-08, and OT-09 were classified as "no further action" sites because data obtained during an extensive review of Base records showed that hazardous materials or wastes were never handled or disposed at these sites.
- PSC DP-24 was removed from the Superfund process because it had mistakenly been included on the list of potentially contaminated sites.
- PSCs SS-15, SS-16, and ST-19 were removed from the Superfund process and placed under the jurisdiction of the ADEQ Underground Storage Tank (UST) section.
- PSC OT-10 was removed from the list of sites requiring field investigations because that site lies completely within the boundaries of PSC DP-13 and the landfill contents of both sites were presumed similar.

Because of its complexity, the OU-1 RI field investigation was divided into three phases, phase I conducted from October 1991 through March 1992, phase II activities conducted from June 1992 through April 1994 and phase III activities conducted in August and September 1996. Phase III activities were required to collect additional data for risk assessment purposes due to Phase I and Phase II laboratory data quality issues. However, the information reported as part of this five-year review is based on a consolidation of the most defensible data collected in conjunction with the overall Superfund process in terms of quality control and assurance (QA/QC) protocol.

¹³ Geraghty & Miller, 1997 Final Remedial Investigation Report OU-1, Luke Air Force Base, Arizona. Volumes 1 and 2. October 1997.

The OU-1 RI/FS was conducted in accordance with USEPA guidance² and approved work plans^{3,14,15,16,17}. OU-1 RI investigation results are detailed in the OU-1 RI report¹³.

As part of the OU-1 FS, a risk-based assessment was performed regarding acceptability of PSCs for residential land use given current conditions. Residential land use implies that a site can be developed and used for any purpose, including residential development. If a PSC was deemed unsuitable for residential land use, remedial alternatives were developed for that site. Remedial alternatives were also developed for any site that could potentially impact underlying groundwater resources in the future. The OU-1 FS results are detailed in the OU-1 FS report¹⁸. OU-1 RI/FS results are summarized in Table 3-2.

¹⁴ Geraghty & Miller, 1993c. OU-1 Phase II Remedial Investigation/Feasibility Study Field Sampling and Analysis Plan, Luke Air Force Base, Arizona.

¹⁵ Geraghty & Miller, 1994. Bioventing Treatability Study Sampling and Analysis Plan for PSC SS-42. Luke Air Force Base, Arizona.

¹⁶ Geraghty & Miller, 1995. Final Sampling and Analysis Plan Environmental Evaluation in Support of the Ecological Risk Assessment, Luke Air Force Base, Arizona.

¹⁷ Geraghty & Miller, 1995. Final Sampling and Analysis for the Additional Sampling Investigations in Support of the Luke AFB CERCLA investigation, Luke AFB, Arizona.

¹⁸ Geraghty & Miller, 1998. Final OU-1 Feasibility Study Report, Luke Air Force Base, Arizona. March 1998.

PSC	Description	COCs Evaluated	COCs Outside Risk- Based Range	Selected Remedial Alternative
OT-01	Old incinerator site	An extensive data review of base records indicated that hazardous materials and wastes were never handled or disposed at this location.	RI not required	No action
RW-02	Wastewater treatment annex landfill	2-methylnaphthalene, benzo(a)anthracene, benzo(a)pyrene, benzo(b)fluoranthene, benzo(g,h,i)perylene, benzo(k)fluoranthene, bis(2-ethylhexyl)phthalate, chrysene, di-n- butyl phthalate, di-n-octyl phthalate, fluoranthene, indeno(1,2,3-cd)pyrene, phenanthrene, pyrene, TRPH, arsenic, barium, beryllium, cadmium, chromium, copper, lead, mercury, nickel, silver, uranium, zinc, gross alpha, gross beta, radium-226, radium-228	Radionuclides are currently be monitored, however the COCs were below risk standards	Institutional controls, radiological monitoring and fencing
LF-03	Outboard runway landfill	TRPH, arsenic, barium, beryllium, cadmium, chromium, copper, lead, nickel, silver, zinc	COCs below risk standard	Institutional controls
FT-07E	Eastern portion of north fire training area	acetone, ethylbenzene, toluene, xylenes, TRPH, arsenic, barium chromium, copper, lead, nickel, zinc	TRPH are below non-residential ADEQ SRLs	Institutional controls
OT-08	F-15 burial site	An extensive data review of base records indicated that hazardous materials and wastes were never handled or disposed at this location.	RI not required	No action
OT-09	Canberra burial site	An extensive data review of base records indicated that hazardous materials and wastes were never handled or disposed at this location.	RI not required	No action

Table 3-2 Summary of OU-1 RI/FS Results

PSC	Description	COCs Evaluated	COCs Outside Risk- Based Range	Selected Remedial Alternative
OT-10	Concrete rubble burial site	This site lies completely within the boundaries of DP-13.	RI not required	No action
SS-11	Former outside transformer storage	PCBs	Risk below risk range	No action
OT-12	Old explosive ordnance division (EOD) burial site	Acenaphthylene, anthracene, benzo(a)anthracene, benzo(a)pyrene, benzo(b)fluoranthene, benzo(g,h,i)perylene, benzo(k)fluoranthene, carbazole, chrysene, dibenz(a,h)anthracene, fluoranthene, indeno(1,2,3-c,d)pyrene, pentachlorophenol, phenanthrene, pyrene, TRPH, arsenic, barium, beryllium, cadmium, chromium, copper, lead, nickel, zinc, cyanide	Risk below risk range, with the exception of Benzo(a)pyrene	No action
DP-13	Drainage ditch disposal area	acenaphthene, anthracene, benzo(a)anthracene, benzo(a)pyrene, benzo(b)fluoranthene, benzo(g,h,i)perylene, benzo(k)fluoranthene, bis(2-ethylhexyl)phthalate, carbazole, chrysene, dibenz(a,h)anthracene, dibenzofuran, fluoranthene, fluorine, indeno(1,2,3-cd)pyrene, phenanthrene, pyrene, TRPH, antimony, arsenic, barium, beryllium, cadmium, chromium, copper, lead, mercury, nickel, zinc, cyanide	Chromium Lead Benzo(a)pyrene	Institutional controls
LF-14	Old salvage yard burial site	Xylenes, benzo(a)anthracene, benzo(a)pyrene, benzo(b)fluoranthene, benzo(g,h,i)perylene, benzo(k)fluoranthene, butyl benzyl phthalate, chrysene, fluoranthene, indeno(1,2,3-cd)pyrene, phenanthrene, pyrene, PCBs, TRPH, arsenic, barium, beryllium, cadmium, chromium, copper, lead, nickel, silver, zinc, cyanide,	Polychlorinated Biphenyls (PCB)	Institutional controls

PSC	Description	COCs Evaluated	COCs Outside Risk- Based Range	Selected Remedial Alternative
SS-15	Facility 328 spill site	This site were removed from the superfund process and placed under the ADEQ UST jurisdiction.	RI not required	ADEQ jurisdiction
SS-16	Facility 321 underground storage tank (UST).	This site were removed from the superfund process and placed under the ADEQ UST jurisdiction.	RI not required	ADEQ jurisdiction
SS-17	Former defense property disposal office (DPDO) yard.	chrysene, di-n-octyl phthalate, fluoranthene, pyrene, PCBs, TRPH, arsenic, barium, beryllium, cadmium, chromium, copper, lead, nickel, silver, zinc	Risk below risk range	No action
ST-19	Base exchange (BX) leaking USTs.	This site were removed from the superfund process and placed under the ADEQ UST jurisdiction.	RI not required	ADEQ UST jurisdiction
SD-20	Oil/water separator canal and earth fissures.	Toluene, benzo(a)anthracene, benzo(a)pyrene, benzo(b)fluoranthene, benzo(g,h,i)perylene, bis(2- ethylhexyl)phthalate, chrysene, di-n-octylphthalate, fluoranthene, indeno(1,2,3-c,d)pyrene, phenanthrene, pyrene, TRPH, antimony, arsenic, barium, beryllium, cadmium, chromium, copper, lead, nickel, zinc	Risk below risk range, with the exception of Benzo(a)pyrene	No action
SD-21	Sewage treatment plant effluent canal	Anthracene, benzo(a)anthracene, benzo(b)fuoranthene, benzo(k)fluoranthene, benzo(g,h,i)perylene, benzo(a)pyrene, chrysene, dibenzo(a,h)anthracene, fluoranthene, indeno (1,2,3-c,d)pyrene, phenanthrene, pyrene, TRPH, arsenic, barium, beryllium, cadmium, chromium, copper, lead, nickel, silver, zinc	Risk below risk range, with the exception of Benzo(a)pyrene	No action
DP-24	Base ammunition storage area	Removed from the Superfund process because this site was mistakenly included on the list of potentially contaminated sites.	RI not required	No action

PSC	Description	COCs Evaluated	COCs Outside Risk- Based Range	Selected Remedial Alternative
LF-25	Northwest landfill	Xylenes, benzo(a)anthracene, benzo(a)prene, benzo(b)fluoranthene, benzo(k)fluoranthene, chrysene, fluoranthene, pyrene, TRPH, antimony, arsenic, barium, beryllium, cadmium, chromium, copper, lead, nickel, silver, zinc, cyanide	Lead and antimony were above risk range	Lead shot recovery Institutional controls
SD-26	Hush house canal	ethylbenzene, toluene, xylenes, 2-methylnaphthalene, benzo(a)anthracene, benzo(a)pyrene, benzo(b)fluoranthene, benzo(g,h,i)perylene, benzo(k)fluoranthene, bis(2-ethylhexyl)phthalate, chrysene, di-n-butylphthalate, fluoranthene, indeno(1,2,3-cd)pyrene, naphthalene, phenanthrene, pyrene, TRPH, arsenic, barium, beryllium, cadmium, chromium, copper, lead, nickel, silver, zinc	Risk below risk range	No action
LF-37	Northeast landfill	benzo(a)anthracene, benzo(a)pyrene, benzo(b)fluoranthene, benzo(g,h,i)perylene, benzo(k)fluoranthene, bis(2-ethylhexyl)phthalate, butylbenzylpthalate, chrysene, fluoranthene, pyrene, TRPH, arsenic, barium, beryllium, cadmium, chromium [b] copper, lead, nickel, silver, zinc	Risk below risk range, with the exception of Benzo(a)pyrene	No action
SD-38	Southwest oil/water separator at the auto hobby shop	TRPH, antimony, arsenic, barium, beryllium, cadmium, chromium, copper, lead, nickel, zinc	TRPHs were below non-residential ADEQ SRLs	Institutional controls
SD-39	Waste discharge at the old Lockheed site	diethyl phthalate, TRPH, arsenic, barium, cadmium, chromium, copper, lead, nickel, zinc	Risk below risk range	No action
OT-41	Skeet range	Lead	Risk below risk range	No Action

PSC	Description	COCs Evaluated	COCs Outside Risk- Based Range	Selected Remedial Alternative
SS-42	Bulk fuels storage	Benzo(a)anthracene, benzo(b)fluoranthene, bis(2-ethylhexyl)phthalate, chrysene, di-n-butylphalate, fluoranthene, pyrene, TPH, lead	Benzene, Toluene, Total Xylenes, TPH	Soil vapor extraction (SVE) and five-year groundwater monitoring
3.3 OU-2 ROD Summary

The description of the remedy in the OU-2 ROD is summarized in Table 3-3.

PSC	Selected Remedy	Implemented Remedial Components
ST-18	Capping, Surface Controls, and Groundwater Monitoring.	Concrete CAP installed in 1987, Annual Inspection and maintenance of a concrete cap and groundwater monitoring during each 5-year review
DP-23	Excavation, ex-situ biological treatment, confirmation sampling, and on-site disposal of impacted soils from the canal portion.	Design and implementation of excavation and on-site ex- situ biological treatment of soils impacted by PAHs above industrial PRGs.

 Table 3-3 Description of the Remedy for OU-2

3.4 OU-1 ROD Summary

The description of the remedy in the OU-1 ROD is summarized in Table 3-4.

Table 3-4 Description	of the Reme	dy for OU-1
------------------------------	-------------	-------------

PSC	Selected Remedy	Implemented Remedial Components
RW-02	Institutional Controls, Radiological Monitoring, and Fencing	 Voluntary Environmental Mitigation Use Restriction (VEMUR) executed and recorded to restrict land use to non-residential purposes. Base General Plan (BGP) modified to place constraints on future residential development of the site. Geophysical monitoring program to ensure safety of potential receptors and warning mechanism in case subsurface conditions change. Perimeter fencing. Institutional Control Plan (ICP) to maintain and document required institutional controls.
LF-03	Institutional Controls	 A VEMUR to restrict land use to non-residential. The BGP modified to place constraints on future residential development. An ICP to document required institutional controls.
FT-07E	Institutional Controls	 A VEMUR to restrict land use to non-residential. The BGP modified to place constraints on future residential development. An ICP to document required institutional controls.

PSC	Selected Remedy	Implemented Remedial Components
DP-13	Institutional Controls	 A VEMUR to restrict land use to non-residential. The BGP modified to place constraints on future residential development. Work practices requiring the use of Personal Protective Equipment (PPE) while excavating the site. An ICP to document required institutional controls.
LF-14	Institutional Controls	 A VEMUR to restrict land use to non-residential. The BGP modified to place constraints on future residential development. An ICP to document required institutional controls.
LF-25	Institutional controls/ex-situ physical treatment/metal recovery	 The area of impacted soils containing COCs in excess of evaluated criteria to be further delineated. Surficial soils with COCs in excess of Arizona soil remediation standards to be excavated and disposed. Remediation of metal shot via mechanical sifting and gravimetric separation. Recovered metal shot recycled or disposed. Soil material returned to excavated areas. A VEMUR to restrict land use to non-residential. The BGP modified to place constraints on future residential development. Work practices requiring the use of PPE while excavating the site. An ICP to document required institutional controls.
SD-38	Institutional Controls	 A VEMUR to restrict land use to non-residential. The BGP modified to place constraints on future residential development. An ICP to document required institutional
SS-42	Soil Vapor Extraction and Groundwater Monitoring	 Install SVE System. Monitor soil and groundwater to confirm effectiveness of remedy.

4.0 Summary of Base-wide Risk Assessment

This section of the report provides a summary of the approach used in the development of the Base-wide risk assessment¹⁹. The risk assessment evaluated current and potential future risks to human health and the environment from exposure to the constituents of potential concern (COPCs) in soil, sediment, groundwater, surface water, and ambient air at each of the PSCs. The following summarizes the results of

¹⁹ Geraghty & Miller, 1997. Final Remedial Investigation Report OU-1, Appendix B – Baseline Base Wide Risk Assessment, Luke Air Force Base, Arizona. Volumes 1 and 2. October 1, 1997.

the risk assessment. Luke AFB is an active military facility, and is expected to remain active in the foreseeable future. Therefore, the most likely type of exposure is for industrial workers rather than residents. Exposure to soil and sediments included the ingestion, dermal contact, and inhalation of fugitive dusts and/or vapors. Risks from exposure to soils and sediments were evaluated using either surficial, base worker scenario (0 to 2 feet bgs) or combined surface and subsurface, excavation worker (0 to 16 feet bgs) data. Exposure to groundwater was evaluated through the ingestion and dermal contact pathways. Potential groundwater exposure was evaluated using production well sampling data as well as data collected from groundwater monitoring wells.

Potentially exposed populations considered in the risk assessment included the following:

- Base workers
- Excavation workers
- Military personnel
- Child visitors for sites which extend off the base property
- Base residents.

The risk assessment considered both average and reasonable maximum exposure conditions to characterize current and future risks. During the five-year review, exposure point concentrations (EPC) were taken from the 95 percent upper confidence limit (one-tailed) on the mean, assuming a normal distribution. For PSCs with post-remediation data, the EPC was taken from the maximum concentration from the appropriate soil horizon or medium.

The USEPA established risk-based guidance goals as an aid in determining which sites would be acceptable for use in an industrial setting in the absence of remediation. Sites at which a non-cancer hazard index (HI) greater than or equal to 1.0 or an excess lifetime cancer risk (ELCR) greater than the risk of 1×10^{-6} to 1×10^{-4} , as determined by risk assessment, generally would require remediation and would therefore be recommended for inclusion in the FS.

Arsenic and beryllium were found to be constituents that potentially contributed most significantly to the estimates of risk in the assessment. The results of the risk assessment were reevaluated to determine the impact background had on the level of risk at the various PSCs. When background was considered in the evaluation, it was found that most of the naturally occurring inorganic constituents (e.g., arsenic,

beryllium) were present at background levels. Remediation to concentrations below background is not typically required by USEPA. Therefore, based on this reevaluation of the risk assessment results, only two PSCs, LF-25 and SS-42, were found to pose a risk above the target risk range.

To determine whether the PSC areas at Luke AFB are suitable for future residential land use, risks from exposure to soil by a hypothetical future resident were evaluated. Because Luke AFB is an active military facility, and is expected to remain active in the foreseeable future, calculating risks for residential exposure to soils at each of the PSCs is highly conservative; it is unlikely that the active portions of Luke AFB will be used for residential purposes in the future. The ADEQ proposed soil remediation levels (SRLs)²⁰ and the USEPA Region IX PRGs were used in the residential exposure evaluation.

Risks from residential exposure to combined surface and subsurface soil were calculated using both the USEPA Region IX PRGs and the ADEQ SRLs. Based on the results of the evaluation, all of the PSC areas evaluated were determined to be suitable for unrestricted, or residential land use with the exception of the following PSCs:

- RW-02
- LF-03
- FT-07 E
- DP-13
- LF-14
- ST-18
- LF-25DP-23 N
- DP-23 f
 SD-38

In addition to evaluating potential human exposure at Luke AFB, an ecological risk assessment was also performed. Prior to completing the ecological risk assessment, a Base-wide ecological inventory (EI) was conducted to collect data on:

- Biotic communities present on the base.
- Evidence of biological stress.
- Pathways of potential exposure to impacted media.

²⁰ ADEQ, 1996. A.A.C. Title 18, Chapter 7, Article 2, Appendix A. Soil Remediation levels.

• The presence of species of special concern.

Luke AFB is in the lower Colorado River Valley of the Sonoran Desert. However, little vegetation characteristics of this area were identified during the EI. Instead, flora was dominated by vegetation characteristic of urban, disturbed areas at similar elevations in the Sonoran Desert. This is consistent with current and past land use at the Base.

No species of special concern were observed during the EI. Animal species observed at the Base during the EI are more tolerant of urban and disturbed conditions. Because vegetative growth at the Base is sparse due to physical activities associated with normal operations, the diversity and abundance of animals observed were less than that typical in more native conditions.

Potential risks to ecological receptors were assessed quantitatively by using the round-tailed ground squirrel, desert cottontail, western whiptail lizard and side-blotched lizard as indicator species. The desert cottontail was used to represent herbivorous primary consumers; the round-tailed ground squirrel to represent herbivorous/insectivorous primary consumers; and the western whiptail lizard and side-blotched lizard to represent insectivorous secondary consumers. HQs were calculated for the indicator species by comparing an estimated intake of site-related constituents of ecological concern (COECs) with a toxicity reference value derived for the specific indicator species and for the specific COEC. Hazard quotients (HQs) were determined for the ingestion of food sources and for the incidental ingestion of soil where appropriate for the indicator species. The HQs were then added to obtain a HI for each PSC.

Based on previous investigations at Luke AFB and coordination with USEPA representatives, the following PSCs were determined to be representative of site conditions and were selected for study in the ecological risk assessment:

- LF-25
- FT-07
- Combined portions of SS-17 and LF-14
- SD-20

This selection was based on a combination of observations of ecosystems at the PSCs, detected COEC concentrations, and potential risks to higher trophic level organisms.

COECs evaluated in the ecological risk assessment included the following:

- PAHs
- TPH
- PCBs
- Antimony
- Cadmium
- Lead

Data used to assess potential adverse effects to ecological receptors included chemical analysis of soil, plant tissue, and insect tissue. Based on the results of the ecological assessment, it is unlikely that site-related COEC concentrations would pose a risk to ecological receptors at Luke AFB.

5.0 Groundwater Summary

This section of the five-year review report provides information on the hydrogeology of Luke AFB and surrounding region. Also provided is information on the status of monitor wells and a summary of groundwater monitoring results for PSCs.

5.1 Groundwater Hydrology

5.1.1 Aquifer Units

The occurrence and movement of groundwater at Luke AFB is affected by hydraulic characteristics of the aquifer units, and the magnitude and distribution of groundwater recharge and discharge for agriculture and other uses. Aquifer units include the upper alluvial unit (UAU), middle fine unit (MFU) and lower conglomerate unit (LCU). Withdrawals in excess of recharge have created declines in water levels in the Luke AFB area of 300-feet²¹. Structural changes associated with the Luke Salt Body significantly affect local groundwater conditions south and east of the Luke AFB.

Interpolation of data from the regional study of Brown and Pool²² indicates that the UAU has been completely dewatered in the Luke AFB area, except for localized

²¹ US Geological Survey, 1994. Hydrogeological Characterization and Land Subsidence Investigation for Luke Air Force Base, Arizona.

²² US Geological Survey, 1998. Water Resources Investigation Report 88-4202, 1989, Hydrogeology of the Western Part of the Salt River Valley, Maricopa County, Arizona (Brown, JamesG. Pool, D.R.)

areas along the Agua Fria River, near the Luke AFB Waste Water Treatment Plant (WWTP). Partial dewatering of the MFU has also occurred in the Luke AFB area. The upper most aquifer is now the MFU.

5.1.2 Regional Recharge

Groundwater recharge in the WSRV is affected by natural as well as artificial sources. Groundwater is naturally recharged by infiltration through the beds of river channels during stormwater events or releases from upstream impoundments. Water levels in Monitoring Well MW-101, situated near the Aqua Fria River, rose approximately 25-feet between December 28, 1992, and March 14, 1993, in response to upstream releases from Lake Pleasant Reservoir ²³.

Artificial sources of groundwater recharge include infiltration of excess irrigation water applied to fields and seepage losses from irrigation ditches and canals. Infiltration of treated effluent from the Luke AFB WWTP may also provide recharge in the immediate area of the releases to the Aqua Fria River floodplain. Potential recharge due to other activities at the Luke AFB is discussed in greater detail in the Vadose Zone leaching model presented in the Base-wide risk assessment.

5.1.3 Regional Discharge

Groundwater discharge from the regional aquifer in the Luke AFB area occurs primarily from cultural uses. Owing to the depth to the water table, there is no natural discharge due to evapotranspiration or discharge to surface water bodies. Discharge of groundwater occurs principally from pumpage from numerous wells, primarily for irrigation with the remainder for municipal, military, and light industrial consumption.

The amount of groundwater discharge for municipal usage is anticipated to increase dramatically in response to the growing population of the area (Water Resources Associates, 1994). As the population increases in the area it is anticipated that groundwater discharged for agricultural uses will decrease (Water Resources Associates, 1994). Comparison of the increased withdrawals for municipal uses and decreased withdrawals for irrigation uses shows that the demand for groundwater in the area will remain generally the same into the foreseeable future. However, the

²³ Geraghty & Miller, Inc. 1993d. OU-1, Phase II, Remedial Investigations/Feasibility Study Planning Documents for PSC SS-42, Luke Air Force Base, Arizona.

transition from irrigation uses to municipal uses will put greater importance on water quality.

5.1.4 Historical Trends in Regional Groundwater Levels

Groundwater levels declined more than 300 feet in the vicinity of Luke AFB over a 40+ year period from 1923 to the late 1970s, primarily because of significant overdraft in response to pumpage for irrigation requirements. The greatest declines occurred west, north, and south of Luke AFB. A large cone of depression has existed southwest of the Luke AFB prior to 1964. The regional groundwater flow direction is to the south-southwest modified by the cone of depression.

Water levels from selected wells for which data were adequate were plotted to show groundwater declines over time at a given location. Analysis of these hydrographs suggests that water levels have declined substantially over most of the study area through at least 1980. After 1980, many of the hydrographs show a leveling off of the decline trend, or a groundwater rise of up to 40 to 60 feet. Groundwater table altitudes in the study area have continued to rise due to reduction in pumpage and increased recharge related to above average precipitation over the early 1980s in the Phoenix area. The availability of Colorado River water via the Central Arizona Project (CAP) canal (especially for agricultural irrigation) has greatly lessened the demands placed on groundwater in the Phoenix area, and has resulted in the groundwater table rising throughout much of the area.

Water level data for the period 1991 to 1995, documents a continued rise in the groundwater table throughout the study area. Altitudes had increased up to 20 feet, in large part due to above average precipitation for 1992 and 1993. Overall, the historical groundwater altitude data for the study area shows a consistent pattern of water level decline over time despite the limitations in the data previously described.

5.1.5 Groundwater Occurrence, Apparent Gradient, and Estimated Flow Directions in the Luke AFB Area

Water Level Measurements

The water level measurement program was established for the Luke AFB RI and included monthly water level measurements and continuous water level measurements. Monthly water level data were collected to evaluate seasonal water level responses from regional stresses on the groundwater system. Continuous water level measurement were collected at selected monitoring wells using pressure transducers

and data loggers. Continuous water level data were collected to evaluate local water responses from regional as well as local pumping stresses due to both off base and on-Base production well pumping²⁴. Because several different production wells are in use, the data loggers and transducers were periodically moved to collect data from all parts of the Base.

The monthly water level program included measurements of all monitoring wells and selected off Base wells. The program began in October 1990, and extended through December 1995.

Continuous water level recorders were installed periodically in Monitoring Wells MW-3, MW-5, MW-102, MW-103, MW-104, MW-106 through MW-111, MW-112S, MW-112D, MW-113, MW-117, MW-118, MW-119, MW-121, MW-123 and Inactive Production Well IP-PW-12. Nearby on Base production wells, which potentially affected water levels, include Production Wells PW-4, PW-7, PW-9, PW-10, PW-11, PW-12, PW-13, and PW-14.

Well Perforated Intervals and Measured Water Levels.

All of the monitoring wells at the main part of Luke AFB are screened entirely within the MFU. Interpolation of data from the regional study indicates that all monitor well locations near the Luke AFB WWTP (MW-101, MW-115, MW-116, and MW-124) are screened in the UAU. Monitoring Well MW-101 is primarily screened in the UAU with the lower portion of the screen extending into the MFU. Monitoring Wells MW-115, MW-116, MW-116, and MW-124 are screened in the lower portions of the UAU. All of the monitoring wells at the main Base, except for Monitoring Wells MW-102, MW-103, and MW-112D are screened in upper parts of the saturated thickness of the MFU.

The Luke AFB production wells are screened typically in the LCU with some wells also screened in the MFU. Seven of the 15 off Base wells included in the monthly water level network are exclusively perforated within the MFU, and five other off Base wells are perforated within both the MFU and LCU. Six of the off Base wells may have casing collapses below the current indicated depth of the well, which may suggest that these wells may be open to the formations below the indicated depth of

²⁴ Geraghty & Miller, Inc., 1992. First Quarter 1992 Quarterly Well Measurement Report, Hydrogeological Survey, Luke Air Force Base, Arizona.

perforations. Limited hydraulic connection may exist in the well bore with deeper portions of the formation at the location of the casing collapse.

Limitations in the Construction of Water Level Altitude Contour Plots

Water level altitude contour plots are typically used to infer groundwater flow directions. Evidence from water levels measured at site-specific PSC wells at Luke AFB suggest that semi-independent groundwater zones have developed at the local scale as the result of long-term water level declines in response to regional groundwater withdrawals in excess of recharge. Water level measurements also suggest that head differences between zones have created vertical gradients within and between these zones. The development of these semi-independent groundwater zones makes it imperative that water level measurements used in the construction of water level altitude contours be from wells with perforated intervals which extend and penetrate into similar lithologic portions of these zones or when compatible data is not available, that the data be used with an understanding of its comparability and hence the accuracy of the resulting plot. One approach is to use contour intervals that are large enough to limit the effect of small vertical head gradients within these zones.

Evidence which supports the existence of semi-independent groundwater zones at Luke AFB includes anomalous water levels within similar areas, limited or non-response of water levels in wells near pumping wells, and limited or non-response of water levels in wells to seasonal water-level changes experienced by other nearby wells. Anomalous water levels occur at PSC SD-20 where water levels are as much as 50-feet lower than at PSCs immediately to the northwest. These anomalous water levels are suspected to be attributed to the geologic structure associated with the Luke Salt Body.

Non-responsive water levels in wells near a pumping well occur at PSC FT-07 where pumpage from production well PW-12 (approximately 1,000 gallons per minute) does not cause any direct water level response in nearby monitoring wells. The lack of response of the water levels is suspected to be attributed to an aquitard which most likely occurs between the largest penetrating perforated interval of the monitoring wells at the PSC (453 feet bgs at MW-109) and the top of the perforated interval of the production well (600 feet bgs at PW-12). This aquitard limits the vertical hydraulic connection between the perforated intervals of these wells.

Water level changes at PSC FT-07 also experience the smallest amplitude of seasonal change of any of the monitoring wells at Luke AFB. Seasonal water level amplitudes in the monitoring wells at PSC FT-07 average approximately five feet. Seasonal water level amplitudes in monitoring wells at other Luke AFB PSCs range from

approximately five feet at PSCs FT-06/ST-18 to 23 feet at PSC SD-20. The limited response of water levels in these wells to regional changes in water levels is likely caused by the aquitard, which is suspected to exist at this location.

Evidence which supports the existence of vertical head gradients within groundwater zones include water level differences between nearby wells with slight to significant differences in saturated extent and penetration of perforated intervals. This is evident at PSC FT-07 and SD-20. Monitoring wells MW-110 (saturated perforated interval from approximately 362 to 398 feet bgs [screened interval from approximately 362 to 398 feet bgs]) and MW-123 (saturated perforated interval from approximately 340 to 395 feet bgs [total screened interval from 295 to 395 feet bgs]) at PSC FT-07 located within 30 feet of each other have consistent differences in water levels of approximately 290 to 342 feet bgs [screened interval from 780 to 722 feet above mean sea level]) and MW-112D (saturated perforated interval from 381 to 428 feet bgs [screened interval from 682 to 632 feet above mean sea level]) at PSC SD-20 have indicated differences in water levels of 0.5 to 8.5 feet (Figure 4-48).

The hydraulic effects associated with the development of the semi-independent groundwater zones at Luke AFB requires special precautions in the construction of site-specific PSC water level altitude contour plots. Water level data used to construct water level contours needs to be relatively comparable in that data should be from wells, which measure similar lithologic portions of the water zones. However, most PSCs have limited number of wells, which measure similar lithologic portions of the water zones. Water-level altitude contour plots for the site specific PSCs have therefore been constructed with the use of site-specific and regional water level measurements, and a contour interval of ten feet to limit the impact of the effects. Regardless of these precautions, groundwater flow directions inferred from these contour plots should still be viewed with a degree of scrutiny, and should only be used as the direction of a general groundwater flow at the time of water level measurement.

Contour intervals at site specific PSCs at Luke AFB were constructed using a regional ten feet contour interval. Because of the inherent effects on measured water levels from the vertical component of flow, the regional contours provide the best representation of the lateral groundwater flow direction a two dimensional contour plot can achieve.

Impact of Luke Salt Body on Groundwater Flow

The Luke Salt Body has impacted groundwater movement in the study area. Groundwater flow in the vicinity of the salt mass has been affected in several ways. Doming of the Luke Salt Body has deformed the overlying and peripheral sediments through high angle faulting and folding, as mentioned previously in Section 4.6.5. Furthermore, permeability and anisotropy of the basin fill sediments in the study area have been altered by depositional changes from proximal coarse-grained to distal fine-grained sediments peripheral to the Luke Salt Body, and compaction of the fine-grained sediments overlying the Luke Salt Body. In conjunction with the structural complexities, the result has been to create a complex groundwater flow regime in the vicinity of the Luke Salt Body. The effect on groundwater movement and aquifer properties along the southern portions of the Base is not fully understood.

Regionally, the localization of fine-grained sediments and the additional compaction over the Luke Salt Body have resulted in reduced transmissivities east and south of Luke AFB within the basin fill units ^{25,26}. This reduction has impacted regional groundwater movement into the pumping areas west of Luke AFB, with higher water levels to the east of the Base (near WWTP) and lower water levels in wells in the western part of Luke AFB. Although the groundwater flow across the site is generally southwest, near monitoring wells MW-112 and MW-113, the groundwater flow is northwest because of the Luke Salt Body.

5.2 PSC Groundwater Summary

The status of groundwater monitoring wells at Luke AFB is summarized in Table 5-1. The location and status of groundwater monitoring wells is in Figure 5-1. Groundwater sampling results are summarized in Table 5-2 and Figure 5-2. In addition, a table that provides groundwater monitoring results including sampling dates and concentrations of COCs for all wells for the period of record is in Appendix B. In general, a review of the data in Appendix B indicate that groundwater at Luke AFB is not impacted as there

²⁵ Stulik, R.S. and F.R. Twenter, 1964. Geology and Groundwater of the Luke Area, Maricopa County, Arizona. USGS Water Supply Paper #1799P.

²⁶ Eaton, G.P., D.L. Peterson, and D.L. Schumann, 1972. Geophysical, Geohydrological, and Geochemical Reconnaissance of the Luke Salt Body in Central Arizona, USGS Professional Paper #753.

are no constituents in groundwater that currently exceed applicable water quality standards and it appears all potential sources of constituents have been controlled or eliminated through the institution of pollution prevention measures or remedial activities. However, there have been some occurrences of COCs above detection limits during the period of record. Therefore, some monitoring will be conducted as part of future five-year reviews to verify that remedies are protective of groundwater. MW-124 (RW-02), MW-118 and MW-123 (FT-07),), MW-112S, MW-112D and MW-113 (SD-20) and MW-121 and MW-125 (SS-42) will be monitored for VOCs as part of future five-year reviews. MW-114 and MW-122 (ST-18) will be monitored annually for VOCs and the results evaluated as part of the next five-year review.

The following sections provide a more detailed description of groundwater conditions associated with specific PSCs.

Table 5-1 Status of Groundwater Monitoring Wells

Well ID	Site ID	Site Location	Status	Comments
MW-1	ST-18	Facility 993	Abandoned	Unknown location
MW-2	ST-18	Facility 993	Abandoned	COC concentrations did not warrant additional monitoring.
MW-3	ST-18	Facility 993		Possible RCRA point of compliance
MW-4	ST-18	Facility 993	Abandoned	COC concentrations did not warrant additional monitoring.
MW-5	ST-18	Facility 993		Possible RCRA point of compliance
MW-101	SD-21	Aqua Fria River	ACTIVE	Needed for Compliance Sampling
MW-102	SD-20	Head of O/W Separator Canal	Abandoned	COC concentrations did not warrant additional monitoring.
MW-103	SD-20	Near Ammo. Storage	Abandoned	COC concentrations did not warrant additional monitoring.
MW-104	DP-05	Eastern edge of Site	Abandoned	COC concentrations did not warrant additional monitoring.
MW-105	DP-05	Southern edge of Site	Abandoned	COC concentrations did not warrant additional monitoring.
MW-106	DP-05	Western edge of Site	Abandoned	COC concentrations did not warrant additional monitoring.
MW-107	FT-06	Near Bldg. 1031	Abandoned	COC concentrations did not warrant additional monitoring.
MW-108	FT-06	North of DP-23	Active	

Well ID	Site ID	Site Location	Status	Comments	
MW-109	FT-07	Western edge of Site	Active		
MW-110	FT-07	Central portion of Site	Abandoned	COC concentrations did not warrant additional monitoring.	
MW-111	FT-07	North of perimeter road.	Abandoned	COC concentrations did not warrant additional monitoring.	
MW-112S	SD-20	Earth Fissures area	Active		
MW-112D	SD-20	Earth Fissures area	Active		
MW-113	SD-20	Earth Fissures area	Active		
MW-114	ST-18	Point of release at ST-18		Possible RCRA point of compliance	
MW-115	RW-02	Church parking lot	Abandoned	COC concentrations did not warrant additional monitoring.	
MW-116	RW-02	Tanner property		To be Abandoned	
MW-117	SD-38	South of Auto Hobby Shop	Active		
MW-118	FT-07	Between FTPs	Active		
MW-119	SS-42	Near central portion of site.	DETERIORATED	To be Abandoned	
MW-120	SS-42	CE yard	DETERIORATED	To be Abandoned	
MW-121	SS-42	Point of release at SS-42	ACTIVE	Needed for Sampling per OU-1 ROD	
MW-122	ST-18	Downgradient of ST-18		Possible RCRA point of compliance	

Well ID	Site	Site	Status	Comments
	ID	Location		
MW-123	FT-07	Downgradient of FTPs	Active	
MW-124	RW-02	DRMO yard	Active	
MW-125	SS-42	Downgradient of SS-42	ACTIVE - Replaced	Needed for Sampling per OU-1 ROD

Operable Unit	PSC	COCs in Soil (based on post remediation when available)	COCs Detected in Groundwater	Source of Detected Constituents	Attributable to PSC
OU-1	RW-02	Benzo(a)pyrene, arsenic, beryllium, cadmium, copper	bis(2- ethylhexyl)phthalate (BEP), arsenic, barium, chromium, copper, nickel, lead, zinc	BEP - Laboratory contaminant chromium, copper, nickel and lead were found within the range of naturally occurring concentrations.	No constituents attributable to PSC
OU-2	DP-05	ethylbenzene, xylenes, bis(2- ethylhexyl)phthalate, 2- methylnaphthalene, naphthalene, TRPH, copper, lead	Acetone, toluene, BEP barium, lead, silver, arsenic, barium, chromium, copper, lead, zinc	Acetone and toluene were detected in one well during one sampling event and not detected again. BEP is a common laboratory contaminant. Barium and lead were found within the range of naturally occurring concentrations.	No constituents attributable to PSC
OU-1	FT-07	arsenic, TRPH	Chloroform, 1,2- dichloropropane (DCP), 1,2-dichloroethane (DCA), toluene, BEP barium, chromium, copper, lead, zinc	Chloroform, DCP, DCA, and BEP were found to laboratory contaminants. Toluene source unknown but detected in only one well. Barium, chromium, copper, zinc and lead were found within the range of naturally occurring concentrations.	No constituents attributable to PSC

Table 5-2 Groundwater Sampling Summary

Operable Unit	PSC	COCs in Soil (based on post remediation when available)	COCs Detected in Groundwater	Source of Detected Constituents	Attributable to PSC
OU-2	ST-18	benzene, 1,1 dichloroethene, ethylbenzene, 1,1,2,2- tetrachlorethane, tetrachloroethene, toluene, trichloroethene, xylenes, benzo(a)anthracene, benzo(b)fluoranthene, benzo(g,h,i)perylene, benzo(a)pyrene, benzyl alcohol, bis(2-ethylhexyl)phthalate, chrysene, fluoranthene, indeno(1,2,3-c,d)pyrene, 2- methylnaphthalene, naphthalene, phenanthrene, pyrene, TRPHs, copper, lead	Toluene BEP, DBCP, 2-butanone, arsenic, barium, chromium, copper, lead, nickel, selenium, zinc and TCE.	Toluene was detected in one well during one sampling event and not detected again. BEP is a common laboratory contaminant and DBCP is an agricultural contaminant. Arsenic, barium, chromium, copper, nickel, selenium, zinc and lead were found within the range of naturally occurring concentrations.	Potentially attributable to PSC (monitoring continuing).
OU-1	SD-20	toluene, benzo(a)anthracene, benzo(a)pyrene, benzo(b)fluoranthene, benzo(g,h,i)perylene, bis(2- ethylhexyl)phthalate, chrysene, di-n-octylphthalate, fluoranthene, indeno(1,2,3-c,d)pyrene, phenanthrene, pyrene, TRPH, antimony, arsenic, barium, beryllium, cadmium, chromium, Copper, lead, nickel, zinc	TCE, PCE, DCA, DCP, BEP, acetone, bromodichloromethane, bromoform, chloroform, methylene chloride, arsenic, barium, boron, chromium, copper, lead, nickel, selenium, and zinc	TCE and DCA source unknown but not detected in wells at source area. DCP detected one time. BEP is a common laboratory contaminant. Arsenic, barium, boron, chromium, copper, nickel, selenium, zinc and lead were found within the range of naturally occurring concentrations.	It is possible that TCE originating from points on-base that discharged to the oil water separator canal may have migrated to the groundwater. The fissures at the end of the canal may have provided a pathway for these constituents to reach groundwater.

Operable Unit	PSC	COCs in Soil (based on post remediation when available)	COCs Detected in Groundwater	Source of Detected Constituents	Attributable to PSC
OU-1	SD-21	anthracene, benzo(a)anthracene, benzo(b)fluoranthene, benzo(k)fluoranthene, benzo(g,h,i)perylene, benzo(a)pyrene, chrysene, dibenzo(a,h)anthracene, fluoranthene, indeno(1,2,3- c,d)pyrene, phenanthrene, pyrene, TRPH, arsenic, barium, beryllium, cadmium, chromium, copper, lead, nickel, silver, zinc	Acetone, carbon disulfide, and BEP arsenic, barium, boron, copper, lead, and zinc	Acetone, carbon disulfide and BEP were found to be laboratory contaminant. Arsenic, barium, boron, copper, zinc and lead were found within the range of naturally occurring concentrations.	No constituents attributable to PSC
OU-1	SD-38	arsenic, beryllium	Barium, copper, lead, and zinc	Barium, copper, zinc and lead were found within the range of naturally occurring concentrations.	No constituents attributable to PSC
OU-1	SS-42	Benzene, ethylbenzene, toluene, xylene	PCE, DCP, TRPH, TPH, BTEX, arsenic, barium, boron, chromium, copper, lead, nickel, selenium, and zinc. (non-detect for past two years).	DCP is a common laboratory contaminant. TRPH, TPH, and BTEX have been detected inconsistently at the site. The latest sampling results did not contain hydrocarbons. Arsenic, barium, boron, chromium, copper, nickel, selenium, zinc and lead were found within the range of naturally occurring concentrations.	TPH and BTEX were attributable to PSC. Remediation system has been effective in reducing constituents to residual concentrations that will not impact groundwater. Five year monitoring implemented.

5.2.1 PSC RW-02

PSC RW-02 is located approximately 2 miles east of Luke AFB at the WWTP. Three monitoring wells, MW-115, MW-116, and MW-124, were installed at this site. Monitoring wells MW-115 and MW-116 were installed in 1992, and monitoring well MW-124 was installed in 1994. The screened interval has not been submerged during any of the sampling events. Monitoring well MW-115 was abandoned in February 1996. Monitoring wells MW-116 and MW-124 are still located on the site. Based on 1995 water level measurements, groundwater flow is towards the west.

Concentrations of naturally occurring constituents were compared with site-specific background concentrations as well as with regional background (within Maricopa County). These two comparisons were performed to ensure that the background was well characterized. Arsenic, chromium, copper, nickel, and lead were detected within the range of naturally occurring concentrations. BEP was detected in three samples during the May 1994 sampling event, however, the results were qualified as non-detect because BEP was also detected in the method blank. All VOC and BNA analyses of groundwater samples collected prior to August 1995, were analyzed by ATI Phoenix laboratory and do not satisfy data validation requirements for quantitative data. However, eight validated samples collected from August 1995 to 1998 were non-detect.

The most recent sampling event at the site was in May 1998. A sample was collected from monitoring well MW-124. No analytes were detected above laboratory detection limits. Samples have not been collected from monitoring well MW-116 since June 1996 and from MW-115 since February 1996. MW-115 is abandoned. No analytes were detected above laboratory detection limits during these sampling events. MW-124 will be monitored as part of the next five-year review period.

5.2.2 PSC DP-05

PSC DP-05 is located on the southwestern portion of Luke AFB, near Taxiway I. Three monitoring wells, MW-104, MW-105, and MW-106, were installed at this site in 1986. The screened intervals on these wells have been submerged since or right after installation. Based on 1995 water level measurements, groundwater flow is towards the southwest.

Barium and lead were detected within the range of naturally occurring concentrations, as defined by the site-specific and countywide background determinations, described

above. Toluene and acetone were detected in one sample from monitoring well MW-104 and BEP was detected in one sample from monitoring well MW-106 during the third quarter 1992 sampling event. These contaminants were not reported at detectable concentrations in any other sampling events. All VOC and BNA analyses of groundwater samples collected prior to August 1995, were analyzed by ATI Phoenix laboratory and do not satisfy data validation requirements for quantitative data. However, validated samples for five sample events conducted from October 1995 to May 1998 were non-detect.

The most recent sampling event at the site was in May 1998, when a sample was collected from monitoring well MW-105. No analytes were detected above laboratory detection limits. Samples have not been collected from monitoring wells MW-104 and MW-106 since June 1996. No analytes were detected above laboratory detection limits during this sampling event. All three wells have been abandoned.

5.2.3 PSC FT-06

PSC FT-06 is located on the southern portion of Luke AFB. Two monitoring wells, MW-107 and MW-108, were installed at this site in 1986. The screened intervals on these wells have been submerged since installation. Both wells are still located on the site. Based on 1995 water level measurements, groundwater flow is towards the west-southwest.

Chromium, selenium, and lead were detected within the range of naturally occurring concentrations, as defined by the site-specific and countywide background determinations, described above. DBCP, an agricultural contaminant, was detected in samples collected during the fourth quarter 1992 sampling event in concentrations, below the 2000 PRGs, ranging from 0.02 mg/L through 0.05 mg/L. The 2000 PRGs for DBCP are 0.45 mg/L for residential land use and 4.0 mg/L for industrial land use. This was the only sampling event in which DBCP was analyzed. These detections have been attributed to agricultural fields located up-gradient of the site. The most recent sampling events at the site were conducted in 1996. Samples were collected from monitoring wells MW-107 and MW-108. No analytes were detected above laboratory detection limits. MW-107 has been abandoned. MW-108 is scheduled to be abandoned.

5.2.4 PSC FT-07

PSC FT-07 is located on the northern portion of Luke AFB. Five monitoring wells, MW-109, MW-110, MW-111, MW-118, and MW-123, were installed at this site. Monitoring wells MW-109, MW-110, and MW-111 were installed in 1986, monitoring well MW-118 was installed in 1993, and monitoring well MW-123 was installed in 1994. The screened intervals in monitoring wells MW-109, MW-110, and MW-111 have been submerged since installation. Monitoring wells MW-110 and MW-111 were abandoned in 1996. Monitoring wells MW-109, MW-118, and MW-111 were abandoned in 1995 water level measurements, groundwater flow is towards the southwest.

Barium, chromium, copper, zinc, and lead were detected within the range of naturally occurring concentrations, as defined by the site-specific and countywide background determinations, described above. Chloroform was detected consistently in monitoring wells MW-110 and MW-123 through the first quarter of 1995. It was again detected in monitoring well MW-123 in 1997 and 1998. The concentrations were well below the MCL and AWQS. Toluene was detected in monitoring well MW-110 during the fourth quarter 1993 and first quarter 1993 sampling events. Toluene was not reported at detectable concentrations in any other sampling events. DCA and DBCP were detected in samples collected during the third quarter of 1992. These analytes were not reported at detectable concentrations in any other sampling events. BEP was detected in the duplicate sample collected from monitoring well MW-110 and the primary sample from MW-111 during the June 1993 sampling event. The presence of BEP appears to be anomalous because the primary/duplicate pair did not contain BEP. All VOC and BNA analyses of groundwater samples collected prior to August 1995, were analyzed by ATI Phoenix laboratory and do not satisfy data validation requirements for quantitative data. However, samples from 14 sampling events conducted from August 1995 through May 1999 were non-detect for all but three events. DBCP, an agricultural contaminant, was detected in samples collected during the fourth quarter 1992 sampling event. This was the only sampling event in which DBCP was analyzed. These detections have been attributed to agricultural fields located up-gradient of the site. Samples from three sample events conducted from October 1995 through June 1996 were non-detect.

The most recent sampling event at the site was in May 1999. A sample was collected from monitoring well MW-123. No analytes were detected above laboratory detection limits. Methylene chloride was detected during the November 1998 sampling event but was qualified because it was also detected in the blank. Monitoring well MW-118

was sampled in May 1998. No analytes were detected above laboratory detection limits. Samples have not been collected from monitoring wells MW-109, MW-110, and MW-111 since 1996. No analytes were detected above laboratory detection limits during this sampling event. MW-118 and MW-123 will be monitored as part of the next five-year review period.

5.2.5 PSC ST-18

PSC ST-18 is located on the southern portion of Luke AFB. Six monitoring wells, MW-2, MW-3, MW-4, MW-5, MW-114, and MW-122, were installed at this site. Monitoring wells MW-2 and MW-3 were installed in 1985. Monitoring wells MW-4 and MW-5 were installed in 1987. Monitoring well MW-114 was installed in 1991, and monitoring well MW-122 was installed in 1994. The screened intervals in monitoring wells MW-2, MW-3, MW-4, and MW-5 have been submerged since or just after installation. The screened intervals in MW-114 and MW-122 have not been submerged. Monitoring well MW-2 was abandoned in October 1993. Monitoring well MW-4 was abandoned in October 1994. Monitoring wells MW-3, MW-5, MW-114, and MW-122 are still located on the site. Based on 1995 water level measurements, groundwater flow is towards the west-southwest.

Arsenic, barium, chromium, copper, nickel, selenium, zinc, and lead were detected within the range of naturally occurring concentrations, as defined by the site-specific and countywide background determinations, described above. BEP was detected in monitoring well MW-114 during the first quarter of 1992, and the second quarter of 1993. BEP was not reported at detectable concentrations in any other sampling events. All VOC and BNA analyses of groundwater samples collected prior to August 1995, were analyzed by ATI Phoenix laboratory and do not satisfy data validation requirements for quantitative data. However, samples collected from 13 sample events conducted from August 1995 to October 2001 were non-detect for all but two events. DBCP, an agricultural contaminant, was detected in a sample collected from monitoring well MW-3 during the fourth quarter 1992 sampling event. This was the only sampling event in which DBCP was analyzed. This detection has been attributed to agricultural fields located up-gradient of the site.

The most recent sampling event at the site was in October 2001. Samples were collected from monitoring wells MW-114 and MW-122. No analytes were detected above laboratory detection limits in MW-122. Trichloroethene, 1,1-dichloroethene and tetrachloroethene were above detection limits but well below applicable water quality standards in MW-114 Methylene chloride was detected in monitoring wells MW-114

and MW-122 during the November 1998 sampling event but was qualified as estimated because it was also detected in the blank. 2-Butanone was also detected in monitoring well MW-114 during this sampling event and was qualified as estimated. MW-114 and MW-122 will be monitored on an annual basis for the next five-year period to evaluate VOC trends, although, no trends are anticipated given the overall monitoring results for the period of record. Samples have not been collected from monitoring wells MW-3 and MW-5 since June 1996. No analytes were detected above laboratory detection limits during this sampling event.

5.2.6 PSC SD-20

PSC SD-20 is located on the southern portion of Luke AFB. Five monitoring wells, MW-102, MW-103, MW-112S, MW-112D, and MW-113, were installed at this site. Monitoring wells MW-102 and MW-103 were installed prior to 1991. Monitoring wells MW-112D, and MW-113 were installed in 1991. The screened intervals in monitoring wells MW-103 and MW-112D have been submerged the majority of the time since installation. The screened intervals in monitoring wells MW-103 and MW-112D have been submerged the majority of the time since installation. The screened intervals in monitoring wells MW-102, MW-112S and MW-113 have not been submerged the majority of the time since installation. Based on 1995 water level measurements, groundwater flow is towards the northwest.

Arsenic, barium, boron, chromium, copper, nickel, selenium, zinc, and lead were detected within the range of naturally occurring concentrations, as defined by the site-specific and countywide background determinations, described above. BEP was detected in two samples during the fourth quarter of 1991. BEP was not reported at detectable concentrations in any other sampling events. TCE and DCA were detected in monitoring well MW-113 consistently through the first quarter of 1995. TCE, DCA, PCE, and methylene chloride were detected in monitoring well MW-113 in November 1998. TCE was consistently detected in monitoring well MW-112S through the fourth quarter of 1993. TCE, DCA, PCE, acetone, and methylene chloride were detected in monitoring well MW-112S in November 1998. Since groundwater flow is towards the northwest, the source of these analytes may be attributable to an off-site source rather than the oil/water separator located northeast of the wells. The presence of fissures in this area further complicates the behavior of groundwater. Alternatively, it is possible TCE and other constituents that originated from points on-base that discharged to the oil water separator canal may have migrated to the groundwater. The fissures at the end of the canal may have provided a pathway for these constituents to reach groundwater. The institution of pollution prevention control measures at Luke AFB has eliminated any future potential sources of constituents in this regard. All VOC and BNA

groundwater samples collected prior to August 1995 were analyzed by ATI Phoenix laboratory and did not satisfy data validation requirements for quantitative data. However, samples from 16 sampling events conducted from August 1995 to May 1999 were non-detect for all but six events.

The most recent sampling event at the site was in May 1999. Samples were collected from monitoring wells MW-112S and MW-113. Bromodichloromethane and chloroform were detected in monitoring well MW-113. No other analytes were detected above laboratory detection limits. Monitoring well MW-112D was last sampled in May 1998. No analytes were detected above laboratory detection limits during this sampling event. Samples have not been collected from monitoring wells MW-102 and MW-103 since June 1996. No analytes were detected above laboratory detection limits during this sampling event. MW-102 and MW-103 have been abandoned. MW-112S, MW-112D and MW-113 will be monitored as part of the next five-year review period.

5.2.7 PSC SD-21

PSC SD-21 is located approximately two miles from Luke AFB south of the WWTP. One monitoring well, MW-101, was installed at this site in 1986. The screened interval has been submerged since installation. Monitoring well MW-101 is still located on the site. Based on 1995 water level measurements, groundwater flow is towards the west.

Arsenic, barium, boron, copper, zinc, and lead were detected within the range of naturally occurring concentrations, as defined by the site-specific and countywide background determinations, described above. BEP, acetone, and carbon disulfide was detected in one sample during the second quarter of 1994. These analytes were not reported at detectable concentrations in any other sampling events. All VOC and BNA analyses of groundwater samples collected prior to August 1995 were analyzed by ATI Phoenix laboratory and do not satisfy data validation requirements for quantitative data. However, samples from two sampling events conducted from August 1995 to June 1996 were non-detect.

The most recent sampling event at the site was in June 1996. No analytes were detected above laboratory detection limits during this sampling event.

5.2.8 PSC SD-38

PSC SD-38 is located on the eastern portion of Luke AFB. One monitoring well, MW-117, was installed at this site. The screened interval has not been submerged during any of the sampling events. Based on 1995 water level measurements, groundwater flow is towards the southwest.

Barium, copper, zinc, and lead were detected within the range of naturally occurring concentrations, as defined by the site-specific and countywide background determinations, described above. The most recent sampling event at the site was in June 1996. No analytes were detected above laboratory detection limits during this sampling event.

5.2.9 PSC SS-42

PSC SS-42 is located in the northeastern portion of Luke AFB. Four monitoring wells, MW-119, MW-120, MW-121, and MW-125, were installed at this site. Monitoring wells MW-119, MW-120, and MW-121 were installed in 1993. Monitoring well MW-125 was installed in 1995. The screened interval has not been submerged during any of the sampling events. Based on 1995 water level measurements, groundwater flow is towards the southwest.

Arsenic, barium, boron, chromium, copper, nickel, selenium, zinc, and lead were detected within the range of naturally occurring concentrations, as defined by the site-specific and countywide background determinations, described above. DCP, a common component of insecticides, was detected in several samples collected from monitoring wells MW-119, MW-120, and MW-121 between November 1993 and February 1995. DCP was again detected in 1997 and 1998 in all four monitoring wells on the site. This data was qualified as estimated. TPH was detected at the site beginning in the first quarter of 1995. BTEX was detected in monitoring well MW-121 in 1997 and 1998. Methylene chloride and PCE were detected in monitoring well MW-121 in 1998. Toluene and methylene chloride were detected in monitoring well MW-120 in November 1998. This data was qualified as estimated. Methylene chloride was detected in monitoring well MW-125 in November 1998. These data were qualified as estimated. All VOC and BNA analyses of groundwater samples collected prior to August 1995 were analyzed by ATI Phoenix laboratory and do not satisfy data validation requirements for quantitative data. However, samples from 19 sampling events conducted from August 1995 to November 2001 were non-detect for all but five events.

The most recent sampling events at the site were in August and November 2001. Samples were collected from monitoring well MW-121 and a replacement to well MW-125 designated, as MW-125R MW-125 had to be replaced because the casing had collapsed. The well that collapsed had steel casing. The deterioration of the steel casing is attributed to the reactivity of the steel with the underlying geologic materials. No analytes were detected above laboratory detection limits. Monitoring well MW-119 was last sampled in July 1997. DCP was detected above laboratory detection limits during this sampling event. Monitoring well MW-120 was last sampled in November 1998. DRO, DCP, methylene chloride, and toluene were detected above laboratory detection limits during this sampling event. MW-119 and MW-120 have collapsed due to corrosion and are scheduled to be abandoned. MW-121 and MW-125 are still active and will be monitored as part of the next five-year review.

6.0 Investigative Site History

This section of the five-year review report provides a historical and five year review process summary for OU-1 and OU-2 PSCs for which remedies were selected. The five-year review process primarily consisted of a site inspection, interviews and a review of relevant documents and data. The five-year review for the site was led by Jeff Rothrock of Luke AFB. The following team members assisted in the review:

- Jon Sherrill, ARCADIS G&M, Inc.
- Kent Lang, ARCADIS G&M, Inc.
- Stephanie Armijo, ARCADIS G&M, Inc.
- Nichole Cherry, ARCADIS G&M, Inc.
- Monique Ostemann, USACE
- Greg Mellema, USACE
- Dan Stralka, USEPA
- Nancy Lou Minkler, ADEQ

6.1 Five Year Review Process Summary

The five-year review process includes the following primary elements:

- Remedy selection and implementation is reviewed and summarized for each OU-1 and OU-2 PSC for which a remedy was selected.
- Changes in standards were evaluated with respect to the continued effectiveness of the remedies that were implemented based on cancer risks and

non-cancer hazards for applicable COCs for Base worker or excavation worker scenarios.

- Groundwater monitoring results are compared to groundwater standards established for the project.
- Representatives of Luke AFB, USEPA, US Army Corps. of Engineers, ADEQ and ARCADIS G&M performed a site inspection of each PSC for which a remedy was selected on May 22, 2001.
- The results of interviews with individuals knowledgeable about the project.

6.2 Review of OU-1 PSCs

OU-1 PSCs for which there was no action taken (refer to discussion in Section 1.1.3 and information in Table 3-2) include the following:

- OT-01
- OT-08
- OT-09
- OT-10
- SS-15
- SS-16
- SS-19
- SD-21
- DP-24
- SD-26
- LF-37
- SD-39
- OT-41

A comparison of the EPC for a given COC in the combined surface and subsurface soil, with USEPA Region IX industrial PRGS (1996 and 2000) and ADEQ residential and non-residential SRLs is in Appendix C. EPCs were taken from the 95 percent upper confidence limit (one-tailed) on the mean assuming a normal distribution unless indicated otherwise.

OU-1 PSCs for which remedies were selected based on the results of risk assessment include the following:

- RW-02
- LF-03

- FT-07
- DP-13
- LF-14
- LF-25
- SD-38
- SS-42

For OU-1 PSCs, PRGs were not established. Alternatively, PSC specific cancer risks and non-cancer hazards were calculated using 1996 USEPA Region IX PRG guidance to develop a site-specific industrial scenario. To evaluate changes in standards as part of this five-year review, cancer risk and non-cancer hazards were recalculated using 2000 USEPA Region IX industrial PRGs and post remediation exposure point concentrations for base worker and excavation worker scenarios as applicable (USEPA Region IX PRGS for 1996 and 2000 are in Appendix D). EPCs were taken from the 95 percent upper confidence limit (one-tailed) on the mean assuming a normal distribution, or from the maximum concentration for PSCs with post-remediation data in surface soils for the base worker and in all soils collected to a depth of 16 feet bgs for excavation workers. ADEQ SRLs were also reviewed since they were used to determine risk under a residential land use scenario.

Changes in standards are evaluated with respect to the continued effectiveness of the remedies that were implemented based on a non-cancer HI less than or equal to 1.0 or an ELCR greater than the risk range of 1×10^{-6} to 1×10^{-4} .

6.2.1 PSC RW-02 Wastewater Treatment Annex

Remedial Actions

Remedy Selection

As stated in the OU-1 ROD²⁷, the remedy selected for PSC RW-02 consisted of institutional controls listed as follows:

²⁷ ARCADIS Geraghty & Miller, Inc., 1999. Final Record of Decision, Operable Unit 1, Luke AFB, Arizona, January 20, 1999.

- Land use restrictions consisting of a VEMUR and constraints within the Base General Plan to limit future development and residential use at the site.
- Geophysical monitoring conducted annually for 30 years to ensure safety of potential receptors and to provide a warning mechanism in case of a change in subsurface conditions.
- Perimeter fencing installed around the low-level waste containment structure to provide a physical barrier.

Remedy Implementation

The establishment of PSC RW-02 was part of overall remediation of an associated landfill and bank stabilization for the Agua Fria River. The implementation of the remedy for PSC RW-02 is summarized as follows:

- A VEMUR was placed on the radiological waste portion of the site on June 15, 2000, to restrict residential development on the site.
- The BGP was revised on January 5, 2000, to place restraints on the residential development of the site.
- An ICP^{28} was prepared and implemented as part of the BGP to facilitate training and education of all personnel involved with the implementation and enforcement of the required institutional controls.
- The ICP details the objectives and rationales for establishing institutional controls and described the procedures that were implemented to ensure the required institutional controls are enforced.
- The ICP includes provisions for annual reviews and updates of the BGP, thus ensuring regular checks and balances are in place into the foreseeable future.
- Four monitoring points were installed to a depth of 20 feet in December 1999. According to the monitoring plan²⁹, radiological logging will be conducted on an annual basis at the three monitoring points for a period of 30 years. The third annual radiological monitoring event was conducted on August 8, 2001.

²⁸ ARCADIS Geraghty & Miller, Inc., 2000. Institutional Control Plan, Luke Air Force Base, Arizona, December 15, 2000.

²⁹ ARCADIS Geraghty & Miller, Inc., 2000. Long Term Radiological Monitoring Plan. November 14, 2000.

The background levels for soil in the area were measured at 11,558 through 19,618 counts per minute (cpm). The radiological levels for all four measuring points were between 10,310 and 20,434 cpm. These results are similar to background levels, indicating that the soil surrounding the buried radiological waste has not been impacted and the radiological waste containment structure remains protective.

System Operation and Maintenance

No operation and maintenance was required for the remedy selected. The cost of the remediation at PSC RW-02 to date has been \$23,560. The annual monitoring is expected to cost \$2,000.

Progress Since the Last Five-year Review

This is the first five-year review conducted for the site.

Five-year Review Process

Five-year Review Findings

Site Inspection

No change in land use had occurred since implementation of the remedy for the site. Stabilization work conducted on the Aqua Fria River as part of remediation of a former landfill at the site was also inspected and discussed. Photographs of PSC RW-02 taken during the site inspection are in Appendix E.

Changes to Standards

A comparison of exposure point concentrations (taken to be the maximum detected concentration within the soil profile considered) in surface soils and USEPA Region IX PRGs for a base worker and soils to a depth of 16 feet bgs and USEPA Region IX PRGs for an excavation worker are in Tables 6-1 and 6-2, respectively. The comparison indicates that exposure to constituents detected at PSC RW-02 is still within the acceptable risk range.

Changes in Exposure Pathways, Toxicity and Other Contaminant Characteristics

There were no changes in exposure pathways, toxicity, or other contaminant characteristics, with the exception of beryllium. Beryllium was reclassified by USEPA for risk assessments as non-carcinogenic based on the types of exposure. This difference can be seen in a comparison of the 1996 PRGs and the 2000 PRGs.

Data Evaluation

TRPH was detected to a depth of ten feet in the test pit with the highest concentration at 4,100 mg/kg. TRPH was detected in all eight-soil borings advance in 1993. Radiochemical results indicated that concentrations were consistent with natural soils. Additional soil borings were advance in 1996. BNAs were detected to a depth of 16 feet. The risk assessment conducted for the site concluded that the most conservative ELCR and HI were 2 x 10⁻⁷ and 0.6, respectively³⁰. Based on the results of the risk assessment, COPCs identified at PSC RW-02 were not present in concentrations high enough to cause adverse health effects under industrial or residential land use. However, the decision to use institutional controls was based on the presence of the low-level radioactive waste containment structure limits potential future land usage.

Assessment

Is the remedy functioning as intended by decision documents?

The objective of the institutional controls was to prevent exposure to radioactive material and to prevent residential development at the site. The institutional controls have functioned as intended.

Are the assumptions used at the time of the remedy selection still valid?

The assumptions used in performing the risk assessment at the site are still valid.

³⁰ Geraghty & Miller, 1997. Final Remedial Investigation Report OU-1, Appendix B – Baseline Base wide Risk Assessment, Luke Air Force Base, Arizona. Volumes 1 and 2. October 1, 1997.

Table 6-1

REASONABLE MAXIMUM SOIL EXPOSURE POINT CONCENTRATIONS IN SOIL AND USEPA REGION IX PRGS FOR SURFACE SOIL (BASE WORKER) PSC RW-02 Luke Air Force Base, Arizona

	Exposure Point Concentration	USEPA I Industrial P	Region IX RG (mg/kg)	*Risk or	ADE (m	EQ SRL g/kg)
Constituent	(mg/kg)	1996	2000	Hazard	Residential	Non-Residential
Cancer Effects				ELCR		
<u>BNAs</u> Benzo(a)pyrene	0.10	0.26	0.29	3.4E-07	0.61	2.6
<u>Metals</u> Arsenic	5.3	2.4	2.7	2.0E-06	10	10
Total Site Risk or Hazard				2E-06		
Non-Cancer Effects				HQ		
BNAs						
Benzo(a)pyrene [a]	0.10	800	190	5.3E-04	0.61	2.6
TRPHs [b]	330	NA	NA	NA	4,100	18,000
<u>Metals</u>						
Arsenic	5.3	2.4	2.7	2.0E+00	10	10
Hazard Index				2		

All Units in milligrams per kilogram (mg/kg).

ADEQ SRL	Arizona Department of Environmental Quality 2000 Soil Remediation Level.
BNAs	Base-neutral and extractable acids.
ELCR	Excess lifetime cancer risk.
HQ	Hazard quotient.
NA	Not available.
PRG	Preliminary Remediation Goal.
TRPHs	Total recoverable petroleum hydrocarbons.
USEPA	U.S. Environmental Protection Agency.

[a] The PRG is based on naphthalene non-carcinogenic effects.

[b] n-Hexane was used as a surrogate in the risk assessment for toxicity. A PRG for TRPHs is not provided. *Risk and Hazards are based upon 2000 Industrial PRGs

Surface soil samples were taken from a depth of 0 to 2 feet below ground surface. Subsurface soil samples were taken from a depth of 2 to 16 feet below ground surface. Exposure point concentrations were taken from the 95 percent upper confidence limit (one-tailed) on the mean, assuming a normal distribution.

Arsenic background levels in the State of Arizona regularly exceed the PRGs. Arsenic is present at this site within background levels, and when taken out of the equation the ELCR and hazard index are within acceptable risk ranges.

Table 6-2

REASONABLE MAXIMUM EXPOSURE POINT CONCENTRATIONS IN SOIL AND USEPA REGION IX PRGS FOR SURFACE AND SUBSURFACE (EXCAVATION WORKER) PSC RW-02 Luke Air Force Base, Arizona

	Exposure Point Concentration	USEPA Region IX Industrial PRG (mg/kg)		*Risk or	ADEQ SRL (mg/kg)	
Constituent	(mg/kg)	1996	2000	Hazard	Residential	Non-Residential
Cancer Effects				ELCR	-	
BNAs						
Benzo(a)pyrene	0.10	0.26	0.29	3.4E-07	0.61	2.6
Metals						
Arsenic	6.0	2.4	2.7	2.2E-06	10	10
Beryllium	0.27	1.1	2,200	1.2E-10	1.4	11
Cadmium [a]	5.0	850	3,000	1.7E-09	38	850
Total Site Risk				3E-06		
Non-Cancer Effects				HQ		
BNAs						
Benzo(a)pyrene [b]	0.10	800	190	5.3E-04	0.61	2.6
TRPHs [c]	530	NA	NA	NA	4,100	18,000
Metals						
Arsenic	6.0	NA	440	1.4E-02	10	10
Beryllium	0.27	NA	3,700	7.3E-05	1.4	11
Cadmium	5.0	850	810	6.2E-03	38	850
Copper	370	63,000	76,000	4.9E-03	2,800	63,000
Hazard Index				0.03		

All Units in milligrams per kilogram (mg/kg).

ADEQ SRL	Arizona Department of Environmental Quality 2000 Soil Remediation Level.
BNAs	Base-neutral and extractable acids.
ELCR	Excess lifetime cancer risk.
HQ	Hazard quotient.
NA	Not available.
PRG	Preliminary Remediation Goal.
TRPHs	Total recoverable petroleum hydrocarbons.
USEPA	U.S. Environmental Protection Agency.

[a] PRG is based on non-carcinogenic effects.

[b] The PRG is based on naphthalene non-carcinogenic effects.

[c] n-Hexane was used as a surrogate in the risk assessment for toxicity. A PRG for TRPHs is not provided.

*Risk and Hazards are based upon 2000 Industrial PRGs

Surface soil samples were taken from a depth of 0 to 2 feet below ground surface.

Subsurface soil samples were taken from a depth of 2 to 16 feet below ground surface.

Exposure point concentrations were taken from the 95 percent upper confidence limit (one-tailed) on the mean, assuming a normal distribution.

Arsenic background levels in the State of Arizona regularly exceed the PRGs. Arsenic is present at this site within background levels, and when taken out of the equation the ELCR and hazard index are within acceptable risk ranges.

Has any other information come up that could question the protectiveness of the remedy?

No additional information has come to light that could call into question the protectiveness of the remedy.

Issues

No issues were discovered during this five-year review.

Recommendations and Follow-up Activities

MW-124 will be monitored as part of future five-year reviews. No other recommendations or follow-up activities are suggested at this time.

Protectiveness Statement

The remedy at PSC RW-02 is protective of human health and the environment and exposure pathways that could result in unacceptable risks are being controlled.

6.2.2 PSC LF-03 Outboard Runway Landfill

Remedial Actions

Remedy Selection

As stated in the OU-2 ROD, the remedial action selected for PSC LF-03 consisted of institutional controls list as follows:

• Land use restrictions consisting of a VEMUR and constraints within the Base General Plan to limit future development and residential use at the site.

Remedy Implementation

The implementation of the remedy for PSC LF-03 is summarized as follows:

• A VEMUR was placed on the site on June 15, 2000, to restrict residential development on the site.

- The BGP was revised on January 5, 2000, to place restraints on the residential development of the site.
- The ICP was incorporated as part of the BGP to facilitate training and education of all personnel involved with the implementation and enforcement of the required institutional controls.
- The ICP details the objectives and rationales for establishing institutional controls and describes the procedures that were implemented to ensure the required institutional controls are enforced.
- The ICP includes provisions for annual reviews and updates of the BGP, thus ensuring regular checks and balances are in place into the foreseeable future.

System Operation and Maintenance

No operation and maintenance was required for the remediation selected. The cost of the institutional controls implemented at PSC LF-03 was \$347.

Progress Since the Last Five-year Review

This is the first five-year review conducted for the site.

Five-year Review Process

Five-year Review Findings

Site Inspection

No changes in land use since implementation of the remedy for the site were observed during the inspection. Photographs of PSC LF-03 taken during the site inspection are in Appendix E.

Changes to Standards

A comparison of exposure point concentrations in surface soils and USEPA Region IX PRGs for a base worker and surface and sub-surface soils and USEPA Region IX PRGs for an excavation worker are in Tables 6-3 and 6-4, respectively. The comparison indicates PSC LF-03 is still with in the acceptable risk range.
REASONABLE MAXIMUM EXPOSURE POINT CONCENTRATIONS IN SOIL AND USEPA REGION IX PRGS FOR SURFACE SOIL (BASE WORKER) PSC LF-03

Luke Air Force Base, Arizo	na
----------------------------	----

	Exposure Point Concentration	USEPA Region IX Industrial PRG (mg/kg)		*Risk or	ADEQ SRL (mg/kg)	
Constituent	(mg/kg)	1996	2000	Hazard	Residential	Non-Residential
Cancer Effects				ELCR	-	
Metals						
Arsenic	4.8	2.4	2.7	1.8E-06	10	10
Beryllium	0.7	1.1	2,200	3.2E-10	1.4	11
Total Site Risk				1.8E-06		
Non-Cancer Effects				HQ		
Metals						
Arsenic	4.8	NA	440	1.1E-02	10	10
Beryllium	0.7	NA	3,700	1.9E-04	1.4	11
Hazard Index				0.01		

All Units in milligrams per kilogram (mg/kg).

Arizona Department of Environmental Quality 2000 Soil Remediation Level.
Base-neutral and extractable acids.
Excess lifetime cancer risk.
Hazard quotient.
Preliminary Remediation Goal.
U.S. Environmental Protection Agency.

*Risk and Hazards are based upon 2000 Industrial PRGs

Surface soil samples were taken from a depth of 0 to 2 feet below ground surface.

Subsurface soil samples were taken from a depth of 2 to 16 feet below ground surface.

Exposure point concentrations were taken from the 95 percent upper confidence limit (one-tailed) on the mean, assuming a normal distribution.

Arsenic background levels in the State of Arizona regularly exceed the PRGs. Arsenic is present at this site within background levels, and when taken out of the equation the ELCR and hazard index are within acceptable risk ranges.

REASONABLE MAXIMUM EXPOSURE POINT CONCENTRATIONS IN SOIL AND USEPA REGION IX PRGS FOR SURFACE AND SUBSURFACE SOIL (EXCAVATION WORKER) PSC LF-03 Luke Air Force Base, Arizona

	Exposure Point Concentration	USEPA Region IX Industrial PRG (mg/kg)		*Risk or	ADEQ SRL (mg/kg)	
Constituent	(mg/kg)	1996	2000	Hazard	Residential	Non-Residential
Cancer Effects				ELCR		
<u>Metals</u> Arsenic Beryllium Chromium [a]	6.9 0.42 140	2.4 1.1 450	2.7 2,200 450	2.6E-06 1.9E-10 3.1E-07	10 1.4 2,100	10 11 4,500
Total Site Risk				2.9E-06		
Non-Cancer Effects				HQ		
<u>Metals</u> Arsenic Beryllium Chromium [b] Copper	6.9 0.42 140 1,100	NA NA NA 63,000	440 3,700 6,100 76,000	1.6E-02 1.1E-04 2.3E-02 1.4E-02	10 1.4 2,100 2,800	10 11 4,500 63,000
Hazard Index				0.1		

All Units in milligrams per kilogram (mg/kg).

ADEQ SRL	Arizona Department of Environmental Quality 2000 Soil Remediation Level.
BNAs	Base-neutral and extractable acids.
ELCR	Excess lifetime cancer risk.
HQ	Hazard quotient.
PRG	Preliminary Remediation Goal.
USEPA	U.S. Environmental Protection Agency.

[a] The chromium PRG is based on total chromium carcinogenic effects and a 1:6 ratio of chromium IV to chromium III.[b] The chromium PRG is based on chromium VI non-carcinogenic effects.

*Risk and Hazards are based upon 2000 Industrial PRGs

Surface soil samples were taken from a depth of 0 to 2 feet below ground surface. Subsurface soil samples were taken from a depth of 2 to 16 feet below ground surface. Exposure point concentrations were taken from the 95 percent upper confidence limit (one-tailed) on the mean, assuming a normal distribution.

Arsenic background levels in the State of Arizona regularly exceed the PRGs. Arsenic is present at this site within background levels, and when taken out of the equation the ELCR and hazard index are within acceptable risk ranges.

Changes in Exposure Pathways, Toxicity and Other Contaminant Characteristics

There were no changes in exposure pathways, toxicity, or other contaminant characteristics.

Data Evaluation

The determination to use institutional controls was based on the ELCR for chromium concentrations at the site of 5 x 10^{-6} , above the allowable residential risk of 1 x 10^{-6} . The elevated ELCR for chromium was caused by two samples with high chromium concentrations collected at test pit TP-5. The risk assessment used the conservative assumption that all of the chromium was in the hexavalent state.

Assessment

Is the remedy functioning as intended by decision documents?

The objective of the institutional controls was to prevent residential development at the site. The institutional controls have functioned as intended.

Are the assumptions used at the time of the remedy selection still valid?

The assumptions used in performing the risk assessment at the site are still valid.

Has any other information come up that could question the protectiveness of the remedy?

No additional information has come to light that could call into question the protectiveness of the remedy.

Issues

No issues were discovered during this five-year review.

Recommendations and Follow-up Activities

There are no further recommendations or follow-up activities suggested at this time.

Protectiveness Statement

The remedy at PSC LF-03 is protective of human health and the environment and exposure pathways that could result in unacceptable risks are being controlled.

6.2.3 PSC FT-07E Eastern Portion of North Fire Training Area

Remedial Actions

Remedy Selection

A SVE system was installed in 1992 at a cost of \$395,000 and was done independent of the OU-I ROD. The system was operational from April 1992 through December 1992 and approximately 14,000 pounds of contaminants were removed. During the RI, an investigation was conducted to determine the effectiveness of the removal. The conclusions of the investigation were that the SVE effectively removed contaminants greater than 16 feet bgs. However, high levels of contaminants still remained in the shallow soils. As stated in the OU-1 ROD, and based on the risk assessment for the shallow soils, the remedial action selected for PSC FT-07E consisted of institutional controls. Institutional controls at PSC FT-07E consisted of the following:

• Land use restrictions consisting of a VEMUR and constraints within the Base General Plan to limit future development and residential use at the site.

Remedy Implementation

The implementation of the remedy for PSC FT-07E is summarized as follows:

- A VEMUR was placed on the site on June 15, 2000, to restrict residential development on the site.
- The BGP was revised on January 5, 2000, to place restraints on the residential development of the site.
- The ICP was incorporated as part of the BGP to facilitate training and education of all personnel involved with the implementation and enforcement of the required institutional controls.

- The ICP details the objectives and rationales for establishing institutional controls and describes the procedures that were implemented to ensure the required institutional controls are enforced.
- The ICP includes provisions for annual reviews and updates of the BGP, thus ensuring regular checks and balances are in place into the foreseeable future.

System Operation and Maintenance

No operation and maintenance was required for the remediation selected. The cost of the institutional controls implemented at PSC FT-07E was \$347.

Progress Since the Last Five-year Review

This is the first five-year review conducted for the site.

Five-year Review Process

Five-year Review Findings

Site Inspection

There were no changes in land use at the site since implementation of the remedy was observed. Photographs of PSC FT-07E taken as part of the site inspection are in Appendix E.

Changes to Standards

A comparison of exposure point concentrations in surface soils and USEPA Region IX PRGs for a base worker and surface and sub-surface soils and USEPA Region IX PRGs for an excavation worker are in Tables 6-5 and 6-6, respectively. The comparison indicates that PSC FT-07E is still within the acceptable risk range.

Changes in Exposure Pathways, Toxicity and Other Contaminant Characteristics

There were no changes in exposure pathways, toxicity, or other contaminant characteristics.

Table 6-5

REASONABLE MAXIMUM EXPOSURE POINT CONCENTRATIONS IN SOIL AND USEPA REGION IX PRGS FOR SURFACE AND SUBSURFACE SOIL (EXCAVATION WORKER) PSC FT-07 Luke Air Force Base, Arizona

	Exposure Point Concentration	USEPA Region IX Industrial PRG (mg/kg)		*Risk or	ADEQ SRL (mg/kg)	
Constituent	(mg/kg)	1996	2000	Hazard	Residential	Non-Residential
Cancer Effects				ELCR		
<u>Metal</u> Arsenic	5.2	2.4	2.7	1.9E-06	10	10
Total Site Risk				1.9E-06		
Non-Cancer Effects				HQ		
TRPH [a]	7,500	NA	NA	NA	4,100	18,000
<u>Metal</u> Arsenic	5.2	NA	440	1.2E-02	10	10
Hazard Index				0.01		

All Units in milligrams per kilogram (mg/kg).

[a] n-Hexane was used as a surrogate in the risk assessment for toxicity. A PRG for TRPHs is not provided. *Risk and Hazards are based upon 2000 Industrial PRGs

Surface soil samples were taken from a depth of 0 to 2 feet below ground surface. Subsurface soil samples were taken from a depth of 2 to 16 feet below ground surface. Exposure point concentrations were taken from the 95 percent upper confidence limit (one-tailed) on the mean, assuming a normal distribution.

Arsenic background levels in the State of Arizona regularly exceed the PRGs. Arsenic is present at this site within background levels, and when taken out of the equation the ELCR and hazard index are within acceptable risk ranges.

Table 6-6

REASONABLE MAXIMUM EXPOSURE POINT CONCENTRATIONS IN SOIL AND USEPA REGION IX PRGS FOR SURFACE SOIL (BASE WORKER) PSC FT-07 Luke Air Force Base, Arizona

USEPA Region IX ADEQ SRL **Exposure** Point Industrial PRG (mg/kg) (mg/kg) Concentration *Risk or 1996 2000 Constituent (mg/kg) Hazard Residential Non-Residential Cancer Effects ELCR Metal 7.9 2.4 2.7 2.9E-06 10 10 Arsenic **Total Site Risk** 2.9E-06 Non-Cancer Effects HQ TRPH [a] 18,000 1,600 NA NA NA 4,100 Metal Arsenic 7.9 440 10 10 NA 1.8E-02 Hazard Index 0.02

All Units in milligrams per kilogram (mg/kg).

ADEO SDI	Arizona Donartment of Environmental 2000 Quality Soil Remediation Level
ADEQ SKL	Arizona Department of Environmental 2000 Quarty Son Remediation Level.
ELCR	Excess lifetime cancer risk.
Н	Hazard quotient.
NA	Not applicable.
PRG	Preliminary Remediation Goal.
R	Excess lifetime cancer risk.
TRPH	Total recoverable petroleum hydrocarbons.
USEPA	U.S. Environmental Protection Agency.
VOCs	Volatile organic compounds.

[a] n-Hexane was used as a surrogate in the risk assessment for toxicity. A PRG for TRPHs is not provided. *Risk and Hazards are based upon 2000 Industrial PRGs

Surface soil samples were taken from a depth of 0 to 2 feet below ground surface. Subsurface soil samples were taken from a depth of 2 to 16 feet below ground surface. Exposure point concentrations were taken from the 95 percent upper confidence limit (one-tailed) on the mean, assuming a normal distribution.

Arsenic background levels in the State of Arizona regularly exceed the PRGs. Arsenic is present at this site within background levels, and when taken out of the equation the ELCR and hazard index are within acceptable risk ranges.

Data Evaluation

The determination to have no remedial action at the site was based on the results of soil sampling conducted as part of the RI. Soil samples collected in 1991 had concentrations of TRPH ranging up to 3,800 mg/kg. Lead was detected above the background UTL. The highest concentration was 172 mg/kg. The risk assessment conducted for the site concluded that the most conservative ELCR and HI were 4 x 10^{-8} and 0.0002, respectively. The vadose zone transport model also indicated that the COCs would not migrate to and impact groundwater. Due to TPH concentrations, residential land use is was restricted through a VEMUR.

Assessment

Is the remedy functioning as intended by decision documents?

The remedial action is functioning as intended.

Are the assumptions used at the time of the remedy selection still valid?

The assumptions used in performing the risk assessment at the site are still valid.

Has any other information come up that could question the protectiveness of the remedy?

No additional information has come to light that could call into question the protectiveness of the remedy.

Issues

No issues were discovered during this five-year review.

Recommendations and Follow-up Activities

MW-118 and MW-123 will be monitored as part of future five-year reviews. No other recommendations or follow-up activities are suggested at this time.

Protectiveness Statement

The remedy at PSC FT-07E is protective of human health and the environment and exposure pathways that could result in unacceptable risks are being controlled.

6.2.4 DP-13 Drainage Ditch Disposal Area

Remedial Actions

Remedy Selection

As stated in the ROD for OU-1, institutional controls were the selected remedy for PSC DP-13. Institutional controls implemented at PSC DP-13 consisted of the following:

- A VEMUR to restrict land use to non-residential.
- The BGP modified to place constraints on future residential development.
- Work practices requiring the use of Personal Protective Equipment (PPE) while excavating the site.
- An ICP to document required institutional controls.

Remedy Implementation

The implementation of the remedy for PSC DP-13 is summarized as follows:

- A VEMUR was implemented at site on June 15, 2000, to restrict residential development. The BGP was revised on January 5, 2000, to restrict residential development and to require the use of PPE by workers in the event soils are excavated at the site.
- An Institutional Control Plan was implemented on December 15, 2000, which was designed to facilitate training and education of all personnel involved with the implementation and enforcement of the required institutional controls.
- The ICP included provisions for annual reviews and updates of the BGP, thus ensuring regular checks and balances are in place into the foreseeable future.

System Operation and Maintenance

No operation and maintenance was required for the selected remedy. The cost of the institutional controls implemented at PSC DP-13 was \$347.

Progress Since the Last Five-year Review

This is the first five-year review conducted for the site.

Five-year Review Process

Five-year Review Findings

Site Inspection

During the inspection, the feasibility of removing landfill materials at some future date was raised as an issue. There were no other comments. No changes in land use had occurred since implementation of the remedy for the site. Photographs of PSC DP-13 taken as part of the inspection are in Appendix E.

Changes to Standards

A comparison of exposure point concentrations in surface soils and USEPA Region IX PRGs for a base worker and surface and sub-surface soils and USEPA Region IX PRGs for an excavation worker are in Tables 6-7 and 6-8, respectively. The comparison indicates that PSC DP-13 is still within the acceptable risk range.

Changes in Exposure Pathways, Toxicity and Other Contaminant Characteristics

There were no changes in exposure pathways, toxicity, or other contaminant characteristics.

Data Evaluation

The determination to use institutional controls was based on the ELCR at the site of 3 x 10^{-5} , which is greater than the allowable residential risk of 1 x 10^{-6} , and the HI of 2, which is greater than the allowable residential risk of 1. The risk assessment assumed that all of the chromium was in the hexavalent state. Mean blood lead levels for sensitive populations that included children up to seven years old, were calculated using the IEUBK model. The predicted blood lead level for exposure to subsurface soils at PSC DP-13 were 21.4 µg/dL, which exceeds the concern limit of 10 µg/dL.

REASONABLE MAXIMUM EXPOSURE POINT CONCENTRATIONS IN SOIL AND USEPA REGION IX PRGS FOR SURFACE SOIL (BASE WORKER) PSC DP-13 Luke Air Force Base, Arizona

	Exposure Point Concentration	USEPA Region IX Industrial PRG (mg/kg)		*Risk or	ADEQ SRL (mg/kg)	
Constituent	(mg/kg) 1996 2000 Hazard	Hazard	Residential	Non-Residential		
Cancer Effects				ELCR		
<u>Metals</u> Arsenic Beryllium	6.3 0.47	2.4 1.1	2.7 2,200	2.3E-06 2.1E-10	10 1.4	10 11
Total Site Risk				2.3E-06		
Non-Cancer Effects				HQ		
TRPHs [a]	530	NA	NA	NA	4,100	18,000
<u>Metals</u> Arsenic Beryllium	6.3 0.47	NA NA	440 3,700	1.4E-02 1.3E-04	10 1.4	10 11
Hazard Index				0.01		

All Units in milligrams per kilogram (mg/kg).

Arizona Department of Environmental Quality 2000 Soil Remediation Level.
Base-neutral and extractable acids.
Excess lifetime cancer risk.
Hazard quotient.
Preliminary Remediation Goal.
U.S. Environmental Protection Agency.

[a] n-Hexane was used as a surrogate in the risk assessment for toxicity. A PRG for TRPHs is not provided. *Risk and Hazards are based upon 2000 Industrial PRGs

Surface soil samples were taken from a depth of 0 to 2 feet below ground surface.

Subsurface soil samples were taken from a depth of 2 to 16 feet below ground surface.

Exposure point concentrations were taken from the 95 percent upper confidence limit (one-tailed) on the mean, assuming a normal distribution.

Arsenic background levels in the State of Arizona regularly exceed the PRGs. Arsenic is present at this site within background levels, and when taken out of the equation the ELCR and hazard index are within acceptable risk ranges.

REASONABLE MAXIMUM EXPOSURE POINT CONCENTRATIONS IN SOIL AND USEPA REGION IX PRGS FOR SURFACE AND SUBSURFACE SOIL (EXCAVATION WORKER) PSC DP-13

Luke A	Air F	orce	Base,	Arizona
--------	-------	------	-------	---------

	Exposure Point Concentration	USEPA Region IX Industrial PRG (mg/kg)		*Risk or	ADEQ SRL (mg/kg)	
Constituent	(mg/kg)	1996	2000	Hazard	Residential	Non-Residential
Cancer Effects				ELCR		
BNAs						
Benzo(a)anthracene	0.33	2.6	2.9	1.1E-07	6.1	26
Benzo(b)fluoranthene	0.36	2.6	2.9	1.2E-07	6.1	26
Benzo(a)pyrene	0.32	0.26	0.29	1.1E-06	0.61	2.6
Dibenzo(a,h)anthracene	0.11	0.26	0.29	3.8E-07	0.61	2.6
Metals						
Arsenic	5.1	2.4	2.7	1.9E-06	10	10
Bervllium	0.39	1.1	2.200	1.8E-10	1.4	11
Chromium [a]	820	450	450	1.8E-06	2,100	4,500
Total Site Risk				5.4E-06		
Non-Cancer Effects				HQ		
BNAs						
Benzo(a)anthracene [b]	0.33	800	190	1.7E-03	6.1	26
Benzo(b)fluoranthene [b]	0.36	800	190	1.9E-03	6.1	26
Benzo(a)pyrene [b]	0.32	800	190	1.7E-03	0.61	2.6
Dibenzo(a,h)anthracene [b]	0.11	800	190	5.8E-04	0.61	2.6
<u>TRPH [c]</u>	790	NA	NA	NA	4,100	18,000
Metals						
Arsenic	5.1	NA	440	1.2E-02	10	10
Beryllium	0.39	NA	3,700	1.1E-04	1.4	11
Chromium [d]	820	NA	6,100	1.3E-01	2,100	4,500
Copper	250	63,000	76,000	3.3E-03	2,800	63,000
Hazard Index				0.16		

All Units in milligrams per kilogram (mg/kg).

ADEQ SRL	Arizona Department of Environmental Quality 2000 Soil Remediation Level.
BNAs	Base-neutral and extractable acids.
ELCR	Excess lifetime cancer risk.
HQ	Hazard quotient.
PRG	Preliminary Remediation Goal.
USEPA	U.S. Environmental Protection Agency.

[a] The chromium PRG is based on total chromium carcinogenic effects and a 1:6 ratio of chromium IV to chromium III.

[b] The PRG is based on naphthalene non-carcinogenic effects.

[c] n-Hexane was used as a surrogate in the risk assessment for toxicity. A PRG for TRPHs is not provided.
[d] The chromium PRG is based on chromium VI non-carcinogenic effects.
*Risk and Hazards are based upon 2000 Industrial PRGs

Surface soil samples were taken from a depth of 0 to 2 feet below ground surface. Subsurface soil samples were taken from a depth of 2 to 16 feet below ground surface. Exposure point concentrations were taken from the 95 percent upper confidence limit (one-tailed) on the mean, assuming a normal distribution.

Arsenic background levels in the State of Arizona regularly exceed the PRGs. Arsenic is present at this site within background levels, and when taken out of the equation the ELCR and hazard index are within acceptable risk ranges.

The elevated ELCR and HI values for chromium and blood level values predicted by the IEUBK model were the result of one sample with elevated chromium concentrations (15,900 mg/kg) and lead concentrations (36,000 mg/kg) collected from test pit TP-12. The elevated concentrations of chromium were attributed to a paint pail and dried paint observed in this test pit. The paint pail and dried paint was removed from the test pit.

Assessment

Is the remedy functioning as intended by decision documents?

The objective of the institutional controls was to prevent residential development at the site. The institutional controls have functioned as intended.

Are the assumptions used at the time of the remedy selection still valid?

The assumptions used in performing the risk assessment at the site are still valid.

Has any other information come up that could question the protectiveness of the remedy?

No additional information has come to light that could call into question the protectiveness of the remedy.

Issues

No issues were discovered during this five-year review.

Recommendations and Follow-up Activities

There are no further recommendations or follow-up activities suggested at this time.

Protectiveness Statement

The remedy at PSC DP-13 is protective of human health and the environment and exposure pathways that could result in unacceptable risks are being controlled.

6.2.5 PSC LF-14 Old Salvage Yard Burial Site

Remedial Actions

Remedy Selection

As stated in the OU-1 ROD, the remedy selected for PSC LF-14 consisted of institutional controls. Institutional controls implemented at PSC LF-14 consisted of the following:

- A VEMUR to restrict land use to non-residential.
- The BGP modified to place constraints on future residential development.
- Work practices requiring the use of Personal Protective Equipment (PPE) while excavating the site.
- An ICP to document required institutional controls.

Remedy Implementation

The implementation of the remedy for PSC LF-14 is summarized as follows:

- A VEMUR was placed on the site on June 15, 2000, to restrict residential development on the site.
- The BGP was revised on January 5, 2000, to place restraints on the residential development of the site and to require the use of PPE while excavating soils at the site.
- An ICP was developed and implemented at the site on December 15, 2000, as part of the BGP to facilitate training and education of all personnel involved with the implementation and enforcement of the required institutional controls.
- The ICP details the objectives and rationales for establishing institutional controls and describes the procedures that were implemented to ensure the required institutional controls are enforced.
- The ICP includes provisions for annual reviews and updates of the BGP, thus ensuring regular checks and balances are in place into the foreseeable future.

System Operation and Maintenance

No operation and maintenance was required for the remediation selected. The cost of the institutional controls implemented at PSC LF-14 was \$347.

Progress Since the Last Five-year Review

This is the first five-year review conducted for the site.

Five-year Review Process

Five-year Review findings

Site Inspection

There were no changes in land use since implementation of the remedy observed during the site inspection. Photographs of PSC LF-14 taken as part of the site inspection are in Appendix E.

Changes in Standards

A comparison of exposure point concentrations in surface soils and USEPA Region IX PRGs for a base worker, and surface and sub-surface soils and USEPA Region IX PRGs for an excavation worker are in Tables 6-9 and 6-10, respectively. The comparison indicates that PSC LF-14 is still within the acceptable risk range.

Changes in Exposure Pathways, Toxicity and Other Contaminant Characteristics

There were no changes in exposure pathways, toxicity, or other contaminant characteristics.

REASONABLE MAXIMUM EXPOSURE POINT CONCENTRATIONS IN SOIL AND USEPA REGION IX PRGS FOR SURFACE SOIL (BASE WORKER) PSC LF-14

	Luk	e.	Air	F	orce	B	lase,	A	riz	ZOI	na
--	-----	----	-----	---	------	---	-------	---	-----	-----	----

	Exposure Point Concentration	USEPA Region IX Industrial PRG (mg/kg)		*Risk or	ADEQ SRL (mg/kg)		
Constituent	(mg/kg)	1996	2000	Hazard	Residential	Non-Residential	
Cancer Effects				ELCR			
BNAs Benzo(a)pyrene	0.15	0.26	0.29	5.2E-07	0.61	2.6	
PCBs	3.6	0.34	1.0	3.6E-06	2.5	13	
<u>Metals</u> Arsenic Beryllium Chromium [a]	5.8 0.62 100	2.4 1.1 450	2.7 2,200 450	2.1E-06 2.8E-10 2.2E-07	10 1.4 2,100	10 11 4,500	
Total Site Risk				6.5E-06			
Non-Cancer Effects				HQ			
<u>BNAs</u> Benzo(a)pyrene [b]	0.150	800	190	7.9E-04	0.61	2.6	
PCBs [c]	3.6	NA	14.0	2.6E-01	2.5	13	
TRPHs [d]	1100	NA	NA	NA	4,100	18,000	
Metals							
Arsenic	5.8	NA	440	1.3E-02	10	10	
Beryllium Chromium [e]	0.62 100	NA NA	3,700 6,100	1.7E-04 1.6E-02	1.4 2,100	11 4,500	
Hazard Index				0.3			

All Units in milligrams per kilogram (mg/kg).

ADEO SRL Arizona Department of Environmental Quality 2000 Soil Remediation Level. **BNAs** Base-neutral and extractable acids. ELCR Excess lifetime cancer risk. HQ Hazard quotient. NA Not applicable. PRG Preliminary Remediation Goal. USEPA U.S. Environmental Protection Agency.

[a] The chromium PRG is based on total chromium carcinogenic effects and a 1:6 ratio of chromium IV to chromium III.

[b] The PRG is based on naphthalene non-carcinogenic effects. [c] The PRG is based on the Aroclor 1254 non-carcinogenic effects.

[d] n-Hexane was used as a surrogate in the risk assessment for toxicity. A PRG for TRPHs is not provided.

[e] The chromium PRG is based on chromium VI non-carcinogenic effects.

*Risk and Hazards are based upon 2000 Industrial PRGs

Surface soil samples were taken from a depth of 0 to 2 feet below ground surface. Subsurface soil samples were taken from a depth of 2 to 16 feet below ground surface. Exposure point concentrations were taken from the 95 percent upper confidence limit (one-tailed) on the mean, assuming a normal distribution.

Arsenic background levels in the State of Arizona regularly exceed the PRGs. Arsenic is present at this site within background levels, and when taken out of the equation the ELCR and hazard index are within acceptable risk ranges.

REASONABLE MAXIMUM EXPOSURE POINT CONCENTRATIONS IN SOIL AND USEPA REGION IX PRGS FOR SURFACE AND SUBSURFACE SOIL (EXCAVATION WORKER) PSC LF-14 a

	Exposure Point Concentration	USEPA Region IX Industrial PRG (mg/kg)		*Risk or	ADEQ SRL (mg/kg)		
Constituent	(mg/kg)	1996	2000	Hazard	Residential	Non-Residential	
Cancer Effects				ELCR			
<u>BNAs</u> Benzo(a)pyrene	0.16	0.26	0.29	5.5E-07	0.61	2.6	
PCBs	5.2	0.34	1.0	5.2E-06	2.5	13	
<u>Metals</u> Arsenic Beryllium Chromium [a]	5.4 0.53 59	2.4 1.1 450	2.7 2,200 450	2.0E-06 2.4E-10 1.3E-07	10 1.4 2,100	10 11 4,500	
Total Sue Kisk				7.9E-00			
<u>Non-Cancer Effects</u> <u>BNAs</u> Benzo(a)pyrene [b]	0.16	800	190	HQ 8.4E-04	0.61	2.6	
PCBs [c]	5.2	NA	14.0	3.7E-01	2.5	13	
TRPHs [d]	570	NA	NA	NA	4,100	18,000	
<u>Metals</u> Arsenic Beryllium Chromium [e]	5.4 0.53 59	NA NA NA	440 3,700 6,100	1.2E-02 1.4E-04 9.7E-03	10 1.4 2,100	10 11 4,500	
Hazard Index				0.4			

All Units in milligrams per kilogram (mg/kg).

ADEO SRL Arizona Department of Environmental Quality 2000 Soil Remediation Level. **BNAs** Base-neutral and extractable acids. ELCR Excess lifetime cancer risk. HQ Hazard quotient. NA Not applicable. PRG Preliminary Remediation Goal. USEPA U.S. Environmental Protection Agency.

[a] The chromium PRG is based on total chromium carcinogenic effects and a 1:6 ratio of chromium IV to chromium III.

[b] The PRG is based on naphthalene non-carcinogenic effects.

[c] The PRG is based on the Aroclor 1254 non-carcinogenic effects.

[d] n-Hexane was used as a surrogate in the risk assessment for toxicity. A PRG for TRPHs is not provided.

[e] The chromium PRG is based on chromium VI non-carcinogenic effects.

*Risk and Hazards are based upon 2000 Industrial PRGs

Surface soil samples were taken from a depth of 0 to 2 feet below ground surface. Subsurface soil samples were taken from a depth of 2 to 16 feet below ground surface. Exposure point concentrations were taken from the 95 percent upper confidence limit (one-tailed) on the mean, assuming a normal distribution.

Arsenic background levels in the State of Arizona regularly exceed the PRGs. Arsenic is present at this site within background levels, and when taken out of the equation the ELCR and hazard index are within acceptable risk ranges.

Data Evaluation

The determination to use institutional controls was based on the ELCR at the site of 3×10^{-5} , above the allowable residential risk of 1×10^{-6} . The elevated ELCR was caused by two samples with high chromium concentrations. The risk assessment used the conservative assumption that all of the chromium was in the hexavalent state. Additionally high concentrations of PCBs elevated the ELCR. The highest concentration of PCBs was found at 20 feet bgs. Because exposure to soils beneath 16 feet bgs is not likely concentration of PCBs detected below 16 feet were not used in calculating the ELCR.

Assessment

Is the remedy functioning as intended by decision documents?

The objective of the institutional controls was to prevent residential development at the site. The institutional controls have functioned as intended.

Are the assumptions used at the time of the remedy selection still valid?

The assumptions used in performing the risk assessment at the site are still valid.

Has any other information come up that could question the protectiveness of the remedy?

No additional information has come to light that could call into question the protectiveness of the remedy.

Issues

No issues were discovered during this five-year review.

Recommendations and Follow-up Activities

No follow-up activities are suggested at this time.

Protectiveness Statement

The remedy at PSC LF-14 is protective of human health and the environment and exposure pathways that could result in unacceptable risks are being controlled.

6.2.6 PSC LF-25 Northwest Landfill

Remedial Actions

Remedy Selection

As stated in the OU-1 ROD, the remedy selected for PSC LF-25 consisted of the following:

- Ex-situ physical treatment/metals recovery
- Institutional controls

Remedy Implementation

Implementation of ex-situ physical treatment/metals recovery at PSC LF-25 is summarized as follows:

- Shot recovery activities, conducted from December 16-19, 1999³¹, included removal of surficial soil from an area approximately 375 feet by 375.
- The soil was fed into a metal recovery processor, which sorted out the metal shot and returned that soil to the ground.
- Approximately 2,800 pounds of shot was recovered.
- Confirmation sampling was conducted to ensure that site remediation was effective.
- The analytical results showed that all soil samples were below the residential SRLs of 31 mg/kg for antimony and 400 mg/kg for lead.

Implementation of institutional controls at PSC LF-25 is summarized as follows:

³¹ ARCADIS Geraghty & Miller, Inc., 2000. Shot Recovery Summary Report for PSC LF-25, June 1, 2000.

- A VEMUR was placed on the site on June 15, 2000, to restrict residential development on the site.
- The BGP was revised on January 5, 2000, to place restraints on the residential development of the site and to require the use of PPE while excavating soils at the site.
- An ICP was developed and implemented at the site on December 15, 2000, as part of the BGP to facilitate training and education of all personnel involved with the implementation and enforcement of the required institutional controls.
- The ICP details the objectives and rationales for establishing institutional controls and describes the procedures that were implemented to ensure the required institutional controls are enforced.
- The ICP includes provisions for annual reviews and updates of the BGP, thus ensuring regular checks and balances are in place into the foreseeable future.

System Operation and Maintenance

No operation and maintenance was required for the remediation selected. The cost of the institutional controls implemented at PSC LF-25 was \$347. The cost of the ex-situ physical treatment/metals recovery was \$42,985.

Progress Since the Last five-year Review

This is the first five-year review conducted for the site.

Five-year Review Process

Five-year Review Findings

Site Inspection

There were no changes in land use since implementation of the remedy observed during the site inspection. Photographs of PSC LF-25 taken as part of the site inspection are in Appendix E.

Changes to Standards

A comparison of exposure point concentrations in surface and subsurface soils and USEPA Region IX PRGs for Base and excavation worker is in Table 6-11. The comparison indicates that PSC LF-25 is still within the acceptable risk range.

POST REMEDIATION CONCENTRATIONS IN SOIL AND USEPA REGION IX PRGS FOR SURFACE AND SUBSURFACE SOIL (BASE AND EXCAVATION WORKER) PSC LF-25 Luke Air Force Base, Arizona

	Post Remediation Concentration	nediation USEPA Region IX Industrial PRG (mg/kg)		*Risk or	ADEQ SRL (mg/kg)	
Constituent	(mg/kg)	1996	2000	Hazard	Residential	Non-Residential
Cancer Effects				ELCR	_	
Metals						
Antimony	10	NC	NC	NC	NA	NA
Lead	240	400	750	3.2E-07	400	2,000
Total Site Risk or Hazard				3E-07		
Non-Cancer Effects				HQ	-	
Metals						
Antimony	10	NA	820	1.2E-02	NA	NA
Lead [a]	240	400	750	0.32	400	2,000
Hazard Index				0.3		

All Units in milligrams per kilogram (mg/kg).

Arizona Department of Environmental Quality 2000 Soil Remediation Level.
Base-neutral and extractable acids.
Excess lifetime cancer risk.
Iazard quotient.
Not Applicable
Non Carcinogenic
Preliminary Remediation Goal.
J.S. Environmental Protection Agency.

[a] The PRG for lead is based on acceptable levels of lead in the blood stream rather than a traditional toxicity approach.

The hazard quotient presented under non-cancer effects is actually just the ratio of the PRG to the soil concentration.

*Risk and Hazards are based upon 2000 Industrial PRGs

Results were taken from post remediation conformational soil sampling constituent concentrations. Post remediation results were obtained from the Results and Conclusions section of the Remedial Action Report for PSC LF-25 by ARCADIS Geraghty & Miller dated June 1, 2000.

Surface soil samples were taken from a depth of 0 to 2 feet below ground surface.

Subsurface soil samples were taken from a depth of 2 to 16 feet below ground surface.

Changes in Exposure Pathways, Toxicity and Other Contaminant Characteristics

There were no changes in exposure pathways, toxicity, or other contaminant characteristics.

Data Evaluation

The determination to implement institutional controls was based on failed site-specific industrial risk standards that was a result of one sample with high antimony concentrations. This sample contained a piece of metal shot, resulting in the high concentration of antimony. The metal shot came from the nearby skeet range. Removal of the metal shot from the site was conducted in December 1999 and subsequent soil sampling indicated that soil levels were below residential SRLs³¹. Since no carcinogens were identified as COCs, an ELCR was not calculated for the site.

Mean blood lead levels for sensitive populations, children up to seven years old, were calculated using the IEUBK model. The predicted blood lead level for exposure to subsurface soils at LF-25 was 14.5 μ g/dL. This is above the limit of 10 μ g/dL. The high concentration of lead in one sample (10,100 mg/kg) elevated the predicted blood lead level. This sample contained a piece of metal shot, resulting in the high concentration of lead. The metal shot came from the nearby skeet range. Removal of the metal shot from the site was conducted in December 1999, and subsequent soil sampling indicated that soil levels were below residential SRLs. Even though antimony and lead concentrations are below residential SRLs, institutional controls (VEMUR) are still required because the site is still utilized as an active skeet range and there is still a potential source of these metals.

Assessment

Is the remedy functioning as intended by decision documents?

The objective of the institutional controls was to prevent residential development at the site. The institutional controls have functioned as intended. Removal of the metal shot from the site was conducted in December 1999, and subsequent soil sampling indicated that soil levels were below residential SRLs³¹.

Are the assumptions used at the time of the remedy selection still valid?

The assumptions used in performing the risk assessment at the site are still valid.

Has any other information come up that could question the protectiveness of the remedy?

No additional information has come to light that could call into question the protectiveness of the remedy.

Issues

No issues were discovered during this five-year review.

Recommendations and Follow-up Activities

No follow-up activities are suggested at this time.

Protectiveness Statement

The remedy at PSC LF-25 is protective of human health and the environment and exposure pathways that could result in unacceptable risks are being controlled.

6.2.7 PSC SD-38 Southwest Oil/Water Separator at the Auto Hobby Shop

Remedial Actions

Remedy Selection

As stated in the OU-1 ROD, the remedy selected for PSC SD-38 consisted of institutional controls. Institutional controls implemented at PSC SD-38 consisted of the following:

- A VEMUR to restrict land use to non-residential.
- The BGP modified to place constraints on future residential development.
- Work practices requiring the use of PPE while excavating the site.
- An ICP to document required institutional controls.

Remedy Implementation

The implementation of the remedy for PSC SD-38 is summarized as follows:

- A VEMUR was placed on the site on June 15, 2000, to restrict residential development on the site.
- The BGP was revised on January 5, 2000, to place restraints on the residential development of the site and to require the use of PPE while excavating soils at the site.
- An ICP was developed and implemented at the site on December 15, 2000, as part of the BGP to facilitate training and education of all personnel involved with the implementation and enforcement of the required institutional controls.
- The ICP details the objectives and rationales for establishing institutional controls and describes the procedures that were implemented to ensure the required institutional controls are enforced.
- The ICP includes provisions for annual reviews and updates of the BGP, thus ensuring regular checks and balances are in place into the foreseeable future.

System Operation and Maintenance

No operation and maintenance was required for the remediation selected. The cost of the institutional controls implemented at PSC SD-38 was \$347.

Progress Since the Last Five-year Review

This is the first five-year review conducted for the site.

Five-year Review Process

Five-year Review Findings

Site Inspection

There were no changes in land use since implementation of the remedy observed during the site inspection. Photographs of PSC SD-38 taken as part of the site inspection are in Appendix E.

Changes to Standards

A comparison of exposure point concentrations in surface and subsurface soils and USEPA Region IX PRGs for an excavation worker is in Table 6-12. The comparison indicates PSC SD-38 is still within the acceptable risk range.

REASONABLE MAXIMUM EXPOSURE POINT CONCENTRATIONS IN SOIL AND USEPA REGION IX PRGS FOR SURFACE AND SUBSURFACE SOIL (EXCAVATION WORKER) PSC SD-38 Luke Air Force Base, Arizona

	Exposure Point Concentration	USEPA Region IX Industrial PRG (mg/kg)		*Risk or	ADEQ SRL (mg/kg)		
Constituent	(mg/kg)	1996	2000	Hazard	Residential	Non-Residential	
Cancer Effects				ELCR	-		
Metals							
Arsenic	7.8	2.4	2.7	2.9E-06	10	10	
Beryllium	0.37	1.1	2,200	1.7E-10	1.4	11	
Total Site Risk				3E-06			
Non-Cancer Effects				HQ	-		
TRPHs [a]	16,000	NA	NA	NA	4,100	18,000	
Metals							
Arsenic	7.8	NA	440	1.8E-02	10	10	
Beryllium	0.37	NA	3,700	1.0E-04	1.4	11	
Hazard Index				0.02			

All Units in milligrams per kilogram (mg/kg).

ADEQ SRL	Arizona Department of Environmental Quality 2000 Soil Remediation Level.
BNAs	Base-neutral and extractable acids.
ELCR	Excess lifetime cancer risk.
HQ	Hazard quotient.
NA	Not available.
PRG	Preliminary Remediation Goal.
TRPHs	Total recoverable petroleum hydrocarbons.
USEPA	U.S. Environmental Protection Agency.

[a] n-Hexane was used as a surrogate in the risk assessment for toxicity. A PRG for TRPHs is not provided. *Risk and Hazards are based upon 2000 Industrial PRGs

Surface soil samples were taken from a depth of 0 to 2 feet below ground surface.

Subsurface soil samples were taken from a depth of 2 to 16 feet below ground surface.

Exposure point concentrations were taken from the 95 percent upper confidence limit (one-tailed) on the mean, assuming a normal distribution.

Arsenic background levels in the State of Arizona regularly exceed the PRGs. Arsenic is present at this site within background levels, and when taken out of the equation the ELCR and hazard index are within acceptable risk ranges.

Changes in Exposure Pathways, Toxicity and Other Contaminant Characteristics

There were no changes in exposure pathways, toxicity, or other contaminant characteristics.

Data Evaluation

The determination to use institutional controls was based on the HI, which was above the allowable residential risk of 1.0. The elevated HI was caused by several samples with high TRPH concentrations.

Assessment

Is the remedy functioning as intended by decision documents?

The objective of the institutional controls was to prevent residential development at the site. The institutional controls have functioned as intended.

Are the assumptions used at the time of the remedy selection still valid?

The assumptions used in performing the risk assessment at the site are still valid.

Has any other information come up that could question the protectiveness of the remedy?

No additional information has come to light that could call into question the protectiveness of the remedy.

Issues

No issues were discovered during this five-year review.

Recommendations and Follow-up Activities

No recommendations or follow-up activities are suggested at this time.

Protectiveness Statement

The remedy at PSC SD-38 is protective of human health and the environment and exposure pathways that could result in unacceptable risks are being controlled.

6.2.8 PSC SS-42 Bulk Fuels Storage

Remedial Actions

Remedy Selection

As stated in the OU-1 ROD, the remedy selected for PSC SS-42 consisted of the following:

- Soil vapor extraction
- Groundwater monitoring

Remedy Implementation

Implementation of the remedial action selected for PSC SS-42 is summarized as follows:

- In August 1996, the Base initiated a SVE removal action at PSC SS-42.
- A highly modified ICE was used to draw contaminated vapors from the ground and to treat the off-gas prior to discharge.
- The SVE removal action continued through November 1998.
- In June 1997, an initial confirmation boring was advanced to a depth of 181 feet bgs near the former UST location.
- The analytical results indicated that BTEX and TPH had been decreased in the subsurface. However, the results also indicated that BTEX was detected at depths below 150 bgs at concentrations higher than they had been originally detected.
- A second confirmation boring, located approximately eight feet northwest of monitoring well MW-121, was advanced to a depth of 310 feet bgs.
- TPH and benzene were detected above their respective residential SRLs. Benzene was also detected above the industrial SRLs. Toluene, ethylbenzene,

and total xylenes were detected in several samples but below their respective SRLs³².

- Analytical data indicates that SVE operation has removed approximately 399,514 pounds of TPH (approximately 66,584 gallons of hydrocarbons) and reduced BTEX concentrations by 87%.
- Although benzene was detected above the AWQS during the November 1998 groundwater sampling event, the May 1999, May 2000 and August 2001 samples did not contain benzene above laboratory detection limits.
- Groundwater monitoring has continued at the site³².

System Operation and Maintenance

Operation and maintenance was performed monthly during operation of the SVE. This included sampling, field measurements, readings from the system, and engine service. Also, any problems with the system between monthly visits were addressed as needed. The cost of the operation and maintenance was \$65,910. There was no cost for remediation because the internal combustion engine (ICE) was provided to Luke AFB without charge by AFCEE.

Progress Since the Last Five-year Review

This is the first five-year review conducted for the site.

Five-year Review Findings

Site Inspection

There were no changes in land use since implementation of the remedy observed during the site inspection. Photographs of PSC SS-42 taken as part of the site inspection are in Appendix E.

³² ARCADIS Geraghty & Miller, Inc. 2000. Soil Vapor Extraction and Confirmation Sampling Summary Report, PSC SS-42, May 22, 2000.

Changes to Standards

A comparison of exposure point concentrations in surface and subsurface soils and USEPA Region IX PRGs for an excavation worker is in Table 6-13. The comparison indicates that PSC SS-42 is still within the acceptable risk range.

Changes in Exposure Pathways, Toxicity and Other Contaminant Characteristics

There were no changes in exposure pathways, toxicity, or other contaminant characteristics.

Data Evaluation

Based on the laboratory analysis of soil samples collected from confirmation soil boring number 2 (CB-2), detected TPH concentrations in the soil directly beneath the former Leaking Under Storage Tank (LUST) range from 250 to 7,400 mg/kg. With the exception of the 7,400 mg/kg concentration, all other detected TPH values are below the residential SRL of 4,100 mg/kg. The TPH concentration of 7,400 mg/kg is above the residential SRL but below the non-residential SRL of 14,000 mg/kg. Detected benzene concentrations ranged from 0.23 to 150 mg/kg. The benzene concentrations detected at 140 feet bgs (150 mg/kg) and 150 feet bgs (2.5 mg/kg) were above both the residential SRL (0.62 mg/kg) and non-residential SRL (1.4 mg/kg), respectively. Detected toluene concentrations were below both the residential SRL (790 mg/kg) and the non-residential SRL (2,700 mg/kg), respectively. Detected ethylbenzene concentrations were below both the residential SRL (1,500 mg/kg) and the non-residential SRL (2,700 mg/kg), respectively. Detected total xylenes concentrations were below both the residential (2,800mg/kg) and non-residential SRL (2,800mg/kg). Analytical data indicates that SVE operation has removed approximately 399, 514 pounds of TPH (approximately 66,584 gallons of hydrocarbons) and reduced BTEX concentrations by 87 percent. Although benzene was detected above AWQSs during the November 1998 groundwater-sampling event, the May 1999, May 2000 and August 2001 samples did not contain benzene above laboratory detection limits. Groundwater monitoring has continued at the site 32 .

POST REMEDIATION CONCENTRATIONS IN SOIL AND USEPA REGION IX PRGS FOR SURFACE AND SUBSURFACE SOIL (EXCAVATION WORKER) PSC SS-42 Luke Air Force Base, Arizona

	Post Remediation Concentration	USEPA Region IX Industrial PRG (mg/kg)		*Risk or	ADEQ SRL (mg/kg)		
Constituent	(mg/kg)	1996	2000	Hazard	Residential	Non-Residential	
Cancer Effects				ELCR			
VOCs							
Benzene	ND	1.4	1.5	NA	0.62	1.4	
Ethylbenzene [a]	ND	NC	NC	NA	1,500	2,700	
Toluene [a]	ND	NC	NC	NA	790	2,700	
(total) Xylenes [a]	ND	NC	NC	NA	2,800	2,800	
Total Site Risk				NA			
Non-Cancer Effects				<u>HQ</u>			
<u>TPH [b]</u>	ND	NA	NA	NA	4,100	18,000	
VOCs							
Benzene	ND	NA	24	NA	0.62	1.4	
Ethylbenzene [a]	ND	230	230	NA	1,500	2,700	
Toluene [a]	ND	880	520	NA	790	2,700	
(total) Xylenes [a]	ND	320	210	NA	2,800	2,800	
Hazard Index				NA			

All Units in milligrams per kilogram (mg/kg).

ADEQ SRL	Arizona Department of Environmental Quality 2000 Soil Remediation Level.
BNAs	Base-neutral and extractable acids.
ELCR	Excess lifetime cancer risk.
HQ	Hazard quotient.
PRG	Preliminary Remediation Goal.
USEPA	U.S. Environmental Protection Agency.

ND - Non Detect

NA - Not Applicable

[a] The PRG is based on soil saturation.

[b] n-Hexane was used as a surrogate in the risk assessment for toxicity. A PRG for TRPHs is not provided.

*Risk and Hazards are based upon 2000 Industrial PRGs

Exposure point concentrations are based upon second confirmation sampling analytical results from 1999 of the Soil Vapor Extraction and Confirmation Sampling Summary Report by ARCADIS Geraghty & Miller dated May 22, 2000.

Surface soil samples were taken from a depth of 0 to 2 feet below ground surface. Subsurface soil samples were taken from a depth of 2 to 16 feet below ground surface.

Groundwater Protection Level (GPL) Modeling

As detailed in the ROD, vadose zone fate-and-transport modeling was previously conducted at the site during the OU-1 remedial investigation¹³. Results of this modeling indicate that petroleum related compounds (i.e. TPH and BTEX) could eventually leach to the groundwater. However, the vadose zone modeling results conducted as part of the OU-1 remedial investigation did not predict whether these petroleum related compounds could cause a violation of the AWQS at a point of compliance. As a result, groundwater protection levels (GPLs) had not been previously established for the site. GPLs could not be calculated for TPH because there are no numeric water quality standards established for TPH. GPLs can only be calculated for individual constituents with AWQSs. Of the petroleum-related constituents with established AWQSs detected at the site, BTEX compounds posed the greatest potential risk to human health. For these reasons, GPLs calculated for BTEX are considered representative values established for the petroleum release at the site.

As a consequence of the limited depth of incorporation range presented in the ADEQ "Alternative GPL" tables, a site-specific model had to be used to determine GPLs for the site. The ADEQ screening model was selected for use in this evaluation. Several model runs were conducted using varying depths of incorporation and varying depths to groundwater. These additional runs were conducted so that GPLs could be established for a variety of potential site conditions in the event confirmation sampling at the site yields a different depth of incorporation and depth to groundwater than indicated by previously collected site characterization data. The results of the additional modeling runs are summarized below:

- GPLs calculated for benzene ranged from 8,685 mg/kg (180.45 ft depth of incorporation and 295.28 ft depth to groundwater) to 400,600 mg/kg (180.45 ft depth of incorporation and 328.10 ft depth to groundwater).
- GPLs calculated for the ethylbenzene ranged from 679 mg/kg (180.45 ft depth of incorporation and 229.66 ft depth to groundwater) to GWNT at variable depths.
- GPLs calculated for toluene ranged from 35,310 mg/kg (180.45 ft depth of incorporation and 229.66 ft depth to groundwater) to GWNT at variable depths.
- GPLs calculated for xylenes ranged from 23,580 mg/kg (180.45 ft depth of incorporation and 229.66 ft depth to groundwater) to GWNT at variable depths.

Concentrations of BTEX remaining in the soils are protective of groundwater. Analytical results and the GPL model also indicate that remediation has decreased hydrocarbon concentrations to this protective point and further remediation is not needed. However, because constituents of concern were detected at a depth of 140 feet bgs, it was prudent to conduct groundwater monitoring.

Assessment

Is the remedy functioning as intended by decision documents?

The objective of the remedial action was to clean up impacted soil and prevent migration to groundwater. Analytical data indicates that SVE operation has removed approximately 399,514 pounds of TPH (approximately 66,584 gallons of hydrocarbons) and reduced BTEX concentrations by 87 percent. TPH and BTEX were not detected above laboratory detection limits during the most recent groundwater results.

Are the assumptions used at the time of the remedy selection still valid?

The assumptions made at the time of the remedy selection are still valid.

Has any other information come up that could question the protectiveness of the remedy?

No additional information has come to light that could call into question the protectiveness of the remedy.

Issues

No issues were discovered as part of this five-year review.

Recommendations and Follow-up Activities

MW-121 and MW-125R will be monitored as part of future five-year reviews. No other recommendations or follow-up activities are suggested at this time.

Protectiveness Statement

The remedy at PSC SS-42 is protective of human health and the environment and exposure pathways that could result in unacceptable risks are being controlled.

6.3 Review Process for OU-2 PSCs

OU-2 PSCs for which there was no action taken (refer to discussion in Section 1.1.3 and information in Table 3-1) include the following:

- OT-04
- DP-05
- FT-06
- FT-07W
- DP-22
- SD-40

A comparison of the EPC for a given COC in the combined surface and subsurface soil, with USEPA Region IX industrial PRGS (1996 and 2000) and ADEQ residential and non-residential SRLs is in Appendix C. EPCs were taken from the 95 percent upper confidence limit (one-tailed) on the mean assuming a normal distribution.

OU-2 PSCs for which remedies were selected based on the results of risk assessment include the following:

- ST-18
- DP-23

For OU-2 PSCs, 1991 USEPA Region IX PRGs were originally used to establish performance standards. To evaluate changes in standards as part of this five-year review, cancer risks and non-cancer hazards were recalculated for each COC using 2000 USEPA Region IX industrial PRGs and post remediation exposure point concentrations for Base worker and excavation worker scenarios as applicable. The analysis of standard changes also included a review of 1996 USEPA industrial PRGs. ADEQ SRLs were also reviewed since they were used to evaluate residential use standards. USEPA Region IX PRGs for 1991, 1996 and 2000 are in Appendix D. Changes in standards are evaluated with respect to the continued effectiveness of the

remedies that were implemented based on a non-cancer HI greater than or equal to 1.0 or an ELCR greater than the risk of 1×10^{-6} to 1×10^{-4} .

6.3.1 PSC ST-18 Facility 993

Remedial Actions

Remedy Selection

As stated in the OU-2 ROD, the remedy selected for PSC ST-18 consisted of the following:

- Inspection and maintenance of concrete cap
- Institutional controls
- Monitoring of groundwater every five years

Remedy Implementation

The implementation of the remedy for PSC ST-18 is summarized as follows:

- The site was capped with nine inches of concrete, underlain by six inches of base course and a 30-mil HDPE liner as part of the RCRA closure requirement in 1987.
- The integrity of the cap has been maintained through annual inspections of the concrete and joints and repairs as needed in accordance with the Air Force design guidance for airfield pavement maintenance and recommendations contained in the annual inspection report. Annual inspection reports are maintained at the Environmental Flight office of Luke AFB. A visual inspection was conducted in August 2000 and the need for some repairs was identified. Recommended repairs were performed in August 2001. The annual inspection report dated October 2001 documents the successful completion of repairs.
- According to the ROD, a deed restriction would be placed on the site as part of the surface controls to prevent removal of the cap and excavation of the soil. A DEUR has been filed with the ADEQ for this site.
- The other surface control at the site is the Base perimeter fence monitored 24-hours a day which prevents public access and exposure.

- Groundwater at the site has been monitored semiannually since 1991. A review of the site data was conducted in 2000 and it was concluded that groundwater monitoring was not necessary at the site³³.
- The FCOR states that groundwater monitoring will be conducted at the site as part of each five-year review. Well MW-114 was monitored in October 2001 as part of the five-year review process. No constituent s exceeded standards in the sample from MW-114.

System Operation and Maintenance

Operation and maintenance has included annual inspections of the cap. The integrity of the cap has been maintained through annual inspections of the concrete and joints and repairs have been conducted as needed in accordance with the Air Force design guidance for airfield pavement and maintenance and recommendations contained in the annual inspection reports. The cost of the cap in 1987 was \$122,300. The annual cap inspection is \$2,500. To date, the costs of repairs to the cap have been \$12,118. An additional \$3,880 in repairs is scheduled for 2001.

Progress Since the Last Five-year Review

This is the first five-year review conducted for the site.

Five-year Review Process

Five-year Review Findings

Site Inspection

Comments made during the site inspection of PSC ST-18 are noted under recommendations and follow-up activities below. No changes in land use were observed since implementation of the remedy. Photographs of PSC ST-18 taken as part of the site inspection are in Appendix E.

³³ Geraghty & Miller, Inc., 2000. RCRA Facility Investigation Summary Report Facility #993 (PSC ST-18), Luke Air Force Base, Arizona, December 19, 2000.
Changes to Standards

A comparison of exposure point concentrations in surface and subsurface soils and USEPA Region IX PRGs for an excavation worker is in Table 6-14. The comparison indicates that PSC ST-18 is still within the acceptable risk range.

Changes in Exposure Pathways, Toxicity and Other Contaminant Characteristics

There were no changes in exposure pathways, toxicity, or other contaminant characteristics.

POST REMEDIATION CONCENTRATIONS IN SOIL AND USEPA REGION IX PRGS FOR SURFACE AND SUBSURFACE SOIL (EXCAVATION WORKER) PSC ST-18 Լոե ona

uke Air Force Base, Ariz	0
--------------------------	---

$\begin{array}{c c c c c c c c c c c c c c c c c c c $		Post Remediation Concentration	USEPA F Industrial P	Region IX RG (mg/kg)	*Risk or	ADE (m	EQ SRL g/kg)
$\begin{tabular}{ c c c c c c c } \hline Cancer Effects & ELCR & & & & & & & & & & & & & & & & & & &$	Constituent	(mg/kg)	1996	2000	Hazard	Residential	Non-Residential
$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	Cancer Effects				ELCR		
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	<u>VOCs</u> Ethylbenzene Xylenes	0.15 1.3	NC NC	NC NC	NA NA	1,500 2,800	2,700 2,800
Total Site Risk 2E-06 Non-Cancer Effects HQ VOCs Ethylbenzene 0.15 230 230 6.5E-04 1,500 2,700 Xylenes 1.3 320 210 6.2E-03 2,800 2,800 BNAs Benzo(a)anthracene [b] 0.43 800 190 2.3E-03 6.1 26 Benzo(g),hiperylene [a] 0.56 100 54,000 1.0E-05 NA NA Benzo(a),hiperylene [b] 0.43 800 190 2.3E-03 0.61 2.6 Benzo(a),hiperylene [b] 0.43 800 190 0.00 NA 320 1,400 Chrysene [c] 0.075 140 180 NA	BNAs Benzo(a)anthracene Benzo(b)fluoranthene Benzo(a)pyrene Benzyl alcohol Bis(2-Ethylhexyl)phthalate Chrysene Indeno (1,2,3-cd)pyrene	$\begin{array}{c} 0.43 \\ 0.77 \\ 0.43 \\ 0.42 \\ 0.075 \\ 0.92 \\ 0.34 \end{array}$	2.62.60.26100,0001407.20.61	2.92.90.29100,000180620.62	1.5E-07 2.7E-07 1.5E-06 4.2E-12 4.2E-10 1.5E-08 5.5E-07	$ \begin{array}{c} 6.1 \\ 6.1 \\ 0.61 \\ 20,000 \\ 320 \\ 610 \\ 6.1 \\ \end{array} $	26262.6200,0001,4002,60026
Non-Cancer Effects HQ VOCs Ethylbenzene 0.15 230 230 6.5E-04 1,500 2,700 Xylenes 1.3 320 210 6.2E-03 2,800 2,800 BNAs Benzo(a)anthracene [b] 0.43 800 190 2.3E-03 6.1 26 Benzo(g),hi)perylene [a] 0.56 100 54,000 1.0E-05 NA NA Benzo(g),hi)perylene [a] 0.43 800 190 2.3E-03 6.1 26 Benzo(g),hi)perylene [a] 0.43 800 190 2.3E-03 0.61 2.6 Benzo(g),hi)perylene [a] 0.43 800 190 2.3E-03 0.61 2.6 Benzo(g),hi)perylene [a] 0.43 800 190 2.3E-03 0.61 2.6 Benzo(g),hi)perylene [a] 0.43 800 190 0.2B-05 NA 610 2,600 Benzo(g),hi)perylene [a] 0.42 100,000 100,000 4.2E-06 20,000 2,600 Chrys	Total Site Risk				2E-06		
VOCs Ethylbenzene 0.15 230 230 6.5E-04 1,500 2,700 Xylenes 1.3 320 210 6.2E-03 2,800 2,800 BNAs Benzo(a)anthracene [b] 0.43 800 190 2.3E-03 6.1 26 Benzo(b)fluoranthene [b] 0.77 800 190 4.1E-03 6.1 26 Benzo(a),hiperylene [a] 0.56 100 54,000 1.0E-05 NA NA Benzo(a),hiperylene [b] 0.43 800 190 2.3E-03 0.61 2.6 Benzo(a), pyrene [b] 0.43 800 190 2.3E-03 0.61 2.6 Benzo(a), pyrene [b] 0.43 800 190 2.3E-03 0.61 2.6 Benzo(a), pyrene [b] 0.43 800 190 2.3E-03 0.61 2.6 Chrysene [c] 0.75 140 180 NA 320 1,400 Fluoranthene 0.49 27,000 30,000 1.6E-05 2	Non-Cancer Effects				HQ		
BNAs Benzo(a)anthracene [b] 0.43 800 190 2.3E-03 6.1 26 Benzo(b)fluoranthene [b] 0.77 800 190 4.1E-03 6.1 26 Benzo(g,h,i)perylene [a] 0.56 100 54,000 1.0E-05 NA NA Benzo(a)pyrene [b] 0.43 800 190 2.3E-03 0.61 2.6 Benzyl alcohol 0.42 100,000 100,000 4.2E-06 20,000 200,000 Bis(2-ethylhexyl)phthalate [c] 0.075 140 180 NA 320 1,400 Chrysene [c] 0.92 7.2 62 NA 610 2,600 Fluoranthene 0.49 27,000 30,000 1.6E-05 2,600 27,000 2-Methylnaphthalene [d] 0.35 800 190 0.00 NE NE Phenanthrene [a] 0.18 100 54,000 3.3E-06 NA NA Pyrene 0.56 100 54,000 1.0E-05	<u>VOCs</u> Ethylbenzene Xylenes	0.15 1.3	230 320	230 210	6.5E-04 6.2E-03	1,500 2,800	2,700 2,800
Hazard Index 0.02	BNAs Benzo(a)anthracene [b] Benzo(g),hi)perylene [a] Benzo(a)pyrene [b] Benzyl alcohol Bis(2-ethylhexyl)phthalate [c] Chrysene [c] Fluoranthene 2-Methylnaphthalene [d] Phenanthrene [a] Pyrene TRPHs	0.43 0.77 0.56 0.43 0.42 0.075 0.92 0.49 0.35 0.18 0.56 ND	800 800 100 800 100,000 140 7.2 27,000 800 100 100 NA	190 190 54,000 190 100,000 180 62 30,000 190 54,000 54,000 54,000 NA	2.3E-03 4.1E-03 1.0E-05 2.3E-03 4.2E-06 NA NA 1.6E-05 0.00 3.3E-06 1.0E-05 NA	6.1 6.1 NA 0.61 20,000 320 610 2,600 NE NA 2,000 4,100	26 26 NA 2.6 200,000 1,400 2,600 27,000 NE NA 20,000 18,000
	Hazard Index				0.02		

All Units in milligrams per kilogram (mg/kg).

Arizona Department of Environmental Quality 2000 Soil Remediation Level.
Base-neutral and extractable acids.
Excess lifetime cancer risk.
Hazard quotient.
Not Applicable
Non-detect
Preliminary Remediation Goal.
Total recoverable petroleum hydrocarbons.
U.S. Environmental Protection Agency

[a] Pyrene is used as a surrogate.
[b] The PRG is based on naphthalene non-carcinogenic effects. The hazard quotient presented under non-cancer effects is actually just the ratio of the PRG to the soil concentration.

[c] The PRG is based on carcinogenic effects so a non-cancer hazard quotient is not calculated.[d] Napthalene is used as a furrogate for comparison to the Region IX PRG.

*Risk and Hazards are based upon 2000 Industrial PRGs

Surface soil samples were taken from a depth of 0 to 2 feet below ground surface. Subsurface soil samples were taken from a depth of 2 to 10 feet below ground surface. Post remediation concentrations were taken from the maximum concentration of the constituent within a medium from a depth of 0 through 10 feet from Figures 2 and 4 of the RCRA Facility Investigation Summary Report Facility #993 (PSC ST-18) by Geraghty & Miller, Inc., date December 19, 2000.

Data Evaluation

Soil samples collected in 1992 had concentrations of TRPH ranging up to 17,000 mg/kg. BTEX, 1,1-DCA, 1,1,2,2-tetrachloroethane, TCE, and PCE were also detected. Lead was detected above the background UTLs. The highest concentration of lead was 32 mg/kg. The risk assessment conducted for the site concluded that the most conservative ELCR and HI were 3 x 10^{-6} and 0.1, respectively. The purpose of the institutional controls at the site are to ensure the integrity of the concrete cap.

Assessment

Is the remedy functioning as intended by decision documents?

The objective of the remedial action was to prevent exposure to the contaminated soil. By maintaining the integrity of the cap, implementing surface controls, and continuing groundwater monitoring, the remedy is functioning as intended.

Are the assumptions used at the time of the remedy selection still valid?

The assumptions made at the time of the remedy selection are still valid.

Has any other information come up that could question the protectiveness of the remedy?

No additional information has come to light that could call into question the protectiveness of the remedy.

Issues

According to the ROD, a deed restriction should be placed on the site as part of the institutional controls to prevent removal of the cap and excavation of the soil. A Declaration of Environmental Use Restriction (DEUR) for PSC ST-18 has been filed with the ADEQ.

Recommendations and Follow-up Activities

The following are the recommendations or follow-up activities suggested at this time:

- According to the ROD, a deed restriction needs to be placed on the site as part of the institutional controls. A DEUR has been filed with the ADEQ to restrict residential land use in the future.
- As part of the inspection, it was noted the concrete cap is in good condition and is well maintained by Luke AFB and that maintenance of cracks in the concrete is less critical given the presence of the geomembrane layer. The cap will continue to be inspected annually.
- MW-114 and MW-122 will be monitored annually for VOCs and the results evaluated as part of the next five-year review.

Protectiveness Statement

The remedy at PSC ST-18 Facility 993 currently protects human health and the environment because the cap prevents exposure in the short term. However, in order for the remedy to be protective in the long term, a DEUR will be placed at the site to ensure long-term protectiveness. In addition, MW-114 and WM-122 will be monitored for VOCs and evaluated as part of the next five-year review.

6.3.2 PSC DP-23 Old Surface Impoundment West of Facility 993

Remedial Actions

Remedy Selection

In accordance with the OU-2 ROD, the selected remedy for the southern portion of PSC DP-23 consisted of the following:

- Excavation
- Ex-situ biological treatment
- On-site disposal
- Monitoring

The selected remedy for the northern portion of PSC DP-23 was no action.

Remedy Implementation

Southern Portion of PSC DP-23

The implementation of the remedy for the southern portion of PSC DP-23 is summarized as follows:

- In 1995, a preliminary soil survey was conducted by Environmental Chemical Corporation (ECC) to determine the exact extent of the impacted soil³⁴.
- Based on the results of the preliminary survey, the USACE requested a more detailed site characterization.
- Additional samples were collected and ECC used the results to determine the area for excavation.
- ECC constructed berms to contain impacted soil and divert surface runoff away from the excavation areas.
- An on-site containment cell was constructed and lined with a 40 mil HDPE liner and topped with approximately six inches of native soil to protect the liner.
- ECC excavated 625 cubic yards of soil, which was transferred to the containment cell.
- The soil was mixed with alfalfa, manure, wood chips and green waste according to ratios established by Woods End Research Laboratory during computer optimization studies to form a compost.
- Composite samples were collected to determine the baseline levels of benzo(a)pyrene.
- The soil was tilled and watered and monitored daily for temperature, oxygen, and moisture content.
- Interim sampling was conducted after the compost had been processed for 60 days from the same locations as the baseline samples with a final sampling event was conducted after 120 days.
- The interim sample results indicated one quarter of the soil remained above PRGs.
- The soil was composted for an additional 60 days.

³⁴ Environmental Chemical Corporation, 1997. Closure Report, Site DP-23, Soil Composting at Luke Air Force Base, Arizona, August 27, 1997.

- Samples collected after the additional 60 days of composting indicated benzo(a)pyrene concentrations were below PRGs.
- Upon completion of the remediation, the site was restored to its original condition and the liner was disposed at a local landfill.
- The PAH concentrations were compared to analytical detection limits and not PRGs. This was done because the evaluation of risk determined that the risk associated with the higher concentrations was acceptable based on the potential for exposure of a base worker or construction worker to PAH at DP-23 south.
- The site was closed based on completion of remediation

Northern Portion of PSC DP-23

While the extent of impacted soil was being determined for the southern portion of the site it became apparent that the contamination extended northward. The implementation of the remedy for the northern portion of PSC DP-23 is summarized as follows:

- In 1996, Dames & Moore performed a risk-based assessment for the northern portion of PSC DP-23 based on two rounds of soil sampling conducted to characterize soil impacts at the site³⁵ and the results of samples collected by ECC in 1995. The extent of contamination to the north was never fully determined due to the tarmac at the northern most reaches of the site. Due to mission impact, no samples were collected from under the tarmac.
- Dames & Moore used the 1996 EPA Region IX PRG tables for soil to calculate the potential risk. Dames & Moore concluded that over the entire extent of the site, the predicted risk associated with exposure to carcinogens from PAHs in the surface soil was 1 x 10⁻⁵ and that predicted risks associated with exposure to subsurface soils ranged from 6 x 10⁻⁶ to 2 x 10⁻⁵.
- These risks calculated by Dames & Moore were within the acceptable range of 1×10^{-6} to 1×10^{-4} for industrial sites according to EPA and ADEQ standards.
- Although Dames & Moore did not recommend soil remediation, they did recommend a VEMUR be implemented on the site.
- In 2001, a DEUR was filed with the ADEQ.

³⁵ Dames & Moore, 1998. Final Site DP-23 Phase II Remedial Design Report, Luke Air Force Base, Arizona, April 1998.

System Operation and Maintenance

No operation and maintenance was required for the remedy selected. Remedial costs for the southern portion of PSC DP-23 were \$735,805. The cost of the risk-based assessment for the northern portion of PSC DP-23 was \$149,159.

Progress Since the Last Five-year Review

This is the first five-year review conducted for the site.

Five-year Review Findings

Site Inspection

There were no comments during the site inspection of PSC DP-23. No changes in land use were observed since implementation of the remedy. Photographs of PSC DP-23 taken as part of the site inspection are in Appendix E.

Changes to Standards

Southern Portion of PSC DP-23

A comparison of exposure point concentrations in sub-surface soils utilizing post-remediation data and USEPA Region IX PRGs is in Tables 6-15. The comparison indicates the southern portion of PSC DP-23 is within the acceptable risk range.

Northern Portion of PSC DP-23

A comparison of exposure point concentrations in surface soils and USEPA Region IX PRG's for a base worker and surface and sub-surface soils and USEPA Region IX PRG's for an excavation worker are in Tables 6-16 and 6-17, respectively. The comparison indicates the northern portion of PSC DP-23 is within the acceptable risk range for industrial land use and outside the acceptable risk range for residential land use. A DEUR for this portion of the site has been filed with the ADEQ to provide long-term protectiveness.

Changes in Exposure Pathways, Toxicity and Other Contaminant Characteristics

There were no changes in exposure pathways, toxicity, or other contaminant characteristics.

Data Evaluation

Southern Portion of PSC DP-23

The impacted soil at the site was remediated. Post-remediation soil samples collected from the walls and floors of the excavation indicate the site has been remediated to residential standards.

POST REMEDIATION CONCENTRATIONS IN SOIL AND USEPA REGION IX PRGS FOR SURFACE SOIL (BASE WORKER) PSC DP-23 (NORTHERN PORTION) Luke Air Force Base, Arizona

	Post Remediation Concentration	USEPA I Industrial P	Region IX RG (mg/kg)	*Risk or	ADEQ SRL (mg/kg)	
Constituent	(mg/kg)	1996	2000	Hazard	Residential	Non-Residential
Cancer Effects				ELCR		
BNAs						
Benzo(a)anthracene	35.0	2.6	2.9	1.2E-05	6.1	26
Benzo(a)pyrene	35.0	0.26	0.29	1.2E-04	0.61	2.6
Benzo(b)fluoranthene	57.0	2.6	2.9	2.0E-05	6.1	26
Benzo(k)fluoranthene	18.0	26	29	6.2E-07	61	260
Chrysene	28.0	7.2	62	4.5E-07	610	2,600
Dibenzo(a,h)anthracene	2.5	0.26	0.29	8.6E-06	0.61	26
Indeno(1,2,3-cd)pyrene	31.0	0.61	0.62	5.0E-05	6.1	26
Total Site Risk				2.1E-04		
Non-Cancer Effects				HQ		
BNAs						
Acenaphthene	1.1	110	38,000	2.9E-05	3,900	41,000
Acenaphthylene	<5	NA	NA	NA	NA	NA
Anthracene	3.50	5.7	100,000	3.5E-05	20,000	200,000
Benzo(g,h,i)perylene [a]	33.00	100	54,000	6.1E-04	NA	NA
Fluoranthene	33	27,000	30,000	1.1E-03	2,600	27,000
Fluorene	5	90	33,000	1.5E-04	2,600	27,000
Napthalene	7.6	240	190	4.0E-02	2,600	27,000
Phenanthrene [a]	14	100	54,000	2.6E-04	NA	NA
Pyrene	40	100	54,000	7.4E-04	2,000	20,000
Hazard Index				0.04		

All Units in milligrams per kilogram (mg/kg).

ADEQ SRL	Arizona Department of Environmental Quality 2000 Soil Remediation Level.
BNAs	Base-neutral and extractable acids.
ELCR	Excess lifetime cancer risk.
HQ	Hazard quotient.
NA	Not available.
NC	Not carcinogenic.
PRG	Preliminary Remediation Goal.
USEPA	U.S. Environmental Protection Agency.

[a] Pyrene is used as a surrogate.

*Risk and Hazard are based upon 2000 Industrial PRGs

Surface soil samples were taken from a depth of 0 to 3 feet below ground surface. Subsurface soil samples were taken from a depth of 3 to 10 feet below ground surface.

The cancer effect concentrations were based on the maximum exposure concentrations in the surface soil. These concentrations came from one sample, which contained the highest concentrations of BNAs, and exceed the risk value of 1.0E-04. The remaining sample concentrations for BNAs do not exceed the risk value of 1.0E-04. Post remediation information was obtained from the Final Site DP-23 Phase II Remedial Design Report by Dames & Moore, dated April 10, 1998.

POST REMEDIATION CONCENTRATIONS IN SOIL AND USEPA REGION IX PRGS FOR SURFACE AND SUBSURFACE SOIL (BASE AND EXCAVATION WORKER) PSC DP-23 (SOUTHERN PORTION) Luke Air Force Base, Arizona

	Post Remediation Concentration	USEPA F Industrial P	Region IX RG (mg/kg)	*Risk or	ADEQ SRL (mg/kg)	
Constituent	(mg/kg)	1996	2000	Hazard	Residential	Non-Residential
Cancer Effects				ELCR		
BNAs						
Benzo(a)pyrene	0.508	0.26	0.29	1.8E-06	0.61	2.6
Benzo(b)fluoranthene	ND	2.6	2.9	NA	6.1	26
Indeno(1,2,3)pyrene	ND	0.61	0.62	NA	6.1	26
Total Site Risk				1.8E-06		
Non-Cancer Effects				HQ		
BNAs						
Benzo(a)pyrene	0.508	0.26	0.29	NA	0.61	2.6
Benzo(b)fluoranthene	ND	2.6	2.9	NA	6.1	26
Indeno(1,2,3)pyrene	ND	0.61	0.62	NA	6.1	26
Hazard Index				NA		

All Units in milligrams per kilogram (mg/kg).

ADEQ SRL	Arizona Department of Environmental Quality 2000 Soil Remediation Level.
BNAs	Base-neutral and extractable acids.
ELCR	Excess lifetime cancer risk.
HQ	Hazard quotient.
NA	Not available.
NC	Not carcinogenic.
ND	Non-detect.
PRG	Preliminary Remediation Goal.
USEPA	U.S. Environmental Protection Agency.

[a] Pyrene is used as a surrogate.

*Risk and Hazard are based upon 2000 Industrial PRGs

Soil samples were taken from depths of 0 to 2 feet, 4 to 6 feet, and 8 to 10 feet below ground surface. Post remediation concentration data was obtained from the Closure Report, Site DP-23, Soil Composting, by Environmental Chemical Corporation, dated August 27, 1997.

POST REMEDIATION CONCENTRATIONS IN SOIL AND USEPA REGION IX PRGS FOR SURFACE AND SUBSURFACE SOIL (EXCAVATION WORKER) PSC DP-23 (NORTHERN PORTION) Luke Air Force Base, Arizona

	Post Remediation Concentration	USEPA I Industrial P	Region IX RG (mg/kg)	*Risk or	ADE (m	EQ SRL ug/kg)
Constituent	(mg/kg)	1996	2000	Hazard	Residential	Non-Residential
Cancer Effects				ELCR		
BNAs						
Benzo(a)anthracene	8.11	2.6	2.9	2.8E-06	6.1	26
Benzo(a)pyrene	8.03	0.26	0.29	2.8E-05	0.61	2.6
Benzo(b)fluoranthene	13.8	2.6	2.9	4.8E-06	6.1	26
Benzo(k)fluoranthene	5.1	26	29	1.8E-07	61	260
Chrysene	6.8	7.2	62	1.1E-07	610	2,600
Dibenz(a,h)anthracene	8.0	0.26	0.29	2.8E-05	0.61	26
Indeno(1,2,3-cd)pyrene	7.3	0.61	0.62	1.2E-05	6.1	26
Total Site Risk				7.5E-05		
Non-Cancer Effects				HQ		
BNAs						
Acenaphthene	1.6	110	38,000	4.2E-05	3,900	41,000
Acenaphthylene	<1.6	NA	NA	NA	NA	NA
Anthracene	9.8	5.7	100,000	9.8E-05	20,000	200,000
Benzo(g,h,i)perylene [a]	145	100	54,000	2.7E-03	NA	NA
Fluoranthene	110	27,000	30,000	3.7E-03	2,600	27,000
Fluorene	1.5	90	33,000	4.5E-05	2,600	27,000
Napthalene	17.9	240	190	9.4E-02	2,600	27,000
Phenanthrene [a]	44.2	100	54,000	8.2E-04	NA	NA
Pyrene	147	100	54,000	2.7E-03	2,000	20,000
Hazard Index				0.1		

All Units in milligrams per kilogram (mg/kg).

ADEQ SRL	Arizona Department of Environmental Quality 2000 Soil Remediation Level.
BNAs	Base-neutral and extractable acids.
ELCR	Excess lifetime cancer risk.
HQ	Hazard quotient.
NA	Not available.
NC	Not carcinogenic.
PRG	Preliminary Remediation Goal.
USEPA	U.S. Environmental Protection Agency.

[a] Pyrene is used as a surrogate.

*Risk and Hazard are based upon 2000 Industrial PRGs

Surface soil samples were taken from a depth of 0 to 3 feet below ground surface.

Subsurface soil samples were taken from a depth of 3 to 10 feet below ground surface.

Post remediation information was obtained from the Final Site DP-23 Phase II Remedial Design Report by Dames & Moore, dated April 10, 1998.

The northern portion of DP-23 is capped and affected soils are inaccessible to base and construction workers, therefore there is no exposure to the impacted soils.

Northern Portion of PSC DP-23

No remedial action was performed on this site. A risk-based assessment was conducted by Dames & Moore³⁶ and concluded that the potential risk from exposure to the carcinogenic PAHs was between 1×10^{-6} and 1×10^{-4} . The potential risk meets the acceptable ranges for industrial sites but does not meet the acceptable limit for residential sites. Therefore, a DEUR should be implemented on the site. The risk-based assessment used 1996 PRGs, which are more stringent than the 2000 PRGs.

Assessment

Is the remedy functioning as intended by decision documents?

The objective of the remedial action established in the OU-2 ROD was to clean up impacted soil in the southern portion of PSC DP-23. Since the soil was successfully remediated to residential standards, the remedy is considered protective.

Are the assumptions used at the time of the remedy selection still valid?

The assumptions made at the time of the remedy selection are still valid.

Has any other information come up that could question the protectiveness of the remedy?

No additional information has come to light that could call into question the protectiveness of the remedy.

Issues

The northern portion of PSC DP-23 requires a DEUR to provide long-term protectiveness.

³⁶ Dames & Moore, 1998. Final DP-23 Phase II Remedial Desgn Report, Luke Air Force Base, Arizona.

Recommendations and Follow-up Activity

The following are the recommendations or follow-up activities suggested at this time:

• A DEUR needs to be finalized for the northern portion of PSC DP-23 since the site was not remediated to residential standards.

Protectiveness Statement

The remedy for the southern portion of PSC DP-23 is protective of human health and the environment. To ensure conditions for the northern portion of PSC DP-23 are protective of human health and the environment in the long-term, a DEUR has been filed with the ADEQ.

6.4 Groundwater Review

A comparison of exposure point concentrations in groundwater (maximum concentrations for the period of record) and USEPA Region IX 2000 PRGs for tap water and ADEQ aquifer water quality standards are in Tables 6-18 through 6-26, respectively, for the following PSCs:

- RW-02
- DP-05
- FT-06
- FT-07
- ST-18
- SD-20
- SD-21
- SD-38
- SS-42

The comparison indicates groundwater is within the acceptable risk range for applicable PSCs.

MAXIMUM EXPOSURE POINT CONCENTRATIONS IN GROUNDWATER AND USEPA REGION IX PRGS FOR PSC RW-02 (MW-115; MW-116; MW-124) Luke Air Force Base, Arizona

Constituent	Exposure Point Concentration (mg/L)	USEPA Region IX Tap Water PRG (mg/L) 2000	*Risk or Hazard	ADEQ 2000 Aquifer Water Qualtiy Standards (mg/L)
Cancer Effects			ELCR	
Inorganics				
Arsenic	0.017	0.000045	3.8E-04	0.05
Barium	0.071	NC	NC	2
Chromium	0.058	NC	NC	0.1
Copper	0.276	NC	NC	NA
Lead	0.018	NC	NC	0.05
Nickel	0.042	NC	NC	NA
Zinc	0.86	NC	NC	NA
Total Site Risk			4E-04	
Non-Cancer Effects			HQ	
Inorganics				
Arsenic	0.017	0.011	1.5E+00	0.05
Barium	0.071	2.6	2.7E-02	2
Chromium [a]	0.058	0.11	5.3E-01	0.1
Copper	0.276	1.4	2.0E-01	NA
Lead	0.018	NA	-	0.05
Nickel [b]	0.042	0.73	5.8E-02	NA
Zinc	0.86	11	7.8E-02	NA
Hazard Index			2	

ADEQ	Arizona Department of Environmental Quality.
ELCR	Excess lifetime cancer risk.
HQ	Hazard quotient.
NA	Not available.
NC	Non-Carcinogenic.
PRG	Preliminary Remediation Goal.
USEPA	U.S. Environmental Protection Agency.

[a] PRG is based on chromium VI non-carcinogenic effects.

[b] The PRG is based on nickel (soluble salts).

*Risk and Hazards are based upon 2000 Tap Water PRGs

MW-115 was abandoned on 2/7/95

MW-116 is to be abandoned.

MW-124 is to be abandoned.

(The metals in this table are based upon the maximum range of detects in the Occurrence of Constituents Detected in Groundwater Samples at PSC RW-02 table in the Basewide Risk Assessment. The analysis of the latest sampling event conducted on May 12, 1998, indicated the constituents analyzed were at non-detect levels.)

MAXIMUM EXPOSURE POINT CONCENTRATIONS IN GROUNDWATER AND USEPA REGION IX PRGS FOR PSC DP-05 (MW-104; MW-105; MW-106) Luke Air Force Base, Arizona

Constituent	Exposure Point Concentration (mg/L)	USEPA Region IX Tap Water PRG (mg/L) 2000	*Risk or Hazard	ADEQ 2000 Aquifier Water Qualtiy Standards (mg/L)
Cancer Effects			ELCR	
Inorganics				
Arsenic	0.014	0.000045	3.1E-04	0.05
Barium	0.27	NC	NC	2
Chromium	0.034	NC	NC	0.1
Copper	0.04	NC	NC	NA
Lead	0.017	NC	NC	0.05
Zinc	1.2	NC	NC	NA
Total Site Risk			3E-04	
Non-Cancer Effects			HQ	
Inorganics				
Arsenic	0.014	0.011	1.3E+00	0.05
Barium	0.27	2.6	1.0E-01	2
Chromium [a]	0.034	0.11	3.1E-01	0.1
Copper	0.04	1.4	2.9E-02	NA
Lead	0.017	NA	NA	0.05
Zinc	1.2	11	1.1E-01	NA
Hazard Index			2	

ADEQ	Arizona Department of Environmental Quality
ELCR	Excess lifetime cancer risk.
HQ	Hazard quotient.
NA	Not available.
NC	Non-Carcinogenic.
PRG	Preliminary Remediation Goal.
USEPA	U.S. Environmental Protection Agency.

[a] PRG is based on chromium VI non-carcinogenic effects.

*Risk and Hazards are based upon 2000 Tap Water PRGs

MW-104 was abandoned.

MW-105 was abandoned.

MW-106 was abandoned.

(The metals in this table are based upon the maximum range of detects in the Occurrence of Constituents Detected in Groundwater Samples at PSC DP-05 table in the Basewide Risk Assessment. The analysis of the latest sampling event conducted on May 12, 1998, indicated the constituents analyzed were at non-detect levels.)

MAXIMUM EXPOSURE POINT CONCENTRATIONS IN GROUNDWATER AND USEPA REGION IX PRGS FOR PSC FT-06 (MW-107; MW-108) Luke Air Force Base, Arizona

	Exposure Point	USEPA Region IX Tap Water PRG (mg/L)	*Risk or	ADEO 2000 Aquifier Water
Constituent	(mg/L)	2000	Hazard	Qualtiy Standards (mg/L)
Cancer Effects			ELCR	
Inorganics				
Arsenic	0.016	0.000045	3.6E-04	0.05
Barium	0.214	NC	NC	2
Chromium	0.054	NC	NC	0.1
Copper	0.019	NC	NC	NA
Lead	0.01	NC	NC	0.05
Nickel	0.022	NC	NC	NA
Selenium	0.011	NC	NC	0.05
Zinc	2.05	NC	NC	NA
Total Site Risk			4E-04	
Non-Cancer Effects			HQ	
Inorganics				
Arsenic	0.016	0.011	1.5E+00	0.05
Barium	0.214	2.6	8.2E-02	2
Chromium [a]	0.054	0.11	4.9E-01	0.1
Copper	0.019	1.4	1.4E-02	NA
Lead	0.01	NA	-	0.05
Nickel [b]	0.022	0.73	3.0E-02	NA
Selenium	0.011	0.18	6.1E-02	0.05
Zinc	2.05	11	1.9E-01	NA
Hazard Index			2	

ADEQ	Arizona Department of Environmental Quality
ELCR	Excess lifetime cancer risk.
HQ	Hazard quotient.
NA	Not available.
NC	Non-Carcinogenic.
PRG	Preliminary Remediation Goal.
USEPA	U.S. Environmental Protection Agency.

[a] The PRG is based on chromium VI non-carcinogenic effects.

[b] The PRG is based on nickel (soluble salts).

*Risk and Hazards are based upon 2000 Tap Water PRGs

MW-107 is abandond.

MW-108 is to be abandoned.

(The metals in this table are based upon the maximum range of detects in the Occurrence of Constituents Detected in Groundwater Samples at PSC FT-06 table in the Basewide Risk Assessment. The analysis of the latest sampling event conducted on June 6, 1996, indicated the constituents analyzed were at non-detect levels.)

MAXIMUM EXPOSURE POINT CONCENTRATIONS IN GROUNDWATER AND USEPA REGION IX PRGS FOR PSC FT-07 (MW-109; MW-110; MW-111; MW-118; MW-123) Luke Air Force Base, Arizona

Constituent	Exposure Point Concentration (mg/L)	USEPA Region IX Tap Water PRG (mg/L) 2000	*Risk or Hazard	ADEQ 2000 Aquifier Water Qualtiy Standards (mg/L)
Cancer Effects			ELCR	
Inorganics				
Barium	0.32	NC	NC	2
Chromium	0.024	NC	NC	0.1
Copper	0.032	NC	NC	NA
Lead	0.008	NC	NC	0.05
Zinc	1.07	NC	NC	NA
Total Site Risk			NC	
Non-Cancer Effects			HQ	
Inorganics				
Barium	0.32	2.6	1.2E-01	2
Chromium [b]	0.024	0.11	2.2E-01	0.1
Copper	0.032	1.4	2.3E-02	NA
Lead	0.008	NA	-	0.05
Zinc	1.07	11	9.7E-02	NA
Hazard Index			0.5	

ADEQ	Arizona Department of Environmental Quality
ELCR	Excess lifetime cancer risk.
HQ	Hazard quotient.
NA	Not available.
NC	Non-Carcinogenic.
PRG	Preliminary Remediation Goal.
USEPA	U.S. Environmental Protection Agency.

[a] Compund was detected but reported value is estimated.

[b] The PRG is based on chromium VI non-carcinogenic effects.

*Risk and Hazards are based upon 2000 Tap Water PRGs

MW-109 is to be abandoned.

MW-110 was abandoned 2/7/95.

MW-111 was abandoned 11/21/96.

MW-118 is to be abandoned.

MW-123 is to be abandoned.

(The metals in this table are based upon the maximum range of detects in the Occurrence of Constituents Detected in Groundwater Samples at PSC FT-07 table in the Basewide Risk Assessment. The analysis of the latest sampling event conducted on May 19, 1999, indicated the constituents analyzed were at non-detect levels.)

MAXIMUM EXPOSURE POINT CONCENTRATIONS IN GROUNDWATER AND USEPA REGION IX PRGS FOR PSC ST-18 (MW-2; MW-3; MW-4; MW-5; MW-114; MW-122) Luke Air Force Base, Arizona

	Exposure Point Concentration	USEPA Region IX Tap Water PRG (mg/L)	*Risk or	ADEQ 2000 Aquifier Water
Constituent	(mg/L)	2000	Hazard	Quality Standards (mg/L)
Cancer Effects			ELCR	
VOCs				
cis-1,2-Dichloroethene	0.00094	NC	NC	0.07
Trichloroethene	0.002	0.0016	1.3E-06	0.005
Tetrachloroethene	0.00095	0.0011	8.6E-07	0.005
Inorganics				
Arsenic	0.019	0.000045	4.2E-04	0.05
Barium	0.26	NC	NC	2
Chromium	0.12	NC	NC	0.1
Copper	0.13	NC	NC	NA
Lead	0.026	NC	NC	0.05
Nickel	0.051	NC	NC	NA
Selenium	0.006	NC	NC	0.05
Zinc	8.7	NC	NC	NA
Total Site Risk			4E-04	
Non-Cancer Effects			HQ	
NOCo				
vocs	0.00094	61	1.5E.05	0.07
Trichloroethene	0.00094	0.0016	1.3E+00	0.005
Tetrachloroethene	0.0002	0.0010	8.6E-01	0.005
retuentoroethene	0.00075	0.0011	0.01 01	0.000
Inorganics				
Arsenic	0.019	0.011	1.7E+00	0.05
Barium	0.26	2.6	1.0E-01	2
Chromium [c]	0.12	0.11	1.1E+00	0.1
Copper	0.13	1.4	9.3E-02	NA
Lead	0.026	NA		0.05
Nickel [d]	0.051	0.73	7.0E-02	NA
Selenium	0.006	0.18	3.3E-02	0.05
Zinc	8.7	11	7.9E-01	NA

ADEQ	Arizona Department of Environmental Quality
ELCR	Excess lifetime cancer risk.
HQ	Hazard quotient.
NA	Not available.
NC	Non-Carcinogenic.
PRG	Preliminary Remediation Goal.
USEPA	U.S. Environmental Protection Agency.

[a] Compound was detected but reported value is estimated.

[b] The PRG is based on bromodichloromethane carcinogenic effects, a non-carcinogenic value is not available.

[c] The PRG is based on chromium VI non-carcinogenic effects.

[d] The PRG is based on nickel (soluble salts).

*Risk and Hazards are based upon 2000 Tap Water PRGs

MW-2 was abandoned 10/1993.

MW-3 is active.

MW-4 was abandoned 10/4/94.

MW-5 is active.

MW-114 is active.

MW-122 is active.

(The metals in this table are based upon the maximum range of detects in the Occurrence of Constituents Detected in Groundwater Samples at PSC ST-18 table in the Basewide Risk Assessment and the analysis of the latest sampling event conducted on October 25, 2001.

Arsenic background levels in the State of Arizona regularly exceed the PRG's. Arsenic is present at this site within background levels.

G:\ENV\PROJ\800\891\draftfinalfiveyearreview2.rtf

MAXIMUM EXPOSURE POINT CONCENTRATIONS IN GROUNDWATER AND USEPA REGION IX PRGS FOR PSC SD-20 (MW-102; MW-103; MW-112D; MW-112S; MW-113) Luke Air Force Base, Arizona

	Exposure Point Concentration	USEPA Region IX Tap Water PRG (mg/L)		ADEQ 2000 Aquifer Water Quality Standards
Constituent	(mg/L)	2000	*Risk or Hazard	(mg/L)
Cancer Effects			ELCR	
VOCs				
Bromodichloromethane	0.006	0.00018	3.3E-05	NA
**Chloroform	0.005	0.00016	3.1E-05	0.1
Inorganics				
Arsenic	0.026	0.000045	5.8E-04	0.05
Barium	0.47	NC	NC	2
Boron	0.23	NC	NC	NA
Chromium	0.11	NC	NC	0.1
Copper	0.17	NC	NC	NA
Lead	0.048	NC	NC	0.05
Nickel	0.071	NC	NC	NA
Zinc	1.66	NC	NC	NA
Total Site Risk			6E-04	
Non-Cancer Effects			HQ	
VOCs				
Bromodichloromethane	0.006	0.12	5.0E-02	NA
**Chloroform	0.005	0.00063	7.9E+00	0.1
Inorganics				
Arsenic	0.026	0.011	2.4E+00	0.05
Barium	0.47	2.6	1.8E-01	2
Boron	0.23	3.3	7.0E-02	NA
Chromium [b]	0.11	0.11	1.0E+00	0.1
Copper	0.17	1.4	1.2E-01	NA
Lead	0.048	NA	NA	0.05
Nickel [c]	0.071	0.73	9.7E-02	NA
Zinc	1.66	11	1.5E-01	NA
Hazard Index			12	

Arizona Department of Environmental Quality.
Excess lifetime cancer risk.
Hazard quotient.
Not available.
Non-Carcinogenic.
Preliminary Remediation Goal.
U.S. Environmental Protection Agency.

[a] Compound was detected but reported value is estimated.

[b] The PRG is based on chromium VI non-carcinogenic effects.

[c] The PRG is based on nickel (soluble salts).

*Risk and Hazards are based upon 2000 Tap Water PRGs

MW-102 was abandoned.

MW-103 was abandoned.

MW-112D is to be abandoned.

MW-112S is to be abandoned.

MW-113 is to be abandoned.

(The metals in this table are based upon the maximum range of detects in the Occurrence of Constituents Detected in Groundwater Samples at PSC SD-20 table in the Basewide Risk Assessment and the analysis of the latest sampling event conducted on May 19, 1999.)

**Chloroform is a known common laboratory contaminate. Chloroform was found in only one sampling event, May 19, 1999 in MW-113, from 1991 through 1999 from five different monitoring wells. Arsenic background levels in the State of Arizona regularly exceed the PRG's. Arsenic is present at this site within background levels.

G:\EVN\PROJ\800\891\draftfinalfiveyearreview2.rtf

MAXIMUM EXPOSURE POINT CONCENTRATIONS IN GROUNDWATER AND USEPA REGION IX PRGS FOR PSC SD-21 (MW-101) Luke Air Force Base, Arizona

Constituent	Exposure Point Concentration (mg/L)	USEPA Region IX Tap Water PRG (mg/L) 2000	- *Risk or Hazard	ADEQ 2000 Aquifer Water Quality Standards (mg/L)
Cancer Effects			ELCR	
Inorganics				
Arsenic	0.011	0.000045	2.4E-04	0.05
Barium	0.117	NC	NC	2
Boron	0.25	NC	NC	NA
Copper	0.092	NC	NC	NA
Lead	0.007	NC	NC	0.05
Zinc	0.5	NC	NC	NA
Total Site Risk			2E-04	
Non-Cancer Effects			HQ	
Inorganics				
Arsenic	0.011	0.011	1.0E+00	0.05
Barium	0.117	2.6	4.5E-02	2
Boron	0.25	3.3	7.6E-02	NA
Copper	0.092	1.4	6.6E-02	NA
Lead	0.007	NA	-	0.05
Zinc	0.5	11	4.5E-02	NA
Hazard Index			1	

ADEQ Anzona Department of Environmental Quar	ny.
ELCR Excess lifetime cancer risk.	
HQ Hazard quotient.	
NA Not available.	
NC Non-Carcinogenic.	
PRG Preliminary Remediation Goal.	
USEPA U.S. Environmental Protection Agency.	

*Risk and Hazards are based upon 2000 Tap Water PRGs MW-101 is active.

(The metals in this table are based upon the maximum range of detects in the Occurrence of Constituents Detected in Groundwater Samples at PSC SD-21 table in the Basewide Risk Assessment. The analysis of the latest sampling event conducted on June 4, 1996, indicated the constituents analyzed were at non-detect levels.)

MAXIMUM EXPOSURE POINT CONCENTRATIONS IN GROUNDWATER AND USEPA REGION IX PRGS FOR PSC SD-38 (MW-117) Luke Air Force Base, Arizona

	Exposure Point Concentration	USEPA Region IX Tap Water PRG (mg/L)	- *Risk or	ADEQ 2000 Aquifer Water Quality Standards
Constituent	(mg/L)	2000	Hazard	(mg/L)
Cancer Effects			ELCR	
Inorganics				
Barium	0.146	NC	NC	2
Copper	0.012	NC	NC	NA
Lead	0.003	NC	NC	0.05
Zinc	0.378	NC	NC	NA
Total Site Risk			NC	
Non-Cancer Effects			HQ	
Inorganics				
Barium	0.146	2.6	5.6E-02	2
Copper	0.012	1.4	8.6E-03	NA
Lead	0.003	NA	-	0.05
Zinc	0.378	11	3.4E-02	NA
Hazard Index			0.1	

All Units in milligrams per Liter (mg/L).

ADEQ	Arizona Department of Environmental Quality.
ELCR	Excess lifetime cancer risk.
HQ	Hazard quotient.
NA	Not available.
NC	Non-Carcinogenic.
PRG	Preliminary Remediation Goal.
USEPA	U.S. Environmental Protection Agency.

*Risk and Hazards are based upon 2000 Tap Water PRGs

MW-117 is to be abandoned.

(The metals in this table are based upon the maximum range of detects in the Occurrence of Constituents Detected in Groundwater Samples at PSC SD-38 table in the Basewide Risk Assessment. The analysis of the latest sampling event conducted on May 29, 1996, indicated the constituents analyzed were at non-detect levels.)

MAXIMUM EXPOSURE POINT CONCENTRATIONS IN GROUNDWATER AND USEPA REGION IX PRGS FOR PSC SS-42 (MW-119; MW-120; MW-121; MW-125) Luke Air Force Base, Arizona

	Exposure Point Concentration	USEPA Region IX Tap Water PRG (mg/L)	*Disla or	ADEQ 2000 Aquifer Water
Constituent	(mg/L)	2000 Hazard		Quality Standards (mg/L)
Cancer Effects			ELCR	
Inorganics				
Arsenic	0.007	0.000045	1.6E-04	0.05
Barium	0.139	NC	NC	2
Chromium	3.84	NC	NC	0.1
Copper	0.036	NC	NC	NA
Nickel	0.254	NC	NC	NA
Selenium	0.008	NC	NC	0.05
Zinc	3.09	NC	NC	NA
Total Site Risk			2E-04	
Non-Cancer Effects			HQ	
Inorganics				
Arsenic	0.007	0.011	6.4E-01	0.05
Barium	0.139	2.6	5.3E-02	2
Chromium [b]	3.84	0.11	3.5E+01	0.1
Copper	0.036	1.4	2.6E-02	NA
Nickel [c]	0.254	0.73	3.5E-01	NA
Selenium	0.008	0.18	4.4E-02	0.05
Zinc	3.09	11	2.8E-01	NA
Hazard Index			36	

All Units in milligrams per Liter (mg/L).

ADEQ	Arizona Department of Environmental Quality
ELCR	Excess lifetime cancer risk.
HQ	Hazard quotient.
NA	Not available.
NC	Non-Carcinogenic.
PRG	Preliminary Remediation Goal.
TPH	Total Petroleum Hydrocarbons
USEPA	U.S. Environmental Protection Agency.

[a] Compound was detected but reported value is estimated.

[b] The PRG is based on chromium VI non-carcinogenic effects.

[c] The PRG is based on nickel (soluble salts).

*Risk and Hazards are based upon 2000 Tap Water PRGs

MW-119 deteriorated.

MW-120 deteriorated.

MW-121 is active.

MW-125 deteriorated.

(The metals in this table are based upon the maximum range of detects in the Occurrence of Constituents Detected in Groundwater Samples at PSC SS-42 table in the Basewide Risk Assessment. The analysis of the latest sampling event conducted on May 16, 2000, indicated the constituents analyzed were at non-detect levels.)

Arsenic background levels in the State of Arizona regularly exceed the PRG's. Arsenic is present at this site within background levels.

6.5 Interviews

The following individuals were solicited for interviews by questionnaire as part of this five-year review:

- Belle Matthews, Luke AFB Project Manager
- Sean Hogan, EPA Project Manager
- Nancy Lou Minkler, ADEQ Project Manager
- Dan Salzler, Citizens Advisory Board (CAB) Community Co-Chairperson
- Joyce Clark, CAB member
- Martin Jeffries, CAB member

The only individuals who responded to the questionnaire were Belle Matthews, Nancy Lou Minkler and Martin Jeffries. Interview results for these individuals are in Appendix F.

In addition to solicitation of interviews by questionnaire, the following individuals were interviewed in person as part of the May 22, 2001 site inspection:

- Chris Christoffer, Luke AFB Environmental Analyst
- Sergeant Anthony Michels, Luke AFB Infrastructure Superintendent

Chris Christoffer and Sergeant Michels were interviewed relative to procedures that ensure compliance with the BGP and ICP. As part of these interviews, the BGP was reviewed and it was verified that the ICP had been implemented. Also verified were approval and record keeping procedures for digging permits relative to environmental constraints at Luke AFB.







SON

SHERR ö B

50

ÿ

FILE 423.011 Å PRJ#: 17:36 666 01/19/1

DATE:

FIGURE



-1rod\figures\BASE ings\az\luke\ou draw DWG 8 AZ000760 PRJ∰: 09:33 2001 07/24/:

SES

11-97 ND	5-12- ND	98									
W-1	19	3-96 7-17	7-07								
PH 11	IOJ TPH	26J DCP	1.0J								
-96	5-23-96	MW- 7-17-97	121	97 5-14	-98	11-0	5-98	5-20-99	5-16	-00	80201
	ND	DCP 1.4J BZ 1.8J EB 4.4J	DCP 1. DRO 13	7J DCP Ovr 8Z EB 2	1.8J 1.8 2.3J	DRO GRO BZ	630 200JB 17	ND	N	D	ND
25R		XYL 12 DRO' 970	1	XYL DRO GRO	6.9 120 44B	MC 0 PCE TOL	23 1.34JB 0.20J 36				
-01				<u> </u>		DRO GRO	61 250 490JB				
						BZ EB MC 0	19 25 40JB 0.25J				
	47. 47.	<u>MW-120</u>			_	TOL	38 64				
5 10	ND	2-8-96 ND	6-1-96 ND	7-17-97 DCP 1.7J	GR D	-05-9 0 270 CP 1.4	8 18 1				
				<u>.</u>		0L 0.11	J				
5 1 10	-31-95	2-9-96	6-11-96	IW-125	T 11-	13-97	5-14-	-08 11-0	4_081	5-20	- 99]
	ND	TPH 11J	ND	DCP 1.0J GRO 23J	DCF	• 1.2 J	DCP 1	.6J GRO MC 0	7.2 1.2J .21JB	N	D
				DCP 1.1J BZ 3.4J DRO 53J	DCF	י 1.3 J				N	⊵
							•				
995	2-2-9	6 6-6-90	MW-122	2	-98	11-03-	98 5-	-18-99 10)-25-0	n l	
ID	ND	ND	ND	ND)	MC 0.2	9J8	ND	ND		
98 30J 0J8 18J 6J	5-19-99 Chil 5.0 Brimo 6.0]									
7											
J											
Ţ											
.N.	ALY	TICA		ESU	JL	TS				FI	GURE
.N/	ALY	TIC	\L R	ESL	JL	TS				FI	gure 5-2

Appendix A

PSC – Detailed Descriptions and Background Information

PSC RW-02 Wastewater Treatment Annex

PSC RW-02 consists of a former 28-acre landfill at the Luke AFB wastewater treatment plant (WWTP) annex located north of Glendale Avenue and, two miles east of the main Base. The former landfill is located in the northwestern portion of the WWTP annex, adjacent to the western bank of the Agua Fria River. The site served as the Base's main landfill for the disposal of refuse from 1953 until 1970.

A small quantity of low-level radioactive electron tubes and dials were buried at the site in 1956. The radioactive material was encased in concrete and disposed in a pit 12 feet deep with 4 feet of concrete cover and 6 feet of earth cover. The radioactive material burial site is currently located within the boundaries of the Defense Reutilization Marketing Office (DRMO) storage yard.

During the OU-1 RI, two soil borings (SB-1 and SB-2) were advanced and sampled near the radiological waste containment structure to assess its integrity. The borings were located approximately 15 feet north and south of the radiological monument marker and advanced to a depth of 17 feet below ground surface (bgs). A third soil boring (SB-11) was also advanced 30 feet north of the radiological waste containment to assess background radiological conditions. Two soil samples were collected from each of the borings at depths between 10-12 feet and 15-17 feet bgs. The samples were submitted to IT laboratory and analyzed for total uranium, radium-226, radium-228, gross alpha, and gross beta.

Radiochemical analyses of soil samples adjacent to the monument were not significantly different from the background boring SB-11, and the results of the radiochemical analyses for all samples are within the background ranges for natural soils. Specifically, samples adjacent to the containment structure contained uranium concentrations of 0.3 to 1.0 micrograms/per gram (μ g/g). These concentrations are within the background range for natural geologic materials (up to 4.8 μ g/g). Radium-226 (alpha radiation emitters associated with the uranium decay series), and radium-228 (a beta radiation emitter associated with the thorium decay series) are also present at concentrations expected in natural soils.

Natural gamma ray geophysical logging was conducted in borings drilled adjacent to the monument (SB-1, SB-2) and at a nearby groundwater monitoring well (MW-115). The natural gamma logs displayed total gamma counts that ranged up to 235 American Petroleum Institute (PSI) units. Gamma radiation in natural geologic materials range from a few API units to more than 300 API units. Thus, no gamma radiation anomalies were noted.

While natural gamma logs and soil sampling results indicate that radioactive materials have not impacted soils adjacent to the containment structure, the mere presence of the low-level radioactive waste containment structure warrants concern. As a result, remedial alternatives were developed and evaluated for PSC RW-02.

PSC LF-03 Outboard Runway

PSC LF-03 consists of a former construction debris landfill located on the western side of the Base near the central part of the outboard runway, south of Taxiway F. The site occupies approximately 21 acres. The outboard runway currently covers 60 percent of the site. The remainder of the site consists of a bare low-lying area with sparse vegetation. The Base reportedly used the site for limited disposal of refuse from 1951 to 1953. Land filling operations at this site ceased when the outboard runway was constructed.

During the OU-1 RI, geophysical and soil gas surveys were conducted to define the landfill boundaries and to select locations for test pits. Six test pits were excavated and sampled to characterize its extent and contents. Two additional soil borings were advanced and sampled in August 1996 to collect additional VOC and BNA data for risk assessment purposes.

Numerous metallic wastes were unearthed at the central portion of PSC LF-03 during test pit excavation. Samples of the wastes collected from Test Pit TP-5 at depths of 8 feet bgs and a 7-8 feet bgs contained chromium at concentrations of 349 and 386 mg/kg, respectively. Because the metallic wastes containing elevated concentrations of chromium are buried and extend below the outboard runway, direct exposure is not likely under current land use scenarios.

The risk assessment concluded that the site does not present unacceptable health risks given its current land uses. However, long-term exposure in unacceptable health risks could result if the runways were removed and the site was developed for residential purposes. As a result, remedial alternatives were developed for the site.

PSC FT-07E East Portion of North Fire Training Area

PSC FT-07E is located in the northern portion of the Base, west of Fire Department Training Facility 1355. Fire training activities in the eastern portion of PSC FT-07E began in 1973 when the Base constructed three fire-training pits (FTPs). The two largest training pits were constructed with sprinkler systems to dispense flammable POL waste onto mock aircraft or similar structures. According to Base records, the three pits were active from 1973 until 1989. The two largest pits were designated as Fire Training Pit #3 (FTP-3) and Fire Training Pit #4 (FTP-4). The third pit was identified as Fire Training Pit #6 (FTP-6).

Luke AFB conducted a soil vapor extraction (SVE) removal action at fire training pits FTP-3 and FTP-4 from April 1992 through December 1992. Calculations indicate that over 14,000 pounds of contaminants were removed from the soil and destroyed by a thermal oxidizer treatment system. The objectives of the OU-1 RI at PSC FT-07E were to assess effectiveness of the removal action, to further evaluate the vertical extent of any constituents still remaining in the soils, and to assess the potential for groundwater impacts beneath the site. Fourteen soil borings were advanced and sampled at the two fire training pits where vapor extraction was performed (FTP-3 and FTP-4). Three soil borings were also advanced and samples at fire training pit FTP-6. Two groundwater monitoring wells (MW-118 and MW-123) were installed during the OU-1 investigation to assess groundwater quality at the site.

Soil sampling results indicated that residual hydrocarbon contamination was effectively reduced at depths greater than 16 feet bgs. Groundwater sampling results indicate the underlying groundwater resources have not been impacted. Vadose zone transport modeling also indicated that residual petroleum hydrocarbon contaminants in the soil would not leach to the underlying groundwater. However, relatively high concentrations (27,000 mg/kg) of total petroleum hydrocarbons (TPH) remained in the soils near the surface. While the current site conditions do not pose a threat to human health or the environment given the its current land use (military/industrial), unacceptable health risks would occur if the site were developed and used for residential purposes. For this reason, remedial alternatives were developed.

PSC SS-11 Former Outside Transformer Storage

PSC SS-11 consists of a 0.79-acre site located in the northeastern portion of Luke AFB, northeast of Facility 328 and west of Building 360. The Luke AFB exterior electric shop used the site prior to 1981 for temporary storage of out-of-service electrical transformers, some of which may have contained polychlorinated biphenyls (PCBs). Approximately 20-percent of the site is covered by bare ground with no vegetation, and the remaining 80-percent is covered with degraded asphalt which has been present for the past 40 years. The transformers were reportedly stored on the bare ground. The shallow soils at this PSC contain PCBs at low concentrations. The Basewide risk assessment concluded that these levels pose no risk to human health or the environment.

PSC OT-12 Old EOD Site

PSC OT-12 consists of a 15-acre former landfill area located between the outboard runway and the west perimeter road. The majority of the site lies in a low depression covered with exposed soil and grass. The exact dates of operation of the pit could not be determined, however, it was reportedly in existence in the early 1970s. The site is located just south of the EOD Demolition and Burn Facility #1047, which was constructed in 1963. The pit was probably excavated at that time to dispose of residue from the incineration or detonation of unused or outdated ordinance. Currently, all unexploded ordnance is taken to the Luke Air Force Base Range at Gila Bend for demolition and disposal. Prior to remedial investigations, this area was surveyed by air force explosion ordnance disposal technicians and found the site to be free of UneXploded Ordnance (UXO). The soils at this PSC contain TPH, PAHs, arsenic and beryllium. The Base-wide risk assessment concluded these levels pose no risk to human health or the environment.

PSC DP-13 Drainage Ditch Disposal Area

PSC DP-13 is located in the northwest corner of the Base (Figure 8). During the 1940s, this site was the location of a drainage ditch that was reportedly used for refuse disposal. The ditch was filled and covered when the Base was deactivated in 1946. Asphalt and concrete rubble stored in the northwest corner of the site was disposed in a burial pit in 1974. No known or suspected industrial-type wastes or hazardous wastes were disposed at this site. Currently, a majority of the site is covered with bare ground. The northern portion of the site is used as a bivouac area for preparedness training.

Objectives of the RI at PSC DP-13 were to define the boundaries of the former landfill and characterize its contents. Geophysical and soil gas surveys were conducted to define the landfill boundaries and to select locations for test pits. Fifteen test pits were excavated to characterize the extent and contents of the landfill. Ten soil borings were advanced to further define the vertical and lateral extent of constituents of potential concern detected in the test pit samples. In August 1996, three additional soil borings were advanced to collect supplemental VOC and BNA data for risk assessment purposes.

Test Pits TP-12 (located near the side of a maintained road within the bivouac area) intercepted an inactive underground utility line. A paint pail and dried paint residue were also observed in Test Pit TP-12. Wastes collected from that test pit at a depth of 5 feet bgs contained chromium at 15,900 mg/kg and lead at 36,000 mg/kg. Because these wastes are buried and the surface area is maintained, direct exposure is not likely under current land use scenarios. However, exposure to these buried wastes could result if excavation were to occur or if the site were developed for residential purposes.

For this reason, remedial alternatives were developed for PSC DP-13 as a protective measure.

PSC LF-14 Old Salvage Yard Burial Site

PSC LF-14 consists of a former landfill site located in the northeastern corner of the Base. In the 1940s, this site was part of the main drainage canal for the north end of the Base. The canal was abandoned when the drainage was changed in the 1950s. The abandoned canal may have been used as a landfill and was completely filled and covered by 1962. According to interviews with Base personnel, PCB-containing transformer fluids may have been disposed in the ditch in the northern portion of this site. The site is currently unpaved and covered with bare ground.

The objectives of the RI at PSC LF-14 were to define the boundaries of the former drainage ditch landfill and to characterize its content. Geophysical and soil gas surveys were conducted to define the landfill boundaries and to select locations for test pits. Phase II activities consisted of excavating four test pits and sampling 10 soil borings. Two additional soil borings were advanced in August 1996 to collect supplemental VOC and BNA data for risk assessment purposes.

Relatively high PCB concentrations (2,300 mg/kg) were detected at the site, however, the depth at which this concentration was detected was greater than 16 feet bgs and exposure is unlikely. Based on the results of the Base-wide risk assessment, contaminants identified at PSC LF-14 were not present at areas of potential exposure at concentrations high enough to cause adverse health effects under current land use scenarios. However, the concentrations of PCBs and chromium present in soils 0 to 16 feet bgs could theoretically cause adverse health affects in unlikely event that PSC LF-14 were developed for residential purposes in the future. For this reason, remedial alternatives were developed for the site.

PSC SS-17 Former Defense Property Disposal Office (DPDO) Yard

PSC SS-17 consists of the former DPDO yard facility located in the northeastern corner of Luke AFB and occupies approximately 13-acres. Forty percent of the site is paved with old asphalt and concrete pads and 60 percent is soil ground cover. During the 1950s and 1960s, hazardous materials and 55-gallon drums of industrial wastes were stored on the floor of the former DPDO building. The hazardous waste included spent thinners and strippers, paint, solvents, mercury-contaminated rages, and asbestos-containing material. In 1986, all wastes were shipped from the site for proper disposal

in California. Soil samples and samples of the concrete pad were collected in May 1986. None of the samples contained detectable concentrations of potential contaminants. The DPDO yard was listed as "closed" on September 21, 1988, with closure acknowledged by ADEQ on September 30, 1988. Despite its "closed" status, PSC SS-17 was included in the OU-1 RI. The soils at this PSC contain TPH, PCBs, arsenic and beryllium. The Base-wide risk assessment concluded these levels pose no risk to human health or the environment.

PSC SD-20 Oil/Water Separator Canal and Earth Fissures

PSC SD-20 consists of a drainage canal located on the southern side of Luke AFB. This unlined canal originates at the Oil/Water Separator 912, approximately 100-feet north of N Street, and extends southward. The 912 oil/water separator system serves two drainage systems, a 30-inch diameter system for the areas to the northwest and a 43-inch diameter system for an area to the northeast. In some instances during past storm events, stagnant oily water in the 30-inch diameter system overflowed into the oil/water separator canal. Recent upgrades to Luke AFB sewer system have eliminated the potential for additional discharges to the canal. Two earth fissures, apparently resulting from differential land subsidence, are present at the end of the drainage canal. The soils at PSC SD-20 contain TRPH, benzo(a)pyrene, arsenic, and beryllium at low concentrations. Groundwater samples collected at the site were found to contain TCE, arsenic and lead. The Base-wide risk assessment concluded these levels pose no risk to human health or the environment.

PSC SD-21 WWTP Effluent Canal

PSC SD-21 is located approximately 3-miles east of the Base, south of Glendale Avenue, adjacent to the west bank of the Agua Fria River. Prior to1997, treated effluent was discharged to this canal from the Base WWTP. The canal and associated wetlands comprise approximately 33-acres. The water in the canal is categorized as effluent dominated surface water according to the ADEQ. In 1997 effluent discharge to the canal was discontinued and discharge was piped to the new Luke AFB golf course for irrigation. The soils at PSC SD-21 contained BNAs, arsenic and beryllium. Sediment samples collected at this PSC contained arsenic and beryllium. Surface water samples collected at this PSC contained arsenic and lead. Samples collected from groundwater monitoring wells at the site contained arsenic and lead. The Basewide risk assessment concluded these levels pose no risk to human health or the environment.

PSC LF-25 Northwest Landfill

PSC LF-25 consists of an area formerly used for land filling and is located along the southwest boundary of the Base, between the west perimeter and the northwest runway. This narrow site occupies approximately 43-acres. Portions of PSC LF-25 are located immediately downrange of the Base skeet shooting range. Small, localized sections of the site were used as a landfill for construction debris in the past for an undetermined length of time, but it has not been used since 1989.

The objectives of the RI at PSC LF-25 were to define the boundaries of any former landfills and to characterize their content. During the OU-1 RI investigations, geophysical and soil gas surveys were conducted to define landfill boundaries and to select locations for test pits and soil borings.

Lead and antimony were detected in the surface soils adjacent to the skeet range at concentrations that could cause adverse health effects if prolonged exposure, such as excavation work or residential occupation, were to occur. The lead and antimony are present in the form of metal shot that was fired from the adjacent Base skeet shooting range. Metal shot continues to impact the site because the skeet range is still active. As a protective measure, remedial alternatives were developed for the site.

PSC SD-26 Hush House Canal

PSC SD-26 consists of a surface drainage canal located southeast of the Hush Houses. This canal merges with the Oil/Water Separator canal (PSC SD-20) at a location southwest of the Base Ammunition Storage Area. The combined flows discharge to an area of subsidence fissures. From the mid-1960s until 1993, the oil/water separators attached to the Hush Houses discharged directly into PSC SD-26. The oil/water separators were connected to the Base WWTP in 1993 and no longer discharge to the canal. Drainage from the runway and taxiway to the west, and most of the facilities for the 944th Tactical Air Group are also channeled into the Hush House canal. This site was not included in any IRP documents or reports. The soils at this PSC contain TRPH, benzo(a)pyrene, arsenic, and beryllium. The Base-wide risk assessment concluded these levels pose no risk to human health or the environment.

PSC LF-37 Northeast Landfill

PSC LF-37 is located in the northeast corner of Luke AFB and occupies approximately 11.9 acres. The site is currently unpaved except for the perimeter road. Luke AFB canal and a railroad spur are located adjacent to the north side of the site. This site was not investigated in any IRP documents or reports. The soils at PSC LF-37 contain TRPH, benzo(a)pyrene, arsenic, and beryllium. The Base-wide risk assessment concluded these levels pose no risk to human health or the environment.

PSC SD-38 Oil/Water Separator at Auto Hobby Shop

PSC SD-38 is located near the middle of the Base at the northwest corner of "D" Street and 3rd Street. The site consists of the former oil/water separator serving Building 248, the old Base Auto Hobby Shop. In March 1991, the SD-38 oil/water separator was inspected as part of the RCRA Facilities Assessment (RFA). It was discovered that this oil/water separator did not have a concrete bottom. This separator has since been removed. The Base for laboratory analysis submitted samples of the sludge from the bottom of the oil/water separator. Other than the sludge sampling, no previous investigations or environmental sampling was performed at this site prior to the OU-1 RI.

PSC SD-38 was originally assigned to the OU-2 investigation. Because OU-2 data indicated a deep soil impact and thus, a potential threat to groundwater, the site was reclassified as an OU-1 PSC. In May 1992, three soil borings were advanced and sampled to assess the nature and extent of any impacts at the site. During the OU-1 investigation, three soil borings were advanced and sampled to further evaluate the vertical and horizontal extent of any impact. A groundwater monitoring well (MW-117) was also installed and sampled at this time to evaluate groundwater quality at the site. In August 1996, one additional boring was advanced and sampled to collect supplemental VOC and BNA data for use in the risk assessment.

Soil samples collected directly beneath the former oil/water separator at a depth of 8 feet bgs contained TRPH at a concentration of 58,000 mg/kg. Based on the results of the Base-wide risk assessment, prolonged exposure to this concentration of TRPH could potentially cause adverse health affects. Because the soils containing elevated concentrations of TRPH are located at depth, direct exposure is not likely under current land use scenarios. However, prolonged exposure to the TRPH in the subsurface soils could result if the site were developed for residential purposes in the future. For this reason, remedial alternatives were developed for PSC SD-38.

PSC SD-39 Waste Discharge at the Old Lockheed Site
PSC SD-39 consists of two separate areas located near the northern end of the inboard runway. According to information obtained during the RFA conducted in March 1991, Luke AFB used the facilities in the area for aircraft ground equipment maintenance prior to 1964. Lockheed Aircraft Company occupied the facilities in the area from 1964 to 1982. Presently, the 405th TPW Maintenance Shop occupies the facilities. This site was identified as a PSC because of the lack of information on the composition and quantity of wastes released. The soils at PSC SD-39 contain TRPH and arsenic. The Base-wide risk assessment concluded these levels pose no risk to human health or the environment.

PSC OT-41 Skeet Range Canal

PSC OT-41 consists of Luke AFB Skeet Range. The site occupies approximately 3.27 acres located along the western side of Luke AFB near the southern end of the outboard runway in a triangular extension of the western boundary of Luke AFB. The paved west perimeter road comprises 5 percent of the site. The remainder of the site is desert soil and grass, except for an unlined irrigation canal, which passes through the site. The irrigation canal originates off Base and flows south along the west boundary and exits Luke AFB to the south. The site was identified as a PSC because lead shot from skeet shooting could potentially enter the canal and could be transported off of Luke AFB property. The area where lead shot and broken clay pigeons primarily fall is not within the boundary of PSC OT-41. Rather, the impact areas for the skeet range are further to the east of the irrigation canal within the boundaries of PSC LF-25. The boundary of PSC OT-41 was established as such because the irrigation canal was the point of interest for the investigation, not the impact area. Detected lead concentrations were all below the USEPA Region IX residential PRGs, which is 400 mg/Kg.

PSC SS-42 Bulk Fuels Storage Area

PSC SS-42 consists of a former leaking UST site located within the eastern portion of the bulk fuels storage area of Luke AFB. The leaking UST was part of an oil/water separator system that received condensate from the two large aboveground fuel tanks.

In March 1993, the leak detection system for the oil/water separator UST sounded, indicating a release had occurred. According to Base personnel, unusually heavy rains caused the soil around the UST to settle. The settling apparently caused the fill line to dislodge from the tank. In response, the oil/water separator and fiberglass UST were removed from service and excavated.

Environmental investigations by Environmental Engineering Consultants, Inc. (EEC) in response to the release from the oil/water separator UST began in March 1993. EEC advanced seven soil borings (UST-1 through UST-7) adjacent to the oil/water separator and leaking UST. Several of the borings advanced to define the horizontal extent of the impact contained detections of TRPH and benzene, toluene, ethylbenzene, and total xylenes in samples collected at depths between 70 feet and 160 feet bgs. Because of these unexpected detections, the horizontal extent of the impact was not defined by the seven borings advanced by EEC.

After review of the EEC data, the FFA parties added this site as a PSC in the CERCLA investigation. Because of the depth of the impact and magnitude of the release, the FFA parties agreed that additional investigations were warranted because of the potential for groundwater impact. Therefore, PSC SS-42 was assigned to OU-1 in August 1993.

The objectives of the RI at PSC SS-42 were to define the horizontal extent of the impact detected at the former oil/water separator UST, identify other potential sources of contamination at the site, and to assess the groundwater quality. Initial activities included conducting a geophysical survey to identify underground lines and utilities. A soil-gas scan was also conducted to assess the integrity of the underground distribution system and identify other potential sources of contamination. Sixteen soil borings were advanced and sampled to determine the horizontal and vertical extent of the impacts identified at the site. Four groundwater monitoring wells (MW-119 through MW-121, and MW-125) were also installed and sampled to evaluate the groundwater quality.

TPH and BTEX concentrations were detected at depths ranging from 10 to 160 feet bgs. The highest detected concentration of TPH was 33,900 mg/kg at a depth of 70 feet bgs. BTEX compounds were also detected at their highest concentrations at this depth. Based on the results of the Base-wide Risk Assessment, contaminants identified at PSC SS-42 were not present at areas of potential exposure at concentrations high enough to cause adverse health effects under current land use scenarios, or even under residential land use scenarios. However, results of the vadose zone transport modeling indicated that petroleum related contaminants (TPH and BTEX) detected in the soil could migrate to the underlying groundwater resources. For this reason, remedial alternatives were developed for the site.

The remedial alternative selected for PSC SS-42 in the OU-1 ROD was S-11 (In-situ Soil Vapor Extraction with Long-term Groundwater Monitoring). The remedial components included:

- Installing, operating, and maintaining a Soil Vapor Extraction (SVE) System.
- Monitoring soil and groundwater to confirm effectiveness and potential migration of the contaminants.

Because the Base-wide risk assessment concluded that the site did not pose a threat to human health, the only remedial objective was to reduce TPH and BTEX concentrations in the soil to levels that would no longer pose a threat to the underlying groundwater resources. More specifically, ARARs for the site (Arizona Soil Remediation Standards) required that soil remediation continue until contaminants remaining in the soil did not cause or threaten to cause a violation of Aquifer Water Quality Standards at a point of compliance. The Arizona Department of Environmental Quality Groundwater Protection Limit (GPL) screening model was to be used for determining whether residual contaminant concentrations in the soil were protective of groundwater.

Prior to the signing of the OU-1 ROD, the Base initiated a SVE removal action at PSC SS-42. The SVE removal action was performed using a highly modified internal combustion engine (ICE) to create necessary vacuum to draw the contaminated soil vapors from the subsurface. The ICE used the petroleum laden vapors as a fuel source, effectively treating the soil vapors prior to discharge. An on-board computer adjusted carburation to ensure emissions met air quality standards. Supplemental propane was used to fuel the engine as petroleum concentrations in the soil gas decreased. The SVE removal action continued through November 1998. Results of the removal action were documented in a series of letter reports produced by Parsons Engineering Science, Inc.

Data collected during the removal action between August 6, 1996, and November 2, 1998, indicated that over 399,514 pounds of total volatile hydrocarbons (TVH) (approximately 66,586 gallons) were removed from the subsurface soils. BTEX concentrations in the soil gas decreased from 4,590 parts per million by volume (ppmv) to 608 ppmv. This calculates to a reduction of soil gas BTEX concentrations by 87 percent.

In January 1999, following completion of the removal action, a confirmation boring was installed and sampled to evaluate the residual TPH and BTEX concentrations in soil. While TPH and BTEX were still detectable at reduced concentrations at depths between 50 and 180 feet bgs, residual TPH and BTEX were not detected in the upper 40 feet of soil.

OU-1 SiteDescriptions

Vadose zone transport modeling was conducted following completion of the SVE removal action to determine whether the remaining hydrocarbon contamination could migrate beneath the site and impact the groundwater resources. The ADEQ Groundwater Protection Limit (GPL) model was used for this evaluation. The model results indicated that the residual TPH and BTEX concentrations would not impact groundwater at concentrations above Arizona Aquifer Water Quality Standards, and furthermore, additional remediation was not needed to satisfy all applicable, relevant and appropriate requirements (ARARs). For this reason, the first part of the remedial alternative selected for the site (In-situ Soil Vapor Extraction) was not implemented.

The remedial alternative selected for PSC SS-42 also involved a groundwater monitoring program. At a minimum, groundwater monitoring was to be conducted at the site annually for 5 years.

PSC OT-04 OLD PERIMETER ROAD WASTE DISPOSAL SITE

PSC OT-04 consists of the old perimeter road waste application site. From 1951 until approximately 1970, petroleum, oil, and lubricant (POL) wastes generated during Base operations were sprayed on this section of road to control excessive dust. The old perimeter road disposal site begins approximately 200 feet north of Facility 1080 and runs southwest around the southern portion of the runway, then turns northeast before terminating immediately adjacent to Facility 1082. PSC OT-04 is unpaved and approximately 6800 feet long and 15 to 20 feet wide.

According to the IRP Phase I investigation, the total volume of POL waste generated at the Base prior to 1954 was relatively small and was disposed mainly through fire department training exercises. After 1954, the total volume of POL waste generated by the Base increased significantly. Available records show that up to 50,000 gallons per year of POL wastes were disposed on the perimeter road. The majority of the POL wastes disposed at the site consisted of contaminated JP-4, but may also have included aviation gasoline (AVGAS), diesel fuel, waste engine oils, and waste solvents. Other wastes disposed in this manner included wastes from the Facility 912 oil/water separator and tank sludge from the periodic cleaning of fuel storage tanks. Some of the tank sludge contained lead from cleaning AVGAS storage tanks.

During the IRP, Phase II, Stage 1 investigation, eight two-foot deep soil borings were advanced along the road. Samples were collected from each of the borings at one-foot intervals. In December 1991 during the OU-2 RI, twelve 40-foot deep soil borings (SB-1 through SB-12) were drilled along the length of the roadway. The borings were spaced at approximately 800-foot intervals to provide coverage across the entire length of the site. A total of 51 soil samples (48 primary and 3 duplicate) were collected from the borings and submitted for laboratory analysis. The contract laboratory analyzed samples for TRPH (EPA Method 418.1), VOCs (EPA Method 8240), BNAs (EPA Method 8270), and metals (EPA Method 7421). The subsurface sample from each boring was also analyzed for PCBs (EPA Method 8080). Detailed descriptions of the sampling methodologies and analytical results are presented in the OU-2 RI report.

The surface soils in the southern section of the site were found to contain TRPH at concentrations ranging up to 250 mg/Kg. Although soil borings were advanced to 40 feet bgs, samples collected below 10 feet bgs did not show detectable concentrations of TRPH. VOC compounds were not detected in any of the samples, and the only detected BNA compounds were common laboratory contaminants at low concentrations. With only two exceptions, the metals concentrations detected in soil samples were below their respective background upper confidence limits (UCLs).

During 1992 and 1993, the data collected during the OU-2 field investigation were used to evaluate remedial alternatives for the soils at this site. Based on these results of the OU-2 Risk Assessment and Feasibility Study, no further action alternative was proposed. This alternative was officially adopted in the OU-2 ROD that was signed in January 1994.

Later in 1994, concerns arose about the quality of some of the VOC and BNA data used in the original evaluation of PSC OT-04. In response, seven additional samples were collected in August 1996 to replace the original data of unknown quality. The additional samples were collected from three borings, which were located at the areas of the site, which showed the greatest signs of impact.

The three additional soil borings (SB-13 through SB-15) were located adjacent to Soil Borings SB-5, SB-9, and SB-10, respectively. Three surface and four subsurface samples were collected from the borings and submitted to Quanterra laboratories for analysis. VOC and BNA compounds were not detected in any of the additional samples.

These new sampling data were used with previously collected data of known quality to re-calculate the risk assessment for the site. The results of the risk assessment showed that exposure to the soil at this site would not cause a risk to human health of the environment. Both current and hypothetical future excess lifetime cancer rate (ELCR) and Hazard Index (HI) for exposure to soils at these sites were below the ADEQ and USEPA's residential risk-based remediation benchmarks (ELCR less than 10⁻⁴, HI below 1.0). As a result, the original remedial alternative selected for the site (no further action) was re-affirmed.

PSC DP-05 POL DISPOSAL AREA

PSC DP-05 consists of an 18-acre triangular-shaped area located on the southeast side of Taxiway I. Base Production Well 11 and PSC SD-26 (the Hush House Canal) are located adjacent to the site. PSC DP-05 currently consists of bare ground covered with sparse vegetation According to the IRP, Phase I investigation results, this area was used for the disposal of petroleum, oil, and lubricant (POL) waste from approximately 1970 until 1972. POL wastes were delivered to the site in 5,000-gallon tanker trucks and dumped in shallow (1.5 feet deep) trenches. The waste was allowed to weather for 4 to 6 weeks and was then covered with soil. Eleven trenches ranging from approximately 200 to 550 feet in length were identified on aerial photographs taken

between 1977 and 1989. A shallow lagoon was also identified at the northeast corner of the site. The lagoon was apparently excavated for additional waste disposal. Undocumented estimates of the POL waste volumes, mostly JP-4, were as high as 100,000 gallons per year at this disposal site from 1970 to 1972.

During the IRP, Phase II, Stage 1 investigation, ten soil borings were drilled and sampled to a depth of 20 feet. A soil gas survey, geophysical survey, and soil boring investigation were conducted during the IRP, Phase II, Stage 2 investigation. The results of the soil gas survey and geophysical survey were used to determine the locations of nine 100-foot soil borings. Monitoring wells MW-104, MW-105, and MW-106 were also installed at the site during the IRP Phase II, Stage 2 investigation. The locations for the monitoring wells were selected to encircle the site as completely as possible.

The OU-2 RI field activities at PSC DP-05 began in December 1991 and continued through June of 1992. Initially, twenty 20-foot deep soil borings (SB-1 through SB-20) were drilled and sampled at the areas of concern identified on aerial photographs. Drilling or two 150-foot deep borings (SB-21 and SB-22) was also initiated in February 1992, however, the borings were not completed due to problems caused by heavy rain. Soil Borings SB-21 and SB-22 were only advanced to depths of 77 feet and 20 feet, respectively. Two 150-foot deep borings were completed as Soil Borings SB-23 and SB-24 in April 1992. After review of the data, borings SB-25, SB-26, SB-27, and SB-28 were drilled as contingency borings in June 1992. The contingency borings were drilled to further characterize the organic compounds detected in samples collected from Soil Boring SB-9. Descriptions of the sampling methodologies and analytical results are presented in the OU-2 RI report

A total of 100 samples (95 primary and 5 duplicate) were collected from the borings and submitted for laboratory analysis. Samples were analyzed for TRPH (EPA Method 418.1), VOCs (EPA Method 8240), BNAs (EPA Method 8270), and metals (EPA Method 7421). The 0-2 foot bgs sample from each boring was also analyzed for PCBs (EPA Method 8080). With the exceptions of two samples collected from Soil Boring SB-8 and three samples collected from Soil Boring SB-22. The ATI Phoenix laboratory conducted analytical procedures. The ATI San Diego laboratory analyzed the three samples collected from Soil Boring SB-22 and both samples collected from Soil Boring SB-8.

Samples from the majority of the soil borings drilled at the site did not contain detectable concentrations of organic compounds or detections were limited to near surface soils. TRPH, BNA, and VOC compounds were detected at their highest

concentrations in soil samples collected from the center site where the former disposal pits are located. Soil Boring SB-9 and four contingency borings (SB-25 through SB-28) were drilled in this area of the site during the OU-2 RI.

Soil samples collected from Soil Boring SB-9 at a depth of 6 feet bgs contained the highest concentrations of TRPH (8,300 mg/Kg). TRPH was not detected below the depth of 22-feet in any of the samples collected at the site. Ethyl benzene and xylenes were the only detected VOC compounds. The highest detected concentration of ethyl benzene was 0.9 mg/Kg in the 12-14 foot bgs samples collected from Soil Borings SB-25 and SB-27. The highest detected concentration of xylenes was 86 mg/kg in the 6-foot bgs sample collected from Soil Boring SB-9. VOC compounds were not detected below the depth of 14 feet. Three BNA compounds were detected in the central portion of the site near SB-9. The highest detected concentrations were naphthalene at 4.6 mg/Kg, 2-methylnapthalene at 4.7 mg/Kg, and BEP at 3.7 mg/Kg. BNA compounds were not detected below the depth of 22 feet.

With the exception of lead, all metals results for soil samples collected from PSC DP-05 were either below their respective background UCLs or were within naturally occurring background ranges. The maximum detected concentration of lead (115 mg/Kg) does exceed its background UCL and the upper range of concentrations included in the background data set. Only three samples contained elevated lead concentrations. The surface sample collected from Soil Boring SB-7 contained 115 mg/kg of lead. The two shallowest samples collected from Soil Boring SB-9 contained lead concentrations of 72 mg/kg and 39 mg/kg, respectively. These samples also contained some of the highest concentration of TRPH detected at this site.

Groundwater samples collected from PSC DP-05 did not contain detectable concentrations of organic compounds with just two exceptions. VOC compounds were detected in one sample collected from Monitoring Well MW-104 in November1992. The only VOC compounds detected were acetone and toluene. These compounds were never detected in any of the other samples collected from Monitoring Well MW-104. BEP, a common laboratory contaminant, was the only other organic chemical detected in groundwater samples collected from this site. This BNA compound was not detected in any of the other groundwater samples. Total silver (0.018 mg/L) was detected in one groundwater sample collected from Monitoring Well MW-106. Silver was not detected in any of subsequent groundwater samples events or in any of the other samples collected at the site. All other metals detected in groundwater samples collected from PSC DP-05 were either below their respective background UCLs or were within the range of naturally occurring concentrations included in the background data set. During 1992 and 1993, the data collected during the OU-2 field investigation were used to evaluate remedial alternatives for the soils at this site. Based on these results of the OU-2 Risk Assessment and Feasibility Study, a no further action alternative was proposed. This alternative was officially adopted in the OU-2 ROD that was signed in January 1994.

Later in 1994, concerns arose about the quality of some of the VOC and BNA data used in the original evaluation of PSC DP-05. Because only a limited area near the center of the site was impacted, the FFA parties determined that two additional soil borings (SB-29 and SB-30) would provide the needed data. Both borings were located in near the center of the former disposal pits at the area of the site that showed the greatest sign of impact. Soil Boring SB-29 was located adjacent to Soil Boring SB-27. Soil Boring SB-30 was located adjacent to Soil Boring SB-25. None of the samples collected from the additional soil borings contained detectable concentrations of VOCs or BNAs.

These new sampling data were used with previously collected data of known quality to re-calculate the risk assessment for the site. The results of the risk assessment showed that exposure to the soil at this site would not cause a risk to human health of the environment. Both current and hypothetical future ELCRs and HIs for exposure to soils at these sites were below the ADEQ and USEPA's residential risk-based remediation benchmarks (ELCR less than 10^{-4} , HI below 1.0). As a result, the original remedial alternative selected for the site (no further action) was re-affirmed.

PSC FT-06 SOUTH FIRE TRAINING AREAPSC FT-06 SOUTH FIRE TRAINING AREAPSC FT-06 SOUTH FIRE TRAINING AREAPSC FT-06 SOUTH FIRE TRAINING AREA

PSC FT-06 was the original fire department training area for the Base. PSC FT-06 is located in the southern portion of the Base, east of the Facility 1009 power check pad. The PSC is a rectangular area approximately 8 acres in size. Buildings 1031, 988, and 1018 are located on the site. Eighty percent of the PSC is paved; this includes portions that are under building foundations, parking lot asphalt, and a concrete lined storm drain canal. Twenty percent of the PSC is unpaved including landscaped areas around buildings, parking lots that are covered with gravel, and a bare area north of the perimeter road.

During past operations, standard practice was to transport petroleum, oil, and lubricant waste in 55-gallon drums to the fire department training site. The POL wastes were poured onto an old aircraft or simulated aircraft in a cleared, unlined, bermed circular

pit approximately 100-feet in diameter. The structures were then set on fire and extinguished with water and aqueous film forming foam (AFFF). Fire training operations were conducted at this site from 1941 until deactivation of the Base in 1946, and again from the time of Base reactivation in 1951 until approximately 1973. Aerial photographs from 1953, 1962, and 1970 indicate that 13 pits of various dimensions were present at this site.

Ten 20-foot deep soil borings were originally proposed in the IRP, Phase II, and Stage 1 investigation. However, due to construction at the site, drilling of the proposed soil borings was prohibited. Instead, four shallow soil samples (2 to 3-feet total depth) were collected from soil exposed by construction activities. In addition, two 100-foot borings were drilled, and a total of 40 subsurface soil samples were collected during the IRP Phase II, Stage 2 soils investigation.

Two groundwater monitoring wells (MW-107 and MW-108) were installed at this site during the IRP, Phase II, and Stage 2 groundwater investigation. Monitoring well locations were chosen so that MW-107 was located between the former pits and MW-108 was in the presumed down gradient direction. Monitoring Wells MW-107 and MW-108 were included in the Base wide groundwater monitoring and sampling program, which continued through the second quarter of 1996.

From December 1991 through April 1992, 18 soil borings (SB-1 through SB-18) were drilled and sampled at the site during the OU-2 field investigation. Thirteen borings were advanced to 100 feet bgs and five borings were advanced to 20 feet bgs. The locations were selected so that at least one 100-foot deep soil boring would be advanced in each of the former fire training pits identified on the aerial photographs.

TRPH was detected in 14 of the 18 soil borings, with the highest concentration of 18,000 mg/Kg being in the 2-4 foot bgs sample from Soil Boring SB-5. The TRPH values from the remaining borings ranged from 10 to 12,000 mg/Kg. With the exception of Soil Borings SB-5 and SB-18, TRPH was not detected below 20-feet. In Soil Borings SB-5, and SB-18, TRPH was detected at maximum depths of 38 and 24 feet bgs, respectively. VOC compounds including BTEX, TCE, PCE, MEK and methyl isobutyl ketone were detected at depths up to 14 feet bgs. BNA compounds, such as phenanthrene, chrysene, fluoranthene, pyrene, benzo(b)fluroanthene, and benzo(a)pyrene, were detected in samples from 16 borings. Sample collected from Soil Boring SB-8 contained he highest concentrations of BNAs, with 27 BNA compounds detected in the 0-2 foot bgs sample. BNA compounds were detected at depths up to 24 feet bgs. PCBs were not detected in any of the samples collected and analyzed at PSC FT-06.

Three surface soil samples were also collected from PSC FT-06 for dioxin and furan analysis. Two composite soil samples and one background sample were collected and submitted for analysis. Dioxins and furans were not detected in either of the composite samples. However, the background sample did contain heptachlorodibenzo-p-dioxin (HpCDD) at a concentration of 1.2 nanograms per gram (ng/g), octachlordibenzo-p-dioxin (OCDD) at 4.6 ng/g, heptachlorodibenzo-p-furan (HpCDF) at 1.1 ng/g, and octachlordibenzo-p-furan (OCDF) at 2.0 ng/g. Because this sample was not collected in the former fire training pits, the presence of the dioxins and furans is most likely not related to site activities.

The results of the analyses conducted on groundwater samples collected from Monitoring Wells MW-107 and MW-108 indicate VOCs, BNA compounds, and EDB, are not present in the groundwater at this PSC. The agricultural pesticide DBCP (0.05 micrograms per liter) was present in groundwater. Detected metals concentrations in groundwater were within background ranges.

During 1992 and 1993, the data collected during the OU-2 field investigation were used to evaluate remedial alternatives for the soils at this site. Based on these results of the OU-2 Risk Assessment and Feasibility Study, a no further action alternative was proposed. This alternative was officially adopted in the OU-2 ROD that was signed in January 1994.

Later in 1994, concerns arose about the quality of some of the VOC and BNA data used in the original evaluation of PSC FT-06. The FFA parties elected to collect additional data at the site to re-evaluate the original remedial alternative. A total of six additional subsurface soil samples were collected in August of 1996 as part of this additional soil sampling investigation. Surface samples were not collected because the former fire training pits are no longer present and exposure to their surface soils are not possible. For each additional soil boring, the sampling depth was determined based on visual evidence and field screening results noted on the original soil boring logs. Attempts were made to collect soil samples immediately beneath the former fire training pits.

None of the samples collected from additional Soil Borings SB-20, SB-22, or SB-23 contained detectable concentrations of either VOCs or BNAs. The sample collected from Soil Boring SB-19 did not contain detectable concentrations of VOCs; however, di-n-octylphthalate was detected at a concentration of 0.16 mg/Kg. This was the only BNA compound detected at Soil Boring SB-19. Similarly, the sample collected from Soil Boring SB-21 did not contain VOCs, but nine different BNA compounds were detected in the sample collected from this borehole. The highest detected

concentration was 0.33 mg/Kg of pyrene. Seven VOC compounds were detected in the subsurface sample collected from Soil Boring SB-24. The highest detected concentration was 23 mg/Kg of xylenes. BNA compounds 2-methylnapthalene (15 mg/Kg) and naphthalene (33 mg/Kg) were also detected in this sample.

These new sampling data were used with previously collected data of known quality to re-calculate the risk assessment for the site. The results of the risk assessment showed that exposure to the soil at this site would not cause a risk to human health of the environment. Both current and hypothetical future ELCRs and HI's for exposure to soils at these sites were below the ADEQ and USEPA's residential risk-based remediation benchmarks (ELCR less than 10⁻⁴, HI below 1.0). As a result, the original remedial alternative selected for the site (no further action) was re-affirmed.

PSC FT-07W WESTERN PORTION OF THE NORTH FIRE TRAINING AREA

PSC FT-07W is located in the northern portion of the Base, west of Fire Department Training Facility 1355. PSC FT-07W was previously identified as Site Number 7 in the IRP Phase I, Records Search Report (CH2M HILL, 1982). The North Fire Training Area (NFTA) was divided into an eastern and western portion during the RI. The soils in the western portions were included in the OU-2 investigation. The eastern portion of the site was investigated during the OU-1 RI. Section 13 of this report details the investigative results of the eastern portion of the site.

The western portion of the NFTA occupies approximately 14 acres west of Facility 1356. Approximately 50 percent of the site is currently occupied by a new fire training facility that was constructed in the spring of 1996. The remaining portion of the site is covered by bare ground with sparse vegetation. During past operations at the site, POL waste was poured into circular unlined bermed areas containing mock aircraft and then set on fire for fire fighting training. These fires were extinguished with water and aqueous film forming foam.

During the IRP Phase II, Stage I investigation, Weston identified three former firetraining pits in the western portion of PSC FT-07. Four 20-foot deep soil borings were drilled at these pits. Two 20-foot deep borings were advanced within the biggest pit, while a single 20-foot deep boring was advanced in each of the smaller pits. VOC and Oil & Grease were detected in several samples collected from the pits. Based on the results of this contaminant verification sampling, additional investigation was recommended at two of the pits in the IRP Phase II, Stage 2 investigation. During the IRP Phase II, Stage 2 investigation, Weston advanced a 100-foot soil boring in the two pits recommended for further investigation. Three monitoring wells (MW-109, MW-110, and MW-111) were also installed and sampled by Weston during the IRP Phase II, Stage 2 investigation.

Following completion of Weston's activities, EA Engineering Science and Technology, was contracted to perform additional soil investigations across the entire site at PSC FT-07W. The main objective of the EA Engineering's investigation was to design a soil vapor extraction system for the eastern portion of the site (FTP-3 and FTP-4). However, they did advance four additional borings in each of the three pits identified in the western portion of the site. EA Engineering designated the three pits in the western portion of the site as FTP-1, FTP-2, and FTP-5.

For each of the pits, a deep boring was drilled in the center of the pit and three shallow borings were drilled around its perimeter. The deep borings in FTP-1 and FTP-5 were advanced to a depth of 100 feet bgs. The deep boring for FTP-2 was only advanced to a depth of 50-feet bgs. The shallow borings were all drilled to a depth of 30 feet bgs. Samples collected from these borings were analyzed for TRPH (EPA Method 418.1), VOCs (EPA Method 8240), and metals (EPA Method 7421) by Southwest laboratories of Oklahoma.

Soil sampling data indicated that only low concentrations of acetone and methylene chloride were present in the samples. Although numerous detections of these constituents were attributed to laboratory contamination, the deepest samples collected from FTP-1 and FTP-5 that were analyzed for VOCS did contain detectable concentrations of acetone and methylene chloride that were not flagged as laboratory contaminants. EA Engineering did not detect pOL-related constituents, such as BTEX and TRPH, in any of the samples collected form the western portion of the site.

During the compilation of the planning documents for the OU-2 investigation, Geraghty & Miller identified seven additional fire-training locations in the western portion of the site. These pits were identified on 1965, 1970, and 1973 aerial photographs. A total of 10 former fire-raining pits (FTP-1, FTP-2, FTP-5, and seven un-numbered pits) were included in the OU-2 investigation. During the OU-2 RI, Geraghty & Miller advanced 20 soil borings in the western portion of PSC FT-07. Ten of the borings were drilled to a depth of 100 feet, and ten drilled to a depth of 20-feet. A deep boring was advanced in each of the ten pits. The ten shallow borings were advanced at various locations around the pits to define the horizontal extent of any detected constituents. Soil samples collected from six of the former fire training pits contained detectable concentrations of organic chemicals. The six pits with organic chemical detections include FTP-1, FTP-2, and FTP-5 and three smaller un-numbered pits. Detected organic constituents are limited to TRPH and BNA compounds. Soils containing detectable concentrations of TRPH range to depths of 100 feet bgs at FTP-2 and 60 feet bgs at FTP-5, but to depths of less than 16 feet bgs in all other areas. BNA compounds were only detected in three samples and were not detected below the depth of 6 feet bgs. Lead was the only inorganic constituent detected above background ranges. Only two samples contained elevated lead concentrations. Both borings with samples containing elevated lead levels also contained detectable concentrations of TRPH.

Based on a review of EA Engineering's data and the data produced during the OU-2 RI, the vertical extent of organic constituents were not defined to within laboratory detection limits at three areas of the site. Samples collected at FTP-2 during the OU-2 RI contained detectable concentrations of TRPH at depths of 100 feet bgs. Samples collected at the center of FTP-1 by EA Engineering contained methylene chloride at a depth of 120 feet bgs. Although methylene chloride is a common laboratory contaminant, this detection at this depth was not qualified. Similarly, samples collected from the center of the FTP-5 at a depth of the 120 feet bgs contained detections of methylene chloride and acetone that were not qualified.

Following completion of the OU-2 investigation, three additional soil borings were advanced and sampled at the site. The purpose of the additional sampling was to define, to laboratory non-detectable levels, the maximum vertical extent of the constituents of potential concern in the soil. TPH was not detected below a depth of 10 feet in any of these three borings. VOCs were not detected in samples collected from Soil Boring SB-24 or SB-25. The only VOC detected in samples collected from Soil Boring SB-27 was acetone; however, based on data validation criteria, all acetone detections were qualified as a laboratory contaminant. Based on the results of the additional sampling investigation, the vertical extent of organic chemicals in the soils beneath each of the former fire training pits at PSC FT-07 have been defined to be less than 120 feet bgs.

These new sampling data were used with previously collected data of known quality to re-calculate the risk assessment for the site. The results of the risk assessment showed that exposure to the soil at this site would not cause a risk to human health of the environment. Both current and hypothetical future ELCR's and HI's for exposure to soils at these sites were below the ADEQ and USEPA's residential risk-based

remediation benchmarks (ELCR less than 10^{-4} , HI below 1.0). As a result, the original remedial alternative selected for the site (no further action) was re-affirmed.

PSC ST-18 FACILITY 993

PSC ST-18 consists of a former liquid waste storage facility (Facility 993) located in the southern part of the Base. Facility 993 originally consisted of a single 5,000-gallon refueling tank truck that was coated and buried in 1968. This underground storage tank (UST) was used for the temporary storage of all liquid petroleum, oil, lubricant (POL) and solvent wastes generated at the Base. Prior to 1972, liquid wastes stored at this facility were disposed during road oiling and dust suppression activities (PSC OT-04), in narrow trenches (PSC DP-05), and in fire training activities (PSCs FT-06 and FT-07). In 1972, two 10,000-gallon capacity USTs were installed at the facility, and the area around all three USTs, approximately 0.2 acres, was enclosed with a fence. Also at this time, the Base began selling the liquid wastes to private contractors for recycling.

This facility was classified as an interim status treatment, storage, and disposal (TSD) facility under RCRA in 1979. Part A of a Hazardous Waste Permit application was submitted in 1980. However, closure of this facility began in 1982 to facilitate the construction of a new USAF Reserve maintenance building.

The USAF to direct the closure activities and related subsurface investigations retained Raymond E. Kary, Ph.D., in association with Guitierrez-Palmenberg, Inc. (GPI). Initially, twelve 50-foot deep soil borings were drilled adjacent to the USTs during July and August 1983. The analytical results showed no contaminants. Based on the results of the initial investigation, a partial closure plan was submitted to the Arizona Department of Health Services. The closure plan was approved on October 4, 1983.

The three USTs were removed on October 19, 1983 following this initial investigation. Soil samples collected directly beneath the 5,000-gallon tanker truck and one of the 10,000-gallon USTs showed signs of impact from past releases. The tank pit was excavated to a depth of 16 feet bgs in an attempt to assess the extent of contamination. Based on field observations, highly impacted soils were manifested to a hazardous waste landfill. The moderately contaminated soils were aired for several weeks and replaced in the pit, and the minimally contaminated soils were placed directly back into the pit.

Because this site was an active facility in 1981, it was not identified in the IRP Phase I investigation which was focused on historic waste disposal activities. However, the Base decided to include this site in the IRP Phase II investigation because of the sampling results of the UST closure activities. Between November 4, 1985 and February 6, 1986, Weston continued the investigation of the site during the IRP, Phase II, and Stage 1 investigation. During this investigation, five soil borings were advanced in and around Facility 993. Soil Boring depths ranged from 100-145 feet bgs. In addition, five groundwater monitoring wells (MW-1 through MW-5) were installed. Conclusions presented in the IRP, Phase II, Stage 1 report (Roy F. Weston, 1986) indicated that the soil beneath the former USTs had been impacted by fuel and organic solvents, and the impacted soil extended to the maximum depth of 56.5 feet bgs.

The site was capped with concrete in 1987 as part of the RCRA post-closure requirements for the site. In a letter dated May 13, 1988, the ADEQ stated that they had inspected the concrete cap covering the facility and it was satisfactory. Currently, the Base continues to inspect and maintain the cap to ensure integrity of the concrete and sealed joints.

In September 1990, the Federal Facilities Agreement for Luke AFB was signed and regulatory authority for Facility 993 was transferred from the RCRA program to the CERCLA program as part of RCRA/CERCLA integration. The FFA parties elected to include Facility 993 in the OU-2 remedial investigation as PSC ST-18.

From February to June of 1992, eight soil borings were drilled during the OU-2 investigation to define the horizontal and vertical extent of the soil impact. A total of 37 samples were collected from the borings and submitted to the ATI Phoenix laboratory for analysis. The highest detected concentrations of TRPH were in the samples collected from Soil Borings SB-1 and SB-2, which were drilled at the former leaking UST locations. The highest detected TRPH value in Soil Boring SB-1 was 4,900 mg/Kg in the 12-14 feet bgs sample. Samples collected from Soil Boring SB-2 contained TRPH at concentrations of 10,000 mg/Kg and 17,000 mg/Kg in the 12-14 foot bgs and 20-22 foot bgs samples, respectively. VOC compounds were only detected in Soil Borings SB-1 and SB-2. Detected compounds include: BTEX, 1,1-dichloroethane, 1,1,2,2-tetrachloroethane, TCE, and PCE. The highest concentrations of VOCs were detected in the 20-22 foot bgs sample from Soil Boring SB-2, which was also the sample with the highest TRPH value. BNAs were detected in samples from all eight borings. In general, the detected BNA compounds were all found in samples that also contained TRPH.

Groundwater quality beneath PSC ST-18 was evaluated using analytical results from groundwater samples collected at Groundwater Monitoring Wells MW-2, MW-3, MW-4, MW-5, MW-114, and MW-122. VOCs and BNAs were not detected in Monitoring Wells MW-2, MW-3, MW-5, or MW-122. EDB and DBCP were not reported in any samples collected from Monitoring Wells MW-2, MW-4, MW-5 and MW-114. EDB was also not detected in the samples analyzed from Monitoring Well MW-3.

Toluene was detected at a concentration of 4 µg/L during the first quarter 1992 groundwater sampling event at Monitoring Well MW-4. Toluene has not been reported in subsequent sampling events at Monitoring Well MW-4 or in any of the other wells monitored at ST-18. BEP, a common laboratory contaminant, was detected during the first quarter 1992 and second quarter 1993 in Monitoring Well MW-114. However, BEP was reported in the equipment blank during the first quarter of 1992.

DBCP was reported in Monitoring Well MW-3 at a concentration of 0.07 μ g/L in the only sampling event (fourth quarter 1992) for which analyses for DBCP were performed. DBCP is a pesticide that was commonly used in citrus groves from the mid-1950s until it was banned by the USEPA in 1980. Citrus groves may have been located on private lands adjacent to Luke AFB. No citrus groves are currently present at Luke AFB. Although DBCP was not detected in any other wells at ST-18, it was reported in Monitoring Wells MW-107, MW-108 and MW-110.

Throughout 1992 and 1993, the data collected during the OU-2 field investigation were evaluated to determine the appropriate remedial alternative for this site. Based on the results of the OU-2 Risk Assessment and OU-2 Feasibility Study, the remedial action proposed for implementation at PSC ST-18 was capping, surface controls, and monitoring. The FFA parties in the OU-2 ROD that was signed in January 1994 officially adopted this alternative. Although a cap had already been installed at the site, the monitoring requirements for PSC ST-18 were not identified in the OU-2 ROD because they are dependent on the results and conclusions of the groundwater investigation which is included as part of theOU-1 Remedial Investigation Report.

Later in 1994, concerns arose about the quality of some of the VOC and BNA data used in the original evaluation of PSC FT-06. Because CERCLA guidance requires that only data of known quality be used to evaluate remedial alternatives for a site, the FFA parties elected to collect additional data at the site to re-evaluate the original remedial alternatives. Additional data was also to be collected to refine the delineation of the extent of the soil impact. A more refined delineation of the extent of impact was needed to assist in the establishment of long term groundwater monitoring requirements for the site, as required by the original OU-2 ROD.

Nine additional soil borings were advanced and sampled in August of 1996 because of concerns of the quality of the original VOC and BNA data and to refine the delineation of the horizontal and vertical extent of the impacted soil. A total of 36 samples (33 primary and 3 duplicate) were collected from the nine borings and submitted for laboratory analysis.

Analytical results from the additional sampling indicated that TPH concentrations are highest in the area of the former UST pit. TPH concentrations were detected to depths of 48-50 feet bgs in SB-11 and 78-80 feet bgs in SB-10. In both borings, the samples containing the highest detected TPH concentrations (6,800 for SB-10 and 18,000 mg/Kg for SB-11) were collected at a depth of 18-20 foot bgs. This depth corresponds to just below the depth excavated during the removal of the USTs. VOC compounds (including BTEX, TCE and PCE) were detected in Soil Borings SB-10 and SB-11 to depths of 60 feet bgs. BNAs were detected in Soil Borings SB-10 and SB-11, but only in those samples, which also contained TPH. BEP, a common laboratory contaminant, was also detected in Soil Borings SB-13, SB-14, and SB-15. Most values were qualified as estimated values.

These new sampling data were used with previously collected data of known quality to re-calculate the risk assessment for the site. The results of the risk assessment showed that exposure to the soil at this site would not cause a risk to human health of the environment. Both current and hypothetical future ELCRs and HIs for exposure to soils at these sites were below the ADEQ and USEPA's residential risk-based remediation benchmarks (ELCR less than 10⁻⁴, HI below 1.0). As a result, the original remedial alternative selected for the site (capping, surface controls, and monitoring) was re-affirmed as protective of human health and the environment.

PSC DP-22 POL TRENCH AT NORTHEAST RUNWAY

PSC DP-22 is an irregular-shaped area located at the north end of the inboard runway. The site occupies approximately 4.6 acres. Approximately 30 percent of the PSC is covered with concrete (the inboard runway), 20 percent is covered with bituminous cover, and 50 percent of the site is covered by gravel with sparse vegetation.

This site may have been used for the disarmament and de-fueling of aircraft during the 1940s and 1950s. Reportedly, waste POL was dumped into shallow trenches at this site. Based on interviews with Base personnel, off-loaded fuel may have been drained into trenches perhaps 600 to 800 feet long and a few feet deep. No evidence of trenches was visible on examination of aerial photographs. During the 1950s, the configuration of the east runway was different than it is at present. The reported disposal site was located approximately 800 feet southwest of where the runway presently ends. Construction of the runway extension in the late 1950s ended disposal of POL in this area. There were no environmental investigations conducted at this site prior to the OU-2 RI.

In January and February of 1992, five soil borings (SB-1 through SB-5) were drilled at PSC DP-22 during the OU-2 investigation. TRPH concentrations were detected in the surface or near surface samples collected at each of the borings. Detected TRPH concentrations generally decreased with increasing depth. The surface sample collected from Soil Boring SB-4 contained the highest detectable concentration of TRPH (970 mg/Kg). Although TRPH was generally not detected in subsurface soils, TRPH was detected at concentrations near the laboratory detection limits in the 98-100 foot bgs samples from Soil Borings SB-3 and SB-4. The only VOC detected was acetone at a concentration of 1.0 mg/Kg in the 10-12 foot bgs sample collected from Soil Boring SB-2. BNA compounds were not detected in any of the 21 soil samples submitted for laboratory analysis. The highest detected concentrations of barium (407 mg/Kg) and lead (30 mg/Kg) do exceed their respective background UCLs. The surface sample collected from Soil Boring SB-3 contained both of these elevated metals concentrations. This sample also contained TRPH. None of the other samples collected from the site contained metals at concentrations above their background UCLs.

During 1992 and 1993, the data collected during the OU-2 field investigation were used to evaluate remedial alternatives for the soils at this site. Based on these results of the OU-2 Risk Assessment and Feasibility Study, a no further action alternative was proposed. This alternative was officially adopted in the OU-2 ROD that was signed in January 1994.

Later in 1994, concerns arose about the quality of some of the VOC and BNA data used in the original evaluation of PSC DP-22. The FFA parties determined these data were of unknown quality. In response, six additional samples (5 primary and one duplicate) were collected in August 1996 to replace the original data of unknown quality. The additional samples were collected from three borings, which were located at the areas of the site, which showed the greatest signs of impact. The three additional soil borings (SB-6 through SB-8) were located adjacent to Soil Borings SB-3, SB-1, and SB-4, respectively. Three surface and three subsurface samples (5 primary and one duplicate) were collected from the borings and submitted to Quanterra laboratories for analysis. None of the samples collected from the additional soil borings contained detectable concentrations of VOCs. The primary subsurface sample collected from Soil Boring SB-7 contained a concentration of 0.17 mg/Kg of BEP, a common laboratory contaminant. This was the only BNA compound detected in any of the six additional soil samples.

These new sampling data were used with previously collected data of known quality to re-calculate the risk assessment for the site. The results of the risk assessment showed that exposure to the soil at this site would not cause a risk to human health of the environment. Both current and hypothetical ELCRs and HIs for exposure to soils at these sites were below the ADEQ and USEPA's residential risk-based remediation benchmarks (ELCR less than 10^{-4} , HI below 1.0). As a result, the original remedial alternative selected for the site (no further action) was re-affirmed.

PSC DP-23 OLD SURFACE IMPOUNDMENT AREA

PSC DP-23 consists of the Old Surface Impoundment and associated wash located west of Building 999 and adjacent to the former south fire training area. The old surface impoundment portion of the site is a rectangular-shaped area that occupies approximately 3.3 acres. Currently, 80 percent of this area is either paved with asphalt, under tarmac, or under concrete, which includes the AGE equipment yard. In the late 1940's, an impoundment dam was constructed along an old natural drainage system, which flowed south off of the Base. This area may have been used for the disposal site for POL waste until construction covered the site in 1969. The dam used to create the surface impoundment was buried, but not removed. The wash portion of the site is located to the south of the impoundment area and occupies approximately 19.4 acres. The wash extends off Base and flows south to an area of earth fissures (See PSC SD-20).

In February of 1992, two 150-foot deep borings (SB-2 and SB-4) and four 40-foot deep soil borings (SB-1, SB-3, SB-5, and SB-6) were drilled and sampled at PSC DP-23 during the OU-2 investigation. Sediment samples were collected from ten locations (SD-1 through SD-10) in December of 1991 and February of 1992. A total of 26 soil samples (23 primary and 3 duplicate) and 21 sediment samples (20 primary and 1 duplicate) were collected and submitted for laboratory analysis.

The highest detected concentration of TRPH was 2,000 mg/kg in the 2 to 4 foot bgs sample collected from Soil Boring SB-4. The only detected VOC compounds (trace concentrations of toluene and ethyl benzene) were also detected in this sample. TRPH was generally confined to shallow soils, and the deepest sample with detectable TRPH concentrations was collected at a depth of 8 to 10 feet bgs from SB-5.

Six soil and five sediment samples collected during the OU-2 investigation contained detectable concentrations of BNA compounds. Four samples contained concentrations of Benzo(a)pyrene in excess of its Preliminary Remediation Goal (PRG) of 0.78 mg/kg. These four samples include the surface sample and its duplicate collected from SB-4, the 2 to 4 foot bgs sample collected from SB-4, and the surface sample collected from SB-5. None of the other samples contained BNA compounds at concentrations in excess of their PRGs

The data collected during the OU-2 field investigation were evaluated to determine the appropriate remedial alternative for this site. Based on the results of the OU-2 Risk Assessment and OU-2 Feasibility Study, the remedial action proposed for PSC DP-23 was excavation of all soils with benzo(a)pyrene concentrations above the PRGs, biological treatment to reduce contaminant concentrations, monitoring to confirm effectiveness, and return of the treated soils to the excavation for final disposal. The FFA parties in the OU-2 ROD that was signed in January 1994 officially adopted this alternative.

PSC SD-40 TAXIWAY FUEL DISCHARGE AREA

PSC SD-40 consists of the areas located on both sides of the southeastern end of Taxiway F and on both sides of the south-central section of Taxiway E. The southern area of the PSC (along Taxiway F) covers approximately 3 acres and the northern area (along Taxiway E) covers approximately 7.6 acres. The areas adjacent to the taxiways are covered with a bituminous dust cover of 2-inch thick asphalt. The site has been used to perform limited service of aircraft since the present runway layout was complete in the 1950s. De-fueling of jet aircraft onto the bituminous cover was reportedly conducted for fuel tank maintenance. This de-fueling practice occurred on Taxiway F from the early 1970s until 1990.

Information from interviews with Base personnel indicates that during maintenance activities fuel was drained from the aircraft fuel tanks onto the dust cover adjacent to the taxiways. De-fueling of jets is believed to have been the primary source of releases from the 1970s until the de-fueling procedure was modified in 1990 to control those releases. The amounts of fuel involved in the individual events varied depending on

the maintenance involved. In addition, hydrazine may have been discharged onto the surface adjacent to the Foxtrot Extension during F-16 maintenance. Staining of the taxiways and asphalt along the taxiways is documented by historical aerial photographs dating back to 1964. No staining was visible in the 1958 aerial photographs so the maintenance activities may have begun between 1958 and 1964. Information from interviews indicates that for at least 20 years these areas were used as Alternate F-15 fuel tank maintenance areas.

In April of 1992, eleven 100-foot deep borings were drilled at PSC SD-40 during the OU-2 investigation. TRPH was detected in 21 of the 47 samples and in samples collected from ten of the eleven borings. The highest detected concentration of TRPH was 1,200 mg/Kg in the 0-2 feet bgs sample from Soil Boring SB-7. TRPH concentrations were generally confined to 4 feet bgs, with concentrations decreasing with increasing depth. Only three samples collected below the depth of 4 feet contained detectable concentrations of TRPH. The two deepest detections of TRPH were in soil borings SB-3 and SB-7. The 98 to 100 foot bgs sample collected from both of these borings contained 20 mg/kg of TRPH. VOCs were only detected in two samples, SB-2 at 98-100 feet bgs and the surface at SB-7. These results in Soil Boring SB-2 at 98-100 feet bgs are most likely related to either laboratory or field sampling contamination. VOCs (toluene, ethyl benzene, and xylenes) were detected in the 0-2 feet bgs sample collected from Soil Boring SB-7. The BNA compounds naphthalene and 2-methylnapthalene were also detected in this sample. This was the only sample with detectable BNA compounds. PCBs were not detected in any of the samples.

The highest detected concentrations of barium (402 mg/Kg), copper (42.8 mg/Kg), and nickel (35 mg/Kg) slightly exceed their background UCLs. The only sample that contained these metals at levels above the average for the site was collected from SB-8 at a depth of 98 to 100 feet bgs. This sample also contained the highest detected concentration of zinc. No trends were observed with respect to metals results and depth of sample collection.

Throughout 1992 and 1993, the data collected during the OU-2 field investigation were evaluated to determine the appropriate remedial alternative for the soils at this site. Based on the results of the OU-2 Risk Assessment and OU-2 Feasibility Study, a no further action alternative was proposed. The FFA parties in the OU-2 ROD that was signed in January 1994 officially adopted this alternative.

Later in 1994, concerns arose about the quality of some of the VOC and BNA data used in the original evaluation of PSC SD-40. In response, three additional soil borings were advanced at the site in August of 1996.

A total of seven soil samples (three surface and four subsurface) were collected in August of 1996 during the additional sampling investigation. None of the samples collected from the additional soil borings contained detectable concentrations of VOCs. The 14-16 foot bgs sample collected from Soil Boring SB-12 contained a concentration of 0.075 mg/Kg of bis(2-ethylhexyl)phthalate, a common laboratory contaminant.

These new sampling data were used with previously collected data of known quality to re-calculate the risk assessment for the site. The results of the risk assessment showed that exposure to the soil at this site would not cause a risk to human health of the environment. Both current and hypothetical future ELCRs and HIs for exposure to soils at these sites were below the ADEQ and USEPA's residential risk-based remediation benchmarks (ELCR less than 10^{-4} , HI below 1.0). As a result, the original remedial alternative selected for the site (no further action) was re-affirmed.

Appendix B

Groundwater Monitoring Results for the Period of Record

Sample	Location		Organ	ic Consti	tuents				Ino	rganic (Constitue	nts (mg	y/L)		Inc	organic	Consti	ituents (Continu	ed(mg/l	L)
Date	ļ	VOCs	BNAs	ТРН	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)															
12-12-91 Total Metals Dissolved	MW-104	ND [UQ]	ND [UQ]	NA	NA	NA	<0.01 <0.01	<0.005 <0.005	0.267 0.285	<0.005 <0.005	<0.005 <0.005	<0.01 <0.01	0.017 <0.01	<0.0002 <0.0002	<0.02 <0.02	0.003 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.354 0.320	
07-16-92 Total Metals Dissolved	MW-104	ND [UQ]	ND J [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.171	<0.005 <0.005	<0.0005 <0.0005	0.014 <0.010	0.012 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.006 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	1.23 0.071	NA NA
11-19-92 Total Metals	MW-104	ACE 15 TOL 5	ND	NA	<0.01	<0.01	<0.010	<0.005	0.257	<0.005	<0.0005	<0.010	<0.010	<0.0002	<0.020	0.011	<0.05	<0.005	<0.005	0.977	<0.10
11-19-92 Total Metals	MW-104 Duplicate	ACE 13 TOL 5	ND	NA	<0.01	<0.01	<0.010	<0.005	0.255	<0.005	<0.0005	<0.010	<0.010	<0.0002	<0.020	0.011	<0.05	<0.005	< 0.005	0.985	<0.10
Dissolved 03-09-93	MW-104	[UQ] ND	[UQ] ND	NA	NA	NA	<0.010	<0.005	0.264	<0.005	< 0.0005	<0.010	<0.010	< 0.0002	<0.020	0.011	<0.05	< 0.005	< 0.005	0.920	<0.10
Total Metals Dissolved	MW7 ***	[UQ]	[UQ]	¥7 •	¥7.		<0.010 <0.010	<0.005 <0.005	0.229 0.250	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	0.011 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.003 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.484 0.491	NA NA
06-09-93 Total Metals Dissolved	MW-104	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.239 0.266	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	0.011 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.473 0.452	NA NA
11-10-93 Total Metals 1 Dissolved In	MW-104 Fime-series itial Sample	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.240 0.241	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.285 0.262	NA NA
11-10-93 Total Metals T Dissolved 4	MW-104 Fime-series I-hr sample	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.270 0.286	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.220 0.218	NA NA
05-11-94 Total Metals Dissolved	MW-104	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.254 0.279	<0.004 <0.004	<0.005 <0.005	<0.010 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.386 0.380	NA NA
02-10-95 Total Metals Dissolved	MW-104	ND [UQ]	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
10-04-95 Total Metals Dissolved	MW-104	ND	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
06-10-96 Total Metals Dissolved	MW-104	ND	ND	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
12-13-91 Total Metals Dissolved	MW-105	ND [UQ]	ND [UQ]	NA	NA	NA	<0.01 <0.01	<0.005 <0.005	0.099	<0.005 <0.005	<0.005 <0.005	0.014 <0.01	<0.01 <0.01	<0.0002 <0.0002	<0.02 <0.02	0.003	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.393 0.238	
07-17-92 Total Metals Dissolved	MW-105	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.057 0.037	<0.005 <0.005	<0.0005 <0.0005	<0.010 <0.010	0.028 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.003 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.146 <0.010	NA NA

Sample	Location		Organ	ic Consti	ituents				Ino	rganic	Constitue	ents (mg	;/L)		In	organic	Consti	tuents (Continu	ed(mg/l	L)
Date		VOCs	BNAs	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo		
		(ug/L)	(ug/L) (ug/L) (mg/L) (mg/L) (mg/L)																		

Sample	Location		Organ	ic Consti	tuents		—		Ino	rganic (Constitue	ents (mg	•/L)		Inc	organic	Consti	tuents (Continue	d(mg/l	L)
Date		VOCs	BNAs	ТРН	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Ću	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
	I	(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)		' <u> </u>	' <u> </u>	<u> </u>	' I			<u> </u>	I	<u> _ </u>	' <u> </u>	<u> </u>	<u> </u>	' <u> </u>	!
12-16-92 Total Metals Dissolved	MW-105	ND [UQ]	ND [UQ]	NA	<0.01	<0.01	<0.010 <0.010	0.006 <0.005	0.111 0.100	<0.005 <0.005	<0.0005 <0.0005	0.014 <0.010	0.014 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.007 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.508 0.375	<0.10 <0.10
03-09-93 Total Metals Dissolved	MW-105	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.108 0.103	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.006 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.363 0.350	NA NA
06-09-93 Total Metals Dissolved	MW-105	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.106 0.103	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.003 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.304 0.230	NA NA
11-05-93 Total Metals Dissolved	MW-105	ND [UQ]	ND * [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.101 0.104	<0.005 <0.005	<0.005 <0.005	<0.010J <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020J <0.020	0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.341 0.220	NA NA
05-11-94 Total Metals Dissolved	MW-105	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.114 0.114	<0.004 <0.004	<0.005 <0.005	<0.010 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.310 0.257	NA NA
05-11-94 Total Metals Dissolved	MW-105 ADEQ Duplicate	ND [UQ]	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
02-10-95 Total Metals Dissolved	MW-105	ND [UQ]	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
10-04-95 Total Metals Dissolved	MW-105	ND	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
06-07-96 Total Metals Dissolved	MW-105	ND	ND	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
11-12-97 Total Metals Dissolved	MW-105	ND	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
5-12-98 Total Metals Dissolved	MW-105	ND	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
12-12-91 Total Metals Dissolved	MW-106	ACE TrU [UQ]	ND [UQ]	NA	NA	NA	<0.01 <0.01	0.014 <0.005	0.182 0.155	<0.005 <0.005	<0.005 <0.005	0.032 <0.01	0.040 <0.01	<0.0002 <0.0002	<0.02 <0.02	0.017 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	1.04 0.356	
12-12-91 Total Metals Dissolved	MW-106 Duplicate	ND [UQ]	ND [UQ]	NA	NA	NA	<0.01 <0.01	0.014 <0.005	0.18 0.158	<0.005 <0.005	<0.005 <0.005	0.034 <0.01	0.018 <0.01	<0.0002 <0.0002	<0.02 <0.02	0.011 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.986 0.356	
07-16-92 Total Metals Dissolved	MW-106	ND [UQ]	BEP 17 [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.064 0.117	<0.005 <0.005	<0.0005 <0.0005	<0.010 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.242	NA NA

Sample	Location		Organ	ic Consti	ituents				Ino	rganic	Constitue	ents (mg	;/L)		In	organic	Consti	tuents (Continu	ed(mg/l	L)
Date		VOCs	BNAs	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo		
		(ug/L)	(ug/L) (ug/L) (mg/L) (mg/L) (mg/L)																		

Sample	Location		Orga	nic Const	ituents				In	organic	Constitu	ents (m	g/L)		In	organic	Const	ituents (Continu	ed(mg/l	L)
Date		VOCs	BNAs	ТРН	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)	_							_							
11-24-92	MW-106	ND	ND	NA	< 0.01	< 0.01	.0.010	.0.005	0.174	.0.005	.0.0005	0.010	.0.010		.0.020	0.005	.0.05	.0.005	.0.005	0.401	.0.10
I otal Metal	S	[UQ]	[UQ]				<0.010	< 0.005	0.1/4	<0.005	< 0.0005	0.012	< 0.010	< 0.0002	<0.020	0.005	< 0.05	< 0.005	< 0.005	0.421	<0.10
Dissolved							<0.010	<0.003	0.167	<0.005	<0.0003	<0.010	<0.010	0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0.380	<0.10
03-22-93	MW-106	ND	ND	NA	NA	NA															
Total Metal	s	[UQ]	[UQ]				< 0.010	< 0.005	0.17	< 0.005	< 0.0005	< 0.010	< 0.010	< 0.0002	< 0.020	0.004	< 0.05	< 0.005	< 0.005	0.410	NA
Dissolved							< 0.010	< 0.005	0.203	< 0.005	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.304	NA
06-10-93 Tatal Matal	MW-106	ND	ND	NA	NA	NA	<0.010	<0.005	0.154	<0.005	<0.005	0.011	<0.010	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0.204	NIA
Dissolved	5	[UQ]	[UQ]				<0.010	<0.003	0.154	<0.005	<0.003	<0.011	<0.010	< 0.0002	<0.020	<0.002	<0.05	<0.003	<0.005	0.284	NA
Dissolved							-0.010	-0.005	0.105	-0.005	-0.005	-0.010	-0.010	-0.0002	-0.020	-0.002	-0.05	-0.005	-0.005	0.224	1471
11-05-93	MW-106	ND	ND *	NA	NA	NA															
Total Metal	s						< 0.010	< 0.005	0.173	< 0.005	< 0.005	0.012	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.587	NA
Dissolved							< 0.010	< 0.005	0.170	< 0.005	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.522	NA
05 11 04	NUV 107	ND	ND *	214		N14															
US-11-94 Total Metal	MW-106	IUO1	TTOJ	NA	NA	NA	0.018	<0.005	0 161	<0.004	<0.005	<0.010	<0.010	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0.281	NΔ
Dissolved	5	[00]	[00]				<0.010	<0.005	0.172	<0.004	<0.005	<0.010	<0.010	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0.231	NA
02-10-95	MW-106	ND	NA	NA	NA	NA															
Total Metal	s	[UQ]	[UQ]				NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
10.04.95	MW-106	ND	NA	NA	NA	NA															
Total Metal	s	ND	11/1	INA	na.	na.	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
06-07-96	MW-106	ND	ND	NA	NA	NA															
Total Metal	s						NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Notes:																					
NA	Not analyzed				TCE	Trichloro	bethene.														
VOCs	Volatile orga				EDB	Ethylene	Dibrom	ide.													
TPH	Total Petrole	um Hydro	carbons		DBCP	1,2-Dibr	omo-3-c	hloropro	pane.												
Tr	Trace, unqua	ntifiable a	mount det	ected.	PCB	Pentachl	orobenze	ene													
ACE	Acetone.				BA	Benzoic	Acid														
BEP	Bis(2-ethylhe	exyl)phtha	late.		DCP	1,2-Dich	loroprop	ane													
CHL	Chloroform.				Во	Boron.															
CDS	Carbon Disu	fide			*	TICs pre	sent.														
DCA	1,2-Dichloro	ethane.			J	Data are	qualitati	ve or es	timated												
DBCM	Dibromochlo	romethane	e.		R	Data are	rejected	and unu	isable												
ND	Not detected				U	Data are	qualitati	ve and c	onside	ed to be	not detec	ted at									

qua BNAs Base/neutral and acid extractable compounds. the value reported. TOL Toluene. [UQ] ATI-Phoenix data of unknown quality as determined by

Well screen submerged below top of water table. the FFA Parties (1)

Method detection limits (mdls) for VOCs and BNAs varied. See laboratory reports for specific mdls. Metals are as follows: silver (Ag), arsenic (As), beryllium (Be), barium (Ba), cadmium (Cd), copper (Cu), chromium (Cr), mercury (Hg), nickel (Ni), lead (Pb), antimony (Sb), selenium (Se), thallium (Tl), and zinc (Zn).

Sample	Location									Inorganic	Constitue	nts (mg/L)				Inorg	anic Cons	stituents Co	ontinued (n	ng/L)	
Date		VOCs	BNAs	ТРН	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)															
01-22-92 Total Metals Dissolved	MW-107	ND [UQ]	ND [UQ]	NA	NA	NA	<0.01 <0.01	<0.05 <0.05	0.159 0.137	<0.005 <0.005	<0.005 <0.005	<0.01 <0.01	<0.01 <0.01	<0.0002 <0.0002	<0.02 <0.02	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.435 0.430	NA NA
01-22-92 Total Metals Dissolved	MW-107 Duplicate	ND [UQ]	ND [UQ]	NA	NA	NA	<0.01 <0.01	<0.05 <0.05	0.149 0.144	<0.005 <0.005	<0.005 <0.005	<0.01 <0.01	<0.01 <0.01	<0.0002 <0.0002	<0.02 <0.02	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.446 0.432	NA NA
07-17-92 Total Metals Dissolved	MW-107	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	0.016 <0.005	0.144 0.044	<0.005 <0.005	<0.0005 <0.0005	0.054 <0.010	0.018 <0.010	<0.0002 <0.0002	0.022 <0.020	0.010 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	2.05 0.019	NA NA
12-09-92 Total Metals Dissolved	MW-107 Time-series Initial Sample	ND [UQ]	ND [UQ]	NA	<0.01	0.05	<0.010 <0.010	<0.005 <0.005	0.114 0.117	<0.005 <0.005	<0.0005 <0.0005	<0.010 <0.010	0.019 0.022	<0.0002 <0.0002	<0.020 <0.020	0.002 <0.002	<0.05 <0.05	0.006 <0.005	<0.005 <0.005	0.415 0.357	<0.10 <0.10
12-09-92 Total Metals Dissolved	MW-107 Time-series 4-hr Sample	ND [UQ]	NA	NA	<0.01	0.04	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
03-08-93 Total Metals Dissolved	MW-107	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.112 0.120	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	0.018 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.003 <0.002	<0.05 <0.05	0.011 0.010	<0.005 <0.005	0.605 0.591	NA NA
06-16-93 Total Metals Dissolved	MW-107	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.115 0.124	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	0.017 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.003 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.530 0.484	NA NA
11-04-93 Total Metals Dissolved	MW-107	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.116 0.114	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.600 0.596	NA NA
05-18-94 Total Metals Dissolved	MW-107	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.115 0.120	<0.004 <0.004	<0.005 <0.005	<0.010 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.290 0.287	NA NA
02-23-95 Total Metals Dissolved	MW-107	ND [UQ]	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
10-20-95 Total Metals Dissolved	MW-107	ND	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
6-6-96 Total Metals Dissolved	MW-107	ND	ND	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
01-23-92 Total Metals Dissolved	MW-108	ND [UQ]	ND [UQ]	NA	NA	NA	<0.01 <0.01	<0.05 <0.05	0.14 0.14	<0.005 <0.005	<0.005 <0.005	<0.01 <0.01	<0.01 <0.01	<0.0002 <0.0002	<0.02 <0.02	<0.002 <0.002	<0.05 <0.05	0.006 <0.005	<0.005 <0.005	0.443 0.413	NA NA

Sample	Location									Inorganic	Constitue	nts (mg/L)				Inors	ganic Cons	tituents Co	ontinued (n	ng/L)	
Date		VOCs	BNAs	ТРН	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
		$(n\sigma/L)$	$(n\sigma/L)$	(mg/L)	(mg/L)	(mg/L)															-
7-21-92 Total Metals Dissolved	MW-108	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.132 0.128	<0.005 <0.005	<0.0005 <0.0005	<0.010 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.281 0.073	NA NA
11-20-92 Total Metals Dissolved	MW-108	ND [UQ]	ND [UQ]	NA	<0.01	0.02	<0.010 <0.010	<0.005 <0.005	0.125 0.129	<0.005 <0.005	<0.0005 <0.0005	<0.010 <0.010	0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.009 0.009	<0.05 <0.05	0.005 <0.005	<0.005 <0.005	0.790 0.828	<0.10 <0.10
03-11-93 Total Metals Dissolved	MW-108	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.109 0.120	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.712 0.463	NA NA
06-14-93 Total Metals Dissolved	MW-108	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.118 0.127	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	0.016 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.502 0.479	NA NA
11-04-93 Total Metals Dissolved	MW-108	ND [UQ]	ND * [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.114 0.124	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.570 0.568	NA NA
05-18-94 Total Metals Dissolved	MW-108	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.107 0.118	<0.004 <0.004	<0.005 <0.005	<0.010 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.282 0.280	NA NA
2-3-95 Total Metals Dissolved	MW-108	ND [UQ]	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
10-20-95 Total Metals Dissolved	MW-108	ND	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
5-31-96 Total Metals Dissolved	MW-108	ND	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
Notes: NA VOCS TPH Tr ACE BEP CHL CDS DCA DBCM ND BNAS TOL	Not analyzed. TCE Tr Volatile organi EDB EdI Total Petroleum Hydrocarbons DBCP 1,2 Trace, unquantifiable amount detected. PCB Pee Acetone. BA Be Bis(2-arbylhexyl)phthalate. DCP 1,2 Chloroform. Bo Be Carbon Disulfide * TH 1,2-Dichloroethane. J Da Not detected. U Da Base/neutral and acid extractable compounds. th Toluene. [UQ] AT Well screen submerged below top of water table. de						ethene. Dibromide mo-3-chle orobenzene Acid oropropan sent. qualitative rejected an qualitative reported. enix data o	e. oropropane e or estimat id unusable and consi of unknown	ed. e dered to b	e not detec s	ted at										

Method detection limits (mdls) for VOCs and BNAs varied. See laboratory reports for specific mdls. Metals are as follows: silver (Ag), arsenic (As), beryllium (Be), barium (Ba), cadmium (Cd), copper (Cu), chromium (Cr), mercury (Hg), nickel (Ni), lead (Pb), antimony (Sb), selenium (Se),

Groundwater Monitoring Results for the Period of Record at PSC FT-06, OU-2

thallium (Tl), and zinc (Zn).

Sampla	Location									Inorgania	Constituer	ote (ma/L)				Inora	anic Cons	tituente Co	ontinued (r	ад/I.)	
Dete	Location	VOCe	BNA 6	три	FDP	DRCP	٨a	Ac	Po	Bo	Cd	ns (ng/L)	Cu	Ца	Ni	Ph	sh Sh	So So	TI	7n 7	Bo
Date		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)	Ag		Da	БС	Cu	CI	Cu	ng	141	10	30	50		211	bu
		(ug/L)	(ug/L)	(ing/L)	(ing/L)	(ing/L)				I				<u> </u>							
01-27-92(1)	MW-109	ND	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				< 0.010	< 0.005	0.146	< 0.005	< 0.005	0.013	< 0.01	< 0.0002	< 0.020	0.002	< 0.05	< 0.005	< 0.005	0.284	NA
Dissolved			r d				< 0.010	< 0.005	0.150	< 0.005	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.290	NA
07-17-92 (1)	MW-109	ND	ND	NA	NA	NA															
Total Metals Dissolved		[UQ]	[UQ]				<0.010	<0.005	0.044	< 0.005	<0.0005	0.011	0.011	<0.0002	<0.020	0.002	<0.05	<0.005	<0.005	1.06	NA
Dissolved							-0.010	-0.005	0.125	-0.005	-0.0005	-0.010	-0.010	-0.0002	0.020	~0.002	-0.05	-0.005	-0.005	0.124	1011
11-30-92 (1)	MW-109	ND	ND	NA	< 0.01	< 0.01															
Total Metals		[UQ]	[UQ]				<0.010	<0.005	0.167	< 0.005	<0.0005	0.016	<0.010	<0.0002	<0.020	0.004	<0.05	<0.005	<0.005	0.565	<0.10
Dissolved							-0.010	-0.005	0.100	-0.005	~0.0005	-0.010	-0.010	-0.0002	~0.020	~0.002	-0.05	-0.005	-0.005	0.550	-0.10
03-24-93 (1)	MW-109	ND	ND *	NA	NA	NA															
Total Metals Dissolved		[UQ]	[UQ]				<0.010 <0.010	<0.005 <0.005	0.146	<0.005	<0.0005	<0.010	<0.010	<0.0002	<0.020 <0.020	0.003	<0.05 <0.05	<0.005 <0.005	<0.005	0.421	NA NA
Dissolved							-0.010	-0.005	0.175	-0.005	-0.000	0.011	-0.010	-0.0002	-0.020	.0.002	-0.02	-0.005	-0.005	0.155	
06-15-93 (1)	MW-109	ND	ND	NA	NA	NA															
Total Metals Dissolved		[UQ]	[UQ]				<0.010	<0.005	0.207	<0.005	<0.005	0.015	0.010	<0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005	<0.005	0.453	NA NA
Dissolved							-0.010	-0.005	0.221	-0.005	.0.005	0.015	-0.010	-0.0002	-0.020	-0.002	-0.02	-0.005	-0.002	0.112	
11-12-93 (1)	MW-109	ND	ND	NA	NA	NA	-0.010	-0.005	0.146	-0.005	-0.005	-0.010	-0.010	-0.0002	-0.020	-0.000	-0.05	-0.005	-0.005	0.000	
Dissolved		[UQ]	[UQ]				<0.010	<0.005	0.146	<0.005	<0.005	<0.010	<0.010	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0.296	NA
Dissolved							-0.010	-0.005	0.107	-0.002	-0.000	-0.010	-0.010	-0.0002	-0.020	-0.002	-0.02	-0.002	-0.005	0.277	
05-12-94 (1)	MW-109	ND	ND R	NA	NA	NA															
Dissolved		[UQ]	[UQ]				<0.010	<0.005 <0.005	0.182	< 0.004	<0.005 <0.005	<0.010 <0.010	< 0.010	<0.0002	<0.020	<0.002	<0.05 <0.05	<0.005 <0.005	<0.005	0.305	NA
02-01-95 Total Matala	MW-109	ND	NA	NA	NA	NA	NA	NA	NIA	NIA	NA	NA	NIA	NA	NIA	NIA	NA	NA	NA	NA	NA
Dissolved		[UQ]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
10-25-95 Total Metals	MW-109	ND	NA	NA	NA	NA	NA	NΔ	NΔ	NA	NΔ	NΔ	NΔ	NΔ	NΔ	NΔ	NΔ	NΔ	NΔ	NΔ	NΔ
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
		100																			
5-29-96 Total Metals	MW-109	ND	ND	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
01 22 02 (1)	MW 110	CHI 1	ND	NA	NA	NA															
Total Metals	WIW-110	[UQ]	[UQ]	INA	INA	INA	< 0.010	< 0.005	0.240	< 0.005	< 0.005	0.024	0.026	< 0.0002	< 0.020	0.005	< 0.05	< 0.005	< 0.005	0.560	NA
Dissolved			r d				< 0.010	< 0.005	0.237	< 0.005	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.220	NA
01-23-92 (1)	MW-110	CHL 1	ND	NΔ	NΔ	NΔ															
Total Metals	Duplicate	[UQ]		14/1	1411	1411	< 0.010	< 0.005	0.223	< 0.005	< 0.005	0.013	0.025	< 0.0002	< 0.020	0.003	< 0.05	< 0.005	< 0.005	0.512	NA
Dissolved							< 0.010	< 0.005	0.231	< 0.005	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.471	NA
07-15-92 (1)	MW-110	CHL 2	ND	NA	NA	NA															
Total Metals		DCA 3	[UQ]				< 0.010	< 0.005	0.125	< 0.005	< 0.0005	0.023	0.013	< 0.0002	< 0.020	0.005	< 0.05	< 0.005	< 0.005	0.422	NA
Dissolved		[UQ]					< 0.010	< 0.005	0.099	< 0.005	< 0.0005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.056	NA
12-08-92 (1)	MW-110	TOL 3	ND	NA	<0.01	0.11															
Total Metals		[UQ]	[UQ]	1111	-0.01	0.11	< 0.010	< 0.005	0.233	< 0.005	< 0.0005	0.021	0.021	< 0.0002	< 0.020	0.006	< 0.05	< 0.005	< 0.005	1.07	< 0.10
Dissolved							< 0.010	< 0.005	0.215	< 0.005	< 0.0005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.543	< 0.10

Sample	Location									Inorganic	Constitue	nts (mg/L)	-			Inorg	anic Cons	stituents Co	ontinued (r	ng/L)	
Date		VOCs	BNAs	ТРН	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
		(ug/L)	(ug/L)	(mg/L) (mg/L)	(mg/L)															
03-16-93 (1) Total Metals Dissolved	MW-110	CHL 1 TOL 2 [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.215 0.228	<0.005 <0.005	<0.005 <0.005	0.014 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.005 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.591 0.403	NA NA
03-16-93 (1) Total Metals Dissolved	MW-110 Duplicate	CHL 1 [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.213 0.223	<0.005 <0.005	<0.005 <0.005	0.013 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.006 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.501 0.486	NA NA
06-10-93(1) Total Metals Dissolved	MW-110	CHL 1 [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.214 0.226	<0.005 <0.005	<0.005 <0.005	0.013 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.003 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.457 0.393	NA NA
06-10-93 (1) Total Metals Dissolved	MW-110 Duplicate	CHL 1 [UQ]	BEP 33 [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.214 0.226	<0.005 <0.005	<0.005 <0.005	0.013 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.003 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.437 0.394	NA NA
11-09-93 (1) Total Metals Dissolved	MW-110 Time-series Initial Sample	CHL 1 [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.230 0.224	<0.005 <0.005	<0.005 <0.005	0.027 <0.010	0.012 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.009 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.530 0.180	NA NA
11-09-93 (1) Total Metals Dissolved	MW-110 Time-series 3.5-hr sample	CHL 1 [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.214 0.228	<0.005 <0.005	<0.005 <0.005	0.014 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.377 0.296	NA NA
05-12-94(1) Total Metals Dissolved	MW-110	CHL 2 [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.226 0.241	<0.004 <0.004	<0.005 <0.005	0.013 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.432 0.406	NA NA
01-31-95(1) Total Metals Dissolved	MW-110	CHL 2 [UQ]	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
5-4-95 Total Metals Dissolved	MW-110	ND [UQ]	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
5-4-95 Total Metals Dissolved	MW-110 (D)	ND [UQ]	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
2-3-96 Total Metals Dissolved	MW-110	ND	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
01-27-92 (1) Total Metals Dissolved	MW-111	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.223 0.231	<0.005 <0.005	<0.005 <0.005	0.013 <0.010	0.025 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.003 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.512 0.471	NA NA
7/15/1992 (1) Total Metals Dissolved	MW-111	DCA 1 DBCM 1 [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.207 0.135	<0.005 <0.005	0.0016 <0.0005	0.022 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.190 0.022	NA NA
11-25-92 (1) Total Metals Dissolved	MW-111	ND [UQ]	ND [UQ]	NA	<0.01	<0.01	<0.010 <0.010	<0.005 <0.005	0.298 0.319	<0.005 <0.005	<0.0005 <0.0005	0.011 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.006 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.757 0.752	0.26 0.26

Sample	Location									Inorganic	Constitue	nts (mg/L)				Inorg	ganic Cons	tituents Co	ontinued (r	ng/L)	
Date		VOCs	BNAs	TPH	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)															
03-16-93 (1) Total Metals Dissolved	MW-111	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.247 0.269	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	0.012 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.535 0.517	NA NA
06-16-93 (1) Total Metals Dissolved	MW-111	ND [UQ]	BEP 15 [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.288 0.276	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	0.024 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.003 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.722 0.570	NA NA
06-16-93 (1) Total Metals Dissolved	MW-111 Duplicate	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.272 0.277	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	0.012 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.003 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.696 0.574	NA NA
11-12-93 (1) Total Metals Dissolved	MW-111	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.263 0.285	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.440 0.416	NA NA
5/12/1994 (1) Total Metals Dissolved	MW-111	ND [UQ]	ND R [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.244 0.256	<0.004 <0.004	<0.005 <0.005	0.015 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.490 0.467	NA NA
2-1-95 (1) Total Metals Dissolved	MW-111	ND [UQ]	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
10-25-95 Total Metals Dissolved	MW-111	ND	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
5-22-96 Total Metals Dissolved	MW-111	ND	ND	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
08-04-93 Total Metals Dissolved	MW-118	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.323 0.335	<0.005 <0.005	<0.005 <0.005	0.017 <0.010	0.032 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.740 0.730	NA NA
08-04-93 Total Metals Dissolved	MW-118 Duplicate	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.320 0.328	<0.005 <0.005	<0.005 <0.005	0.020 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.714 0.708	NA NA
11-09-93 Total Metals Dissolved	MW-118	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.276 0.295	<0.005 <0.005	<0.005 <0.005	0.043 J <0.010	<0.010 <0.010	<0.0002 <0.0002	0.023 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.499 0.476	NA NA
11-09-93 Total Metals Dissolved	MW-118 Duplicate	ND [UQ]	ND * [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.294 0.322	<0.005 <0.005	<0.005 <0.005	0.020 J <0.010	0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.551 0.496	NA NA
05-17-94 Total Metals Dissolved	MW-118	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.271 0.250	<0.004 <0.004	<0.005 <0.005	0.014 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.360 0.339	NA NA
01-31-95 Total Metals Dissolved	MW-118	ND [UQ]	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA

Sample	Location									Inorganic	Constitue	nts (mg/L)				Inore	anic Cons	tituents C	ontinued (r	ng/L)	
Date	Location	VOCs	BNAs	трн	FDR	DRCP	Åσ	As	Ra	Be	Cd	Cr	Cu	Ha	Ni	Ph	Sh	Se Se	TI	Zn	Bo
Date		(ug/L)	(ng/L)	(mg/L)	(mg/L)	(mg/L)	As	113	Da	ы	Cu	ci	Cu	ns	14	10	50	54		2.11	50
		(ug/L)	(ug/L)	(ing/L)	(ing/L)	(ing/L)															
5-1-95	MW-118	ND	NA	NA	NA	NA															
Total Metals		[UQ]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
8-11-95	MW-118	ND	NA	NA	NA	NA															
Total Metals		[UQ]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
10.04.05	NOV 110	ND																			
10-26-95 Total Metals	MW-118	ND	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved		[00]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-3-96	MW-118	ND	NA	NA	NA	NA															
Total Metals		[UQ]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
5-22-96	MW-118	ND	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
11-11-97	MW-118	ND	ND	NΔ	NΔ	NΔ															
Total Metals	IVI VV -110	ND	ND	INA	NA.	INA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
5-12-98	MW-118	ND	ND	NA	NA	NA	NIA	NA	NIA	NIA	NIA	NA	NA	NA	NIA	NIA	NIA	NIA	NIA	NIA	NIA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
07-14-94	MW-123	BROM 2.3	ND	NA	NA	NA															
Total Metals		CHL 7.9 DRCM 11	[UQ]				NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolveu		[UO]					NA	INA	INA	INA	INA	NA	INA	INA	INA	INA	INA	INA	INA	INA	INA
12-07-94	MW-123	CHL 1.4	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				<0.010UJ	0.0022	0.250	< 0.004	< 0.005	0.0092U	< 0.010 U	< 0.0002	0.031U	< 0.002	< 0.05	0.003	< 0.005	0.440	NA
Dissolved							<0.010 UJ	0.0021	0.260	< 0.004	< 0.005	0.0057 U	<0.010U	< 0.0002	<0.001U	< 0.002	< 0.05	0.0029	< 0.005	0.440	NA
12-07-94	MW-123	CHL16	ND	NA	NA	NA															
Total Metals	Duplicate	[UQ]	[UQ]				<0.010UJ	0.0023	0.250	< 0.004	< 0.005	0.0098U	0.0043U	< 0.0002	0.031U	< 0.002	< 0.05	0.0024	< 0.005	0.540	NA
							<0.010UJ	0.0019	0.26	< 0.004	< 0.005	<.0029U	<.0034U	< 0.0002	0.022U	0.0012	< 0.05	0.003	< 0.005	0.41	NA
02-23-95	MW-123	CHL 2	NA	NA	NA	NA	NIA	NA	NIA	NIA	NIA	NA	NA	NIA	NIA	NIA	NIA	NIA	NIA	NIA	NIA
Dissolved		[UQ]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissorred																					
5-4-95	MW-123	ND	NA	NA	NA	NA															
Total Metals		[UQ]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
8-11-95	MW-123	ND	NA	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
10 26 95	MW 122	ND	NA	NA	NA	NA															
Total Metals	141 99 -123	ND	110	inn.	nn.	in a	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Sample	Location									Inorganic	Constitue	nts (mg/L)				Inorg	ganic Cons	tituents Co	ontinued (r	ng/L)	
--------------	----------------	--------------	--------	--------	----------	-------------	------------	------------	----------	-----------	-----------	------------	----	----	----	-------	------------	-------------	-------------	-------	-----
Date		VOCs	BNAs	TPH	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)	-														
10-26-95	MW-123(D)	ND	NA	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	INA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	INA
2-6-96	MW-123	ND	NA	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
5-23-96	MW-123	ND	ND	NΔ	NΔ	NΔ															
Total Metals	1111-125	ПD	нь	14/1	101	1471	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
11-11-97	MW-123	CHL 2.7J	NA	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
5-13-98	MW-123	CHL 2.4J	NA	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
		CHL 1.6J																			
11-05-98	MW-123	MC 0.32JB	NA	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	INA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
5-19-99	MW-123	ND	NA	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Notes: NA	Not analyzed.	,		ND	Not dete	ected.		. 11													
vocs	voiauie organi	c compounds.		BINAS	Base/ne	utrai and a	icia extra	clable com	ipounds.												

VOCs	Volatile organic compounds.	BNAs	Base/neutral and acid extractable compounds
TPH	Total Petroleum Hydrocarbons	TOL	Toluene.
Tr	Trace, unquantifiable amount detected.	TCE	Trichloroethene.
ACE	Acetone.	EDB	Ethylene Dibromide.
BEP	Bis(2-ethylhexyl)phthalate.	DBCP	1,2-Dibromo-3-chloropropane.
CHL	Chloroform.	PCB	Pentachlorobenzene
CDS	Carbon Disulfide	BA	Benzoic Acid
DCA	1,2-Dichloroethane.	DCP	1,2-Dichloropropane
DBCM	Dibromochloromethane.	Bo	Boron.
MC	Methylene Chloride	[UQ]	ATI-Phoenix data of unknown quality as
*	TICs present.		determined by the FFA Parties
J	Data are qualitative or estimated.		
R	Data are rejected and unusable		
II.	Data are qualitative and considered to be	not data	atad at the velue reported

U Data are qualitative and considered to be not detected at the value reported.

(1) Well screen submerged below top of water table.

Method detection limits (mdls) for VOCs and BNAs varied. See laboratory reports for specific mdls. Metals are as follows: silver (Ag), arsenic (As), beryllium (Be), barium (Ba), cadmium (Cd), copper (Cu), chromium (Cr), mercury (Hg), nickel (Ni), lead (Pb), antimony (Sb), selenium (Se), thallium (Tl), and zinc (Zn).

Sample	Location		Orgai	ic Const	ituents	-			Inorganic (Constituen	ts (mg/L)					Inorganic	Constitu	ents Conti	nued (mg/l	L)	
Date		VOCs	BNAs	ТРН	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)															
07-23-92 Total Metals Dissolved	MW-115	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	0.008 0.006	0.064 0.052	<0.005 <0.005	<0.0005 <0.0005	0.043 <0.010	0.068 <0.010	<0.0002 <0.0002	0.034 <0.020	0.004 J <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.205 0.182	NA NA
07-23-92 Total Metals Dissolved	MW-115 Duplicate	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	0.007 0.006	0.071 0.051	<0.005 <0.005	<0.0005 <0.0005	0.058 <0.010	0.096 0.012	<0.0002 <0.0002	0.042 0.024	0.018 J <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.236 0.182	NA NA
11-17-92 Total Metals Dissolved	MW-115	ND [UQ]	ND [UQ]	NA	<0.01	<0.01	<0.010 <0.010	0.006 0.006	0.048 0.053	<0.005 <0.005	<0.0005 <0.0005	0.024 0.014	<0.010 <0.010	<0.0002 <0.0002	0.029 0.026	0.008 0.008	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.860 0.811	<0.10 <0.10
03-25-93 Total Metals Dissolved	MW-115	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	0.007 0.005	0.049 0.193	<0.005 <0.005	<0.005 <0.005	0.018 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.003 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.422 0.302	NA NA
06-17-93 Total Metals Dissolved	MW-115	ND [UQ]	BEP 63 [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.054 0.063	<0.005 <0.005	<0.005 <0.005	0.016 0.013	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.373 0.445	NA NA
11-08-93 Total Metals Dissolved	MW-115	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	0.006 0.006	0.053 0.055	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.529 0.551	NA NA
05-19-94 Total Metals Dissolved	MW-115	ND [UQ]	BEP 5 J [UQ]	NA	NA	NA	<0.010 <0.010	0.007 0.006	0.051 0.057	<0.004 <0.004	<0.005 <0.005	0.012 0.011	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.003 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.232 0.235	NA NA
02-22-95 Total Metals Dissolved	MW-115	ND [UQ]	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
10-24-95 Total Metals Dissolved	MW-115	ND	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
02-06-96 Total Metals Dissolved	MW-115	ND	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
07-23-92 Total Metals Dissolved	MW-116	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	0.016 0.015	0.045 0.021	<0.005 <0.005	<0.0005 <0.0005	0.016 0.010	0.276 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.017 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.437 0.232	NA NA
07-23-92 Total Metals Dissolved	MW-116 Duplicate	ND [UQ]	ND [UQ]	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA

11-18-92 MW-116 ND [UQ] ND [UQ] NA <0.01 <0.01

Sample	Location		Orga	nic Const	ituents				Inorganic	Constituen	ts (mg/L)					Inorgani	c Constitu	ents Conti	nued (mg/l	L)	
Date		VOCs	BNAs	TPH	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)															
Total Metals							< 0.010	0.017	0.020	< 0.005	< 0.0005	0.019	< 0.010	< 0.0002	< 0.020	0.004	< 0.05	< 0.005	< 0.005	0.414	< 0.10
Dissolved							< 0.010	0.014	0.020	< 0.005	< 0.0005	0.012	< 0.010	< 0.0002	< 0.020	0.004	< 0.05	< 0.005	< 0.005	0.414	< 0.10
3/23/93	MW-116	ND [UQ]	ND [UQ]	NA	NA	NA															
Total Metals							< 0.010	0.015	0.022	< 0.005	< 0.005	0.014	0.013	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.782	NA
Dissolved							< 0.010	0.014	0.035	< 0.005	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.687	NA
06 17 02	MW 116			NA	NA	NA															
Total Metale	IVI VV - 1 1 O	ND[UQ]	ND[UQ]	INA	INA	INA	<0.010	0.012	0.032	<0.005	<0.005	0.013	<0.010	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0.758	NA
Dissolved							<0.010	0.012	0.034	<0.005	<0.005	<0.015	<0.010	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0.758	NA
Dissolved							~0.010	0.005	0.054	~0.005	~0.005	~0.010	~0.010	<0.0002	<0.020	~0.002	~0.05	~0.005	~0.005	0.054	11A
11-08-93	MW-116	ND [UQ]	ND [UQ]	NA	NA	NA															
Fotal Metals							< 0.010	0.014	0.026	< 0.005	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.548	NA
Dissolved							< 0.010	0.014	0.026	< 0.005	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.584	NA
05-19-94	MW-116	ND [UQ]	BEP 8 J	NA	NA	NA															
Fotal Metals			[UQ]				< 0.010	0.016	0.023	< 0.004	< 0.005	0.016	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.382	NA
Dissolved							< 0.010	0.016	0.022	< 0.004	< 0.005	0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.384	NA
02-22-05	MW-116	ND IUOI	NA	NA	NA	NA															
Total Metals	101 00 -1110	ND[0Q]	nn.	na.	nn.	na.	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissorred																					
10-24-95	MW-116	ND	NA	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
06-01-96	MW-116	ND	ND	NA	NA	NA															
Fotal Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
12-07-94	MW-124	ND ILIO1	ND IUO1	NA	NA	NA															
Fotal Metals							<0.010 UJ	0.0068	0.047	< 0.004	< 0.005	0.018U	<0.010U	< 0.0002	<.002U	0.0013	< 0.05	0.0024	< 0.005	0.130	NA
Dissolved							<0.010UJ	0.0065	0.041	< 0.004	< 0.005	0.012U	<0.010U	< 0.0002	<0.02U	< 0.001	< 0.05	0.002	< 0.005	0.120	NA
02-23-95	MW-124	ND [UQ]	NA	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
5-4-95	MW-124	ND [UQ]	NA	NA	NA	NA															
Fotal Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
8-14-05	MW-124	ND	NA	NA	N۸	N۸															
Fotal Metals	191 99 -124	14D	11/1	11/1	1474	11/1	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA

Sample	Location		Orga	nic Const	ituents				Inorganic	Constituent	s (mg/L)					Inorganic	Constitue	ents Conti	nued (mg/l	.)	
Date		VOCs	BNAs	ТРН	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)															
10-24-95	MW-124	ND	NA	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-08-96	MW-124	ND	NA	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
6-5-96	MW-124	ND	ND	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
11-11-97	MW-124	ND	NA	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
5-12-98	MW-124	ND	NA	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA

Notes:

NA	Not analyzed.	TCE	Trichloroethene.
VOCs	Volatile orga	EDB	Ethylene Dibromide.
TPH	Total Petroleum Hydrocarbons	DBCP	1,2-Dibromo-3-chloropropane.
Tr	Trace, unquantifiable amount dete	ect PCB	Pentachlorobenzene
ACE	Acetone.	BA	Benzoic Acid
BEP	Bis(2-ethylhexyl)phthalate.	DCP	1,2-Dichloropropane
CHL	Chloroform.	Во	Boron.
CDS	Carbon Disulfide	*	TICs present.
DCA	1,2-Dichloroethane.	J	Data are qualitative or estimated.
DBCM	Dibromochloromethane.	R	Data are rejected and unusable
ND	Not detected.	U	Data are qualitative and considered to be not detected at the
BNAs	Base/neutral and acid extractable	compound	s value reported.
TOL	Toluene.		
IT IOI		1. 1	III d EEAD C

ATI-Phoenix data of unknown quality as determined by the FFA Parties [UQ]

(1) Well screen submerged below top of water table.

Method detection limits (mdls) for VOCs and BNAs varied. See laboratory reports for specific mdls.

Metals are as follows: silver (Ag), arsenic (As), beryllium (Be), barium (Ba), cadmium (Cd),

copper (Cu), chromium (Cr), mercury (Hg), nickel (Ni), lead (Pb), antimony (Sb), selenium (Se), thallium (Tl), and zinc (Zn).

Sample	Location		Organ	nic Const	tituents					Inorganic	Constituer	ts (mg/L)				Ino	rganic Con	stituents Co	ntinued (m	g/L)	
Date		VOCs	BNAs	TPH	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)															
12-12-91	MW-102	ND	BEP 14	NA	NA	NA															
Total Metals		[UQ]	[UQ]				< 0.010	0.026	0.022	< 0.005	< 0.005	0.026	0.025	< 0.0002	< 0.020	0.004	< 0.05	< 0.005	< 0.005	0.025	NA
Dissolved							< 0.010	0.028	0.017	< 0.005	< 0.005	0.020	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	< 0.010	NA
07-18-92	MW-102	ND	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				< 0.010	0.016	0.021	< 0.005	< 0.0005	0.027	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.022	NA
Dissolved							< 0.010	0.016	0.019	< 0.005	< 0.0005	0.021	< 0.010	< 0.0002	<0.020	< 0.002	<0.05	< 0.005	< 0.005	< 0.010	NA
11-23-92	MW-102	ND	ND	NA	<0.01	<0.01															
Total Metals		TUOI	TUOI		-0.01	-0.01	<0.010	0.015	0.019	<0.005	<0.0005	0.024	<0.010	<0.0002	<0.020	0.006	<0.05	<0.005	<0.005	0 492	0.19
Dissolved		1.4 41	1.40				< 0.010	0.015	0.018	< 0.005	< 0.0005	0.022	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.216	0.19
03-11-93	MW-102	ND	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				< 0.010	0.020	0.014	< 0.005	< 0.005	0.023	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.147	NA
Dissolved							< 0.010	0.019	< 0.010	< 0.005	< 0.005	0.012	< 0.010	< 0.0002	< 0.020	< 0.002	<0.05	< 0.005	< 0.005	0.058	NA
06-11-93	MW-102	ND	ND	NΔ	NA	NΔ															
Total Metals		TUOI	TUOI				<0.010	0.016	0.018	<0.005	<0.005	0.018	0.016	<0.0002	<0.020	0.003	<0.05	<0.005	<0.005	1.66	NA
Dissolved		1.4 41	1.40				< 0.010	0.015	0.018	< 0.005	< 0.005	0.016	< 0.010	< 0.0002	< 0.020	0.002	< 0.05	< 0.005	< 0.005	0.296	NA
11-11-93	MW-102	ND	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				< 0.010	0.016	0.017	< 0.005	< 0.005	0.023	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	1.06	NA
Dissolved							< 0.010	0.015	0.018	< 0.005	< 0.005	0.022	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.447	NA
05 20 04	NUV 102	ND	NID	NIA	NIA	NIA															
Total Metals	WIW-102	IUO1		INA	INA	INA	<0.010	0.016	0.015	<0.004	<0.005	0.027	<0.010	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0.329	NΔ
Dissolved		[00]	[00]				<0.010	0.016	0.015	<0.004	<0.005	0.027	<0.010	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0.084	NA
Billsonrou							-0.010	0.010	0.012	-0.001	.0.002	0.020	-0.010	-0.0002	-0.020	-0.002	-0.05	-0.005	.0.005	0.001	
02-06-95	MW-102	ND	NA	NA	NA	NA															
Total Metals		[UQ]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
10 11 95	MW 102	ND	NA	NA	NA	NA															
Total Metals	WI W-102	ND	INA	INA	na.	INA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
5-31-96	MW-102	ND	ND	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
12-09-91 (1)	MW-103	ND	BEP 32	NA	NA	NA															
Total Metals		[UQ]	[UQ]				< 0.010	0.015	0.092	< 0.005	< 0.005	0.025	0.027	< 0.0002	< 0.020	0.003	< 0.05	< 0.005	< 0.005	0.338	NA
Dissolved							< 0.010	0.014	0.090	< 0.005	< 0.005	0.015	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.321	NA
12-09-91 (1)	MW-103	ND	ND	NA	NA	NA															
Total Metals	Duplicate	[UQ]	[UQ]				< 0.010	0.014	0.090	< 0.005	< 0.005	0.024	0.043	< 0.0002	< 0.020	0.004	< 0.05	< 0.005	< 0.005	0.374	NA
Dissolved															0.000		0.0.5	0.005			NT A
							< 0.010	0.014	0.090	< 0.005	< 0.005	0.015	< 0.010	<0.0002	<0.020	< 0.002	<0.05	<0.005	<0.005	0.319	NA
7-18-92(1)	MW-103	ND	ND	NA	NA	NA	<0.010	0.014	0.090	<0.005	<0.005	0.015	<0.010	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0.319	NA

<0.010 0.013 0.086 <0.005 <0.0005 0.018 <0.010 <0.0002 <0.020 <0.002 <0.05 <0.005 <0.005 <0.010

Dissolved

NA

VOCs (ug/L)

ND

[UQ]

ND

[UQ]

ND

[UQ]

ND

[UQ]

ND

[UQ]

ND

[UQ]

ND

Location

MW-103

MW-103

MW-103

MW-103

MW-103

MW-103

MW-103

Sample

Date

11-24-92(1)

Total Metals

Dissolved 03-11-93 (1)

Total Metals

Dissolved 06-11-93 (1)

Total Metals

Dissolved

11-06-93 (1)

Total Metals

Dissolved 5/20/94 (1)

Total Metals

Dissolved 02-06-95

Total Metals

Dissolved 10-10-95

Total Metals Dissolved

Organ	ic Const	ituents				1	Inorganic	Constituen	ts (mg/L)				Ino	rganic Con	stituents Co	ntinued (mg	g/L)	
BNAs	TPH	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
(ug/L)	(mg/L)	(mg/L)	(mg/L)															
ND	NA	< 0.01	<0.01															
[UQ]				< 0.010	0.019	0.040	< 0.005	< 0.0005	0.024	< 0.010	< 0.0002	< 0.020	0.004	< 0.05	< 0.005	< 0.005	0.474	0.19
				< 0.010	0.016	0.037	< 0.005	< 0.0005	0.011	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.381	0.18
ND	NIA	NIA	NIA															
IUOI	INA	INA	INA	<0.010	0.023	0.032	<0.005	<0.005	0.016	<0.010	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0.150	NA
[00]				<0.010	0.020	0.032	<0.005	<0.005	0.014	<0.010	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0.104	NA
				-0.010	0.020	0.052	-0.005	-0.005	0.014	-0.010	-0.0002	-0.020	-0.002	-0.05	-0.005	~0.005	0.104	1011
ND	NA	NA	NA															
[UQ]				< 0.010	0.018	0.032	< 0.005	< 0.005	0.014	0.015	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.090	NA
				< 0.010	0.015	0.036	< 0.005	< 0.005	0.014	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.132	NA
ND	NA	NA	NA															
[UQ]				< 0.010	0.020	0.033	<0.005	<0.005	0.020	< 0.010	< 0.0002	< 0.020	< 0.002	<0.05	<0.005	<0.005	0.148	NA
				<0.010	0.019	0.032	<0.005	<0.005	0.017	<0.010	<0.0002	<0.020	< 0.002	<0.05	<0.005	<0.005	0.095	NA
ND	NA	NA	NA															
шот				<0.010	0.025	0.032	<0.004	<0.005	0.026	<0.010	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0.079	NA
1				< 0.010	0.023	0.028	< 0.004	< 0.005	0.022	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.059	NA
NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
				NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
				NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
ND	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
				NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
ND	NA	NA	NA															
[UQ]				< 0.010	0.011	0.470	< 0.005	< 0.005	0.062	0.073	< 0.0002	0.035	0.048	< 0.05	< 0.005	< 0.005	0.501	NA
				< 0.010	0.008	0.066	< 0.005	< 0.005	0.011	< 0.01	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.152	NA
ND	NA	NA	NA				0.00-				0.000-				0.00-			
[UQ]				< 0.010	0.010	0.119	< 0.005	< 0.0005	0.020	0.082	< 0.0002	< 0.020	0.011	<0.05	< 0.005	< 0.005	0.749	NA
				< 0.010	0.008	0.078	< 0.005	<0.0005	0.014	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.426	NA

6-6-96	MW-103	ND	ND	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved																					
12-09-91	MW-112S	TCE Tr	ND	NA	NA	NA															
Total Metals		TOL Tr	[UQ]				< 0.010	0.011	0.470	< 0.005	< 0.005	0.062	0.073	< 0.0002	0.035	0.048	< 0.05	< 0.005	< 0.005	0.501	NA
Dissolved		[UQ]					< 0.010	0.008	0.066	< 0.005	< 0.005	0.011	< 0.01	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.152	NA
07-22-92	MW-112S	ND	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				< 0.010	0.010	0.119	< 0.005	< 0.0005	0.020	0.082	< 0.0002	< 0.020	0.011	< 0.05	< 0.005	< 0.005	0.749	NA
Dissolved							< 0.010	0.008	0.078	< 0.005	< 0.0005	0.014	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.426	NA
12-01-92	MW-112S	TCE 1	ND	NA	< 0.01	< 0.01															
Total Metals		[UQ]	[UQ]				< 0.010	0.007	0.075	< 0.005	< 0.0005	0.011	< 0.010	< 0.0002	< 0.020	0.002	< 0.05	< 0.005	< 0.005	0.089U	0.22
Dissolved							< 0.010	0.007	0.075	< 0.005	< 0.0005	0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.064U	0.22
03-19-93	MW-112S	TCE 1	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				< 0.010	0.010	0.291	< 0.005	< 0.0005	0.026	0.012	< 0.0002	< 0.020	0.004	< 0.05	< 0.005	< 0.005	0.378	NA
Dissolved							< 0.010	0.008	0.106	< 0.005	< 0.005	0.013	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.134	NA
06-08-93 (1)	MW-112S	TCE 1	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				< 0.010	0.009	0.078	< 0.005	< 0.005	0.013	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.149	NA
Dissolved							< 0.010	0.007	0.090	< 0.005	< 0.005	0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.149	NA

Sample	Location		Organ	ic Const	ituents					Inorgania	Constituer	ts (mg/L)				Ino	reanic Con	tituents Co	ntinued (mo	vT.)	
Deto	Location	VOCe	DNAs	три	FDP	DRCP	٨a	Ac	Po	Bo	Cd		Cn	Ца	Ni	Ph	Sh	So So	TI	7n	Po
Date		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)	Ag	A3	Ба	БС	Cu	CI	Cu	ng	141	10	30	30		Za	Bu
11-03-93 (1)	MW-112S	ND	(ug/L) BA 40 I	NA	NA	NA															
Total Metals	1111-1125	[10]	LTIO1	101	1474	1471	<0.010	0.010	0.086	<0.005	<0.005	0.016	<0.010	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0.620	NΔ
Dissolved		[00]	[00]				<0.010	0.007	0.080	<0.005	<0.005	<0.010	<0.010	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0.020	NA
Dissolved							<0.010	0.007	0.089	<0.005	~0.005	<0.010	~0.010	<0.0002	~0.020	~0.002	~0.05	~0.005	~0.005	0.239	NA.
05-10-94	MW-112S	ND	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				< 0.010	0.026	0.097	< 0.004	< 0.005	0.053	0.022	< 0.0002	< 0.020	0.008	< 0.05	< 0.005	< 0.005	0.153	NA
Dissolved							< 0.010	0.009	0.038	< 0.004	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	< 0.020	NA
02-07-95	MW-112S	ND	NA	NA	NA	NA															
Total Metals		[UQ]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
5-3-95	MW-112S	ND	NA	NA	NA	NA															
Total Metals		[UQ]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
8-8-95	MW-112S	ND	NA	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
10 (05	MW 1120	ND	NIA	NIA	NIA	NA															
Total Matala	MW-1125	ND	NA	NA	NA	INA	NIA	NIA	NIA	NIA	NIA	NIA	NIA	NT A	NIA	NA	NIA	NIA	NA	NIA	NTA
Discolar Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	INA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-8-96	MW-112S	ND	NA	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
5-30-96	MW-112S	ND	NA	ND	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
11-12-97	MW-112S	ND	NA	ND	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
5-11-98	MW-112S	TCE 1.5J	NA	ND	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
		ACE 1.4J																			
		DCA 0.28J																			
		MC 0.24J																			
12-22-98	MW-1128	TCE 0.555 TCE 1.2J	NA	ND	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
05-19-99	MW-1128	ND	NA	ND	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA

Sample Date 12-10-91 (1)

Total Metals Dissolved 7-22-92(1)

Total Metals Dissolved 12-01-92(1)

Total Metals Dissolved 03-19-93 (1)

Total Metals Dissolved 06-08-93 (1)

Total Metals Dissolved 11-03-93 (1)

Total Metals Dissolved 5-10-94 (1)

Total Metals Dissolved 5-10-94(1)

Total Metals

Dissolved 02-07-95

Total Metals Dissolved 02-07-95

Total Metals Dissolved 10-6-95

Total Metals Dissolved 5-30-96

Total Metals

Dissolved

11-10-97

Total Metals

Dissolved

MW-112D

BF 1.1J

NA

ND NA

Location		Organ	ic Const	tituents					Inorganie	c Constituer	ts (mg/L)				Ino	rganic Con	stituents Co	ntinued (m	g/L)	
	VOCs	BNAs	TPH	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
	(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)						-		-							
MW-112D	ND	ND	NA	NA	NA															
	[UQ]	[UQ]				< 0.010	0.009	0.044	< 0.005	< 0.005	0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.101	NA
						< 0.010	0.009	0.035	< 0.005	< 0.005	0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.099	NA
MW-112D	ND	ND	NA	NA	NA															
	[UQ]	[UQ]				< 0.010	0.016	0.097	< 0.005	< 0.0005	0.032	0.084	< 0.0002	< 0.020	0.009	< 0.05	< 0.005	< 0.005	0.208	NA
						< 0.010	0.007	0.039	< 0.005	< 0.0005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.025	NA
MW 112D	ND	ND	NLA	-0.01	-0.01															
MW-112D	ND	ND	NA	<0.01	<0.01	<0.010	0.017	0.07(-0.005	<0.0005	0.022	<0.010	-0.0002	<0.020	0.002	-0.05	<0.005	<0.005	0.1401	0.22
	[UQ]	[UQ]				<0.010	0.017	0.076	<0.005	<0.0005	0.032	<0.010	<0.0002	<0.020	0.003	<0.05	<0.005	<0.005	0.1400	0.25
						<0.010	0.006	0.055	<0.005	<0.0005	<0.010	<0.010	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0.038	0.22
MW-112D	ND	ND	NA	NA	NA															
	[UQ]	[UQ]				< 0.010	0.013	0.065	< 0.005	< 0.005	0.027	< 0.010	< 0.0002	< 0.020	0.005	< 0.05	< 0.005	< 0.005	0.162	NA
						< 0.010	0.008	0.056	< 0.005	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	< 0.010	NA
MW-112D	ND	ND	NA	NA	NA															
	[UQ]	[UQ]				< 0.010	0.016	0.072	< 0.005	< 0.005	0.030	0.012	< 0.0002	< 0.020	0.003	< 0.05	< 0.005	< 0.005	0.151	NA
						< 0.010	0.007	0.045	< 0.005	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.014	NA
MW-112D	ND	ND	NA	NA	NA	-0.010	0.022	0.005	-0.005	-0.005	0.040	-0.010	-0.000	-0.020	0.004	-0.05	-0.005	-0.005	0.1.40	
	[UQ]	[UQ]				< 0.010	0.023	0.095	< 0.005	< 0.005	0.049	< 0.010	< 0.0002	< 0.020	0.004	< 0.05	< 0.005	< 0.005	0.149	NA
						< 0.010	0.009	0.039	<0.005	<0.005	< 0.010	< 0.010	<0.0002	<0.020	0.002	<0.05	< 0.005	<0.005	<0.010	NA
MW 112D	ND	ND	NIA	NIA	NIA															
WIW-112D		IUOI	INA	INA	INA	<0.010	0.000	0.005	<0.004	<0.005	0.018	0.029	<0.0002	<0.020	0.010	<0.05	<0.005	<0.005	0.600	NA
	[00]	[UQ]				<0.010	0.009	0.095	<0.004	<0.005	0.012	<0.029	<0.0002	<0.020	<0.010	<0.05	<0.005	<0.005	0.099	NA
						-0.010	0.000	0.007	-0.004	~0.005	0.012	-0.010	×0.0002	-0.020	-0.002	-0.05	~0.005	-0.005	0.204	1474
MW-112D	ND	ND	NA	NA	NA															
Duplicate	[UQ]	[UQ]				< 0.010	0.009	0.096	< 0.004	< 0.005	0.014	0.020	< 0.0002	< 0.020	0.007	< 0.05	< 0.005	< 0.005	0.715	NA
·						< 0.010	0.008	0.086	< 0.004	< 0.005	0.014	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.269	NA
MW-112D	ND	NA	NA	NA	NA															
	[UQ]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
						NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
MW-112D	ND	NA	NA	NA	NA															
	[UQ]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
						NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
MW-112D	ND	NA	NA	NA	NA															
						NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
						NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
MW-112D	ND	ND	NA	NA	NA															

NA

Sample	Location		Organ	ic Const	ituents					Inorganie	c Constituer	nts (mg/L)				Ino	rganic Con	stituents Co	ntinued (m	2/L)	
Date		VOCs	BNAs	ТРН	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)	-							_							
5-11-98	MW-112D	ND	NA	ND	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
01-24-92	MW-113	ND	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				< 0.010	0.009	0.064	< 0.005	< 0.005	0.027	0.016	< 0.0002	< 0.020	0.004	< 0.05	< 0.005	< 0.005	0.327	NA
Dissolved							< 0.010	0.009	0.016	< 0.005	< 0.005	0.013	< 0.010	< 0.0002	< 0.020	0.004	< 0.05	< 0.005	< 0.005	0.212	NA
07-21-92	MW-113	TCE 1	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				<0.010	0.020	0.447	< 0.005	< 0.0005	0.109	0.174	< 0.0002	0.071	0.023	<0.05	<0.005	<0.005	0.470	NA
Dissolved							< 0.010	0.008	0.014	< 0.005	<0.0005	0.012	< 0.010	< 0.0002	0.030	< 0.002	<0.05	<0.005	<0.005	0.018	NA
12-17-92	MW-113	DCA 1	ND	NA	< 0.01	<0.01															
Total Metals		TCE 2	[UQ]				<0.010	0.012	0.077	<0.005	< 0.0005	0.041	0.017	< 0.0002	< 0.020	0.005	<0.05	<0.005	<0.005	0.626	0.19
Dissolved		TOL 1					< 0.010	0.011	0.014	< 0.005	< 0.0005	0.012	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.299	0.21
		[UQ]																			
03-18-93	MW-113	DCA 1	ND	NΔ	NA	NΔ															
Total Metals	Time-series	TCE 2	IUOI	1471	101	1474	<0.010	0.010	0.019	<0.005	<0.005	0.019	<0.010	<0.0002	<0.020	0.002	<0.05	<0.005	<0.005	0.355	NA
Discolved	Initial Sample	TUO1	[00]				<0.010	0.010	0.015	<0.005	<0.005	0.019	<0.010	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0.355	NA
Dissolved	mitiai Sample	[00]					<0.010	0.010	0.015	~0.005	<0.005	0.014	~0.010	~0.0002	~0.020	~0.002	~0.05	<0.005	~0.005	0.250	nn.
03-18-93	MW-113	DCA 1	ND	NA	NA	NA															
Total Metals	Time-series	TCE 2	[UO]				<0.010	0.010	0.020	<0.005	<0.005	0.017	0.011	<0.0002	<0.020	0.002	<0.05	<0.005	<0.005	0.326	NA
Dissolved	Duplicate	[UO]	[00]				<0.010	0.010	0.014	<0.005	<0.005	0.016	<0.011	<0.0002	<0.020	0.004	<0.05	<0.005	<0.005	0.181	NA
Dissorred	Bupfieute	[04]					-0.010	0.010	0.011	-0.002	-0.002	0.010	-0.010	.0.0002	.0.020	0.001	-0.05	-0.002	-0.002	0.101	
03-18-93	MW-113	DCA 1	ND	NA	NA	NA															
Total Metals	Time-series	TCE 2	[UQ]				< 0.010	0.011	0.014	< 0.005	< 0.005	0.019	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.333	NA
Dissolved	4-hr Sample	[UQ]					< 0.010	0.010	0.014	< 0.005	< 0.005	0.014	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.288	NA
06-07-93 (1)	MW-113	DCA 1	ND	NA	NA	NA															
Total Metals		TCE 2	[UQ]				< 0.010	0.010	0.017	< 0.005	< 0.005	0.020	< 0.010	< 0.0002	< 0.020	0.002	< 0.05	< 0.005	< 0.005	0.188	NA
Dissolved		[UQ]					< 0.010	0.010	0.015	< 0.005	< 0.005	0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.137	NA
06-07-93 (1)	MW-113	DCA 1	ND	NA	NA	NA															
Total Metals	Duplicate	TCE 2	[UQ]				< 0.010	0.011	0.017	< 0.005	< 0.005	0.022	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.192	NA
Dissolved		[UQ]					< 0.010	0.011	0.015	< 0.005	< 0.005	0.015	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.130	NA
11.02.02.(1)	MW 112	DCA 1	ND	NIA	NIA	NIA															
T1-02-93 (1)	MW-113	DCA I	ND	NA	NA	NA	-0.010	0.011	0.017	-0.005	-0.005	0.101	-0.010	-0.0002	0.052	-0.002	-0.05	-0.005	-0.005	0.701	
I otal Metals		TUO1	[UQ]				<0.010	0.011	0.016	< 0.005	<0.005	0.121	<0.010	<0.0002	0.052	< 0.002	<0.05	<0.005	< 0.005	0.721	NA
Dissolved		[UQ]					<0.010	0.008	0.015	<0.005	<0.005	0.014	<0.010	<0.0002	0.022	<0.002	<0.05	<0.005	<0.005	0.652	NA
5/9/94 (1)	MW-113	TCE 2	ND	NA	NA	NA															
Total Metals	MIW-115	TUO1	TUOI	1471	101	1474	<0.010	0.012	0.018	<0.004	<0.005	0.042	<0.010	<0.0002	0.025	<0.002	<0.05	<0.005	<0.005	0.628	NA
Dissolved		[00]	[00]				<0.010	0.009	0.017	<0.004	<0.005	<0.010	<0.010	<0.0002	<0.025	<0.002	<0.05	<0.005	<0.005	0.595	NA
Dissoured							-0.010	0.007	0.017	-0.004	-0.005	-0.010	-0.010	-0.0002	-0.020	-0.002	-0.05	-0.005	-0.005	0.070	1471
02-11-95	MW-113	TCE 2	NA	NA	NA	NA															
Total Metals		[UQ]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
5-3-95	MW-113	DCP 4	NA	NA	NA	NA															
Total Metals		[UQ]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA

Sample	Location		Organ	nic Const	ituents					Inorganic	c Constituer	nts (mg/L)				Inc	rganic Con	stituents Co	ntinued (m	g/L)	
Date		VOCs	BNAs	TPH	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)															
8-8-95	MW-113	ND	NA	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
10-10-95	MW-113	ND	NA	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
10-10-95	MW-113	ND	NA	NA	NA	NA															
Total Metals	DUPLICATE	112					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved	Dertheitit						NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-8-96	MW-113	ND	NA	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-8-96	MW-113	ND	NA	NA	NA	NA															
Total Metals	DUPLICATE						NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
5 24 96	MW 112	ND	ND	NA	NA	NA															
Total Metals	WIW-115	ND	ND	INA	INA	NA.	NA	NA	NΔ	NA	NA	NA	NΔ	NA	NΔ	NA	NA	NA	NA	NA	NΔ
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							Turi,	1471	101	101	1471	101	1011	1471	101	1111	11/1	1111	1011	1414	11/1
11-12-97	MW-113	ND	ND	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
11-12-97	MW-113	ND	ND	NA	NA	NA															
Total Metals	Duplicate						NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
5-12-98	MW-113	TCE 1.9J	ND	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
		DCA 0.30J																			
		MC 0.40JB PCE 0 181																			
11-03-98	MW-113	TCE 1.6J	NA	ND	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA

Sample

Date

5-19-99

Total Metals

Dissolved

Notes: NA

VOCs

TPH

	Location		Orga	nic Cons	tituents					Inorganic	Constituer	nts (mg/L)				Inc	rganic Cons	stituents Co	ntinued (m	g/L)
		VOCs	BNAs	TPH	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)														
	MW-113	CHL 5.0 BRMO 6.0	NA	ND	NA	NA														
							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	
							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA]
Not analyzed. ND Not detected					cted.															
Volatile organic compounds. BNAs Base/neutral and						tral and ac	id extractat	ole compour	nds.											
Total Petroleum Hydrocarbons TOL Toluene.																				
Trace, unquantifiable amount detected.				TCE	Trichlore	oethene.														
Acetone. E				EDB	Ethvlene	Dibromide														

	···· ·· · · · · · · · · · · · · · ·		
Tr	Trace, unquantifiable amount detected.	TCE	Trichloroethene.
ACE	Acetone.	EDB	Ethylene Dibromide.
BEP	Bis(2-ethylhexyl)phthalate.	DBCP	1,2-Dibromo-3-chloropropane.
CHL	Chloroform.	PCB	Pentachlorobenzene
CDS	Carbon Disulfide	BA	Benzoic Acid
DCA	1,2-Dichloroethane.	DCP	1,2-Dichloropropane
DBCM	Dibromochloromethane.	Во	Boron.
MC	Methylene Chloride	PCE	Tetrachloroethelene
Bromo	Bromodichloromethene	[UQ]	ATI-Phoenix data of unknown quality as
*	TICs present.		determined by the FFA Parties
J	Data are qualitative or estimated.		

R Data are rejected and unusable

U Data are qualitative and considered to be not detected at the value reported.

(1) Well screen submerged below top of water table.

Method detection limits (mdls) for VOCs and BNAs varied. See laboratory reports for specific mdls. Metals are as follows: silver (Ag), arsenic (As), beryllium (Be), barium (Ba), cadmium (Cd), copper (Cu), chromium (Cr), mercury (Hg), nickel (Ni), lead (Pb), antimony (Sb), selenium (Se), thallium (Tl), and zinc (Zn).

Zn

NA

NA

Bo

NA

NA

Sample	Location		Organ	ic Const	ituents					Inorgani	c Constituer	ts (mg/L)				Inc	organic Con	stituents Co	ontinued (m	g/L)	
Date		VOCs	BNAs	TPH	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)	_														
01-21-92(1)	MW-101	ND	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				< 0.010	< 0.005	0.103	< 0.005	< 0.005	< 0.010	0.026	< 0.0002	< 0.020	0.002	< 0.05	< 0.005	< 0.005	0.229	NA
Dissolved							< 0.010	< 0.005	0.104	< 0.005	< 0.005	< 0.010	0.034	< 0.0002	< 0.020	0.003	< 0.05	< 0.005	< 0.005	0.236	NA
01-21-92(1)	MW-101	ND	ND	NΔ	NΔ	NΔ															
Total Metals	Duplicate	[UQ]	[UQ]	1474	1471	101	< 0.010	< 0.005	0.100	< 0.005	< 0.005	< 0.010	0.034	< 0.0002	< 0.020	0.003	< 0.05	< 0.005	< 0.005	0.236	NA
Dissolved			1.0				< 0.010	< 0.005	0.100	< 0.005	< 0.005	< 0.010	0.034	< 0.0002	< 0.020	0.002	< 0.05	< 0.005	< 0.005	0.236	NA
07-23-92 (1)	MW-101	ND	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				<0.010	<0.005	0.095	<0.005	<0.0005	< 0.010	0.092	<0.0002	<0.020	0.007	<0.05	<0.005	<0.005	0.187	NA
Dissolved							<0.010	<0.005	0.100	<0.005	<0.0005	<0.010	<0.010	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0.122	NA
11-18-92(1)	MW-101	ND	ND	NA	ND	ND															
Total Metals		[UQ]	[UQ]				< 0.010	< 0.005	0.089	< 0.005	< 0.0005	< 0.010	< 0.010	< 0.0002	< 0.020	0.005	< 0.05	< 0.005	< 0.005	0.500	0.250
Dissolved							< 0.010	< 0.005	0.094	< 0.005	< 0.0005	< 0.010	< 0.010	< 0.0002	< 0.020	0.004	< 0.05	< 0.005	< 0.005	0.489	0.240
3 18 93 (1)	MW 101	ND	ND	NA	NA	NA															
Total Metals	101 00 - 101	IUOI	TUO	INA	INA	na.	<0.010	<0.005	0 117	<0.005	<0.005	<0.010	<0.010	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0 219	NA
Dissolved		1.0 (1	1.0 (1)				< 0.010	< 0.005	0.096	< 0.005	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.197	NA
06-17-93 (1)	MW-101	ND	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				<0.010	0.011	0.114	<0.005	<0.005	< 0.010	0.012	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0.184	NA
Dissolved							<0.010	0.011	0.117	~0.005	~0.005	~0.010	~0.010	~0.0002	~0.020	~0.002	~0.05	~0.005	~0.005	0.174	nn.
11-08-93 (1)	MW-101	ND	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				< 0.010	< 0.005	0.103	< 0.005	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.286	NA
Dissolved							< 0.010	< 0.005	0.114	< 0.005	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.268	NA
5-19-94(1)	MW-101	ACE 23	BEP 5 I	NΔ	NΔ	NΔ															
Total Metals	11111-101	CDS 25	* [UQ]	1474	1471	101	< 0.010	< 0.005	0.092	< 0.004	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.231	NA
Dissolved		[UQ]					< 0.010	< 0.005	0.095	< 0.004	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.227	NA
7-14-94 (1)	MW-101	ND	ND	NA	NA	NA	<0.010	<0.005	0.007	-0.004	<0.005	<0.010	<0.010	<0.0002	<0.20	<0.000	<0.05	<0.005	<0.005	0.202	NIA
Dissolved		[UQ]	[UQ]				<0.010	<0.005	0.097	<0.004	<0.005	< 0.010	<0.010	<0.0002	<0.20	<0.002	<0.05	<0.005	<0.005	0.302	NA NA
Dissolved							-0.010	-0.005	0.105	-0.004	-0.005	-0.010	-0.010	-0.0002	-0.20	-0.002	-0.05	-0.005	-0.005	0.525	1411
7-14-94 (1)	MW-101	ND	ND	NA	NA	NA															
Total Metals	Duplicate	[UQ]	[UQ]				< 0.010	< 0.005	0.09	< 0.004	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.367	NA
Dissolved							< 0.010	< 0.005	0.103	< 0.004	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.321	NA
03-18-95	MW-101	ND	NA	NA	NA	NA															
Total Metals		[UQ]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							INA	INPA	INA	INA	INA	1974	INA	INA	INA	INA	INA	INA	INA	INPA	INA
10-25-95	MW-101	ND	NA	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA

Sample	Location		Orga	nic Const	ituents					Inorganic	Constituen	ts (mg/L)				Ino	rganic Cons	stituents Co	ntinued (mg	g/L)	
Date		VOCs	BNAs	ТРН	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)															
6-4-96	MW-101	ND	ND	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Notes:																					
NA	Not analyzed.			ND	Not deter	eted.															
VOCs	Volatile organic compounds.BN.Total Petroleum HydrocarbonsTOI				Base/neu	tral and ac	id extractab	le compou	nds.												
TPH	Volatile organic compounds. Biv Total Petroleum Hydrocarbons TO.				Toluene.																
Tr	Trace, unquanti	TCE	Trichloro	ethene.																	
ACE	Acetone.			EDB	Ethylene	Dibromide	÷.														
BEP	Bis(2-ethylhexy	l)phthalate		DBCP	1,2-Dibro	omo-3-chlo	ropropane.														
CHL	Chloroform.			PCB	Pentachle	orobenzene	;														
CDS	Carbon Disulfid	le		BA	Benzoic	Acid															
DCA	1,2-Dichloroeth	ane.		DCP	1,2-Dich	loropropan	e														
DBCM	Dibromochloror	nethane.		Bo	Boron.																
*	Dibromochloromethane. Bo H TICs present. [UQ] A					enix data o	f unknown	quality as													
J	Data are qualitative or estimated.						FA Parties														
R	Data are rejecte	Data are rejected and unusable																			
U	Data are qualita	tive and co	nsidered t	o be not d	etected at	the value i	reported.														
(1)	Well screen submerged below top of water table.																				

Method detection limits (mdls) for VOCs and BNAs varied. See laboratory reports for specific mdls. Metals are as follows: silver (Ag), arsenic (As), beryllium (Be), barium (Ba), cadmium (Cd), copper (Cu), chromium (Cr), mercury (Hg), nickel (Ni), lead (Pb), antimony (Sb), selenium (Se), thallium (Tl), and zinc (Zn).

Sample	Location		Organ	ic Const	ituents					Inorganic	Constituen	ts (mg/L)				Inor	rganic Cons	tituents Co	ntinued (mg	g/L)	
Date		VOCs	BNAs	ТРН	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)															
05-26-93 Total Metals Dissolved	MW-117	ND	ND	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.146 0.143	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	0.012 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.003 0.003	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.378 0.364	NA NA
05-26-93 Total Metals Dissolved	MW-117 Duplicate	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.140 0.150	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.002 0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.348 0.344	NA NA
11-12-93 Total Metals Dissolved	MW-117	ND [UQ]	ND * [UQ]	NA	NA	NA	<0.010 <0.010	0.006 0.005	0.137 0.145	<0.005 <0.005	<0.005 <0.005	0.017 <0.010	<0.010 <0.010	<0.0002 <0.0002	0.042 0.049	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.309 0.298	NA NA
5-17-93 Total Metals Dissolved	MW-117	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.149 0.158	<0.004 <0.004	<0.005 <0.005	0.011 <0.010	0.017 <0.010	<0.0002 <0.0002	0.030 0.024	0.003 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.317 0.292	NA NA
02-13-95 Total Metals Dissolved	MW-117	ND [UQ]	NA [UQ]	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
10-4-95 Total Metals Dissolved	MW-117	ND	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
5-29-96 Total Metals Dissolved	MW-117	ND	ND	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA

Notes:

Not analyzed. ND Not detected.

NA [UQ] ATI-Phoenix data of unkno

determined by the FFA Parties

* TICs present.

Method detection limits (mdls) for VOCs and BNAs varied. See laboratory reports for specific mdls. Metals are as follows: silver (Ag), arsenic (As), beryllium (Be), barium (Ba), cadmium (Cd), copper (Cu), chromium (Cr), mercury (Hg), nickel (Ni), lead (Pb), antimony (Sb), selenium (Se), thallium (Tl), and zinc (Zn).

Г

_

Sample	Location		Org	anic Constit	uents					Inorganic	Constituer	nts (mg/L)				Inc	organic Cor	stituents Co	ontinued (m	g/L)	
Date		VOCs	BNAs	TPH	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)															
08-04-93	MW-119	DCP 1	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				< 0.010	< 0.005	0.097	< 0.005	< 0.005	0.034	0.019	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	1.88	NA
Dissolved							< 0.010	< 0.005	0.106	< 0.005	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	1.97	NA
11-05-93	MW-119	DCP 2	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				< 0.010	0.007	0.095	< 0.005	< 0.005	3.84J	0.036	< 0.0002	0.103 J	< 0.002	<0.05	0.007	< 0.005	2.77	NA
Dissolved							< 0.010	<0.005	0.095	<0.005	<0.005	<0.010	<0.010	<0.0002	0.039	<0.002	<0.05	0.007	< 0.005	3.00	NA
11-05-93	MW-119	DCP 2	ND *	NΔ	NA	NA															
Total Metals	Duplicate	IU01	TUOI				< 0.010	< 0.005	0.090	< 0.005	< 0.005	1.64 J	0.022	< 0.0002	0.053 J	< 0.002	< 0.05	0.008	< 0.005	3.09	NA
Dissolved	1	1.4	1. 0				< 0.010	< 0.005	0.089	< 0.005	< 0.005	< 0.010	< 0.010	< 0.0002	0.025	< 0.002	< 0.05	0.008	< 0.005	3.16	NA
05-17-94	MW-119	DCP 2	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				< 0.010	< 0.005	0.077	< 0.004	< 0.005	0.073	< 0.010	< 0.0002	0.254	< 0.002	< 0.05	< 0.005	< 0.005	0.640	NA
Dissolved							< 0.010	< 0.005	0.076	< 0.004	< 0.005	< 0.010	< 0.010	< 0.0002	0.250	< 0.002	< 0.05	< 0.005	< 0.005	0.656	NA
02.02.05	MW 110	DCD 2	ND	NA	NIA	NA															
Total Metals	WIW-119	IUO1	IUOI	INA	INA	INA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved		[00]	[00]				NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
5-2-95	MW-119	ND	NA	ND	NA	NA															
Total Metals		[UQ]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
0 14 05	NOV 110	ND		NID																	
Total Metals	WIW-119	ND	INA	ND	INA	INA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
10-28-95	MW-119	ND	NA	ND	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-10-96 Total Matala	MW-119	ND	NA	DRO	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved				DRO			NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissorred																					
5-23-96	MW-119	ND	ND	ND	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
5-23-96	MW-119	ND	ND	26J	NA	NA															
Total Metals	DUPLICATE			GRO			NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							INA	INA	INA	INA	INA	INA	IN/A	INA	INA	INA	INA	IN/A	INA	INA	INA
07-17-97	MW-119	DCP 1.0J	NA	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA

Sample	Location		Org	anic Constit	uents					Inorganic	c Constituer	nts (mg/L)				Ino	ganic Con	stituents Co	ntinued (mg	y/L)	
Date		VOCs	BNAs	TPH	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)															
01-05-94	MW-120	ND	ND	NA	NA	NA															
Fotal Metals		[UQ]	[UQ]				< 0.010	< 0.005	0.088	< 0.005	< 0.005	0.092	< 0.010	< 0.0002	0.086	< 0.002	< 0.05	0.006	< 0.005	2.34	NA
Dissolved		1.0	1.0				< 0.010	< 0.005	0.088	< 0.005	< 0.005	< 0.010	< 0.010	< 0.0002	0.081	< 0.002	< 0.05	0.005	< 0.005	2.32	NA
01-05-94	MW-120	ND	ND *	NA	NA	NA															
Fotal Metals	Duplicate	[UQ]	[UQ]				< 0.010	< 0.005	0.090	< 0.005	< 0.005	0.099	< 0.010	< 0.0002	0.093	< 0.002	< 0.05	0.005	< 0.005	2.36	NA
Dissolved							< 0.010	< 0.005	0.088	< 0.005	< 0.005	< 0.010	< 0.010	< 0.0002	0.086	< 0.002	< 0.05	< 0.005	< 0.005	2.34	NA
05-16-94	MW-120	DCP 1	ND	NA	NA	NA						0.000			0.071						
l otal Metals		[UQ]	[UQ]				< 0.010	<0.005	0.116	<0.004	<0.005	0.089	<0.010	<0.0002	0.071	<0.002	<0.05	< 0.005	< 0.005	2.15	NA
Dissolved							<0.010	<0.005	0.120	<0.004	<0.005	<0.010	<0.010	<0.0002	0.071	<0.002	<0.05	<0.005	<0.005	2.34	NA
02-01-95	MW-120	ND	NA	ND	NA	NA															
Fotal Metals		IUOI		110			NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved		1.40					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
5-2-95	MW-120	ND	NA	ND	NA	NA															
Fotal Metals		[UQ]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
8-9-95	MW-120	ND	NA	ND	NA	NA															
Fotal Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
10 27 05	MW 120	ND	NA	ND	NA	NA															
Total Metals	WIW-120	ND	1874	ND	1874	1974	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-9-96	MW-120	ND	NA	ND	NA	NA															
Fotal Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
6-11-96	MW-120	ND	ND	ND	NA	NA															
Fotal Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
07 17 07	MW 120	DCR 1 71	NA	ND	NA	NA															
Fotal Metals	WIW-120	DCI 1.75	1874	ND	1874	1974	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
										- // •										•	
		DCP 1,4J																			
		MC 0.25JB																			
11-05-98	MW-120	10L 0.11J	NA	270JB	NA	NA										27.4					
otal Metals				GRO			NA	NA	NA	NA	NA	NA	NA	NA NA	NA	NA	NA	NA	NA	NA NA	NA

Sample	Location		Org	anic Constitu	uents					Inorganic	Constituer	nts (mg/L)				Inc	rganic Con	stituents Co	ntinued (mg	g/L)	
Date		VOCs	BNAs	TPH	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)															
01-05-94	MW-121	ND	ND *	NA	NA	NA															
Total Metals		[UQ]	[UQ]				< 0.010	< 0.005	0.139	< 0.005	< 0.005	0.074	< 0.010	< 0.0002	0.144	< 0.002	< 0.05	0.005	< 0.005	1.85	NA
Dissolved							< 0.010	< 0.005	0.136	< 0.005	< 0.005	< 0.010	< 0.010	< 0.0002	0.143	< 0.002	< 0.05	< 0.005	< 0.005	1.76	NA
04-06-94	MW-121	ND	NA	ND	NA	NA															
Total Metals	Time-series	[UQ]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved	Initial Sample						NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
04-07-94	MW-121	ND	NA	ND	NA	NA															
Total Metals	Time-series	[UQ]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved	24-hr Sample						NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
04-07-94	MW-121	ND	NA	ND	NA	NA															
Total Metals	Time-series	IUOI					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved	24-hr Duplicate	1.40					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
05-13-94	MW-121	DCP 2	NA	NA	NA	NA															
Total Metals	ADEQ	[UQ]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved	Duplicate						NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
05-16-94	MW-121	DCP 2	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				< 0.010	< 0.005	0.081	< 0.004	< 0.005	0.016	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	<0.005	< 0.005	1.61	NA
Dissolved							<0.010	< 0.005	0.084	<0.004	<0.005	<0.010	< 0.010	< 0.0002	< 0.020	< 0.002	<0.05	<0.005	<0.005	1.48	NA
05 16 04	MW 121	DCP 2	ND	NA	NA	NA															
Total Metals	Duplicate	11101	IUOI	1974	1974	1974	<0.010	<0.005	0.082	<0.004	<0.005	0.026	<0.010	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	1.69	NA
Dissolved	Dupileute	[04]	[0.4]				< 0.010	<0.005	0.080	< 0.004	<0.005	0.011	< 0.010	< 0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	2.12	NA
02-02-95	MW-121	DCP 2	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
02-02-95	MW-121	DCP 2	ND	NA	NA	NA															
Total Metals	Duplicate	[UQ]	[UQ]				NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
5 2 05	MW 121	ND	NA	ND	NA	NA															
J=2-95 Total Metale	ivi vv =1.2.1	TUOI	INA	ND	INA	18/4	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved		[00]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
2.3501700																					
8-15-95	MW-121	ND	NA	ND	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA

Sample	Location		Or	ganic Constitu	ients					Inorganio	Constituer	nts (mg/L)				In	organic Cor	stituents Co	ntinued (mg	g/L)	
Date		VOCs	BNAs	TPH	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)															
10-31-95	MW-121	ND	NA	ND	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-10-96	MW-121	ND	NA	ND	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
6 22 06	101/101	ND	ND	ND																	
5-23-96 Tatal Matala	MW-121	ND	ND	ND	NA	NA	NA	NIA	NA	NIA	NIA	NIA	NIA	NIA	NIA	NA	NIA	NA	NIA	NIA	NA
Discolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							INA	INA	INA	INA	INA	INA	INA	INA	INA	INPA	INA	INA	INA	INA	INA
		DCP 1.4J																			
		BZ 1.8J																			
		EB 4.4J																			
07 17 07	MW 121	TOL 6.3 XVI 12	NIA	0701	NIA	NIA															
Total Metale	IVI W = 1 2 1	ATE 12	INA	9700 DRO	INA	INA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved				DRO			NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							1974	1974	19/4	INA	1974	1874	19/4	1974	1974	1974	1974	1974	1874	1974	19/4
11-13-97	MW-121	DCP 1.7J	NA	130vr	NA	NA															
Total Metals				DRO			NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
		DCP 1.8J																			
		BZ 1.8																			
		TOL 2.5J																			
05-14-98	MW-121	XYL 6.9	NA	120. 44B	NA	NA															
Total Metals				DRO, GRO			NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
		BZ 17																			
		EB 23																			
		MC 34JB PCE 0.021																			
		TOL 36																			
11-05-98	MW-121	XYL 61	NA	630, 200JB	NA	NA															
Total Metals				DRO,GRO			NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
		BZ 19																			
		EB 25 MC_04IR																			
		PCE 0.251																			
		TOL 38																			
11-05-98	MW-121	XYL 64	NA	250, 490JB	NA	NA															
Total Metals	Duplicate			DRO, GRO			NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA

Г

Sample	Location		Org	anic Consti	tuents					Inorgani	c Constitue	nts (mg/L)				In	organic Cor	nstituents Co	ontinued (m	g/L)	
Date		VOCs	BNAs	TPH	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
1	1	(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)											1	1			
05-20-00	MW-121	ND	NA	ND	NA	NA															
Total Metale	1V1 VV = 1 2 1	ND	13/3	IND.	13/3	13/4	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							1111	1111	11/4	1174	11/1	11/1	1111	1114	1174	1174	1474	nn.	1474	11/1	1111
05-16-00	MW-121	ND	NA	ND	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
05-16-00	MW-121	ND	NA	ND	NA	NA															
Total Metals	Duplicate						NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
8-2-01	MW-121	ND	NA	ND	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
8-2-01	MW-121	ND	NA	ND	NA	NA															
Total Metals	Duplicate						NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
5-30-95	MW-125	ND	NA	ND	NA	NA												214			
I otal Metals		[UQ]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
8 15 05	MW 125	ND	NA	ND	NA	NA															
Total Metale	10100-125	ND ND	1174	ND	1474	1474	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved																					
8-15-95	MW-125	ND	NA	ND	NA	NA															
Total Metals	DUPLICATE						NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
10-31-95	MW-125	ND	NA	ND	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-9-96	MW-125	ND	NA	11 J	NA	NA															
Total Metals				GRO			NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
6-11-96	MW-125	ND	NA	ND	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
07-17-97	MW-125	DCP 1.0J	NA	23J	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Total Metals				GRO			NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA

Г

Sample	Location		Org	anic Constitu	ients					Inorganic	Constituen	ts (mg/L)				Ino	rganic Cons	stituents Co	ntinued (mg	/L)	
Date		VOCs	BNAs	TPH	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	Tl	Zn	Bo
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)															
		DCP 1.1J																			
07-17-97	MW-125	BZ 3.4J	NA	53J	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Total Metals	Duplicate			DRO			NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
11-13-97	MW-125	DCP 1.2J	NA	ND	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
11-13-97	MW-125	DCP 1.3J	NA	ND	NA	NA															
Total Metals	Duplicate						NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
05-14-98	MW-125	DCP 1.6J	NA	ND	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
		DCB 1 21																			
11-04-98	MW-125	MC 0.21JB	NA	7.2	NA	NA															
Total Metals				GRO			NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
05-20-99	MW-125	ND	NA	ND	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
05-16-00	MW-125	ND	NA	ND	NA	NA															
Discolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							18/4	INA	INA	INA	18/4	18/4	18/4	INA	INA	INA	INA	18/4	INA	INA	INA
05-16-00	MW-125	ND	NA	ND	NA	NA															
Total Metals	Duplicate						NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved	-						NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA

Sample	Location		Or	ganic Constitu	uents					Inorganic	Constituer	its (mg/L)				Inc	rganic Cons	stituents Co	ntinued (mg	g/L)	
Date		VOCs	BNAs	TPH	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)															
11-09-01	MW-125R	ND	NA	ND	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Notes:																					
NA	Not analyzed.			ND	Not detec	ted.															
VOCs	Volatile organic	compounds.		BNAs	Base/neu	tral and aci	d extractab	le compoun	ds.												
TPH	Total Petroleum	Hydrocarbons		TOL	Toluene.																
Tr	Trace, unquantifi	iable amount d	etected.	TCE	Trichloro	oethene.															
ACE	Acetone.			EDB	Ethylene	Dibromide															
BEP	Bis(2-ethylhexyl)phthalate.		DBCP	1,2-Dibro	mo-3-chlo	ropropane.														
CHL	Chloroform.			PCB	Pentachlo	orobenzene															
CDS	Carbon Disulfide			BA	Benzoic /	Acid															
DCA	1,2-Dichloroetha	ne.		DCP	1,2-Dichl	oropropan															
DBCM	Dibromochlorom	nethane.		Bo	Boron.																
BZ	Benzene			EB	Ethyl Ber	nzene															
PCE	Tetrachloroethen	ie		XYL	Total Xyl	enes															
MC	Methylene Chlor	ide		[UQ]	ATI-Phoe	enix data o	unknown o	quality as													
*	TICs present.			DRO	Diesel Ra	inge															
GRO	Gasoline Range				determine	ed by the F	FA Parties														
В	Compound also	detected in the	blank.																		
J	Data are qualitati	ive or estimated	1 .																		
R	Data are rejected	and unusable																			
U	Data are qualitati	ive and conside	ered to be 1	not detected at	the value re	eported.															
v	Reliable identific	ation of a proc	luct could	not be achieve	d																

Well screen submerged below top of water table.

Method detection limits (mdls) for VOCs and BNAs varied. See laboratory reports for specific mdls. Metals are as follows: silver (Ag), arsenic (As), beryllium (Be), barium (Ba), cadmium (Cd), copper (Cu), chronium (Cr), mercury (Hg), nickel (Ni), lead (Pb), antimony (Sb), selenium (Se), thallium (TI), and zinc (Zn).

Sample	Location		Orgai	nic Cons	tituents					Inorgani	c Constituer	ts (mg/L)				Ino	rganic Cons	stituents Co	ntinued (mg	g/L)	
Date		VOCs (ug/L)	BNAs (ug/L)	TPH (mg/L)	EDB (mg/L)	DBCP (mg/L)	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	ті	Zn	Bo
07-22-92 Total Metals Dissolved	MW-2	ND	ND J	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.139 0.140	<0.005 <0.005	<0.0005 <0.0005	<0.010 <0.010	0.064 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.011 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.338 0.253	NA NA
11-25-92 Total Metals Dissolved	MW-2	ND [UQ]	ND [UQ]	NA	<0.01	<0.01	<0.010 <0.010	<0.005 <0.005	0.127 0.137	<0.005 <0.005	<0.0005 <0.0005	<0.010 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.006 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.629 0.628	<0.10 <0.10
03-24-93 Total Metals Dissolved	MW-2	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.126 0.15	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	0.011 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.004 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.393 0.392	NA NA
06-18-93 Total Metals Dissolved	MW-2	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.130 0.138	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	0.013 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.462 0.453	NA NA
07-16-92 Total Metals Dissolved	MW-3	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.028 0.038	<0.005 <0.005	<0.0005 <0.0005	<0.010 <0.010	0.015 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.454 <0.010	NA NA
11-24-92 Total Metals Dissolved	MW-3	ND [UQ]	ND [UQ]	NA	<0.01	0.07	<0.010 <0.010	<0.005 <0.005	0.098 0.106	<0.005 <0.005	<0.0005 <0.0005	<0.010 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.005 <0.002	<0.05 <0.05	0.006 0.006	<0.005 <0.005	0.764 0.770	<0.10 <0.10
03-12-93 Total Metals Dissolved	MW-3	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.096 0.096	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	0.006 0.006	<0.005 <0.005	0.469 0.465	NA NA
06-15-93 Total Metals Dissolved	MW-3	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.099 0.110	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	0.012 <0.010	<0.0002 <0.0002	0.028 <0.020	0.003 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.638 0.629	NA NA
11-04-93 Total Metals Dissolved	MW-3	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.093 0.105	<0.005 <0.005	<0.005 <0.005	<0.010 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.365 0.380	NA NA
05-13-94 Total Metals Dissolved	MW-3	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.101 0.098	<0.004 <0.004	<0.005 <0.005	<0.010 <0.010	0.034 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.003 <0.002	<0.05 <0.05	0.006 0.005	<0.005 <0.005	0.488 0.480	NA NA
2-13-95 Total Metals Dissolved	MW-3	ND [UQ]	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
10-11-95 Total Metals Dissolved	MW-3	ND	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
6-4-96 Total Metals Dissolved	MW-3	ND	ND	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
07-23-92 Total Metals Dissolved	MW-4	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	0.005 <0.005	0.121 0.127	<0.005 <0.005	<0.0005 <0.0005	<0.010 <0.010	0.029 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.004 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.204 0.188	NA NA
12-05-92 Total Metals Dissolved	MW-4 Time-series Initial Sample	ND [UQ]	ND [UQ]	NA	<0.01	<0.01	<0.010 <0.010	<0.005 <0.005	0.127 0.134	<0.005 <0.005	<0.0005 <0.0005	<0.010 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.611 0.352	<0.10 <0.10
12-05-92 Total Metals Dissolved	MW-4 Time-series 4-hr Sample	ND [UQ]	NA	NA	<0.01	<0.01	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
03-24-93 Total Metals Dissolved	MW-4	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.005 <0.005	0.134 0.154	<0.005 <0.005	<0.005 <0.005	0.010 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	0.003 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.360 0.331	NA NA
06-21-93	MW-4	ND	ND	NA	NA	NA															

Samula	Location	1	0	in Come	414					Inormal	Constituou	to (ma/L)				Inc		atitu anta Ca		./I.)	
Sample	Location		Orgai	tie Cons	tituents					inorgani	c Constituer	its (mg/L)	-			Inc	organic Con	stituents Co	ntinued (mg	yL)	-
Date		VOCs	BNAs	TPH	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	РЬ	Sb	Se	TI	Zn	Bo
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)															
Total Matala		TUO1	TUO1				<0.010	<0.005	0.114	<0.005	<0.005	0.011	0.021	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0.250	NA
Discolved		[UQ]	[UQ]				<0.010	<0.005	0.114	<0.005	<0.005	<0.011	<0.021	<0.0002	<0.020	<0.002	< 0.05	<0.005	< 0.005	0.350	NA
Dissolved							~0.010	~0.005	0.124	<0.005	~0.005	<0.010	<0.010	~0.0002	~0.020	~0.002	<0.05	~0.005	~0.005	0.559	1974
11-15-93	MW-4	ND	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				< 0.010	< 0.005	0.114	< 0.005	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.355	NA
Dissolved							< 0.010	< 0.005	0.139	< 0.005	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.401	NA
05-13-94	MW-4	ND	ND	NA	NA	NA	.0.010	.0.005	0.120	.0.004	.0.005	.0.010	0.011		.0.020	.0.000	-0.05	-0.005	.0.005	0.250	
Discolved		[UQ]	[UQ]				<0.010	<0.005	0.130	<0.004	<0.005	<0.010	<0.011	<0.0002	<0.020	<0.002	< 0.05	<0.005	< 0.005	0.379	NA
Dissolved							~0.010	~0.005	0.138	~0.004	~0.005	~0.010	~0.010	~0.0002	~0.020	~0.002	~0.05	~0.005	~0.005	0.375	INA
07-18-92	MW-5	ND	ND J	NA	NA	NA															
Total Metals		IUOI	[UO]				< 0.010	0.019	0.204	< 0.005	< 0.0005	0.115	0.052	< 0.0002	0.051	0.026	< 0.05	< 0.005	< 0.005	8.70	NA
Dissolved		1.0	1. 0				< 0.010	< 0.005	0.082	< 0.005	< 0.0005	< 0.010	< 0.010	< 0.0002	< 0.020	0.004	< 0.05	< 0.005	< 0.005	0.116	NA
11-20-92	MW-5	ND	ND	NA	< 0.01	< 0.01															
Total Metals		[UQ]	[UQ]				< 0.010	< 0.005	0.199	< 0.005	< 0.0005	< 0.010	< 0.010	< 0.0002	< 0.020	0.008	< 0.05	< 0.005	< 0.005	0.514	< 0.10
Dissolved							< 0.010	< 0.005	0.221	< 0.005	< 0.0005	< 0.010	< 0.010	< 0.0002	< 0.020	0.007	< 0.05	< 0.005	< 0.005	0.572	< 0.10
03-23-93	MW-5	ND	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				< 0.010	< 0.005	0.199	< 0.005	< 0.005	< 0.010	0.012	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.439	NA
Dissolved							< 0.010	<0.005	0.222	<0.005	<0.005	<0.010	< 0.010	<0.0002	< 0.020	< 0.002	<0.05	<0.005	<0.005	0.399	NA
06 21 02	MW 5	ND	NID	NIA	NIA	NIA															
Total Metals	WI W-5	IUOI	TUOL	INA	INA	INA	<0.010	<0.005	0.182	<0.005	<0.005	<0.010	0.015	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0.454	NΔ
Dissolved		[00]	[UQ]				<0.010	<0.005	0.182	<0.005	<0.005	<0.010	<0.015	<0.0002	<0.020	0.016	<0.05	<0.005	<0.005	0.4.04	NA
Dissolved							-0.010	-0.005	0.174	-0.005	-0.005	-0.010	~0.010	~0.0002	~0.020	0.010	~0.05	-0.005	~0.005	0.500	1474
11-11-93	MW-5	ND	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				< 0.010	< 0.005	0.196	< 0.005	< 0.005	0.011	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	0.007	< 0.005	0.395	NA
Dissolved		1.0	1. 0				< 0.010	< 0.005	0.217	< 0.005	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	0.006	< 0.005	0.400	NA
05-18-94	MW-5	ND	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				< 0.010	< 0.005	0.174	< 0.004	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.288	NA
Dissolved							< 0.010	< 0.005	0.174	< 0.004	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.286	NA
2-13-95	MW-5	ND	NA	NA	NA	NA															
I otal Metals		[UQ]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
10-19-95	MW-5	ND	NA	NΔ	NΔ	NA															
Total Metals	14144-5	ND	11/1	11/1	1174	1474	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
6-5-96	MW-5	ND	ND	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
07-22-92 Total Mate1-	MW-114	ND	ND	NA	NA	NA	<0.010	0.007	0.049	<0.005	<0.0005	0.022	0.122	<0.0002	<0.020	0.011	<0.05	<0.005	<0.005	0.407	NA
Discolved		[UQ]	[UQ]				<0.010	0.007	0.048	<0.005	< 0.0005	0.022	0.133	<0.0002	<0.020	0.011	< 0.05	< 0.005	< 0.005	0.407	NA
Dissolved							~0.010	0.000	0.040	<0.005	~0.0005	0.015	0.020	~0.0002	~0.020	0.005	~0.05	~0.005	~0.005	0.370	1974
11-30-92	MW-114	ND	ND	NA	< 0.01	< 0.01															
Total Metals		[UQ]	[UQ]				< 0.010	0.006	0.063	< 0.005	< 0.0005	0.017	< 0.010	< 0.0002	< 0.020	0.004	< 0.05	< 0.005	< 0.005	0.344	< 0.10
Dissolved		1. 1	r - «J				< 0.010	0.005	0.060	< 0.005	< 0.0005	0.018	< 0.010	< 0.0002	< 0.020	0.002	< 0.05	< 0.005	< 0.005	0.310	< 0.10
11-30-92	MW-114	ND	ND	NA	< 0.01	< 0.01															
Total Metals	Duplicate	[UQ]	[UQ]				< 0.010	0.006	0.062	< 0.005	< 0.0005	0.023	< 0.010	< 0.0002	< 0.020	0.003	< 0.05	< 0.005	< 0.005	0.337	< 0.10
Dissolved							< 0.010	0.006	0.062	< 0.005	< 0.0005	0.015	< 0.010	< 0.0002	< 0.020	0.002	< 0.05	< 0.005	< 0.005	0.348	<0.10
03-25-93	MW-114	ND	ND	NA	NA	NA	.0.01-	-0 00 -	0.045	-0 00 -	-0 00 -	.0.017	.0.017	-0.000-	.0.027	.0.007	-0.05	-0 00 -	-0.005	0.464	
Total Metals		[UQ]	[UQ]				< 0.010	< 0.005	0.262	< 0.005	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.464	NA
Dissolved							< 0.010	< 0.005	0.285	< 0.005	< 0.005	< 0.010	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	<0.005	0.453	NA

Sampla	Logation		Organ	ia Cone	ituante					Inorganic	Constituer	ts (mg/L)				Ino	rannia Con	ctituante Co	ntinued (m	·/I)	
Sample	Location	VOC	DNA	TDH	EDD	DBCB	4.		n.	norganic	Cul	its (ing/L)	6	п.	N.''	ni	rganic Con		nunucu (mį	() ()	р.
Date		vocs	BNAS	IPH	EDB	DBCP	Ag	AS	ва	ве	Ca	Cr	Cu	нg	INI	PD	SD	Se		Zn	во
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)															
06-18-93	MW-114	ND	BEP 40	NA	NA	NA	-0.010	0.005	0.071	-0.005	-0.005	0.020	0.012	-0.0003	-0.020	-0.003	-0.05	-0.005	-0.005	0.240	
Dissolved		[UQ]	[UQ]				<0.010	<0.005	0.071	< 0.005	<0.005 <0.005	0.020	<0.012	<0.0002	<0.020	<0.002	<0.05	< 0.005	<0.005	0.349	NA
11-11-93	MW-114	ND	ND	NA	NA	NA															
Total Metals		[UQ]	[UQ]				< 0.010	< 0.005	0.183	< 0.005	< 0.005	0.011	< 0.010	< 0.0002	< 0.020	< 0.002	< 0.05	< 0.005	< 0.005	0.247	NA
Dissolved							<0.010	<0.005	0.273	<0.005	<0.005	0.010	<0.010	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0.181	NA
05-13-94	MW-114	ND	ND	NA	NA	NA															
Total Metals Dissolved		[UQ]	[UQ]				<0.010	<0.005 <0.005	0.295	<0.004	<0.005	<0.010	<0.010	<0.0002	<0.020	<0.002	<0.05	<0.005	<0.005	0.373	NA
Dissorred							-0.010	-0.005	0.511	-0.001	-0.005	-0.010	-0.010	-0.0002	-0.020	-0.002	-0.02	-0.005	-0.005	0.100	
05-13-94 Total Matala	MW-114	ND	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved	Duplicate	[00]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
2 14 05	MW 114	ND	NIA	NIA	NIA	NIA															
Total Metals	WI W-114	[UQ]	1974	1974	1N/A	N/A	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
5-5-95	MW-114	ND	NA	NA	NA	NA															
Total Metals		[UQ]					NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
8-9-95	MW-114	ND	NA	NA	NA	NA															
Total Metals Dissolved							NA	NA NA	NA NA	NA NA	NA NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA	NA NA	NA NA
10-19-95 Total Metals	MW-114	ND	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
2-2-96	MW-114	ND	NA	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
6-5-96	MW-114	ND	ND	NA	NA	NA															
Total Metals Dissolved							NA	NA NA	NA NA	NA NA	NA NA	NA	NA NA	NA NA	NA	NA NA	NA NA	NA NA	NA	NA NA	NA NA
11-11-97 Total Metals	MW-114	ND	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
5-13-98	MW-114	ND	NA	NA	NA	NA															
Total Metals							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
		MC 0.35JB																			
11-03-98	MW-114	MEK 1.1J	NA	NA	NA	NA															
Total Metals Dissolved							NA	NA	NA NA	NA	NA NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA	NA	NA NA	NA
5-18-99 Total Metals	MW-114	ND	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
		cis-1 2-Dichloroethene																			
		0.89																			
10.25.01	MW 114	TCE 2.0 PCE 0.95	NIA	NIA	NIA	NIA															
Total Metals	WI W -1 14	1020.00	1974	IN/A	1N/A	NA.	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
Dissolved							NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA
		cis-1,2-Dichloroethene																			
		0.94 TCF 2.0																			
10-25-01	MW-114	PCE 0.90	NA	NA	NA	NA															

Sample	Location		Organ	ic Consti	ituents					Inorganic	Constituen	ts (mg/L)				Ino	rganic Con	stituents Co	ntinued (mg	g/L)	
Date		VOCs	BNAs	ТРН	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)															
Total Metals Dissolved	Duplicate						NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
07-14-94 Total Metals Dissolved	MW-122	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.006 <0.005	0.065 0.062	<0.004 <0.004	<0.005 <0.005	0.013 0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.313 0.268	NA NA
07-14-94 Total Metals Dissolved	MW-122 Duplicate	ND [UQ]	ND [UQ]	NA	NA	NA	<0.010 <0.010	<0.006 <0.005	0.064 0.062	<0.004 <0.004	<0.005 <0.005	0.012 <0.010	<0.010 <0.010	<0.0002 <0.0002	<0.020 <0.020	<0.002 <0.002	<0.05 <0.05	<0.005 <0.005	<0.005 <0.005	0.342 0.264	NA NA
2-14-95 Total Metals Dissolved	MW-122	ND [UQ]	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
5-5-95 Total Metals Dissolved	MW-122	ND [UQ]	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
8-9-95 Total Metals Dissolved	MW-122	ND	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
10-19-95 Total Metals Dissolved	MW-122	ND	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
2-2-96 Total Metals Dissolved	MW-122	ND	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
6-6-96 Total Metals Dissolved	MW-122	ND	ND	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
11-12-97 Total Metals Dissolved	MW-122	ND	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
5-13-98 Total Metals Dissolved	MW-122	ND	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
11-03-98 Total Metals Dissolved	MW-122	MC 0.29JB	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA

			Organic Constituents																		
Sample	Location		Organ	ic Const	ituents					Inorganic	Constituen	ts (mg/L)				Ino	ganic Cons	tituents Co	ntinued (mg	/L)	
Date		VOCs	BNAs	TPH	EDB	DBCP	Ag	As	Ba	Be	Cd	Cr	Cu	Hg	Ni	Pb	Sb	Se	TI	Zn	Bo
		(ug/L)	(ug/L)	(mg/L)	(mg/L)	(mg/L)															
5-18-99 Total Metals Dissolved	MW-122	ND	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA
10-25-01 Total Metals Dissolved	MW-122	ND	NA	NA	NA	NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA	NA NA

Notes:			
NA	Not analyzed.	ND	Not detected.
VOCs	Volatile organic compounds.	BNAs	Base/neutral and acid extractable compounds.
TPH	Total Petroleum Hydrocarbons	TOL	Toluene.
Tr	Trace, unquantifiable amount detected.	TCE	Trichloroethene.
ACE	Acetone.	EDB	Ethylene Dibromide.
BEP	Bis(2-ethylhexyl)phthalate.	DBCP	1,2-Dibromo-3-chloropropane.
CHL	Chloroform.	PCB	Pentachlorobenzene
CDS	Carbon Disulfide	BA	Benzoic Acid
DCA	1,2-Dichloroethane.	DCP	1,2-Dichloropropane
DBCM	Dibromochloromethane.	Bo	Boron.
TCE	Trichloroethene	[UQ]	ATI-Phoenix data of unknown quality as
PCE	Tetrachloroethene		determined by the FFA Parties
MC	Methylene chloride		
MEK	2-Butanone		
*	TICs present.		
J	Data are qualitative or estimated.		
R	Data are rejected and unusable		

U Data are qualitative and considered to be not detected at the value reported.

Well screen submerged below top of water table.

Method detection limits (mdls) for VOCs and BNAs varied. See laboratory reports for specific mdls. Metals are as follows: silver (Ag), arsenic (As), beryllium (Be), barium (Ba), cadmium (Cd), copper (Cu), chromium (Cr), mercury (Hg), nickel (Ni), lead (Pb), antimony (Sb), selenium (Se), thallium (Tl), and zine (Zn).

Appendix C

Changed Standards Comparison for No-Action PSCs

			Maximum Concentration	USEPA I Industrial P	Region IX RG (mg/kg)	ADEQ S (mg/k
Operable Unit	PSC	COC	of COC in soil	1996	2000	Residential
		<u>BNAs</u> Bis(2-ethylhexyl)phthalate Butylbenzylphthalate	1.3 0.22	140 930	180 100,000	320 13,000
OU-2	OT-04	TRPH [a]	250	NA	NA	4,100
		<u>Metals</u> Copper Lead	30.5 21	63,000 400	76,000 750	2,800 400
		<u>VOCs</u> Ethylbenzene Xylenes	0.9 86	690 320	230 210	1,500 2,800
OU-2	DP-05	BNAs Bis(2-ethylhexyl)phthalate 2-Methylnaphthalene [d] Naphthalene <u>TRPH [a]</u> <u>Metals</u> Copper	3.7 4.7 4.6 8,300 37.8 115	140 240 240 NA 63,000	180 190 190 NA 76,000	320 NE 2,600 4,100 2,800
		Leau	115	400	750	400
		2-Butanone (MEK) Ethylbenzene 2-Hexanone (MBK) 1,1,2,2-Tetrachlorethane Tetrachloroethene Toluene Trichloroethene Xylenes	0.9 6 0.8 0.4 0.05 3 9 43	NA 690 110 1.1 17 880 7 320	NA 230 110 0.9 19 520 6.1 210	NA 1,500 120 4.4 53 790 27 2,800
OU-2	FT-06	Aylenes BNA Acenaphthene Anthracene Benzo(a)anthracene [e] Benzo(b)fluoranthene [e] Benzo(y,h,i)perylene [c] Benzo(a)pyrene [e] Bis(2-ethylhexyl)phthalate Butylbenzylphthalate Chrysene Dibenzo(a,h)anthracene [e] Dibenzofuran Di-n-butylphthalate Fluoranthene Fluorent Indeno(1,2,3-c,d)pyrene 2-Methylphenol Naphthalene Pentachlorophenol Phenol Pyrene TRPH [a] Metals	1.8 2.6 27 46 73 10 30 3.2 0.68 52 10 0.67 0.46 42 0.83 8.1 3 9.1 9.7 3.1 13 3.1 36 18,000	110 5.7 2.6 2.6 26 100 0.26 140 930 7.2 0.26 14,000 68,000 27,000 90 0.61 240 3,400 240 7.9 100 100,000 100 NA	38,000 100,000 2.9 2.9 29 54,000 0.29 180 100,000 62 0.29 5,100 88,000 30,000 33,000 0.62 190 4,400 190 4,400 190 11 54,000 100,000 54,000 NA	3,900 20,000 6.1 6.1 61 NA 0.61 320 13,000 610 0.61 260 6,500 2,600 2,600 2,600 2,600 6.1 NE 330 2,600 25 NE 39,000 2,000 4,100

No Action Sites Combined Suface and Subsurface Soil Constituents

		Copper	40.3	63,000	76,000	2,800
		Lead	101	400	750	400
		BNAs				
		Benzo(b)fluoranthene [e]	0.26	2.6	2.9	6.1
		Benzo(g,h,i)perylene [c]	0.22	100	54,000	NA
		Bis(2-ethylhexyl)phthalate	0.21	140	180	320
		Chrysene	0.29	7.2	62	610
		Fluoranthene	0.22	27,000	30,000	2,600
011-2	ET-07\//	2-Methylnaphthalene [d]	0.91	240	190	NE
00-2	11-07 00	Naphthalene	0.26	240	190	2,600
		Pyrene	0.28	100	54,000	2,000
		TRPH [a]	3,800	NA	NA	4.100
			0,000			.,
		Metals	07.0	c2 000	70.000	0.000
		Copper	37.3	63,000	76,000	2,800
		Lead	172	400	750	400
		VOCs				
		Acetone	1	8,800	6,200	2,100
011.0		TRPH [a]	970	NA	NA	4,100
00-2	DP-22	Motols				
			25.8	63 000	76.000	2 800
		Lead	23.0	400	70,000	2,800
		Lead	50	400	750	400
		VOCs				
		Acetone	1.8	8,800	6,200	2,100
		Benzene	0.13	1.4	1.5	0.62
		Ethylbenzene	1	690	230	1,500
		loluene	0.2	880	520	790
		Xylenes	2.4	320	210	2,800
OU-2	SD-40	<u>BNAs</u>				
		2-Methylnaphthalene [d]	2	240	190	NE
		Naphthalene	0.98	240	190	2,600
			1 200	NΔ	ΝΔ	4 100
			1,200			4,100
		Metals	10.0		70.000	0.000
		Copper	42.8	63,000	76,000	2,800
		Lead	20	400	750	400
OU-1	SS-11	PCBs	0.22	0.066	1	2.5
		BNAs				
		Acenaphthylene [c]	0.046	100	54,000	NE
		Anthracene	0.083	5.7	100,000	20,000
		Benzo(a)anthracene [e]	0.66	2.6	2.9	6.1
		Benzo(a)pyrene [e]	0.87	0.26	0.29	0.61
		Benzo(b)fluoranthene [e]	0.97	2.6	2.9	6.1
		Benzo(g,h,i)perylene [c]	0.48	100	54,000	NA
		Benzo(k)fluoranthene	0.69	26	29	61
		Carbazole	0.13	95	120	220
		Chrysene	1.1	7.2	62	610
		Dibenz(a,h)anthracene	0.2	0.26	0.29	0.61
		Fluoranthene	1.1	27,000	30,000	2,600
		Indeno(1,2,3-c,d)pyrene	0.47	0.61	0.62	6.1
		Pentachlorophenol	0.045	7.9	11	25
OU-1	OT-12	Phenanthrene [c]	0.43	100	54,000	NE
		Pyrene	1.4	100	54,000	2,000
		TRPH [a]	1,400	NA	NA	4,100
		Metals				
		Arsenic	11.0	24	27	10
		Barium	276.0	۲.4 100 000	، <u>م</u> 100 مم	5 300
		Bervllium	0.7	1 1	2 200	1 4
		Cadmium	1.0	850	810	38
		Chromium [b]	33.0	450	450	2.100
		· · · · · · · · · · · · · · · · · · ·				,

		Copper Lead Nickel	29.7 330.0 17.1	63,000 400 34,000	76,000 750 41,000	2,800 400 1,500	
		Zinc Cvanide	76.3 2.0	100,000 35	100,000 35	23,000 1.300	
		BNAc	2.0	00	00	1,000	
		Chrysene Di-n-octyl phthalate Fluoranthene Pyrene	0.20 0.17 0.23 0.18	7.2 10,000 27,000 100	62 10,000 30,000 54,000	610 1,300 2,600 2,000	
		PCBs	0.30	0.34	1	2.5	
		<u>TRPH</u> [a]	7,000.00	NA	NA	4,100	
OU-1	SS-17	Metals Arsenic Barium Beryllium Cadmium Chromium [b] Copper Lead Nickel Silver Zinc <u>Cyanide</u>	12.40 230.00 2.60 24.60 28.40 189.00 169.00 20.00 2.00 366.00 2.50	2.4 100,000 1.1 850 450 63,000 400 34,000 8,500 100,000 35	2.7 100,000 2,200 810 450 76,000 750 41,000 10,000 35	10 5,300 1.4 38 2,100 2,800 400 1,500 380 23,000 1,300	
		<u>VOC</u>	0.40	000	500	700	
OU-1	SD-20	Toluene <u>BNAs</u> Benzo(a)anthracene [e] Benzo(b)fluoranthene [e] Benzo(g,h,i)perylene [c] Bis(2-ethylhexyl)phthalate Chrysene Di-n-octylphthalate Fluoranthene Indeno(1,2,3-c,d)pyrene Phenanthrene [c] Pyrene	0.10 0.30 0.32 0.22 0.54 0.41 0.18 0.65 0.20 0.32 0.64 2.700 0	880 2.6 2.6 100 140 7.2 10,000 27,000 0.61 100 100	520 2.9 0.29 2.9 54,000 180 62 10,000 30,000 0.62 54,000 54,000	790 6.1 0.61 6.1 NA 320 610 1,300 2,600 6.1 NE 2,000	
		<u>Metals</u> Antimony Arsenic Barium Beryllium Cadmium Chromium [b] Copper Lead Nickel Zinc	0.6 26.0 532.0 0.9 4.3 81.5 36.2 118.0 26.3 157.0	NA 2.4 100,000 1.1 850 450 63,000 400 34,000 100,000	820 2.7 100,000 2,200 810 450 76,000 750 41,000 100,000	31 10 5,300 1.4 38 2,100 2,800 400 1,500 23,000	
		BNAs					
		Anthracene Benzo(a)anthracene [e] Benzo(b)fluoranthene [e] Benzo(k)fluoranthene Benzo(g,h,i)perylene [c] Benzo(a)purceno [a]	0.085 0.48 1.5 1.4 0.42	5.7 2.6 2.6 26 100 0.26	100,000 2.9 2.9 29 54,000	20,000 6.1 6.1 61 NA 0.61	
		Chrysene	0.67	7.2	62	610	
		Dibenzo(a,h)anthracene [e]	0.085	0.26	0.29	0.61	

OU-1	SD-21	Fluoranthene Indeno(1,2,3-c,d)pyrene Phenanthrene [c] Pyrene TRPH [a]	0.97 0.57 0.4 0.88 10	27,000 0.61 100 100 NA	30,000 0.62 54,000 54,000 NA	2,600 6.1 NE 2,000 4.100
		Metals Arsenic Barium Beryllium Cadmium Chromium [b] Copper Lead Nickel Silver Zinc	8.2 148 0.6 1.2 19.4 32.7 16 20 2 69.5	2.4 100,000 1.1 850 450 63,000 400 34,000 8,500 100,000	2.7 100,000 2,200 810 450 76,000 750 41,000 10,000 100,000	10 5,300 1.4 38 2,100 2,800 400 1,500 380 23,000
		<u>VOCs</u> Ethylbenzene Toluene Xylenes (total)	4 3 18	690 880 320	230 520 210	1,500 790 2,800
OU-1	SD-26	BNAs 2-Methylnaphthalene [d] Benzo(a)anthracene [e] Benzo(a)pyrene [e] Benzo(b)fluoranthene [e] Benzo(g,h,i)perylene [c] Benzo(k)fluoranthene Bis(2-ethylhexyl)phthalate Chrysene Di-n-butylphthalate Fluoranthene Indeno(1,2,3-cd)pyrene Naphthalene Phenanthrene [c] Pyrene TRPH [a] Metals Arsenic Barium Beryllium Cadmium Chromium [b] Copper Lead Nickel Silver	6.5 0.097 0.11 0.18 0.066 0.086 1.7 0.14 7.3 0.23 0.053 1.7 0.085 0.23 19,000 20 742 0.8 3.7 41.6 35.1 20 21 1.4	240 2.6 0.26 2.6 100 26 140 7.2 68,000 27,000 0.61 240 100 100 NA 2.4 100,000 1.1 850 450 63,000 400 34,000 8,500	190 2.9 0.29 2.9 54,000 29 180 62 88,000 30,000 0.62 190 54,000 54,000 NA 2.7 100,000 2,200 810 450 76,000 750 41,000 10,000	NE 6.1 0.61 6.1 NA 61 320 610 6,500 2,600 6.1 2,600 NE 2,000 4,100 10 5,300 1.4 38 2,100 2,800 400 1,500 380 000 000 000 000 000 000 0
		<u>BNAs</u> Benzo(a)anthracene [e]	0.054	2.6	2.9	6.1
OU-1	LF-37	Benzo(a)pyrene [e] Benzo(b)fluoranthene [e] Benzo(g,h,i)perylene [c] Benzo(k)fluoranthene Bis(2-ethylhexyl)phthalate Butylbenzylpthalate Chrysene Fluoranthene Pyrene <u>TRPH</u> [a]	0.425 0.425 0.425 0.425 0.425 1.2 0.062 0.425 0.425 0.425 540	0.26 2.6 100 26 140 930 7.2 27,000 100 NA	0.29 2.9 54,000 29 180 100,000 62 30,000 54,000 NA	0.61 6.1 NA 61 320 13,000 610 2,600 2,000 4,100

		Arsenic	9.6	2.4	2.7	10	
		Barium	334	100,000	100,000	5,300	
		Bervllium	0.8	1.1	2.200	1.4	
		Cadmium	29.5	850	810	38	
		Chromium [b]	28.2	450	450	2.100	
		Copper	561	63,000	76.000	2,800	
		Lead	597	400	750	400	
		Nickel	58 5	34 000	41 000	1 500	
		Silver	3.4	8 500	10,000	380	
		Zinc	2 270	100 000	100,000	23 000	
			2,210	100,000	100,000	20,000	_
		BNA					
		Diethyl phthalate	0.042	100,000	100,000	52,000	
		<u>TRPH</u> [a]	2,000	NA	NA	4,100	
		Metals					
		Arsenic	14	2.4	2.7	10	
011-1	SD-39	Barium	220	100.000	100.000	5.300	
00 1		Cadmium	1.6	850	810	38	
		Chromium [b]	22.5	450	450	2,100	
		Copper	40.1	63,000	76.000	2,800	
		Lead	125	400	750	400	
		Nickel	25	34,000	41.000	1.500	
		Zinc	62.8	100,000	100,000	23,000	
				,	,		_
OU-1	OT-41	Metal					
		Lead	22	400	750	400	
	OT-01		f haaa waaanda kad	in a tarak tarak taran a	udavia urata dala		
OU-1	OT-08	An extensive data review of	or base records ind	icated that haza	rdous materiais a	and wastes were	
	OT-09	never handled of disposed	at these locations	•			
011.1	OT 10	This site lies completely w	ithin the houndarie	c of DP 12			_
00-1	01-10	This site lies completely w		S 01 DF=13.			_
	SS-15	These three sites were ren	noved from the sur	perfund process	and placed unde		
OU-1	SS-16	iurisdiction					
	ST-19	Junisaletion.					
		Removed from the Superfu	und process becau	se this site was	mistakenly inclu	ded on the list of	
00-1	DP-24	potentially contaminated s	ites.				
	[a]	n-Hexane us used as a su	rrogate for compar	ison to the Regio	on IX PRG.		
	ri. 1	The chromium PRG is bas	ed on total chromi	um carcinogenic	enects and		
	[a]	a 1:6 ratio of chromium IV	to chromium III.				
	[c]	Pyrene is used as a surrog	pate for comparisor	n to the Region I	X PRG.		
	[a]	Napthalene is used as a si	urrogate for compa	rison to the Reg	ION IX PRG.		
	[e]	The PRG is based on napl	nthalene non-carci	nogenic effects.			
	NE	Not Established					
	NA	Not Available					

SRL	
Non-Residential	
1,400 140,000 18,000	
63,000 2,000	
2,700 2,800	
1,400 NE 27,000 18,000	
63,000 2,000	
NA 2,700 400 11 170 2,700 70 2,800	
41,000 200,000 26 26 260 NA 2.6 1,400 140,000 2,600 26 2,700 68,000 27,000 26 NE 3,400 27,000 79 NE 410,000	
20,000	

,
2,000
26
20
INA 1 100
1,400
2,600
27,000
NE
27,000
20,000
18 000
10,000
63,000
2,000
0 000
8,800
18,000
·
00.000
63,000
2,000
8 800
1 4
2 700
2,700
2,700
2,800
NF
27 000
21,000
18,000
18,000
18,000
18,000 63,000
18,000 63,000 2,000
18,000 63,000 2,000 13
18,000 63,000 2,000 13
18,000 63,000 2,000 13
18,000 63,000 2,000 13 NE
18,000 63,000 2,000 13 NE 200,000
18,000 63,000 2,000 13 NE 200,000 26
18,000 63,000 2,000 13 NE 200,000 26 2.6
18,000 63,000 2,000 13 NE 200,000 26 2.6 2.6 26
18,000 63,000 2,000 13 NE 200,000 26 2.6 2.6 26 NA
18,000 <u>63,000</u> <u>2,000</u> <u>13</u> NE 200,000 <u>26</u> <u>2.6</u> <u>26</u> <u>2.6</u> <u>26</u> NA <u>260</u>
18,000 <u>63,000</u> <u>2,000</u> <u>13</u> NE 200,000 26 2.6 26 26 26 NA 260 950
18,000 63,000 2,000 13 NE 200,000 26 2.6 26 NA 260 950 2,600
18,000 <u>63,000</u> <u>2,000</u> <u>13</u> NE 200,000 26 2.6 26 NA 260 950 2,600 26
18,000 <u>63,000</u> <u>2,000</u> <u>13</u> <u>NE</u> 200,000 <u>26</u> <u>2.6</u> <u>26</u> <u>26</u> <u>8</u> <u>8</u> <u>8</u> <u>9</u> <u>50</u> <u>2</u> ,600 <u>26</u> <u>2</u> ,600 <u>26</u> <u>27,000</u>
18,000 <u>63,000</u> <u>2,000</u> <u>13</u> <u>NE</u> 200,000 <u>26</u> <u>2.6</u> <u>26</u> <u>26</u> <u>26</u> <u>26</u> <u>26</u> <u>26</u> <u>950</u> <u>2,600</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u>
18,000 <u>63,000</u> <u>2,000</u> <u>13</u> NE 200,000 <u>26</u> <u>2.6</u> <u>2.6</u> <u>26</u> <u>2.6</u> <u>26</u> <u>26</u> <u>950</u> <u>2,600</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u>
18,000 63,000 2,000 13 NE 200,000 26 2.6 26 2.6 26 NA 260 950 2,600 26 27,000 26 27,000 26 27,000 26 27,000 26 27,000 26 27,000 26 27,000 26 27,000 26 27,000 26 27,000 26 27,000 26 27,000 26 27,000 26 27,000 26 27,000 26 26 26 27,000 26 27,000 26 26 26 26 26 26 26 26 26 26
18,000 63,000 2,000 13 NE 200,000 26 2.6 26 2.6 26 950 2,600 26 27,000 26 27,000 26 79 NE 20,000
18,000 63,000 2,000 13 NE 200,000 26 2.6 26 26 26 950 2,600 26 27,000 26 27,000 26 79 NE 20,000
18,000 63,000 2,000 13 NE 200,000 26 2.6 26 26 26 NA 260 950 2,600 26 27,000 26 27,000 26 79 NE 20,000 18,000
18,000 <u>63,000</u> <u>2,000</u> <u>13</u> <u>NE</u> 200,000 <u>26</u> <u>2.6</u> <u>26</u> <u>26</u> <u>8</u> <u>8</u> <u>8</u> <u>8</u> <u>9</u> <u>8</u> <u>8</u> <u>8</u> <u>8</u> <u>8</u> <u>8</u> <u>8</u> <u>8</u>
18,000 <u>63,000</u> <u>2,000</u> <u>13</u> <u>NE</u> 200,000 <u>26</u> <u>2.6</u> <u>26</u> <u>26</u> <u>8</u> <u>8</u> <u>8</u> <u>8</u> <u>9</u> <u>8</u> <u>8</u> <u>8</u> <u>8</u> <u>8</u> <u>8</u> <u>8</u> <u>8</u>
18,000 <u>63,000</u> <u>2,000</u> <u>13</u> <u>NE</u> 200,000 <u>26</u> <u>2.6</u> <u>26</u> <u>26</u> <u>26</u> <u>26</u> <u>26</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>79</u> <u>NE</u> <u>20,000</u> <u>18,000</u> <u>10</u>
18,000 <u>63,000</u> <u>2,000</u> <u>13</u> <u>NE</u> 200,000 <u>26</u> <u>2.6</u> <u>26</u> <u>26</u> <u>26</u> <u>950</u> <u>2,600</u> <u>26</u> <u>27,000</u> <u>26</u> <u>79</u> <u>NE</u> <u>20,000</u> <u>18,000</u> <u>10</u> <u>110,000</u>
18,000 <u>63,000</u> <u>2,000</u> <u>13</u> <u>NE</u> 200,000 <u>26</u> <u>2.6</u> <u>26</u> <u>26</u> <u>26</u> <u>26</u> <u>26</u> <u>26</u> <u>26</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>18,000</u> <u>10</u> <u>110,000</u> <u>11</u>
18,000 <u>63,000</u> <u>2,000</u> <u>13</u> <u>NE</u> 200,000 <u>26</u> <u>2.6</u> <u>26</u> <u>26</u> <u>26</u> <u>26</u> <u>26</u> <u>26</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>27,000</u> <u>26</u> <u>79</u> <u>NE</u> <u>20,000</u> <u>18,000</u> <u>10</u> <u>110,000</u> <u>11</u> <u>850</u>

63,000
2,000
34,000
510,000
14,000
2,600
14,000
20,000
20,000
13
18,000
10
110 000
11
850
4,500
63,000
2,000
34,000
8,500
510,000
14,000
2,700
26
2.6
26
NA
1,400
2,600
14,000
27,000
26
NE
20,000
18,000
680
10
11
850
4.500
63,000
2,000
34,000
510,000
200,000
20 26
20 260
NΔ
2.6
2.600

27,000 26 NE 20,000 18,000	
10 110,000 11 850 4,500 63,000 2,000 34,000 8,500 510,000	
2,700 2,700 2,800	
NE 26 2.6 26 NA 260 1,400 2,600 68,000 27,000 26 27,000 NE 20,000 18,000	
10 110,000 11 850 4,500 63,000 2,000 34,000 8,500 510,000	
26 2.6 26 NA 260 1,400 140,000 2,600 27,000 20,000 18,000	
$ \begin{array}{r} 110,000\\ 11\\ 850\\ 4,500\\ 63,000\\ 2,000\\ 34,000\\ 8,500\\ 510,000\\ \hline 550,000\\ 18,000\\ 10\\ 110,000\\ 850\\ 4,500\\ 63,000\\ 2,000\\ 34,000\\ 510,000\\ \hline \end{array} $	

11 850 4,500 63,000 2,000 34,000 8,500 510,000 18,000 10 10 110,000 850 4,500 63,000 2,000 34,000 510,000	
850 4,500 63,000 2,000 34,000 8,500 510,000 550,000 18,000 10 110,000 850 4,500 63,000 2,000 34,000 510,000	
4,500 63,000 2,000 34,000 8,500 510,000 550,000 18,000 10 110,000 850 4,500 63,000 2,000 34,000 510,000	
63,000 2,000 34,000 8,500 510,000 550,000 18,000 18,000 10 110,000 850 4,500 63,000 2,000 34,000 510,000	
2,000 34,000 8,500 510,000 550,000 18,000 10 110,000 850 4,500 63,000 2,000 34,000 510,000	
34,000 8,500 510,000 550,000 18,000 10 110,000 850 4,500 63,000 2,000 34,000 510,000	
8,500 510,000 550,000 18,000 10 110,000 850 4,500 63,000 2,000 34,000 510,000	
510,000 550,000 18,000 10 110,000 850 4,500 63,000 2,000 34,000 510,000	
550,000 18,000 10 110,000 850 4,500 63,000 2,000 34,000 510,000	
550,000 18,000 10 110,000 850 4,500 63,000 2,000 34,000 510,000	
18,000 10 110,000 850 4,500 63,000 2,000 34,000 510,000	
10 110,000 850 4,500 63,000 2,000 34,000 510,000	
10 110,000 850 4,500 63,000 2,000 34,000 510,000	
110,000 850 4,500 63,000 2,000 34,000 510,000	
850 4,500 63,000 2,000 34,000 510,000	
4,500 63,000 2,000 34,000 510,000	
63,000 2,000 34,000 510,000	
2,000 34,000 510,000	
34,000 510,000	
510,000	
2,000	

Appendix D

USEPA Region IX PRGs for 1991, 1996 and 2000

NECEIVED

NOV 0 9 2001

ARCADIS Geraghty & Miller



UNITED STATES ENVIRONMENTAL PROTECTION AGENCY

REGION IX 75 Hawthorne Street San Francisco, CA 94105-3901

August 1, 1996



Subject: Region 9 Preliminary Remediation Goals (PRGs) 1996

From: Stanford J. Smucker, Ph.D. Regional Toxicologist (H-9-3) Technical Support Team

To: PRG Table Mailing List

Please find the annual update to the Region 9 PRG table. The table has been revised to reflect the most current EPA toxicological and risk assessment information. Updates to EPA toxicity values were obtained from IRIS through July 1996, HEAST through May 1995, and EPA's National Center for Environmental Assessment (NCEA, formerly ECAO).

Region 9 PRGs are "evergreen" and have evolved as new methodologies and parameters have been developed. In several cases the models, equations, and assumptions presented in RAGS HHEM, Part B, Development of Risk-Based Preliminary Remediation Goals (1991) have been replaced with new information that is consistent with the document, Soil Screening Guidance, recently issued by the Office of Solid Waste and Emergency Response (OSWER), dated April 1996.

The updated PRG table also contains soil screening levels (SSLs) for protection of groundwater. The SSLs were obtained directly from EPA/OSWER's *Soil Screening Guidance* document which is available from NTIS as EPA/540/R-96/018 and EPA/540/R-95/128. Please note that because R 9 PRGs currently evaluate intermedia transfer of volatile organic chemicals (VOCs) and heavy metals from soil to air, the PRG table does not include a separate list of SSLs for the air pathway.

To help users rapidly identify substances with new PRGs, these contaminants are printed in boldface type. Changes in PRG values are either due to new toxicity constants or new physico-chemical information. This version of the table contains revised toxicity values for acetaldehyde, chlorine cyanide, 1,3-dichlorobenzene, 2-dichloroethane, endosulfan, manganese, phosphoric acid, and 1,1,1-trichloroethane. Also, 23 additional VOCs have been identified and evaluated for inhalation exposures resulting from intermedia transfer from soil and water to air.

EPA Region 9 has established a homepage on the World Wide Web which you can find at http://www.epa.gov/region 9/. Our homepage will soon include the PRG table in downloadable form. The electronic table contains additional information not presented in the printed table (e.g. physico-chemical constants, non-cancer PRGs for carcinogens, pathway-specific PRGs, and volatilization factors for VOCs). Meanwhile, we still provide the electronic PRG table (PRG96.zip) on California Regional Water Board's BBS (510.286.0404) for those of you who have a modem.

Before relying on any number in the table, it is recommended that the user verify the numbers with an agency toxicologist or risk assessor because the toxicity / exposure information in the table may contain errors or default assumptions that need to be refined based on further evaluation. If you find an error please send me a note via email at Smucker.Stan@epamail.epa.gov or fax at 415.744.1916.

Printed on Recycled Paper

EXHIBIT 1-1 - TYPICAL EXPOSURE PATHWAYS BY MEDIUM FOR RESIDENTIAL AND INDUSTRIAL LAND USES*

EE	XPOSURE PATHWAYS, ASSU	MING:				
MEDIUM	RESIDENTIAL LAND USE	INDUSTRIAL LAND USE				
Ground Water	Ingestion from drinking	Ingestion from drinking				
	Inhalation of volatiles	Inhalation of volatiles				
	Dermal absorption from bathing	Dermal absorption				
Surface Water	Ingestion from drinking	Ingestion from drinking				
	Inhalation of volatiles	Inhalation of volatiles				
	Dermal absorption from bathing	Dermal absorption				
	Ingestion during swimming					
	Ingestion of contaminated fish					
Soil	Ingestion	Ingestion				
	Inhalation of particulates	Inhalation of particulates				
	Inhalation of volatiles	Inhalation of volatiles				
	Exposure to indoor air from soil gas	Exposure to indoor air from soil gas				
	Exposure to ground water contaminated by soil leachate	Exposure to ground water contaminated by soil leachate				
	Ingestion via plant, meat, or dairy products	Inhalation of particulates from trucks and heavy equipment				
	Dermal absorption	Dermal absorption				

Footnote:

*Exposure pathways considered in the PRG calculations are indicated in boldface italics.

pollutant risks.

In addition to Region 9 PRGs, the PRG table also includes California EPA PRGs ("CAL-Modified PRGs") for specific chemicals where CAL-EPA values may be more restrictive than the federal values; and, soil screening levels (SSLs) for protection of groundwater (see Section 2.3 below).

2.2 Toxicity Values

EPA toxicity values, known as noncarcinogenic reference doses (RfD) and carcinogenic slope factors (SF) were obtained from IRIS through July 1996, HEAST through May 1995, and EPA's National Center for Environmental Assessment (NCEA, formerly ECAO). The priority among sources of toxicological constants used are as follows: (1) IRIS (indicated by "i"), (2) HEAST ("h"), (3) NCEA ("n"), and (4) withdrawn from IRIS or HEAST and under review ("x").

Route-to-route extrapolations ("r") were frequently used when there were no toxicity values available for a given route of exposure. Oral cancer slope factors ("SFo") and reference doses ("RfDo") were used for both oral and inhaled exposures for organic compounds lacking inhalation values. Inhalation slope factors ("SFi") and inhalation reference doses ("RfDi") were used for both inhaled and oral exposures for organic compounds lacking oral values. An additional route extrapolation is the use of oral toxicity values for evaluating dermal exposures. Although route-to-route methods are a useful screening procedure, the appropriateness of these default assumptions for specific contaminants should be verified by a toxicologist.

To help users rapidly identify substances with new PRGs, these contaminants are printed in boldface type. This version of the table contains revised toxicity values for acetaldehyde, chlorine cyanide, 1,3-dichlorobenzene, 2-dichloroethane, endosulfan, manganese, phosphoric acid, and 1,1,1-trichloroethane.

2.3 Soil Screening Levels

Generic soil screening levels (SSLs) for the protection of groundwater have been included in the PRG table for 100 of the most common contaminants at Superfund sites. Generic SSLs are derived using default values in standardized equations presented in *Soil Screening Guidance* (available from NTIS as document numbers PB96-963502 and PB96-963505 or EPA/540/R-95/128 and EPA/540/R-96/018).

The SSLs were developed using a default dilution-attenuation factor (DAF) of 20 to account for natural processes that reduce contaminant concentrations in the subsurface. Also included are generic SSLs that assume no dilution or attenuation between the source and the receptor well (i.e., a DAF of 1). These values can be used at sites where little or no dilution or attenuation of soil leachate concentrations is expected at a site (e.g., sites with shallow water

3.1 Developing a Conceptual Site Model

The primary condition for use of PRGs is that exposure pathways of concern and conditions at the site match those taken into account by the PRG framework. Thus, it is always necessary to develop a conceptual site model (CSM) to identify likely contaminant source areas, exposure pathways, and potential receptors. This information can be used to determine the applicability of PRGs at the site and the need for additional information. For those pathways not covered by PRGs, a risk assessment specific to these additional pathways may be necessary. Nonetheless, the PRG lookup values will still be useful in such situations for focusing further investigative efforts on the exposure pathways not addressed.

To develop a site-specific CSM, perform an extensive records search and compile existing data (e.g. available site sampling data, historical records, aerial photographs, and hydrogeologic information). Once this information is obtained, CSM worksheets such as those provided in ASTM's *Standard Guide for Risk-Based Corrective Action Applied at Petroleum Release Sites* (1995) can be used to tailor the generic worksheet model to a site-specific CSM. The final CSM diagram represents linkages among contaminant sources, release mechanisms, exposure pathways and routes and receptors. It summarizes our understanding of the contamination problem.

As a final check, the CSM should answer the following questions:

- Are there potential ecological concerns?
- Is there potential for land use other than those covered by the PRGs (that is, residential and industrial)?
- Are there other likely human exposure pathways that were not considered in development of the PRGs (e.g. impact to groundwater, local fish consumption; raising beef, dairy, or other livestock)?
- Are there unusual site conditions (e.g. large areas of contamination, high fugitive dust levels, potential for indoor air contamination)?

If any of these four conditions exist, the PRG may need to be adjusted to reflect this new information. Suggested references for evaluating pathways not currently evaluated by Region 9 PRG's are presented in Exhibit 3-1.

consulting a staff toxicologist at state and / or federal regulatory agencies.

Where anthropogenic background levels exceed PRGs and EPA has determined that a response action is necessary and feasible, EPA's goal will be to develop a comprehensive response to the widespread contamination. This will often require coordination with different authorities that have jurisdiction over the sources of contamination in the area.

3.3 Risk Screening

A suggested stepwise approach for screening sites with PRGs is as follows:

Perform an extensive records search and compile existing data.

Identify site contaminants in the PRG Table. Record the PRG concentrations for various media and note whether PRG is based on cancer risk (indicated by "ca") or noncancer hazard (indicated by "nc"). Segregate cancer PRGs from non-cancer PRGs and exclude (but don't eliminate) non-risk based PRGs ("sat" or "max").

For cancer risk estimates, take the site-specific concentration (maximum or 95 UCL) and divide by the PRG concentrations that are designated for cancer evaluation ("ca"). Multiply this ratio by 10⁻⁶ to estimate chemical-specific risk. For multiple pollutants, simply add the risk for each chemical :

$$Risk = \left[\left(\frac{CONC_x}{PRG_x}\right) + \left(\frac{CONC_y}{PRG_y}\right) + \left(\frac{CONC_z}{PRG_z}\right)\right] \times 10^{-6}$$

For non-cancer hazard estimates. Divide concentration term by its respective non-cancer PRG designated as "nc" and sum the ratios for multiple contaminants. [Note that carcinogens may also have an associated non-cancer PRG that is not listed in the printed copy of the table and these will also need to be obtained in order to complete the non-cancer evaluation.] The non-cancer ratio represents a hazard index (HI). A hazard index of 1 or less is generally considered safe. A ratio greater than 1 suggests further evaluation:

Hazard Index =
$$\left[\left(\frac{CONC_x}{PRG_x}\right) + \left(\frac{CONC_y}{PRG_y}\right) + \left(\frac{CONC_z}{PRG_z}\right)\right]$$

For more information on screening site risks, the reader should contact EPA Region 9's Technical Support Team.

Document (USEPA 1996a,b).

To address the soil-to-air pathways the PRG calculations incorporate volatilization factors (VF₄) for volatile contaminants and particulate emission factors (PEF) for nonvolatile contaminants. These factors relate soil contaminant concentrations to air contaminant concentrations that may be inhaled on-site. The VF₄ and PEF equations can be broken into two separate models: an emission model to estimate emissions of the contaminant from the soil and a dispersion model to simulate the dispersion of the contaminant in the atmosphere.

It should be noted that the box model in RAGS Part B has been replaced with a dispersion term (Q/C) derived from a modeling exercise using meteorological data from 29 locations across the United States because the box model may not be applicable to a broad range of site types and meteorology and does not utilize state-of-the-art techniques developed for regulatory dispersion modeling. The dispersion model for both volatiles and particulates is the AREA-ST, an updated version of the Office of Air Quality Planning and Standards, Industrial Source Complex Model, ISC2. However, different Q/C terms are used in the VF and PEF equations. Los Angeles was selected as the 90th percentile data set for volatiles and Minneapolis was selected as the 90th percentile data set for fugitive dusts (USEPA 1996 a,b). A default source size of 0.5 acres was chosen for the PRG calculations. This is consistent with the default exposure area over which Region 9 typically averages contaminant concentrations in soils. If unusual site conditions exist such that the area source is substantially larger than the default source size assumed here, an alternative Q/C could be applied (see USEPA 1996a,b).

Volatilization Factor for Soils

Volatile chemicals, defined as those chemicals having a Henry's Law constant greater than 10^{-5} (atm-m³/mol) and a molecular weight less than 200 g/mole, were screened for inhalation exposures using a volatilization factor for soils (VF₃). Please note that VF₃'s are available in the electronic version of the PRG table.

The emission terms used in the VF, are chemical-specific and were calculated from physicalchemical information obtained from a number of sources including Superfund Exposure Assessment Manual (SEAM, EPA 1988), Subsurface Contamination Reference Guide (EPA 1990a), Fate and Exposure Data (Howard 1991), and Superfund Chemical Data Matrix (USEPA 1994c). In those cases where Diffusivity Coefficients (Di) were not provided in existing literature, Di's were calculated using Fuller's Method described in SEAM. A surrogate term was required for some chemicals that lacked physico-chemical information. In these cases, a proxy chemical of similar structure was used that may over- or under-estimate the PRG for soils. Physico-chemical information is available in the electronic version of the PRG table. To access this information, the user should display the hidden columns in the table.

Equation 4-9 forms the basis for deriving generic soil PRGs for the inhalation pathway. The following parameters in the standardized equation can be replaced with specific site data to

Note: the generic PEF evaluates windborne emissions and does not consider dust emissions from traffic or other forms of mechanical disturbance that could lead to greater emissions than assumed here.

4.2 Dermal Contact with Contaminants in Soil

Much uncertainty surrounds the determination of hazards associated with skin contact with soils. One important data gap is the lack of EPA verified toxicity values for the dermal route. For screening purposes it is assumed that dermal toxicity values can be route-to-route extrapolated from oral values but this may not always be an appropriate assumption and should be checked.

Thus far, chemical-specific absorption values for skin have been recommended for only five chemicals (arsenic, cadmium, pentachlorophenol, PCBs, and dioxin) by EPA's Office of Research and Development. For all other chemicals, default absorption values for inorganics and organics are assumed to be 1 and 10 percent, respectively. At 10 % skin absorption, the dermal dose is estimated to equal an ingestion dose for adults, using the best estimate default values in *Dermal Exposure Assessment: Principles and Applications* (EPA 1992). At 1 % absorption, the dermal dose is estimated to be 10% of the oral dose (i.e. based on an adult ingestion rate of 100 mg per day). Note: worker and children intake rates (50 and 200 mg per day, respectively) yield somewhat different results.

dermal dose = ingestion dose

 $C_{SOIL} \times ABS \times AF \times SA = C_{SOIL} \times IR$

 $ABS = \frac{(100mg/day)}{[(0.2mg/cm^2-day)(5000cm^2)]} = 0.10$

4.3 SSLs for the Migration to Groundwater Pathway

The methodology for calculating SSLs for the migration to groundwater was developed to identify chemical concentrations in soil that have the potential to contaminate groundwater. Migration of contaminants from soil to groundwater can be envisioned as a two-stage process: (1) release of contaminant in soil leachate and (2) transport of the contaminant through the underlying soil and aquifer to a receptor well. The SSL methodology considers both of these fate and transport mechanisms.

SSLs are backcalculated from acceptable ground water concentrations (i.e. nonzero MCLGs, MCLs, or risk-based PRGs). First, the acceptable groundwater concentration is multiplied by a dilution factor to obtain a target leachate concentration. For example, if the dilution factor

(2) skin contact([mg•yr]/[kg•d]:

$$SFS_{adj} = \frac{ED_c \times AF \times SA_c}{BW_c} + \frac{(ED_r - ED_c) \times AF \times SA_a}{BW_a}$$

(3) inhalation $([m^3 \cdot yr]/[kg \cdot d])$:

$$InhF_{adj} = \frac{ED_c \times IRA_c}{BW_c} + \frac{(ED_r - ED_c) \times IRA_a}{BW_a}$$

4.5 **PRG Equations**

The equations used to calculate the PRGs for carcinogenic and noncarcinogenic contaminants are presented in Equations 4-1 through 4-8. The PRG equations update RAGS Part B equations. Briefly, PRGs are risk assessments run in reverse. The methodology backcalculates a soil, air, or water concentration level from a target risk (for carcinogens) or hazard quotient (for noncarcinogens). For completeness, the soil equations combine risks from ingestion, skin contact, and inhalation simultaneously. Note: the electronic version of the table also includes pathway-specific PRGs, should the user decide against combining specific exposure pathways; or, the user wants to identify the relative contribution of each pathway to exposure.

To calculate PRGs for volatile chemicals in soil, a chemical-specific volatilization factor is calculated per Equation 4-9. Because of its reliance on Henry's law, the VF, model is applicable only when the contaminant concentration in soil is at or below saturation (i.e. there is no free-phase contaminant present): Soil saturation ("sat") corresponds to the contaminant concentration in soil at which the adsorptive limits of the soil particles and the solubility limits of the available soil moisture have been reached. Above this point, pure liquid-phase contaminant is expected in the soil. If the PRG calculated using VF, was greater than the calculated sat, the PRG was set equal to sat, in accordance with *Soil Screening Guidance* (USEPA 1996 a,b). The updated equation for deriving sat is presented in Equation 4-10.

PRG EQUATIONS

Soil Equations: For soils, equations were based on three exposure routes (ingestion, skin contact, and inhalation).

Equation 4-1: Combined Exposures to Carcinogenic Contaminants in Residential Soil

$$C(mg/kg) = \frac{TR \times AT_{c}}{EF_{r} \left[\left(\frac{IFS_{adj} \times CSF_{o}}{10^{6}mg/kg} \right) + \left(\frac{SFS_{adj} \times ABS \times CSF_{o}}{10^{6}mg/kg} \right) + \left(\frac{InhF_{adj} \times CSF_{i}}{VF_{o}^{4}} \right) \right]}$$

Equation 4-2: Combined Exposures to Noncarcinogenic Contaminants in Residential Soil

$$C(mg/kg) = \frac{THQ \times BW_c \times AT_n}{EF_x \times ED_c \left[\left(\frac{1}{RfD_o} \times \frac{IRS_c}{10^6 mg/kg}\right) + \left(\frac{1}{RfD_o} \times \frac{SA_c \times AF \times ABS}{10^6 mg/kg}\right) + \left(\frac{1}{RfD_i} \times \frac{IRA_c}{VF_s^a}\right) \right]$$



$$C(mg/kg) = \frac{TR \times BW_a \times AT_c}{EF_o \times ED_o \left[\left(\frac{IRS_o \times CSF_o}{10^6 mg/kg} \right) + \left(\frac{SA_a \times AF \times ABS \times CSF_o}{10^6 mg/kg} \right) + \left(\frac{IRA_a \times CSF_i}{VF_a^6} \right) \right]}$$

Equation 4-4: Combined Exposures to Noncarcinogenic Contaminants in Industrial Soil

$$C(mg/kg) = \frac{THQ \times BW_a \times AT_n}{EF_o \times ED_o[(\frac{1}{RfD_o} \times \frac{IRS_o}{10^6 mg/kg}) + (\frac{1}{RfD_o} \times \frac{SA_a \times AF \times ABS}{10^6 mg/kg}) + (\frac{1}{RfD_i} \times \frac{IRA_a}{VF_a^a})]$$

^{*}Use VF, for volatile chemicals (defined as having a Henry's Law Constant [atm-m³/mol] greater than 10⁻⁶ and a molecular weight less than 200 grams/mol) or PEF for non-volatile chemicals.

SOIL-TO-AIR VOLATILIZATION FACTOR (VF.)

Equation 4-9: Derivation of the Volatilization Factor

$$VF_{g}(m^{3}/kg) = (Q/C) \times \frac{(3.14 \times D_{A} \times T)^{1/2}}{(2 \times \rho_{b} \times D_{A})} \times 10^{-4} (m^{2}/cm^{2})$$

where:

$$D_{A} = \frac{\left[\left(\Theta_{a}^{10/3}D_{i}H' + \Theta_{w}^{10/3}D_{w}\right)/n^{2}\right]}{\rho_{B}K_{d} + \Theta_{w} + \Theta_{a}H'}$$

<u>Parameter</u>	Definition (units)	Default
VF,	Volatilization factor (m ³ /kg)	_ ·
D,	Apparent diffusivity (cm ² /s)	-
Q/C	Inverse of the mean conc. at the center of a 0.5-acre square source (g/m²-s per kg/m³)	68.81
т.	Exposure interval (s)	9.5 x 10 ^a
Ръ	Dry soil bulk density (g/cm³)	1.5
Θ	Air filled soil porosity (لمن المنابع)	0.28 or n-⊖ _w
n	Total soil porosity (L _{pore} /L _{soil})	0.43 or 1 - (ρ⊾/ρ_)
Θ"	Water-filled soil porosity (L _{weter} /L _{soi})	0.15
ρ,	Soil particle density (g/cm ³)	2.65
Di	Diffusivity in air (cm²/s)	Chemical-specific
н	Henry's Law constant (atm-m³/mol)	Chemical-specific
H'	Dimensionless Henry's Law constant	Calculated from H by multiplying by 41 (USEPA 1991a)
D,	Diffusivity in water (cm ² /s)	Chemical-specific
Kª	Soil-water partition coefficient (cm ³ /g) = $K_{\infty}f_{\infty}$	Chemical-specific
K _∞	Soil organic carbon-water partition coefficient (cm ³ /g)	Chemical-specific
f _{oc}	Fraction organic carbon in soil (g/g)	0.006 (0.6%)

SOIL-TO-AIR PARTICULATE EMISSION FACTOR (PEF)

Equation 4-11: Derivation of the Particulate Emission Factor

$\mathbf{n} = \mathbf{n} \left(-\frac{3}{4} \right)$	$- \alpha c x$	3600 <i>s/h</i>
$PEF(m^2/Kg)$	= <u>0</u> / C X	$0.036 \times (1-V) \times (U_m/U_t)^3 \times F(x)$

<u>Parameter</u>	<u>Definition (units)</u>	Detaurt
PEF	Particulate emission factor (m ³ /kg)	1. 316 x 10°
Q/C	Inverse of the mean concentration at the center of a 0.5-acre-square source (g/m ² -s per kg/m ³)	90.80
V	Fraction of vegetative cover (unitless)	0.5
U _m	Mean annual windspeed (m/s)	4.69
U,	Equivalent threshold value of windspeed at 7 m (m/s)	11.32
F(x)	Function dependent on U _m /U _t derived using Cowherd (1985) (unitless)	0.194

REFERENCES

- ASTM. 1995. Standard Guide for Risk-Based Corrective Action Applied at Petroleum Release Sites. Designation E 1739 - 95. Philadelphia, Pennsylvania.
- California EPA. 1994. Preliminary Endangerment Assessment Guidance Manual. Department of Toxic Substances Control, Sacramento, California.
- Cowherd, C., G. Muleski, P. Engelhart, and D. Gillette. 1985. *Rapid Assessment of Exposure to Particulate Emission from Surface Contamination*. EPA/600/8-85/002. Prepared for Office of Health and Environmental Assessment, U.S. Environmental Protection Agency, Washington, DC. NTIS PB85-192219 7AS.
- Howard, P.H. 1990. Handbook of Environmental Fate and Exposure Data for Organic Chemicals. Lewis Publishers, Chelsea Michigan.
- U.S. EPA. 1988. Superfund Exposure Assessment Manual. EPA/540/1-88/001. Office of Emergency and Remedial Response, Washington, DC.
- U.S. EPA. 1990a. Subsurface Contamination Reference Guide. EPA/540/2-90/011. Office of Emergency and Remedial Response, Washington, DC.

						FOR PLANNING	BPUR	POSE	S			
T	DXICITY IN	FORMA	TION			IL POLICONTAMINANT	PRELIM	INARY REMED	AL GOALS (P	RGs)	SOIL SCREEN	ING LEVELS
SFo 1/(mg/kg-d)	RfDo (mg/kg-d)	6F1 /(mg/kg-d)	(mg/kg-d)	V skin O sbs C soil	CAS		III Residential . III Soli (mg/kg)	f Industrial (). Soll (mg/kg)	Amblent Alr	Tap Walar Main (ug/)	Migration to G DAF 20 ; ; ; ; ; ; ; ; ; ; ; ; ; ; ; ; ; ;	DAF 1
8.7E-03 I	4.0E-03 I	8.7E-03 r	4.0E-03 r	0 0.1	0 30580-1	1 Acephate	5.1E+01 ca**	2.2E+02 ca*	7.7E-01 ca*	7.7E+00 ca*		
7.7E-03 r	2.6E-03 r	7.7E-03 I	2.6E-03	1 0.1	0 75-07-0	Acetaldehyde	9.2E+00 ca**	2.1E+01 ca**	8.7E-01 ca*	1.5E+00 ca*		:
	2.0E-02		2.0E-02 r	0 0.1	0 34258-82	1 Acetochlor	1.3E+03 nc	<u>1.4E+04 nc</u>	7.3E+01 nc	7.3E+02 nc		
	1.0E-01 i		1.0E-01 r	1 0.1	0 67-64-1	Acetone	2.1E+03 nc	8.8E+03 nc	3.7E+02 nc	6.1E+02 nc	1.6E+01	8.0E-01
	8.0E-04 h		2.9E-03 x	0 0,1	0 75-86-5	 Acetone cyanohydrin 	5.2E+01 nc	5.5E+02 nc	1.0E+01 nc	2.9E+01 nc		
···	6 0E-03 I		1.4E-02 h	1 0.1	0 75-05-8	Acetonitrile	2.2E+02 nc	1.2E+03 nc	5.2E+01 nc	7.1E+01 nc		
	1.0E-01 I		5.7E-06 x	1 0.1	0 98-86-2	Acetophenone	4.9E-01 nc	1.6E+00 nc	2.1E-02 nc	4.2E-02 nc		
	1.3E-02 I		1.3E-02 r	0 0.1	0 50594-60	6 Acifluorfen	8.5E+02 nc	8.9E+03 nc	4.7E+01 nc	4.7E+02 nc		
	2.0E-02 h		5.7E-06 I	1 0.1	0 107-02-8	Acrolein	<u>1.0E-01 nc</u>	<u>3.4E-01 nc</u>	2.1E-02 nc	4.2E-02 nc		
4 8E+00 I	2.0E-04 I	4.6E+00 i	2.0E-04 r	0 0.1	0 79-08-1	Acrylamide	9.8E-02 ca*	4.2E-01 ca	1.5E-03 ca	1.5E-02 ca		
	5.0E-01 i		2.9E-04	0 0.1	0 78-10-7	Acrylic acid	3.1E+04 nc	2.9E+05 nc	1.0E+00 nc	1.8E+04 nc	1	
5.4E-011	1.0E-03 h	2.4E-01 I	5.7E-04 I	1 0.1	0 107-13-1		1.9E-01 ca*	4.7E-01 ca*	2.8E-02 ca*	3.7E+00 ca*		
8.1E-02 h	1.0E-02 I	6.0E-02 r	1.0E-02 r	0 0.1	0 15972-60	s Alachior	5.5E+00 ca*	2.4E+01 ca	8.4E-02 ca	8.4E-01 ca		
	1.5E-01 i		1.5E-01 r	0 0.1	0 1596-84	Alar	9.8E+03 nc	1.0E+05 nc	5.5E+02 nc	5.5E+03 nc		
	1.0E-03 I		1.0E-03 r	0 0.1	0 116-08-3	Aldicarb	6.5E+01 nc	6.8E+02 nc	3.7E+00 nc	<u>3.7E+01 nc</u>		i
	1.0E-03 I		1.0E-03 r	0 0.1	0 1646-88-	Aldicard sulfone	6.5E+01 nc	6.8E+02 nc	3.7E+00 nc	3.7E+01 nc		
1.7E+011	3 0E-05 I	1.7E+01 I	3.0E-05 r	0 0.1	0 309-00-2	Aldrin	2.6E-02 ca*	1.1E-01 ca	3.9E-04 ca	4.0E-03 ca	1.2E+04	5.9E+02
	2.5E-011		2.5E-01 r	0 0.1	0 5585-64-		1.6E+04 mc	1.0E+05 max	9.1E+02 nc	9.1E+03 nc		
	5.0E-03 x		5.0E-03 r	0 0.1	0 107-18-0	Ally alconol	3.3E+02 nc	3.4E+03 nc	1.8±+01 nc	1.8E+02 nc		
	5 0E-02 h		2.9E-04 I	0 0.1	0 107-05-1	Aliyi chionde	3.2E+03 nc	3.3E+04 nc	1.0E+00 nc	1.8E+03 nc		
}	1.0E+00 A			0 0.0	1 7429-90-		1.1E+04 mc	1.0E+03 max		3.7E+04 ng		
	4 05-04 1			0 0.0	01 20859-73	a Aluminum prospride	3.1E+01 nc	0.0E+02 nc	4 45 . 00	1.5E+01 nc		
	3 0E-04 I		3.0E-04 r	0 0.1	0 67485-2	4 Amaro	2.0E+01 nc	2.0E+02 nc	1.1E+00 nc	1.1E+01 nc		
	0.0E-03 I		9.0E-03 r	0 0.1	0 834-12-6	Ametryn	0.9E+02 nc	0.1E+03 nc	3.3E+01 nc	3.3E+02 nc		·····
	7.0E-02 h		7.0E-02 r	0 0.1	10 591-27-5	In-Aminophenoi	4.02+U3 nc	4.0E+04 ng	2.0E+U2 nc	2.0E+U3 nc		
1	20E-05 h		2.02-05 1	0 0.	10 504-24-2	4-Aminopynaine	1.3E+00 m	1.4C+U1 nc	7.3E-02 m	7.3E-01 nc		
	2.52-031		2.62-03 /	0 0.	0 33089-8	Ammonia	1.0C+U2 nc	1./E+U3 ng	9.1E+00 mc	9.1E+U1 no		
	206.011		4.9E-02 I		10 7773 04	Ammonium sulfamate	1 35+04	1 05+05	1.UETUZ ne	7 25-02		
57EMI	2.95-047	575 M .	2 95.041	0.0	0 42.53.3		1.05+04 he	2 0E+03 mbc	1 05+00	1.3E+03 M		
0.12.001	4.05.041	3.72-031	2.02.041	0 0.	1 7440 38	Antimony and compounds	3 1 5+01 -	6 8 5 102 nc	1.02+00 hc	1.12+01 10	5 05+00	2 05 01
	5 0E 04 b			0 0.	1 1314 80	Antimony pentovide	3.85+01	8 55+02 16		1.95401 nc	5.0E+00	3.02-01
1	9 0E-04 h			0.01	1 28300.7	s Antimony poincelus	6.9E+01 m	1 55+03 -		3 35+01		
	4 0E-04 h			0.0	1 1332.41	Antimony polassian lands	3 1E+01 m	6.8E+02 m		1.55+01 m		· · · · · · · · · · · · · · · · · · ·
	4 0E-04 h			0.00	11 1309.64	Antimony Identical	3 1E+01 ~	6.8E+02 m		1.5E+01		
	1 35-02 1		135-027	0.0	10 74115.2	s Apollo	8.5E+02 m	8.9E+03 m	4 7E+01 m	4 7E+02 m		
2 5E-02 1	5.0E-02 h	2 5E-02 1	5 0E-02 (0 0	10 140-57-4	Aramite	1 8F+01 cm	7.6F+01 cm	2 7F-01 ca	2 7E+00 ca		
	3 DE-04 I			0 0.	03 7440-38	Arsenic (noncancer endpoint)	2.2E+01 m					
1 5E+001	3 0E-04 I	1.5E+01 I		0.0	03 7440-38	Arsenic (cancer endpoint)	3.8E-01 ca*	2.4E+00 ca	4.5E-04 ca	4.5E-02 ca	2.9E+01	1.0E+00
			1.4E-05 I	N/8 N/	7784-42	Arsine			5.2E-02 m		1	
	9.0E-03 I		9.0E-03 r	0 0.	10 76576-1	a Assure	5.9E+02 m	6.1E+03 na	3.3E+01 ne	3.3E+02 №	1	
1	5.0E-02 I		5.0E-02 r	0 0.	10 3337-71	Asulam	3.3E+03 nc	3.4E+04 na	1.8E+02 nc	1.8E+03 nc		
2.2E-01 h	3.5E-02 h	2.2E-01 r	3.5E-02 h	0 0.	10 1912-24	Atrazine	2.0E+00 ca	8.6E+00 ca	3.1E-02 ca	3.0E-01 ca		
	4.0E-04 I		4.0E-04 r	0 0.	10 71751-4	2 Avermectin B1	2.6E+01 ng	2.7E+02 nc	1.5E+00 nc	1.5E+01 nc		
1.1E-01 I		1.1E-01 I		00.	10 103-33-:	Azobenzene	4.0E+00 ca	1.7E+01 ca	6.2E-02 ca	6.1E-01 a		
1	7.0E-021		1.4E-04 h	0 0.	C1 7440-38	Barium and compounds	5.3E+03 m	1.0E+05 max	5.2E-01 nc	2.6E+03 nc	1.6E+03	8.2E+01
1	4.0E-03 I		4.0E-03 r	00.	10 114-26-	Baygon	2.6E+02 mc	2.7E+03 nc	1.5E+01 nc	1.5E+02 nc	1	

.

.

, ¹³ г.

Key : HIRIS H-HEAST N-NCEA X-WITHDRAWN	r=ROUTE EXTRAPOLATION ca=CANCER PRG	nc=NONCANCER PRG sat=SOIL SATUR/	TION max=CEILING LIMIT *(w	here: nc < 100X ca) ™h	where: nc < 10X ca)

.

) (******) 	FOR PLANNING	PUR	POSE	S	Antonio antonio antonio antonio Contenta de la contenta de la content Contenta de la contenta de la content Contenta de la contenta de la content Contenta de la contenta de la content Contenta de la contenta de la content Contenta de la contenta	ta provinsi sa se	
T	DXICITY I	NFORMA [®]	TION	S.:	. (6)	n en	CONTAMINANT	PRELIMI	NARY REMED	AL GOALS (P	RGs)	SOIL SCREENIN	IG LEVELS
SFo 1/(mg/kg-d)	RfDo (mg/kg-d)	SFI 3	RfDi (mg/kg-d)	V I O I	ikin ibs, iolis	CAS No.		Residential (Soli (mg/kg)	Industrial Soil (mg/kg)	Amblent Air (ug/m^3)	. Tap Wulor (ug/)	Migration to Grou DAF 20 (mg/kg)	Ind Water DAF 1 (mg/kg)
	1.0E-02 i		1.0E-02 r	0	0,10	55285-14-8	Carbosulfan	6.5E+02 nc	6.8E+03 nc	3.7E+01 nc	3.7E+02 nc		
	1.0E-01 I		1.0E-01 r	0	0.10	5234-68-4	Carboxin	6.5E+03 nc	6.8E+04 nc	3.7E+02 nc	3.7E+03 nc		
	2.0E-03 (2:0E-03 r	0	0.10	302-17-0	Chloral	1.3E+02 nc	1.4E+03 nc	7.3E+00 nc	7.3E+01 nc		
	1.5E-02 I		1.5E-02 r	0	0.10	133-90-4	Chloramben	9.8E+02 nc	1.0E+04 nc	5.5E+01 nc	5.5E+02 nc		
4.0E-01 h		4.0E-01 r		0	0.10	118-75-2	Chloranil	1.1E+00 a	4.7E+00 ca	1.7E-02 a	1.7E-01 a		
1.3E+00 I	6.0E-05 I	1.3E+00 I	6.0E-05 r	0	0.10	67-74-9	Chlordane	3.4E-01 ca**	1.5E+00 ca*	5.2E-03 ca*	5.2E-02 cm	1.0E+01	5.0E-01
1	2 0E-02 I		2.0E-02 r	٥	0.10	90982-32-4	Chlorimuron-ethyl	1.3E+03 nc	1.4E+04 nc	7.3E+01 nc	7.3E+02 nc		
	1.0E-01 I			0	0.01	7782-50-5	Chlorine	7.7E+03 nc	1.7E+05 nc		3.7E+03 nc		
			5.7E-05 I	r/a	n/a	10049-04-4	Chlorine dloxIde			<u>2.1E-01 m</u>			
[1	0.10	107-20-0	Chloroacetaldehyde					1	
	2.0E-03 h		2.0E-03 r	0	0.10	79-11-8	Chloroacetic acid	1.3E+02 nc	1.4E+03 nc	7.3E+00 nc	7.3E+01 nc		
	8.6E-06 r		8.6E-06 I	1	0.10	532-27-4	2-Chloroacetophenone	3.2E-02 nc	1.1E-01 no	3.1E-02 nc	5.2E-02 nc		
	4.0E-03 I		4.0E-03 r	0	0.10	106-47-8	4-Chloroaniline	2.6E+02 nc	2.7E+03 nc	1.5E+01 nc	1.5E+02 nc	7.0E-01	3.0E-02
1	2.0E-02 I		5.7E-03 h	1	0.10	108-90-7	Chlorobenzene	6.5E+01 nc	2.2E+02 nc	2.1E+01 nc	3.9E+01 nc	1.0E+00	7.0E-02
2.7E-01 h	2.0E-02 I	2.7E-01 h	2.0E-02 r	0	0.10	510-15-6	Chlorobenzilate	1.6E+00 ca	7.1E+00 ca	2.5E-02 ca	2.5E-01 ca		
	2.0E-01 h		2.0E-01 r	0	0.10	74-11-3	p-Chlorobenzoic acid	1.3E+04 nc	1.0E+05 max	7.3E+02 nc	7.3E+03 nc		
1	2.0E-02 h		2.0E-02 r	0	0.10	98-56-6	4-Chlorobenzotrifluoride	1.3E+03 nc	1.4E+04 nc	7.3E+01 nc	7.3E+02 nc		
	2 0E-02 h		2.0E-03 h	1	0.10	126-99-8	2-Chloro-1,3-butadlene	3.6E+00 nc	1.2E+01 no	7.3E+00 nc	1.4E+01 nc		
	4 0E-01 h		4.0E-01 r	1	0.10	109-09-3	1-Chlorobutane	4.8E+02 sat	4.8E+02 sat	1.5E+03 nc	2.4E+03 nc		
	1.4E+01 r		1.4E+01 l	1	0.10	75-68-3	1-Chloro-1,1-difluoroethane (HCFC-142b)	3.4E+02 sat	3.4E+02 sat	5.2E+04 na	8.7E+04 nc		
				1	0.10	110-75-8	2-Chloroethyl vinyl ether						
	1.4E+01 r		1.4E+01 i	1	0.10	75-45-8	Chlorodifluoromethane	3.4E+02 sat	3.4E+02 sat	5.1E+04 nc	8.5E+04 nc		
8.1E-03 I	1.0E-02	8.1E-02 I	1.0E-02 r	1	0,10	67-66-3	Chloroform	2.5E-01 ca	5.3E-01 📾	8.4E-02 ca	1.6E-01 🖬	6.0E-01	3.0E-02
1.3E-02 h		6.3E-03 h		1	0.10	74-87-3	Chloromethane	1.2E+00 ca	2.6E+00 ca	<u>1.1E+00 ca</u>	1.5E+00 ca		
5.8E-01 h		5.8E-01 r		0	0.10	95-69-2	4-Chloro-2-methylaniline	7.7E-01 ca	3.3E+00 ca	1.2E-02 ca	1.2E-01 🚥		
4.6E-01 h		4.6E-01 r		0	0.10	3105-93-3	4-Chloro-2-methylaniline hydrochloride	9.7E-01 ca	4.1E+00 ca	1.5E-02 ca	1.5E-01 ca		
	8 0E-02 I		1 20-30 B	1	0.10	91-58-7	beta-Chloronaphthalene	1.1E+02 sat	1.1E+02 sat	2.9E+02 nc	4.9E+02 nc		
2.5E-02 h		2.5E-02 r	r	0	0.10	88-73-3	o-Chloronitrobenzene	1.8E+01 ca	7.6E+01 ca	2.7E-01 ca	2.7E+00 🛤		
1.8E-02 h		1.8E-02 r	ſ	0	0.10	100-00-5	p-Chloronitrobenzene	2.5E+01 cm	1.1E+02 ca	3.7E-01 🛤	3.7E+00 📾		
l	5 0E-03 I		5.0E-03 r	1	0.10	95-57-8	2-Chlorophenol	9.1E+01 nc	<u>3.7E+02 nc</u>	<u>1.8E+01 nc</u>	<u>3.8E+01 nc</u>	4.0E+00	<u>2.0E-01</u>
1	2.9E-02 r		2.9E-02 h	i 1	0.10	75-29-6	2-Chloropropane	1.7E+02 nc	5.8E+02 nc	1.0E+02 nc	1.7E+02 nc		
1.1E-02 h	1.5E-02 I	1.1E-02 r	1.5E-02 r	0	0.10	1897-45-6	Chlorothalonil	4.0E+01 ca**	1.7E+02 ca•	6.1E-01 cm	6.1E+00 😅		
	2.0E-02 I		2.0E-02 r	1	0.10	95-49-8	o-Chlorotoluene	1.6E+02 nc	5.5E+02 nc	7.3E+01 m	1.2E+02 nc	1	
	2.0E-011		2.0E-01 r	0	0.10	101-21-3	Chlorpropham	1.3E+04 nc	1.0E+05 max	7.3E+02 nc	7.3E+03 nc		
	3.0E-03 I		3.0E-03 r	0	0.10	2921-88-2	Chlorpyrifos	2.0E+02 nc	2.0E+03 nc	1.1E+01 nc	1.1E+02 no		
	1.0E-02 h		1.0E-02 r	0	0.10	6598-13-0	Chlorpyrifos-methyl	6.5E+02 nc	6.8E+03 no	3.7E+01 nc	3.7E+02 nc		
	5.0E-02 I		5.0E-02 r	0	0.10	64902-72-3	Chlorsulfuron	3.3E+03 nc	3.4E+04 nc	1.8E+02 nc	1.8E+03 na		
	8 0E-04 h		8.0E-04 r	0	0.10	602-38-58-4	Chlorthiophos	5.2E+01 nc	5.5E+02 nc	2.9E+00 nc	2.9E+01 nc		
		4.2E+011		0	0.01	<u>n/a</u>	Total Chromium (1/6 ratio Cr VI/Cr III)	2.1E+02 ca	4.5E+02 ca	1.6E-04 ca		3.8E+01	2.0E+00
1	5.0E-03 I	2.9E+02 I		0	0.01	7440-47-3	Chromium VI	3.0E+01 ca	6.4E+01 ca	2.3E-05 ca	1.8E+02 nc	3.8E+01	2.0E+00
							CAL-Modified PRG" (PEA, 1994)	2.0E-01			1.6E-01		
	6.0E-02 n		2.9E-04 m	<u> 0</u>	0.01	7440-48-4	Cobalt	4.6E+03 nc	9.7E+04 nc	1.0E+00 nc	2.2E+03 m		
1		2.2E+00 i		. 0	0.01	8007-45-2	Coke Oven Emissions		0.05	3.1E-03 ca	4 412 - 44		
	3.7E-02 h			٥	0.01	7440-50-8	Copper and compounds	2.8E+03 nc	0.3E+04 nc	0 FT 00	1.4±+03 nc	1	
1.9E+00 h	1.0E-02 x	1.9E+00 x	1.0E-02 r	<u>r 1</u>	0.10	123-73-9	Urotonaldenyde	0.3E-03 ca	1.1E-02 ca	3.5E-03 cs	0.9E-03 ca		· · · · · · · · · · · · · · · · · · ·
	4.0E-02 I		2.6E-03 I	h 1	0.10	98-82-8	Cumene	1.9E+01 nc	0.2E+01 nc	9.4E+00 m	1.9E+01 nc	1	
8.4E-01 h	2.0E-03 h	8.4E-01 r	2.0E-03 r	r 0	0.10	21725-46-2	Cyanazine	0.3E-01 ca*	2.3E+00 ca	8.0E-03 ca	8.0E-02 ca	1	

.

Key: HIRIS h-HEAST n=NCEA x=WITHDRAWN r=ROUTE EXTRAPOLATION ca=CANCER PRG nc=NONCANCER PRG sat=50il SATURATION max=CEILING LIMIT *(where: nc < 100X ca) **(where: nc < 10X ca)

	A		ad abates at here			EXTRAPOLATION CHICANCER PRG ASINONCANCER PRG A	BOIL BATUKATION	max=CEILING LIMP	<u>f *(where: nc < 100X</u>	<u>ca)</u> [∞] (where: nc < 10	IX ca)	· · · · · · · · · · · · · · · · · · ·
1.4.15				1.5411	1 J. N. J. P.		DIID	DOCE			en al la factor y c	it il white
	이 아이 아이	n 1 de jan	a a guid f ui			IUNFLANNINU	FUR	FUJE				
[14] 문우]] (/4).							
	UNICITY	NFURMA		n lin in		CONTAMINANT SILLING	PRELIM	NARY REMED	DIAL GOALS (F	PRGs)	SOIL SCREEN	ING LEVELS
SFo	RíDo	SFI	RIDI	O aba.	CAS No	linderen konstruktion verste anderen die	in Pality and the second	i i i i ili ili ili ili ili ili ili ili		Hard and the state of the	Migration to Gr	ound Water
1/(mg/kg-d)	(mg/kg-d)	1/(mg/kg-d)	(mg/kg-d)	C solls	i de la caración de l		Soli (ma/ka)		Ampient Ar	Tap Water	DAF 20	DAF 1
	1 0E-02 h		1.0E-02 r	1 0.10	158-59-2	1,2-Dichloroethylene (cis)	3.1E+01 nc	1.0E+02 nc	3.7E+01 nc	6.1E+01 cc	1 4 0F-01	2 OF-02
	2 06-02 1		2.0E-02 r	1 0.10	150-60-5	1,2-Dichloroethylene (trans)	7.8E+01 nc	2.7E+02 nc	7.3E+01 nc	1.2E+02 nc	7.0E-01	3 0E-02
	9 0E-03 h		9.0E-03 r	1 0.10	540-59-0	1,2-Dichloroethylene (mixture)	3.5E+01 nc	1.2E+02 nc	3.3E+01 nc	5.5E+01 nc		0.01 01
1	3.0E-03		3.0E-03 r	0 0.10	120-83-2	2,4-Dichlorophenol	2.0E+02 nc	2.0E+03 nc	1.1E+01 nc	1.1E+02 pc	1.0E+00	5 0F-02
	8 OE-03 I		8.0E-03 r	0 0.10	94-82-6	4-(2,4-Dichlorophenoxy)butyric Acid (2,4-DB)	5.2E+02 nc	5.5E+03 nc	2.9E+01 nc	2.9E+02 nc		0.01 01
	1.0E-02 I		1.0E-02 r	0 0.10	94-75-7	2,4-Dichlorophenoxyacetic Acid (2,4-D)	6.5E+02 nc	6.8E+03 nc	3.7E+01 nc	3.7E+02 nc		
6.8E-02 h	1.1E-03 r	8 8E-02 r	1.1E-03 I	1 0.10	78-87-5	1,2-Dichloropropane	3.1E-01 ca*	6.8E-01 ca*	9.9E-02 ca*	1.6E-01 ca*	3.0E-02	1.0E-03
1.82-01 h	3.0E-04 (1.3E-01 h	5.7E-03 I	1 0.10	542-75-6	1,3-Dichloropropene	2.5E-01 ca*	5.5E-01 ca	5.2E-02 ca	8.1E-02 ca	4.0E-03	2.0E-04
	3.0E-03 I		3.0E-03 r	0 0.10	616-23-9	2,3-Dichloropropanol	2.0E+02 nc	2.0E+03 nc	1.1E+01 nc	1.1E+02 nc		,
2.92-011	5.0E-04 I	2.9E-01 r	1.4E-04 i	0 0.10	62-73-7	Dichlorvos	1.5E+00 ca**	6.6E+00 ca*	2.3E-02 ca*	2.3E-01 ca*		
4.4E-01 x		4.4E-01 r		0 0.10	115-32-2	Dicofol	1.0E+00 ca	4.3E+00 🛥	1.5E-02 🛥	1.5E-01 ca		
	3 0E-02 h		5.7E-05 h	1 0.10	77-73-8	Dicyclopentadiene			2.1E-01 nc	4.2E-01 nc		
1.82+011	5.0E-05 I	1.6E+01 I	5.0E-05 r	0 0.10	60-57-1	Dieldrin	2.8E-02 ca*	1.2E-01 ca	4.2E-04 ca	4.2E-03 ca	4.0E-03	2.0E-04
	5.7E-03 h		5.7E-03 x	0 0.10	112-34-5	Diethylene glycol, monobutyl ether	3.7E+02 nc	3.9E+03 nc	2.1E+01 nc	2.1E+02 nc		
·	2.0E+00 h		2.0E+00 r	0 0.10	111-90-0	Diethylene glycol, monoethyl ether	1.0E+05 max	1.0E+05 max	7.3E+03 nc	7.3E+04 nc		
	1.1E-02 h		1.1E-02 r	0 0.10	617-84-5	Diethylformamide	7.2E+02 nc	7.5E+03 ng	4.0E+01 nc	4.0E+02 nc		·····
1.2E-03 I	6 OE-01 I	1.2E-03 r	6.0E-01 r	0 0.10	103-23-1	Di(2-ethylhexyl)adipate	3.7E+02 nc	1.6E+03 nc	5.6E+00 nc	5.6E+01 nc		
	8 0E-01 I		8.0E-01 r	0 0.10	84-66-2	Diethyl phthalate	5.2E+04 nc	1.0E+05 max	2.9E+03 nc	2.9E+04 nc		
4.7E+03 h		4.7E+03 r		0 0.10	50-53-1	Diethylstilbestrol	9.5E-05 ca	4.1E-04 ca	1.4E-06 ca	1.4E-05 ca		
	6 OE-02 I		8.0E-02 r	0 0.10	43222-48-6	Difenzoquat (Avenge)	5.2E+03 nc	5.5E+04 nc	2.9E+02 nc	2.9E+03 nc		
}	2.0E-021		2.0E-02 r	0 0.10	35367-38-5	Diflubenzuron	1.3E+03 nc	1.4E+04 nc	7.3E+01 no	7.3E+02 nc		
	1.1E+01 r		1.1E+01 i	1 0.10	75-37-8	1,1-Difluoroethane			4.2E+04 nc	6.9E+04 nc		
	8.0E-02 (8.0E-02 r	0 0.10	1445-75-8	Disopropyl methylphosphonate	5.2E+03 nc	5.5E+04 nc	2.9E+02 nc	2.9E+03 nc		
	2.0E-021		2.0E-02 r	0 0.10	55290-64-7	Dimethipin	1.3E+03 nc	1.4E+04 nc	7.3E+01 nc	7.3E+02 nc	ł	
	2.02-041		2.0E-04 r	0 0.10	80-51-5	Dimethoate	1.3E+01 nc	1.4E+02 nc	7.3E-01 nc	7.3E+00 nc		
1.4E-02 h		1.4E-02 r		0 0.10	119-90-4	3,3'-Dimethoxybenzidine	3.2E+01 🖬	1.4E+02 ca	4.8E-01 ca	4.8E+00 ca		
	5.7E-06 r		5.7E-06 x	1 0.10	124-40-3	Dimethylamine	6.5E-02 nc	2.4E-01 nc	2.1E-02 nc	3.5E-02 nc		
745.04 1	2.06-031		2.0E-03 r	0 0.10	121-69-7	N-N-Dimethylaniline	1.3E+02 nc	1.4E+03 nc	7.3E+00 nc	7.3E+01 nc		
FAC OF N		7.32-017		0 0.10	95-68-1	2,4-Dimethylaniline	5.9E-01 ca	2.5E+00 ca	9.0E-03 ca	9.0E-02 ca		
0.00-01 h	······	5.8E-01 r	·····	0 0.10	21438-98-4	2.4-Dimethylaniline hydrochloride	<u>7.7E-01 ca</u>	3.3E+00 cm	1.2E-02 ca	1.2E-01 ca		
2.4E+00 h		9.2E+00 r		0 0.10	119-93-7	3,3'-Dimethylbenzidine	4.8E-02 ca	2.1E-01 ca	7.3E-04 ca	7.3E-03 ca		
2.82400 1		3.5E+00 x		0 0.10	67-14-7	1,1-Dimethylhydrazine	1.7E-01 ca	7.3E-01 ca	1.9E-03 ca	2.6E-02 ca	1	
3.7E+01 X		3.7E+01 X		0 0.10	540-73-8	1,2-Dimethylnydrazine	<u>1.2E-02 ca</u>	5.2E-02 ca	1.8E-04 ca	1.8E-03 ca		
	1.0E-01 h		8 6E-03 I	0 0.10	68-12-2	N.N-Dimethylformamide	6.5E+03 nc	6.8E+04 nc	3.1E+01 nc	3.7E+03 nc		
	2.02-021		2.02-02 1	0 0.10	105-67-9	2,4-Dimethylphenol	1.3E+03 nc	1.4E+04 nc	7.3E+01 nc	7.3E+02 nc	9.0E+00	4.0E-01
	0.02-041		8.0E-04 r	0 0.10	570-20-1	2.6-Dimethylphenol	3.9E+01 nc	4.1E+02 ng	2.2E+00 nc	2.2E+01 nc		
1	1.02-031		1.0E-03 r	0 0.10	95-65-8	3,4-Dimethylphenol	6.5E+01 ns	6.8E+02 nc	3.7E+00 no	3.7E+01 nc		
1	1.00+01 h		1.02+01 7	0 0.10	131-11-3		1.0E+05 max	1.0E+05 max	3.7E+04 no	3.7E+05 nc	ĺ	
<u> </u>	2.05.02.1		1.02-017	0 0.10	120-01-0	A 6 Diritico o unicipational de	6.5E+03 nc	6.8E+04 nc	3.7E+02 nc	3.7E+03 nc		
1	4.05.041		1 05 04 -	0 0.10	131-68-5	r.o-Dinino-o-cyclonexyl phenol	1.3E+02 nc	1.4E+03 nc	7.3E+00 nc	7.3E+01 nc		
l .			1.UE-04 F	0 0.10	W#-03-0	1.3 Disitrahannana	0.55+00 nc	6.8E+01 nc	3.7E-01 nc	3.7E+00 nc		
	405.04 5		4.05.04 -	0 0.10	320-29-0		2.0E+01 nc	2.7E+02 nc	1.5E+00 nc	1.5E+01 nc		
1	3.05.034		1.05-01	0 0.10	100-20-4	2 A Dinitronhonol	2.0E+01 nc	2.7E+02 nc	1.5E+00 nc	1.5E+01 nc		
6.65-01	2.00-031	8 85 A1 -	2.02-03 f	0 0.10	81-20-0 05004 44 5	pinitrototuono mixture	1.3E+02 nc	1.4E+03 nc	7.3E+00 nc	7.3E+01 nc	3.0E-01	1.0E-02
0.00-011	205.031	0.00-011	3.05.03 -	0 0.10	20321-14-6		0.5E-01 ca	2.8E+00 cm	9.9E-03 ca	9.9E-02 ca	8.0E-04	4.0E-05
	10501		4.0E-03 f	0 0.10	121-14-2	2.4-Dinitrotoluene (also see Dinitrotoluene mixture)	1.3E+02 nc	1.4E+03 nc	7.3E+00 nc	7.3E+01 nc	8.0E-04	4.0E-05
I	1.02-03 N		1.02-03 f	U 0.10	000-20-2	12,0-Dinirololuene (also see Dinitrololuene mixture)	0.51:+01 nc	6.8E+02 nc	3.7E+00 nc	3.7E+01 nc	7.0E-04	3.0E-05

•

.

Key: HIRIS MAHEAST MANCEA WITHORAWA MOUTE EXTRAPOLATION GA-CANCER PRG MONCANCER PRG AND SOIL BATURATION MAX-CEILING LIMIT "(MINOR: DC 4 100X CR) "(MINOR: DC 4 1

TOXICITY INFORMATION Construction of the second matrix information of the second matrix info		944 (19 <u>1</u> 4	可是这 遭				FURFLANNING						
1:3: 17 DUNCITY INFORMATION 1: 12: 12: 12: 12: 12: 12: 12: 12: 12:				inera s				CORDENIN	LADY DEMENIA	I GOALS (PE	363)	SOIL SCREENING	S LEVELS
Bits Data Display Display <thdisplay< th=""> <thdisplay< th=""> <thdisplay< td=""><td>TO</td><td>XICITY IN</td><td>FORMAT</td><td></td><td></td><td></td><td>CONTAMINANT IN CONTAMINANT</td><td>PRELIMI</td><td>WART REMEDIA</td><td></td><td></td><td>Migration to Ground</td><td>d Water</td></thdisplay<></thdisplay<></thdisplay<>	TO	XICITY IN	FORMAT				CONTAMINANT IN CONTAMINANT	PRELIMI	WART REMEDIA			Migration to Ground	d Water
Bits Construction Construction <thconstruction< th=""> Construction</thconstruction<>	n na na na in Talana	ana in Si	t and the state		V skin 💡	CAG NO		Residential 1	Industrial	Ambient Alr	Tap Water 5	DAF 20	DAF 1
Listent Listent 0 to 1 Prime 20 Constraint Constraint Constraint Constraint Constraint Constraint 18001 18000 18001 18000 18001 18000 18001 18000 18001 18000 18001 18000 18001 18000 18000 18000 18000 18000 18000	SFo //mo/ka-d) /n	RIDo no/ko-ri\ 1	Imaka-da'	(ma/ka-d) = (Cisolis ir			Soll (mg/kg)	Soll (mg/kg)	2 5 5 02	3 55 01		
Discrit 206301 Conto 42224 Fondos 1.46:01 m	1 05 01 1	WWW W	1 9F-01 /		0 0.10	72178-02-0	Fomesafen	2.3E+00 ca	1.0E+01 ca	3.52-02 4	7 35401 -		
132-01 142-01 0.00 2.00 0.00	1.02-011	2 05-03 1		2 0F-03 r	0 0.10	944-22-9	Fonofos	1.3E+02 nc	1.4E+03 nc	7.3E+00 M	7,5E+01 nc		
126:001 26:007 26:007 26:00 0.00		4 65 01 1	4 6E-02 1	1.00 000	0 0.10	50-00-0	Formaldehyde	9.8E+03 nc	1.0E+05 mc	1.02-01 6	7 2 5 + 0.1		
138:00 138:00 0 <th< td=""><td></td><td>1.5E-011</td><td>4 OL OL</td><td>2 0F+00 r</td><td>0 0.10</td><td>64-18-6</td><td>Formic Acid</td><td>1.0E+05 max</td><td>1.0E+05 max</td><td>1.3E+U3 AG</td><td>1 15+05</td><td></td><td></td></th<>		1.5E-011	4 OL OL	2 0F+00 r	0 0.10	64-18-6	Formic Acid	1.0E+05 max	1.0E+05 max	1.3E+U3 AG	1 15+05		
Jack of				3.05+00 r	0 0 10	39148-24-8	Fosetyl-al	1.0E+05 max	1.0E+05 max	1.1E+04 m	1,1E+05 nc		
38:000 bit 00000000000000000000000000000000		4 05 03 1		105-03 r	1 0 10	110-00-9	Furan	2.5E+00 nc	8.5E+00 na	3.7E+00 mc	1 95 02 m		
Jacoba Jacoba Furfural D200402 D200402 Col: 4000 Col: 4000 <thcol: 40000<="" th=""> Col: 40000 <thcol: 4000<="" td=""><td>2 45.00 b</td><td>1.02-03 1</td><td>3.85+00 /</td><td>1.02-031</td><td>0 0.10</td><td>87-45-8</td><td>Furazolidone</td><td>1.2E-01 ca</td><td>5.0E-01 ca</td><td>1.8E-03 ca</td><td>1.05-02 0</td><td></td><td></td></thcol:></thcol:>	2 45.00 b	1.02-03 1	3.85+00 /	1.02-031	0 0.10	87-45-8	Furazolidone	1.2E-01 ca	5.0E-01 ca	1.8E-03 ca	1.05-02 0		
Second Second<	3.8E+00 h	3 05 03 1	3.027001	1.4E-02 h	0 0 10	98-01-1	Furfural	2.0E+02 nc	2,0E+03 nc	5.2E+01 m	1,12+02 16		
Statum Discust 0.10 Discust 0.10 Discust Discust <thdiscust< th=""> <thdiscust< th=""> <thdiscust<< td=""><td>5 65 64 b</td><td>3.02-031</td><td>6 0E+01 4</td><td>1.42-04 10</td><td>0 0 10</td><td>531-82-8</td><td>Furlum</td><td>8.9E-03 ca</td><td>3.8E-02 ca</td><td>1.3E-04 ca</td><td>2 25+00</td><td></td><td></td></thdiscust<<></thdiscust<></thdiscust<>	5 65 64 b	3.02-031	6 0E+01 4	1.42-04 10	0 0 10	531-82-8	Furlum	8.9E-03 ca	3.8E-02 ca	1.3E-04 ca	2 25+00		
30E-071 40E-041 40E-041 40E-041 40E-041 60E-041 10E-010 10E-010 <t< td=""><td>5.0E+01 h</td><td></td><td>3.05.02 4</td><td></td><td>0 0 10</td><td>60566-05-0</td><td>Furmecyclox</td><td>1.5E+01 ca</td><td>6.4E+01 ca</td><td>2.2E-01 ca</td><td>2.2ETUU C</td><td></td><td></td></t<>	5.0E+01 h		3.05.02 4		0 0 10	60566-05-0	Furmecyclox	1.5E+01 ca	6.4E+01 ca	2.2E-01 ca	2.2ETUU C		
440:01 248:01 0 = 00 755:44 Olycholateleyde 2.65:103 mc 6.65:103 mc 6.16:10 mc 1.16:10 mc	3.02-021		3.05-02 1	4 05 04 4	0 0 10	77182-82-2	Glufosinate-ammonium	2.6E+01 nc	2.7E+02 no	1.52+00 m	1.50-01 16	1	
10001 10001 10001 0.0000 0.000 0.000 <t< td=""><td></td><td>4.08-041</td><td></td><td>4.0E-04 I</td><td>0 0.10</td><td>765-34-4</td><td>Glycidaldehyde</td><td>2.6E+01 nc</td><td>2.7E+02 nc</td><td>1.0E+00 mc</td><td>1.5E+U1 m</td><td></td><td></td></t<>		4.08-041		4.0E-04 I	0 0.10	765-34-4	Glycidaldehyde	2.6E+01 nc	2.7E+02 nc	1.0E+00 mc	1.5E+U1 m		
Liberol Liberol Laboratory SLEF402 mc SLEF402 mc Laboratory Laboratory SLEF402 mc Laboratory Laboratory <td></td> <td>4.0E-041</td> <td></td> <td>1.05.01 4</td> <td>0 0.10</td> <td>1071-83-6</td> <td>Givohosale</td> <td>6.5E+03 nc</td> <td>6.8E+04 nc</td> <td>3.7E+02 nc</td> <td>3./E+U3 m</td> <td></td> <td></td>		4.0E-041		1.05.01 4	0 0.10	1071-83-6	Givohosale	6.5E+03 nc	6.8E+04 nc	3.7E+02 nc	3./E+U3 m		
366-031 200-03 0 0 100 127272-8 Harmony 6.65-02 cs 6.95+03 nc 4./2+01 nc 4./2+03 cs 1.65-02 cs		1.02-011		LUE-OT I	0 0 10	49404-40-2	Haloxyfop-methyl	3.3E+00 nc	3.4E+01 no	1.8E-01 m	1.02+00 m		
136:071 136:071 0.00 1207.07 0.00 1707.07 1207.07 <		5.0E-05 I		5.02-05 1	0 0.10	70277-27.1	Harmony	8.5E+02 nc	8.9E+03 nc	4.7E+01 nc	4./E+UZ nc	0.25.01	1 05+00
42E+001 15E+01 15E+02 15E+01 15E+02 15E+01 15E+02 15E+01 15E+02		1.3E-02 I		1.38-02 1	0 0.10	76.44.8	Hentachlor	9.9E-02 ca	4.2E-01 ca	1.5E-03 ca	1.5E-02 ca	2.35401	3 05.02
a [E:601 1.3E:601 0.100 1.3E:601 0.100 1.3E:601 0.100 1.3E:601 0.100 1.3E:601 0.100 1.8E:601 1.1E:002 a.0E:01 ac 1.1E:002 ac 3.0E:01 ac 1.1E:002 ac 3.0E:03 ac 3.7E:02 ac 3.0E:03 5.0E:04 3.0E:03 5.0E:04 ac 3.0E:03 5.0E:02 ac 3.0E:03 5.0E:02 ac 3.0E:03 1.0E:02 ac 3.0E:03 ac 3.7E:02 ac 3.0E:03 5.0E:04 ac 1.1E:02 ac 3.0E:03 ac 3.7E:02 ac 3.0E:03 5.0E:04 ac 1.1E:02 ac 3.0E:03 ac 3.7E:02 ac 3.0E:03 ac <	4.5E+00 I	5.0E-04 I	4.6E+00 I	5.0E-04 F	0 0.10	1004 67 3	Hentachlor epoxide	4.9E-02 a	2.1E-01 a•	7.4E-04 ca*	7.4E-03 ca*	7.02-01	3.00-02
1 202031 20201 0 010 278-01 20201 </td <td>9.1E+00 I</td> <td>1.3E-05 I</td> <td>9.1E+00 I</td> <td>1.3E-05 F</td> <td>0 0.10</td> <td>1024-37-3</td> <td>Hexabromobenzene</td> <td>1.3E+02 nc</td> <td>1.4E+03 ns</td> <td>7.3E+00 nc</td> <td>7.3E+01 nc</td> <td>0.05.00</td> <td>1 05 01</td>	9.1E+00 I	1.3E-05 I	9.1E+00 I	1.3E-05 F	0 0.10	1024-37-3	Hexabromobenzene	1.3E+02 nc	1.4E+03 ns	7.3E+00 nc	7.3E+01 nc	0.05.00	1 05 01
16E-001 06E-041 76E-01 0.010 774-03 1.0E-01 1.0E-02 0.010 3.0E-01 1.1E-02 0.010 3.0E-01 0.010 3.0E-01 <td></td> <td>2.0E-03 I</td> <td></td> <td>2.0E-03 f</td> <td>0 0.10</td> <td>418 74 1</td> <td>Hexachlorobenzene</td> <td>2.8E-01 ca*</td> <td>1.2E+00 ca</td> <td>4.2E-03 ca</td> <td>4.2E-02 ca</td> <td>2.0E+00</td> <td>1.02-01</td>		2.0E-03 I		2.0E-03 f	0 0.10	418 74 1	Hexachlorobenzene	2.8E-01 ca*	1.2E+00 ca	4.2E-03 ca	4.2E-02 ca	2.0E+00	1.02-01
7 166:01 2:06:01 0:00 3:06:01 0:00 3:06:01 0:00 3:06:01 0:00 3:06:01 0:00 3:06:01 0:00 3:06:01 0:00 3:06:01 0:00 3:06:01 0:00 3:06:01 0:00 3:06:01 0:00 3:06:01 0:00 3:06:01 0:00 3:06:01 0:00 3:06:01 0:00 3:06:01 0:00 3:06:01 0:00 3:06:01 0:00 0:00 3:06:01 0:00 0:00 3:06:01 0:00	1.6E+00 i	8.0E-04 I	1.8E+00 I	8.0E-04 r	0 0.10	110-74-1	Hexachlorobutadiene	5.7E+00 ca**	2.4E+01 cm	8.7E-02 😅	8.6E-01 ca*	2.01+00	1.00-01
0 12 E:001 0 21:001 0 101 21:001 0 101 21:001 0 101 21:000 3.7E-02 call 7.7E-01 call 7.7E-01 call 7.7E-01 call 7.7E-01 call 7.7E-01 call 7.7E-02 call 7.7E-02 call 7.7E-02 call 7.7E-02 call 7.7E-01 7.7E-01 7.7E-01 7.7E-01	7.8E-02 I	2.0E-04 h	7.7E-02 I	2.0E-04 r	0 0.10	87-00-3	HCH (sinha)	7.1E-02 a	3.0E-01 ca	1.1E-03 ca	1.1E-02 a	5.01-04	3.02-00
118:001 146:001 0 0.00 3148:07 1011(0017) 3.4E-01 cs 1.5E+00 cs 5.2E-02 cs 3.0E-03 3.0E-03 3.0E-03 3.0E-03 3.0E-03 3.0E-03 3.0E-02 cs 3.0E-03 3.0E-02 cs 3.0E-01 cs 3.0E-0	6.3E+00 I		8.3E+00 I		0 0.10	319-84-0		2.5E-01 ca	1.1E+00 ca	3.7E-03 ca	3.7E-02 ca	3.0E-03	1.0E-04
13E-001 30E-347 0 0.10 63849 FICH (gaining) (include) 2.5E-01 a 1.1E+00 ca 3.8E-03 ca 3.7E-02 ca 3.0E-03 1.1E+00 ca 3.8E-03 ca 3.7E-02 ca 3.0E-03 ta 3.8E-03 ca 3.7E-02 ca 6.8E+03 ca da	1.8E+001		1.8E+00 i		0 0.10	319-85-7	HCH (camma) Lindana	3.4E-01 ca*	1.5E+00 ca	5.2E-03 ca	5.2E-02 ca	9.0E-03	5.02-04
118E-001 18E-001 0.010 84.98 Incrnetional intermetion of the section of the sectin of the section of the section of the secti	1.3E+00 h	3.0E-04 I	1.3E+00 r	3.0E-04 r	0 0.10	58-89-9		2.5E-01 ca	1.1E+00 cm	3.8E-03 ca	3.7E-02 ca	3.0E-03	1.0E-04
7.0E31 2.0E-65 h 0 0.0 77-47-4 Free Addition Color (1) inter (1) and (1) a	1.8E+00 i		1.8E+00 I		0 0.10	58-89-9	Howen bloss and an antadiana	4 5E+02 m	4.6E+03 nc	7.3E-02 m	2.6E+02 nc	4.0E+02	2.0E+01
12 ± 0:01 46 ± 001 0 0.00 10 ± 000 11 ± 400<		7.0E-03 I		2.0E-05 h	0 0.10	77-47-4		7 2E-05 ca	3.1E-04 ca	1.5E-06 ca	1.1E-05 ca		
14E021 10E031 14E021 10E031 14E021 10E031 0.10 67.7-1 Instantanta 2.0E+101 nc 2.0E+02 nc 1.1E+00 nc 1.1E+01 nc 30E041 30E041 30E041 0.010 121-824 Hexachlorophene 4.0E+00 nc 1.1E+01 nc 1.1E+02 nc 6.1E-01 nc 1.0E-01 nc 20E001 20E01 0.010 1022004 1.6-HexaneHylene dilsocyanate 1.1E+02 nc 1.1E+02 nc 3.5E+02 nc 1.0E-01 nc 316021 316021 0.10 105420-2 1.0E+02 nc 1.1E+02 nc 3.5E+02 nc 2.2E+03 nc 2.2E+04 nc 1.2E+02 nc 3.5E+02 nc 30E031 28E041 0.10 1042-14 Hydrogen chloride Hydrogen chloride 1.0E+01 nc 1.0E+00 nc 2.0E+02 nc 1.2E+03 nc 2.2E+03 nc 2.2E+04 nc 1.2E+02 nc 1.2E+03 nc 30E031 28E041 0.10 1047414 Hydrogen sulfide 1.0E+01 nc	6.2E+03 i		4.6E+03 I		0 0.10	19408-74-	Hexachiorodibenzo-p-dibut hixture (riccob)	3 2E+01 cm	1.4E+02 ca**	4.8E-01 ca*	• 4.8E+00 cat	• <u>5.0E-01</u>	2.0E-02
3 DE 041 3 DE 041 3 DE 041 7 0.504 Plex atchinicipation 1.1E-011 3 DE 041 3 DE 041 0 0.10 121-824 Hexathinicon1, 3,5-trinitro-1, 3,5-t	1.4E-02 I	1.0E-03 I	1.4E-02	1.0E-03 r	0 0.10	67-72-1	Hexachioroelitane	2 0E+01 m	2.0E+02 m	1.1E+00 nc	1.1E+01 nc		
1.1E011 3.0E-031 1.1E01r 3.0E-031 0.10 121-224 Hexanyon-1, 5, 5-timino-1, 5, 5-t		3 0E-04 I		3.0E-04 r	0 0.10) 70-30-4	Hexachiorophene	4 0E+00 cm	1.7E+01 ca	6.1E-02 ca	6.1E-01 🚥		
2 0E-06 r 2 0E-06 r 2 0E-06 r 2 0E-02 h 5.7E-021 h 10.01 105-7E 324 methylene disocyanists 1.1E+02 sat 2.1E+02 nc 3.5E+02 nc 3.5E+02 nc 3 3E-021 3 3E-021 0.00 50E-001 1.7E+011 0.010 5023-042 Hydrazine, hydrazine sulfate 1.5E-01 ca 6.4E-01 ca 3.9E-04 ca 2.2E+03 nc 2.2E+04 nc 1.2E+02 nc 1.2E+03 nc 2.2E-02 ca 1.0E+00 nc 2.2E+03 nc 2.1E+01 nc 1.0E+00 nc 2.1E+01 nc 1.0E+00 nc 2.1E+01 nc 1.0E+00 nc 2.1E+03 nc 1.0E+03 nc 1.1E+03 nc 1.0E+03 nc 1.0E+03 nc 1.0E+03 nc 1.0E+03 nc 1.1E+03 nc 1.0E+03 nc 1.0E+03 nc 1.0E+03 nc 1.0E+03 nc 1.0E+03 nc 1.0E+03 nc 1.0E+	1.1E-01 I	3.0E-03 I	1.1E-01 r	3.0E-03 r	0 0.10) 121-82-4		4.02.00		1.0E-02 m	1.0E-01 nc		
B 0E-02 h 3 3E-021 5.7E 021 1 0.01 110-54-3 5125-04-2 n-Hexalne 1.1E-03 are 1.2E+03 are		2.9E-06 r		2.9E-00 i	0 0.10	822-06-0	1,6-Hexamethylene diisocyanate	1 15+02	1 1E+02 sat	2.1E+02 nc	3.5E+02 m		
3.3E-021 3.3E-02 r 0 0.10 5125-04-2 by draging hydraging hydragin		8.0E-02 h		5.7E-02 i	1 0.10	110-54-3	n-Hexane	2 25+03	2 2E+04 m	1.2E+02 m	1.2E+03 nc		
3 0E+001 1.7E+011 0 0.10 302-01-2 Hydrazine, hydrazine suitate 1.5E+01 ta 0.4E 01 ta		3.3E-02 i		3.3E-02 r	0 0.10	51235-04	Hexazinone	1 65.01	64E-01 ca	3.9E-04 m	2.2E-02 ca		
5.7E-031 0 0.10 7647-01-0 Hydrogen chloride 1.0E+00 nc 2.0E+00 nc 3.0E-031 2.9E-041 1 0.10 7783-064 Hydrogen sulfide 1.0E+00 nc 2.0E+03 nc 1.5E+03 nc 1.5E+03 nc 4.0E-02 h 4.0E+02 r 0 0.10 3555444-0 Imazalil 8.5E+02 nc 8.9E+03 nc 4.7E+01 nc 4.7E+02 nc 1.3E+02 i 1.3E+02 r 0 0.10 3555444-0 Imazalil 8.5E+02 nc 8.9E+03 nc 4.7E+01 nc 4.7E+02 nc 2.5E+01 i 2.5E+01 r 0 0.10 355344-0 Imazalil 1.6E+04 nc 1.0E+05 max 9.1E+03 nc 3.0E-011 2.0E+01 r 0 10 3374-19-7 Iprodione 1.1E+04 nc 1.0E+05 max 1.1E+03 nc 1.8E+03 nc 3.0E-011 3.0E=01 r 0 10 78-59-1 Isophorone 9.8E+02 nc 1.0E+04 nc 5.5E+01 nc 5.5E+02 nc 1.5E+021 1.5E+02 r 0 0.10 33820-53-0 Isophorone 9.8E+02 nc 1.0E+04 nc 1.8E+03 nc 1.5E+021 r 0.50E-02 r 0	3.0E+00 I		1.7E+01 I		0 0.10	302-01-2	Hydrazine, hydrazine suirate		V. 10. VI V8	2.1E+01 m			
3 0E-031 2 9E-041 1 0.10 7783-08-4 Hydrogen sulfide 2.6E+03 no 2.7E+04 nc 1.5E+02 nc 1.5E+02 nc 1.5E+03 nc 4 0E-02 h 4 0E-02 r 0 0.10 123-31-9 p-Hydroguinone 1.3E-02 nc 1.5E+02 nc 1.5E+02 nc 1.5E+02 nc 1.5E+03 nc 1.3E-02 i 1.3E-02 r 0 0.10 35554-44-0 Imazalil 1.6E+04 nc 1.0E+05 max 9.1E+03 nc 4.7E+01 nc 4.7E+03 nc 2.5E-01 i 2.5E-01 r 0 0.10 8734-19-7 Iprodione 1.6E+04 nc 1.0E+05 max 9.1E+03 nc 1.8E+03 nc 3.0E-01 i 3.0E-01 r 0 0.10 8734-19-7 Iprodione 1.1E+04 nc 1.0E+05 max 1.1E+03 nc 1.8E+03 nc 3.0E-01 i 3.0E-01 r 0 0.10 78-3-1 Isobutanol 1.1E+04 nc 1.0E+05 max 1.1E+03 nc 5.0E-01 3.0E-01 9.5E-041 2.0E-01 r 0.50-02 r 0.010 78-59-1 Isophorone 9.8E+02 nc 1.0E+04 nc 5.5E+01 nc 5.0E-01 3.0E-01 9.5E+02 r 0.010 3320-53-0 Isopropalin Isopropalin 6.5E+03 nc 3.3E+03 nc <td></td> <td></td> <td></td> <td>5.7E-03 I</td> <td>0 0.10</td> <td>0 7647-01-0</td> <td>Hydrogen chloride</td> <td></td> <td></td> <td>1.0E+00 m</td> <td>2.0E+00 nc</td> <td>1</td> <td></td>				5.7E-03 I	0 0.10	0 7647-01-0	Hydrogen chloride			1.0E+00 m	2.0E+00 nc	1	
4 0E-02 h 4 0E-02 r 0 0.10 123-31-9 p-Hydroquinone 2.0E403 mc 2.12E-03 mc 1.12E-01 mc 4.7E+02 mc 1.3E-021 1.3E-02 r 0 0.10 35554-44-0 Imazalii 8.5E+02 mc 8.9E+03 mc 4.7E+02 mc 9.1E+03 mc 2.5E-011 2.5E-01 r 0 0.10 83553-37-7 Imazaquin 1.6E+04 mc 1.0E+05 max 9.1E+03 mc 1.5E+03 mc 4 0E-02 I 4 0E-02 r 0 0.10 8734-19-7 Iprodione 2.6E+03 mc 2.7E+04 mc 1.5E+02 mc 1.5E+03 mc 3 0E-011 3 0E-01 r 1 0.10 76-83-1 Isobutanol 1.1E+04 mc 1.0E+05 max 1.1E+03 mc 1.8E+03 mc 9.5E-041 2 0E-01 r 0 0.10 78-59-1 Isophorone 9.8E+02 mc 1.0E+04 mc 5.5E+01 mc 5.0E-01 3.0E-01 1.5E-021 1.5E-02 r 0 0.10 78-59-1 Isophorone 9.8E+02 mc 1.0E+04 mc 1.0E+02 mc 3.7E+03 mc 1.5E-021 1.5E-02 r 0 0.10 182-54 a Isopropyl methyl phosphonic acid 6.5E+0		3 OE-03 I		2.9E-04 I	1 0.10	0 7783-08-4	Hydrogen sulfide	2 65103	2 7E+04 🔤	1.5E+02 m	1.5E+03 no		
1.3E-021 1.3E-02 r 0.010 35554-44-0 Imazalil 0.010 0.010 35554-44-0 Imazalil 1.0E+02 re 0.0E+05 max 9.1E+02 re 9.1E+03 re 1.0E+03 re 1.0E+04 re 1.0E+04 re 1.0E+03 re 1.5E+02 re 1.5E+03 re 1.5E+03 re 1.5E+03 re 1.0E+04 re 1.0E+05 max 1.1E+03 re 1.5E+03 re 1.5E+01 re 5.5E+01 re 5.0E+01 re 5.0E+02 re 1.5E+02 re 1.5E+02 re 1.5E+02 re 1.5E+02 re 1.5E+03 re 5.0E+03 re 3.3E+03 re 3.3E+03 re 3.3E+03 re 3.3E+03 re 3.7E+03		4 0E-02 h		4.0E-02 r	0 0.1	0 123-31-8	p-Hydroquinone	8 55±02 m	8.9E+03.m	4.7E+01 m	4.7E+02 nc	-	
2.5E-011 2.5E-011 0.00 81335-37-7 Imazaquin 1.0E+04 mc 1.0E+04 mc 1.5E+02 mc 1.5E+03 mc 4.0E-021 4.0E+02 r 0.00 36734-19-7 Iprodione 2.6E+03 mc 2.7E+04 mc 1.5E+02 mc 1.5E+03 mc 1.8E+03 mc 3.0E-011 3.0E-01 r 1.010 76-83-1 Isobutanol 1.1E+04 mc 1.0E+04 mc 1.0E+03 mc 1.8E+03 mc 1.8E+03 mc 8.5E-041 2.0E-011 0.00 76-83-1 Isobutanol 1.7E+04 mc 1.0E+04 mc 1.0E+04 mc 1.6E+03 mc 7.1E+01 ca 5.0E-01 3.0E-01 9.5E-041 2.0E-011 0.00 76-59-1 isophrone 9.8E+02 mc 1.0E+04 mc 5.5E+01 mc 5.0E-01 3.0E-01 3.0E-01 1.5E-021 1.5E+02 r 0.010 78-59-1 isophropalin Isopropyl methyl phosphonic acid 6.5E+03 mc 6.8E+04 mc 1.8E+02 mc 1.8E+03 mc 1.0E+011 1.1E+01 r 0.010 1832-54-8 isophropyl methyl phosphonic acid 6.5E+03 mc 3.4E+04 mc 1.8E+03 mc 3.7E+03 mc 1.8E+01 n 5.0E-02 r 0.010 1332-54-8 <td></td> <td>1.3E-02 I</td> <td></td> <td>1.3E-02 r</td> <td>0 0.1</td> <td>0 35554-44</td> <td>Imazalii</td> <td>1 65104 -</td> <td>1 0E+05 max</td> <td>9.1E+02 ne</td> <td>9.1E+03 nc</td> <td></td> <td></td>		1.3E-02 I		1.3E-02 r	0 0.1	0 35554-44	Imazalii	1 65104 -	1 0E+05 max	9.1E+02 ne	9.1E+03 nc		
4 0E-02 i 4.0E-02 r 0 0.10 36734-19-7 Iprodione 2.0E403 mc 2.12E405 max 1.1E+03 mc 1.8E+03 mc		2.5E-01 i		2.5E-01 r	0 0.1	0 61335-37-	r Imazaquin	2.65+03	2 7E+04 m	1.5E+02 nc	1.5E+03 nc		
3 0E-011 3 0E-01 r 1 0.10 76-83-1 Isobutanol 1.1E+04 mc 1.0E+04 mc 1.1E+00 mc 7.1E+01 mc 5.0E-01 3.0E-01 9.5E-041 2.0E-011 9.5E-04 r 2.0E-01 r 0 0.10 76-59-1 Isophorone 1.0E+02 mc 2.0E+03 mc 5.5E+02 mc 5.0E-01 3.0E-01		4 0E-02 I		4.0E-02 r	0 0.1	0 36734-19	r Iprodione	1 1 1 1 1 1 1 1 1 1	1 05+05 may	1.1E+03 m	1.8E+03 m		
9.5E-041 2.0E-011 9.5E-04 r 2.0E-01 r 0 0.10 78-59-1 isophorone 9.8E+02 nc 1.0E+04 nc 5.5E+01 nc 5.5E+02 nc 1.5E-02 i 1.5E-02 r 0 0.10 33820-53-0 isophorone 9.8E+02 nc 1.0E+04 nc 5.5E+01 nc 5.5E+02 nc 1.0E-011 1.1E-01 r 0 0.10 1832-54-8 isopropyl methyl phosphonic acid 6.5E+03 nc 6.8E+04 nc 4.0E+02 nc 1.8E+03 nc 5.0E-02 i 5.0E-02 r 0 0.10 1832-54-8 isopropyl methyl phosphonic acid 3.3E+03 nc 3.3E+04 nc 1.8E+02 nc 1.8E+03 nc 1.8E+01 n 5.0E-02 r 0 0.10 143-50-0 Kepone 2.5E-02 ca 1.1E-01 ca 3.7E-04 ca 3.7E-03 ca 1.8E+01 n 1.8E+01 r 0 0.10 77501-83-4 Lactofen 1.3E+02 nc 1.4E+03 nc 7.3E+00 nc 7.3E+01 nc		3.0E-01 I		3.0E-01 r	1 0.1	0 76-83-1	Isobutanol	1.10104 16	205+03 ~*	7.1E+00 ca	7.1E+01 c	5.0E-01	3.0E-0
1.5E-02 I 1.5E-02 r 0.010 33820-53-0 Isopropalin 9.0E+02 rc 1.0E+01 rc 0.021-04 rc 0.021-	9.55-041	2.0E-011	9.5E-04	r 2.0E-01 r	0 0.1	0 78-59-1	Isophorone	4./E+U2 CM	1 05+04	5 5F+01 m	5.5E+02 m		
1.0E-011 1.1E-01 r 0.010 1832-54-8 isopropyl methyl phosphonic acid 6.5E+03 nc 0.6E+04 nc 1.8E+02 nc 1.8E+03 nc 5.0E-021 5.0E-02 r 0.010 143-50-0 Isoxaben 2.5E-02 ca 1.1E-01 ca 3.7E-04 ca 3.7E-03 ca 1.8E+01 n 1.8E+01 r 0.010 143-50-0 Kepone 1.3E+02 nc 1.4E+03 nc 7.3E+00 nc 7.3E+01 nc	0.00.041	1 5E-02 i		1.5E-02 r	0 0.1	0 33820-53	o Isopropalin	9.0E+U2 nc	6 95±04	4 0F+02 ~~	3.7E+03 m		
5.0E-021 5.0E-02 r 0.010 82558-50-7 Isoxaben 3.3E+03 rc 3.4E+04 rc 1.0E+02 rc 1.0E+03 rc 3.7E-03 ca 1.8E+01 n 1.8E+01 r 0.010 143-50-0 Kepone 2.5E-02 ca 1.1E-01 ca 3.7E-04 ca 3.7E-03 ca 2.0E-03 r 0.010 77501-83-4 Lactofen 1.3E+02 rc 1.4E+03 rc 7.3E+00 rc 7.3E+01 rc		1.0E-01 I		1.1E-01 r	0 0.1	0 1832-54	isopropyl methyl phosphonic acid	0.011403 m	0.0CTU4 NG	1 85+02 ~~	1.8E+03 ~		
1.8E+01 n 1.8E+01 r 0.10 143-50-0 Kepone 2.0E-02 ca 1.1E-01 ca 3.7E-04 ca 5.7E-04 ca 2.0E-01 L 2.0E-03 r 0.10 77501-83-4 Lactofen 1.3E+02 nc 1.4E+03 nc 7.3E+00 nc 7.3E+01 nc		5 05-02 1		5.0E-02 r	0 0.1	0 82558-50	7 isoxaben	3.3E+U3 no	J.4ETU4 NG	3 7 5-04	37F-03		
20E031 20E03 0 0.10 77501-83-4 Lactofen 1.3E+U2 nc 1.4E+U3 nc 7.5E+U0 nc 7.5E	1.85+01 0	w	1.8E+01	r	0 0.1	10 143-50-0	Kepone	2.5E-02 ca	1.12-01 0	7 36+00	7 3E+01 ~		
	1.02+01	205.03		2.0E-03 r	0 0.1	10 77501-63	4 Lactofen	1.3E+02 nc	1.40103 06	1.3ETUU N			-

07

. . .

Kay : HIRIS h-HIEAST n=NCEA x=WITHDRAWN (=ROUTE EXTRAPOLATION ca=CANCER PRG nc=NONCANCER PRG sat=SOIL SATURATION max=CEILING LIMIT "(where: nc < 100X ca) "(where: nc < 10X ca)

						FOR PLANNING	PUR	POSE	S			
	OXICITY	NFORMA	TION	het de).;;;; : .4.4	CONTAMINANT	PRELIM	NARY REMEDI	AL GOALS (F	PRGs)	SOIL SCREENIN	IG LEVELS
SFo /(mg/kg-d)	RfDo (mg/kg-d)	5FI	RfDi [V skin O abs. C solls	CAS No.		Residential	industrial : Soll (mg/kg)	Ambient Alr	Tap Water	Migration to Grou DAF 20 (mg/kg) to the second	nd Water DAF 1
1.1E+00 h		1.1E+00 r		0 0.10	60-34-4	Methyl hydrazine	4.0E-01 ca	1.7E+00 ca	6.1E-03 ca	6.1E-02 ca	1	
	8.0E-02 h		2.3E-02 h	1 0.10	108-10-1	Methyl Isobutyl ketone	7.7E+02 nc	2.8E+03 nc	8.3E+01 nc	1.6E+02 nc		
	8 0E-02 h		8.0E-02 r	1 0.10	80-62-6	Methyl methacrylate	7.6E+02 nc	2.8E+03 nc	2.9E+02 nc	4.9E+02 nc		
3.3E-02 h		3.3E-02 r		0 0.10	99-55-8	2-Methyl-5-nitroaniline	1.3E+01 ca	5.8E+01 ca	2.0E-01 ca	2.0E+00 ca		
	2.5E-04 I		2.5E-04 r	0 0.10	298-00-0	Methyl parathion	1.6E+01 nc	1.7E+02 nc	9.1E-01 nc	9.1E+00 nc		
	5.0E-02 x		5.0E-02 r	0 0.10	85-46-7	2-Methylphenol	3.3E+03 nc	3.4E+04 nc	1.8E+02 nc	1.8E+03 nc	1.5E+01	8.0E-01
	5.0E-02 x		5.0E-02 r	0 0.10	108-39-4	3-Methylphenol	3.3E+03 nc	3.4E+04 nc	1.8E+02 nc	1.8E+03 nc		
	5.0E-03 h		5.0E-03 r	0 0.10	106-44-5	4-Methylphenol	3.3E+02 nc	3.4E+03 nc	1.8E+01 nc	1.8E+02 nc	}	
	6 0E-03 h		1.1E-02 h	1 0.10	25013-15-4	Methyl styrene (mixture)	1.2E+02 no	5.2E+02 nc	4.2E+01 ns	6.0E+01 nc		
	7.0E-02 h		7.0E-02 r	1 0.10	98-83-9	Methyl styrene (alpha)	6.8E+02 sat	6.8E+02 set	2.6E+02 nc	4.3E+02 nc		
	5.0E-03 n		8.6E-01 I	1 0.10	1834-04-4	Methyl tertbutyl ether (MTBE)			3.1E+03 nc	1.8E+02 nc		
	1.5E-01 i		1.5E-01 r	0 0.10	51218-45-2	Metolaclor (Dual)	9.8E+03 nc	<u>1.0E+05 max</u>	5.5E+02 nc	<u>5.5E+03 nc</u>		
	2.5E-02 I		2.5E-02 r	0 0.10	21087-64-9	Metribuzin	1.6E+03 nc	1.7E+04 nc	9.1E+01 nc	9.1E+02 nc		
1.6E+00 h	2.0E-04 I	1.8E+00 r	2.0E-04 r	0 0.10	2385-85-5	Mirex	2.5E-01 ca*	1.1E+00 ca	3.7E-03 🛤	3.7E-02 📾		
	2 0E-03 I		2.0E-03 r	0 0.10	2212-87-1	Molinate	1.3E+02 nc	1.4E+03 nc	7.3E+00 nc	7.3E+01 nc		
	5.0E-03 h			0 0.01	7439-98-7	Molybdenum	3.8E+02 nc	8.5E+03 nc		1.8E+02 nc		
	1.0E-01 h		1.0E-01 h	0 0.10	10599-90-3	Monochloramine	6.5E+03 nc	6.8E+04 nc	3.7E+02 nc	3.7E+03 nc		
	2.0E-03 i		2.0E-03 r	0 0.10	300-76-5	Naled	1.3E+02 nc	1.4E+03 nc	7.3E+00 nc	<u>7.3E+01 nc</u>		· .
	1.0E-01 I		1.0E-01 r	0 0.10	15299-99-7	Napropamide	6.5E+03 nc	6.8E+04 nc	3.7E+02 nc	3.7E+03 nc		
1	2.0E-02 i			0 0.01	7440-02-0	Nickel (soluble salts)	1.5E+03 nc	3.4E+04 nc		7.3E+02 nc	1.3E+02	7.0E+00
						"CAL-Modified PRG" (PEA, 1994)	1.5E+02					
		8.4E-01 J		0 0.01	n/a	Nickel refinery dust			8.0E-03 ca			
		1.7E+00 i		0 0.01	12035-72-2	Nickel subsulfide		1.1E+04 cm	4.0E-03 ca			
	1.5E-03 x		1.5E-03 r	0 0.10	1929-82-4	Nitrapyrin	9.8E+01 nc	1.0E+03 nc	5.5E+00 nc	5.5E+01 nc		
	1.6E+00 i			0 0.10	14797-55-8	Nitrate				5.8E+04 nc		
	1.0E-01 x			0 0.10	10102-43-9	Nitric Oxide	6.5E+03 nc	1.0E+05 max		3.7E+03 nc		
	1.0E-01 I			0 0.10	14797-65-0	Nitrite	6.5E+03 na	1.0E+05 max		3.7E+03 no		
	0.0E-05 r		5.7E-05 h	0 0.10	88-74-4	2-Nitroaniline	3.9E+00 nc	4.1E+01 nc	2.1E-01 nc	2.2E+00 nc		
				0 0.10	99-09-2	3-Nitroaniline						
				0 0.10	100-01-8	4-Nitroaniline						
	5 OE-04 I		5.7E-04 h	1 0.10	98-95-3	Nitrobenzene	1.8E+01 nc	9.4E+01 nc	2.1E+00 nc	3.4E+00 nc	1.0E-01	7.0E-03
	7.0E-02 h	•	7.0E-02 r	0 0.10	67-20-9	Nitrofurantoin	4.6E+03 nc	4.8E+04 nc	2.6E+02 nc	2.6E+03 nc		
1.5E+00 h		9.4E+00 h		0 0.10	59-87-0	Nitrofurazone	3.0E-01 ca	1.3E+00 ca	7.2E-04 ca	4.5E-02 ca		
1	1.0E+00 x			0 0.10	101102-44-0	Nitrogen dioxide						
	1.0E-01 i		1.0E-01 r	0 0.10	550-88-7	NitroguanIdine	6.5E+03 nc	6.8E+04 nc	3.7E+02 ns	3.7E+03 ne		
				0 0.10	100-02-7	4-Nitrophenol						
9.4E+00 r	5.7E-03 r	9.4E+00 h	5.7E-03 I	1 0.10	79-40-9	2-Nitropropane			7.2E-04 ca	3.5E+01 ca		
5.4E+001		5 6E+00 i		1 0.10	924-18-3	N-Nitrosodi-n-butylamine	2.2E-02 ca	5.5E-02 ca	1.2E-03 ca	2.0E-03 ca		
2.8E+00 i		2.8E+00 r		0 0.10	1110-54-7	N-Nitrosodiethanolamine	1.6E-01 ca	<u>6.8E-01 ca</u>	<u>2.4E-03 ca</u>	2.4E-02 ca		
1.5E+02 i		1.5E+02 I		0 0.10	55-18-5	N-Nitrosodiethylamine	3.0E-03 ca	1.3E-02 ca	4.5E-05 ca	4.5E-04 ca	1	
5.1E+01 I		4.9E+01 i		0 0.10	62-75-9	N-Nitrosodimethylamine	8.7E-03 ca	3.7E-02 ca	1.4E-04 ca	1.3E-03 ca		
4.9E-031		4.9E-03 r		0 0.10	86-30-6	N-Nitrosodiphenylamine	9.1E+01 ca	3.9E+02 cm	1.4E+00 ca	<u>1.4E+01 ca</u>	1.0E+00	6.0E-02
7.0E+001		7.0E+00 r		0 0.10	621-64-7	N-Nitroso di-n-propylamine	6.3E-02 ca	2.7E-01 ca	9.6E-04 ca	9.6E-03 ca	5.0E-05	2.0E-06
2.2E+01		2.2E+01 r		0 0.10	10595-95-8	N-Nitroso-N-methylethylamine	2.0E-02 ca	8.7E-02 ca	3.1E-04 ca	3.1E-03 a		
2.1E+00 I		2.1E+00 i		0 0.10	930-55-2	N-Nitrosopyrrolidine	2.1E-01 cm	9.1E-01 ca	3.1E-03 a	3.2E-02 ca	ļ	
	1.0E-02 h		1.0E-02 r	0 0.10	99-08-1	m-Nitrotoluene	6.5E+02 nc	6.8E+03 nc	3.7E+01 nc	3.7E+02 nc		
1	106-02 6		1.0E-02 r	0 0 10	99-99-0	io-Nitrotoluene	6.5E+02 mc	6.8E+03 nc	3.7E+01 nc	3.7E+02 nc	1	

Kay : LIRIS DEHEAST DENCEA TRANSFER TO ANTHORADAN DEROTE EXTRAPOLATION CONCANCER PRG DO NONCANCER PRG BATESOIL BATURATION MAX CEILING LIMIT "(where: nc < 100X ca) "(where: nc < 10X ca)

	NOY: PING	IS IN-MEAS	I HANGEA X		THI PROVIEC	and a second single of a fight faile from the single of the second second second second second second second se					Letter Ar aba	an shasi
	后的物理的	医复数菌	e e ca se ca	5 C C C U	11 1 5 10 5 10		PIIRL	JUSE				
	h WIMEN	the th	astrusii			FURFLANNING	IUVINI			NELENAL, SEL		79.44
		YARAA		ritinit				n san kana ka			SOUL SCREENIN	GIEVELS
TO	XICITY INF	ORMAT	ION	(Charle		THE PLET CONTAMINANT AT HISTORY AND THE	MART PRELIMI	NARY REMEDIA	L GUALS (PH	031	Mioration Lo Grou	nd Water
	in a fail fit at	محمد معمد معمد مانان از از محمد م		V skin		The second of the second s	Residential	Y Industrial . F. S. C	Ambient Alr	Tap Waler	DAF 20	DAF 1
SF0	NDO	ori noko-d) (r	ະ ແມ່ນ 😳 😳	C solls	LAS NO. 2		Soll (mg/kg)	Soil (mg/kg)	. (ug/m^3) 11. ++	(uo/l)	(mg/kg)	(mg/kg) 😳 🗄 🗤
adam when a	www.www.	un un un de la la			the state of the s	PAHs continued					0.05.00	105.01
735+001	7.	3E+00 r		0 0.10	50-32-8	Benzo(a)pyrene	6.1E-02 ca	2.6E-01 🚥	9.2E-04 ca	9.2E-03 ca	8.0E+00	4.02-01
	4.					"CAL-Modified PRG" (PEA, 1994)				1.5E-03	4.00.00	
7 3E-01 0		3E-03 r		0 0.10	218-01-9	Chrysene	7.2E+00 sat	7.2E+00 sat	9.2E-01 ca	9.2E+00 ca	1.0E+02	0.012+00
F.00-03 II						"CAL-Modified PRG" (PEA, 1994)	6.1E+00		• • • ·	0.077.00	0.00.00	0 AE AA
7 35-00 -		3E+00 *		0 0.10	53-70-3	Dibenz[ah]anthracene	6.1E-02 ca	2.6E-01 ca	9.2E-04 ca	9.2E-03 ca	2.02+00	245+02
1.00100	4.05-021		4.0E-02 /	0 0.10	200-44-0	Fluoranthene	2.6E+03 nc	2.7E+04 nc	1.5E+02 nc	1.51:+03 m	4.3E+03	2.10+02
	4.0E-021		4.0E-02 /	1 0.10	86-73-7	Fluorene	9.0E+01 sat	9.0E+01 sat	1.5E+02 nc	2.41:+02 m	5.02+02	2.0E+U1
735.01 -	7.02741	3E-01 r		0 0.10	193-39-5	Indeno[1,2,3-cd]pyrene	6.1E-01 ca	2.6E+00 ca	9.2E-03 ca	9.2E-02 ca	1.4±+01	1.02-01
1.36-011	4 0E-02 n		4 0E-02 r	1 0.10	91-20-3	Naphthalene	2.4E+02 sat	2.4E+02 sat	1.5E+02 nc	2.41+02 nc	0.41+01	4.UETUU
	3 05-02 1		3.0E-02	1 0.10	129-00-0	Pyrene	1.0E+02 sat	1.0E+02 sat	1.1E+02 nc	1.85+02 ns	4.26+03	2.10+02
1.5E.011	9.0E-03.1 4	5E-01 r	9.0E-03	0 0.10	67747-09-5	Prochloraz	3.0E+00 ca	1.3E+01 ca	4.5E-02 ca	3.3E+02 ca		
	6 0F-03 h		6.0E-03 r	0 0.10	26399-36-0	Profluralin	3.9E+02 nc	4.1E+03 nc	2.2E+01 nc	2.2E+02 no	1	
	1 5F-02 I		1.5E-02 ·	0 0.10	1610-18-0	Prometon	9.8E+02 nc	1.0E+04 nc	5.5E+01 nc	5.5E+02 nc	1	
1	4 05-03 1		4.0E-03 r	0 0.10	7267-19-6	Prometryn	2.6E+02 mc	2.7E+03 nc	1.5E+01 m	1.5E+02 m		
	7 6E-02 1		7.5E-02 -	0 0 10	23950-58-5	Pronamide	4.9E+03 nc	5.1E+04 nc	2.7E+02 nc	2./E+03 m	l	
	1 35-02 1		1.3E-02 4	0 0.10	1918-16-7	Propachlor	8.5E+02 nc	8.9E+03 nc	4.7E+01 nc	4.7E+02 nc		
	5 OF-01 1		5 0E-03 /	0 0.10	709-98-8	Propanil	3.3E+02 nc	3.4E+03 nc	1.8E+01 ns	1.85+02 m	l	
	2 0F-02 1		2.0E-02 r	0 0.10	2312-35-8	Propargite	1.3E+03 no	1.4E+04 nc	7.3E+01 nc	7.3E+02 nc		
1	2 05-011		2.0E-03 /	0 0.10	107-19-7	Propargyl alcohol	1.3E+02 nc	1.4E+03 nc	7.3E+00 nc	7.3E+01 nc	1	
	2 05.021		2.0E-02 /	0 0 10	139-40-2	Propazine	1.3E+03 nc	1.4E+04 nc	7.3E+01 nc	7.3E+02 nc		
	205-021		2.0E-02 -	0 0 10	122-42-9	Propham	1.3E+03 nc	1.4E+04 ns	7.3E+01 nc	7.3E+02 nc		
1	135.021		1.35.02 /	0 0 10	60207-90-1	Propiconazole	8.5E+02 nc	8.9E+03 nc	4.7E+01 nc	4.7E+02 nc	1	
	205-04 5		205+01 -	0 0.10	57-65-6	Propylene glycol	1.0E+05 max	1.0E+05 max	7.3E+04 nc	7.3E+05 nc		
 	7.05.01 5		7.0E-01 -	0 0 10	111-35-3	Propylene glycol, monoethyl ether	4.6E+04 nc	1.0E+05 max	2.6E+03 nc	2.6E+04 nc	ļ	
1	7.05-01 0		575.011	0 0 10	107-98-2	Propylene glycol, monomethyl ether	4.6E+04 nc	1.0E+05 max	2.1E+03 no	2.6E+04 №	1	
245.011	8.8F.01 P	1.35-02 1	8.6E-03 1	1 0.10	76-56-9	Propylene oxide			5.2E-01 ca	2.2E-01 ca		
	2 55-01 1		2.5E-01 r	0 0.10	81335-77-5	Pursuit	1.6E+04 nc	1.0E+05 max	9.1E+02 nc	9.1E+03 nc		
1	2 45-02 1		2 6F-07	0 0 10	51030-58-1	Pydrin	1.6E+03 no	1.7E+04 no	9.1E+01 no	9.1E+02 m		
1	106-031		1.0E-03 /	0 0 10	110-86-1	Pyridine	6.5E+01 nc	6.8E+02 m	3.7E+00 nc	3.7E+01 m		
	5 0E-03 1		5 OF A4 -	0 0 10	13593-03-8	Quinalphos	3.3E+01 nc	3.4E+02 no	1.8E+00 nc	1.8E+01 nc	1	
125-04-	J.UE-04 I	1 25+01 -	0.00.041	0 0 10	91-22-5	Quinoline	3.7E-02 🚥	1.6E-01 ca	5.6E-04 ca	5.6E-03 ca	}	
1.4E+U1 N	3.05.03.1	1 16.01 -	3 0F-01 -	0 0 10	121-82-4	RDX (Cyclonite)	4.0E+00 ca*	1.7E+01 ca	6.1E-02 ca	6.1E-01 ca		
1.10-011	305-031	1.16-011	3 05-02 4	0 0 10	10453-80-8	Resmethrin	2.0E+03 nc	2.0E+04 nc	1.1E+02 nc	1.1E+03 nc		
1	3.0E-021		505-021	0.10	299-84-3	Ronnel	3.3E+03 nc	3.4E+04 nc	1.8E+02 nc	1.8E+03 nc		
1	5.02-02 h		4 NF-01 -	0 0.10	83-79-4	Rotenone	2.6E+02 nc	2.7E+03 m	1.5E+01 nc	1.5E+02 mc		
	3 46.031		2 5F-02 -	0 0 10	78578-05-0	Savey	1.6E+03 nc	1.7E+04 nc	9.1E+01 nc	9.1E+02 m		
	4.30-04 I			0 0 10	7783-00-8	Selenious Acid	3.3E+02 nc	3.4E+03 nc		1.8E+02 AC	E 05.00	3 05 04
	5.02-03 I			0 0 01	7782-49-2	Selenium	3.8E+02 m	8.5E+03 m		1.85+02 nc	5.0E+00	3.02-01
	505.031			0 0 10	630-10-4	Selenourea	3.3E+02 nc	3.4E+03 nc		1.8E+02 m		
	9.0E.02 I		9.0E-02 r	0 0.10	74051-80-2	Sethoxydim	5.9E+03 nc	6.1E+04 nc	3.3E+02 nc	3.35+03 nc	9.45.04	3 05.00
1	5 0F-01 1			0 0.01	7440-22-4	Silver and compounds	3.8E+02 nc	8.5E+03 nc		1.8E+02 m	J	2.00+00
125.01 +	5 0F-01 1	1.2E-01 /	2.0E-03 r	0 0.10) 122-34-9	Simazine	3.7E+00 ca*	1.6E+01 ca*	5.6E-02 ca	5.6E-01 ca		
1.40-01 0	405-031		4.0E-03 /	0 0.10) 26628-22-8	Sodium azide	2.6E+02 nc	2.7E+03 nc	1.5E+01 nc	1.5E+02 nc	1	
275.01 -	3 0F-07 1	2.7E-01 r	3.0E-02 r	0 0.10	0 148-18-5	Sodium diethyldithlocarbamate	1.6E+00 ca	7.1E+00 a	2.5E-02 ca	2.5E-01 ca	- <u> </u>	
L.12-010	2 05-05 1		2.0E-05 /	0 0.10	0 02-74-8	Sodium fluoroacetate	1.3E+00 ng	1.4E+01 nc	7.3E-02 nc	7.3E-01 nc	1	
1	1 DE-03 h		1.0E-03 /	0 0.10	0 13718-26-8	Sodium metavanadate	6.5E+01 m	6.8E+02 nc	3.7E+00 nc	3./E+01 nc	J	
1	1.00.000 11					A DESCRIPTION OF THE OWNER OWNER						

.

.

			1. 21		FUR PLANNING	FUN		\mathbf{v}			
						PRELIMI	NARY REMEDI	AL GOALS (PR	(Gs)	SOIL SCREENIN	G LEVELS
<u> </u>	JAILITY INFORM	ATION	V skin					A mbleat Alc	Teo Water	Migration to Grour	DAF 1
SFo	RIDo SFI	RIDI 📊	O abs.	CAS No.		Soil (ma/ka)	Soll (ma/ka)	(ug/m^3)	(ug/) 📲 🔛	(mg/kg)	(mg/kg)
1/(mg/kg-d)	(mg/kg-d) 1/(mg/kg-d) (mg/kg-d) 👘	C BOILS	70.00 6	1 1 2 Trichloroethane	6.5E-01 ca	1.5E+00 ca	1.2E-01 ca	2.0E-01 ca	2.0E-02	9.0E-04
5.7E-02 I	4.0E-031 50E-021	4.0E-03 r	1 0.10	79-00-5	Trichloroethylene (TCF)	3.2E+00 cm	7.0E+00 ca*	1.1E+00 ca*	1.6E+00 ca*	6.0E-02	3.0E-03
1.1E-02 n	6.0E-03 n 6.0E-03 r	6.0E-03 r	1 0.10	/9-01-0	Trichlorofluoromethane	3.8E+02 nc	1.3E+03 nc	7.3E+02 nc	1.3E+03 nc		
	3.0E-011	2.0E-01 h	1 0.10	75-09-4	2.4.5 TrichloronbAnol	6.5E+03 nc	6.8E+04 nc	3.7E+02 nc	2.7E+03 nc	2.7E+02	1.4E+01
4	1.0E-01 J	1.0E-01 r	0 0.10	95-95-4	2.4.5-Trichlerophenol	4 0E+01 ca	1.7E+02 cm	6.2E-01 ca	6.1E+00 ca	2.0E-01	8.0E-03
1.1E-02 I	1.1E-02 I		0 0.10	88-06-2	2,4,0-1 fichlorophenovacetic Acid	6 5E+02 m	6.8E+03 nc	3.7E+01 nc	3.7E+02 nc		
	1.0E-02 I	1.0E-02 r	0 0.10	93-78-5	2,4,5- Hichorophanoxy) propionic acid	52F+02 m	5.5E+03 nc	2.9E+01 nc	2.9E+02 nc		
	8 DE-03 i	8.0E-03 r	0 0.10	93-72-1	2-(2,4,5-11chlorophenoxy) propionic acid	1.5E+01 m	5.0E+01 ng	1.8E+01 nc	3.0E+01 nc		
	5.0E-03 I	5.0E-03 r	1 0.10	598-77-6	1, 1, 2- Themoropropane	1 4E-03 ca	3.1E-03 ca	9.6E-04 ca	1.6E-03 ca		
7.0E+00 h	6 OE-03 1 7.0E+00	5.0E-03 r	1 0.10	96-18-4	1,2,3-Thenloropropane	1 1E+01 m	3.8E+01 m	1.8E+01 nc	3.0E+01 nc		
1	5.0E-03 h	5.0E-03 r	1 0.10	96-19-5	1,2,3-1 fichiorophopene	5.6E+03 M	5.6E+03 aat	3.1E+04 nc	5.9E+04 nc		
	3.0E+011	6.6E+00 h	1 0.10	76-13-1	11,1,2-111Chioro-1,2,2-11110010611an6	2 0E+02 m	2 0F+03 m	1.1E+01 nc	1.1E+02 nc		
	3 0E-03 I	3.0E-03 r	0 0.10	58138-08-2	The formation of the fo	2 3E+01 m	8 4F+01 m	7.3E+00 m	1.2E+01 nc		
1	2 0E-03 r	2.0E-03 i	1 0.10	121-44-8	I netnyiamine	5.85+01	2.5E+02 cm	8.7E-01 ca*	8.7E+00 ca		
7.7E-03 I	7.5E-03 I 7.7E-03	r 7.5E-03 r	0 0.10	1582-09-8		1 25+01	5 2E+01 m	1.8E-01 ca	1.8E+00 ca		
3.7E-02 h	3.7E-02	r	0 0.10	512-58-1	I rimetnyi phosphate	3 3 5 + 00	3.4E+01 m	1.8E-01 m	1.8E+00 m		
	5.0E-05 I	5.0E-05 r	0 0.10	99-35-4	1,3,5-I rinitrobenzene	5.5E+02 m	8 8E+03 m	3.7E+01 m	3.7E+02 nc	1	
1	1.0E-02 h	1.0E-02 r	0 0.10	478-45-8	I rinitrophenyimethyinitramine		6.4E+01 cam	2.2E-01 ca*	2.2E+00 cm	-	
3.0E-02 1	5.0E-041 3.0E-02	r 5.0E-04 r	0 0.10	118-96-7		1.52701 Ca	0.42.01.4				
	3.0E-03 l		0 0.01	7440-61-1	Uranium (soluble saits)	5 4 E+02 m	1 2E+04 m		2.6E+02 m	6.0E+03	3.0E+0
1	7.0E-03 h		0 0.01	7440-82-2		6 9E+02 m	1.5E+04 m		3.3E+02 nc	6.0E+03	3.0E+0
	9 0E-03 I		0 0.01	1314-62-1		1 55102 10	3 4 E+04 m		7.3E+02 nc	6.0E+03	3.0E+0
	2.0E-02 h		0 0.01	13701-70-7	Vanadium suirate	8.5E+01	6.8E+02 m	3.7E+00 m	3.7E+01 nc		
	1.0E-03 I	1.0E-03 r	0 0.10	1929-77-7		1.65+03	1 7E+04	9.1E+01 m	9.1E+02 ng		
	2.5E-02 I	2.5E-02 r	0 0.10	50471-44-8		7 85+02 ~	2.6E+03.cc	2 1E+02 m	4.1E+02 ng	1.7E+02	8.0E+0
	1.0E+00 h	5.7E-02 I	1 0.10	108-05-4	Vinyi acetate	1 95-01	A 1E-01	6 1E-02 -	1.0E-01 ca	•	
1.1E-01 r	8.6E-04 r 1.1E-01	h 8.6E-04 i	1 0.10	593-80-2	Vinyi promide (promoethene)	1.65-02	3.5E-02	2.2E-02	2.0E-02 c	1.0E-02	7.0E-0-
1.9E+00 h	3.0E-01	h	1 0.10	75-01-4		2 05+01	2 0E+02 ~~	1.1E+00 m	1.1E+01 m		
	3.0E-04 I	3.0E-04 r	0 0.10	81-81-2	warrann	2.01.001 00	3 2 5+02	7 3E+02 m	1.4E+03 m	2.1E+02	1.0E+0
	2.0E+00 I	2.0E-01 x	1 0.10	108-38-3	m-Xylene	3.26102 18	3 25+02	7 3E+02 ~~	1.4E+03 m	1.9E+02	9.0E+0
	2.0E+00 I	2.0E-01 x	1 0.10	95-47-8	o-Xylene	3.2ETU2 ##	3 26+02	T.ULTUL IN	1,-14-00 1	2.0E+02	1.0E+0
			1 0.10) 106-42-3	p-Xylene	3.2ETU2 ##	3 26102 14	7 3E+02 ~~	1.4E+03 ~	2.0E+02	1.0E+0
	2.0E+00 I	2.0E-01 x	1 0.10	1330-20-7	Xylene (mixed)	3.25+02 ##	1 05105		1.1E+04 ~	1.2E+04	6.2E+0
	3.0E-01 I		0 0.0	1 7440-66-6	Zinc	2.3E+04 MG	5 1 ELO2	·	1 1E+01 ~		
	3.0E-04 I		0 0.0	1 1314-84-7	Zinc phosphide	2.35401 00	3.15TUA NG	1 85+02	1 8F+03 ~		
1	5 0E-02 I	5.0E-02 r	0 0.1	12122-87-7	Zineb	3.32+03 nc	J.4ETU4 NG	1.01.704 16	1.02.00 18	<u></u>	

.

Key =1RIS h=	HEAS'I n=NCE	A x=WITHDRAWN	o=Other EPA D	DOCUMENTS	renoute Extr.	APOLATION COSCANCER PRG ACENONCANCER PRG SITESOIL SATU	RATION max=CEILING	SLIMIT "(where: nc < 1	100X ca) **(where:	nc < 10X ca)		
an shekara San shekara San shekara				TA S		FOR PLANNING	PURP	OSES	Skap da	ann -	\$	
ing size and	TOXICI	TY INFORM	ATION	en kinder der konstruction Kinder	ille back in this	CONTAMINANT	PRELIMIN		ATION GOAL	S (PRGs)	SOIL SCRE	ENING LEVELS
5 SFo	RIDo	SFI	RIDi	V skin O atis	CAS No	· · · · · · · · · · · · · · · · · · ·	Queidontial				Migration to Gr	ound Water
1/(mg/kg·d)	(my/ky-d)	1/(mg/kg-d)	(my/kg·d)	C soils	NDS KARANA		Soil (ing/kg)	Soil (mg/kg)	Ampient Air (ug/m*3)	Tap Water	DAI 20	(mg/kg)
L			الايىنى ويوقد الم	ې د د د د واد ورونه			WI- FURA- All Catalors	ter i din kana dari karak	N Pilori din Mary	anana an Kata	Pla.	
87E-03 i	4 0E-03	1 87E-03 1	4 0E-03	10	30560-19-1	Acephate	5.6E+01 ca**	2.8E+02 ca*	7.7E-01 cm	7.7E+00 ca		
	2 0F.02	7 /E-03 I	2 6E-03	1 1	75-07-0	Acetaldehvde	1.1E+01 ca**	2.3E+01 ca**	8.7E-01 ca	1.7E+00 ca		
	1 DE-01	•	1 0E-01	1 1	67-64-1	Acelone	1.2E+03 nc	6.2E+03 ac	7.3E+01 nc	7.3E+02 nc	1 65+01	8.05.01
	8 UE-04	ь	6 UE-04	1001	75 86 5	Acetone cvanohvdrin	4.9E+01 nc	7.0E+02 nc	2.9E+00 nc	2.9E+01 nc	1.02+01	0.02-01
	1 0E-03	<u>.</u>	1 7E-02	<u></u>	75-05-8	Acetonitrile	2.7E+02 nc	1.7E+03 nc	6.2E+01 no	7.9E+01 nc		
1 1E-01 o	1 3E-02	. 11E-01 /	1 3E-02	x 1 1 0 01	50594-66-6	Aceiophenone	4.9E-01 nc	1.6E+00 nc	2.1E-02 nc	4.2E-02 nc		
	2 0E-02	h	5 7E-06	1.1	107-02-8	Acrolein	1.0E-01 nc	3.4E-01 nc	2.1E-02 no	4.2E-02 nc		
4 6E + 00 I	2 OE -04	4 6E+00 +	2 0E -04	1 0 0.1	79-06-1	Acrylamide	1.1E-01 a	5.4E-01 ca	1.5E-03 ca	1.5E-02 ca		
5 4E-01 I	5 DE-D1 1 DE-D3	ı h 2.4E-01 ı	2 9E-04 5 7E-04	1001 11	79-10-7	Acrylic acid	2.9E+04 nc	1.0E+05 max	1.0E+00 nc	1.8E+04 nc		
8 1E-02 h	1 0E-02	1 8 0E-02 r	1 0E-02	1 0 01	15972-60-8	Alachlor	6 0F+00 ca	3.1E-01 ca*	2.8E-02 cm	3.9E-02 ca		
	1 5E-01	•	1 5E-01	1 0 01	1596-84-5	Alar	9.2E+03 nc	1.0E+05 max	5.5E+02 nc	5.5E+03 nc		
	1 0E-03	!	1 0E-03	1 0 01	116-06-3	Aldicarb	6.1E+01 nc	8.8E+02 nc	3.7E+00 nc	3.6E+01 nc		
176.01	1 DE-03 3 DE-05	, 17 5-01 ,	1 0E-03	1 0 01	1646-88-4	Aldicarb sulfone	6.1E+01 nc	8.8E+02 nc	3.7E+00 nc	3.6E+01 nc		
	2 5E-01		2 5E-01	1 0 01	5585-64-8	Ally	1.5E+04 oc	1.5E-01 ca 1.0E+05 max	91E+02 or	4.0E-03 ca	5.0E-01	2.0E-02
	5 0E-03	1	5 OE -03	1 0 01	107-18-6	Alivi alcohol	3.1E+02 nc	4.4E+03 nc	1.8E+01 nc	1.8E+02 ac		
ł	5 0E-02	*	2 9E-04	1 0 01	107-05-1	Allyl chloride	3.0E+03 nc	4.3E+04 nc	1.0E+00 nc	1.8E+03 nc		
	4 0E-04	n	14E-03	<u>n U</u>	20850.73 9	Aluminum Aluminum phosobide	7.6E+04 nc	1.0±+05 max	5.1E+00 AC	3.6E+04 nc		
	3 0E-04		3 0E-04	1 0 01	20639-73-8 67485-29-4	Amdro	1.8E+01 nc	0.2E+02 AC	1 1E+00	1.5E+01 nc	1	
	9 0E-03	•	9 0E-03	1001	834-12-8	Ametryn	5.5E+02 nc	7.9E+03 ne	3.3E+01 ac	3.3E+02 oc		
	7 0E-02	h	7 0E-02	r 0 01	591-27-5	m-Aminophenol	4.3E+03 nc	6.2E+04 nc	2.6E+02 nc	2.6E+03 nc		·····
	2 0E-05 2 5E-03	n	2 0E-05	1 0 0.1	504-24-5	4-Aminopyridine	1.2E+00 nc	1.8E+01 nc	7.3E-02 AC	7.3E-01 nc		
			2 9E-02	; ; ; ; ;	7664 41-7	Ammonia	I.DE+U2 nc	2.2E+03 nc	9.1E+00 nc	9.1E+01 nc		
	2 0E 01	•		0 01	7773 06-0	Ammonium sulfamate	1.2E+04 AC	1.0E+05 max	1.0L102 AC	7.3E+03 or		·
5 7E-03 ;	7 UE-03	n 57E-03 /	2 9E-04	1 0 01	62-53-3	Aniline	8.5E+01 co**	4.3E+02 cm*	1.0E+00 nc	1.2E+01 ca*		
	4 0E-04			0	7440-36-0	Antimony and compounds	3.1E+01 nc	8.2E+02 nc		1.5E+01 nc	5.0E+00	3 0E-01
ł	9 DE-04	n ħ		0	1314-60-9	Antimony perioxide	3.9E+01 AC	1.0E+03 nc		1.8E+01 nc		
	4 0E-04	h		0	1332-81-6	Antimony tetroxide	31E+01 nc	8 2E+02 oc		1.5E+01 mc		
	4 0E-04	h	5 7E-05	1 0	1309-64-4	Antimony trioxide	3.1E+01 nc	8.2E+02 nc	2.1E-01 nc	1.5E+01 nc		
	1 3E-02	<u> </u>	1 3E-02	1001	74115-24-5	Apollo	7.9E+02 nc	1.1E+04 nc	4.7E+01 nc	4.7E+02 nc		
2 5E-02 I	5 0E-02 3 0E-04	h 25E-02 i	5 OE -02	1 0 01	140-57-8		1.9E+01 ca	9.9E+01 ca	2.7E-01 ca	2.7E+00 ca		
1 SE+00 .	3 DE U4	, 15E+01 i		0 003	7440-38-2	Arsenic (cancer endpoint)	3.9E-01 ca	4.4E+02 AC 2.7E+00 ca	4 5E-04 c	4.5E-02 a	2 9E+01	1.0E+00
			1 4F-05	• •	7784 42 1	Arsine (see arsenic for cancer endpoint)			5.2E-02 nc			1.02.00
	9 UE 03	•	9 OE -03	1 01	76578-12-6	Assure	5.5E+02 AC	7.9E+03 nc	3.3E+01 nc	3.3E+02 nc		
2.2E-01	5 0E-02	1 b 2.25.01 /	5 0E-02	<u>1001</u>	3337-71-1	Asulam	3.1E+03 nc	4.4E+04 nc	1.8E+02 nc	1.8E+03 nc		
	4 DE-04		4 0E-04	1 0 01	71751-41-2	Avermectin B1	2.2E+00 ca	3.5E+02 m	3.1E-02 ca	3.0E-01 ca		
1 1E-01		1 1E-01 i		0 01	103-33-3	Azobenzene	4.4E+00 c	2.2E+01 cm	6.2E-02 ca	6.1E-01 a		
1	7 0E-02	•	1 4E-04	h 0	7440-39-3	Barium and compounds	5.4E+03 nc	1.0E+05 max	5.2E-01 nc	2.6E+03 nc	1.6E+03	8.2E+01
1	4 0E-03 3 0E-02	1	4 0E-03 3 0E-02	100.1	114-26-1	Bayleton	2.4E+02 nc	3.5E+03 nc	1.5E+01 nc	1.5E+02 nc		
	2.5E-02	!	2 5E-02	1 0 01	68359-37-5	Baythroid	1.5E+03 AC	2.0E+04 nc	91E+01	1.1C+U3 nc		
	3 OE-01	1	3 0E-01	1 0 0.1	1861-40-1	Benefin	1.8E+04 nc	1.0E+05 max	1.1E+03 nc	1.1E+04 nc		
 	5 0E-02	1	5 OE -02	. 0 0.1	17804-35-2	Benomyl	3.1E+03 nc	4.4E+04 nc	1.8E+02 nc	1.8E+03 nc		
	1 0E-01	•	3 UE -02	1 U 01	25057-89-0	Benzaldehyde	1.8E+03 nc	2.6E+04 nc	1.1E+02 nc	1.1E+03 AC		
5 5E-02	3 0E 03	n 27E-02	17E-03	A 1	71-43-2	Benzene	6.5E-01	0.0E+04 AC	3./E+U2 nc	3.5E+03 nc	3 05 03	2 05 02
										0.00-01 64	0.02-02	2.00-03

· . .

Key HIRIS A-H	EAST n=NCF	EA x=WITHDRAW	N 0=Other EPA (DOCU	MENTS	I=ROUTE EXTRA	POLATION CAECANCER PRG ACENONCANCER PRG & LESOIL SATUF	RATION max=C	EILING	LIMIT "(where: nc < 1)	JOX ca) **(whe	re. nc <	10X ca)			
				鹕	閷		FOR PLANNING	PUR	٢P	OSES	開始		er an		İ nteri i	
	TOXIC	ITY INFORM	IATION	2444.4	Procession Procession	an fallena. The Japa	CONTAMINANT	PREI		IARY REMEDIA	TION GO.	ALS (PRGs)	HE GU	SOIL SCRE	
SFo	KIDo	SFI	RIDi	· v	/skin ∋abs	CAS No	······································	Parida	-tint "	in distant		<u> </u>		×αε; [−]	Migration to Gro	ound Water
1/(mg/kg·d)	(mg/kg∙d)	1/(mg/kg-d)	(mg/kg-d)	č	soils	endition and a	าก และ 10% และ เป็นเหตุลายัง (และสาวเหตุลายัง) และ (กรุง (กรุง (ประกูลาย์เป็นกรุงสายไม่เป็นเป็นกรุงและ เป็นเรื	Soil (m	ntiai ig/kg)	Industrial Soil (mg/kg)	Ambient Ai (ug/m^3)	ſ	Tap Water (ug/l)		EIAF 20	DAF 10 2000
L <u></u>			<u> </u>		<u></u> Ø	關基礎研究		Atrian with	Lindikingd	e werden de de la constante	Adlini i ginisi	<u>MGAING NI</u>	summer of	ll ar	Č.	
2 3E+02	3 0E-03	2 3E +02	. 305.03			07.97.6	IPagyidiag	1045.00			2.05.0r					
	4 UE+00	1 232.02	4 0E+00		0 0 1	92-87-3 65-85-0	Benzoic acid	2.1E-03	CA	1.1E-02 cs	2.9E-05	ca 2	1.9E-04	Ca	405+00	0.05+01
1 3E+01 i		1 3E+01	1	0	1 01	98-07-7	Benzotrichloride	3.7E-02	Ca	1.9E-01 ca	5.2E-04	nc i ca É	5.2E-03	^C C 1	4.0ETU2	2.02+01
	3 0E-01	h	3 0E-01	r O	01	100-51-6	Benzyl alcohol	1.8E+04	nc	1.0E+05 max	1.1E+03	nc 1	.1E+04	nc	[
17E-01	2 OF -03	17E-01	1 5 7E-06		~	100-44-7	Benzyl chloride	8.9E-01	Ca	2.3E+00 ca	4.0E-02	co 6	3.6E-02	Ca		
	1 0E 04		1 0E-04	$\frac{1}{7}$	01	141-66-2	TRidin	1.5E+02		2.2E+03 ca*	8.0E-04	<u>ca' /</u>	.3E+01	nc	6.3E+01	3.0E+00
	1 5E-02		1 5E-02	1 0	J 01	82657-04-3	Biohenthrin (Talstar)	9 2E+02	/ nc	13F+04 or	3.7E-01	nc J	5E+00	nc	i	
	5 UE-02	1	5 0E-02	1 1	1	92-52-4	1,1-Biphenyl	3.5E+02	sat	3.5E+02 sat	1.8E+02	nc J	0E+02	0c	I	
1 1E+00 i		1 2E +00	•	1		111-44-4	Bis(2-chloroethyl)ether	2 1E-01	ca	6.2E-01 ca	5.8E-03	ca 9	18E-03		4 0E-04	2 0E-05
7 0E-02 h	4 UE-02	+ 3.5E.02 F	h 4.0E-02	11		108 60-1	Bis(2-chloroisopropyl)ether	2.9E+00	a l	8.1E+00 co	1.9E-01	cu 2	2.7E-01	C 4		
7 0E-02 h	4 0E-02		1 4 05 02			542-88-1	Bis(chloromethyl)ether	1.9E-04	Ca	4.4E-04 ca	3.1E-05	<u>. 5</u>	.2E-05	C à		
1 4E 02 1	2 0E-02	1 14E-02	· 2 2E-02	- 1 - 1 	3 O I	108-00-1	DIS(2-Childro-1-methylethylether Ris(2-athylbayyl)phibalate (DEHP)	2.91+00) ca	8.1E+00 ca	1.9E-01	ca Z	7E-01	Ca		
	5 0E-02	1	5 0E 02	1 0	0 0 1	80 05 7	Bisphenol A	3.1E+03	CB"	1.8c+02 ca	4.8E+02	ca 4	.8E+00	ca		
	9 OE 02	•	5 7E 03	h 0	01	7440 42-8	Boron	5.5E+03	 	7.9E+04 nc	21E+01	ac 3	3F+03	-nc		
			2 UE-04	n 0	, 01	7637-07-2	Boron trifluoride		•••		7.3E-01	nc	.01	""		
5 25.02	2 DE -02	n 6 25 02	2 96 03	<u>n 1</u>		108-86-1	Bromobenzene	2.8E+01	nc	9.2E+01 nc	1.0E+01	nc 2	.0E+01	nc		
7 9E-03	2 06:02	1 02002 1	1 2 UE-U2		· 01	75-27-4	Bromodichioromeinane	1.0E+00) CB	2.4E+00 ca	T1E-01	ca 1	BE-01	¢.3	6.0E-01	3 0E-02
	1 4E-03	1 332.03	1 46-03	1 1		73-23-2 74-83-9	Bromomethane (Methyl bromide)	0.2E+01	C9.	3.1E+02 ca* 1.3E+01	1.7E+00	ca* 0	.5E+00	6a*	8.0E-01	4.0E-02
		*****		0	01	101-55-3	14-Bromophenyl phenyl ether		nc	1.JETUI no	5.2ETUU	nc O	./E+00	- 10	2.0E-01	1.0E-02
1	5 OE-03	h	5 0E-03	10	01	2104-96-3	Bromophos	3.1E+02	nc	4.4E+03 nc	1 8E+01	nc 1	8E+02			
	2 0E-02	<u> </u>	2 0E -02	1 0	, 01	1689-84-5	Bromoxynil	1.2E+03	nc	1.8E+04 nc	7.3E+01	nc 7	.3E+02	00		
	2 0E-02	1	2 0E -02	1 0	01	1689-99-2	Bromoxynil octanoate	1.2E+03	nc	1.8E+04 nc	7.3E+01	nc 7	3E+02	nc		
18E+00 (18E+00	1	1		106-99-0	1.3-Butadiene	3.5E-03	C8	7.6E-03 ca	3.7E-03	ca 6	i.2E-03	Ca		
}	5 0F-02	<u></u>	1 0E-01		0.1	71-36-3	1-Butanol	6.1E+03	nc nc	8.8E+04 nc	3.7E+02	nc 3	.6E+03	nc	1.7E+01	9.0E-01
1	1 0E-02	n	1 0E-02			2008-41-5	n-Butylbenzene	3.1E+U3	nc i	4.4E+04 nc	1.8E+02	nc 1	8E+03	nc		
	1 0E-02	н	1 0E-02	1 1	i	135-98-8	Isec-Butylbenzene	1.1E+02	, nc	2.4ETU2 54 2.2F+0.2 54	3.7E+01	nc 0.	1E+01	nc		
	1 OE 02	0	1 OE D2	11	,	98 06 6	lert-Bulyibenzene	1.3E+02	nc	3.9E+02 Fal	37E+01	nc 6	TETOT			
	2 0E-01	4	2 0E-01	1 0	0 1	85-68-7	Butvi benzvi phthalate	1.2E+04	i nc	1.0E+05 max	7.3E+02	nc 7	3E+03	AC	9.3E+02	8.1E+02
	1 0E+00	1	1 ÚE +00	1 0	01	85-70-1	Butylphthalyl butylglycolate	6.1E+04	nc	1.0E+05 max	3.7E+03	nc 3	.6E+04	nc		
	3 0E-03	h	3 OE -03	- i 0	01	75-60-5	Cacodylic acid	1.8E+02	nc	2.6E+03 nc	1.1E+01	nc 1	1E+02	nc		
	5 UE-04	1 63E+D0	1	0	0 00 1	7440-43-9	Cadmium and compounds	3.7E+01	nc	8.1E+02 nc	1.1E-03	ca 1	.8E+01	nc	8.0E+00	4.0E-01
·	5 DE-01		5 0E-01		01	105-60-2	Caprolactam	9.02+00		1 06+05	1 951/19		00104			
86E-03 h	2 0E-00	i 86E-03	2 0E-03	1 0	J 01	2425-06-1	Captafol	5.7E+01	- nc	2.9E+02 ca**	7.8E-01	nc 1.	8E+04	nc		
3 5E-03 h	1 36 01	i 0.5E.00	/ 1 JE-01	1 0	01	133-06-2	Captan	1.4E+02	. ca*	7.0E+02 .	1.9E+00	a 1	9E+01			
Γ	1 01: 01	1	1 16 01	1 0	01	63 25 2	Carbaryl	6.1E+03	ne	8 812+04 nc	4.0E+02	nc 3	6E+03	AC .		
20E-02 n	5.0E-03	2 UE 02	1 6.05.03	0	01	86-74-8	Carbazole	2.4E+01	C N	1.2E+02 •	3.4E-01	ca 3	.4E+00	Ca	6.0E-01	3.0E-02
	1 DE-01		2 OE -01			75-15-0		3.12+02	. nc	4.412+03 nc	795107	nc I	.8E+02	<u>nc</u>		
13E01 I	7 OE -04	1 531∈02	1 / UE-04			56-23-5	Carbon tetrachloride	2.4E-01	. nc	5.3E-01 cat	1 3E-01	nc 1.	.0E+03	nc cat	3 2E+01 7 0E-02	202+00
	1 UE-02	1	1 OE-02	1 0	/ 01	55265-14-8	Carbosulfan	6.1E+02	i nc	8.8E+03 nc	3.7E+01	nc 3	.6E+02	00	1.06-02	3.02-03
	1 06-01	1	1 0E-01	r 0	01	5234-68-4	Carboxin	6.1E+03	nc	8.8E+04 nc	3.7E+02	nc 3	6E+03	nc		
4.05.01	1 5E-02	1	1 5E-02	1 0	01	133-90-4	Chloramben	9.2E+02	nc	1.3E+04 nc	5.5E+01	nc 5	.5E+02	nc		
3.5E-01	5.0F-04	4 UE-01	1 2 0E 04	<u> </u>	01	118-75-2		1.2E+00	<u> </u>	6.1E+00 ca	1.7E-02	<u>ca</u> 1	.7E-01	C 8		
	2 0E-02	1 332.01	2 UE 02		0 0 U 0 1 0 1	90982-32-4	Chloringeron-ethyl	1.0E+00) CB*	1.1E+01 ca*	1.9E-02	· · 1	3E-01	·••	1.0E+01	5.0E-01
	1 0E-01	1	5 7E-05	n		7782-50-5	Chlorine	1.20.00	AC		2 1F-01	AC /.	.3E+U2	NC		
			57E-05			10049-04-4	Chlorine dioxide			· · · · · · · · · · · · · · · · · · ·	215-01	nc		-+		
1				1	1	107-20-0	Chloroacetaldehyde				4.1			1		
	2 0E-03	h	2 0E-03		J 0.1	79-11-8	Chloroacetic acid	1.2E+02	nc	1.8E+03 nc	7.3E+00	nc 7.	.3E+01	AC		

- -

Key I=IRIS h	TOXIC	A X=WIHDRAW	N o=Other EPA C			ROUTE EXTRA	FOR PLANNING F				SULL (where ATION GOA	ALS (PRGs	() () () () ()	SOIL SCRE	ENING LEVELS
` SFo 1/(mg/kg-⊄)	R1Do (mg/kg-d)	SFi 1/(nig/kg-d)	RfDi (mg/kg-d)	- v 0 c	/ skin) abs ; soils	CAS No		Residen Soil (mg	tial /kg)	Industrial Soil (mg/kg)	Ambient Air (ug/m*3)	Tap W (ug	aler [7]] /)	Migration to Gro (JAF 20 (Mg/kg)	Dund Water DAF 1 (My/kg) (My/kg)
					1911.11.190			A	0.7.9 4 74440				ALLANNE POS		
	8 6E-06	1	8 6E-06	1 1		532-27-4	2-Chloroacetophenone	3.3E-02	nc	1.1E-01 nc	3.1E-02	nc 5.2E-0	J2 nc	7 05 01	2 05 02
	4 0E-03	•	4 0E-03	1 0	01	106-47-8	4-Chioroaniline Chiorobenzene	2.4E+02	nc	3.5E+U3 Ac	1.5E+01	nc 1.512+1	JZ nc D2 nc	1.0E+00	7 0E-02
2.7E-01	2 0E-02	2 7E-01	h 2.0E-02	<u> </u>	01	510-15-6	Chlorobenzilate	1 8F+00		9 1F+00 ca	2 5E-02	- 2 5E-C	<u>1 a</u>	1.02.00	1.02.02
	2 0E-01	h	2 ÚE -01	1 0	01	74-11-3	p-Chlorobenzoic acid	1.2E+04	nc	1.0E+05 max	7.3E+02	nc 7.3E+0	03 nc	1	
	2 OE-02	h	2 UE -02	1 0	01	98-56 6	4-Chlorobenzolrifluoride	1.2E+03	nc	1.8E+04 nc	7.3E+01	nc 7.3E+(02 nc		
	2 0E-02	h	2 0E -03	h 1		126-99-8	2-Chloro-1.3-butadiene	3.6E+00	n¢	1.2E+01 nc	7.3E+00	nc 1.4E+0	01 nc		
	4 6E-01	h	4 UE-01	1 1		109-69-3	1-Chlorobutane	4.8E+02	sat	4.81=+02 sat	1.5E+03	nc 2.4E+0	JJ nc		
	1 4E+01	- <u></u>	1 4E+01	. 1		75-68-3	Chlorodilluoromethane	3.46+02	531	3.4ETU2 681	5.2E+04	NC 0.7ET			
295.03	0 4.05.01	0 295.03	140.00			75-45-0	Chloroethane	3 0E+00	581	6.5E+00 cm	2.3E+00	nc 0.3L+1	00 0		
			1 152.00			110.75-8	2-Chloroethyl vinyl ether	0.02.00		0.02.00 11	2.02.00			1	
6 1E-03	1 0E-02	1 8 1E-02	1 8 6E-05	n 1		67-66-3	Chloroform	2.4E-01	ca**	5.2E-01 cam	8.4E-02 d	- 1.6E-C	J1 ca**	6 0E-01	3.0E-02
1 3E-02	n	6 3E 03	h 86E-02	n 1	i i	74-87-3	Chloromethane	1.2E+00	ca	2.7E+00 ca	1.1E+00	ca 1.5E+6	00 ca	1	
58E-01	h	5 8E-01	1	0	01	\$5-69-2	4-Chloro-2-methylaniline	8.4E-01	ca	4.3E+00 ca	1.2E-02	ca 1.2E-0)1 ca		
46E01	h	4 6E 01	1	0	0 1	3165-93-3	4-Chloro-2-methylaniline hydrochloride	1.1E+00	C8	5.4E+00 ca	1.5E-02	a 1.5E-0)1 ca		
255.02	8 UE-02	2 55 02	8 UE -02	. 1		91-58-7	Deta-Unioronaphinalene	3.9E+03	00	2.7E+04 nc	2.9E+02 2.7E-01	nc 4.9E+0	JZ nc 01 cm		
1 8E-02	h	1 8E-02	;			100-00-5	n-Chloronitrobenzene	1 1E+01		3.2E+01 ca	3.7E-01	6.2E-0	ji		
	5 0E-03		5 0E-03	1 1		95-57-8	2-Chlorophenol	6.3E+01	AC	2.4E+02 nc	1.8E+01	nc 3.0E+6	01 nc	4.0E+00	2.0E-01
	2 9E-02		2 9E-02	h 1	1	75-29-6	2-Chloropropane	1.7E+02	AC	5.9E+02 nc	1.0E+02	nc 1.7E+0	02 nc		
1 1E-02	h 15E-02	i 1 1E-02	r 15E-02	1 0	0 1	1897-45-6	Chlorothalonil	4.4E+01	ca*	2.2E+02 ca*	6.1E-01	6.1E+	00 ca.		
	2 0E-U2	1	2 0E-02	1 1	1	95-49-8	o-Chlorololuene	1.6E+02	AC	5.7E+02 nc	7.3E+01	nc 1.2E+	02 nc		
	2 0E-01		2 UE-01	1 0	0 1	101-21-3	Chiorpropham	11.22+04	nc	1.0E+05 max	7.32+02	nc 7.3E+	03 nc		
	3 0E-03	1	3 0E-03	(0	0 0 1	2921-88-2	Chlorpyrilos Chlorpyrifos matbul	1.00002	nc	2.0E+03 nc	1.1ETU1 3.7E±01	AC 1.1ET	02 nc		
1	1 UE-02	n	1 0E-02		0 01	5598-13-0	Chlorsulfuron	3 1E+03	NC 00	44F+04 m	1 8E+02	ac 3.02+	03		
	8 0E 04	h	8 OE 04		0 01	60238-56-4	Chlorthiophos	4 9E+01		7.0E+02 nc	2.9E+00	nc 2.9E+	01 00		
		4 2E+01	1	6	0		Total Chromium (1:6 ratio Cr VI:Cr III)	2.1E+02	CA	4.5E+02 ca	1.6E-04	ca		3.8E+01	2.0E+00
	1 5E+00	1				16065-83-1	Chromium III	1.0E+05	តាន×	1.0E+05 max	0.0E+00	5.5E+	04 nc	1	
	3 OE 03	1 2 9E + 02	1		0	18540 29 9	Chromium VI	3 0E+01	ca	6.4E+01 ca	2.3E-05	ca 1.1E+	02 nc	3.8E+01	2.0E+00
							"CAL-Modified PRG" (PEA. 1994)	2 0E-01		4.05.05		1.6E-0	J1		
	6 0E 02	n 			<u> </u>	7440-48-4	Coball	4.76703	nc	1.0E+03 mix	7 1 - 04	2.201			
	3.75.00	2 26 100	1		u n	8007-45-2 7440-50-8	Conner and compounds	2 9E+03	05	7.6E+04 ac	J. 12-03	1 4F+	03		
1 9E+00	572.02	1 9E+00	,		0 1	123-73-9	Crotonaldehyde	5.3E-03	ca	1.1E-02 ca	3.5E-03	ca 5.9E-0	03		
	1 0E-01	+	1 1E-01	. 1	1	98 82 8	Cumene (isopropylbenzene)	1.6E+02	n¢	5.2E+02 nc	4.0E+02	nc 6.6E+	02 nc		
8 4E-01	h 2.0E-03	h 84E01	1 2 0E-03	1 (0 0 1	21/25 46-2	Cvanazine	5.8E-01	Câ	2.9E+00 ca	8.0E-03	ca 8.0E-6	02 🕫		
	2 OE-02	•	8 6E 04	+	1	74-90-8	Cyanide and compounds	1.1E+01	nc	3.5E+01 nc	3.1E+00	nc 6.2E+	00 nc		
	4 0E 02	1	4 0E -02	1	1	460-19-5	Cyanogen	1 3E+02	nc	4.3E+02 nc	1.5E+02	nc 2.4E+	02		
	9 0E-02		9 0E-02		1	506-68-3	Cvanogen chloride	1 6E+02	. nc	54E+02 m	1.8E+02	m 3.0E+	02		
	5 7E+00		5 7E+00	<u> </u>	1	110.82.7	Cyclohexane	1.4E+02	sat	1.4E+02 sat	2.1E+04	ac 3.5E+	04		
1	5 0E+00		5 0E+00		0 01	108-94-1	Cvclohexanone	1.0E+05	max	1.0E+05 max	1.8E+04	nc 1.8E+	05 nc		
1	2 0E-01		2 OE-01		0 01	108-91-8	Cyclohexylamine	1.2E+04	nc	1.0E+05 max	7.3E+02	nc 7.3E+	03 nc		
	5 0E-03	ı	5 0E-03		0 01	68085-85-8	Cyhalothrin/Karate	3.1E+02	nc	4.4E+03 nc	1.8E+01	nc 1.8E+	02 nc		
	1 UE-02	•	1 UE-02	r	0 01	52315-07-8	Cypermethrin	16.1E+02	2 nc	8.8E+03 ac	3.7E+01	nc. 3.6E+	02 nc		
	7 5E-03		/ 5E-03		0 01	66215-27-8	Dachal	16 1E+02	. nc	8 8E+03 hc	375+01	nc 2.7ET	02 10		
1	3 DE-02		3 0E-02		0 0.1	75-99-0	Dalapon	1.8E+03	. nc	2.6E+04 nc	1.1E+02	nc 1.1E+	03 nc		
	2 5E-02	1	2 5E-02	1	0 01	39515-41-8	Danitol	1.5E+03	nc l	2.2E+04 nc	9.1E+01	nc 9.1E+	02 nc		
2 4E-01	•	2 4E-01	1		0 0 03	72-54-8	DDD	2.4E+00) ca	1.7E+01 ca	2.8E-02	ca 2.8E-0	01	1.6E+01	8 0E-01
3 4E-01	1	3 4E 01	1		0 0 03	72-55-9	DDE	1.7E+00) ca	1.2E+01 ca	2.0E-02	a 2.0E-0	01	5.4E+01	3.0E+00
3 4E-01	1 5 0E-04	1 34E-01	1 5 0E-04	1	0 0.03	50-29-3	IDDT	1.7E+00) (8*	1.2E+01 ca*	2.0E-02	-2.0E	01 a*	3.2E+01	2.0E+00

			St.	DOCOME	INIS (BERGUIE EXTRA	POLATION CE-CANCER PRG ACENONCANCER PRG SITESOIL SATURA	TION max	=CEILING	S LIMIT *(where: nc <	100X ca) **(wher	e: nc < 10X ca)	NE COL	
an da i		이 가지 않는 것이 같이 있다. 이 가지 않는 것이 같이 있는 것이 같이 있는 것이 같이 있는 것이 같이 있다.		ri ista			FOR PLANNING F	וטי	RP	POSES	Share	1416426281		法国际和保证
読み機論で	See 14	tadi dhi i ku		<u></u>	11102						MEAN	F-BALLID PC	20.610 . A	
1	TOXIC	ITY INFORM	ATION				CONTAMINANT	PRI	ELIMI	NARY REMEDI	ATION GO	ALS (PRGs)	SOIL SCRI	ENING LEVELS
SFu	RfDo	SFi	RIDi	V s O a	ikin ibs	CAS No.	· · · · · · · · · · · · · · · · · · ·	Rese	dential	Industrial	Ambient Ai	Tap Water	Migration to G	ound Water
1/(mg/kg-d)	(mg/kg∙d)	l/(ing/kg-d)	(mg/kg-d)	C s	oils	Martina		Soil	(ពរម្ូ/kg)	Soil (mg/kg)	(ug/m^3)	(ug/l)	(mg/kg)	(mg/kg)
	<u> </u>		6*****	, i pilijania sji	ia _{l te} 63	and the second second		. ૪ વર્ષી)	in Na Airtain	and Provide a children of the second	Malar del ceretari	and the second	arma and a second s	net belles of BARRA
	1 0E-02	1	1 0E 02	1 0	01	1163-19-5	Decabromodiphenyl ether	6.1E+	02 <u></u>	8.8E+03 ac	3.7E+01	ac 3 6E+02 c	<u>.</u>	
	4 DE 05		4 0E-05	1 0	01	8065-48-3	Demeton	2.4E+	00 nc	3.5E+01 nc	1.5E-01	nc 1.5E+00 r	c	
6 1E-02	h 0.05.04	6 1E -02		0	01	2303-16-4	Dialiate	8.0E+	00 ce	4 0E+01 ca	1.1E-01	ca 1.1E+00 c	•	
	4 UE-U3	n n	9 0E-04 4 0E-03	r 0 F 1	01	333-41-5 132-64-9	Dibenzoturan	2.9E+	01 nc 02 nc	7.9E+02 nc 5.1E+03 nc	3.3E+00 1.5E+01	nc 3.3E+01 r	c	
	1 UE-02		1 UE U2	10	01	106-37-6	1.4-Dibromobenzene	6.1E+	02 nc	8.8E+03 ne	3.7E+01	nc 3.6E+02		
8 4E-02	2 DE-02	8 4E 02	2 0E-02	r 1		124-48-1	Dibromochloromelhane	1.1E+	00 ca	2.7E+00 c+	8.0E-02	ca 1.3E-01 c	4 0E-01	2.0E-02
146+00	n 57E-05	4 24E-03 1	57E-05	• •		96-12-8	1.2-Dibromo-3-Chloropropane	4.5E-0	01 ca*	• 4.0E+00 ca**	2.1E-01	nc 4.8E-02 ci		
8 5E+01	5 7E-05	1 7 7E-01	5 7E-05	b 1		106-93-4	1 2-Dibromoethane	6 9F-0	13	4 8F-02	9.0E-04	4.7E-03		
	1 0E-01	4	1 0E-01	1 0	01	84.74.2	Dibutyl ohthalate	6.1E+	03 AC	8.8E+04 ec	3.7E+02	AC 3.6E+03	2.3E+03	2.7E+02
	3 0E-02	4	3 0E -02	1 0	01	1918-00-9	Dicamba		nc	2.6E+04 nc	1.1E+02	nc 1.1E+03 m	c	
	9 DE -02		5 7E-02	h 1		95-50-1	1.2-Dichlorobenzene	3.7E+	02 sai	3.7E+02 sat	2.1E+02	nc 3.7E+02 n	c 1.7E+01	9.0E-01
2 4E 02	9 UE-04	n n 2.25.02 /	9 0E-04			541-73-1	1.3-Dichlorobenzene	1.3E+	01 nc	5.2E+01 nc	3.3E+00	nc 5.5E+00 n		1 05 04
4 5E-01	, , , , , , , , , , , , , , , , , , , ,	4 5E 01			01	91-94-1	13 3-Dichlorobenzidine	1 1 F +	00 1	5.1E+00 ce	3.10-01	a 5.0E-01 a	2.0E+00	3.05-01
	3 OE -02	0	3 DE -02	r	01	90-98-2	4.4'-Dichlorobenzophenone	1.8E+	03 nc	2.6E+04 nc	1.1E+02	nc 1.1E+03 m		002 04
6 3E+00		6 3E+00 1	h	1		764-41-0	1,4-Dichloro-2-butene	7.9E-0	03	1.8E-02 co	7.2E-04	co 1.2E-03 c	•	
	2 0E-01	1	5 7E-02	h 1		75-71-8	Dichlorodilluoromelhane	9.4E+	01 nc	3.1E+02 nc	2.1E+02	nc 3.9E+02 m	· 0.25+01	4.05.00
5 7E-03	100-01	5 7E-03	140-01	1		73-34-3	"CAL-Modified PRG"	3 3E+	02 nc 00 ca	7 1F+00 ca	5.2E+02	a 20E+02 a	c 2.3E+UI	1.02+00
9 1E-02	3 0E-02	n 91E-02	1 4E-03	n 1		107-06-2	1,2-Dichloroelhane (EDC)	3 5E-0	01	7.6E-01 c	7.4E-02	ca* 1.2E-01 c	2.0E-02	1 0E-03
6 0E-01	9 0E-03	1 8E-01	9 0E-03	7-1		75-35-4	1.1-Dichloroethylene	5.4E-0	02 ca	1.2E-01 cm	3.8E-02	4.6E-02 c	6.0E-02	3.0E-03
	1 0E-02	h	1 0E-02	11		156 59-2	1,2-Dichloroethylene (cis)	4.3E+	01 nc	1.5E+02 nc	3.7E+01	nc 6.1E+01 r	<u>4.0E-01</u>	2.0E-02
	2 05 02	,	2 UE 02 3 UE 03	(1 (0		156-60 5	2.4-Dichlorophenol	1 954	01 nc 02	2.1E+02 nc	7.3E+01	nc 1.2E+02 r	· / /.UE-01	3.0E-02
	8 0E-03		8 0E-03	10	01	94-82.6	4-(2,4-Dichlorophenoxy)butyric Acid (2,4-DB)	4.9E+	02 nc	7.0E+03 nc	2.9E+01	nc = 2.9E+02	C 1.0E+00	5.UE-U2
	1 UE-02	1	1 0E 02	1 0	0 05	94-75 7	2.4-Dichlorophenoxyacetic Acid (2.4-D)	6.9E+	02 nc	1.2E+04 nc	3.7E+01	nc 3.6E+02 r	c	
6 8E 02	h 11E-03	1 6 8E 02	1 11E-03	1 1		78-87-5	1.2-Dichloropropane	3.5E-0	01 ca*	7.7E-01 a*	9.9E-02	ca 1.6E-01 c	- 3.0E-02	1.0E-03
	3 0E-02	1 142-02	3 0F-03		0.1	516.23.0	2 3-Dichloropropanol	1 9E1	01 6	1.0E+00 cm	4.88-01	ca 4.0E-01 c	4.0E-03	2.0E-04
2 9E-01	1 5 0E-04	2 9E-01	1 14E-04		01	62-73-7	Dichlorvos	1 7E+	02 nc	8.5F+00 m	2 3E-02	a 23F-01		
4 4E-01	x	4 4E-01	1	0	01	115-32-2	Dicotol	1.1E+	00 ca	5.6E+00 ca	1.5E-02	a 1.5E-01 a		
	3 0E-02	h	5 7E-05	h 1		77-73-6	Dicyclopentadiene	5.4E-0	01 nc	1.8E+00 nc	2.1E-01	nc 4.2E-01 r	c	
1 6E+01	1 5 0E-05	16E+01	50E-05	10	01	60-57-1	Diethylene alycol, monobytyl ether	3.0E-0	02 na	1.5E-01 ce	4.2E-04	a 4.2E-03 a	4.0E-03	2.0E-04
	2 DE + 00		2 0E+00	, 0	01	111.90.0	Diethylene divcol, monoelbyl ether	10F+	02 AC	10E+05 m	736+03	nc 2.1E+U2 r	c	
	1 1E-02	h	1 1E-02	1 0	01	617-84-5	Diethylformamide	6.7E+	02 110	9.7E+03 nc	4.0E+01	nc 4.0E+02 r		
1 2E-03	1 6 0E-01	i 1.2E-03	6 0E-01	1 0	01	103-23-1	Di(2-ethylhexyl)adipate	4.1E+	02 ca	2.1E+03 ca	5.6E+00	ca 5.6E+01 c	a	
4.7E+03	8 OE -01	4 76 - 03	8 CC -01	10	01	84 66 2	Diethyl phinalale	4.9E+	04 nc	1.0E+05 max	2.9E+03	nc 2.9E+04 r	c	
112.03	 8 0E-02	1	8 0E-02	(0	01	43222-48-6	Difenzoguat (Avenge)	4.9E+	04 ce 03 no	7.0E+04 c	2.9E+02	a 1.4E-05 a		
	2 0E-02	+	2 0E-02	1 0	01	35367-38-5	Dillubenzuron	1.2E+	03 nc	1.8E+04 nc	7.3E+01	nc 7.3E+02 n		
	1 16+01	I.	1 IE+01	i 1		75-37-6	1.1-Difluoroethane				4.2E+04	nc 6.9E+04 m	c	
	2 0E-02	n	2 0E-02	<u> </u>	01	28553-12-0	Dilsononyl phthalate	1.2E+	03 nc	1.8E+04 nc	7.3E+01	nc 7.3E+02 r	c	
	2 0E-02	1	8 UE-02	, U	0.1	1445-75-6	Directhioin	1 25+	03	1.0E+04 nc	2.92+02	nc 2.9E+03 m	c	
	2 0E-04		2 0E-04		01	60-51-5	Dimethoate	1.2E+	01	1.8E+02 nc	7.3E-01	nc 7.3E+00 m		
1 4E-02	h	1 4E 02	1	0	01	119-90-4	3.3'-Dimethoxybenzidine	3.5E+	01 c	1.8E+02 ce	4.8E-01	c. 4.8E+00 c		
	5 7E-06		57E-06	x 1	• -	124-40-3	Dimethylamine	6.7E-	02 ng	2.5E-01 nc	2.1E-02	nc 3.5E-02 n	۰]	
7.55-01	2 UE-03	7.55.01	2 UE-03	·(0	0.1	95.68.1	24.Dimethylandine	1.2E+	02 nc	1.82+03 nc	1.32+00	nc /.3E+U1 n	<u> </u>	
5 8E-01		5 8E-01	•	0	01	21436-96-4	2.4-Dimethylaniline hydrochloride	8.4E-	01	4.3E+00	1.2E-02	a 12F-01		
9 2E+00	h	9 2E+00	1	0	01	119-93-7	3,3'-Dimethylbenzidine	5.3E-	02 c	2.7E-01 a	7.3E-04	• 7.3E-03 •		

S J Simucher

.

Key i=1RIS h=	HEAST #=NCEA	x=WITHDRAWN	o=Other EPA D	OCUME	NTS r=	ROUTE EXTRA	POLATION DESECANCER PRG INCENONCANCER PRG SITESOIL SATURA	TION max=CE	LING	LIMIT "(where: nc < 10	OX ca) **(where	nc <	10X ca)		
		۵۵ کې د ۲۰ ۲۰ کې د ۲۰ کې د ۲۰ ۲۰ کې د ۲۰ کې د ۲۰					FOR PLANNING F	PUR	Ρ	OSES	S IM S		a an chui		d'Adha
र विभिन्न सम्बद्ध	1021 — 422 ТОХІСІ			a zadat	2.9.1.2	and a state	CONTAMINANT	PRELI	MIN	ARY REMEDIA	TION GOA	LS (PRGs)	SOIL SCREE	NING LEVELS
				V s	kin .									Migration to Grou	ind Water
> SFo	RIDo (ovultra.dt)	SFi Marchined	RIDi	0 a	bs	CAS No.		Resident Soit (ma	tiat /ko)	industrial Soit (mo/kg)	Ambient Air (ug/m^3)		Tap Water; 21 (ug/l)	DAF 20	DAF 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
n(mg/kg-a)	(mg/kg·a)	windwâ.at	(119/69-0)	C 5	0115 191	制物的制度的	sense hitte water and the sense of the sense		i da seco	nanta si seko san fili	www.ce.ac.com.wi	kina.			
L						****									
2 6E+00 ×		3 5E+00 ×		0	01	57-14-7	1.1-Dimethylhydrazine	1.9E-01	ca	9.5E-01 ca	1.9E-03 c	- 2	2.6E-02 ca		
376+01 ×		37E+01 ×		0	01	540-73-8	1.2-Dimethylhydrazine	1.3E-02	ca	6.7E-02 ca	1.8E-04 c	. 3	.8E-03 ca		
}	1 0E-01	n	8 6E-03	1 0	01	68-12-2	IN N-Dimelinykolmankide	6 1E+01	0C	8.8E+02 ac	37E+00	- 3	6E+01 nc		
1	2 05-02		2 0E-02	. 0	01	122-09-8	2.4-Dimethylohenol	1.2E+03	nc	1.8E+04 nc	7.3E+01	c 7	.3E+02 nc	9.0E+00	4.0E-01
1	6 UE-04	1	6 0E-04	1 0	01	576-26-1	2,6-Dimethylphenol	3.7E+01	nc	5.3E+02_nc	2.2E+00	<u>c</u> 2	.2E+01 nc		
	1 DE 03	1	1 0E-03	1 0	01	95 65-8	3.4-Dimethylphenol	6 1E+01	nc	8.8E+02 nc	3.7E+00 #	c 3	6E+01 nc		
1	1 UE+01	* .	1 0E+01	1 0	01	131-11-3	Dimethyl ohthalate	1.0E+05	max	1.0E+05 max	376+04	c 3	6E+03 ac		
	1 0E-01	•	1 0E-01	1 0	01	120-61-6	Manual Contraction of the second seco	1725+02	AC	1.8E+03 ac	73E+00	- 7	3E+01 oc		
	2 UE-U3	н Б. ²	2 0E-03 4 0F-04		0.1	528-25-0	1 2-Dinitrobenzene	2.4E+01	00	3.5E+02 nc	1.5E+00	c 1	.5E+01 nc		
	1 DE-04		1 DE-04	1 0	01	99-65-0	1,3-Dinitrobenzene	6.1E+00	nc	8.8E+01 nc	3.7E-01	c 3	.6E+00 nc		
	4 0E-04	h	4 0E-04	1 0	01	100 25 4	1.4-Dinitrobenzene	2.4E+01	nc	3.5E+02 nc	1.5E+00 r	ic 1	.5E+01 nc		
	2 0E 03		2 UE-03	10	01	51-28-5	2.4-Dinitrophenol	1.2E+02	NC	1.8E+03 nc	7.3E+00	с 7	.3E+01 nc	3.0E-01	1.0E-02
6 BE 01		<u>เช่8E</u> -01 เ		0	01	25321-14-6	Dinitrotoluene mixture	7.2E-01	ca	3.6E+00 ca	9.92-03 6		99E-02 ca	8.0E-04	4 0E-05
ļ	2 OF 03	1	2 06 03	1 0	01	121 14 2	12.4-Dinitrololuene (see Dinitrololuene mixture)	1.2E+02	nc	8.8E+02 mc	37E+00	ic i ic 3	6E+01 nc	7.0E-04	3.0E-05
	105-03	n	106-03	10	01	88 85 7	Dinoseb	6.1E+01	nc	8.8E+02 nc	3.7E+00	. 3	.6E+01 nc		
	2 0E-02	h	2 UE 02	1 0	01	117-84-0	di-n-Octyl phthalate	1 2E+03	nc	1.0E+04 sol	7.3E+01	ic 1	.3E+02 nc	1.0E+04	1 0E+04
1 1E 02	ı.	1 1E-02 i		0	01	123-91-1	1.4-Dioxane	4.4E+01	C2	2.2E+02 ca	6.1E-01	a 6	i.1E+00 ca		
15E+05 #	۱	15 <u></u> +05 n		0	0 03	1746-01-6	Dioxin (2,3,7,8-TCDD)	3.9E-06	C.a	2.7E-05 ca	4.5E-08		4.5E-07 ca		
	3 0E-02	4	3 0E-02	1 0	01	957-51-7	Diphenamid	1.8E+03	nc	2.6E+04 nc	1.1E+02 /	10 I	1E+03 nc		
	2 5E-02		2 5E-02	f 0 .	01	122-39-4	Dionenviamine	1 8E+01	nc	2.2E+04 nc 2.6E+02 nc	1 1E+00	16 3 16 1	1E+02 AC		
	3 UE-04	7.75.01	3 OE -04	<u> </u>	01	14-31-7	112.Diphenylbydrazine	6 1E-01		3 1E+00 ct	87E-03		8 4E-02 ca		
8 DE-U1	1 0.05.03	77E-01 I	6 0E-03		01	122:00-7	Diobenyl sulfone	5.5E+02	 AC	7.9E+03 nc	3.3E+01	10	3.3E+02 AC		
	2 2E 03		2 2E-03	1 0	01	85-00-7	Diquat	1.3E+02	06	1.9E+03 nc	8.0E+00	nc é	3.0E+01 nc		
8 6E+00	h	8 6E+00	r	0	01	1937-37-7	Direct black 38	5.7E-02	CB	2.9E-01 ca	7.8E-04	:2	7.8E-03 ca		
8 1E+00	h	8 1E+00	r	0	01	2602-46-2	Direct blue 6	6.0E-02	CB.	3.0E-01 ca	8.3E-04	a	8.3E-03 ca		
9 3E+00	h	9 3E+00		0	0.1	16071-86-6	Direct brown 95	5.2E-02	C3	2.7E-UI ca	1.20-04		1.2E-03 ca		
1	4 0E-05	•	4 OE -05	10	01	298 04 4	Distinoton	6 1E+02	nc	8.8E+03	3.76+01	nc	36E+02 m		
	1 0E-02	•	10E-02	10	01	505-29-3	Diuron	1 2E+02	nc nc	1.8E+03 as	7.3E+00	nc .	7.3E+01 ac		
	4 0E-03		4 0F-03	10	01	2439-10-3	Dodine	2.4E+02	nc	3.5E+03 nc	1.5E+01	nc	.5E+02 nc	h	
	2 0E-01	n				7429-91-6	Dysprosium	1.6E+04	nc	1.0E+05 max			7.3E+03 nc		
	6 0E-03		6 OE -03	1 0	0.1	115-29-7	Endosulfan	3.7E+02	nc	5.3E+03 nc	2.2E+01	nc i	2.2E+02 nc	1.8E+01	9.0E-01
	2 0E-02	1	2 0E -02	, 0	01	145-73-3	Endothall	1.2E+03	nc	1.812+04 nc	7.3E+01	nc	1.3E+02 nc	1.05+00	5 0E-02
	3 0E-04	1	3 0E-04	1 0	01	72-20-8	Enarin	7.6E+00	nc	2.0E+02 Ac	1.0E+00	nc nc	2.0E+00 nc	1.02.100	5.02-02
9.9E-03	5 7E 03	h 426-03	5 7E-03		0.1	106 69 6	12-Enoxybulane	3 5E+02	nc	5.0E+03 nc	2.1E+01	nc	2.1E+02 nc		
1	2 5E-02		2 5E-02	r 0	01	759-94-4	EPTC (S-Ethyl dipropylthiocarbamate)	1.5E+03	nc	2.2E+04 nc	9.1E+01	nc (9.1E+02 nc		
	5 0E-03		5 0E-03	r O	01	16672-87-0	Ethephon (2-chloroethyl phosphonic acid)	3.1E+02	nc	4.4E+03 nc	1.8E+01	nc	1.8E+02 nc		
	5 0E-04	1	5 0E-04	1 0	01	563-12-2	Ethion	3.1E+01	nc	4.4E+02 AC	1.8E+00	nc	1.8E+01 nc	1	
	4 0E-01	в	5 7E-02	, 0	01	110 80-5	2-Ethoxyethanol	2.46+04	nc	1.0E+05 mex	2.1E+02 1 1E+03	nc	1.5C+04 nc	1	
	3 0E-01	<u>h</u>	3 0E-01		01	111-15-9		1 95+04	nC	375+04	3.3E+03	nc	5.5E+03	<u> </u>	
4 85 00	9 DE-01	1	a∩⊱-01	11		141-78-0	Ethyl acrylate	2.1E-01	 Ca	4.5E-01 a	1.4E-01	ca	2.3E-01 G		
4 00 02	" 1 0E-01	4 02-02	2 9E-01	i 1		100-41-4	Ethylbenzene	2.3E+02	sal	2.3E+02 sat	1.1E+03	n¢	1.3E+03 nc	1.3E+01	7.0E-01
2 9E 03	n 40E-01	n 29E03	1 2 9E+00	i 1		75-00-3	Ethyl chloride	3.0E+00	C a	6.5E+00 ca	2.3E+00	Ca ·	4.6E+00 G	1	
	3 0E-01	n	3 UE-01	1 0	01	109-78-4	Ethylene cyanohydrin	1.8E+04	nc	1.0E+05 mex	1.1E+03	nc	1.1E+04 ac	1	
·	2 0E-02	h	2 0E-02	10	01	107-15-3	Enviene diamine	1.26+03	0 00	1.0E+U4 AC	7.36+01	nc	7 3E+02 AC		
1	2 OE +00	i	2 0E+00	1 0	01	107-21-1	Ethylene alycol monobutyl ether	3 1E+04	i maa I maa	10E+05 max	1 4 - + 04	11C		1	
105.00	50E-01	1 3.5F-01	37E+00	1 0	01	111-70-2 75-21-8	Elhylene oxide	1.4E-01	 CA	3.6E-01 cm	1.9E-02	63	2.4E-02 ca	1	
	••														

- - -

St. mmp of prop		TOXICI		N 0=Gihei EPA MATION	DOCUM		FROUTE EXTRA	FOR PLANNING F			OLIMIT 'Iwhere: nc <	Signal Parts	nc < 10X ca)	SOIL SCRE	ENING LEVEL
Tit G1 a 66.64 116.01 c 66.67 c 66.67 c 66.67 c 66.67 c 66.67 c 67.67	⊴; SFo 1/(mg/kg-d)	RíDo (mg/kg-d)	SFi 1/(mg/kg-d)	RíDi (mg/kg-d)	- v 0 c	skin abs soils	CAS No.		R S	esidential oil (mg/kg)	Industrial Soil (mg/kg)	Ambient Air (ug/m^3)	Tap Water (Migration to Gro DAF 20 (mg/kg)	und Water, DAF 122(1) (mg/kg)
116:01 16:03 16:01 <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>• • • • • • • • •</td><td></td><td></td><td></td></t<>												• • • • • • • • •			
store store <th< td=""><td>116-01</td><td>206-01</td><td>1 11E-01</td><td>r 80E-05</td><td>10</td><td>01</td><td>96-45-7</td><td>Ethyleher</td><td>4.4</td><td>⊑+00 ca E+03</td><td>" 2.2E+U1 ca"</td><td>0.1E-02 ca**</td><td>1 25+03</td><td>1</td><td></td></th<>	116-01	206-01	1 11E-01	r 80E-05	10	01	96-45-7	Ethyleher	4.4	⊑+00 ca E+03	" 2.2E+U1 ca"	0.1E-02 ca**	1 25+03	1	
10 10<		9 0E-02	н h	9 0E-02			97-63-2	Ethyl melhacivlate	1.0	E+03 si F+02 si	14F+02 sat	3 3F+02 m	5.5F+02 ac	i i	
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		1 0E-05		1 0E-05	1 0	01	2104-64-5	Ethyl p-nitrophenyl phenylphosphorothioate	6.1	E-01 0	8.8E+00 ac	3.7E-02 ac	3.6E-01 nc		
4 6 € 0. * 6 € 0. * 6 € 0. * 9 € 0. 0 1 37202 * 7.0 € 103 2.0 € 102 # 7.0 € 103 # 2.0 € 101 # 2.0 € 112 # 1 3 € 02 1 3 € 02 1 3 € 02 0 01 225 € 40 Findminbos 7 5 € 102 # 1 1 € 40 # 4.7 € 10 # 4.7 € 10 # 4.7 € 10 # 4.7 € 10 # 4.7 € 10 # 4.7 € 10 # 4.7 € 10 # 4.7 € 10 # 4.7 € 10 # 4.7 € 10 # 4.7 € 103 # 4.7 € 102 # 2.0 € 10 # 4.0 € 11 # 4.0 € 11 # 4.0 € 11 # 4.0 € 11 # 4.0 € 11 # 4.0 € 11 # 4.0 € 11 # 4.0 € 11 # 4.0 € 11 # 4.0 € 11 # 4.0 € 11 # 4.0 € 11 # 4.0 € 11 # 4.0 € 11 # 4.0 € 11 # 4.0 € 11 # 4.0 € 11 # 4.0 € 11 # #		3 0E+00	•	3 OE • 00	1 0	01	84-72-0	Ethylphthalyl ethyl glycolate	1.0	E+05 m	x 1.0E+05 max	1.1E+04 nc	1.1E+05 ac	\$	
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		8 0E-03	1	8 UE -03	10	01	101200-48-0	Express	4.9	E+02 n	c 7.0E+03 nc	2.9E+01 nc	2.9E+02 nc		
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		2 5E 04	•	2 5E-04	1 0	01	22224-92-6	Fenamiphos	1.5	E+01 n	c 2.2E+02 nc	9.1E-01 nc	9.1E+00 nc		
1 0		1 3E-02	•	1 3E-02	10	01	2164 17-2	Fluometuron	17.9	E+02 n E⊥02	c 1.1E+04 nc	4./E+01 nc	4.7E+02 nc		
2 86:00 - 3 66:00 - 2 66:00 - 7 36:00 1 <th< td=""><td></td><td>8 DE-02</td><td><u>.</u></td><td>8 05 02</td><td></td><td>0.1</td><td>50755 60 4</td><td></td><td>13.1</td><td>E+03 A</td><td>- 70E+04 nc</td><td>201-+02</td><td>2.2E+03 AC</td><td>[</td><td></td></th<>		8 DE-02	<u>.</u>	8 05 02		0.1	50755 60 4		13.1	E+03 A	- 70E+04 nc	201-+02	2.2E+03 AC	[
0.66.92 0.66.92 0.1 0.932/06.5 Fluidani 37/E+03 5/E+04 2/E+07 a 2/E+03 a 356:00 1.06:01 1.06:00 r 0.1 0.00 Fluidani 61 14/E+02 a 18/E+03 a 7/E+03 a 18/E+01 a 18/E+01 <t< td=""><td></td><td>2 0E-02</td><td></td><td>2 0E-02</td><td>10</td><td>01</td><td>56425-91-3</td><td>Flurorimidol</td><td>12</td><td>E+03 a</td><td>18F+04 m</td><td>7 3E+01 m</td><td>7.3E+02 oc</td><td></td><td></td></t<>		2 0E-02		2 0E-02	10	01	56425-91-3	Flurorimidol	12	E+03 a	18F+04 m	7 3E+01 m	7.3E+02 oc		
106 40 100 10 106 40 100 10<		6 0E-02	1	6 0E -02	1 0	01	66332-96-5	Flutolanil	3.7	E+03 n	5.3E+04 nc	2.2E+02 nc	2.2E+03 nc	1	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		1 0E-02	1	1 0E-02	(0	01	69409-94-5	Fluvalinate	6.1	E+02 n	8.8E+03 nc	3.7E+01 nc	3.6E+02 AC		
116:01 16:01 10:01 7217020 Formesalen 2.66+00 c 3.26+01 c 3.56+02 c 3.56+01 c c 5.56+01 c 5.56+01 c 5.56+01 c 5.56+01 c 5.56+01 c 5.52+01 c 6.16+00 c 1.16+02 c 1.16+02 c 6.16+00 c 1.16+02 c	3 5E-03	1 0E-01	3 5E-03	1 0E-01	r 0	01	133-07-3	Folget	1.4	E+02 🖬	• 7.0E+02 ca	1.9E+00 ca	1.9E+01 a	1	
206:00 . 206:00 . 0 01 944:20 Fonlos 1224:00 nc 7.3E+00 nc 7.3E+	19E-01	·	1 9E-01	1	٥	01	72178-02-0	Fomesafen	2.6	ء E+00	1.3E+01 ca	3.5E-02 ca	3.5E-01 ca	l	
126.00 1.06.00 1.06.00 me 1.06.00 1.06.00		2 0E-03	•	2 OE -03	1 0	01	944-22 9	Fonotos	1.2	E+02 n	c 18E+03 nc	7.3E+00 nc	7.3E+01 nc	1	
36:00 30:00 10:00 <th< td=""><td>1</td><td>1.5E-01</td><td>1 101 02</td><td>206.00</td><td>0</td><td>01</td><td>50 00 0</td><td>Formic Acid</td><td>1 0</td><td>E+03 n E+05 m</td><td>: 1.0E+05 ms</td><td>7 3E+03</td><td>5.5E+03 mc</td><td>1</td><td></td></th<>	1	1.5E-01	1 101 02	206.00	0	01	50 00 0	Formic Acid	1 0	E+03 n E+05 m	: 1.0E+05 ms	7 3E+03	5.5E+03 mc	1	
3 0E 001 6 0E 00 h 1 70130 From 113 From 113 5 0E 103 still 5 0E 103		3 0E • 00	1	3 0E + 00	10	01	39146-24-8	Fosetyl-al	110	E+05 m	x 1.0E+05 max	1.1E+04 nc	1.1E+05 ac		
10E 03 1 100 00 Furan 2.5E+00 ne 8.5E+00 ne 6.1E+00 ne 6.1E+00 ne 3 8E+00 h 3 8E+00 r 0 0 1 97458 Furan 1.3E+01 ne 6.5E+01 ne 1.8E+02 ne 1.8E+02 ne 1.8E+02 ne 1.1E+02 ne 1.3E+04 ne <td></td> <td>3 0E+01</td> <td>1</td> <td>6 6E + 00</td> <td>h i</td> <td></td> <td>76-13-1</td> <td>Freon 113</td> <td>5.6</td> <td>E+03 s</td> <td>1 5.6E+03 sat</td> <td>3.1E+04 nc</td> <td>5.9E+04 nc</td> <td>l</td> <td></td>		3 0E+01	1	6 6E + 00	h i		76-13-1	Freon 113	5.6	E+03 s	1 5.6E+03 sat	3.1E+04 nc	5.9E+04 nc	l	
3 8E+00 h 3 8E+00 i 0 01 674-54 Furdizolidone 13E-01 nc 13E-01 nc 13E-02 nc 1.8E+02 nc 1.8E+03		1 OE-03	1	1 UE-03	1-1		110-00-9	Furan	2.5	E+00 n	c 8.5E+00 nc	3.7E+00 nc	6.1E+00 nc	1	
30603 14602 n 0 0 9801-1 Furfural 1.8E+02 nc 2.2E+01 nc 1.1E+02 nc 506:01 n 506:01 r 0 0 1 5322.6 1.8E-03 nc 5.2E+01 nc 1.2E+02 nc 1.8E-03 nc 1.8E+02 nc 1.8E+02 nc 1.8E+02 nc 1.8E+02 nc 1.8E+04 nc 1.2E+01 nc 2.2E+01 nc 2.2E+01 nc 2.2E+01 nc 2.2E+01 nc 3.2E+01 nc 1.5E+01 nc 40E-04 1 29E+04 n 0 1071336 Givphosate Givphosate Bithon 3.1E+00 nc 3.2E+01 nc 3.2E+02 nc 1.8E+02 nc 3.8E+03 nc 3.8E+03 nc 7.8E+02 nc 1.8E+01 nc 4.8E+01 nc 3.2E+01 nc 4.8E+01 nc 1.8E+02 nc 3.8E+03 nc 7.8E+02 nc 1.8E+02 nc 3.8E+03 nc 7.8E+02 nc 1.8E+03 nc	3 8E+00 I	h	3 8E+00	r	0	01	67-45-8	Furazolidone	1.3	E-01 n	c 6.5E-01 nc	1.8E-03 nc	1.8E-02 ca		
5 6E 01 n 5 0E 02 i 5 12E 03 c 9 / E - 03 c 4 0E - 04 c 1 3 - 03 ca 1 - 3 - 03		3 0E-03	1	1 4E-02	n O	01	98-01-1	Furfural	1.8	E+02 n	с 2.6E+03 пс	5.2E+01 nc	1.1E+02 nc	Í	
3 0 € 02 i 3 0 € 02 i 0 1 8 056 05 0 Full meter (SVC)(X) 1 0 E + 01 ca 2 2 E + 01 ca 1 2 E	5 0E+01	h	5 0E+01	1	0	01	531-82-8	Funum	9.7	E-03 c	a 4.9E-02 ca	1.3E-04 ca	1.3E-03 ca		
40E-04 24E-01 nc 3.5E+02 nc 1.5E+01 nc 10E-01 10E-01 0 01 1071-83.6 Glycidaldehyde 2.4E+01 nc 3.5E+02 nc 1.5E+01 nc 10E-01 10E-01 0 01 1071-83.6 Glycidaldehyde 2.4E+01 nc 3.7E+02 nc 3.6E+03 nc 50E-05 50E-04 0 01 6600-02 Haixylop-methyl 3.1E+00 nc 4.4E+01 nc 3.7E+02 nc 1.5E+01 nc 4.7E+02 nc 1.5E+01 nc 4.7E+02 nc 1.5E+02	3 0E-02	1	3 0E-02	1	0	01	80568-05-0		1.6	E+01 ℃	a 8.2E+01 ca	2.2E-01 ca	2.2E+00 ca		
10E-00 10E-01 10E-101		4 0E-04	•	4 0E-04	10	01	77182-82-2	Gluiosinale-ammonium Glucidaldebude	2.4	E+01 n E+01 -	c 3.5E+U2 ∩c	1.5E+00 nc	1.5E+01 nc	1	
30E 05 i 50E 05 i 50E 05 i 50E 05 i 50E 06 i 0 01 50E 04 i 12E 02 i 13E 00 i 13E 03 i 13E 02 i 13E 00 i 13E 03 i 13E 02 i 13E 00 i		1.0E-01	· · · · · · · · · · · · · · · · · · ·	1.05-01		01	1071.83.6	Givohosale	- 61	E+03	c 8.8E+04 ac	37E+02 ac	3.6E+03 oc		······
13E-02 13E-02 1 13E-02 1 0 1 7927-27-3 Harmony 79E+02 nc 1.1E+04 nc 4.7E+01 nc 4.7E+02 nc 45E+00 50E-04 46F+00 50E-04 66F+00 50E-04 0 0 1 79E+02 nc 1.1E+04 nc 4.7E+01 nc 4.7E+02 nc 91E+00 1 3E+05 91E+00 1 3E+05 0 01 784+0 HeplaChlor 1.1E+04 nc 4.7E+01 nc 4.7E+03 ca 1.5E+02 ca 1.5E+00 ca 1.5E+00 ca 1.5E+00 ca 1.5E+00 ca 7.4E+03 ca 7.0E+01 3.0E+01 1.0E+0 3.0E+01 nc 7.3E+00 nc 7.3E+01 nc 7.3E+00	1	5 0E 05		5 UE-05	1 0	01	69806-40-2	Haloxyfop-methyl	3.1	E+00 n	c 4.4E+01 nc	1.8E-01 nc	1.8E+00 ac		
4 56 + 00 i 5 0E - 04 i 4 6E + 00 i 5 0E - 04 i 0 - 0 1 - 76 + 4.8 HeplaChlor 1 - 1 E + 00 ca 1 - 5 E - 03 ca 1 - 5 E - 04 ca 7 - 4 E - 04	1	1 3E-02		1 3E-02	1 0	01	79277-27-3	Harmony	7.9	E+02 n	c 1.1E+04 nc	4.7E+01 nc	4.7E+02 nc		
9 1E+00 1 3E+05 1 9 1E+00 1 3E+05 1 0 0 1 1024:57.3 Heptachlor epoxide 5.3E+02 ca* 7.4E+04 ca* 7.4E+03 ca* 7.0E+01 3.0E+01 nc 1 8E+00 1 86±04 1 6E+00 i 80±04 0 0 1 iiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiiii	4 5E +00	5 0E-04	1 4 6E +00	1 5 0E-04	1 0	01	76-44-8	Heptachlor	1.1	E-01 c	5.5E-01 ca	1.5E-03 ca	1.5E-02 G	2.3E+01	1.0E+00
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	9 1E+00	1 3E-05	+ 91E+00	i 13E-05	1 0	01	1024-57-3	Heptachlor epoxide	5.3	E-02 a	· 2.7E-01 a	7.4E-04 ca	7.4E-03 cm	7.0E-01	3.0E-02
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		2 OE 03	1	2 0E-03	1 0	01	87-62-1	Hexabromobenzene	1.2	E+02 n	<u>с 1.8E+03 пс</u>	7.3E+00 nc	7.3E+01 nc		
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	1 6E • 00	8 0E-04	1 6E+00	i 8 0E-04	(0	01	118-74-1	Hexachlorobenzene	3.0	E-01 c	a 1.5E+00 ca	4.2E-03 ca	4.2E-02 ca	2.0E+00	1.0E-01
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	7 8E-02	1 3 DE-04	n 78E-02	1 3 0E-04	1 0	01	87-68-3		0.2	E+00 G	- J.2E+UI ca*	0.0E-02 ca	1 1E 02	2.0E+00	1.0E-01 3.0E-05
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	1 95+00		0 JE+00	- <u>-</u>		0.04	319-84-0		- 32	E-01	215+00 0	375-03	375-02 0	3 05 03	1 0F-04
1 BE+00 1 BE+00 0 004 608/31 HCH-technical 3.2E-01 ca 2.1E+00 ca 3.8E-03 ca 3.7E-02 ca 3.0E-03 1.0E-04 7 0E 03 2 0E 05 h 0 01 77.47 Hexachlorocyclopentadiene 4 2E+02 nc 5.9E+03 nc 7.3E-02 nc 2 0E+02 nc 4 0E+02 2 0E+02 nc 4 0E+02 2 0E+02 nc 4 0E+02 2 0E+02 nc 1 0E-04 1 0E-04 </td <td>1 3E+00</td> <td>h 30E-04</td> <td>1 36:00</td> <td>4 3 0E-04</td> <td>, 0</td> <td>0.04</td> <td>58-89-9</td> <td>HCH (gamma) Lindane</td> <td>4.4</td> <td>E-01 c</td> <td>2.9E+00 c</td> <td>5.2E-03 a</td> <td>5.2E-02 .</td> <td>9.0E-03</td> <td>5.0E-04</td>	1 3E+00	h 30E-04	1 36:00	4 3 0E-04	, 0	0.04	58-89-9	HCH (gamma) Lindane	4.4	E-01 c	2.9E+00 c	5.2E-03 a	5.2E-02 .	9.0E-03	5.0E-04
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	1 8E+00	1	1 86 - 00			0.04	608-73 1	HCH-technical	3.2	E-01 6	a 2.1E+00 ca	3.8E-03 ca	3.7E-02 G	3.0E-03	1.0E-04
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		7 OE 03	1	2 OE -05	h 0	01	77 47 4	Hexachlorocyclopentadiene	4.2	E+02 6	c 5.9E+03 nc	7.3E-02 nc	2.6E+02 nc	4 0E+02	2 0E+01
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	6 2E+03	•	4 6E+03	•	0	01	19408-74-3	Hexachlorodibenzo-p-dioxin mixture (HxCDD)	7.8	E-05 a	• 4.0E-04 c•	1.5E-06 G	1.1E-05 ca	6 05 04	0.05.00
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	1 4E-02	1 0E-03	1 14E-02	1 1 0E-03	1 0	01	67-72-1	Hexachloroethane	3.5	E+01 ca	" 1.8E+02 ca"	4.8E-01 ca*	• 4.8E+00 ca**	5.0E-01	2.0E-02
		3 0E-04	1	3 0E-04	10	01	70-30-4	Hexachiorophene	1.0	E+00 -	C 2.0ETU2 AC		6.1E-01 m	1	
1 = 296.06 ($2.96.06$) 0.1 $822.06.0$ (1.6 -Hexamelty/lene diisocyanate $1.7E-01$ or $2.5E+00$ or $1.0E-02$ or $1.0E-01$ or 1	1 16:01	1 3 0E 03	1 116-01	1 3 DE-03	. 0	01	822-06-0	1 6-Hexamethylene diisocyanate	1.7	E-01 a	с 2.5E+00 лс	1.0E-02 oc	1.0E-01 oc	İ	
6 0E 02 h 57E 02 i 1 110.54.3 [0-HEXANE 1.1E+02 sat 1.1E+02 sat 2.1E+02 nc 3.5E+02 nc		6 0E-02	h	5 7E-02			110-54-3	n-Hexane	-1.7	E+02 .	at 1.1E+02 sat	2.1E+02 nc	3.5E+02 nc	1	
3 3E 42 1 3 3E 42 1 0 0.1 51235-04-2 Hexazinone 2.0E+03 nc 2.9E+04 nc 1.2E+02 nc 1.2E+03 nc	1	3 3E-02	1	3 3E-02	10	0.1	51235-04-2	Hexazinone	2.0	E+03 r	c 2.9E+04 nc	1.2E+02 nc	1.2E+03 nc	1	
5 0E-02 1 5 0E-02 1 0 0 1 2681-41-0 HMX 3.1E+03 nc 4.4E+04 nc 1.8E+03 nc		5 0E-02	1	5 DE-02	1 0	01	2691-41-0	НМХ	3.1	E+03 /	ic 4.4E+04 nc	1.8E+02 nc	1.8E+03 nc	I	
3 0E+00 i 17E+01 i 0 0 1 30201-2 Hydrazine hydrazine sulfate 1.6E-01 cs 8.2E-01 cs 3.9E-04 cs 2.2E-02 cs	3 0E+00	i	1 7E+01	i	0	01	302-01-2	Hydrazine, hydrazine sulfate	110	E-01 a	• 8.2E-01 a	3.9E-04 ca	2.2E-02		
100000 n 17600 n 01 60344 [Hydrazine, monomethy] [1.0E-0] ca $8.2E-0$] ca $40E-04$ ca $2.2E-02$ ca $1000000000000000000000000000000000000$	3 0E+00	n -	17E+01	A		01	60-34-4	Hydrazine, monomethyl Hydrazine, dimethyl	1.6	E-01 6	a 0.21±-01 ca 9.21±-01	4.012-04 ca	2.2E-02 ca	1	
JUE-DU n 01 57.14-7 Inversion 1.0E-DI ca 0.2E-DI ca 4.0E-DI	302,00		172.01	6 75 00		01	2/-14-/	Hydrogen chloride		1-01 6	a 0.2E-UI ca	4.0E-04 ca	2.20-02 63	ł	
3 0E-03 2 0E-04 778-05-4 Hydrogen sulfde 10F+00 1 1E+02	1	3 0E-03		2 9F-04			7783-06-4	Hydrogen sulfide				105+00	1 1E+02	1	
40E-02 h 40E-02 / 0 01 123-31-0 p-Hydroquinone 2.4E+03 nc 3.5E+04 nc 1.5E+02 nc 1.5E+03 nc		4 UE -02	h	4 0E-02	10	01	123-31-8	p-Hydroquinone	2.4	E+03 r	c 3.5E+04 nc	1.5E+02 nc	1.5E+03 nc	1	

- - -

Key FIRIS A			o=Other EPA D			EROUTE EXTRA	FOR PLANNING P			OSES	100X ca) "(where	nc < 10X cm)			
a. SFo 1/(mg/kg⋅d)	RIDo (my/kg-d)	SFi 1/(mg/kg-d)	RfDi (mg/kg-d)	v o c	skin abs soils _{éa}	CAS No.		Resid	ential ng/kg)	IARY REMEDI Industrial Soil (mg/kg)	A HON GOA Ambient Air (ug/m^3)	Tap Wa (ug/) .ter? () 1)	SOIL SCREE Migration to Grou DAF 20 (mg/kg)	INING LEVELS Ind Water Daf 143 144 (mg/kg)
	<u></u>		<u></u>	2.00	. . K			a to site in the	o includes	AF INTERNIE IN INTERNIE	err solar konstalasi.	distantine incom	, Robil	Maling Constant in	an an an Art States
	1 3E-02	1	1 3E-02	1 0	01	35554-44-0	Imazalil	17.9E+0	2 nc	1.1E+04 nc	4.7E+01	4.7E+0	2 nc	[
	2 5E-01	•	2 5E-01	1 0	01	81335-37-7	Imazaquin	1.5E+0	4 nc	1.0E+05 max	9.1E+02 r	. 9.1E+0	3 nc		
	4 0E-02	1	4 0E-02	10	0.1	36734-19-7	Iprodione	2.4E+0	3 nc	3.5E+04 nc	1.5E+02 r	₁c 1.5E+0	3 nc		
	3 0E-01	n.	2.00 01	0		7439-89-6	Iron	12.3E+0	14 nc	1.0±+05 max	1 15+02	1.12+0	4 nc	1	
95E-04 i	2 0E-01	' I 9.5E-04 I	2 0E-01	1 1	01	78-59-1	isophorone	15 1E+0	14 nc 12 cm	2.6E+03 cm	7.1E+00	a 7.1E+0	13 nc	5.0E-01	3.0E-02
	1 5E 02	1	1 5E-02	1 0	01	33820-53 0	Isopropalin	9.2E+C	2 nc	1.3E+04 nc	5.5E+01	5.5E+0	2 nc		
	1 OE-01	•	1 1E-01	r 0	01	1832-54-8	Isopropyl methyl phosphonic acid	6.1E+0	3 nc	8.8E+04 nc	4.0E+02	₀ 3.6E+0	13 nc		
	5 0E-02	1	5 OE -02	1 0	01	82558-50-7	Isoxaben	3.1E+0	13 nc	4.4E+04 nc	1.8E+02 (1.8E+0	13 nc		
18E+01 n		1 8E+01		0	01	143-50-0	Kepone	2.7E-0	2 64	1.4E-01 ca	3.7E-04 4	a 3.71±-0	13 ca		
PRGs Based on	2 UE-U3 FPA Models (IF	IBK 1994 and TRM	2 0E-03	10	01	7/501-63-4	Laciolen Lead	1.2E+0	12 nc	7.5E+03 nc	1.32+00 /	% /.JE+U	1 NC		
	1 0E-07	1		0	01	78-00-2	Lead (tetraethyl)	6 1E-0	3 00	B 8E-02 ac		3 6E-0	3 ac		
	2 DE-03		2 0E-03	1 0	01	330-55-2	Linuron	1.2E+0	2 116	1.8E+03 nc	7.3E+00	. 7.3E+0)1 nc		
	2 0E 02	*		0		7439-93-2	Lithium	1.6E+0	3 nc	4.1E+04 nc		7.3E+0	12 nc		
	2 0E-01	1	2 OE -01	1 0	01	83055 99-6	Londax	1.2E+0	4 nc	1.0E+05 max	7.3E+02 a	- 7.3E+0	3 nc	1	
ļ	2 0E-02	•	2 0E-02	1 0	01	121-75-5	Malathion	1.2E+0	13 nc	1.8E+04 nc	7.3E+01	w 7.3E+0	12 nc		
	1 DE 01	- '	1 0E-01	1 0	01	108-31-6	Maleic annydride	1 761	D RC	0.0E+U4 nc	3.7E+02 /	3.05+0	S nc		
	5 0E-01 2 0E-05	1 N	5 UE U1	1 1	0 1	123 33 1	Maloponitrile	112E+0	nc nc	1.8E+01 m	7 3E-02	~ 7.3E-0	13 nc		
1	2 0E-03	 D	2 0E-03	, 0	01	8018-01-7	Mancozeb	1.8E+0)3 nc	2.6E+04 ac	1.1E+02	ic 1.1E+0	13 nc	1	
6 0E-02 o	5 0E-03	1 6 0E-02	r 5.0E-03	(0	01	12427-38-2	Maneb	8.1E+C	0	4.1E+01 ca	1.1E-01	- 1.1E+C	0 63		
	2 4E-02	4	1 4E-05	1 0		7439-96-5	Manganese and compounds	1.8E+0)3 nc	3.2E+04 nc	5.1E-02	ac 8.8E+C)2 nc		
	9 0E-05	h	9 UE -05	1 0	01	950-10-7	Mephosfolan	5.5E+0)0 nc	7.9E+01 nc	3.3E-01	nc 3.3E+0)0 nc		
	3 0E-02		3 0E -02	r O	01	24307-26-4	Mepiquat	1.8E+0	13 nc	2.6E+04 nc	1.1E+02	nc 1.1E+C	13 nc		
2 9E-02	10E-01	n 29E-02	1 0E-01	ſ O	01	149-30-4	2-Mercaptobenzothiazole	1.7E+0)1 ca	8.5E+01 ca	2.3E-01	a 2.3E+0	63 O		
	3 UE-04	•	. of of	0		7487-94-7	Mercury and compounds	2.32+0	Di nc	0.1E+02 ne	715.01	1.12+0	1] nc		
	1.05.04		8 6E-05	۰ ۵	0.1	7439-97 0	Mercury (methyl)	6 1E+(0	8.8E+01 or	3.12-01	36E+(0 ~		
	3 0E-05		3 0E-05	10	01	150-50-5	Merphos	1.8E+0)0 nc	2.6E+01 nc	1.1E-01	nc 1.1E+0	00 nc	1	
	3 DE 05	4	3 0E 05	(0	0 1	78-48 8	Merphos oxide	1.8E+0)() nc	2.6E+01 nc	1.1E-01	nc 1.1E+()0 nc	1	
1	6 0E-02	1	6 0E-02	1 0	01	57837-19-1	Metalaxvi	3.7E+0)3 nc	5.3E+04 nc	2.2E+02	nc 2.2E+()3 nc		
	1 UE-04	1	2 UE-04	h 1		126-98-7	Methacrylonitrile	2.1E+0)0 nc	8.8E+00 nc	7.3E-01	nc 1.0E+()0 nc		
	5 0E-05	•	5 0E -05	r 0	01	10265-92-6	Methamidophos	3.1E+0)0 nc	4.4±+01 nc	1.8E-01	nc 1.8E+(00 nc	1	
	5 UE-01	1	5 0E-01	1 0	01	67-56-1	Methidathian	13.1E+0)4 nc)1	1.0E+05 max	1.8E+03	nc 1.0E+(3.6E+()4 nc)1 nc		
	1 UE-03		102-03			16752 77.5	Methomy	4 4F+0	11 00	1.5E+02 ac	91E+01	0.0E1	$\frac{12}{12}$		
1	2 JE-02 5 DE-03		5 UE-02	, 0	01	72-43-5	Methoxychlor	3.1E+)2 nc	4.4E+03 nc	1.8E+01	nc 1.8E+0)2 nc	1.6E+02	8.0E+00
	1 DE 03	h	5 7E-03	+ 0	01	109-86-4	2-Methoxyethanol	6.1E+0)1 nc	8.8E+02 ns	2.1E+01	nc 3.6E+(D1 nc		
	2 00-03	h	2 OE 03	1 0	01	110 49 6	2-Methoxyethanol acetate	1.2E+)2 nc	1.8E+03 nc	7.3E+00	nc 7.3E+0	31 nc		
4 6E-02	h	4 6E-02	1	0	01	99-59-2	2-Methoxy-5-nitroaniline	1.16+0)] n)/	5.4E+01 cm	1.56-01	a 1.5E+l 6.1E+(10 61		
 	1 0E+00	h	1 DE+00	11		79-20-9	Melhyl acolate	12.2ET	11 nc	235+02	1 1 1 + 02	1 8E+0	12		
246.01	3 UE-U2	n 2.46.01	3 02-02	1 1	0.1	96-33-3	2-Methylaniline (o-toluidine)	2.0E+)) a	1.0E+01 cm	2.8E-02	a 2.8E-0)1		
1 8E-01	n	1 8E-01	,	0	01	636-21-5	2-Methylaniline hydrochloride	2.7E+		1.4E+01 ca	3.7E-02	- 3.7E-0	1		
	1 0E+00	×	1 0E+00	1 0	01	79-22-1	Methyl chlorocarbonate	6.1E+)4 nc	1.0E+05 max	3.7E+03	nc 3.6E+0	J4 nc	1	
1	5 0E-04		5 0E -04	1 0	0 1	94-74 6	2-Methyl-4-chlorophenoxyacetic acid	3.1E+	01 nc	4.4E+02 nc	1.8E+00	nc 1.8E+0	01 nc		
	1 UE-02		1 0E-02	10	01	94-81-5	4-(2-Melhyl-4-chlorophenoxy) butyric acid	6.1E+	J2 no	8.8E+03 nc	3.7E+01	nc 3.61+1	J2 nc		
1	1 OE 03	<u>.</u>	1 DE 03	1 0	01	93 65 2	2-(2-Methyl-4-Chlorophenoxy) propionic acid	0.1E+	01 nd 01	8.8E+02	3.76+00	nc 3.0E+0	JI ne 31		
	1 UE-03 8 6 F-01	1	8 6F-01	1 U h 1	01	10484-77-8	Methylcyclohexane	2.6E+	03 ~	8.8E+03 m	3.1E+03	NC 5.2E+	01 nc 03 nc]	
2 5E-01	h	2 5E 01	1	ò	01	101-77-9	4,4'-Methylenebisbenzeneamine	1.9E+	00 ca	9.9E+00 ca	2.7E-02	ca 2.7E-0	1	+	·····
1 3E-01	h 7.0E-04	h 136-01	h 7.0E-04	1 0	01	101-14-4	4.4'-Methylene bis(2-chloroaniline)	3.7E+	00 ca	• 1.9E+01 co*	5.2E-02	a• 5.2E-0)1 ca*		
4 6E-02		4 6E-02	1	0	01	101-61-1	4,4'-Methylene bis(N,N'-dimethyl)aniline	1.1E+	01 .	5.4E+01 ca	1.5E-01	ca 1.5E+(60 CB	1	

Key == IRIS IN=1	HEAST N=NCE	A .=WITHDRAWN	e=Other EPA D	OCUMENT	S 1=ROUTE EXTRA	POLATION ca=CANCER PRG nc=NONCANCER PRG SIT=SOIL SATUR	RATION max=CEILING LIM	IT "(where: nc < 10	OX ca) **(where	nc < 10X ca)		
slater a com				QUENEL.	KWA PAR	FOD DI ANIMINO I			WEATER			an pris pris de la companya de la companya de la companya de la companya de la companya de la companya de la co
				(ES145)		FOR PLANNING	PURP(JSES	人民的 有54人	a ann an		
	fill Back									na Orzalista (C. C. C. C. C. C. C. C. C. C. C. C. C. C
				(in an allowing	LAND ADDIES - CONSTR	CONTAMINANT			TION COM			Chimit Guidikkining NINC I EVELS
		IT INFORM	ATION			CONTAMINANT	PRELIMINAL	AT REMEDIA	TION GOAL	.5 (PRGS)	SUL SUREE	NING LEVELS
SEO	RIDo	SE	R (Di	O abs	CAS No.	THE A STREET AND A REPORT OF A REPORT OF A REPORT OF A	Residential	Industrial	Ambient Air	Tan Water	Migration to Grou	na vvater
1/(mg/kg-d)	(mg/kg-d)	1/(mg/kg-d)	(mu/kg-d)	C soil		an an an an an an an an an an an an an a	Soil (mu/kg)	Soil (mg/kg)	(ug/m^3)	(ug/l) (ug/l)	(mg/kg)	(mg/kg)
				1.000.02.0	M. C. Starter	Service and the state of the st	and the second state of the second					
							WATCH COMPANY AND A DESCRIPTION	LINESSEE STREET SHORE S	And the second second second	CONTRACT AND IN A DATA OF A CONTRACT		
	1 0E 02	h	1 OF -02	()	74-95-3	Methylene bromide	167F+01 2	4E+02	3 7E+01	6.1E+01 nc		
7 5E-03 1	6 0E-02	1 6E-03	8 6E-01	n 1	75-09-2	Methylene chloride	8 9E+00 ca 2	1E+01 ca	4.1E+00 ca	4.3E+00 ca	2.0E-02	1.0E-03
	17E-04	1	17E-04	. 0 01	101-68-8	4.4'-Methylene diphenyl diisocyanate	1.0E+01 nc 1	.5E+02 nc	6.2E-01 no	6.2E+00 nc		
	6 0E-01		2 9E-01	1 1	78-93-3	Methyl ethyl kelone	7 3E+03 nc 2	8E+04 nc	10E+03 ad	1.9E+03 nc		
116.00 0		1 16:+06		0 01	60 34 4	Methyl hydrazine	4.4E-01 cm 2	.2E+00 ca	6.1E-03 a	6.1E-02 ca		
	8 0E-02	h	2 3E-02	h 1	108-10-1	Methyl isobulyl ketone	7.9E+02 nc 2	9E+03 nc	8.3E+01 no	1.6E+02 nc	1	
	5 7E-04	1	5 7E-04	A 0 01	74-93-1	Methyl Mercaptan	3.5E+01 ne 5	0E+02 nc	2.1E+00 n	2.1E+01 nc		
	1 4E+00		2 0E-01	1.1	80 62 6	Methyl methacrylate	2.2E+03 nc 2	.7E+03 sat	7.3E+02 nd	1.4E+03 nc	1	
33E-02 n		3 3E-02	•	0 01	99-55-8	2-Methyl-5-nitroaniline	1.5E+01 a 7	'.5E+01 ⊶	2.0E-01 ca	2.0E+00 ca		
	2 5E-04	1	2 5E -04	1001	298-00-0	Methyl parathion	1.5E+01 nc 2	2E+02 nc	9.1E-01 nd	9.1E+00 nc		
1	5 UE-02	1	5 0E-02	1 O O 1	95-48-7	2-Methylphenol	3.1E+03 nc 4	.4E+04 nc	1.8E+02 no	1.8E+03 nc	1.5E+01	8.0E-01
	5 UE-02	<u>.</u>	5 0E 02	1001	108-39-4	3-Methylphenol	3.1E+03 nc 4	.4E+04 nc	1.8E+02 nd	: 1.8E+03 nc	l	
	5 0E 03	h	5 0E-03	0 01	106-44 5	4-Methylphenol	3.1E+02 nc 4	.4E+03 nc	1.8E+01 no	1.8E+02 nc		
ļ	2 0E-02	n	2 UE -02	10 01	993 13-5	Methyl phosphonic acid	1.2E+03 nc 1	.8E+04 nc	7.3E+01 na	7.3E+02 nc		
	6 0E-03	h -	1 1E 02	h 1	25013-15-4	Methyl styrene (mixture)	1.3E+02 nc 5	.6E+02 nc	4.2E+01 nd	6.0E+01 nc		
	7 UF-02	ь.	7 OE -02	1 1	98 83 9	Methyl styrene (alpha)	6 8E+02 sal 6	8E+02 sat	2.6E+02 nd	4.3E+02 nc		
ļ			86E 01	1 1	1634 04 4	Methvl tertbutyl ether (MTBE)			3.1E+03 no	2.0E+01 nd/ca	1	
1 8E 03		1 8E-03		1		"CAL-Modified PRG"	1.7E+01 cm 3	0.7E+01 ca	3.7E+00 ca	6.2E+00 ca		
ļ	1 5E-01	1	1 5E -01	100	51218 45 2	Metolaclor (Dual)	9.2E+03 nc 1	.0E+05 max	5.5E+02 nd	5.5E+03 nc		
1	2 5E-02	1	2 5E-02	100	21087-64-9	Metribuzin	1.5E+03 nc 2	2.2E+04 nc	9.1E+01 no	: 9.1E+02 nc	Í.	
18E+00 x	2 0E-04	1 8E+00	r 2.0E-04	100	2385-85-5	Mirex	2.7E-01 co* 1	.4E+00 ca	3.7E-03 ca	3.7E-02 ca		
	2 0E-03	1	2 0E-03	100	2212-67-1	Molinate	1.2E+02 nc 1	.8E+03 nc	7.3E+00 n	: 7.3E+01 nc		
	5 0E-03	n		٥	7439-98-7	Molybdenum	3.9E+02 nc 1	.0E+04 nc		1.8E+02 nc	1	
	1 0E-01	h	1 0E -01	h 0 0	10599 90-3	Monochloramine	6.1E+03 nc 8	3.8E+04 nc	3.7E+02 n	; 3.6E+03 nc		
	2 0E -03	1	2 0E-03	100	300-76-5	Naled	1.2E+02 nc 1	.8E+03 nc	7.3E+00 n	: 7.3E+01 nc		
	1 0E-01	1	1 OE -01	100	15299-99-7	Napropamide	6.1E+03 nc 8	8.8E+04 nc	3.7E+02 n	: 3.6E+03 nc		
	2 OE -02	1		0	7440-02 0	Nickel (soluble salts)	1.6E+03 ns 4	1.1E+04 nc		7.3E+02 nc	1.3E+02	7.0E+00
					-	"CAL-Modified PRG" (PEA, 1994)	1.5E+02					
		8 4E-01	•	0		Nickel refinery dust		45.04	8.0E-03 a	•		
		1 7E • 00	1	0	12035-72-2			1.1E+04 cm	4.02-03 6	·		
	1 5E-03	x	1 5E-03	100	1 1929-82-4	Nitrapyrin	9.2E+01 nc 1	1.3E+03 ne	5.5E+00 n	: 5.5E+01 nc		
Tap Water PRG	Based on Infant	NOAEL (see IRIS)			14797-55-8	Nitrate				1.0E+04 nc		
	1 0E-01	×			10102-43-9	Nitric Oxide	7.8E+03 nc 1	1.0E+05 max		3.6E+U3 nc		
Tap Water PRG	Based on Infant	NOAEL (see IRIS)			14797-65-0	Nitrite	0.55.00	05.04	045.04	1.UE+U3 AC		
	57E-05	1	5 7E-05	h 0 0	1 88-74-4	2-Nitroaniline	13.5E+00 nc 5	0.012+01 nc	2.12-01	2.1E+00 AC	105.01	7 05-03
L	5 0E-04	1	5 7E-04	h 1	98-95-3	Nilrodenzene	2.0E+01 nc		2.1ETUU A	0.4ETUU AC	1.00-01	1.01-03
	7 0E-02	h	7 OE -02	100	1 67-20-9	Nitroturantoin	14.3E+U3 nc 0	J.2ETU4 NG	2.0ETU2 N	45E.02		
156+00	h	9 4E + UD	ła –	0 0	1 59-87-0	Nitroiurazone	3.20-01 4	1.85+02	4.85-01	4.8E+00		
145.02		1 41 02	•	0 0	1 55 63 0	Nitroquanidune	-161E+03	BE +04	375+02	3.6E+03 **		
1	1 OE-01	1	1 UE-01	100	1 550-887	A Nitrophenol	4 9F+02 ac	7 0F+03	29E+01	c 2.9E+02 nc		
0.45.00	8 06-03		8 UE-U3	100	1 100-02-7	2-Nitropropage	1.02.02 10		7.2E-04 c	1.2E-03 G		
9 4E+00	1 57E-03	1 94E+00	n 572-03	<u> </u>	7940-9	N-Nitrosodi-n-butylamine	24F-02 m	6 1E-02 G	121-03	2 0E-03	 	
346+00	•	3 66 +00			924-10-3	N-Nitrosodiethagolamine	17E-01	8.8F-01	2.4E-03 c	2.4E-02 ca	1	
1.55+02	•	1 5E+02	•	00	1 55-18-5	N-Nitrosodiethylamine	3.2E-03 ==	1.6E-02 ca	4.5E-05 c	4.5E-04 ca		
5 15 +01	<u>.</u>	4 95.01	<u>.</u>		1 62.75.9	IN-Nitrosodimethylamine	9.5E-03 cz	4.8E-02 a	1.4E-04 e	1.3E-03 ca	1	
4 95.03		1 95.01	:	0 0	1 86.30.6	N-Nitrosodiohenvlamine	9.9E+01	5.0E+02	1.4E+00	1.4E+01 ca	1.0E+00	6.0E-02
7 06.00		7 66 100		0 0	1 621-64-7	N-Nitroso di-n-propylamine	6.9E-02	3.5E-01 a	9.6E-04	9.6E-03	5.0E-05	2 0E-06
2 26 + 01	· · · ·	2 2E + 01	······································	0 0	1 10595-95-6	N-Nitroso-N-methylethylamine	2.2E-02	1.1E-01 ==	3.1E-04	3.1E-03 ca	1	
2 1E+00		2 1E+00		0 0	1 930-55-2	N-Nitrosopyrrolidine	2.3E-01 ca	1.2E+00 ca	3.1E-03 a	3.2E-02 G		
1	1 0E-02	6	1 UE-02		89-08-1	m-Nilrotoluene	3.7E+02 nc	1.0E+03 sat	3.7E+01	c 6.1E+01 nc		
	1 DE -02	h	1 0E-02	1 1	88-72-2	o-Nitrololuene	3.7E+02 nc	1.0E+03 sal	3.7E+01	6.1E+01 nc	1	
	1 0E-02	h	1 0E-02	1.1	98-99-0	p-Nitrotoluene	3.7E+02 nc	1.0E+03 sat	3.7E+01	c 6.1E+01 oc	1	
1	4 0E-02		4 UE-02		1 27314-13-2	Norflurazon	2.4E+03 nc	3.5E+04 nc	1.5E+02	c 1.5E+03 nc	1	
Long to the second second second second second second second second second second second second second second s	and the second se					the second second second second second second second second second second second second second second second se					1	

•

Key 1=1RIS 1.=	HEAST NENCE	A .=WITHDRAWN	o=Other EPAL	DOCUMENTS	I FROUTE EXTRA	POLATION CATCANCER PRG ACTINONCANCER PRG SITESOIL SATURAT	ION Max=CEILING	LIMIT (Whore, he <)	Non ca) (where i	atrice in Marine		
	astat. t				iki Parimui i	FOR PLANNING P	NIRP	OSES	Shear chi	Ration in the second		· 9 24 1
na dina kata sa	1.803 M 2014	الى بىرى ئەلەرەتە تىرىلام كە					0111	0010			<u>. 1954. – 1</u> . – 1	K. S. KARACH
	TOXICI	TY INFORM	ATION	a, lorot or relative description	New Strate and the Constant of Strategy of	CONTAMINANT	PRELIMIN	ARY REMEDIA	ATION GOAL	S (PRGs)	SOIL SCREE	INING LEVELS
				V skin	6.6.N		Queidentint	laduation	Ambiant Ak	Too Mate	Migration to Grou	Ind Water
SFo 1/(mo/ko-d)	RIDo (ma/ka-d)	SFi 1/(ma/ka-d)	RIDi (mu/ku-di)	C solis	CAS No	the second second second second second second second second second second second second second second second s	Soil (ing/kg)	Soil (mg/kg)	(ug/m*3)	(ug/l)	(mg/kg)	(mg/kg)
	((ليد به عد	關約中世的	· · · · · · · · · · · · · · · · · · ·	<u>in suit Maralitata p</u>	ARTERIAL CONTRACTOR	adata mitra Pri D	REAL PROPERTY IN CALLER OF CALLER OF CALLER OF CALLER OF CALLER OF CALLER OF CALLER OF CALLER OF CALLER OF CALL	and a second second second	
	7 0E 04	•	7 0E-04	r 0 01	85509-19-9	NuStar	4.3E+01 nc	6.2E+02 nc	2.6E+00 nc	2.6E+01 nc		
	3 UE-03	1 1	3 0E-03	1 0 01	32536-52-0	Octabromodibnenvi einer	1.0E+02 nc 1.2E+02 nc	1.8E+03 nc	7.3E+00 nc	7.3E+01 nc		
	5 0E-02	<u></u>	5 0E-02	1 0 01	19044-88-3	Oryzalin	3.1E+03 nc	4.4E+04 nc	1.8E+02 nc	1.8E+03 nc		
	5 UE-03		5 OE -03	r 0 01	19666 30-9	Oxadiazon	3.1E+02 AC	4.4E+03 nc	1.8E+01 nc	1.8E+02 nc	1	
	2 5E-02	1	2 5E-02	1 0 01	23135-22-0	Oxamyi	1.5E+03 nc	2.2E+04 nc	9.1E+01 ac	9.1E+02 nc		
	3 DE-03		3 0E-03	1001	42874-03-3	Paclobutrazol	7.9E+02 ac	1.1E+04 nc	4.7E+01 nc	4.7E+02 nc	1	
	4 5E-03		4 5E-03	1 0 01	4685-14-7	Paraquat	2.7E+02 NG	4.0E+03 nc	1.6E+01 nc	1.6E+02 nc		
	6 0E-03	h	6 0E -03	1 0 01	56-38-2	Parathion	3.7E+02 nc	5.3E+03 nc	2.2E+01 nc	2.2E+02 nc	1	
	5 DE 02	в	5 OE-02	1 0 01	1114-71-2	Pebulate	3.1±+03 nc	4.41±+04 nc 3.51±+04 nc	1.8E+02 ac 1.5E+02 ac	1.8E+03 nc		
2 35.02	4 UE-02	2 36.02	4 UE-02	0 01	87.84.3	Pentabromo-6-chloro cyclobexane	215+01	1.1E+02 G	2.9E-01 G	2.9E+00 ca		
2 35 -02	2.01-03	2 35 02	2.01-03	1 0 01	32534 81-9	Pentabromodiphenyl ether	1.2E+02 #c	1.8E+03 nc	7.3E+00 nc	7.3E+01 nc		
	8 OE-04	1	8 0E-04	1 0 01	608-93-5	Pentachlorobenzene	4.9E+01 nc	7.0E+02 nc	2.9E+00 nc	2.9E+01 nc	I	
2 6E-01	h 3.0E-03	1 2 6F 01	1 3 0E-03	1 0 01	82 68 8	Pentachloronitrobenzene	19E+00 ca.	9.5E+00 ca	2.6E-02 ca	26E-01 a		1.05.00
1 2E-01	3 06-02	i 12E-01	1 3 OE 02	1 0 025	8/805	Pentachiorophenol	3.0E+00	1.1E+01 ca	5.6E-02 ca	5.6E-01	3.01:-02	1.01:-03
	5 UE 04	*	5 05 03	0	62616 62 1	Permelhrin	3 1E+03 oc	4 4F+04 oc	18F+02	1.8E+03 oc	}	
	2 5E-01		2 58-01	1001	13684-63-4	Phenmedioham	1.5E+04 nc	1.0E+05 max	9.1E+02 nc	9.1E+03 nc	ł	
	6 0E-01		6 0E-01	1 0 01	108-95-2	Phenol	3.7E+04 nc	1.0E+05 max	2.2E+03 nc	2.2E+04 nc	1.0E+02	5.0E+00
	2 0E-03	n	2 0E-03	1 0 01	92-84-2	Phenothiazine	1.2E+02 nc	1.8E+03 ac	7.3E+00 nc	7.3E+01 nc		
	6 UE-U3	•	6 0E-03	1001	108-45-2	m-Phenvlenediamine	3.7E+02 nc	5.3E+03 nc	2.2E+01 AC	2.2E+02 nc		
	1 9E-01	h	1 9E-01	1 0 01	106-50-3	p-Phenylenediamine	1.2E+04 nc	1.0E+05 max	0.9E+02 AC	795+00 nc		
1.05.03	8 0E-05	1	8 OE-05	r 0 01	62-38-4	2-Phenyintercult acetate	2.5E+02 ca	1.3E+03 cm	3.5E+00 ca	3.5E+01 ca	1	
1 95-03	n 2 0E-04	196.03	2 UE-04	1 0 01	298-02-2	Phorate	1.2E+01 nc	1.8E+02 nc	7.3E-01 nc	7.3E+00 nc		
	2 0E-02	1	2 0E-02	1 0 01	732-11-6	Phosmet	1.2E+03 nc	1.8E+04 nc	7.3E+01 nc	7.3E+02 nc		
	3 0E-04	h	8 6E-05	1001	7803-51-2	Phosphine	1.8E+01 nc	2.6E+02 nc	3.1E-01 AC	1.1E+01 nc		
			2 9E-03		7664-38-2	Phosphoric acid	1.65+00	A 1E+01	1.02+01 AC	7 3E-01 oc	<u> </u>	
	2 0E-05	1	1 OF - 00		7723-14-0	p-Phthalic acid	6 1F+04	10E+05 max	3.7E+03 as	3.6E+04 nc	1	
	2 0E+00	n 1	3 4E-02	h 0 01	85-44-9	Phthalic anhydride	1.0E+05 ma	x 1.0E+05 max	1.2E+02 no	7.3E+04 nc	1	
	7 0E-02	1	7 0E-02	(0 01	1918-02-1	Picloram	4.3E+03 no	6.2E+04 nc	2.6E+02 nd	2.6E+03 nc		
	1 0E-02	1	1 0E-02	1 0 0.1	23505-41-1	Pirimiphos-methyl	6.1E+02 nd	8.8E+03 nc	3.7E+01 no	3.6E+02 nc		
8 9E • 00	h 70E-06	h 89E+00	1 7 0E-06	1001		Polybrominated biphenyls	0.5E-02 ca	- 2.0E-U1 ca*	7.0E-04 a	34F-02 ca	 	
2 06 +00	I 	2 06 +00	1	0 01	4 1336-36-3	Aroclor 1016	3.9E+00 m	2.9E+01 ca**	9.6E-02	• 9.6E-01	1	
2 06 100	1 / UE-US	1 7 UE U2 2 UE 100	1 102-05		4 11104-28-2	Aroclor 1221	2.2E-01 a	1.0E+00 .	3.4E-03 ca	3.4E-02 cs		
2 0E +00		2 UE +00	,	0 01	4 11141-16 5	Aroclor 1232	2.2E-01 a	1.0E+00 co	3.4E-03 a	3.4E-02 G		
2 0E+00	1	2 0E+00	н. — — — — — — — — — — — — — — — — — — —	0 01	4 53469-21-9	Aroclor 1242	2.2E-01 a	1.0E+00 cm	3.4E-03 a	3.4E-02 cm		
2 0E+00	1	2 0E + 00	1	0 01	4 12672-29-6	Alocior 1240	2.2L-01 a	• 1 0E+00 ca	3.4E-03 a	· 3.4E-02 ca	1	
2 05+00	1 2 0E-05	1 2 0E+00	1 2 DE-05	(0 01	4 11097-69-1	Aroclor 1260	2.2E-01	1.0E+00 .	3.4E-03 a	3.4E-02 c		
2 02 100	'	202.00	•	01	3	Polynuclear aromatic hydrocarbons (PAHs)	ł					
	6 0E-02	ì	6 0E -02	r 1	83-32-9	Acenaphthene	3.7E+03 n	c 3.8E+04 nc	2.2E+02 n	: 3.7E+02 nc	5.7E+02	2.9E+01
	3 OE-01	i i	3 ÚE-01	i 1	120-12-7	Anthracene	2.2E+04 m	c 1.0≿+05 max	1.1E+U3 m	1.812+03 nc	1.2E+04	5.9E+02 8.0E-02
7 3E-01	n	3 1E 01	0	0 01	3 56-55-3	Benziajaninracene	6 2E-01 a	2.95+00 4	2 2E-02 4	9.2E-02 a	1-50E+00-	
7 3E-01 7 3E-02	n	- 3 1E 01 3 1E 02	n 11	0 01	13 205 99-2	Benzolklifuoranthene	6.2E+00	2.9E+01 c	2.2E-01	9.2E-01 a	4.9E+01	2.0E+00
1 36-02		5 12 -02				"CAL-Modified PRG" (PEA, 1994)	6.1E-01					
7 3E+00	•	3 1E+00	n	0 0	13 50-32-8	Benzolalpyrene	6.2E-02 c	a 2.9E-01 ca	2.2E-03 c	9.2E-03 ci	8.0E+00	4.0E-01
						Chausene Chausene (PEA, 1994)	6 25+01	. 20E+02	2 2E+00	1.52-03	1.6E+02	8 0E+00
7 3E-03	<u>^</u>	3 1E-03	n	0.0	13 218-01-9	Chirysche	10.20101 6	2.31.102 (8	4.4L. UU G	0.2L 00 C		0.01.00

Key IFIRIS N=	HEAST N=NCEA	A x=WITHDRAWN	o=Other EPA D	OCUME	NTS r	ROUTE EXTRA	OLATION CO-CANCER PRG nC=NONCANCER PRG SI I=SOIL SATURA	TION max=	CEILING	LIMIT "(where, nc < 1	OX ca) **(where	nc < 10X ca)			
							FOR PLANNING P	PUF	RP	OSES	S 4 4 4				
ar ar chirate a	TOXICI	TY INFORM	ATION	- Fillen fil	ine for in and 2004	papenal (S. e. ell. al	CONTAMINANT	PRE	LIMIN	ARY REMEDIA	TION GOA	LS (PRGs)	state and the state	OIL SCREE	NING LEVELS
, SFo 1/(mg/kg-d)	RíDo (mg/kg-d)	SFı 1/(mg/kg-d)	RIDi (mg/kg-d)	Vs Oa Cs	kin bs oils	CAS No		Resid Soil (i	ential ng/kg)	Industrial Soil (mg/kg) N994553444	Ambient Air (ug/m^3)	Tap Water (ug/l)		igration to Grou DAF 20 (mg/kg)	nd Water DAF 14-)+**** (mg/kg)
7.35+00	······	3 15 +00			0.13	53.70.3	"CAL-Modified PRG" (PEA, 1994) Dibeozlablanthracene	6.1E+0	0	2.9E-01 m	2 2E-03	9 2E-03		2.0E+00	8.0E-02
732400 1	4 0E-02	1 312,000 1	4 0E-02	1 0	013	206-44-0	Fluoranthene	2.3E+0	3 nc	3.0E+04 nc	1.5E+02	nc 1.5E+03	AC .	4.3E+03	2.1E+02
2.15.04	4 0E-02	1	4 0E-02	11		86-73-7	Fluorene	2.6E+0	3 nc	3.3E+04 nc	1.5E+02	a 2.4E+02	nc	5.6E+02	2.8E+01
73E-01 7	2 0E-02	312-01 /	n 86E-04	0	0 13	193-39-5 91-20-3	Naphthalene	5.6E+0	l ca I nc	1.9E+02 nc	3.1E+00	6.2E+02	63 AC	8.4E+01	4.0E+00
	3 0E -02	4	3 0E -02	11		129-00-0	Pyrene	2.3E+0	3 nc	5.4E+04 nc	1.1E+02	nc 1.8E+02	nc	4.2E+03	2.1E+02
1 5E-01	9 UE-03	1 1 5E 01	9 0E-03	r 0	01	67747-09-5	Prochloraz	3.2E+0	0 ca	1.6E+01 ca	4.5E-02	4.5E-01	C 8		
	6 0E-03	h	6 0E-03	1 0	01	26399-36-0	Profluralin	3.7E+0	2 nc	5.3E+03 nc	2.2E+01	nc 2.2E+02	nc		
	1 5E-02	1	1 5E-02	1 0	01	1610-18-0	Prometon	19.2E+0	12 nc	1.31±+04 nc	5.5E+01	nc 5.5E+02	nc		
	4 0E-03	۰	4 0E-03	10	01	7287-19-6	Pronetryn	2.4E+0	12 nc	3.3E+U3 nc	276+02	nc 1.0E+U2	AC		
	1 35.02		1 35 02		01	1018 16 7	Pronachlor	7 9F+0	2	1 1 1 + 04 00	4 7F+01	47E+02			
	5.02		5.06.03		01	709-98-8	Propacil	3 1E+0	2	44F+03 ac	1.8F+01	1 8E+02	00		
	2 0E-02		2 0E-02	1 0	01	2312-35-8	Propargite	1.2E+0	3 nc	1.8E+04 nc	7.3E+01	nc 7.3E+02	AC		
	2 0E-03	1	2 0E-03	1 0	01	107-19-7	Propargyl alcohol	1.2E+C	2 nc	1.8E+03 nc	7.3E+00	nc 7.3E+01	nc		
	2 DE-02	1	2 UE-02	10	01	139-40-2	Propazine	1.2E+0	3 nc	1.8E+04 nc	7.3E+01	nc 7.3E+02	ne		
	2 0E-02	•	2 DE -02	1 0	01	122-42-9	Propham	1.2E+0	13 ac	1.8E+04 AG	7.3E+01	nc 7.3E+02	nc		
	1 3E-02	•	1 3E 02	1 0	01	60207-90-1	Propiconazole	1.90+0	12 nc	1.1E+U4 nc	4.76+01	nc 4.7E+U2	nc		
	105-01		1 1E-01			98-82-8	n-Pronylbenzene	1.0E+C	12 ac	24E+02 AC	3.7E+01	6 1E+01	nc		
	2.05+01		2.05+01		0.1	57.55.6	Pronviene divcol	110F+0	15 max	10F+05 max	735+04	nc 7.3E+05			
	7 06-01		2 DE -01		01	111-35-3	Propylene givcol monoethyl ether	4.3E+0)4 nc	1.0E+05 max	2.6E+03	ac 2.6E+04	nc		
Į	7 UE 01	h	5 /E-01	. 0	01	107-98-2	Propylene glycol, monomethyl ether	4.3E+0)4 nc	1.0E+05 max	2.1E+03	nc 2.6E+04	nc		
2 4E 01	8 6E-03	1 1 3E-02	1 8 6E-03	1 1		75-56-9	Propylene oxide	1.9E+0	10 ca*	9.1E+00 ca*	5.2E-01	2.2E-01	c. 2		
	2 5E-01	•	2 5E-01	1 0	01	81335-77-5	Pursuit	1.5E+0)4 nc	1.0E+05 max	9.1E+02	nc 9.1E+03	nc		
	2 5E-02		2 5E-02	1 0	01	51630-58-1	Pydrin	1.5E+0)3 nc	2.2E+04 nc	9.1E+01	nc 9.1E+02	nc		
1	1 0E-03	1	1 OE -03	r 0	01	110-86-1	Pyridine	6.1E+0)1 nc	8.8E+02 nc	3.7E+00	nc 3.6E+01	nc		
	5 0E-04	1	5 0E-04	10	01	13593-03-8		3.12+0	J] nc	4.4E+02 Ac	1.8E+00	nc 1.0E+U1	nc		
1 22:01	n 	1 2E+01	2.05.03		01	121.02.4	RDX (Cyclonia)	1 A AF +1	10	225+01 4	6 15-02	6 1F-01			
1 12:01	3.05.02	1 112-01	305-03		01	121 02 1	Resmethrin	1 8E+0	3	2.6E+04 of	1.1E+02	as 1.1E+03	AC		
1	5 0E-02	n	5 0E 02	10	0.1	299-84-3	Ronnel	3.1E+0)3 nc	4.4E+04 ns	1.8E+02	nc 1.8E+03	nc		
	4 0E-03	1	4 OE -03	1 0	01	83-79-4	Rolenone	2.4E+()2 nc	3.5E+03 nc	1.5E+01	nc 1.5E+02	nc		
	2 56-02	1	2 5E-02	1 0	01	78587-05-0	Savev	1.5E+0)3 nc	2.2E+04 AC	9.1E+01	nc 9.1E+02	nc		
	5 0E-03	1		٥	01	7783-00-8	Selenious Acid	3.1E+0	02 nc	4.4E+03 nc		1.8E+02	nc		0.01 04
	5 0E-03	I.		0		7782-49-2	Selenium	3.9E+(02 nc	1.0E+04 nc		1.8E+02	nc	5.0E+00	3.0E-01
1	5 0E-03	h		0	01	630-10-4	Selenourea	13.1E+0	J2 nc	4.412+03 nc	3 36+02	1.000+02	nc		
1	9 UE 02	. <u>'</u>	6 OE 02	1 0	01	74051 80-2	Seinoxyaim	1 2 0 - 1	12 ac	1 0F+04 mc	J.JL 102	1 85+02	- 00	3.4E+01	2 0E+00
1.05.01	5 0E-03	1			••	7440 22 4	Simazine	4.1E+	00	2.1E+01 m	5.6E-02	a 5.6E-01	c.a		
122.01	4 0E-03	1 122 01	1 206.03			26628-22-8	Sodium azide								
2 7E-01	h 3 0E-02	2 7E-01	1 3 0E-02	(0	01	148-18-5	Sodium diethyldithiocarbamate	1.8E+	00 ca	9.1E+00 ca	2 5E-02	a 2.5E-01	ca		
	2 0E 05	•	2 UE-05	r O	01	62·74 8	Sodium fluoroacetate	1.2E+	00 nc	1.8E+01 nc	7.3E-02	nc 7.3E-01	AC		
L	1 0E-03	h	1 OE 03	10	0.1	13718-26-8	Sodium metavanadate	6.1E+	U1 nc	8.8E+02 nc	3.7E+00	nc 3.6E+01	nc		
	6 0E 01	i		0		7440-24-6	Strontium, stable	14./12+	U4) nc 0.1	1.012+03 max	1 15+00	2.20104	nc		
	3 DE 04	1	3 0E-04	• •	0.1	57-24-9	Styrepe	1.001	01 10	2.0E702 AC	1.1E+03	nc 1.1E701		4.0E+00	2.0E-01
	1 OF 03	· · · · · · · · · · · · · · · · · · ·	1 05.03			80.07.9	1.1'-Sultonylbis (4-chlorobenzene)	7 8E+	01	2.0E+03 nc	3.7E+00	nc 3.6E+01	00		
	2 5E-02	•	2 5E-02		01	88671-89-0	Systhane	1.5E+	03 nc	2.2E+04 nc	9.1E+01	nc 9.1E+02	AC		
1 5E+05	h	1 5E+05	h	0	0 03	1746-01-6	2,3,7,8-TCDD (dioxin)	3.9E-	06 ca	2.7E-05 ca	4.5E-08	ca 4.5E-07	C8		
	7 0E-02	i	7 0E-02	1 0	01	34014-18-1	Tebuthiuron	4.3E+	03 nc	6.2E+04 nc	2.6E+02	nc 2.6E+03	nc		
1	2 ØE-02	h	2 0E-02	1 0	01	3383-96-8	Temephos	1.2E+	03 nc	1.8E+04 nc	7.3E+01	ne 7.3E+02	AC .		
1	1 3E-02	1	1 3E-02	1 0	01	5902-51-2	I erbacil	17.9E+	UZ nc	1.1E+04 Ac	4./E+01	nc 4.7E+02	nc		

							FOR PLANNING I		P	OSES			< 10X ca)			
	TOXIC	ITY INFORM	ATION	-		a - 15 a - 17	CONTAMINANT	PREL	MIN	ARY REMEDI	ATION GC	DALS	i (PRGs)	aran da ka	SOIL SCREI	ENING LEVEL
SFo 1/(mg/kg·d)	RiDo (mg/kg-d)	SFi 1/(mg/kg-d)	RIÐi (mg/kg-d)	Vs Oa Cs	kin bs oils <u>8</u>	CAS No		Residen Sod (mg	tial /kg) diacto	Industriai Soil (ing/kg)	Ambient / (ug/m^3	Air I) Isac4:21	Tap Wate (ug/l)		Migration to Gro DAF 20 K Mg (mg/kg)	und Water DAF 1 (mg/kg)
	2 5E-05	h	2 5E-05	()	0.1	12071 70 0	Terbulos	11 55+00		2.25+01	0.15.00		0.15.01		· · · · · · · · · · · · · · · · · · ·	
	1 UE-03		1 UE-03	10	01	886-50-0	Terbultyn	6 1E+01	AC	2.2E+U1 nc	9.1E-02 3.7E+00	nc	9.1E-01 3.6E+01	nc		
	3 0E-04	1	3 0E -04	1 0	01	85-94-3	1,2,4,5-Tetrachlorobenzena	1.8E+01	nc	2.6E+02 AC	1.1E+00	nc	1.1E+01	nc nc		
2 6E-02	3 0E 02	2 6E-02	1 3 0E-02	r 1		630-20 6	1.1.1.2-Tetrachloroethane	3.0E+00	ca	7.0E+00 ca	2.6E-01	ca	4.3E-01	ca		
2 0E-01 5 2E-02	i 60E-02	n 20E-D1	1 6 UE-02	()		79-34-5	1.1.2.2-Tetrachloroethane	3.8E-01	ca	9.0E-01 ca	3.3E-02	ca	5.5E-02	ca	3.0E-03	2.0E-04
		1 202.03	N 112-01	<u> </u>		127-18-4		5.7E+00	c	1.9E+01 ca	3.31:+00	CB.	1.1E+00	C 8	6.0E-02	3.0E-03
	3 UE-02		3 OE -02	. 0	01	58-90-2	2.3.4.6-Tetrachlorophenol	1 8E+03	00	2.6E+04	1 1E+02		1 1E+03			
2 0E+01	h	2 0E •01	1	0	01	5216-25-1	p,a,a,a-Tetrachlorotoluene	2.4E-02	ca	1.2E-01 ca	3.4E-04	C.2	3.4E-03	нс са		
2 4E-02	h 3.0E-02	1 2 4E-02	1 3 0E-02	1 0	01	961-11-5	Tetrachlorovinphos	2.0E+01	ca*	1.0E+02 ce	2.8E-01	ca	2.8E+00	ca		
	5 0E-04	ı	5 0E-04	10	01	3689-24-5	Tetraethyldithiopyrophosphate	3.1E+01	nc	4.4E+02 nc	1.8E+00	nc	1.8E+01	nc		
7 6E-03	n 21E-01	n 68E-03	n 86E-02	n 0	01	109-99-9	Tetrahydrofuran	6.4E+01	ca	3.2E+02 ca	9.9E-01	ca	8.8E+00	63		
	0 02-05	1	4 05 00	0		7446-18-6	Thisbasses	15.2E+00	nc	1.3E+02 nc			2.4E+00	nc		
	102-02	1	1 0E-02	, 0	01	28249-77-6	Thiopencarp	10.1E+02	nc	8.8E+03 nc	3.76+01	nc	3.6E+02	UC		
	3 0E-04	h	3 OE -04	, 0	01	39196 18 4	Thiofanox	1185+01	nc DC	26E+02 max	115+00	<u>nc</u>	3.06+03	AC	····-	
	6 0E 02		8 OE 02	1 0	01	23564-05 8	Thiophanate-methyl	4.9E+03	40	7.0E+04 nc	2.9E+02	110	2.9E+03	0C		
	5 OE -0'1	•	5 GE 03	10	01	137 26 8	Thiram	3.1E+02	41C	4.4E+03 nc	1.8E+01	nc	1.8E+02	nc		
	6 0E-01	h		0			Lin (inorganic, see tributyltin oxide for organic tin)	4.7E+04	nc	1.0E+05 max			2.2E+04	nc		
3 25+00	200:01	3 2E+00	1 (E-01	n 1 0	0.1	105-88-3	Toluene 2 4 diamine	5.2E+02	sat	5.2E+02 sat	4.0E+02	nc	7.2E+02	nc	1.2E+01	6.0E-01
	6 0E-01	h	6 0E-01	, 0	01	95-70 5	Toluene-2 5-diamine	37E+04	C8	105+05 mm	2.12-03	<u></u>	2.16-02	63		
	2 0E-01	ħ	2 0E-01	1 0	01	823-40-5	Toluene-2.6-diamine	1.2E+04		1.0E+05 max	7.3E+02	80	7.3E+03	00		
1 9E-01	I	1 9E-01	(0	01	106-49-0	p-Toluidine	2.6E+00	Câ	1.3E+01 ca	3.5E-02	CB	3.5E-01	68		
1 1E+00	1	1 1E+00	•	0	0.1	8001-35-2	Toxaphene	4.4E-01	ca	2.2E+00 ca	6.0E-03	C3	6.1E-02	ca	3.1E+01	2.0E+00
	7 5E-03	1	7 5E-03	1 0	01	66841-25-6	Traiomethrin	4.6E+02	nc	6.6E+03 nc	2.7E+01	AC.	2.7E+02	nc		
	1 0E-02	<u></u>	1 JE-02	1 0	01	2303-17-5		17.92+02	nc	1.112+04 nc	4.76+01	AC	4.7E+02	nc		
	5 DE-03		5 0E-03		01	615-54-3	1.2.4-Tribromobenzene	3 1E+02	100	4 4 E + 03 m	1.8E+01	nc oc	1.8E+02	nc		
	3 DE 04			0	01	56-35-9	Tributyltin oxide (TBTO)	1.8E+01	nc	2.6E+02 nc			1.1E+01	nc	l .	
3 4E-02	h	0.4E-02	1	0	01	634-93-5	2.4.6-Trichloroaniline	1.4E+01	C 2	7.3E+01 ca	2.0E-01	C2	2.0E+00	C a		
2 9E-02	h	5 AE-03	1	0	01	33663-50-2	2.4.6-Trichloroaniline hydrochloride	1.7E+01	Cù	8.5E+01 ca	2.3E-01	ca	2.3E+00	ca		
	1 OE-02	<u> </u>	5 7E-02	h 1		120-82-1	1,2,4-Trichlorobenzene	6.5E+02	nc	3.0E+03 sat	2.1E+02	nc	1.9E+02	nc	5.0E+00	3.0E-01
6.75.02	2 DE 02	D	2 9E-01	n 1		71-55-6	1,1,1-Irichloroethane	6.3E+02	nc	1.41:+03 sai	1.0E+03	nc	5.4E+02	UC	2.0E+00	1.0E-01
1 15-02	1 40E-03	1 305-03	0 605-03	1 1		79.01.6	Trichloroethylene (TCE)	2.8E+00	C8-	6 1E+00 at	1 1 1 + 00	68	1.6F+00	C3	6 0E-02	3.0E-04
	3 0E-01		2 0E-01	h 1		75.69.4	Trichlorofluoromethane	3 9E+02		2 0E+03 sat	7.3E+02		1 3E+03		0.01.02	0.02.00
	1 0E 01	1	1 0E 01	1 0	01	95-95-4	2.4.5-Trichlorophenol	6.1E+03	nc	8.8E+04 nc	3.7E+02	nc	3.6E+03	AC	2.7E+02	1.4E+01
1 1E-02	1	1 1E-02	1	0	01	88 06 2	2,4,6-Trichlorophenol	4.4E+01	69	2.2E+02 ca	6.2E-01	C a	6.1E+00	ca	2 0E-01	8.0E-03
	1 OE 02	1	1 UE-02	1 0	01	93-76-5	2.4.5-Trichlorophenoxyacetic Acid	6.1E+02	nc	8.8E+03 nc	3.7E+01	nc	3.6E+02	nc		
	8 0E 03	•	8 0E-03		01	93-72-1	2-(2.4.5-) richlorophenoxy) propionic acid	1 9E+02	nc	7.0E+03 AC	2.9E+01	nc	2.9E+02 3.0E+01	nc		
7 0E+00	5 0E-03	1 7.0E+00	5 0E-03			96.18.4	1 2 3- Lichloropropane	145-03	ne C2	3 1E-03 0	9.6E-04		1 6F-03		·	
	5 UE U3	n	5 UE-03	6 1		96-19-5	1.2.3-Trichloropropene	1.2E+01	'nc	3.9E+01 nc	1.8E+01	nc	3.0E+01	AC		
L	3 0E + 01	1	8 6E + 00	h 1		76-13-1	1,1,2-Trichloro-1,2,2-trifluoroethane	5.6E+03	6 A (5.6E+03 sat	3.1E+04	UC	5.9E+04	nc	1	
	3 OE 03		3 OE -03	10	01	58138 08-2	Tridiphane	1.8E+02	nc	2.6E+03 nc	1.1E+01	nc	1.1E+02	nc		
7 76-03	2 0E-03	1 1 / 7E-03	2 0E-03	11	0.1	121-44-8	Trifluralio	12.3E+01	AC .	0.01+01 AC	7.3E+00	AC	1.2E+01	nc	1	
}	1 4E-04	1	1 4E-04		01	552-30 7	Trimellitic Anhydride (TMAN)	8 6E+00		1.2E+02 m	5.1E-01	00°	5.16+00	c1.	l	
	5 OE-02	•	1 7E-03	n 1		95 63 6	1.2.4-Trimethylbenzene	5.2E+01	AC	1.7E+02 no	6.2E+00	ne	1.2E+01	nc	1	
175.02	6 06-02	11 2 75 62	176-03	A 1		108-87-8	1,3,5-Trimethylbenzene	2.1E+01	ne	7.0E+01 ne	6.2E+00	nc	1.2E+01	nc		
3,2.02	" 3 0E-02	37C-U2	3 05.07	0 1 0	01	012-00-1 99-35-4	1.3.5-Trinitrobenzene	1.36+01	ca	0./E+U1 ca	1.86-01	ca	1.8E+00	ca		
1	1 DE -02	h	1 0E-02		01	479-45-8	Trinitrophenylmethylnitramine	6 1F+02	nc nc	8.8E+03	3.76+01	ec.	1.12103	nc		

Key i=IRIS I=	HEAST N=NC	EA .=WITHDR	AWN	G=OULEI EPA	DOCL	MENTS	-ROUTE EXTR	POLATION CHECANCER PRG ACENONCANCER PRG STIESOIL SATURA	ATION max=	CEILING	GLIMIT "(where: Ac	< 100X ca) **(wh	ere: nr	c < 10X ca)			
	ΤΟΧΙΟ	ITY INFO	RMA	ATION				FOR PLANNING F			POSE		DAL:	S (PRGs)		SOIL SCREE	
j SFo 1/(mg/kg∘d)	ftiDo (mg/kg-d)	SFi 1/(mg/kg-	-a)	RtDi (mg/kg-d)		/ skin D abs C soils	CAS No		Resid Soil (tential mg/kg)	Industria) Soil (mg/kg	Ambient / (ug/m^:	∖ir i) jjølanni	Tap Wate (ug/l)	翻	Migration to Grou DAF 20 (mg/kg)	ind Water DAF נוסיינים (חיזיאין)
3.0F-02	5.0E-04	1 3 0F 02	·····	5.05.04		0 01	119.06.7	12.4.6-Tripitrololuene	1166+0	1	B 2E+01	2 2 2 5 .01		2 25+00			
	1.06-01	0	•	104.01			701.28.6	Trinbenvinboshine oxide	6 1E+0	13	8.8E+04	376+02	C8	3.65+03	Ca		
1 4E-02 n	3 0E-01	n 14E-02		3 0E-01	,	01	115-96-8	Tris(2-chloroethyl) phosphate	3.5E+0)1	18E+02	4 8E-01	11C	4 8E+00			
	2 0E-04	0				<u> </u>	7440-61-0	Uranium (chemical toxicity only)	1 6F+0	11	4 1E+02			7 3E+00			
1	101 01					u	7440 62 2	Vanadium and compounds	5 5E+0	2 14	1.4E+04			2 6E+02		6 0E+03	3 0E+02
	106-03			1 OE-03		0 01	1029 77-7	Vernam	6 1E+0	01	8.8E+02 n	3.7E+00	AC	3 6E+01			
	2 5E-02	4		2 5E-02	1	0 01	50471-44-8	Vinclozolin	1.5E+0) <u>)</u> "	2.2E+04 n	9.1E+01	nc	9.1E+02			
	1 UE+00	n		5 7E -02		1	108-05-4	Vinvl acetate	4.3E+0	02	1.4E+03 n	: 2.1E+02	nc	4.1E+02	AC	1.7E+02	8.0E+00
1 1E-01 /	8 6E-04	r 1.1E-01	h	8 6E-04		1	593-60-2	Vinyl bromide (bromoethene)	1.9E-0)1 ca	4.2E-01 a	• 6.1E-02	ca.	1.0E-01	ca.		
1 5E+00 i	3 DE -03	1 3 1E-02	-	2 9E-02	1	1	75-01-4	Vinyl chloride (child/adult)	1.5E-0	11 ca		2.2E-01	ca	4.1E-02	Ca	1.0E-02	7.0E-04
7 5E-01 i	3 0E-03	i 1.6€-02		2 9E-02		1	75-01-4	Vinyl chloride (adult)			8.3E-01 a						
	3 UE 04	1		3 UE-04		0 01	81-81-2	Wartarin	1.8E+0)1 🔐	2.6E+02 a	1.1E+00	00	1.1E+01	nc		
1	2 0E+00			2 0E-01	x	1 01	1330-20-7	Xylenes	2.1E+0)2 58	2.1E+02 si	7.3E+02	nc	1.4E+03	nc	2.1E+02	1.0E+01
	3 0E-01	,				0	7440-66-6	Zinc	2.3E+0	14 00	1.0E+05 m	x		1.1E+04	nc	1.2E+04	6.2E+02
	3 UE-04	,				0	1314-84-7	Zinc phosphide	2.3E+0)1	6.1E+02			1.1E+01	nc		
	5 ØE-02			5 0F -02		0 01	12122 67.7	Zineh	13 1E+0	11	44E+04	1 8E+02		1.8E+03			

٠.

EPA/540/R-92/003 Publication 9285.7-01B December 1991

Risk Assessment Guidance for Superfund: Volume I — Human Health Evaluation Manual (Part B, Development of Risk-based Preliminary Remediation Goals)

Interim

Office of Emergency and Remedial Response U.S. Environmental Protection Agency Washington, DC 20460

(XD) Printed on Recycled Paper

NOTICE

The policies set out in this document are intended solely as guidance; they are not final U.S. Environmental Protection Agency (EPA) actions. These policies are not intended, nor can they be relied upon, to create any rights enforceable by any party in litigation with the United States. EPA officials may decide to follow the guidance provided in this document, or to act at variance with the guidance, based on an analysis of specific site circumstances. The Agency also reserves the right to change this guidance at any time without public notice.

This guidance is based on policies in the Final Rule of the National Oil and Hazardous Substances Pollution Contingency Plan (NCP), which was published on March 8, 1990 (55 Federal Register 8666). The NCP should be considered the authoritative source.

CONTENTS

		P	age								
NOT	ICE	······ i	i								
EXH	IBITS .	v	ri -								
DEF	NITIO	4S	ii								
ACR	ONYMS	ABBREVIATIONS	x								
ACK	NOWL	DGEMENTS	di l								
PRE	FACE .	······	ii								
1.0	INTRODUCTION										
	1.1	1.1 DEFINITION OF PRELIMINARY REMEDIATION GOALS									
	1.2	SCOPE OF PART B 1									
	1.3	RELEVANT STATUTES, REGULATIONS, AND GUIDANCE	1								
		1.3.1 CERCLA/SARA 3 1.3.2 National Contingency Plan 3 1.3.3 Guidance Documents 3	5								
	1.4	INITIAL DEVELOPMENT OF PRELIMINARY REMEDIATION GOALS 4	1								
	1.5	MODIFICATION OF PRELIMINARY REMEDIATION GOALS	5								
	1.6	5 DOCUMENTATION AND COMMUNICATION OF PRELIMINARY REMEDIATION GOALS									
	1.7	ORGANIZATION OF DOCUMENT	5								
2.0	IDE	IDENTIFICATION OF PRELIMINARY REMEDIATION GOALS									
	2.1	MEDIA OF CONCERN	1								
	2.2	2 CHEMICALS OF CONCERN									
	2.3	3 FUTURE LAND USE									
	2.4	APPLICABLE OR RELEVANT AND APPROPRIATE REQUIREMENTS)								
		 2.4.1 Chemical-, Location-, and Action-specific ARARs 2.4.2 Selection of the Most Likely ARAR-based PRG for Each Chemical 	10								
	2.5	EXPOSURE PATHWAYS, PARAMETERS, AND EQUATIONS	11								
		2.5.1 Ground Water/Surface Water	13								

CONTENTS (Continued)

				Page								
	2.6	TOX	ICITY INFORMATION	14								
	2.7	7 TARGET RISK LEVELS										
	2.8	MOD	IFICATION OF PRELIMINARY REMEDIATION GOALS	15								
		2.8.1	Review of Assumptions	15								
		2.8.2	Identification of Uncertainties	16								
		2.8.3	Other Considerations in Modifying PRGs	17								
		2.8.4	Post-remedy Assessment	18								
3.0	CALCULATION OF RISK-BASED PRELIMINARY											
	REMEDIATION GOALS											
	3.1	RESI	DENTIAL LAND USE	20								
		31.1	Ground Water or Surface Water	20								
		312	Soil	. 23								
	~~~	L'I L'IL										
	3.2	COM	MERCIAL/INDUSTRIAL LAND USE	24								
		3.2.1	Water	24								
		3.2.2	Soil	25								
	3.3	VOL	ATILIZATION AND PARTICULATE EMISSION FACTORS	26								
		3.3.1	Soil-to-air Volatilization Factor	26								
		3.3.2	Particulate Emission Factor	29								
	3.4	CALC	CULATION AND PRESENTATION OF RISK-BASED PRGs	30								
4.0	RIS	RISK-BASED PRGs FOR RADIOACTIVE CONTAMINANTS										
	4.1	4.1 RESIDENTIAL LAND USE										
		4.1.1	Ground Water or Surface Water	34								
		4.1.2	Soil	35								
	4.2	COM	MERCIAL/INDUSTRIAL LAND USE	36								
		4.2.1	Water	36								
		4.2.2	Soil	36								
		4.2.3	Soil-to-air Volatilization Factor	38								
	4.3	RAD	IATION CASE STUDY	38								
		4.3.1	Site History	40								
		4.3.2	At the Scoping Phase	40								
		4.3.3	After the Baseline Risk Assessment	43								
REFE	RENC	ES		47								
## CONTENTS (Continued)

	Pag	e
APPENDIX A	ILLUSTRATIONS OF CHEMICALS THAT "LIMIT" REMEDIATON 49	
APPENDIX B	RISK EQUATIONS FOR INDIVIDUAL EXPOSURE PATHWAYS	
B.1	GROUND WATER OR SURFACE WATER - RESIDENTIAL LAND USE 51	
	B.1.1 Ingestion	
	B.1.2 Inhalation of Volatiles	
B.2	SOIL - RESIDENTIAL LAND USE	
	B.2.1 Ingestion of Soil	
	B.2.2 Inhalation of Volatiles	
	B.2.3 Inhalation of Particulates	
B.3	SOIL - COMMERCIAL/INDUSTRIAL LAND USE	
	B.3.1 Ingestion of Soil	
	B.3.2 Inhalation of Volatiles	
	B.3.3 Inhalation of Particulates	

## EXHIBITS

Exhibit		Page
1-1	RELATIONSHIP OF HUMAN HEALTH EVALUATION TO THE CERCLA PROCESS	2
2-1	TYPICAL EXPOSURE PATHWAYS BY MEDIUM FOR RESIDENTIAL AND COMMERCIAL/INDUSTRIAL LAND USES	12

### DEFINITIONS

Term	Definition
Applicable or Relevant and Appropriate Requirements (ARARs)	"Applicable" requirements are those clean-up standards, standards of control, and other substantive environmental protection requirements, criteria, or limitations promulgated under federal or state law that specifically address a hazardous substance, pollutant, contaminant, remedial action, location, or other circumstance at a Comprehensive Environmental Response, Compensation, and Liability Act (CERCLA) site. "Relevant and appropriate" requirements are those clean-up standards which, while not "applicable" at a CERCLA site, address problems or situations sufficiently similar to those encountered at the CERCLA site that their use is well-suited to the particular site. ARARs can be action- specific, location-specific, or chemical-specific
Cancer Risk	Incremental probability of an individual's developing cancer over a lifetime as a result of exposure to a potential carcinogen.
Conceptual Site Model	A "model" of a site developed at scoping using readily available information. Used to identify all potential or suspected sources of contamination, types and concentrations of contaminants detected at the site, potentially contaminated media, and potential exposure pathways, including receptors. This model is also known as "conceptual evaluation model".
Exposure Parameters	Variables used in the calculation of intake (e.g., exposure duration, inhalation rate, average body weight).
Exposure Pathway	The course a chemical or physical agent takes from a source to an exposed organism. An exposure pathway describes a unique mechanism by which an individual or population is exposed to chemicals or physical agents at or originating from a site. Each exposure pathway includes a source or release from a source, an exposure point, and an exposure route. If the exposure point differs from the source, a transport/exposure medium (e.g., air) or media (in cases of intermedia transfer) also would be indicated.
Exposure Point	A location of potential contact between an organism and a chemical or physical agent.
Exposure Route	The way a chemical or physical agent comes in contact with an organism (i.e., by ingestion, inhalation, dermal contact).
⁷ inal Remediation Levels	Chemical-specific clean-up levels that are documented in the Record of Decision (ROD). They may differ from preliminary remediation goals (PRGs) because of modifications resulting from consideration of various uncertainties, technical and exposure factors, as well as all nine selection-of-remedy criteria outlined in the National Oil and Hazardous Substances Pollution Contingency Plan (NCP).

### **DEFINITIONS** (Continued)

Term	Definition			
Hazard Index (HI)	The sum of two or more hazard quotients for multiple substances and/or multiple exposure pathways.			
Hazard Quotient (HQ)	The ratio of a single substance exposure level over a specified time period to a reference dose for that substance derived from a similar exposure period.			
"Limiting" Chemical(s)	Chemical(s) that are the last to be removed (or treated) from a medium by a given technology. In theory, the cumulative residual risk for a medium may approximately equal the risk associated with the limiting chemical(s).			
Preliminary Remediation Goals (PRGs)	Initial clean-up goals that (1) are protective of human health and the environment and (2) comply with ARARs. They are developed early in the process based on readily available information and are modified to reflect results of the baseline risk assessment. They also are used during analysis of remedial alternatives in the remedial investigation/feasibility study (RI/FS).			
Quantitation Limit (QL)	The lowest level at which a chemical can be accurately and reproducibly quantitated. Usually equal to the method detection limit multiplied by a factor of three to five, but varies for different chemicals and different samples.			
Reference Dose (RfD)	The Agency's preferred toxicity value for evaluating potential noncarcinogenic effects in humans resulting from contaminant exposures at CERCLA sites. (See RAGS/HHEM Part A for a discussion of different kinds of reference doses and reference concentrations.)			
Risk-based PRGs	Concentration levels set at scoping for individual chemicals that correspond to a specific cancer risk level of 10 ⁻⁶ or an HQ/HI of 1. They are generally selected when ARARs are not available.			
Slope Factor (SF)	A plausible upper-bound estimate of the probability of a response per unit intake of a chemical over a lifetime. The slope factor is used to estimate an upper-bound probability of an individual's developing cancer as a result of a lifetime of exposure to a particular level of a potential carcinogen.			
Target Risk	A value that is combined with exposure and toxicity information to calculate a risk-based concentration (e.g., PRG). For carcinogenic effects, the target risk is a cancer risk of 10 ⁻⁶ . For noncarcinogenic effects, the target risk is a hazard quotient of 1.			

### ACRONYMS/ABBREVIATIONS

Acronym/ Abbreviation	Definition					
ARARs	Applicable or Relevant and Appropriate Requirements					
CAA	Clean Air Act					
CERCLA	Comprehensive Environmental Response, Compensation, and Liability Act					
CFR	Code of Federal Regulations					
CWA	Clean Water Act					
EAG	Exposure Assessment Group					
ECAO	Environmental Criteria and Assessment Office Superfund Health Risk Technical Support Center					
EF	Exposure Frequency					
EPA	U.S. Environmental Protection Agency					
FWQC	Federal Water Quality Criteria					
HEAST	Health Effects Assessment Summary Tables					
HHEM	Human Health Evaluation Manual					
н	Hazard Index					
HQ	Hazard Quotient					
HRS	Hazard Ranking System					
IRIS	Integrated Risk Information System					
LLW	Low-level Radioactive Waste					
MCL	Maximum Contaminant Level					
MCLG	Maximum Contaminant Level Goal					
NCP	National Oil and Hazardous Substances Pollution Contingency Plan					
NPL.	National Priorities List					
OSWER	Office of Solid Waste and Emergency Response					
OERR	Office of Emergency and Remedial Response					

-ix-

### ACRONYMS/ABBREVIATIONS (Continued)

Acronyms/ Abbreviation	Definition	
PA/SI	Preliminary Assessment/Site Inspection	
PEF	Particulate Emission Factor	
PRG	Preliminary Remediation Goal	
RAGS	Risk Assessment Guidance for Superfund	
RCRA	Resource Conservation and Recovery Act	
RfC	Reference Concentration	
RfD	Reference Dose	
RI/FS	Remedial Investigation/Feasibility Study	
RME	Reasonable Maximum Exposure	
ROD	Record of Decision	
RPM	Remedial Project Manager	
SARA	Superfund Amendments and Reauthorization Act	
SDWA	Safe Drinking Water Act	
SF	Slope Factor	
TR	Target Risk	
VF	Volatilization Factor	
WQS	State Water Quality Standards	0 2 <u>—1753</u>

-x-

### ACKNOWLEDGEMENTS

This manual was developed by the Toxics Integration Branch (TIB) of EPA's Office of Emergency and Remedial Response, Hazardous Site Evaluation Division. A large number of EPA Regional and Headquarters managers and technical staff provided valuable input regarding the organization, content, and policy implications of the manual throughout its development. We would especially like to acknowledge the efforts of the staff in the Regions, as well as the following offices:

Guidance and Evaluation Branch, Office of Waste Programs Enforcement;

Remedial Operations and Guidance Branch, Office of Emergency and Remedial Response;

- Policy and Analysis Staff, Office of Emergency and Remedial Response;
- Environmental Response Branch, Office of Emergency and Remedial Response;
- Office of General Counsel; and
- Exposure Assessment Group, Office of Research and Development.

ICF Incorporated (under EPA Contract Nos. 68-01-7389, 68-W8-0098, and 68-03-3452), S. Cohen and Associates (under EPA Contract No. 68-D9-0170), and Environmental Quality Management, Incorporated (under EPA Contract No. 68-03-3482), provided technical assistance to EPA in support of the development of this manual.

### PREFACE

Risk Assessment Guidance for Superfund: Volume I – Human Health Evaluation Manual (RAGS/HHEM) Part B is one of a three-part series. Part A addresses the baseline risk assessment; Part C addresses human health risk evaluations of remedial alternatives. Part B provides guidance on using U.S. Environmental Protection Agency (EPA) toxicity values and exposure information to derive risk-based preliminary remedial goals (PRGs) for a Comprehensive Environmental Response, Compensation, and Liability Act (CERCLA) site. Initially developed at the scoping phase using readily available information, riskbased PRGs generally are modified based on site-specific data gathered during the remedial investigation/feasibility study (RI/FS). This guidance does not discuss the risk management decisions that are necessary at a CERCLA site (e.g., selection of final remediation goals). The potential users of Part B are those involved in the remedy selection and implementation process, including risk assessors, risk assessment reviewers, remedial project managers, and other decision-makers.

This manual is being distributed as an interim document to allow for a period of field testing and review. RAGS/HHEM will be revised in the future, and Parts A, B, and C will be incorporated into a single final guidance document. Additional information for specific subject areas is being developed for inclusion in a later revision. These areas include:

- development of goals for additional land uses and exposure pathways;
- development of short-term goals;
- additional worker health and safety issues; and
- determination of final remediation goals (and attainment).

Comments addressing usefulness, changes, and additional areas where guidance is needed should be sent to:

> U.S. Environmental Protection Agency Toxics Integration Branch (OS-230) Office of Emergency and Remedial Response 401 M Street, SW Washington, DC 20460

Telephone:	202-260-9486
FAX:	202-260-6852

# CHAPTER 1 INTRODUCTION

The purpose of this guidance is to assist risk assessors, remedial project managers (RPMs), and others involved with risk assessment and decisionmaking at Comprehensive Environmental Response, Compensation, and Liability Act (CERCLA) sites in developing preliminary remediation goals (PRGs). This guidance is the second part (Part B) in the series Risk Assessment Guidance for Superfund: Volume I – Human Health Evaluation Manual (RAGS/HHEM).

Part A of this series (EPA 1989d) assists in defining and completing a site-specific baseline risk assessment; much of the information in Part A is necessary background for Part B. Part B provides guidance on using U.S. Environmental Protection Agency (EPA) toxicity values and exposure information to derive risk-based PRGs. Initially developed at the scoping phase using readily available information, risk-based PRGs generally are modified based on site-specific data gathered during the remedial investigation/feasibility study (RI/FS). Part C of this series (EPA 1991d) assists RPMs, sitc engineers, risk assessors, and others in using risk information both to evaluate remedial alternatives during the FS and to evaluate the selected remedial alternative during and after its implementation. Exhibit 1-1 illustrates how the three parts of RAGS/HHEM are all used during the RI/FS and other stages of the site remediation process.

The remainder of this introduction addresses the definition of PRGs, the scope of Part B, the statutes, regulations, and guidance relevant to PRGs, steps in identifying and modifying PRGs, the communication and documentation of PRGs, and the organization of the remainder of this document.

### 1.1 DEFINITION OF PRELIMINARY REMEDIATION GOALS

In general, PRGs provide remedial design staff with long-term targets to use during analysis and selection of remedial alternatives. Ideally, such goals, if achieved, should both comply with applicable or relevant and appropriate requirements (ARARs) and result in residual risks that fully satisfy the National Oil and Hazardous Substances Pollution Contingency Plan (NCP) requirements for the protection of human health and the environment. By developing PRGs carly in the decision-making process (before the RI/FS and the baseline risk assessment are completed), design staff may be able to streamline the consideration of remedial alternatives.

Chemical-specific PRGs are concentration goals for individual chemicals for specific medium and land use combinations at CERCLA sites. There are two general sources of chemical-specific PRGs: (1) concentrations based on ARARs and (2) concentrations based on risk assessment. ARARs include concentration limits set by other environmental regulations (e.g., non-zero maximum contaminant level goals [MCLGs] set under the Safe Drinking Water Act [SDWA]). The second source for PRGs, and the focus of this document, is risk assessment or risk-based calculations that set concentration limits using carcinogenic and/or noncarcinogenic toxicity values under specific exposure conditions.

### 1.2 SCOPE OF PART B

The recommended approach for developing remediation goals is to identify PRGs at scoping, modify them as needed at the end of the RI or during the FS based on site-specific information from the baseline risk assessment, and ultimately select remediation levels in the Record of Decision (ROD). In order to set chemical-specific PRGs in a site-specific context, however, assessors must answer fundamental questions about the site. Information on the chemicals that are present onsite, the specific contaminated media, land-use assumptions, and the exposure assumptions behind pathways of individual exposure is necessary in order to develop chemical-specific PRGs. Part B provides guidance for considering this information in developing chemical-specific PRGs.

### EXHIBIT 1-1

### RELATIONSHIP OF THE HUMAN HEALTH EVALUATION TO THE CERCLA PROCESS



Because Part B focuses on developing chemical-specific PRGs based on protection of human health, there are important types of information that are not considered and that may significantly influence the concentration goals needed to satisfy the CERCLA criteria for selection of a remedy. For example, <u>no</u> consideration is given to ecological effects in this guidance. Other types of remedial action "goals" not addressed in detail include action-specific ARARs (e.g., technology- or performance-based standards) and location-specific ARARs.

Throughout Part B, the term "chemicalspecific" should be understood to refer to both nonradioactive and radioactive chemical hazardous substances, pollutants, or contaminants. Therefore, the process described in this guidance of selecting and modifying PRGs at a site should be applied to each radionuclide of potential concern. Chapter 10 of RAGS/HHEM Part A provides background information concerning radionuclides, and Chapter 4 of RAGS/HHEM Part B includes radionuclide risk-based equations and a case study of a hypothetical radiation site.

This guidance only addresses in detail the initial selection of risk-based PRGs. Detailed guidance regarding other factors that can be used to further modify PRGs during the remedy selection process is presented in other documents (see Section 1.3).

### 1.3 RELEVANT STATUTES, REGULATIONS, AND GUIDANCE

This section provides relevant background on the CERCLA statute and the regulations created to implement the statute (i.e., the NCP). In addition, other CERCLA guidance documents are listed and their relationship to the site remediation process is discussed.

#### 1.3.1 CERCLA/SARA

CERCLA, as amended by the Superfund Amendments and Reauthorization Act of 1986 (SARA), is the authority for EPA to take response actions. (Throughout this guidance, reference to CERCLA should be understood to mean "CERCLA as amended by SARA.")

Several sections of CERCLA, especially section 121 (Clean-up Standards), set out the requirements and goals of CERCLA. Two fundamental requirements are that selected remedies be protective of human health and the environment, and comply with ARARs. CERCLA indicates a strong preference for the selection of remedial alternatives that permanently and significantly reduce the volume, toxicity, or mobility of wastes. To the maximum extent practicable, the selected remedial alternatives should effect permanent solutions by using treatment technologies. Both the law and the regulation (see below) call for cost-effective remedial alternatives.

#### 1.3.2 NATIONAL CONTINGENCY PLAN

Regulations implementing CERCLA are found in Volume 40 of the Code of Federal Regulations (CFR), Part 300, and are referred to collectively as the NCP. Section 300.430 of the NCP, and several portions of the preambles in the Federal Register (55 Federal Register 8666, March 8, 1990 and 53 Federal Register 51394, December 21, 1988), address how the Superfund and other CERCLA programs are to implement the Act's requirements and goals concerning clean-up levels.

Nine criteria have been developed in the NCP to use in selecting a remedy. These criteria are listed in the next box. The first criterion — overall protection of human health and the environment — is the focus of this document. This criterion coupled with compliance with ARARs are referred to as "threshold criteria" and must be met by the selected remedial alternative. PRGs are developed to quantify the standards that remedial alternatives must meet in order to achieve these threshold criteria. See the second box on the next page for highlights from the NCP on remediation goals.

#### 1.3.3 GUIDANCE DOCUMENTS

There are several existing documents that provide gudiance on related steps of the site remediation process. These documents are described in the box on page five. When documents are referenced throughout this guidance, the abbreviated titles, indicated in parentheses after the full titles and bibliographic information, are used.

#### NINE EVALUATION CRITERIA FOR ANALYSIS OF REMEDIAL ALTERNATIVES (40 CFR 300.430(e)(9)(iii))

#### Threshold Criteria:

- · Overall Protection of Human Health and the Environment
- Compliance with ARARs

**Balancing Criteria:** 

- · Long-term Effectiveness and Permanence
- Reduction of Toxicity, Mobility, or Volume Through Treatment
- Short-term Effectiveness
- · Implementability
- Cost
- **Modifying Criteria:** State Acceptance
- · Community Acceptance

### 1.4 INITIAL DEVELOPMENT OF PRELIMINARY REMEDIATION GOALS

The NCP preamble indicates that, typically, PRGs are developed at scoping or concurrent with initial RI/FS activities (i.e., prior to completion of the baseline risk assessment). This carly determination of PRGs facilitates development of a range of appropriate remedial alternatives and can focus selection on the most effective remedy.

Development of PRGs early in the RI/FS requires the following site-specific data:

- media of potential concern;
- chemicals of potential concern; and
- probable future land use.

This information may be found in the preliminary. assessment/site inspection (PA/SI) reports or in the conceptual site model that is developed prior to or during scoping. (When a site is listed on the National Priorities List [NPL], much of this information is compiled during the PA/SI as part of the Hazard Ranking System [HRS] documentation record.) Once these factors are known, all potential ARARs must be identified. When ARARs do not exist, risk-based PRGs are calculated using EPA health criteria (i.e., reference doses or cancer slope factors) and default or sitespecific exposure assumptions.

#### NCP RULE HIGHLIGHTS RISK AND REMEDIATION GOALS (40 CFR 300.430(e)(2))

"In developing and, as appropriate, screening ... alternatives, the lead agency shall: (i) Establish remedial action objectives specifying contaminants and media of concern, potential exposure pathways, and remediation goals. Initially, preliminary remediation goals are developed based on readily available information, such as chemicalspecific ARARs or other reliable information. Preliminary remediation goals should be modified, as necessary, as more information becomes available during the RI/FS. Final remediation goals will be determined when the remedy is Remediation goals shall establish selected. acceptable exposure levels that are protective of human health and the environment and shall be developed by considering the following:

- (A) Applicable or relevant and appropriate requirements ..., and the following factors:
  - (1) For systemic toxicants, acceptable exposure levels shall represent concentration levels to which the human population, including sensitive subgroups, may be exposed without adverse effect during a lifetime or part of a lifetime, incorporating an adequate margin of safety;
  - (2) For known or suspected carcinogens, acceptable exposure levels are generally concentration levels that represent an excess upper-bound lifetime cancer risk to an individual of between 10r4 and 10r6 using information on the relationship between dose and response. The 104 risk level shall be used as the point of departure for determining remediation goals for alternatives when ARARs are not available or are not sufficiently protective because of multiple contaminants at a site or multiple pathways of exposure ..."

It is important to remember that risk-based PRGs (either at scoping or later on) are initial guidelines. They do not establish that cleanup to A risk-based meet these goals is warranted. concentration, as calculated in this guidance, will be considered a final remediation level only after appropriate analysis in the RI/FS and ROD.

#### GUIDANCE DOCUMENTS

- Risk Assessment Guidance for Superfund: Volume I Human Health Evaluation Manual Part A (EPA 1989a) (RAGS/HHEM Part A) contains background information and is particularly relevant for developing exposure and toxicity assessments that are required when refining chemical-specific risk-based concentrations, and accounting for site-specific factors such as multiple exposure pathways.
- Guidance for Conducting Remedial Investigations and Feasibility Studies Under CERCLA (EPA 1988c) (RI/FS Guidance) presents detailed information about implementing the RI/FS and general information on the use of risk-based factors and ARARs in the context of the RI/FS.
- Guidance on Remedial Action for Contaminated Ground Water at Superfund Sites (EPA 1988d) (Ground-water Guidance) details some of the key issues in development, evaluation, and selection of ground-water remedial actions at CERCLA sites.
- CERCLA Compliance with Other Laws Manuals (Part I, EPA 1988a; and Part II, EPA 1989a) (CERCLA Compliance Manuals) provide guidance for complying with ARARs. Part I addresses the Resource Conservation and Recovery Act (RCRA), the Clean Water Act (CWA), and the SDWA; Part II addresses the Clean Air Act (CAA), other federal statutes, and state requirements.
- Methods for Evaluating the Attainment of Cleanup Standards (Volume 1: Soils and Solid Waste) (EPA 1989e) and Methods for Evaluating the Attainment of Cleanup Standards (Volume 2: Water) (Draft, 1988, EPA, Statistical Policy Branch) (Attainment Guidance) provide guidance on evaluating the attainment of remediation levels, including appropriate sampling and statistical procedures to test whether the chemical concentrations are significantly below the remediation levels.
- Interim Final Guidance on Preparing Superfund Decision Documents (EPA 1989b) (ROD Guidance) provides guidance that: (1) presents standard formats for documenting CERCLA remedial action decisions; (2) clarifies the roles and responsibilities of EPA, states, and other federal agencies in developing and issuing decision documents; and (3) explains how to address changes made to proposed and selected remedies.
- Catalog of Superfund Program Publications, Chapter 5 (EPA 1990a) lists all ARARs guidance documents that
  have been issued by EPA, shown in order of date of issuance.
- Role of the Baseline Risk Assessment in Superfund Remedy Selection Decisions (EPA 1991c) provides clarification
  on the role of the baseline risk assessment in developing and selecting CERCLA remedial alternatives.
- Guidance for Data Useability in Risk Assessment (EPA 1990b) (Data Useability Guidance) provides guidance on how to obtain a minimum level of quality for all environmental analytical data required for CERCLA risk assessments. It can assist with determining sample quantitation limits (SQLs) for chemical-specific analyses.
- Guidance on Remedial Actions for Superfund Sites with PCB Contamination (EPA 1990c) describes the
  recommended approach for evaluating and remediating CERCLA sites having PCB contamination.
- Conducting Remedial Investigations/Feasibility Studies for CERCLA Municipal Landfill Sites (EPA 1991a) (Municipal Landfill Guidance) offers guidance on how to streamline both the RI/FS and the selection of a remedy for municipal landfills.

### 1.5 MODIFICATION OF PRELIMINARY REMEDIATION GOALS

The initial list of PRGs may need to be revised as new data become available during the RI/FS. Therefore, upon completion of the baseline risk assessment, it is important to review the media and chemicals of potential concern, future land use, and exposure assumptions originally identified at scoping. Chemicals may be added or dropped from the list, and risk-based PRGs may need to be recalculated using site-specific exposure factors. PRGs that are modified based on the results of the baseline risk assessment must still meet the

-5-

"threshold criteria" of: (1) protection of human health and the environment and (2) compliance with ARARs. However, the NCP also allows for modification of PRGs during final remedy selection based on the "balancing" and "modifying" criteria and factors relating to uncertainty, exposure, and technical feasibility.

Final remediation levels are not determined until the site remedy is ready to be selected; final remediation levels are then set out in the ROD. PRGs are refined into final remediation goals throughout the process leading up to remedy selection. The ROD itself, however, should include a statement of final clean-up levels based on these goals, as noted in NCP section 300.430(e)(2)(i)(A). In the ROD, it is preferable to use the term "remediation <u>level</u>" rather than "remediation <u>goal</u>" in order to make clear that the selected remedy establishes binding requirements.

### 1.6 DOCUMENTATION AND COMMUNICATION OF PRELIMINARY REMEDIATION GOALS

Clear and concise communication of risk-based PRGs among the risk assessor, the RPM, the ARARs coordinator, site engineers, analytical chemists, hydrogeologists, and others is important in the development of PRGs. The involvement of the RPM in the direction and development of risk-based PRGs is important to ensure that communication is facilitated and that the PRGs are used effectively in streamlining the RI/FS process.

Because PRGs are most useful during the RI/FS (e.g., for streamlining the consideration of remedial alternatives), it is important to communicate them to site engineers as soon as possible. A memorandum from either the site risk assessor or the RPM to the site engineers and others concerned with PRGs would be appropriate for transmitting the initial PRGs. A brief cover page could highlight key assumptions, as well as changes, if any, to the standard equations (i.e., those presented in this guidance). Following this brief discussion, the PRGs could be presented using a table similar to that in Section 3.4 of this guidance.

The RI/FS Guidance recommends that "chemical- and/or risk-based remedial objectives associated with the alternative should be documented in the final RI/FS report to the extent possible." Therefore, the RI/FS report is a logical place to present PRGs that have been modified after the baseline risk assessment. A summary table such as the one developed in Section 3.4 of Part B could be incorporated into the RI/FS following the presentation of the baseline risk assessment. Along with the table, a discussion of issues of particular interest, such as assumptions used and the relationship between ARARs and risk-based PRGs at the site, could be included. Also, it is always appropriate to discuss how findings of the baseline risk assessment were incorporated into the calculation of PRGs.

### 1.7 ORGANIZATION OF DOCUMENT

The remainder of this guidance is organized into three additional chapters and two appendices. Chapter 2 discusses the initial identification of PRGs and provides guidance for modifying appropriate values during the RI/FS. Chapter 3 outlines equations that can be used to calculate risk-based PRGs for residential and commercial/ industrial land uses. These equations are presented in both "reduced" format (i.e., incorporating certain default assumptions discussed in Chapter 2) and expanded format (i.e., with all variables included so that the user of this guidance can incorporate site-specific values). Particular considerations regarding radionuclides are provided in Chapter 4.

Appendix A supports several points made in Chapter 2 by providing illustrations of remedial alternatives where one or more chemicals "limit" remediation and, thus, represent a major portion of the residual risk. Appendix B lists equations for media-specific exposure pathways, enabling the risk assessor to derive site-specific equations that differ from those presented in Chapter 3.

Throughout Chapters 2, 3, and 4, case studies are presented that illustrate the process of determining PRGs. These case studies are contained in boxes with a shadow box appearance. Other types of boxed information (e.g., NCP quotes) is contained in boxes such as those in Chapter 1, which have thicker lines on the top and bottom than on the sides.

## CHAPTER 2

## IDENTIFICATION OF PRELIMINARY REMEDIATION GOALS

This chapter provides guidance on the initial identification of PRGs during the scoping phase of the RI/FS. As discussed in Chapter 1, medium-specific PRGs (ARAR-based and/or risk-based) should be identified during scoping for all chemicals of potential concern <u>using readily</u> <u>available information</u>. Sections are provided in this chapter on how to use this information to identify media and chemicals of potential concern, the most appropriate future land use, potential exposure pathways, toxicity information, potential ARARs, and risk-based PRGs. Finally, a section is provided on the modification of PRGs.

When using PRGs developed during scoping, the design engineers should understand that these may be modified significantly depending on information gathered about the site. The subsequent process of identifying key site contaminants, media, and other factors (i.e., during the baseline risk assessment) may require that the focus of the RI/FS be shifted (e.g., chemicals without ARARs may become more or less important). Thus, the design of remedial alternatives should remain flexible until the modified (i.e., more final) PRGs are available.

Prior to identifying PRGs during scoping, a conceptual site model should be developed (see the next box). Originally developed to aid in planning site activities (e.g., the RI/FS), the conceptual site model also contains information that is valuable for identifying PRGs. For example, it can be relied upon to identify which media and chemicals need PRGs. More information on developing and using a conceptual site model during the RI/FS process can be found in Chapter 2 of the RI/FS Guidance and Chapter 4 of RAGS/HHEM Part A.

To illustrate the process of calculating risk-based PRGs at the scoping stage of remediation, hypothetical CERCLA sites will be examined in boxes in appropriate sections throughout Chapters 2, 3, and 4. See the box on

#### CONCEPTUAL SITE MODEL

During project planning, the RPM gathers and analyzes available information and develops the conceptual site model (also called the conceptual evaluation model). This model is used to assess the nature and the extent of contamination. It also identifies potential contaminant sources, potential exposure pathways, and potential human and/or environmental receptors. Further, this model helps to identify data gaps and assists staff in developing strategies for data collection. Site history and PA/SI data generally are extremely useful sources of information for developing this model. The conceptual site model should include known and suspected sources of contamination, types of contaminants and affected media, known and potential routes of migration, and known or potential human and environmental receptors.

the next page for an introduction to the first site. (The radiation case study is addressed in Chapter 4.) The information (e.g., toxicity values) contained in these case studies is for illustration only, and should not be used for any other purpose. These case studies have been simplified (e.g., only ground water will be examined) so that the steps involved in developing risk-based PRGs can be readily discerned.

### 2.1 MEDIA OF CONCERN

During scoping, the first step in developing PRGs is to identify the media of potential concern. The conceptual site model should be very useful for this step. These media can be either:

 currently contaminated media to which individuals may be exposed or through which chemicals may be transported to potential receptors; or

.7-

#### CASE STUDY: INTRODUCTION

The XYZ Co. site contains an abandoned industrial facility that is adjacent to a highdensity residential neighborhood. Remnants of drums, lagoons, and waste piles were found at the site. Ground water in the area of the site is used by residents as a domestic water supply. There is also a small lake downgradient from the site that is used by some of the local residents for fishing and swimming.

 currently uncontaminated media that may become contaminated in the future due to contaminant transport.

Several important media often requiring direct remediation are ground water, surface water, soil, and sediment. Currently, only the first three of these media are discussed in this chapter and addressed by the equations provided in Chapters 3 and 4. If other media that may require the development of risk-based concentrations (e.g., sediments) are identified at scoping, appropriate equations for those media should be developed. Regional risk assessors should be consulted as early as possible to assist with this process.

#### CASE STUDY: IDENTIFY MEDIA OF CONCERN

The PA/SI for the example site indicates that ground water beneath the site is contaminated. The source of this contamination appears to have been approximately 100 leaking drums of various chemicals that were buried in the soil but have since been removed. Lagoons and waste piles also may have contributed to the contamination. Thus, ground water and soil are media of concern.

Although evidence of lake water contamination was not found during the PA/SI, there is a reasonable possibility that it may become contaminated in the future due to contaminant transport either via ground-water discharge or surface water run-off. Thus, surface water (the lake) and sediments also may be media of concern.

### 2.2 CHEMICALS OF CONCERN

This step involves developing an initial list of chemicals for which PRGs need to be developed. <u>Chapters 4 and 5 of RAGS/HHEM Part A provide</u> important additional information on identifying chemicals of potential concern for a site and should be consulted prior to development of the conceptual site model and PRGs at scoping.

Initially, the list of chemicals of potential concern should include any chemical reasonably expected to be of concern at the site based on what is known during scoping. For example, important chemicals previously detected at the site, based on the PA/SI, the conceptual site model, or other prior investigations, generally should be included. In addition, the list may include chemicals that the site history indicates are likely to be present in significant quantities, even though they may not yet be detected. Sources of this latter type of information include records of chemicals used or disposed at the facility, and interviews with current or former employees. The list also may include chemicals that are probable degradation products of site contaminants where these are determined to be potential contributors of significant risk. An environmental chemist should be consulted for assistance in determining the probable degradation products of potential site-related chemicals and their persistence under site conditions. Generally, the chemicals for which PRGs should be developed will correspond to the list of suspected site contaminants included in the sampling and analysis plan.

### 2.3 FUTURE LAND USE

This step involves identifying the most appropriate future land use for the site so that the appropriate exposure pathways, parameters, and equations (discussed in the next section) can be used to calculate risk-based PRGs. RAGS/HHEM Part A (Chapter 6) and an EPA Office of Solid Waste and Emergency Response (OSWER) directive on the role of the baseline risk assessment in remedy selection decisions (EPA 1991b) provide additional guidance on identifying future land use. The standard default equations provided in Chapter 3 of Part B only address residential and commercial/industrial land uses. If land uses other than these are to be assumed (e.g., recreational), then exposure pathways, parameters,

-8-

### CASE STUDY: IDENTIFY CHEMICALS OF CONCERN

The PA/SI for the XYZ Co. site identified the following seven chemicals in ground-water samples: benzene, ethylbenzene, hexane, isophorone, triallate, 1,1,2-trichloroethane, and vinyi chloride. Therefore, these chemicals are obvious choices for chemicals of potential concern.

Although not detected in any of the PA/SI samples, site history indicates that one other solvent — carbon tetrachloride — also was used in significant quantities by the facility that operated at the site. This chemical, therefore, is added to the list of chemicals of potential concern.

and equations will need to be developed for the others as well.

In general, residential areas should be assumed to remain residential. Sites that are surrounded by operating industrial facilities can be assumed to remain industrial areas unless there is an indication that this is not appropriate. Lacking site-specific information (e.g., at scoping), it may be appropriate to assume residential land use. This assumption will generally lead to conservative (i.e., lower concentration) risk-based PRGs. If not enough site-specific information is readily available at scoping to select one future land use over another, it may be appropriate to develop a separate set of risk-based PRGs for each possible land use.

When waste will be managed onsite, land-use assumptions and risk-based PRG development become more complicated because the assumptions for the site itself may be different from the land use in the surrounding area. For example, if waste is managed onsite in a residential area, the risk-based PRGs for the ground water beneath the site (or at the edge of the waste management unit) may be based on residential exposures, but the risk-based PRGs for the site soils may be based on an industrial land use with some management or institutional controls.

If a land-use assumption is used that is less conservative (i.e., leads to higher risk-based concentrations) than another, it generally will be necessary to monitor the future uses of that site. For example, if residential land use is not deemed to be appropriate for a particular site because local zoning laws prohibit residential development, any changes in local zoning would need to be monitored. Such considerations should be clearly documented in the site's ROD.

### CASE STUDY: IDENTIFY FUTURE LAND USE

Based on established land-use trends, local renovation projects, and population growth projections in the area of the XYZ Co. site, the most reasonable future use of the land is determined to be residential use. Thus, sitespecific information is sufficient to show that the generally more conservative assumption of residential land use should serve as the basis for development of risk-based PRGs.

### 2.4 APPLICABLE OR RELEVANT AND APPROPRIATE REQUIREMENTS

Chemical-specific ARARs are evaluated as PRGs because they are often readily available and provide a preliminary indication about the goals that a remedial action may have to attain. This step involves identifying all readily available chemical-specific potential ARARs for the chemicals of potential concern (for each medium and probable land use). Because at scoping it often is uncertain which potential ARAR is the most likely one to become the ARAR-based PRG, all potential ARARs should be included in a tabular summary (i.e., no potential ARAR should be discarded). If there is doubt about whether a value is a potential ARAR, and therefore whether it could be used as a PRG, it should be included at this stage.

This section summarizes the concept of ARARs and identifies the major types of ARARs, but provides only <u>limited</u> guidance on identifying the most appropriate (likely) ARAR of all possible ARARs to use as the chemical-specific PRG. More detailed information about the identification and evaluation of ARARs is available from two important sources:

 the NCP (see specifically 55 Federal Register 8741-8766 for a description of ARARs, and 8712-8715 for using ARARs as PRGs; see also 53 Federal Register 51394); and

 CERCLA Compliance Manuals (EPA 1988a and 1989a).

### 2.4.1 CHEMICAL, LOCATION, AND ACTION-SPECIFIC ARARs

The Agency has identified three general types of federal and state ARARs:

- <u>chemical-specific</u>, are usually health- or risk management-based numbers or methodologics that, when applied to site-specific conditions, result in the establishment of numerical values (e.g., chemical-specific concentrations in a given medium);
- <u>location-specific</u>, are restrictions placed upon the concentration of hazardous substances or the conduct of activities solely because they are in special locations (e.g., wetlands); and
- <u>action-specific</u>, are usually technology- or activity-based requirements or limitations on actions taken with respect to hazardous wastes.

This guidance primarily addresses only <u>chemical-specific</u> ARARs since it focuses on the identification of chemical-specific concentrations that represent target goals (e.g., PRGs) for a given medium.

### 2.4.2 SELECTION OF THE MOST LIKELY ARAR-BASED PRG FOR EACH CHEMICAL

This section briefly describes which, if any, of several potential ARAR values for a given chemical is generally selected as the most likely ARAR-based PRG (and therefore the most likely PRG at this point). Although the process for identifying the most likely ARAR-based PRG is specific to the medium, in general the process depends on two considerations: (1) the applicability of the ARAR to the site; and (2) the comparative stringency of the standards being evaluated. The previously cited documents should be carefully considered for specific recommendations on identifying ARARs.

Ground Water. SDWA maximum contaminant levels (MCLs), non-zero MCLGs, state drinking water standards, and federal water quality criteria (FWQC) are common ARARs (and, therefore, potential PRGs) for ground water. Other types of laws, such as state anti-degradation laws, may be PRGs if they are accompanied by allowable concentrations of a chemical. (Although state anti-degradation laws that are expressed as qualitative standards may also be potential ARARs, they generally would not be considered PRGs.)

As detailed in the NCP (see next box), the first step in identifying ground-water PRGs is to determine whether the ground water is a current or potential source of drinking water. If the aquifer is a potential source of drinking water, then potential ARARs generally will include the federal non-zero MCLG, MCL, or state drinking water standard, and the most stringent (i.e., the lowest concentration) is identified as the most likely ARAR-based PRG.

> NCP ON GROUND-WATER GOALS (NCP Preamble; 55 Federal Register 8717, March 8, 1990)

"Ground water that is not currently a drinking water source but is potentially a drinking water source in the future would be protected to levels appropriate to its use as a drinking water source. Ground water that is not an actual or potential source of drinking water may not require remediation to a 10⁻⁴ to 10⁻⁴ level (except when necessary to address environmental concerns or allow for other beneficial uses; . . .)."

If the aquifer is not a potential source of drinking water, then MCLs, MCLGs, state drinking water requirements, or other health-based levels generally are <u>not</u> appropriate as PRGs. Instead, environmental considerations (i.e., effects on biological receptors) and prevention of plume expansion generally determine clean-up levels. If an aquifer that is not a potential source of drinking water is connected to an aquifer that is a drinking water source, it may be appropriate to use PRGs to set clean-up goals for the point of interconnection.

For chemicals without MCLs, state standards, or non-zero MCLGs, the FWQC may be potentially relevant and appropriate for ground water when that ground water discharges to surface water that is used for fishing or shellfishing. Surface Water. FWQC and state water quality standards (WQS) are common ARARs for surface water. An important determination for identifying ARARs and other criteria as potential PRGs for surface water is the current designated and future expected use of the water body. Because surface water potentially could serve many uses (e.g., drinking and fishing), several ARARs may be identified as potential PRGs for a chemical, with each ARAR corresponding to an identified use. A state WQS is generally the most likely ARAR for surface water unless a federal standard is more stringent.

If surface water is a current or potential source of <u>drinking</u> water, MCLs, state drinking water standards, non-zero MCLGs, and FWQC are potential ARARs. The analysis to determine which of these drinking water standards is the most likely ARAR-based PRG is the same as that conducted for ground water. An FWQC based on ingestion of water and fish might be an ARAR for surface water used for drinking.

If the designated or future expected use of surface water is <u>fishing or shellfishing</u>, and the state has not promulgated a WQS, an FWQC should be considered as a potential ARAR. The particular FWQC (i.e., for water and fish ingestion or fish ingestion alone) selected as the potential ARAR depends on whether exposure from one or both of the routes is likely to occur and, therefore, on the designated use of the water body. If other uses of the water are designated (e.g., swimming), a state WQS may be available.

Soil. In general, chemical-specific ARARs may not be available for soil. Certain states, however, have promulgated or are about to promulgate soil standards that may be ARARs and thus may be appropriate to use as PRGs. In addition, several EPA policies may be appropriate to use in developing PRGs (e.g., see EPA 1990c for guidance on PCB clean-up levels).

### 2.5 EXPOSURE PATHWAYS, PARAMETERS, AND EQUATIONS

This step is generally conducted for each medium and land-use combination and involves identifying the most appropriate (1) exposure pathways and routes (e.g., residential ingestion of drinking water), (2) exposure parameters (e.g., 2 liters/day of water ingested), and (3) equations (e.g., to incorporate intake). The equations include calculations of total intake from a given medium and are based on the identified exposure pathways and associated parameters. Information gathered in this step should be used to calculate risk-based PRGs using the default equations identified in Chapters 3 and 4. Site-specific equations can be derived if a different set of exposure pathways is identified for a particular medium; this option also is discussed in Chapters 3 and 4.

When risk-based concentrations are developed during scoping, <u>readily available</u> site-specific information may be adequate to identify and develop the exposure pathways, parameters, and equations (e.g., readily available information may indicate that the exposure duration should be 40 years instead of the standard default of 30 years). In the absence of readily available site-specific information, the standard default information in Chapters 3 and 4 generally should be used for the development of risk-based PRGs.

Exhibit 2-1 lists a number of the potential exposure pathways that might be present at a CERCLA site. The exposure pathways included in the medium-specific standard default equations (see Chapters 3 and 4) are italicized in this exhibit. Note that Chapters 3 and 4 may not address all of the exposure pathways of possible importance at a given CERCLA site. For example, the consumption of ground water that continues to be contaminated by soil leachate is not addressed. Guidance on goal-setting to address this exposure pathway is currently under development by EPA. In addition, the standard default equations do not address pathways such as plant and animal uptake of contaminants from soil with subsequent human ingestion. Under certain circumstances, these or other exposure pathways may present significant risks to human health. The standard default information, however, does address the quantifiable exposure pathways that are often significant contributors of risk for a particular medium and land use.

Chapters 3 and 4 show how exposures from several pathways are addressed in a single equation for a medium. For example, in the equation for ground water and surface water under the residential land-use assumption, the coefficients incorporate default parameter values for ingestion of drinking water and inhalation of volatiles during

-11-

### EXHIBIT 2-1

### TYPICAL EXPOSURE PATHWAYS BY MEDIUM FOR RESIDENTIAL AND COMMERCIAL/INDUSTRIAL LAND USES^{a,b}

	Exposure Pathways, Assuming:						
Medium	Residential Land Use	Commercial/Industrial Land Use					
Ground Water	Ingestion from drinking	Ingestion from drinking ^d					
	Inhalation of volatiles	Inhalation of volatiles					
	Dermal absorption from bathing	Dermal absorption					
	Immersion - external ^c						
Surface Water	Ingestion from drinking	Ingestion from drinking ^d					
	Inhalation of volatiles	Inhalation of volatiles					
	Dermal absorption from bathing	Dermal absorption					
	Ingestion during swimming						
	Ingestion of contaminated fish						
	Immersion - externale						
Soft	Ingestion	Ingestion					
200 200	Inhalation of particulates	Inhalation of particulates					
	Inhalation of volatiles	Inhalation of volatiles					
	Direct external exposure ^c	Direct external exposure ^e					
	Exposure to ground water contaminated by soil leachate	Exposure to ground water contaminated by soil leachate					
	Ingestion via plant uptake	Inhalation of particulates from trucks and heavy equipment					
and the state of the	Dermal absorption from gardening						

⁸ Lists of land uses, media, and exposure pathways are not comprehensive.

^b Exposure pathways included in RAGS/HHEM Part B standard default equations (Chapters 3 and 4) are italicized.

^c Applies to radionuclides only.

^d Because the NCP encourages protection of ground water to maximize its beneficial use, risk-based PRGs generally should be based on residential exposures once ground water is determined to be suitable for drinking. Similarly, when surface water will be used for drinking, general standards (e.g., ARARs) are to be achieved that define levels protective for the population at large, not simply worker populations. Residential exposure scenarios should guide risk-based PRG development for ingestion and other uses of potable water.

household water use. Full details of parameters used to develop each equation and a summary of the "reduced" standard default equations are provided in the text of these chapters.

Certain modifications of the default equations may be desirable or necessary. For example, if an exposure pathway addressed by an equation in Chapter 3 seems inappropriate for the site (e.g., because the water contains no volatiles and, therefore, inhalation of volatiles is irrelevant), or if information needed for a pathway (e.g., a chemical-specific inhalation slope factor [see Section 2.6]) is not readily available or derivable, then that pathway can be disregarded at this stage.

The decision about whether the risk assessor should collect site-specific human exposure pathway information (e.g., exposure frequency, duration, or intake rate data) is very important. There will frequently be methods available to gather such information, some of which are more expensive and elaborate than others. Determining whether the resulting data are reasonably representative of populations in the surrounding area, however, is often difficult. Collecting data by surveying those individuals most convenient or accessible to RPMs or risk assessors may not present a complete population exposure picture. In fact, poorly planned data gathering efforts may complicate the assessment process. For example, those surveyed may come to believe that their contributions will play a more meaningful role in the risk assessment than that planned by the risk assessors; this can result in significant demands on the risk assessor's time.

Before such data collection has begun, the risk assessor should determine, with the aid of screening analyses, what benefits are likely to result. Collection of the exposure data discussed in this section generally should not be attempted unless significant differences are likely to result in final reasonable maximum exposure (RME) risk estimates. If data collection is warranted, systematic and well-considered efforts that minimize biases in results should be undertaken. Estimates of future exposures are likely to rely heavily on conservative exposure assumptions. By definition, these assumptions will be unaffected by even the most extensive efforts to characterize current population activity.

At this stage, the risk assessor, site engineer, and RPM should discuss information concerning the absence or presence of important exposure pathways, because remediation goals should be designed for specific areas of the site that a particular remedy must address, and exposures expected for one area of the site may differ significantly from those expected in another area.

#### 2.5.1 GROUND WATER/SURFACE WATER

The residential land-use default equations presented in Chapters 3 and 4 for ground water or surface water are based on ingestion of drinking water and inhalation of volatile (vapor phase) chemicals originating from the household water supply (e.g., during dish washing, clothes laundering, and showering).

Ingestion of drinking water is an appropriate pathway for all chemicals with an oral cancer slope factor or an oral chronic reference dose. For the purposes of this guidance, however, inhalation of volatile chemicals from water is considered routinely only for chemicals with a Henry's Law constant of  $1 \times 10^{-5}$  atm-m³/mole or greater and with a molecular weight of less than 200 g/mole. Before determining inhalation toxicity values for a specific chemical (Section 2.6), it should be confirmed that the Henry's Law constant and molecular weight are in the appropriate range for inclusion in the inhalation pathway for water.

Default equations addressing industrial use of ground water are not presented. Because the NCP encourages protection of ground water to its maximum beneficial use, once ground water is determined to be suitable for drinking, risk-based PRGs generally should be based on residential exposures. Even if a site is located in an industrial area, the ground water underlying a site in an industrial area may be used as a drinking water source for residents several miles away due to complex geological interconnections.

#### 2.5.2 SOIL

The residential land-use standard default equations for the soil pathway are based on exposure pathways of ingestion of chemicals in soil or dust. The industrial land-use equations are based on three exposure pathways: ingestion of soil and dust, inhalation of particulates, and inhalation of volatiles. Again, for the purposes of this guidance, inhalation of volatile chemicals is relevant only for chemicals with a Henry's Law constant of 1 x 10⁻⁵ atm-m³/mole or greater and with a molecular weight of less than 200 g/mole. For the inhalation pathways, in addition to toxicity information, several chemical- and site-specific values are needed. These values include molecular diffusivity, Henry's Law constant, organic carbon partition coefficient, and soil moisture content (see Chapter 3 for details).

#### CASE STUDY: IDENTIFY EXPOSURE PATHWAYS, PARAMETERS, AND EQUATIONS

For the potential residential land use identified at the XYZ Co. site, the contaminated ground water (one of several media of potential concern) appears to be an important source of future domestic water. Because site-specific information is not initially available to develop specific exposure pathways, parameters, and equations, the standard default assumptions and equations provided in Chapter 3 will be used to calculate risk-based PRGs. Exposure pathways of concern for ground water, therefore, are assumed to be ingestion of ground water as drinking water and inhalation of volatiles in ground water during household use.

### 2.6 TOXICITY INFORMATION

This step involves identifying readily available toxicity values for all of the chemicals of potential concern for given exposure pathways so that the appropriate slope factors (SFs; for carcinogenic effects) and reference doses (RfDs; for noncarcinogenic effects) are identified or derived for use in the site-specific equations or the standard default equations. Therefore, <u>Chapter 7</u> of <u>RAGS/HHEM</u> Part A should be reviewed carefully before proceeding with this step.

The hierarchy for obtaining toxicity values for risk-based PRGs is essentially the same as that used in the baseline risk assessment. Briefly, Integrated Risk Information System (IRIS) is the primary source for toxicity information; if no verified toxicity value is available through IRIS, then Health Effects Assessment Summary Tables (HEAST) is the next preferred source. When the development of a toxicity value is required (and appropriate data are available), consultation with the Superfund Health Risk Assessment Technical Support Center is warranted. EPA staff can contact the Center by calling FTS-684-7300 (513-569-7300) or by FAX at FTS-684-7159 (513-569-7159). Others must fax to the above number or write to:

Superfund Health Risk Technical Support Center Environmental Criteria and Assessment Office U.S. Environmental Protection Agency Mail Stop 114 26 West Martin Luther King Drive Cincinnati, Ohio 45268

Other toxicity information that should be obtained includes EPA's weight-of-evidence classification for carcinogens (e.g., A, B1) and the source of the information (e.g., IRIS, HEAST).

Note that throughout this document, the term hazard index (HI) is used to refer to the risk level associated with noncarcinogenic effects. An HI is the sum of two or more hazard quotients (HQs). An HQ is the ratio of an exposure level of a single substance to the RfD for that substance. Because RfDs are generally exposure pathway-specific (e.g., inhalation RfD), the HQ is a single substance/ single exposure pathway ratio. An HI, on the other hand, is usually either a single substance/ multiple exposure pathway ratio, a multiple substance/single exposure pathway ratio, or a multiple substance/multiple exposure pathway In this document, however, only one ratio. exposure pathway is included in the default equation for some land-use and medium combinations (e.g., residential soil). In order to remain consistent, the term HI has been used throughout RAGS/HHEM Part B, even though for such a pathway, the term HQ could apply.

### 2.7 TARGET RISK LEVELS

This step involves identifying target risk concentrations for chemicals of potential concern. The standard default equations presented in Chapters 3 and 4 are based on the following target risk levels for carcinogenic and noncarcinogenic effects.

 For carcinogenic effects, a concentration is calculated that corresponds to a 10⁻⁶ incremental risk of an individual developing cancer over a lifetime as a result of exposure to the potential carcinogen from all significant exposure pathways for a given medium.

#### CASE STUDY: IDENTIFY TOXICITY INFORMATION*

Reference toxicity values for cancer and noncancer effects (i.e., SFs and RfDs, respectively) are required for chemicals without ARAR-based PRGs (only the case study chemicals without ARARs are listed here). Considering the ground-water medium only, ingestion and inhalation are exposure pathways of concern. Toxicity information is obtained from IRIS and HEAST, and is shown in the table below.

Chemical	RfD (mg/kg-day)	Source	SF (mg/kg-day) ⁻¹	Weight of Evidence	Source
EXPOSURE ROUTE:	INGESTION		0	0	
Hexane	0.06	HEAST	-	_	$\simeq$
Isophorone	0.2	IRIS	0.0039	С	HEAST
Triallate	0.013	IRIS		-	-
EXPOSURE ROUTE:	INHALATIO	N			
Hexane	0.04	HEAST	121	-	-
Isophorone	-	-	-	с	HEAST
Triallate	-			-	

 For noncarcinogenic effects, a concentration is calculated that corresponds to an HI of 1, which is the level of exposure to a chemical from all significant exposure pathways in a given medium below which it is unlikely for even sensitive populations to experience adverse health effects.

At scoping, it generally is appropriate to use the standard default target risk levels described above and discussed in the NCP. That is, an appropriate point of departure for remediation of carcinogenic risk is a concentration that corresponds to a risk of 10⁻⁶ for one chemical in a particular medium. For noncarcinogenic effects, the NCP does not specify a range, but it generally is appropriate to assume an HI equal to 1.

### 2.8 MODIFICATION OF PRELIMINARY REMEDIATION GOALS

Upon completion of the baseline risk assessment (or as soon as data are available), it is important to review the future land use, exposure assumptions, and the media and chemicals of potential concern originally identified at scoping, and determine whether PRGs need to be modified. Modification may involve adding or subtracting chemicals of concern, media, and pathways or revising individual chemical-specific goals.

#### 2.8.1 REVIEW OF ASSUMPTIONS

Media of Concern. As a guide to determining the media and chemicals of potential concern, the OSWER directive Role of the Baseline Risk Assessment in Superfund Remedy Selection Decisions (EPA 1991c) indicates that action is generally warranted at a site when the cumulative carcinogenic risk is greater than 10⁻⁴ or the cumulative noncarcinogenic HI exceeds 1 based on RME assumptions. Thus, where the baseline risk assessment indicates that either the cumulative current or future risk associated with a medium is greater than 10⁻⁴ or that the HI is greater than 1, that medium presents a concern, and it generally is appropriate to maintain risk-based PRGs for contaminants in that medium or develop risk-based PRGs for additional media where PRGs are not clearly defined by ARARs.

When the cumulative current or future baseline cancer risk for a medium is within the range of 10⁻⁶ to 10⁻⁴, a decision about whether or not to take action is a site-specific determination. Generally, risk-based PRGs are not needed for any chemicals in a medium with a cumulative cancer risk of less than 10⁻⁶, where an HI is less than or

-15-

equal to 1, or where the PRGs are clearly defined by ARARs. However, there may be cases where a medium appears to meet the protectiveness criterion but contributes to the contamination of another medium (e.g., soil contributing to groundwater contamination). In these cases, it may be appropriate to modify existing or develop new riskbased PRGs for chemicals of concern in the first medium, assuming that fate and transport models can adequately predict the impacts of concern on other media. EPA is presently developing guidance on quantifying the impact of soil contamination on underlying aquifers.

Chemicals of Concern. As with the initial media of potential concern, the initial list of specific chemicals of potential concern in a given medium may need to be modified to reflect increased information from the RI/FS concerning the importance of the chemicals to the overall site risk. Chemicals detected during the RI/FS that were not anticipated during scoping should be considered for addition to the list of chemicals of potential concern; chemicals anticipated during scoping that were not detected during the RI/FS should be deleted from the list. Ultimately, the identity and number of contaminants that may require risk-based PRGs depends both on the results of the baseline risk assessment and the extent of action required, given site-specific circumstances.

Following the baseline risk assessment, any chemical that has an associated cancer risk (current or future) within a medium of greater than 10⁻⁶ or an HI of greater than 1 should remain on the list of chemicals of potential concern for that medium. Likewise, chemicals that present cancer risks of less than 10⁻⁶ generally should <u>not</u> be retained on the list unless there are significant concerns about multiple contaminants and pathways.

Land Use. After the RI/FS, one future land use can usually be selected based on the results of the baseline risk assessment and discussions with the RPM. In many cases, this land use will be the same as the land use identified at scoping. In other cases, however, additional information from the baseline risk assessment that was not available at scoping may suggest modifying the initial landuse and exposure assumptions. A qualitative assessment should be made — and should be available from the baseline risk assessment — of the likelihood that the assumed future land use will occur.

Exposure Pathways, Parameters, and Equations. For exposure pathways, this process of modifying PRGs consists of adding or deleting exposure pathways from the medium-specific equations in Chapters 3 and 4 to ensure that the equation accounts for all significant exposure pathways associated with that medium at the site. For example, the baseline risk assessment may indicate that dermal exposure to contaminants in soil is a significant contributor to site risk. In this case, the risk-based PRGs may be modified by adding equations for dermal exposure. EPA policy on assessing this pathway is currently under development; the risk assessor should consult the Superfund Health Risk Technical Support Center (FTS-684-7300 or 513-569-7300) to determine the current status of guidance. Likewise, when appropriate data (e.g., on exposure frequency and duration) have been collected during the RI/FS, site-specific values can be substituted for the default values in the medium-specific equations.

#### 2.8.2 IDENTIFICATION OF UNCERTAINTIES

The uncertainty assessment for PRGs can serve as an important basis for recommending further modifications to the PRGs prior to setting final remediation goals. It also can be used during the post-remedy assessment (see Section 2.8.4) to identify areas needing particular attention.

Risk-based PRGs are associated with varied levels of uncertainty, depending on many factors (e.g., confidence that anticipated future land use is correct). To place risk-based PRGs that have been developed for a site in proper perspective, an assessment of the uncertainties associated with the concentrations should be conducted. This assessment is similar to the uncertainty assessment conducted during the baseline risk assessment (see RAGS/HHEM Part A, especially Chapters 6, 7, In fact, much of the uncertainty and 8). assessment conducted for a site's baseline risk assessment will be directly applicable to the uncertainty assessment of the risk-based PRGs.

In general, each component of risk-based PRGs discussed in this chapter — from media of potential concern to target risk level — should be examined, and the major areas of uncertainty highlighted. For example, the uncertainty

-16-

.

associated with the selected future land use should be discussed. Furthermore, the accuracy of the technical models used (e.g., for volatilization of contaminants from soil) to reflect site-specific conditions (present and future) should be discussed. If site-specific exposure assumptions have been made, it is particularly important to document the data supporting those assumptions and to assess their relevance for potentially exposed populations.

As the chemical- and medium-specific PRGs are developed, many assumptions regarding the RME individual(s) are incorporated. Although PRGs are believed to be fully protective for the RME individual(s), the proximity of other nearby sources of exposure (e.g., other CERCLA sites, RCRA facilities, naturally occurring background contamination) and/or the existence of the same contaminants in multiple media or of multiple chemicals affecting the same population(s), may lead to a situation where, even after attainment of all PRGs, protectiveness is not clearly achieved (e.g., cumulative risks may fall outside the risk The more likely it is that multiple range). contaminants, pathways, operable units, or other sources of toxicants will affect the RME individual(s), the more likely it will be that protectiveness is not achieved. This likelihood should be addressed when identifying uncertainties.

### 2.8.3 OTHER CONSIDERATIONS IN MODIFYING PRGs

The NCP preamble and rule state that factors related to exposure, technical limitations, and uncertainty should be considered when modifying PRGs (see next two boxes) and setting final remediation levels.

While the final remedial action objectives must satisfy the original "threshold criteria" of protection of human health and the environment and compliance with ARARs, the factors in the "balancing and modifying criteria" (listed in Section 1.3.2) also are considered in the detailed analysis for choosing among remedial alternatives. In cases where the alternative that represents the best balance of factors is not able to attain cancer risks within the risk range or an HI of 1, institutional controls may be used to supplement treatment and/or containment-based remedial action to ensure protection of human health and the environment.

#### NCP PREAMBLE: EXPOSURE, TECHNICAL, AND UNCERTAINTY FACTORS (55 Federal Register 8717, March 8, 1990)

"Preliminary remediation goals ... may be revised ... based on the consideration of appropriate factors including, but not limited to: exposure factors, uncertainty factors, and technical factors. Included under exposure factors are: cumulative effect of multiple contaminants, the potential for human exposure from other pathways at the site, population sensitivities, potential impacts on environmental receptors, and crossmedia impacts of alternatives. Factors related to uncertainty may include: the reliability of alternatives, the weight of scientific evidence concerning exposures and individual and cumulative health effects, and the reliability of exposure data. Technical factors may include: detection/quantification limits for contaminants, technical limitations to remediation, the ability to monitor and control movement of contaminants, and background levels of contaminants. The final selection of the appropriate risk level is made when the remedy is selected based on the balancing of criteria...."

#### NCP RULE: EXPOSURE, TECHNICAL, AND UNCERTAINTY FACTORS (40 CFR 300.430(c)(2)(i))

"(i)...Remediation goals...shall be developed by considering the following:

"(A) Applicable or relevant and appropriate requirements...and the following factors:

"(1) For systemic toxicants, acceptable exposure levels...;

*(2) For known or suspected carcinogens, acceptable exposure levels...;

"(3) Factors related to technical limitations such as detection/quantification limits for contaminants;

*(4) Factors related to uncertainty; and

"(5) Other pertinent information."

Note that in the absence of ARARs, the 10⁻⁶ cancer risk "point of departure" is used as a starting point for analysis of remedial alternatives, which reflects EPA's preference for managing risks at the more protective end of the risk range, other things being equal. Use of "point of departure" target risks in this guidance does not reflect a presumption that the final remedial action should attain such goals. (See NCP preamble, 55 Federal Register 8718-9.)

### 2.8.4 POST-REMEDY ASSESSMENT

To ensure that protective conditions exist after the remedy achieves all <u>individual</u> remediation levels set out in the ROD, there generally will be a site-wide evaluation conducted following completion of a site's final operable unit (e.g., during the five-year review). This site-wide evaluation should adequately characterize the residual contaminant levels and ensure that the post-remedy cumulative site risk is protective. More detailed guidance on the post-remedy assessment of site "protectiveness" is currently under development by EPA.

ē

## **CHAPTER 3**

## CALCULATION OF RISK-BASED PRELIMINARY REMEDIATION GOALS

This chapter presents standardized exposure parameters, the derivation of risk equations, and the corresponding "reduced" equations, for calculating risk-based PRGs at scoping for the media and land-use assumptions discussed in Chapter 2 (i.e., ground water, surface water, and soil for residential land use, and soil for commercial/industrial land use). Both carcinogenic and noncarcinogenic effects are addressed. Standardized default exposure parameters consistent with OSWER Directive 9285.6-03 (EPA 1991b) are used in this chapter; where default parameters are not available in that guidance, the references used are cited. If other media requiring risk-based PRGs are identified during the RI/FS, or other exposure parameters or land uses are assumed, then appropriate equations will need to be modified or new ones developed.

Risk-based equations have been derived in order to reflect the potential risk from exposure to a chemical, given a specific pathway, medium, and land-use combination. By setting the total risk for carcinogenic effects at a target risk level of 10-6 (the NCP's point of departure for analysis of remedial alternatives), it is possible to solve for the concentration term (i.e., the risk-based PRG). The total risk for noncarcinogenic effects is set at an HI of 1 for each chemical in a particular medium. Full equations with pathway-specific default exposure factors are presented in boxes with uniformly thin borders. Reduced equations are presented in the standard boxes (i.e., thicker top and bottom borders). At the end of this chapter, the case study that began in Chapter 2 is concluded (by showing how to calculate and present risk-based PRGs).

In general, the equations described in this chapter are sufficient for calculating the risk-based PRGs at the scoping stage of the RI/FS. Note, however, that these equations are based on standard default assumptions that may or may not reflect site-specific conditions. When risk-based PRGs are to be calculated based on site-specific conditions, the risk assessor should modify the full equations, and/or develop additional ones. Risk equations for individual exposure pathways for a given medium are presented in Appendix B of this document, and may be used to develop and/or modify the full equations. (See the introduction to Appendix B for more detailed instructions.)

Before examining the calculation of risk-based PRGs, several important points should be noted:

- Use of toxicity values in the equations as written currently assumes 100 percent absorption effeciency. That is, for the sake of simplicity at scoping, it is assumed that the dose administered to test animals in toxicity studies on which toxicity values are based was fully absorbed. This assumption may need to be revised in cases where toxicity values based on route-to-route extrapolation are used, or there are significant differences in absorption likely between contaminants in site media and the contaminants in the vehicle used in the toxicity study. Chapter 7 and Appendix A in RAGS/HHEM Part A (EPA 1989d) provide additional details on this point.
- The risk-based PRGs should contain at most two significant figures even though some of the parameters used in the reduced equations carry additional significant figures.
- The equations presented in this chapter calculate risk-based concentrations using inhalation reference doses (RfD_is) and inhalation slope factors (SF_is). If only the reference concentration (RfC) and/or inhalation unit risk are available for a particular compound in IRIS, conversion to an RfD_i and/or SF_i will be necessary. Many converted toxicity values are available in HEAST.
- <u>All</u> standard equations presented here incorporate pathway-specific default exposure

factors that generally reflect RME conditions. As detailed in Chapter 8 of RAGS/HHEM Part A (in the discussion on combining pathway risks [Section 8.3]), RME risks from one pathway should be combined with RME risks from another pathway only where there is good reason. Typically, RME from one pathway is not likely to occur with RME from another (unless there is a strong logical dependent relationship between exposures from the two pathways). If risk-based concentrations are developed for both the water and the soil pathways, the risk assessor ultimately may need to adjust exposure assumptions from one pathway (i.e., the one with the lower RME) to less conservative (more typical) values.

### 3.1 RESIDENTIAL LAND USE

#### 3.1.1 GROUND WATER OR SURFACE WATER

Under residential land use, risk from surface water or ground-water contaminants is assumed to be due primarily to direct ingestion and to inhalation of volatiles from household water use. Therefore, only these exposure pathways are considered in this section. Additional exposure pathways (e.g., dermal absorption) are possible and may be significant at some sites for some contaminants, while perhaps only one exposure pathway (e.g., direct ingestion of water only) may be relevant at others. In any case, the risk-based PRG for each chemical should be calculated by considering all of the relevant exposure pathways.

In the case illustrated here, risks from two exposure pathways from ground water or surface water are combined, and the risk-based concentration is derived to be protective for exposures from both pathways. Default risk from ground water or surface water would be calculated as follows ("total" risk, as used below, refers to the combined risk for a single chemical from all exposure pathways for a given medium):

Total risk from water	=	Risk from ingestion of water (adult)	+	Risk from inhala- tion of volatiles from household water (adult)
		water (adult)		from household water (adult)

At scoping, risk from indoor inhalation of volatiles is assumed to be relevant only for chemicals that easily volatilize. Thus, the risk

equation incorporates a water-air concentration relationship that is applicable only to chemicals with a Henry's Law constant of greater than 1 x 10⁻⁵ atm-m³/mole and a molecular weight of less than 200 g/mole. These criteria are not used to screen out chemicals that are not of potential concern for this exposure pathway but only to identify those that generally should be considered for the inhalation pathway when developing riskbased PRGs early in the process. Chemicals that do not meet these criteria may pose significant site risks (and require risk-based goals) through volatiles inhalation. The ultimate decision regarding which contaminants should be considered in the FS must be made on a sitespecific basis following completion of the baseline risk assessment.

Based primarily on experimental data on the volatilization of radon from household uses of water, Andelman (1990) derived an equation that defines the relationship between the concentration of a contaminant in household water and the average concentration of the volatilized contaminant in air. In the derivation, all uses of household water were considered (e.g., showering, laundering, dish washing). The equation uses a default "volatilization" constant (K) upper-bound value of 0.0005 x 1000 L/m3. (The 1000 L/m3 conversion factor is incorporated into the equation so that the resulting air concentration is expressed in mg/m³.) Certain assumptions were made in deriving the default constant K (Andelman 1990). For example, it is assumed that the volume of water used in a residence for a family of four is 720 L/day, the volume of the dwelling is 150,000 L and the air exchange rate is 0.25 m3/hr. Furthermore, it is assumed that the average transfer efficiency weighted by water use is 50 percent (i.e., half of the concentration of each chemical in water will be transfered into air by all water uses [the range extends from 30% for toilets to 90% for dishwashers]). See the Andelman paper for further details.

Concentrations Based on Carcinogenic Effects. Total risk for carcinogenic effects of certain volatile chemicals would be calculated by combining the appropriate inhalation and oral SFs with the two intakes from water:

Total = risk	SF _o	x	Intake from ingestion of	+	SF,	x	Intake from inhalation of
			water				volatiles from
							water

Adding appropriate parameters, and then rearranging the equation to solve for concentration, results in Equation (1).

Equation (1') on the next page is the reduced version of Equation (1) using the standard default parameters, and is used to calculate the risk-based PRG at a prespecified cancer risk level of 10⁻⁶. It combines the toxicity information of a chemical with standard default exposure parameters for residential land use to generate the concentration of that chemical that corresponds to a  $10^{-6}$  carcinogenic risk level due to that chemical. If either the SF₀ or SF₁ in Equation (1') is not available for a particular chemical, the term containing that variable in the equation can be ignored or equated to zero (e.g., for a chemical that does not have SF₁, the term 7.5(SF₁) in Equation (1') is ignored). If any of the default parameter values are changed to reflect site-specific conditions, the reduced equation cannot be used.

		RESIDENTIAL WATER - CARCINOGEN	IC EFFECTS
TR	-	SF. x C x IR. x EF x ED BW x AT x 365 days/yr+SF. x C x K x IR BW x AT x	<u>, x EF x ED</u> 365 days/yr
	-	$\frac{EF \times ED \times C \times [(SF_x \times IR_x) + (SF_x \times K \times IR_x)]}{BW \times AT \times 365 \text{ days/yr}}$	
C (mg based)	/L; risk-	$= \frac{\text{TR x BW x AT x 365 days/yr}}{\text{EF x ED x [(SF_i x K x IR_a) + (SF_o x IR_a)]}}$	.)]
where:	1		
Param	eters	Definition (units)	Default Value
С		chemical concentration in water (mg/L)	물건 방법 가격
TR		target excess individual lifetime cancer risk (unitless)	10-4
SF		inhalation cancer slope factor ((mg/kg-day)')	chemical-specific
SF.		oral cancer slope factor ((mg/kg-day) ⁻¹ )	chemical-specific
BW		adult body weight (kg)	70 kg
AT		averaging time (yr)	70 yr
EF cxposure frequency (days/yr)		exposure frequency (days/yr)	350 days/yr
ED		exposure duration (yr)	30 yr
IR,		daily indoor inhalation rate (m3/day)	15 m ³ /day
IR,		daily water ingestion rate (L/day)	2 L/day
K		volatilization factor (unitless)	0.0005 x 1000 L/m3 (Andeiman 1990)

	REDUCED EQUATION: RESIDENTIAL WATER - CARCINOGENIC EFFECTS	and some
Risk-based I (mg/L; TR	$PRG = \frac{1.7 \times 10^4}{2(SF_o) + 7.5(SF_i)}$	(1^)
where:		
SF.	= oral slope factor in (mg/kg-day) ⁻¹	
SFi	= inhalation slope factor in (mg/kg-day)"	

Concentrations Based on Noncarcinogenic Effects. Total HI would be calculated by combining the appropriate oral and inhalation RfDs with the two intakes from water:

- HI = Intake from oral ingestion RfD_o
  - Intake from inhalation RfD_i

Adding appropriate parameters, and then rearranging the equation to solve for concentration, results in Equation (2). Equation (2') on the next page is the reduced version of Equation (2) using the standard default parameters, and is used to calculate the risk-based PRG at a prespecified HI of 1. It combines the toxicity information of a chemical with standard exposure parameters for residential land use to generate the concentration of that chemical that corresponds to an HI of 1. If either the RfD_o or RfD_i in Equation (2') is not available for a particular chemical, the term containing that variable in the equation can be ignored or equated to zero (e.g., for a chemical that does not have RfD_j, the term 7.5/RfD_j in Equations (2') is ignored).

	RESIDENTIAL WATER - NON	CARCINOGENIC EFFECTS	
THI =	C x IR, x EF x ED + RfD, x BW x AT x 365 days/yr	C x K x IR, x EF x ED RfD _i x BW x AT x 365 daya/yr	
=	EF x ED x C x [(1/RfD, x IR_) + (1/R BW x AT x 365 days	f <u>D, x K x IR_a)}</u> yr	
C (mg/L; risk- based)	$= \frac{\text{THI x BW x AT x 36}}{\text{EF x ED x } [(1/\text{RfD}_i x K x IR_i)]}$	$\frac{5 \text{ days/yr}}{+ (1/\text{RfD}_0 \times 1\text{R}_*)]}$	(2)
where:			
Parameters	Definition	Default Value	
C THI	chemical concentration in water (mg/L) target hazard index (unitless)	1	
RfD, RfD,	oral chronic reference dose (mg/kg-day) inhalation chronic reference dose (mg/kg-da	chemical-specific ay) chemical-specific	
BW	adult body weight (kg) averaging time (yr)	70 kg 30 yr (for noncarcinogens, equal to ED)	
ED	exposure frequency (days/yr) exposure duration (yr)	350 days/yr 30 yr	
IR,	daily indoor inhalation rate (m'/day) daily water ingestion rate (L/day)	15 m²/day 2 L/day	

	REDUCE	EQUATION: RESIDENTIAL WAY	TER - NONCARCINOGENIC EFFECTS	
Risk-based (mg/L; TH	$\frac{PRG}{I = 1}$	= <u>73</u> [7.5/RfD _i + 2/RfD _a ]		(2')
where:			18	
RfD。	-	oral chronic reference dose in mg/kg	-day	
RfD,	-	inhalation chronic reference dose in	mg/kg-day	

### 3.1.2 SOIL

Under residential land use, risk of the contaminant from soil is assumed to be due to direct ingestion of soil only.

Total risk from soil = Risk from ingestion of soil (child to adult)

Because the soil ingestion rate is different for children and adults, the risk due to direct ingestion of soil is calculated using an age-adjusted ingestion factor. The age-adjusted soil ingestion factor (IFsoil/adj) takes into account the difference in daily soil ingestion rates, body weights, and exposure durations for two exposure groups - children of one to six years and others of seven to 31 years. Exposure frequency (EF) is assumed to be identical for the two exposure groups. For convenience, this factor is calculated separately as a time-weighted soil intake, normalized to body weight, that can then be substituted in the total intake equation. Calculated in this manner, the factor leads to a more protective risk-based concentration compared to an adult-only assumption. Note that the ingestion factor is in units of mg-yr/kg-day, and therefore is not directly comparable to daily soil intake rate in units of mg/kg-day. See the box containing Equation (3) for the calculation of this factor.

Additional exposure pathways (e.g., inhalation of particulates, inhalation of volatiles, ingestion of foodcrops contaminated through airborne particulate deposits, consumption of ground water contaminated by soil leachate) are possible at some sites. The risk assessor should evaluate whether inhalation or other exposure pathways are significant at the site. Generally, for many undisturbed sites with vegetative cover such as those found in areas of residential land use, air pathways are relatively minor contributors of risk. Greater concern for baseline risk via air pathways exists under commercial/industrial land-use assumptions, given the increased activity levels likely (see Section 3.2.2). Air pathway risks also tend to be major concerns during remedial action (see RAGS/HHEM Part C). If these other pathways are known to be significant at scoping, Appendix B and/or other information should be used to develop site-specific equations for the riskbased PRGs.

Concentrations Based on Carcinogenic Effects. Total risk for carcinogenic effects would be calculated by combining the appropriate oral SF with the intake from soil:

Total risk = SF_a x Intake from ingestion of soil

Adding appropriate parameters, and then rearranging the equation to solve for concentration, results in Equation (4).

Equation (4') below is the reduced version of Equation (4) using the standard default parameters, and is used to calculate the risk-based PRG at a prespecified cancer risk level of 10⁻⁶. It combines the toxicity information of a chemical with standard exposure parameters for residential land use to generate the concentration of that chemical that corresponds to a 10⁻⁶ carcinogenic risk level due to that chemical.

$Rg-day$ = $\frac{IR_{activage1-6} \times ED_{age1-6}}{BW_{age1-6}} + \frac{IR_{activage7-31} \times BW}{BW}$	ED	(3)
Definition	Default Value	
age-adjusted soil ingestion factor (mg-yr/kg-day) average body weight from ages 1-6 (kg)	114 mg-yr/kg-day 15 kg	
average body weight from ages 7-31 (kg) exposure duration during ages 1-6 (yr)	70 kg 6 yr	
exposure duration during ages 7-31 (yr) ingestion rate of soil age 1 to 6 (mg/day)	24 yr 200 mg/day	
	$(kg-osy) = \frac{1K_{assinget-6} \times ED_{aget-6}}{BW_{age14}} + \frac{1K_{assinget-31} \times BW}{BW}$ <u>Definition</u> age-adjusted soil ingestion factor (mg-yr/kg-day) average body weight from ages 1-6 (kg) average body weight from ages 7-31 (kg) exposure duration during ages 1-6 (yr) exposure duration during ages 7-31 (yr) ingestion rate of soil age 1 to 6 (mg/day) ingestion rate of soil age 1 to 6 (mg/day)	$\frac{BW_{age14}}{BW_{age14}} + \frac{BW_{age131}}{BW_{age131}} + \frac{BW_{age132}}{BW_{age131}} + \frac{BW_{age132}}{BW_{age131}} + \frac{BW_{age132}}{BW_{age131}} + \frac{BW_{age132}}{BW_{age131}} + \frac{BW_{age132}}{BW_{age131}} + \frac{BW_{age132}}{BW_{age132}} + BW_{$

	RESIDENTIAL SOIL - CARCINOGENI	C EFFECTS	
TR =	SF, x C x 10 ⁻⁶ kg/mg x EF x IF _{setVadi} AT x 365 days/yr		
C (mg/kg; risk- based)	= <u>TR x AT x 365 days/year</u> SF _o x 10 ⁻⁶ kg/mg x EF x IF _{solandj}		(4)
where:			
Parameters	Definition (units)	Default Value	
с	chemical concentration in soil (mg/kg)		
TR	target excess individual lifetime cancer risk (unitless)	10*	
SF	oral cancer slope factor ((mg/kg-day)-1)	chemical-specific	
AT	averaging time (yr)	70 yr	
EF	exposure frequency (days/yr)	350 days/yr	
	age-adjusted interstion factor (maan/ka.dav)	114 mg-yr/kg-day (see Equation (3))	



Concentrations Based on Noncarcinogenic Effects. Total HI would be calculated by combining the appropriate oral RfD with the intake from soil:

Adding appropriate parameters, and then rearranging the equation to solve for concentration, results in Equation (5).

Equation (5') is the reduced version of Equation (5) using the standard default parameters, and is for calculating the risk-based PRG at a prespecified HI of 1. It combines the toxicity information of a chemical with standard exposure parameters for residential land use to generate the concentration of that chemical that corresponds to an HI of 1.

### 3.2 COMMERCIAL/INDUSTRIAL LAND USE

#### 3.2.1 WATER

Once ground water is determined to be suitable for drinking, risk-based concentrations should be based on residential exposures. This is because the NCP seeks to require protection of ground water to allow for its maximum beneficial use (see Section 2.3). Thus, under the commercial/ industrial land-use scenario, risk-based PRGs for ground water are calculated according to procedures detailed in Section 3.1.1. Similarly, for surface water that is to be used for drinking, the risk-based PRGs should be calculated for residential populations, and not simply worker populations.

	RESIDENTIAL SOIL - NONCARO	CINOGENIC EFFECTS
THI = Co	x 10 ⁻⁶ kg/mg x EF x IF _{solutio} RfD _o x AT x 365 days/yr	
C (mg/kg; risk- based)	= THI x AT x 365 days/vr 1/RfD _o x 10 ⁻⁶ kg/mg x EF x IF _{soilladj}	(5
where:		
Parameters	Definition (units)	Default Value
с	chemical concentration in soil (mg/kg)	
THI	target hazard index (unitless)	1
RfD	oral chronic reference dose (mg/kg-day)	chemical-specific
AT	averaging time (yr)	30 yr (for noncarcinogens, equal to ED [which is incorporated in IF _{aultat} ])
EF	exposure frequency (days/yr)	350 days/yr
IFacilisti	age-adjusted ingestion factor (mg-yr/kg-day)	114 mg-yr/kg-day (see Equation (3))

REDUCED EQUATION: RESIDENTIAL SOIL - NONCARCINOGENIC EFFECTS

Risk-based PRG (mg/kg; THI = 1)

where:

RfD_o = oral chronic reference dose in mg/kg-day

2.7 x 105 (RfD_)

### 3.2.2 SOIL

Under commercial/industrial land use, risk of the contaminant from soil is assumed to be due to direct ingestion, inhalation of volatiles from the soil, and inhalation of particulates from the soil, and is calculated for an adult worker only. For this type of land use, it is assumed for calculating default risk-based PRGs that there is greater potential for use of heavy equipment and related traffic in and around contaminated soils and thus greater potential for soils to be disturbed and produce particulate and volatile emissions than in most residential land-use areas. Additional exposure pathways (e.g., dermal exposure) are possible at some sites, while perhaps only one exposure pathway (e.g., direct ingestion of soil only) may be relevant at others; Appendix B may be used to identify relevant exposure pathways to be combined. In such cases, the risk is calculated by considering all the relevant exposure pathways identified in the RI.

In the default case illustrated below, intakes from the three exposure pathways are combined and the risk-based PRG is derived to be protective for exposures from all three pathways. In this case, the risk for a specific chemical from soil due to the three exposure pathways would be calculated as follows:

- Total risk = Risk from ingestion of soil (worker) from soil
  - Risk from inhalation of volatiles from soil (worker)

(5')

 Risk from inhalation of particulates from soil (worker)

It is possible to consider only exposure pathways of site-specific importance by deriving a site-specific risk-based PRG (e.g., using the equations in Appendix B).

-25-

Concentrations Based on Carcinogenic Effects. Total risk for carcinogenic effects would be calculated by combining the appropriate inhalation and oral SFs with the three intakes from soil:

- Total risk = SF_o x Intake from ingestion of soil (worker)
  - + SF_i x Intake from inhalation of volatiles from soil (worker)
  - + SF_i x Intake from inhalation of particulates (worker)

Adding appropriate parameters, and then rearranging the equation to solve for concentration, results in Equation (6). As discussed in more detail in Section 3.3.1, Equation (6a) is used to test the results of Equation (6).

Equation (6') is the reduced version of Equation (6) using the standard default parameters, and is used to calculate the risk-based PRG at a prespecified cancer risk level of 10⁻⁶. It combines the toxicity information of a chemical with standard exposure parameters for commercial/industrial land use to generate the concentration of that chemical that corresponds to a 10⁻⁶ carcinogenic risk level due to that chemical.

Concentrations Based on Noncarcinogenic Effects. Total HI would be calculated by combining the appropriate oral and inhalation RfDs with the three intakes from soil:

HI = Intake from ingestion RfD_o (Intake from inhalation of volatiles

> and particulates) RfD:

Adding appropriate parameters, and then rearranging the equation to solve for concentration, results in Equation (7).

Equation (7') is the reduced version of Equation (7) using the standard default parameters, and is used to calculate the risk-based PRG at a prespecified HI of 1. It combines the toxicity information of a chemical with standard exposure parameters for commercial/industrial land use to generate the concentration of that chemical that corresponds to an HI of 1.

### 3.3 VOLATILIZATION AND PARTICULATE EMISSION FACTORS

#### 3.3.1 SOIL-TO-AIR VOLATILIZATION FACTOR

The volatilization factor (VF) is used for defining the relationship between the concentration of contaminants in soil and the volatilized contaminants in air. This relationship was established as a part of the Hwang and Falco (1986) model developed by EPA's Exposure Assessment Group (EAG). Hwang and Falco present a method intended primarily to estimate the permissible residual levels associated with the cleanup of contaminated soils. This method has been used by EPA in estimating exposures to PCBs and 2,3,7,8-TCDD from contaminated soil (EPA 1986; EPA 1988a). One of the pathways considered in this method is the intake by inhalation of volatilized contaminants.

The basic principle of the Hwang and Faico model is applicable only if the soil contaminant concentration is at or below saturation. Saturation is the soil contaminant concentration at which the adsorptive limits of the soil particles and the solubility limits of the available soil moisture have been reached. Above saturation, pure liquid-phase contaminant is present in the soil. Under such conditions, the partial pressure of the pure contaminant and the partial pressure of air in the interstitial soil pore spaces cannot be calculated without first knowing the mole fraction of the contaminant in the soil. Therefore, above saturation, the PRG cannot be accurately calculated based on volatilization. Because of this limitation, the chemical concentration in soil (C) calculated using the VF must be compared with the soil saturation concentration (Csat) calculated using Equation (6a) or (7a). If C is greater than Ctate then the PRG is set equal to Csat-

The VF presented in this section assumes that the contaminant concentration in the soil is homogeneous from the soil surface to the depth of concern and that the contaminated material is not covered by contaminant-free soil material. For the purpose of calculating VF, depth of concern is defined as the depth at which a near impenetrable layer or the permanent ground-water level is reached.

	COMMERCIAL/INDUSTRIAL SOIL - CARCIN	OGENIC EFFECTS	
TR = SF	<u>x C x 10[*] kg/mg x EF x ED x IR_{sol} + <u>SF, x C x EF x</u> BW x AT x 365 days/yr BV</u>	ED x IR _{st} x (1/VF + 1/PEF) V x AT x 365 days/yr	
C (mg/kg; risk- based)	$= \frac{\text{TR x BW x AT x 365}}{\text{EF x ED x [(SF_o x 10^6 kg/mg x IR_{sol}) + (SI)]}}$	$\frac{\text{days/yr}}{F_i \times [R_{sir} \times [1/VF + 1/PEF])]} $ (	6)
where:			
Parameters	Definition (units)	Default Value	
C TR SF _i SF _o BW AT EF ED IR _{sol} IR _{sol}	chemical concentration in soil (mg/kg) target excess individual lifetime cancer risk (unitless) inhalation cancer slope factor ((mg/kg-day) ⁻¹ ) oral cancer slope factor ((mg/kg-day) ⁻¹ ) adult body weight (kg) averaging time (yr) exposure frequency (days/yr) exposure frequency (days/yr) soil ingestion rate (mg/day) workday inhalation rate (m ³ /day) soil-to-air volatilization factor (m ³ /kg)	10 ⁻⁶ chemical-specific chemical-specific 70 kg 70 yr 250 days/yr 25 yr 50 mg/day 20 m ³ /day chemical-specific (see Section 3.3.1)	
rtr C _{er} = (	$K_{x} x s x n_{n} + (s x \Theta_{n})$	4.05 X 10 III /kg (see Section 3.5.2)	ia)
where:			83) -
Parameters	Definition (units)	Default Value	
C _{est} K _d K _{oc} OC s n _m $\Theta_m$	soil saturation concentration (mg/kg) soil-water partition coefficient (L/kg) organic carbon partition coefficient (L/kg) organic carbon content of soil (fraction) solubility (mg/L-water) soil moisture content, expressed as a weight fraction soil moisture content, expressed as L-water/kg-soil		

**REDUCED EQUATION:** COMMERCIAL/INDUSTRIAL SOIL – CARCINOGENIC EFFECTS Risk-based PRG  $= \frac{2.9 \times 10^{-4}}{[((5 \times 10^{-5}) \times SF_{o}) + (SF_{i} \times ((20/VF) + (4.3 \times 10^{-5})))]}$  (6') where:  $SF_{o} = \text{oral slope factor in (mg/kg-day)}^{-1}$   $SF_{i} = \text{inhalation slope factor in (mg/kg-day)}^{-1}$   $VF = \text{chemical-specific soil-to-air volatilization factor in m}^{-3}/kg$  (see Section 3.3.1) If PRG > C_{sat}, then set PRG = C_{sat} (where C_{sat} = soil saturation concentration (mg/kg); see Equation (6a) and Section 3.3.1).

	COMMERCIAL/INDUSTRIAL SOIL - NONCAR	CINOGENIC EFFECTS
THI =	C x 10 ⁴ kg/mg x EF x ED x IR _{sol.} +     C x EF x       RfD _e x BW x AT x 365 days/yr     Rf	$\frac{\text{ED x IR}_{\text{st}} x (1/\text{VF} + 1/\text{PEF})}{\text{fD}_{i} x BW x AT x 365 days/yr}$
C (mg/kg; =	THI x BW x AT x 365 davs/yr	(7)
risk-based)	$ED \times EF \times [((1/RiD_0) \times 10^{\circ} \text{ kg/mg} \times IR_{soil}) + ((1/RiD_i)$	(1/VF + 1/PEF))
where:		
Parameters	Definition (units)	Default Value
с	chemical concentration in soil (mg/kg)	
THI	target hazard index (unitless)	1
RfD.	oral chronic reference dose (mg/kg-day)	chemical-specific
RfD,	inhalation chronic reference dose (mg/kg-day)	chemical-specific
BW	adult body weight (kg)	70 kg
AT	averaging time (yr)	25 yr (always equal to ED)
EF	exposure frequency (days/yr)	250 days/yr
ED	exposure duration (yr)	25 уг
IRani	soil ingestion rate (mg/day)	50 mg/day
IR	workday inhalation rate (m3/day)	20 m ³ /day
VF	soil-to-air volatilization factor (m3/kg)	chemical-specific (see Section 3.3.1)
PEF	particulate emission factor (m ³ /kg)	4.63 x 10 ⁹ m ³ /kg (see Section 3.3.2)
C,	$u_{est} = (K_d x s x n_m) + (s x \Theta_m)$	· . (7a)
where:		
	D. C. March 1	Production of the second second second second second second second second second second second second second se
Parameters	Demition (units)	Default Vasue
C	soil saturation concentration (mg/kg)	-
K.	soil-water partition coefficient (L/kg)	chemical-specific, or K v OC
K	organic carbon partition coefficient (L/kg)	chemical-specific
OC	organic carbon content of soil (fraction)	site-specific, or 0.02
5	solubility (mg/L-water)	chemical-specific
n _m	soil moisture content, expressed as a weight fraction	site-specific
0.	soil moisture content, expressed as L-water/kg-soil	site-specific

Risk-based PRG (mg/kg; THI = 1)	=	$\frac{102}{\left[(5 \times 10^{3}/\text{RfD}_{o}) + ((1/\text{RfD}_{i}) \times ((20/\text{VF}) + (4.3 \times 10^{9})))\right]}$	(7*)
where:			
RfD.		oral chronic reference dose in mg/kg-day	
RfD	-	inhalation chronic reference dose in mg/kg-day	
VF	=	ch. mical-specific soil-to-air volatilization factor in m3/kg (see Section 3.3.1)	
A chemical-specific value for VF is used in the standard default equations (Equations (6), (6'), (7), and (7') in Section 3.2.2) and is developed in Equation (8). The VF value calculated using Equation (8) has been developed for specific use in the other equations in this guidance; it may not be applicable in other technical contexts. Equation (8) lists the standard default parameters for calculating VF. If site-specific information is available, Equation (8) may be modified to calculate a VF that is more appropriate for the particular site. Supporting references should be consulted when substituting site-specific data to ensure that the model and specific parameters can be appropriately applied to the given site.

### 3.3.2 PARTICULATE EMISSION FACTOR

The particulate emission factor (PEF) relates the contaminant concentration in soil with the concentration of respirable particles  $(PM_{10})$  in the air due to fugitive dust emissions from surface contamination sites. This relationship is derived by Cowherd (1985) for a rapid assessment procedure applicable to a typical hazardous waste site where the surface contamination provides a relatively continuous and constant potential for emission over an extended period of time (e.g., years). The particulate emissions from contaminated sites are due to wind erosion and, therefore, depend on the erodibility of the surface



material. The equation presented below, Equation (9), is representative of a surface with "unlimited erosion potential," which is characterized by bare surfaces of finely divided material such as sandy agricultural soil with a large number ("unlimited reservoir") of erodible particles. Such surfaces erode at low wind speeds, and particulate emission rates are relatively time-independent at a given wind speed.

This model was selected for use in RAGS/HHEM Part B because it represents a conservative estimate for intake of particulates; it is used to derive Equations (6) and (7) in Section 3.2.2.

Using the default parameter values given in the box for Equation (9), the default PEF is equal to 4.63 x 10⁹ m³/kg. The default values necessary to calculate the flux rate for an "unlimited reservoir" surface (i.e., G, Um, U, and F(x)) are provided by Cowherd (1985), and the remaining default values (i.e., for LS, V, and DH) are "typical" values (EPA 1986). If site-specific information is available, Equation (9) may be modified to calculate a PEF that is more appropriate for the particular site. Again, the original reference should be consulted when substituting site-specific data to ensure applicability of the model to specific site conditions.

	PARTICULATE EMISS	ION FACTOR
PEF (m ³ /kg) where:	$= \frac{\text{LS x V x DH x 3600 s/hr}}{A}$	x $\frac{1000 \text{ g/kg}}{0.036 \text{ x (1-G) x (U_{ub}/U_1)^3 x F(x)}}$ (5)
Parameter	Definition (units)	Default
PEF	particulate emission factor (m3/kg)	4.63 x 10 ⁹ m ³ /kg
LS	width of contaminated area (m)	45 m
v	wind speed in mixing zone (m/s)	2.25 m/s
DH	diffusion height (m)	2 m
A	area of contamination (m2)	2025 m ²
0.036	respirable fraction (g/m2-hr)	0.036 g/m ² -hr
G	fraction of vegetative cover (unitless)	0
U_	mean annual wind speed (m/s)	4.5 m/s
U _t	equivalent threshold value of wind speed at 10 m (m/s)	12.8 m/s
F(x)	function dependent on U_/U, (unitless)	0.0497 (determined using Cowherd 1985)

### 3.4 CALCULATION AND PRESENTATION OF RISK-BASED PRGs

The equations presented in this chapter can be used to calculate risk-based PRGs for both carcinogenic and noncarcinogenic effects. If both a carcinogenic and a noncarcinogenic risk-based PRG are calculated for a particular chemical, then the lower of the two values is considered the appropriate risk-based PRG for any given contaminant. The case-study box below illustrates a calculation of a risk-based PRG. A summary table — such as that in the final case-study box should be developed to present both the risk-based PRGs and the ARAR-based PRGs. The table should be labeled as to whether it presents the concentrations that were developed during scoping or after the baseline risk assessment.

### CASE STUDY: CALCULATE RISK-BASED PRGs^a

Risk-based PRGs for ground water for isophorone, one of the chemicals detected in ground-water monitoring wells at the site, are calculated below. Initial risk-based PRGs for isophorone (carcinogenic and noncarcinogenic effects) are derived using Equations (1') and (2') in Section 3.1.1. Equations (1') and (2') combine the toxicity information of the chemical (oral RfD of 0.2 mg/kg-day and oral SF of 0.0039 [mg/kg-day]¹; inhalation values are not available and, therefore, only the oral exposure route is considered) with standard exposure parameters. The calculated concentrations in mg/L correspond to a target risk of  $10^{-6}$  and a target HQ of 1, as follows:

Carcinogenic risk-based PRG	=	$\frac{1.7 \times 10^{-4}}{2(SF_{o})}$	Noncarcinogenic risk-based PRG	=	- <u>73</u> 2/RfD _o
	=	<u>1.7 x 10⁻⁴</u> 2(0.0039)		=	73
	=	0.022 mg/L		=	7.3 mg/L

The lower of the two values (i.e., 0.022 mg/L) is selected as the appropriate risk-based PRG. Risk-based PRGs are calculated similarly for the other chemicals of concern.

^a All information in this example is for illustration purposes only.

#### CASE STUDY: PRESENT PRGs DEVELOPED DURING SCOPING^a

Site: XYZ Co. Location: Anytown, Anystate Medium: Ground Water Land Use: Residential Exposure Routes: Water Ingestion, Inhalation of Volatiles

Chemical	Risk-base (mg/	d PRGs L)*	ARA	R-based PRG
Chemical	10-6	HQ = 1	Туре	Concentration (mg/L)
Benzene	_	_	MCL	0.005
Carbon Tetrachloride	_	_	MCL	0.005
Ethylbenzene	_	_	MCLG	0.7***
			MCL	0.7
Hexane		0.33	-	_
Isophorone	0.022**	7.3	_	
Triallate	_	0.47		_
1,1,2-Trichloroethane			MCLG	0.003***
			MCL	0.005
Vinyl chloride	—	_	MCL	0.002

^a All information in this example is for illustration purposes only.

These concentrations were calculated using the standard default equations in Chapter 3.

•• Of the two potential <u>risk-based</u> PRGs for this chemical, this concentration is the selected risk-based PRG. ••• Of the two potential <u>ARAR-based</u> PRGs for this chemical, this concentration is selected as the ARAR-

based PRG.

# **CHAPTER 4**

# **RISK-BASED PRGs FOR RADIOACTIVE CONTAMINANTS**

This chapter presents standardized exposure parameters, derivations of risk equations, and "reduced" equations for calculating risk-based PRGs for radioactive contaminants for the pathways and land-use scenarios discussed in Chapter 2. In addition, a radiation site case study is provided at the end of the chapter to illustrate (1) how exposure pathways and radionuclides of potential concern (including fadioactive decay products) are identified, (2) how initial risk-based PRGs for radionuclides are calculated using reduced equations based on information available at the scoping phase, and (3) how risk-based PRGs can be re-calculated using full risk equations and site-specific data obtained during the baseline risk assessment. Chapters 1 through 3 and Appendices A and B provide the basis for many of the assumptions, equations, and parameters used in this chapter, and therefore should be reviewed before proceeding further into Chapter 4. Also, Chapter 10 in RAGS/HHEM Part A should be consulted for additional guidance on conducting baseline risk assessments at sites contaminated with radioactive substances.

In general, standardized default exposure equations and parameters used to calculate riskbased PRGs for radionuclides are similar in structure and function to those equations and parameters developed in Chapter 3 for nonradioactive chemical carcinogens. Both types of risk equations:

- Calculate risk-based PRGs for each carcinogen corresponding to a pre-specified target cancer risk level of 10⁻⁶. As mentioned in Section 2.8, target risk levels may be modified after the baseline risk assessment based on site-specific exposure conditions, technical limitations, or other uncertainties, as well as on the nine remedy selection criteria specified in the NCP.
- Use standardized default exposure parameters consistent with OSWER Directive 9285.6-03 (EPA 1991b). Where default parameters are

not available in that guidance document, other appropriate reference values are used and cited.

 Incorporate pathway-specific default exposure factors that generally reflect RME conditions.

There are, however, several important areas in which risk-based PRG equations and assumptions for radioactive contaminants differ substantially from those used for chemical contaminants. Specifically, unlike chemical equations, risk equations for radionuclides:

- Accept input quantities in units of activity (e.g., picocuries (pCi)) rather than in units of mass (e.g., milligrams (mg)). Activity units are more appropriate for radioactive substances because concentrations of radionuclides in sample media are determined by direct physical measurements of the activity of each nuclide present, and because adverse human health effects due to radionuclide intake or exposure are directly related to the amount, type, and energy of the radiation deposited in specific body tissues and organs.
- Consider the carcinogenic effects of radionuclides only. EPA designates all radionuclides as Class A carcinogens based on their property of emitting ionizing radiation and on the extensive weight of epidemiological evidence of radiation-induced cancer in humans. At most CERCLA radiation sites, potential health risks are usually based on the radiotoxicity, rather than the chemical toxicity, of each radionuclide present.
- Use cancer slope factors that are best estimates (i.e., median or 50th percentile values) of the age-averaged, lifetime excess total cancer risk per unit intake of a radionuclide (e.g., per pCi inhaled or ingested) or per unit external radiation exposure (e.g., per microRoentgen) to gamma-emitting

radionuclides. Slope factors given in IRIS and HEAST have been calculated for individual radionuclides based on their unique chemical, metabolic, and radiological properties and using a non-threshold, linear dose-response model. This model accounts for the amount of each radionuclide absorbed into the body from the gastrointestinal tract (by ingestion) or through the lungs (by inhalation), the distribution and retention of each radionuclide in body tissues and organs, as well as the age, sex, and weight of an individual at the time of exposure. The model then averages the risk over the lifetime of that exposed individual (i.e., 70 years). Consequently, radionuclide slope factors are not expressed as a function of body weight or time, and do not require corrections for gastrointestinal absorption or Jung transfer efficiencies.

Risk-based PRG equations for radionuclides presented in the following sections of this chapter are derived initially by determining the total riskposed by each radioactive contaminant in a given pathway, and then by rearranging the pathway equation to solve for an activity concentration set equal to a target cancer risk level of 10⁻⁶. At the scoping phase, these equations are "reduced" - and risk-based PRGs are calculated for each radionuclide of concern - using standardized exposure assumptions for each exposure route within each pathway and land-use combination. After the baseline risk assessment, PRGs can be recalculated using full risk equations and sitespecific exposure information obtained during the RL

### 4.1 RESIDENTIAL LAND USE

#### 4.1.1 GROUND WATER OR SURFACE WATER

Under the residential land-use scenario, risk from ground-water or surface water radioactive contaminants is assumed to be due primarily to direct ingestion and inhalation of volatile radionuclides released from the water to indoor air. However, because additional exposure routes (e.g., external radiation exposure due to immersion) are possible at some sites for some radionuclides, while only one exposure route may be relevant at others, the risk assessor always should consider all relevant exposure routes and add or modify exposure routes as appropriate. In the case illustrated below, risks from the two default exposure routes are combined, as follows:

- Total risk = Risk from ingestion of radionuclides from water in water (adult)
  - Risk from indoor inhalation of volatile radionuclides released from water (adult)

At the scoping phase, risk from indoor inhalation of volatile radionuclides is assumed to be relevant only for radionuclides with a Henry's Law constant of greater than  $1 \times 10^{-5}$  atm-m³/mole and a molecular weight of less than 200 g/mole. However, radionuclides that do not meet these criteria also may, under certain site-specific wateruse conditions, be volatilized into the air from water, and thus pose significant site risks (and require risk-based goals). Therefore, the ultimate decision regarding which contaminants should be considered must be made by the risk assessor on a site-specific basis following completion of the baseline risk assessment.

Total carcinogenic risk is calculated for each radionuclide separately by combining its appropriate oral and inhalation SFs with the two exposure pathways for water, as follows:

Total risk	-	SF _e x	Intake from ingestion of of radionuclides
	+	SF _i x	Intake from inhalation of volatile radionuclides

By including appropriate exposure parameters for each type of intake, rearranging and combining exposure terms in the total risk equation, and setting the target cancer risk level equal to  $10^{-6}$ , the risk-based PRG equation is derived as shown in Equation (10).

Equation (10'), presented in the next box, is the reduced version of Equation (10) based on the standard default values listed below. It is used to calculate risk-based PRGs for radionuclides in water at a pre-specified cancer risk level of 10⁻⁶ by combining each radionuclide's toxicity data with the standard default values for residential land-use exposure parameters.

After the baseline risk assessment, the risk assessor may choose to modify one or more of the exposure parameter default values or assumptions



#### REDUCED EQUATION FOR RADIONUCLIDE PRGS: RESIDENTIAL WATER - CARCINOGENIC EFFECTS

Risk-based PRG (pCi/L; TR = 10⁴)

where:

<ul> <li>oral (ingestion) slope factor (risk/pCi)</li> <li>inhalation slope factor (risk/pCi)</li> </ul>

9.5 x 10⁻¹¹

2(SF_) + 7.5(SF_)

in the risk equations to reflect site-specific conditions. In this event, radionuclide PRGs should be calculated using Equation (10) instead of Equation (10^{*}).

### 4.1.2 SOIL

Under residential land-use conditions, risk from radionuclides in soil is assumed to be due to direct ingestion and external exposure to gamma radiation. Soil ingestion rates differ for children and adults, therefore age-adjusted ingestion rate factors are used in the soil pathway equation. Calculation of the risk from the external radiation exposure route assumes that any gamma-emitting radionuclide in soil is uniformly distributed in that soil within a finite soil depth and density, and dispersed in an infinite plane geometry.

The calculation of external radiation exposure risk also includes two additional factors, the gamma shielding factor (Se) and the gamma exposure time factor  $(T_e)$ , which can be adjusted to account for both attenuation of radiation fields due to shielding (e.g., by structures, terrain, or engineered barriers) and for exposure times of less than 24-hours per day, respectively. S, is expressed as a fractional value between 0 and 1, delineating the possible risk reduction range from 0% to 100%, respectively, due to shielding. The default value of 0.2 for  $S_e$  for both residential and commercial/industrial land-use scenarios reflects the initial conservative assumption of a 20% reduction in external exposure due to shielding from structures (see EPA 1981). T_a is expressed as the quotient of the daily number of hours an individual is exposed directly to an external radiation field divided by the total number of exposure hours assumed each day for a given land-

(10')

use scenario (i.e., 24 hours for residential and 8 hours for commercial/industrial). The default value of 1 for  $T_e$  for both land-use scenarios reflects the conservative assumptions of a 24-hr exposure duration for residential populations (i.e., 24/24 = 1) and an 8-hr exposure duration for workers (i.e., 8/8 = 1). Values for both factors can (and, if appropriate, should) be modified by the risk assessor based on site-specific conditions.

In addition to direct ingestion of soil contaminated with radionuclides and exposure to external radiation from gamma-emitting radionuclides in soil, other soil exposure routes are possible, such as inhalation of resuspended radioactive particles, inhalation of volatile radionuclides, or ingestion of foodcrops contaminated by root or leaf uptake. The risk assessor should therefore identify all relevant exposure routes within the soil pathway and, if necessary, develop equations for risk-based PRGs that combine these exposure routes.

In the case illustrated below, the risk-based PRG is derived to be protective for exposure from the direct ingestion and external radiation routes. Total risk from soil due to ingestion and external radiation is calculated as follows:

Total risk	**	Risk from direct ingestion of radio-
from soil		nuclides in soil (child to adult)

 Risk from external radiation from gamma-emitting radionuclides in soil

Total risk for carcinogenic effects from each radionuclide of potential concern is calculated by combining the appropriate oral slope factor,  $SF_{o}$ , with the total radionuclide intake from soil, plus the appropriate external radiation slope factor,  $SF_{e}$ , with the radioactivity concentration in soil:

Total risk	=	SFo	х	Intake from direct ingestion	
				of soil	

+ SF, x Concentration of gammaemitting radionuclides in soil

Adding appropriate parameters, then combining and rearranging the equation to solve for concentration, results in Equation (11).

Equation (11') is the reduced version of Equation (11) based on the standard default values listed below. Risk-based PRGs for radionuclides in soil are calculated for a pre-specified cancer risk level of 10⁻⁶.

The age-adjusted soil ingestion factor  $(IF_{soil/adj})$  used in Equation (11) takes into account the difference in soil ingestion for two exposure groups — children of one to six years and all other individuals from seven to 31 years.  $IF_{soil/adj}$  is calculated for radioactive contaminants as shown in Equation (12). Section 3.1.2 provides additional discussion on the age-adjusted soil ingestion factor.

If any parameter values or exposure assumptions are adjusted after the baseline risk assessment to reflect site-specific conditions, soil PRGs should be calculated using Equation (11).

### 4.2 COMMERCIAL/INDUSTRIAL LAND USE

### 4.2.1 WATER

Under the commercial/industrial land use scenario, risk-based PRGs for radionuclides in ground water (and for radionuclides in surface water used for drinking water purposes) are based on residential exposures and calculated according to the procedures detailed in Section 4.1.1 (see Section 3.2.1 for the rationale for this approach). Risk-based PRGs should be calculated considering the possibility that both the worker and general population at large may be exposed to the same contaminated water supply.

### 4.2.2 SOIL

Under the commercial/industrial land use scenario, four soil exposure routes - direct ingestion, inhalation of volatile radionuclides, inhalation of resuspended radioactive particulates, and external exposure due to gamma-emitting radionuclides - are combined to calculate riskbased radionuclide PRGs in soil for adult worker exposures. Additional exposure routes (e.g., ingestion of foodcrops contaminated by radionuclide uptake) are possible at some sites, while only one exposure route (e.g., external radiation exposure only) may be relevant at others. The risk assessor should therefore consider and combine all relevant soil exposure routes, as necessary and appropriate, based on site-specific conditions.

	RADIONOCLIDE PROS: RESIDENTIAL SOIL - CA	KCINUGENIC EFFECTS
Total risk =	RS x $[(SF_0 \times 10^3 g/mg \times EF \times 1F_{solitadj}) + (SF_e \times 10^3)$	$g/kg \ge ED \ge D \ge SD \ge (1-S_*) \ge T_*$
RS (pCi/g; =	TR	(11)
risk-based)	$(SF_0 \times 10^3 \times EF \times IF_{solitedj}) + (SF_e \times 10^3 \times ED \times D \times 10^3)$	$SD \times (1-S_e) \times T_e)$
where:		50 5 = 550 =
Parameters	Definition (units)	Default Value
RS	radionuclide PRG in soil (pCi/g)	3 <u>-</u>
TR	target excess individual lifetime cancer risk (unitless)	10-6
SFo	oral (ingestion) slope factor (risk/pCi)	radionuclide-specific
SF.	external exposure slope factor (risk/yr per pCi/m2)	radionuclide-specific
EF	exposure frequency (days/yr)	350 days/yr
ED	exposure duration (yr)	30 yr
IFanlind	age-adjusted soil ingestion factor (mg-yr/day)	3600 mg-yr/day (see Equation (12))
D	depth of radionuclides in soil (m)	0.1 m
SD	soil density (kg/m3)	1.43 x 10 ³ kg/m ³
S,	gamma shickling factor (unitless)	0.2 (see Section 4.1.2)
T	samma exposure time factor (unitless)	1 (see Section 4.1.2)

	REDUCED EQUATION FOR RADIONUCLIDE PRGs: RESIDENTIAL SOIL - CARCINOGENIC EFFECTS	
Risk-hased PRG (pCl/g; TR = 10 ⁶ where:	$= \frac{1 \times 10^6}{1.3 \times 10^3 (SF_e) + 3.4 \times 10^6 (SF_e)}$	(11')
SF.	oral (ingestion) slope factor (risk/pCl) external exposure slope factor (risk/yr per pCi/m ² )	

	AGE-ADJUSTED SOIL INGESTION	FACTOR	
IF _{anikodj} (mg-yr/day)	= $(IR_{xeij/age 1.6} \times ED_{age 1.6}) + (IR_{xeij/age 3.57} \times EI$	) _{age 7.51} )	(12)
where:			
Parameters	Definition (units)	Default Value	
IF soiling IR soilings 1-6 IR soilings 7-31 ED age 1-6 ED age 7-31	age-adjusted soil ingestion factor (mg-yr/day) ingestion rate of soil ages I-6 (mg/day) ingestion rate of soil ages 7-31 (mg/day) exposure duration during ages 1-6 (yr) exposure duration during ages 7-31 (yr)	3600 mg-yr/day 200 mg/day 100 mg/day 6 yr 24 yr	

In the case illustrated below, total risk from radionuclides in soil is calculated as the summation of the individual risks from each of the four exposure routes listed above:

- Total risk = Risk from direct ingestion of radiofrom soil nuclides in soil (worker)
  - Risk from inhalation of volatile radionuclides (worker)
  - Risk from inhalation of resuspended radioactive particulates (worker)
  - Risk from external radiation from gamma-emitting radionuclides (worker)

Total risk for carcinogenic effects for each radionuclide is calculated by combining the appropriate ingestion, inhalation, and external exposure SF values with relevant exposure parameters for each of the four soil exposure routes as follows:

Total risk	-	SF _o x	Intake from direct ingestion of radionuclides in soil (worker)
	+	SF _i x	Intake from inhalation of volatile radionuclides (worker)

- SF_i x Intake from inhalation of resuspended radioactive particulates (worker)
- SF_e x Concentration of gamma-emitting radionuclides in soil (worker)

Adding appropriate parameters, and then combining and rearranging the equation to solve for concentration, results in Equation (13).

Equation (13') below is the reduced version of Equation (13) based on the standard default values below and a pre-specified cancer risk level of  $10^{-6}$ . It combines the toxicity information of a radionuclide with standard exposure parameters for commercial/industrial land use to generate the concentration of that radionuclide corresponding to a  $10^{-6}$  carcinogenic risk level due to that radionuclide.

If any parameter default values or assumptions are changed after the baseline risk assessment to reflect site-specific conditions, radionuclide soil PRGs should be derived using Equation (13).

### 4.2.3 SOIL-TO-AIR VOLATILIZATION FACTOR

The VF, defined in Section 3.3.1 for chemicals, also applies for radioactive contaminants with the following exceptions.

- Most radionuclides are heavy metal elements and are non-volatile under normal, ambient conditions. For these radionuclides, VF values need not be calculated and the risk due to the inhalation of volatile forms of these nuclides can be ignored for the purposes of determining PRGs.
- A few radionuclides, such as carbon-14 (C-14), tritium (H-3), phosphorus-32 (P-32), sulfur-35 (S-35), and other isotopes, are volatile under certain chemical or environmental conditions, such as when they are combined chemically with volatile organic compounds (i.e., the socalled radioactively-labeled or "tagged" organic compounds), or when they can exist in the environment in a variety of physical forms, such as C-14 labeled carbon dioxide (CO2) gas and tritiated water vapor. For these radionuclides, VF values should be calculated using the Hwang and Falco (1986) equation provided in Section 3.3.1 based on the chemical species of the compound with which they are associated.
- The naturally occurring, non-volatile radioisotopes of radium, namely Ra-226 and Ra-224, undergo radioactive decay and form inert, gaseous isotopes of radon, i.e., Rn-222 (radon) and Rn-220 (thoron), respectively. Radioactive radon and thoron gases cmanate from their respective parent radium isotopes in soil, escape into the air, and can pose cancer risks if inhaled. For Ra-226 and Ra-224 in soil, use the default values shown in the box on page 40 for VF and for SF_i in Equation (12) and Equation (12').

### 4.3 RADIATION CASE STUDY

This section presents a case study of a hypothetical CERCLA radiation site, the ACME Radiation Co. site, to illustrate the process of calculating pathway-specific risk-based PRGs for radionuclides using the risk equations and assumptions presented in the preceding sections of this chapter. The radiation site case study is modeled after the XYZ Co. site study discussed in

RADIONUCLIDE PRGs: COMMERCIAL/INDUSTRIAL SOIL CARCINOGENIC EFFECTS					
Total =	RS x ED x [(SF _o x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$ g/mg x EF x IR _{soil} ) + (SF _i x $10^{-3}$	2/kg x EF x IR _{air} x 1/VF)			
TUX	+ $(SF_i \times 10^3 g/kg \times EF \times IR_{air} \times 1/PEF)$ + $(SF_e \times 10^3 g/kg \times 1)^3 g/kg \times 10^3 g/$	$D \ge SD \ge (1-S_e) \ge T_e$			
RS =	TR	(13)			
(pCi/g; risk-based)	ED x [( $SF_ox10^3xEFxIR_{soil}$ ) + ( $SF_x10^3xEFxIR_{air}$ ) x (1/VF + 1/PI	F + (SF _e x10 ³ xDxSDx(1-S _e )xT _e )			
where:					
Parameters	Definition (units)	Default Value			
RS	radionuclide PRG in soil (pCi/g)	_			
TR	target excess individual lifetime cancer risk (unitless)	10 ⁻⁶			
SFi	inhalation slope factor (risk/pCi)	radionuclide-specific			
SF。	oral (ingestion) slope factor (risk/pCi)	radionuclide-specific			
SFe	external exposure slope factor (risk/yr per pCi/m ² )	radionuclide-specific			
EF	exposure frequency (days/yr)	250 days/yr			
ED	exposure duration (yr)	25 yr			
IR _{air}	workday inhalation rate of air (m ³ /day)	20 m ³ /day			
IR _{soil}	daily soil ingestion rate (mg/day)	50 mg/day			
VF	soil-to-air volatilization factor (m ⁻ /kg)	radionuclide-specific (see Section 4.2.3)			
PEF	particulate emission factor (m ⁻ /kg)	$4.63 \times 10^{\circ} \text{ m/kg}$ (see Section 3.3.2)			
Y D	depth of radionuclides in soil (m)	0.1 m			
⊮ SD	soil density (kg/m ² )	1.43 x 10° kg/m°			
Y Se	gamma shielding factor (unitless)	0.2 (see Section 4.1.2)			
VI.	gamma exposure factor (unitless)	1 (see Section 4.1.2)			

REDUCED EQUATION FOR RADIONUCLIDE PRGs: COMMERCIAL/INDUSTRIAL SOIL CARCINOGENIC EFFECTS*				
Risk-based PRG = $(pCi/g; TR = 10^{-6})$ where:	$\frac{1 \times 10^{-6}}{[(3.1 \times 10^{2} (SF_{o})) + ((1.3 \times 10^{8}/VF + 2.7 \times 10^{-2}) (SF_{i})) + (2.9 \times 10^{6} (SF_{e}))]}$	(13´)		
$\begin{array}{rcl} SF_{o} & = \\ SF_{i} & = \\ SF_{e} & = \\ VF & = \end{array}$	oral (ingestion) slope factor (risk/pCi) inhalation slope factor (risk/pCi) external exposure slope factor (risk/yr per pCi/m ² ) radionuclide-specific soil-to-air volatilization factor in m ³ /kg (see Section 3.3.1)			
*NOTE: See Section	n 4.2.3 when calculating PRGs for Ra-226 and Ra-224.			

Chapters 2 and 3. It generally follows a two-phase format which consists of a "at the scoping stage" phase wherein risk-based PRGs for radionuclides of potential concern are calculated initially using reduced equations based on PA/SI data, and then a second, "after the baseline risk assessment" phase wherein radionuclide PRGs are recalculated using full equations and modified site-specific parameter values based on RI/FS data.

Following an overview of the history and current status of the site presented in Section 4.3.1, Section 4.3.2 covers a number of important steps taken early in the scoping phase to calculate preliminary risk-based PRGs assuming a specific

SOIL DEF.	AULT VALUES F OR Ra-226 AND	OR VF AND SF, Ra-224
Radium	Default VF Value ( pCim ² Rn*)	Inhalation Slope Factor, SF _i (risk/pCi)**
Ra-226	8	1.1E-11
Ra-224	200	4.7E-11

 Calculated using values taken from NCRP 1976 and UNSCEAR 1982: Assumptions: (1) an average Ra-226 soil concentration of 1 pCi/g associated with an average ambient Rn-222 air concentration of 120 pCi/m³ and (2) an average Ra-224 soil concentration of 1 pCi/g associated with an average ambient Rn-220 air concentration of 5 pCi/m³.

** Slope factor values are for Rn-222 (plus progeny) and for Rn-220 (plus progeny).

land-use scenario. Section 4.3.3 then discusses how initial assumptions and calculations can be modified when additional site-specific information becomes available.

### 4.3.1 SITE HISTORY

The ACME Radiation Co. site is an abandoned industrial facility consisting of a large factory building situated on ten acres of land high-density surrounded by 8 residential neighborhood. Established in 1925, the ACME Co. manufactured luminous watch dials and gauges using radium-based paint and employed approximately 100 workers, mostly women. With the declining radium market, ACME phased out dial production and expanded its operations in 1960 to include brokering (collection and disposal) of low-level radioactive waste (LLW). After the company was issued a state license in 1961, ACME began receiving LLW from various nearby hospitals and research laboratories. In 1975, acting on an anonymous complaint of suspected mishandling of radioactive waste, state officials visited the ACME Co. site and cited the company for numerous storage and disposal violations. After ACME failed to rectify plant conditions identified in initial and subsequent citations, the state first suspended, and then later revoked its operating license in 1978. Around the same time, officials detected radium-226 (Ra-226) contamination at a few neighboring locations off site. However, no action was taken against the company at that time. When ACME filed for bankruptcy in 1985, it closed its facility before completing cleanup.

In 1987, the state and EPA conducted an acrial gamma survey over the ACME Radiation Co. site and surrounding properties to investigate the potential extent of radioactive contamination in these areas. The overflight survey revealed several areas of elevated exposure rate readings, although individual gamma-emitting radionuclides could not be identified. When follow-up ground level surveys were performed in 1988, numerous "hot spots" of Ra-226 were pinpointed at various locations within and around the factory building. Three large soil piles showing enhanced concentrations of Ra-226 were discovered along the southern border. Approximately 20 rusting drums labelled with LLW placards also were discovered outside under a covered storage area. Using ground-penetrating radar, EPA detected subsurface magnetic anomalies in a few locations within the property boundary which suggested the possibility of buried waste drums. Based on interviews with people living near the site and with former plant workers, the state believes that radium contaminated soil may have been removed from the ACME site in the past and used locally as fill material for the construction of new homes and roadbeds. Site access is currently limited (but not entirely restricted) by an existing security fence.

In 1988, EPA's regional field investigation team completed a PA/SI. Based on the PA/SI data, the ACME Radiation Co. site scored above 28.50 using the HRS and was listed on the National Priorities List in 1989. Early in 1990, an RI/FS was initiated and a baseline risk assessment is currently in progress.

### 4.3.2 AT THE SCOPING PHASE

In this subsection, several steps are outlined to show by example how initial site data are used at the scoping phase to calculate risk-based PRGs for radionuclides in specific media of concern. Appropriate sections of Chapters 2 and 3 should be consulted for more detailed explanations for each step considered below. Identify Media of Concern. A large stream runs along the western border of the site and feeds into a river used by some of the local residents for fishing and boating. Supplemental water intake ducts for the municipal water treatment plant are located approximately 300 yards downriver, and the site is situated over an aquifer which serves as the primary drinking water supply for a community of approximately 33,000 people.

Analyses of ground water, soil, and stream sediment samples taken during the PA/SI revealed significant levels of radionuclide contamination. Potential sources of contamination include the soil piles, process residues in soil, and radionuclides leaking from buried drums. Air filter samples and surface water samples from the stream and river showed only background levels of activity. (Background concentrations were determined from analyses conducted on a limited number of air, ground water, surface water, and soil samples collected approximately one mile from the site.)

The data show that the media of potential concern at this site include ground water and soil. Although stream water and river water were not found to be contaminated, both surface water bodies may become contaminated in the future due to the migration of radionuclides from sediment, from the exposed soil piles, or from leaking drums. Thus, surface water is another medium of potential concern.

For simplicity, only soil will be discussed as the medium of concern during the remainder of this case study. Procedures discussed for this medium can nevertheless be applied in a similar manner to all other media of concern.

Identify Initial List of Radionuclides of Concern. The PA/SI for the ACME Radiation Co. site identified elevated concentrations of five radionuclides in soil (Ra-226, tritium (H-3), carbon-14 (C-14), cesium (Cs-137), and strontium (Sr-90)). These comprise the initial list of radionuclides of potential concern.

Site records indicate that radioisotopes of cobalt (Co-60), phosphorus (P-32), sulfur (S-35), and americium (Am-241 and Am-243) were included on the manifests of several LLW drums in the storage area and on the manifests of other drums suspected to be buried onsite. Therefore, although not detected in any of the initial soil samples analyzed, Co-60, P-32, S-35, Am-241, and Am-243 are added to the list for this medium because of their potential to migrate from leaking buried drums into the surrounding soil.

Identify Probable Land Uses. The ACME Radiation Co. site is located in the center of a rapidly developing suburban community comprised of single and multiple family dwellings. The area immediately encircling the site was recently rezoned for residential use only; existing commercial and light industrial facilities are currently being relocated. Therefore, residential use is determined to be the most reasonable future land use for this site.

Identify Exposure Pathways, Parameters, and Equations. During the scoping phase, available site data were neither sufficient to identify all possible exposure pathways nor adequate enough to develop site-specific fate and transport equations and parameters. Therefore, in order to calculate initial risk-based PRGs for radionuclides of potential concern in soil, the standardized default soil exposure equation and assumptions provided in this chapter for residential land use in Section 4.1.2 are selected. (Later in this case study, examples are provided to illustrate how the full risk equation (Equation (11)) and assumptions are modified when baseline risk assessment data become available.)

For the soil pathway, the exposure routes of concern are assumed to be direct ingestion of soil contaminated with radionuclides and exposure to external radiation from gamma-emitting radionuclides. Again, although soil is the only medium discussed throughout this case study, exposure pathways, parameters, equations, and eventually risk-based concentrations would need to be identified and developed for all other media and exposure pathways of potential concern at an actual site.

Identify Toxicity Information. To calculate media-specific risk-based PRGs, reference toxicity values for radiation-induced cancer effects are required (i.e., SFs). As stated previously, soll ingestion and external radiation are the exposure routes of concern for the soil pathway. Toxicity information (i.e., oral, inhalation, and external exposure SFs) for all radionuclides of potential concern at the ACME Radiation Co. site are obtained from IRIS or HEAST, and are shown in the box on the following page.

Radionuclides	Radioactive Half-life (yr)	Decay Mode	ICRP Lung Classification	Inhalation Slope Factor (risk/pCi)	Ingestion Slope Factor (risk/pCl)	External Exposure Slope Factor (risk/yr per pCl/m ² )
Н-3	12	beta	8	7.8E-14	5.5E-14	NA
C-14	5730	beta	8	6.4E-15	9.1E-13	NA
P-32	0.04	beta	D	3.0E-12	3.5E-12	NA
S-35	0.24	beta	D	1.9E-13	2.2E-13	NA
Co-60	5	beta/gamma	Y	1.6E-10	1.5E-11	1.3E-10
Sr-90	29	beta	Ð	5.6E-11	3.3E-11	NA
Cs-137	30	beta	D	1.9E-11	2.8E-11	NA
Ra-226	1600	alpha/gamma	w	3.0E-09	1.2E-10	4.2E-13
Am-241	432	alpha/gamma	w	4.0E-08	3.1E-10	1.6E-12
Am-243	7380	alpha/gamma	w	4.0E-08	3.1E-10	3.6E-12

RADIATION CASE STUDY: TOXICITY INFORMATION FOR RADIONUCLIDES OF POTENTIAL CONCERN*

* Sources: HEAST and Federal Guidance Report No. 11. All information in this example is for illustration only.

NA = Not applicable (i.e., these radionuclides are not gamma-emitters and the direct radiation exposure pathway can be ignored).

42

Calculate Risk-based PRGs. At this step, riskbased PRGs are calculated for each radionuclide of potential concern using the reduced risk Equation (11') in Section 4.1.2, SF values obtained from IRIS and HEAST, and standardized default values for parameters for the residential land-use scenario. To calculate the risk-based PRG for Co-60 at a pre-specified target risk level of 10⁻⁶, for example, its ingestion SF of 1.5 x 10⁻¹¹ and its external exposure SF of 1.3 x 10⁻¹⁰ are substituted into Equation (11'), along with the standardized default values, as follows:

Risk-based PRG =  $\frac{1 \times 10^4}{1.3 \times 10^3 (SF_e) + 3.4 \times 10^6 (SF_e)}$ (pCi/g; TR = 10⁴)

where:

SF_o = oral (ingestion) slope factor for Co-60 = 1.5 x 10⁻¹¹ (risk/pCi)

SF_{*} = external exposure slope factor for Co-60 = 1.3 x 10⁻¹⁰ (risk/yr per pCi/m²)

Substituting the values for SF_o and SF_e for Co-60 into Equation (11') results in:

Risk-based PRG for Co-60 (pCi/g; TR = 10⁻⁶) =

 $\frac{1 \times 10^6}{[(1.3 \times 10^3)(1.5 \times 10^{10}) + (3.4 \times 10^6)(1.3 \times 10^{10})]}$ 

= 0.002 pCi of Co-60/g of soil

In a similar manner, risk-based PRGs can be calculated for all other radionuclides of concern in soil at the ACME Radiation Co. site. These PRGs are presented in the next box.

### 4.3.3 AFTER THE BASELINE RISK ASSESSMENT

In this subsection, several steps are outlined which demonstrate how site-specific data obtained during the baseline risk assessment can be used to recalculate risk-based PRGs for radionuclides in soil. Appropriate sections of Chapters 2 and 3 should be consulted for more detailed explanations for each step considered below.

Review Media of Concern. During the RI/FS, gamma radiation surveys were conducted in the yards of several homes located within a two-block radius of the ACME Radiation Co. site. Elevated exposure rates, ranging from approximately two to four times the natural background rate, were

### RADIATION CASE STUDY: INITIAL RISK-BASED PRGs FOR RADIONUCLIDES IN SOIL*

H.3	14.000
Sr-90 (only)	23
P-32	220
S-35	3,500
C-14	850
Co-60	0.002
Cs-137 (only)	27
Ra-226 (only)	0.6
Am-241	0.2
Am-243 (only)	7.9 x 10 ⁻²

measured on properties immediately bordering the site. Measurements onsite ranged from 10 to 50 times background. In both cases, enhanced soil concentrations of Ra-226 (and decay products) and several other gamma-emitting radionuclides were discovered to be the sources of these elevated exposure rates. Therefore, soil continues as a medium of potential concern.

Modify List of Radionuclides of Concern. During scoping, five radionuclides (Ra-226, H-3, C-14, Cs-137, and Sr-90) were detected in elevated concentrations in soil samples collected at the ACME Radiation Co. site. These made up the initial list of radionuclides of potential concern. Although not detected during the first round of sampling, five additional radionuclides (P-32, S-35, Co-60, Am-241, and Am-243) were added to this list because of their potential to migrate from buried leaking drums into the surrounding soil.

With additional RI/FS data, some radionuclides are now added to the list, while others are dropped. For example, soil analyses failed to detect P-32 (14-day half-life) or S-35 (87day half-life) contamination. Decay correction calculations strongly suggest that these radionuclides should not be present onsite in detectable quantities after an estimated burial time of 30 years. Therefore, based on these data, P-32 and S-35 are dropped from the list. Soil data also confirm that decay products of Ra-226, Sr-90, Cs-137, and Am-243 (identified in the first box below)

-43-

are present in secular equilibrium (i.e., equal activity concentrations) with their respective parent isotopes.

Assuming secular equilibrium, slope factors for the parent isotope and each of its decay series members are summed. Parent isotopes are designated with a "+D" to indicate the composite

slope factors of its decay chain (shown in bold face in the second box below). Thus, Ra-226+D, Sr-90+D, Cs-137+D, and Am-243+D replace their respective single-isotope values in the list of radionuclides of potential concern, and their composite SFs are used in the full soil pathway equation to recalculate risk-based concentrations.

Parent Radionuclide	Decay Product(s) (Half-life)
Ra-226	Rn-222 (4 days), Po-218 (3 min), Pb-214 (27 min), Bi-214 (20 min), Po-214 (<1 s), Pb-210 (22 yr), Bi-210 (5 days), Po-210 (138 days)
Sr-90	Y-90 (14 hr)
Cs-137	Ba-137m (2 min)
Am-243	Np-239 (2 days)

	Slope F	Pactors	
Decay Series	Inhalation	Ingestion	External
Ra-226	3.0E-09	1.2E-10	4.2E-13
Rn-222	7.2E-13	-	2.2E-14
Po-218	5.8E-13	2.8E-14	0.0E+00
Pb-214	2.9E-12	1.8E-13	1.5E-11
Bi-214	2.2E-12	1.4E-13	8.0E-11
Po-214	2.8E-19	1.0E-20	4.7E-15
Pb-210	1.7E-09	6.5E-10	1.8E-13
Bi-210	8.1E-11	1.9E-12	0.0E+00
Po-210	2.7E-09	2.6E-10	4.8E-16
Ra-226+D	7.5E-09	1.0E-09	9.6E-11
Sr-90	5.6E-11	3.3E-11	0.0E+00
Y-90	5.5E-12	3.2E-12	0.0E+00
Sr-90+D	6.2E-11	3.6E-11	0.0E+00
Cs-137	1.9E-11	2.8E-11	0.0E+00
Ba-137m	6.0E-16	2.4E-15	3.4E-11
Cs-137+D	1.9E-11	2.8E-11	3.4E-11
Am-243	4.0E-08	3.1E-10	3.6E-12
Np-239	1.5E-12	9.3E-13	1.1E-11
	1.05 0.5	3 112 10	1 512 11

* All information in this example is for illustration purposes only.

Review Land-use Assumptions. At this step, the future land-use assumption chosen during scoping is reviewed. Since the original assumption of future residential land use is supported by RI/FS data, it is not modified.

Modify Exposure Pathways, Parameters, and Equations. Based on site-specific information, the upper-bound residence time for many of the individuals living near the ACME Radiation Co. site is determined to be 45 years rather than the default value of 30 years. Therefore, the exposure duration parameter used in Equation (11) in Section 4.1.2 is substituted accordingly. It is also determined that individuals living near the site are only exposed to the external gamma radiation field approximately 18 hours each day, and that their homes provide a shielding factor of about 0.5 (i.e., 50%). Therefore, values for  $T_e$  and  $S_e$  are changed to 0.75 (i.e., 18 hr/24 hr) and 0.5, respectively.

Modify Toxicity Information. As discussed above in the section on modifying the list of radionuclides of concern, oral, inhalation, and external exposure slope factors for Ra-226, Sr-90, Cs-137, and Am-243 were adjusted to account for the added risks (per unit intake and/or exposure) contributed by their respective decay series members that are in secular equilibrium.

Recalculate Risk-based PRGs. At this step, risk-based PRGs are recalculated for all remaining radionuclides of potential concern using the full risk equation for the soil pathway (i.e., Equation (11)) modified by revised site-specific assumptions regarding exposures, as discussed above.

To recalculate the risk-based PRG for Co-60 at a pre-specified target risk level of  $10^{-6}$ , for example, its ingestion SF of  $1.5 \times 10^{-11}$ , and its external exposure SF of  $1.3 \times 10^{-10}$  are substituted into Equation (11), along with other site-specific parameters, as shown in the next box.

In a similar manner, risk-based PRGs can be recalculated for all remaining radionuclides of potential concern in soil at the ACME Radiation Co. site. These revised PRGs are presented in the box on the next page. In those cases where calculated risk-based PRGs for radionuclides are below current detection limits, risk assessors should contact the Superfund Health Risk Technical Support Center for additional guidance.

RS for Co-60	(oCile = TR	
risk-based)	$\frac{(SF_o \times 10^3 \times EF \times IF_{solitadj}) + (SF_e \times 10^3 \times IF_{solitadj})}{(SF_e \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times 10^3 \times$	D x D x SD x (1-S,) x T,
	= 0.003 pCi/g	
where:		
Parameters	Definition (units)	Revised Value
RS	radionuclide PRG in soil (pCi/g)	
TR	target excess individual lifetime cancer risk (unitless)	10-6
SF	oral (ingestion) slope factor (risk/pCi)	1.5 x 10 ⁻¹¹ (risk/pCl)
SF.	external exposure slope factor (risk/yr per pCi/m2)	1.3 x 10 ⁻¹⁰ (risk/yr per pCi/m ² )
EF	exposure frequency (days/yr)	350 days/yr
ED	exposure duration (yr)	45 yr
IFacilitati	age-adjusted soil ingestion factor (mg-yr/day)	5100 mg-yr/day
D	depth of radionuclides in soil (m)	0.1 m
SD	soil density (kg/m ³ )	1.43 x 103 kg/m3
S.	gamma shielding factor (unitless)	0.5
T	gamma exposure time factor (unitiess)	0.75

for individuals 7 to 46 years of age.)

-45-

### RADIATION CASE STUDY: REVISED RISK-BASED PRGs FOR RADIONUCLIDES IN SOIL*

Radionuclides	Risk-based Soil PRG (pCi/g)
H-3	10,200
Sr-90+D	20
C-14	620
Co-60	0.003
Cs-137+D	0.01
Ra-226+D	0.004
Am-241	0.2
Am-243+D	0.03

* Calculated for illustration only. Values have been rounded off.

# REFERENCES

Andelman, J.B. 1990. Total Exposure to Volatile Organic Chemicals in Potable Water. N.M. Ram, R.F. Christman, K.P. Cantor (eds.). Lewis Publishers.

Cowherd, C., Muleski, G., Engelhart, P., and Gillete, D. 1985. Rapid Assessment of Exposure to Particulate Emissions from Surface Contamination. Prepared for EPA Office of Health and Environmental Assessment. EPA/600/8-85/002.

Environmental Protection Agency (EPA). 1981. Population Exposures to External Natural Radiation Background in the U.S. Office of Radiation Programs. ORP/SEPD-80-12.

EPA. 1984. Evaluation and Selection of Models for Estimating Air Emissions from Hazardous Waste Treatment, Storage, and Disposal Facilities. Office of Air Quality Planning and Standards. EPA/450/3-84/020.

EPA. 1986. Development of Advisory Levels for PCBs Cleanup. Office of Health and Environmental Assessment. EPA/600/21.

EPA. 1988a. CERCLA Compliance With Other Laws Manual, Part I (Interim Final). Office of Emergency and Remedial Response. EPA/540/G-89/006 (OSWER Directive #9234.1-01).

EPA. 1988b. Estimating Exposures to 2,3,7,8-TCDD (External Review Draft). Office of Health and Environmental Assessment. EPA/600/6-88/005A.

EPA. 1988c. Guidance for Conducting Remedial Investigations and Feasibility Studies Under CERCLA. Interim Final. Office of Emergency and Remedial Response. EPA/540/G-89/004 (OSWER Directive #9355.3-01).

EPA. 1988d. Guidance on Remedial Actions for Contaminated Ground Water at Superfund Sites. Interim Final. Office of Emergency and Remedial Response. EPA/540/G-88/003 (OSWER Directive #9283.1-2).

EPA. 1988f. Superfund Exposure Assessment Manual. Office of Emergency and Remedial Response. EPA/540/1-88/001 (OSWER Directive 9285.5-1).

EPA. 1988. Availability of the Integrated Risk Information System (IRIS). 53 Federal Register 20162.

EPA. 1989a. CERCLA Compliance With Other Laws Manual, Part II: Clean Air Act and Other Environmental Statutes and State Requirements. Office of Emergency and Remedial Response. EPA/540/G-89/009 (OSWER Directive #9234.1-01).

EPA. 1989b. Interim Final Guidance on Preparing Superfund Decision Documents. Office of Emergency and Remedial Response. OSWER Directive 9355.3-02.

EPA. 1989c. Methods for Evaluating the Attainment of Cleanup Standards (Volume 1: Soils and Solid Waste). Statistical Policy Branch. NTIS #PB89-234-959/AS.

EPA. 1989d. Risk Assessment Guidance for Superfund: Volume I – Human Health Evaluation Manual (Part A, Baseline Risk Assessment). Interim Final. Office of Emergency and Remedial Response. EPA/540/1-89/002.

EPA. 1990a. Catalog of Superfund Program Publications. Office of Emergency and Remedial Response. OSWER Directive 9200.7-02A. EPA. 1990b. Guidance for Data Usability in Risk Assessment. Office of Solid Waste and Emergency Response. EPA/540/G-90/008 (OSWER Directive #9285.7-05).

EPA. 1990c. Guidance on Remedial Actions for Superfund Sites with PCB Contamination. Office of Emergency and Remedial Response. EPA/540/G-90/007 (OSWER Directive #9355.4-01).

EPA. 1990d. "National Oil and Hazardous Substances Pollution Contingency Plan (Final Rule)." 40 CFR Part 300; 55 Federal Register 8666.

EPA. 1991a. Conducting Remedial Investigations/Feasibility Studies for CERCLA Municipal Landfill Sites. Office of Emergency and Remedial Response. EPA/540/P-91/001 (OSWER Directive #9355.3-11).

EPA. 1991b. Risk Assessment Guidance for Superfund, Vol. 1, Human Health Evaluation Manual, Supplemental Guidance: "Standard Default Exposure Factors." (Interim Final). Office of Emergency and Remedial Response. OSWER Directive 9285.6-03.

EPA. 1991c. Role of the Baseline Risk Assessment in Superfund Remedy Selection Decisions. Office of Solid Waste and Emergency Response. OSWER Directive 9355.0-30.

EPA. 1991d. Risk Assessment Guidance for Superfund: Volume I – Human Health Evaluation Manual (Part C, Risk Evaluation of Remedial Alternatives). Interim. Office of Emergency and Remedial Response. OSWER Directive 9285.7-01C.

EPA. Health Effects Assessment Summary Tables (HEAST). Published quarterly by the Office of Research and Development and Office of Solid Waste and Emergency Response. NTIS #PB 91-921100.

Hwang, S.T., and Falco, J.W. 1986. Estimation of Multimedia Exposures Related to Hazardous Waste Facilities. Cohen, Y. (ed). Plenum Publishing Corp.

-48-

# **APPENDIX** A

# **ILLUSTRATIONS OF CHEMICALS THAT "LIMIT" REMEDIATION**

In many cases, one or two chemicals will drive the cleanup at a site, and the resulting cumulative medium or site risk will be approximately equal to the potential risk associated with the individual remediation goals for these chemicals. These "limiting chemicals" are generally either chemicals that are responsible for much of the baseline risk (because of either high toxicity or presence in high concentrations), or chemicals that are least amenable to the selected treatment method. By cleaning up these chemicals to their goals, the other chemicals typically will be cleaned up to levels much lower than their corresponding goals. The example given in the box below provides a simple illustration of this principle.

The actual circumstances for most remediations will be much more complex than those described in the example (e.g., chemicals will be present at different baseline concentrations and will be treated/removed at differing rates); however, the same principle of one or perhaps two chemicals limiting the site cleanup usually applies, even in more complex cases.

Unless much is known about the performance of a remedy with respect to all the chemicals present at the site, it may not be possible to determine which of the site contaminants will drive final risk until well into remedy the implementation. Therefore, it generally is not possible to predict the cumulative risk that will be present at the site during or after remediation. In some situations, enough will be known about the site conditions and the performance of the remedy estimate post-remedy concentrations of to chemicals or to identify the chemical(s) that will dominate the residual risk. If this type of information is available, it may be necessary to modify the risk-based remediation goals for individual chemicals.

### SIMPLE ILLUSTRATION OF A CHEMICAL THAT LIMITS REMEDIATION

Two chemicals (A and B) are present in ground water at a site at the same baseline concentrations. Remediation goals were identified for both A and B. Chemical A's goal is 0.5 ug/L, which is associated with a potential risk of  $10^{-6}$ . Chemical B's goal is 10 ug/L, which is also associated with a potential risk of  $10^{-6}$ . The calculated cumulative risk at remediation goals is therefore  $2 \times 10^{-6}$ . Assuming for the purposes of this illustration that A and B are treated or removed at the same rate, then the first chemical to meet its goal will be B. Remediation must continue at this site, however, until the goal for chemical A has been met. When the concentration of A reaches 0.5 ug/L, then remediation is complete. A is at its goal and has a risk of  $10^{-6}$ . B is at 1/20 of its goal with a risk of  $5 \times 10^{-6}$ . The total risk ( $1 \times 10^{-6} + 5 \times 10^{-6}$ ) is approximately  $10^{-6}$  and is due to the presence of A.

This example illustrates that the final risk for a chemical may not be equal to the potential risk associated with its remediation goal, and, in fact, can be much less than this risk. Although the potential risk associated with Chemical B's goal is  $10^{-6}$ , the final residual risk associated with B is  $5 \times 10^{-8}$ . Thus, if one were to calculate the cumulative risk at PRGs prior to remedy implementation, one would estimate total medium risk of  $2 \times 10^{-6}$ , however, the residual cumulative risk after remediation is  $1 \times 10^{-6}$ .

# APPENDIX B

# RISK EQUATIONS FOR INDIVIDUAL EXPOSURE PATHWAYS

This appendix presents individual risk equations for each exposure pathway presented in Chapter 3. These individual risk equations can be used and rearranged to derive full risk equations required for calculating risk-based PRGs. Depending on the exposure pathways that are of concern for a land-use and medium combination, different individual risk equations can be combined to derive the full equation reflecting the cumulative risk for each chemical within the medium. See Chapter 3 for examples of how equations are combined and how they need to be rearranged to solve for risk-based PRGs. Note that in this appendix, the term HQ is used to refer to the risk level associated with noncarcinogenic effects since the equations are for a single contaminant in an individual exposure pathway.

The following sections list individual risk equations for the ground water, surface water, and soil pathways. Risk equations for exposure pathways not listed below can be developed and combined with those listed. In particular, dermal exposure and ingestion of ground water contaminated by soil leachate, for which guidance is currently being developed by EPA, could be included in the overall exposure pathway evaluation.

# B.1 GROUND WATER OR SURFACE WATER — RESIDENTIAL LAND USE

Both the ingestion of water and the inhalation of volatiles are included in the standard default equations in Section 3.1.1. If only one of these exposure pathways is of concern at a particular site, or if one or both of these pathways needs to be combined with additional pathways, a sitespecific equation can be derived.

The parameters used in the equations presented in the remainder of this section are explained in the following text box.

#### B.1.1 INGESTION

The cancer risk due to ingestion of a contaminant in water is calculated as follows:

PAR	AMETERS FOR SURFACE WATER/GROUND WAT	ER - RESIDENTIAL LAND USE
Parameter	Definition	Default Value
с	chemical concentration in water (mg/L)	-
SF	inhalation cancer slope factor ((mg/kg-day)')	chemical-specific
SF	oral cancer slope factor ((mg/kg-day)1)	chemical-specific
RfD	oral chronic reference dose (mg/kg-day)	chemical-specific
RfD;	inhalation chronic reference dose (mg/kg-day)	chemical-specific
BW	adult body weight (kg)	70 kg
AT	averaging time (yr)	70 yr for cancer risk
		30 yr for noncancer HI (equal to ED)
EF	exposure frequency (days/yr)	350 days/yr
ED	exposure duration (yr)	30 yr
K	volatilization factor (L/m ³ )	0.0005 x 1000 L/m ³ (Andelman 1990)
IR,	daily indoor inhalation rate (m3/day)	15 m ³ /day
IR,	daily water ingestion rate (L/day)	2 L/day

Risk from ingestion =  $\underline{SF_x C x IR_x EF x ED}$ of water (adult) BW x AT x 365 days/yr

The noncancer HQ due to ingestion of a contaminant in water is calculated as follows:

HQ due to ingestion =  $\frac{C \times IR_{x} \times EF \times ED}{RfD_{o} \times BW \times AT \times 365 \text{ days/yr}}$ 

### **B.1.2 INHALATION OF VOLATILES**

The cancer risk due to inhalation of a volatile contaminant in water is calculated as follows:

Risk from = <u>SF, x C x K x IR, x EF x ED</u> inhalation BW x AT x 365 days/yr of volatiles in water (adult)

The noncancer HQ due to inhalation of a volatile contaminant in water is calculated as follows:

HQ due to = <u>C x K x IR, x EF x ED</u> inhalation RfD, x BW x AT x 365 days/yr of volatiles in water (adult)

### B.2 SOIL – RESIDENTIAL LAND USE

and/or inhalation of particulates, are of concern at a particular site, then a site-specific equation can be derived.

The parameters used in the equations presented in the remainder of this section are explained in the text box below.

### B.2.1 INGESTION OF SOIL

The cancer risk from ingestion of contaminated soil is calculated as follows:

Risk from = <u>SF_o x C x 10⁻⁶ kg/mg x EF x IF_{solitadj}</u> ingestion AT x 365 days/yr of soil

The noncancer HQ from ingestion of contaminated soil is calculated as follows:

HQ from =  $C \ge 10^{-6} \text{ kg/mg} \ge \text{EF} \ge \text{IF}_{scil/adj}$ ingestion RfD_e x AT x 365 days/yr of soil

### **B.2.2 INHALATION OF VOLATILES**

The cancer risk caused by inhalation of volatiles released from contaminated soil is:

Risk from = <u>SF, x C x ED x EF x IR_{sir} x (1/VF)</u> inhalation AT x BW x 365 days/yr of volatiles

The equation for calculating the noncancer HQ from inhalation of volatiles released from soil is:

Parameter	Definition	Default Value
с	chemical concentration in soil (mg/kg)	-
SF	inhalation cancer slope factor ((mg/kg-day)-1)	chemical-specific
SF	oral cancer slope factor ((mg/kg-day) ⁻¹ )	chemical-specific
RfD	oral chronic reference dose (mg/kg-day)	chemical-specific
RfD	inhalation chronic reference dose (mg/kg-day)	chemical-specific
BW	adult body weight (kg)	70 kg
AT	averaging time (yr)	70 yr for cancer risk
		30 yr for noncancer HI (equal to ED
EF	exposure frequency (days/yr)	350 days/yr
ED	exposure duration (yr)	30 yr
IR,	daily indoor inhalation rate (m3/day)	15 m ³ /day
IFanilati	age-adjusted soil ingestion factor (mg-yr/kg-day)	114 mg-yr/kg-day
VF	soil-to-air volatilization factor (m3/kg)	chemical specific (see Section 3.3.1)
PEF	particulate emission factor (m ³ /kg)	4.63 x 10° m3/kg (see Section 3.3.2)

HQ from =  $C \times ED \times EF \times IR_{str} \times (1/VF)$ inhalation RfD_i x BW x AT x 365 days/yr of volatiles

### **B.2.3 INHALATION OF PARTICULATES**

Cancer risk due to inhalation of contaminated soil particulates is calculated as:

Risk = <u>SF, x C x ED x EF x IR_{sir} x (1/PEF)</u> from AT x BW x 365 days/yr inhalation of particulates

The noncancer HQ from particulate inhalation is calculated using this equation:

HQ from = C x ED x EF x IR_{sic} x (1/PEF) inhalation RfD_i x BW x AT x 365 days/yr of particulates

### B.3 SOIL – COMMERCIAL/ INDUSTRIAL LAND USE

All three of the exposure pathways detailed below are included in the standard default equation in Section 3.2.2. If only one or some combination of these exposure pathways are of concern at a particular site, a site-specific equation can be derived.

The parameters used in the equations presented in the remainder of this section are explained in the text box below.

### B.3.1 INGESTION OF SOIL

The cancer risk from ingestion of contaminated soil is calculated as follows:

Risk from =  $\frac{SF_{x} \times C \times 10^{4} \text{ kg/mg x } EF \text{ x } ED \text{ x } IR_{sell}}{BW \text{ x } AT \text{ x } 365 \text{ days/yr}}$ of soil

The noncancer HQ from ingestion of contaminated soil is calculated as follows:

HQ from =  $\frac{C \times 10^4 \text{ kg/mg x EF x ED x IR_{sol}}{\text{RfD}_0 \text{ x BW x AT x 365 days/yr}}$ of soil

### **B.3.2 INHALATION OF VOLATILES**

The cancer risk caused by inhalation of volatiles released from contaminated soil is:

Risk from	=	SF, x C x ED x EF x IR x (1/VF)
inbalation		AT x BW x 365 days/yr
of volatiles		

The equation for calculating the noncancer HQ from inhalation of volatiles released from soil is:

HQ from	=	C x ED x EF x IR x (1/VF)
inhalation		RfD; x BW x AT x 365 days/yr
of volatiles		

Note that the VF value has been developed specifically for these equations; it may not be applicable in other technical contexts.

PARAMETERS FOR SOIL - COMMERCIAL/INDUSTRIAL LAND USE		
Parameter	Definition	Default Value
С	chemical concentration in soil (mg/kg)	-
SF	inhalation cancer slope factor ((mg/kg-day)-1)	chemical-specific
SF.	oral cancer slope factor ((mg/kg-day)-1)	chemical-specific
RfD。	oral chronic reference dose (mg/kg-day)	chemical-specific
RfD	inhalation chronic reference dose (mg/kg-day)	chemical-specific
BW	adult body weight (kg)	70 kg
AT	averaging time (yr)	70 yr for cancer risk
		30 yr for noncancer HI (equal to ED)
EF	exposure frequency (days/yr)	250 days/yr
ED	exposure duration (yr)	25 yr
IRair	workday inhalation rate (m3/day)	20 m ³ /day
IR _{acdi}	soil ingestion rate (mg/day)	50 mg/day
VF	soil-to-air volatilization factor (m3/kg)	chemical specific (see Section 3.3.1)
DICIC	positivitate emission factor (milita)	4.62 x 10 m36c /can Cartion 2 2 2)

### **B.3.3 INHALATION OF PARTICULATES**

Cancer risk due to inhalation of contaminated soil particulates is calculated as:

Risk from =  $\underline{SF_i x C x ED x EF x IR_{uir} x (1/PEF)}$ inhalation AT x BW x 365 days/yr of particulates The noncancer HQ from particulate inhalation is calculated using this equation:

 $\begin{array}{rcl} HQ \mbox{ from } = & \underline{C \ x \ ED \ x \ EF \ x \ IR_{ir} \ x \ (1/PEF)} \\ \mbox{inhalation} & RfD_i \ x \ BW \ x \ AT \ x \ 365 \ days/yr \end{array}$ 

Appendix E

Photo Log



ST-18



ST-18



Luke Air Force Base Five Year Review Photographs 1 and 2



DP-23 looking south



DP-23 looking north



Photographic Summary

Luke Air Force Base Five Year Review Photographs



RW-02



Inside Tire Barrier



Photographic Summary

Luke Air Force Base Five Year Review Photographs 5 and 6



View of Bank Stabilization at RW-02



SS-42 pad where SVE pad was



Luke Air Force Base Five Year Review Photographs



SS-42 General View



FT-07 E



Luke Air Force Base Five Year Review Photographs 9 and 10







LF-03



Luke Air Force Base Five Year Review Photographs 11 and 12



**OT-4**1



LF-25



Luke Air Force Base Five Year Review Photographs 13 and 14



OT-41



DP-05



Luke Air Force Base Five Year Review Photographs 15 and 16



DP-05



Photographic Summary

Luke Air Force Base Five Year Review Photographs 17 and 18

# Appendix F

Interview Notes

### Lang, Kent

From: `ent: .o: Subject: Rothrock Charles J Civilian 56 CES/CEVC [charles.rothrock@luke.af.mil] Monday, August 13, 2001 7:30 AM klang@arcadis-us.com FW: Five-Year Review Project-Interviews

Kent,

Not exactly in the right format, but I guess we can include it.

Jeff

-----Original Message-----From: MGprts@aol.com [mailto:MGprts@aol.com] Sent: Sunday, August 12, 2001 5:17 PM To: charles.rothrock@luke.af.mil Subject: Re: Five-Year Review Project-Interviews

Jeff; I'm still on the road....For some reason I could not open the file on my laptop and the Original is at home...So here is a letter that I hope will

keep your feet out of the fire...Sorry I let this slip.... Jeff

August 10, 2001

RE: "A Look Back"

Seven years, almost eight; it was early in 1994 that I saw the advertisement looking for Citizens to serve on the Luke AFB RAB. I answered the advertisement with a letter and I am glad now that I did.

I liked the concept of civilian input into the matter of planning the clean up effort.

From the start we enjoyed a spirit of coopertion with both the Command and the Environmental Staff. The attitude was, "we have this problem and let's get it identified and do what we need to do to put it behind us". Going about setting in place a structured system that would prevent these

problems in the future was our goal.

As sites were identified, we looked at the best and most cost effective ways of mitigating the situation at hand.

We always felt we were well informed and included in the decision making

regarding what could and had to be done. Luke stands out as an example of how to do it right. This was very obvious when I attended a conference in San Francisco and found that of all the attendees from across the nation only Luke and one other East Coast Base

enjoyed a relationship with the Command and Staff that we did. In most cases the RAB's had an adversarial relationship that made it hard to get anything done.

In our case at Luke the Base Commanders and the Environmental Staff established a situation of openness and trust that made us a team. Now that the Base is cleaned up some sites require long term monitoring,

so we feel it important that together, we, from time to time sit down and eview just how things are going.

It has been a great joint effort that worked well and the Base and all the people involved should be commended.

Martin Jeffries
### Luke AFB Five-Year Review Interview Questions Belle Matthews (Luke AFB)

1. What is your understanding of the overall project at the Site? I am thoroughly familiar with all the sites.

2. Please describe your involvement or participation at the Site. I am the IRP manager for the last three years

3. What is your general impression of the project (Site)? The projects were thoroughly investigated and appropriate remediation performed.

4. What effects have the site operation had on you (or the surrounding community)? The surrounding community participated in the restoration project. They partnered in the research investigation and choices of remediation.

5. Are you aware of any community concerns regarding the Site or its operation and administration? Give any details.

The community has addressed concerns for future continuity of the environmental program. The Restoration Advisory Board rewrote the charter and developed an ongoing community outreach called the Citizens' Advisory Board (CAB) which will continue to meet and be involved in the reviewing the environmental program.

6. Are you aware of any unusual events, incidents or activities at the Site (vandalism, trespassing, unauthorized activities, or emergency response)? Give details.

No.

7. Do you feel well informed about the Site's activities and progress?

Yes

8. Do you have any other comments, suggestions or recommendations regarding the Site's management or operation?

Development of a newsletter by the CAB to keep the surrounding community informed about environmental operations.

9. Have there been routine communications or activities (site visits, inspections, reports, sampling, etc) conducted by your office? Give pertinent details.

10. Have any problems been noticed which may require changes in the remedial action taken for this Site or in any of the decision documents?

No

- 11. Please describe current O&M procedures.
- 12. Have there been any unexpected O&M difficulties or costs at the Site? Give details.
  - 13. Do you feel there are opportunities for optimizing the O&M or sampling efforts at the Site?

After the Five Year Review Program is completed, a comprehensive list of sampling and O&M should be developed. The sampling program should be orchestrated to reflect optimum savings. By mid-cycle, the sampling plan should be reviewed; and requests for reducing the numbers of sampling episodes should be explored with AzDEQ and US EPA. Funded through HQ AETC should be secured.

## Luke AFB Five-Year Review Interview Questions Nancy Lou Minkler (ADEQ)

#### 1. What is your understanding of the overall project at the Site?

Luke AFB was placed on the EPA's National Priorities List (NPL) in August 1987. This placement identified Luke AFB as a priority site for investigation and cleanup under the Comprehensive Environmental Response, Compensation, and Liability Act (CERCLA), as amended by Superfund Amendments and Reauthorization Act (SARA). The sites were separated into two Operable Units, OU-1 and OU-2. The Records of Decision (ROD) were signed by ADEQ for OU-2 on January 10, 1994, and for OU-1 on August 16, 1999.

Luke AFB has satisfied the criteria for deletion in that the remediation selected in the OU-2 and OU-1 RODs have been fully implemented.

2. <u>Please describe your involvement or participation at the Site.</u>

I am the ADEQ Project Manager, and have worked on this site since September 1997.

#### 3. What is your general impression of the project (Site)?

The Air Force has done/is doing a thorough job of the investigation, remediation and operation and maintenance of areas of concern at this Site.

#### 4. What effects have the site operation had on you (or the surrounding community)?

I have worked with the Air Force during the drafting and finalization of the OU-1 ROD, the implementation of selected remedies and the delisting process.

# 5. <u>Are you aware of any community concerns regarding the Site or its operation and administration?</u> Give any details.

To my knowledge, the Air Force has always been informative and accommodating to the Restoration Advisory Board (RAB), now the Community Advisory Board (CAB), which has been actively involved for several years. This "oversight" has conveyed a greater sense of ease within the general community. Community issues are expressed in the RAB or CAB meetings.

6. <u>Are you aware of any unusual events</u>, incidents or activities at the Site (vandalism, trespassing, unauthorized activities, or emergency response)? Give details.

Not to my knowledge.

7. Do you feel well informed about the Site's activities and progress?

Yes.

8. Do you have any other comments, suggestions or recommendations regarding the Site's management or operation?

No, as long as the Institutional Control Plan (ICP) is enforced. ADEQ will continue to oversee the O&M activities at Luke Air Force Base, and be involved in the 5-year reviews, as long as the base remains open.

9. <u>Have there been routine communications or activities (site visits, inspections, reports, sampling, etc) conducted by your office?</u> Give pertinent details.

The Air Force has been in regular contact with ADEQ. There have been regular site visits and inspections by our office. ADEQ split sample remediated soils during the lead shot removal activities at the shooting range on December 20, 1999. Occasionally, ADEQ will split sample during routine groundwater monitoring events, such as during the May 16, 2000 sampling event for monitor wells at SS-42.

10. <u>Have any problems been noticed which may require changes in the remedial action</u> <u>taken for this Site or in any of the decision documents?</u>

Not at this time.

11. <u>Please describe current O&M procedures.</u>

Long-term monitoring at SS-42 and at RW-02.. Base-wide groundwater monitoring is required during every 5-year review event. (Including this one)

12. <u>Have there been any unexpected O&M difficulties or costs at the Site?</u> Give details.

The only unexpected O&M difficulties or costs associated with the site at this time, to my knowledge, was the request of the EPA to install two additional monitoring wells at RW-02.

13. Do you feel there are opportunities for optimizing the O&M or sampling efforts at the Site?

Probably yes. After the Base-wide groundwater monitoring report is submitted, and after several rounds of monitoring, where required, there may be some modifications necessary. This would be determined later, probably during the next 5-year review.